

Промышленные аккумуляторы Marathon FT

Идеальное решение для стоек и шкафов 19'' и 23''



Технические преимущества

Запатентованный дизайн боковых стенок "Diamond Side-Wall", обеспечивающий высокую прочность и стабильность корпуса в широком диапазоне температур.

Обязательный контрольный разряд каждого блока в процессе производства. Абсорбированный электролит технология АGM

Единственный в мире контроль герметичности с использованием автоматического гелиевого детектора.

■ Универсальное применение: телекоммуникация и связь, агрегаты бесперебойного питания (UPS), производство и распределение электроэнергии, железная дорога.









пластины

емкость 35–200 Ач

исполнение 12 лет

необслужи-

бокому разряду согласно DIN 43 539 T5

MARATH@N™

Технические данные.

Тип	Серийный	Напря-	Емкость	Длина	Ширина	Высота	Bec,	Внутреннее	Тип	Применение
	номер	жение,	C ₁₀	(I),	(b/w),	(h),		сопро-	вывода	в конструк-
			1,8 В/эл.,					тивление,		тивах
			20°C,							
		В	Ач	MM	MM	MM	КГ	мОм		
M 12 V 35 FT	NAMF120035HM0MA	12	35	280	107	189	14,0	12,8	M-M6-90°	19"- 300 мм
M 12 V 50 FT	NAMF120050HM0MA	12	47	280	107	231	18,0	9,2	M-M6-90°	19"- 300 мм
M 12 V 60 FT	NAMF120060HM0MA	12	59	280	107	263	23,0	7,8	M-M6-90°	19"- 300 мм
M 12 V 90 FT	NAMF120090HM0FA	12	86	395	105	270	31,0	4,5	F-M6-90°	19"- 400 мм
M 12 V 105 FT	NAMF120105HM0FA	12	100	511	110	238	35,8	4,0	F-M6-90°	23"- 600 мм
M 12 V 125 FT	NAMF120125HM0FA	12	121	559	124	283	47,6	3,2	F-M6-90°	23"- 600 мм
M 12 V 155 FT	NAMF120155HM0FA	12	150	559	124	283	53,8	3,0	F-M6-90°	23"- 600 мм
M 06 V 200 FT	NAMT060200HM0FA	6	200	376	132	250	34,0	1,6	F-M6	23"- 600 мм

Типы выводов:









Данные для выбора аккумуляторов.* Разряд постоянным током, А.

	Конечное напряжение 1,80 В/эл — разряд, А										
Тип	Серийный номер	15 мин	30 мин	14	2ч	3ч	44	5ч	84	10 ч	20 ч
M 12 V 35 FT	NAMF120035HM0MA	70,3	43,3	26,3	14,3	10,0	7,8	6,4	4,2	3,5	1,9
M 12 V 50 FT	NAMF120050HM0MA	97,3	60,1	33,5	18,7	13,3	10,1	8,6	5,7	4,7	2,4
M 12 V 60 FT	NAMF120060HM0MA	106,7	67,4	39,6	22,6	16,3	12,9	10,8	7,2	5,9	3,1
M 12 V 90 FT	NAMF120090HM0FA	170,0	105,2	63,0	35,0	24,6	19,1	15,7	10,4	8,6	4,7
M 12 V 105 FT	NAMF120105HM0FA	179,5	109,5	68,3	38,8	27,8	22,0	18,3	12,3	10,0	5,3
M 12 V 125 FT	NAMF120125HM0FA	204,4	132,2	85,1	49,3	36,3	27,9	22,8	14,9	12,1	6,4
M 12 V 155 FT	NAMF120155HM0FA	254,1	167,3	99,8	59,6	42,3	33,1	27,4	18,4	15,0	7,9
M 06 V 200 FT	NAMT060200HM0FA*	297,5	210,7	132,7	75,1	53,8	42,5	35,4	24,1	20,0	10,5

Конечное напряжение 1,75 В/эл — разряд, А											
Тип	Серийный номер	15 мин	30 мин	14	2ч	3ч	4ч	5ч	8ч	10 ч	20 ч
M 12 V 35 FT	NAMF120035HM0MA	72,6	44,0	26,5	14,4	10,2	7,9	6,6	4,3	3,5	1,9
M 12 V 50 FT	NAMF120050HM0MA	99,3	61,0	34,3	19,1	13,5	10,6	8,8	5,8	4,7	2,5
M 12 V 60 FT	NAMF120060HM0MA	109,6	68,8	40,1	23,0	16,6	13,1	11,0	7,3	6,0	3,2
M 12 V 90 FT	NAMF120090HM0FA	178,0	108,0	64,0	35,5	24,9	19,4	15,9	10,6	8,7	4,7
M 12 V 105 FT	NAMF120105HM0FA	190,7	115,4	70,0	39,8	28,5	22,5	18,7	12,7	10,3	5,4
M 12 V 125 FT	NAMF120125HM0FA	233,6	141,7	88,1	50,5	37,2	28,6	23,4	15,2	12,4	6,5
M 12 V 155 FT	NAMF120155HM0FA	275,0	174,9	103,2	60,8	43,2	33,9	28,1	18,9	15,4	8,0
M 06 V 200 FT	NAMT060200HM0FA*	323,4	220,5	135,2	76,9	55,2	43,6	36,3	24,6	20,2	10,7

Конечное напряжение 1,65 В/эл — разряд, А											
Тип	Серийный номер	15 мин	30 мин	14	2ч	34	4ч	5ч	84	10 ч	20 ч
M 12 V 35 FT	NAMF120035HM0MA	75,5	45,0	27,0	14,6	10,3	8,1	6,8	4,4	3,6	1,9
M 12 V 50 FT	NAMF120050HM0MA	103,9	62,0	34,9	19,3	13,8	10,9	9,1	5,9	4,8	2,5
M 12 V 60 FT	NAMF120060HM0MA	111,4	70,5	40,9	23,4	16,9	13,4	11,2	7,5	6,1	3,2
M 12 V 90 FT	NAMF120090HM0FA	193,4	112,3	65,1	36,0	25,2	19,6	16,1	10,7	8,8	4,8
M 12 V 105 FT	NAMF120105HM0FA	200,4	120,7	72,7	41,0	29,2	23,0	19,1	12,9	10,5	5,4
M 12 V 125 FT	NAMF120125HM0FA	250,8	152,8	91,3	52,0	37,7	29,3	24,1	15,7	12,8	6,8
M 12 V 155 FT	NAMF120155HM0FA	312,0	182,6	106,9	62,5	44,4	34,7	28,7	19,2	15,6	8,2
M 06 V 200 FT	NAMT060200HM0FA*	357,1	231,3	139,3	78,7	56,3	44,4	37,0	25,1	20,6	10,8

*Дополнительные разрядные характеристики Вы можете запросить в офисах нашей компании.

Deutsche EXIDE GmbH Im Thiergarten 63654 Büdingen Tel.: +49 (0) 60 42/810 Fax.: +49 (0) 60 42/81 398 www.exide.de

Москва: тел.: 095/247 9898, т/ф: 095/247 9888 Санкт-Петербург: т/ф: 812/273 0121; 327 2065 Ростов-на-Дону: т/ф: 863/295 5561; 292 3144 Екатеринбург: т/ф: 343/365 9197; 371 2351 Н. Новгород: т/ф: 8312/28 30 20; 16 06 06 Новосибирск: т/ф: 3832/46 50 59; 12 58 16 Владивосток: т/ф: 4232/41 06 16





Госсан



e-mail: info@exide-technologies.ru www.exide-technologies.ru



Husky IVO 350R

Freedom to scale up with existing IT infrastructure

Throughput Up to 360 Mbit/s Local storage Up to 32 TB

The Husky IVO 350 Rack is the ideal "first rack server" product for those who aim to start small in their video installation — yet want the flexibility to grow within their existing IT infrastructure.

Easy & flexible setup

With an easy setup and flexibility in your XProtect[®] VMS, this unit is a great fit to get you started in your video installation.

- Seamless setup with the Husky Assistant in a few simple steps
- Windows and XProtect VMS optimization according to your needs
- Choose any version of XProtect VMS for your installation
- Extended camera compatibility

 connect more than 10,000
 devices to your XProtect VMS

Peace of mind – now and long into the future

Your video installation should not only be easy and flexible to set up, but also to manage and run – without extra costs or complexity. With the Husky IVO 350R, you get our standard five-year warranty, with seamless single-point support through Milestone and all the benefits of using XProtect as your VMS.

- Single point of contact for the support of both the VMS and the appliances
- Possibility for onsite support interventions for hardware repairs and replacement

Key features

- Rack-mount unit ideal for those with an existing IT infrastructure
- Up to 32TB raw video storage
- Software RAID for increased uptime

Recommended usage

- Up to 50 channels (Based on the following video stream: 1080p, 30 FPS, 5-8 Mbit/s/channel, with video motion detection recording)
- Local live view/playback possible for up to 4 channels through XProtect Smart Client, without affecting the overall performance



Hardware information

Form Factor Rack Mounted 1U

CPU Intel® Core™ i3-9100

GPU Integrated Intel® UHD Graphics 630 (Intel® Quick Sync Video)

RAM 16GB DDR4

Operating system (OS) Windows 10 IoT Enterprise SAC

VMS/OS Storage 1 x 256GB (M.2 PCIe NVMe)

Number of hard-drives 2 x 3.5" (external accessible)

Raw storage (gross) 4/8/16/24/32 TB (Dell Enterprise grade drives)

RAID Support Intel® Rapid Storage Controller – RAID 0,1 (Software RAID)

Display ports 2 x USB 3.1 Type C / iGFX Display Port USB-C to VGA Video Adapter included

Network interface 2 x 1 GbE (R|45)

Available PCIe slots One PCI Express® x16 Gen 3 double width slot full height One PCI Express® x8 Gen 3 slot full height One PCI Express® x4 Gen 3 slot full height

Archiving to external NAS Yes, through LAN port

Credential and Key encription Trusted Platform Module (TPM) 2.0

Redundant Power Supply No

Remote management No

General information

Hardware warranty

5 years full system warranty (incl. HDD) Onsite support next business day¹

Dimensions

42.8 x 482.0 x 577.6 (mm) (without bezel) 1.68" x 18.98" x 22.74" (without bezel)

Weight 8.74 kg (min) 19.27 lb (min)

Power input 100~240V, 50/60 Hz

Max. power consumption 550W

BTU Rating 1,875 BTU/h (max. theoretical)

Protection

OVP (over voltage), OCP (over current), OTP (over temp), SCP (short circuit)



Environmental

Operating temperature 10° - 35° C / 50° - 95° F

Storage temperature -40° - 65° C / -40° - 149° F

Humidity 10 – 85% relative humidity (non-condensing)

Regulatory

Safety & emission standards CE (class A), UKCA, FCC, RCM, UL, Mexico (NOM), VCCI

Trade compliance NDAA Compliant

Ordering information

HE-350R-4TB	Husky IVO 350 Rack mount, Win10, 4TB	2 bays x 2TB
HE-350R-8TB	Husky IVO 350 Rack mount, Win10, 8TB	2 bays x 4TB
HE-350R-16TB	Husky IVO 350 Rack mount, Win10, 16TB	2 bays x 8TB
HE-350R-24TB	Husky IVO 350 Rack mount, Win10, 24TB	2 bays x 12TB
HE-350R-32TB	Husky IVO 350 Rack mount, Win10, 32TB	2 bays x 16TB
HA-RR-3930	Slide Rack Rail kit for Husky IVO 350R	n/a

XProtect VMS software

Milestone Husky appliances are preloaded with Milestone XProtect VMS, which means that they can be used with any XProtect VMS variants. The required VMS licenses are purchased separately. The units can also be used with XProtect Essential+ with up to 8 cameras for free.

¹ Onsite intervention dispatched following Milestone's support processes – hardware Service Tag information accuracy required. Subject to regional availability, dependent on Dell Technologies support network.





ARUBA 2530 24 POE+ SWITCH (J9779A)

Fixed Port L2 Managed Ethernet Switches



WHAT'S NEW

- Cost-effective, reliable and secure Aruba Layer 2 switch series for a better mobile-first experience.
- ACLs, EEE, traffic prioritization.
- 8-, 24-, and 48-port Gigabit or Fast Ethernet models.
- PoE+ models for voice, video and wireless.
- Supports Aruba ClearPass Policy Manager, AirWave and Central.
- Simple deployment with zero touch provisioning (ZTP).

OVERVIEW

The Aruba 2530 Switch Series provides security, reliability, and ease of use for enterprise edge, branch office, and SMB deployments. Fully managed switches deliver Layer 2 capabilities with optional PoE+, enhanced access security, traffic prioritization, sFlow, IPv6 host support, and power savings with Energy Efficient Ethernet. The Aruba 2530 Switch Series is easy to use and deploy, and delivers consistent wired/wireless user experience with unified security and management tools such as Aruba ClearPass Policy Manager, Aruba AirWave and cloud-based Aruba Central.

FEATURES

Cost-Effective, Reliable and Secure Access Layer Switches

The Aruba 2530 Switch Series provides security, reliability, and ease of use for enterprise edge, branch office, and SMB deployments.

Fully managed switches deliver full Layer 2 capabilities with optional PoE+, enhanced access security, traffic prioritization, sFlow, and IPv6 host support.

Right size deployment with choice of 8-, 24-, and 48-port models available with Gigabit or Fast Ethernet ports, and optional PoE+.

Power savings with fanless models, Energy Efficient Ethernet (IEEE 802.3az), and ability to disable LEDs and enable port low power mode.

Delivers consistent wired/wireless user experience with unified security and management tools such as ClearPass Policy Manager, AirWave and cloud-based Central. Provides optimal configuration automatically when connected to Aruba APs for PoE priority, VLAN configuration, and rogue AP containment.

Security and Quality of Service (QoS)

The Aruba 2530 Switch Series supports flexible authentication methods including Local MAC, 802.1X, and MAC and Web for greater policydriven application security.

Advanced denial of service (DOS) protection, such as DHCP Protection, Dynamic ARP protection, and Dynamic IP lockdown, enhance security. Flexible traffic controls include ACLs and QoS.

Traffic prioritization with IEEE 802.1p allows real-time traffic classification with support for eight priority levels mapped to either two or four queues using weighted deficit round robin (WDRR) or strict priority (SP).

Defend your IPv6 network with DHCPv6 Protection.

Simple Deployment and Management

The Aruba 2530 Switch Series supports a choice of management interfaces with Web GUI, command-line interface (CLI), and SNMP with either console or micro USB ports.

Quiet operation with fanless and variable speed fan models.

Flexible deployment with wall, table, and rack mounting options.

Zero touch provisioning (ZTP) provides quick and painless deployment at locations with few or no technical resources.

Single View of the Network

The Aruba 2530 Switch Series supports Aruba ClearPass Policy Manager for unified and consistent policy between wired and wireless users and simplifies implementation and management of guest login, user onboarding, network access, security, QoS, and other network policies.

Supports Aruba AirWave network management software to provide

common platform for zero touch provisioning management, and monitoring for wired and wireless network devices.

RMON and sFlow provide advanced monitoring and reporting capabilities for statistics, history, alarms, and events.

Aruba Central cloud-based management platform offers simple, secure, and cost effective way to manage switches.

Supports both cloud-based Central and on-premise AirWave with the same hardware ensuring change of management platform without ripping and replacing switching infrastructure.

Technical specifications

Aruba 2530 24 PoE+ Switch

Product Number	J9779A
Differentiator	24-port fully managed Fast Ethernet PoE+ switch with Gigabit uplinks. This model has 24x 10/100 PoE+ ports with a power budget of 195W. Additionally the switch has two 10/100/1000 ports and two dedicated Gigabit SFP ports.
Ports	(24) RJ-45 autosensing 10/100 PoE+ ports (2) autosensing 10/100/1000 ports (2) fixed Gigabit Ethernet SFP ports
Memory and processor	ARM9E @ 800 MHz 128 MB flash Packet buffer size: 1.5 MB dynamically allocated 256 MB DDR3 DIMM
Latency	100 Mb Latency: < 1.7 μs 1000 Mb Latency: < 1.1 μs
Throughput	up to 9.5 Mpps
Switching capacity	12.8 Gbps
PoE capability	195 W
Stacking capabilities	Virtual 16 switches
Management features	Aruba Central and Aruba AirWave Network Management IMC—Intelligent Management Center command-line interface Web browser configuration menu out-of-band management (serial RS-232C or Micro USB) IEEE 802.3 Ethernet MIB Repeater MIB Ethernet Interface MIB AirWave Network Management
Input voltage	100 - 127 / 200 - 240 VAC, rated
Operating humidity range	15 to 95% @ 104°F (40°C) (noncondensing)
Power Consumption	237 W (maximum)
Heat dissipation	99 BTU/hr (104.45 kJ/hr)
Product Dimensions (metric)	44.2 x 32.26 x 4.45 cm
Weight	3.81 kg

For additional technical information, available models and options, please reference the QuickSpecs

ARUBA GLOBAL SERVICES

Aruba Global Services simplifies and accelerates the network technology lifecycle, enabling your network to scale with better predictability and cost-effectiveness. Whether you operate your own network and need to improve your IT efficiencies or you want to offload some of the burden, we have the services you need to reach your goals.

Learn more about what Aruba Global Services has to offer at: arubanetworks.com/services/



Support Services from Aruba

Support services reduce complexity and increase your team's productivity, ensuring you keep pace with technology advances and software releases, and obtain break-fix support required to keep your network running. Access to premium services means you have the right help at the right time.

Professional Services from Aruba

With deep intellectual capital and purpose-built tools, our team delivers a range of standard and custom professional services designed to accelerate your value from Aruba technology.

Proactive Engineering Services include:

Customer Experience Management

Intelligent Operations

Network optimization

QuickStart Services include:

- Planning, audit and assessment
- Architecture review and design
- Deployment, migration, and knowledge transfer
- Our **Education Services** allow your team to come up to speed quickly.

Aruba Network as-a-Service (NaaS)

Our NaaS solution, Aruba Managed Connectivity Services, part of the HPE GreenLake services family, simplifies network operations, accelerates equipment handling, and increases the value of your Aruba network. If you need expert guidance and automation-based operations for your team, please explore the NaaS approach from Aruba <u>here</u>.

Make the right purchase decision. Contact our presales specialists.

Find a partner



Buy now
Share now
Get updates

a Hewlett Packard Enterprise company Parts and components that have reached their maximum supported lifetime and/or the maximum usage limitations as set forth in the manufacturer's operating manual, product quick-specs, or the technical product data sheet will not be provided, repaired, or replaced as part of these services.

Image may differ from the actual product PSN5384999IEEN, August, 2022.

[©] Copyright 2022 Hewlett Packard Enterprise Development LP. The information contained herein is subject to change without notice. The only warranties for Hewlett Packard Enterprise products and services are set forth in the express warranty statements accompanying such products and services. Nothing herein should be construed as constituting an additional warranty. Hewlett Packard Enterprise shall not be liable for technical or editorial errors or omissions contained herein.

Parts and Materials: HPE will provide HPE-supported replacement parts and materials required to maintain the covered hardware.



ARUBA 2530 48 POE+ SWITCH (J9778A)

Fixed Port L2 Managed Ethernet Switches



WHAT'S NEW

- Cost-effective, reliable and secure Aruba Layer 2 switch series for a better mobile-first experience.
- ACLs, EEE, traffic prioritization.
- 8-, 24-, and 48-port Gigabit or Fast Ethernet models.
- PoE+ models for voice, video and wireless.

OVERVIEW

The Aruba 2530 Switch Series provides security, reliability, and ease of use for enterprise edge, branch office, and SMB deployments. Fully managed switches deliver Layer 2 capabilities with optional PoE+, enhanced access security, traffic prioritization, sFlow, IPv6 host support, and power savings with Energy Efficient Ethernet. The Aruba 2530 Switch Series is easy to use and deploy, and delivers consistent wired/wireless user

- Supports Aruba ClearPass Policy Manager, AirWave and Central.
- Simple deployment with zero touch provisioning (ZTP).

experience with unified security and management tools such as Aruba ClearPass Policy Manager, Aruba AirWave and cloud-based Aruba Central.

FEATURES

Cost-Effective, Reliable and Secure Access Layer Switches

The Aruba 2530 Switch Series provides security, reliability, and ease of use for enterprise edge, branch office, and SMB deployments.

Fully managed switches deliver full Layer 2 capabilities with optional PoE+, enhanced access security, traffic prioritization, sFlow, and IPv6 host support.

Right size deployment with choice of 8-, 24-, and 48-port models available with Gigabit or Fast Ethernet ports, and optional PoE+.

Power savings with fanless models, Energy Efficient Ethernet (IEEE 802.3az), and ability to disable LEDs and enable port low power mode.

Delivers consistent wired/wireless user experience with unified security and management tools such as ClearPass Policy Manager, AirWave and cloud-based Central. Provides optimal configuration automatically when connected to Aruba APs for PoE priority, VLAN configuration, and rogue AP containment.

Security and Quality of Service (QoS)

The Aruba 2530 Switch Series supports flexible authentication methods including Local MAC, 802.1X, and MAC and Web for greater policy-driven application security.

Advanced denial of service (DOS) protection, such as DHCP Protection, Dynamic ARP protection, and Dynamic IP lockdown, enhance security. Flexible traffic controls include ACLs and QoS.

Traffic prioritization with IEEE 802.1p allows real-time traffic classification with support for eight priority levels mapped to either two or four queues using weighted deficit round robin (WDRR) or strict priority (SP).

Defend your IPv6 network with DHCPv6 Protection.

Simple Deployment and Management

The Aruba 2530 Switch Series supports a choice of management interfaces with Web GUI, command-line interface (CLI), and SNMP with either console or micro USB ports.

Quiet operation with fanless and variable speed fan models.

Flexible deployment with wall, table, and rack mounting options.

Zero touch provisioning (ZTP) provides quick and painless deployment at locations with few or no technical resources.

Single View of the Network

The Aruba 2530 Switch Series supports Aruba ClearPass Policy Manager for unified and consistent policy between wired and wireless users and simplifies implementation and management of guest login, user onboarding, network access, security, QoS, and other network policies.

Supports Aruba AirWave network management software to provide common platform for zero touch provisioning management, and monitoring for wired and wireless network devices.

RMON and sFlow provide advanced monitoring and reporting capabilities for statistics, history, alarms, and events.

Aruba Central cloud-based management platform offers simple, secure, and cost effective way to manage switches.

Supports both cloud-based Central and on-premise AirWave with the same hardware ensuring change of management platform without ripping and replacing switching infrastructure.



Page 4

Technical specifications

Aruba 2530 48 PoE+ Switch

Product Number	J9778A
Differentiator	High density fully managed 48 port Fast Ethernet PoE+ access switch with Gigabit uplinks. This model has 48x 10/100 PoE+ ports with a power budget of 382W. Additionally the switch has two 10/100/1000 ports and two dedicated Gigabit SFP ports.
Ports	(48) RJ-45 autosensing 10/100 PoE+ ports (2) autosensing 10/100/1000 ports (2) fixed Gigabit Ethernet SFP ports
Memory and processor	ARM9E @ 800 MHz 128 MB flash Packet buffer size: 3 MB dynamically allocated 256 MB DDR3 DIMM
Latency	100 Mb Latency: < 6.6 μs 1000 Mb Latency: < 2.2 μs
Throughput	up to 13 Mpps
Switching capacity	17.6 Gbps
PoE capability	382 W
Stacking capabilities	Virtual 16 switches
Management features	Aruba Central and Aruba AirWave Network Management IMC—Intelligent Management Center command-line interface Web browser configuration menu out-of-band management (serial RS-232C or Micro USB) IEEE 802.3 Ethernet MIB Repeater MIB Ethernet Interface MIB AirWave Network Management
Input voltage	100 - 127 / 200 - 240 VAC, rated
Operating humidity range	15 to 95% @ 104°F (40°C) (noncondensing)
Power Consumption	441 W (maximum)
Heat dissipation	170 BTU/hr (179.35 kJ/hr)
Product Dimensions (metric)	44.2 x 32.26 x 4.45 cm
Weight	4.58 kg

For additional technical information, available models and options, please reference the QuickSpecs

ARUBA GLOBAL SERVICES

Aruba Global Services simplifies and accelerates the network technology lifecycle, enabling your network to scale with better predictability and costeffectiveness. Whether you operate your own network and need to improve your IT efficiencies or you want to offload some of the burden, we have the services you need to reach your goals.

Learn more about what Aruba Global Services has to offer at: arubanetworks.com/services/



Support Services from Aruba

Support services reduce complexity and increase your team's productivity, ensuring you keep pace with technology advances and software releases, and obtain break-fix support required to keep your network running. Access to premium services means you have the right help at the right time.

Professional Services from Aruba

With deep intellectual capital and purpose-built tools, our team delivers a range of standard and custom professional services designed to accelerate your value from Aruba technology.

QuickStart Services include:

- Planning, audit and assessment
- Proactive Engineering Services include:
- Intelligent Operations
- Customer Experience Management
- Architecture review and Network optimization
- design
 Deployment, migration, and knowledge transfer

Our **Education Services** allow your team to come up to speed quickly.

Aruba Network as-a-Service (NaaS)

Our NaaS solution, Aruba Managed Connectivity Services, part of the HPE GreenLake services family, simplifies network operations, accelerates equipment handling, and increases the value of your Aruba network. If you need expert guidance and automation-based operations for your team, please explore the NaaS approach from Aruba here.





© Copyright 2022 Hewlett Packard Enterprise Development LP. The information contained herein is subject to change without notice. The only warranties for Hewlett Packard Enterprise products and services are set forth in the express warranty statements accompanying such products and services. Nothing herein should be construed as constituting an additional warranty. Hewlett Packard Enterprise shall not be liable for technical or editorial errors or omissions contained herein.

Parts and Materials: HPE will provide HPE-supported replacement parts and materials required to maintain the covered hardware.

Parts and components that have reached their maximum supported lifetime and/or the maximum usage limitations as set forth in the manufacturer's operating manual, product quick-specs, or the technical product data sheet will not be provided, repaired, or replaced as part of these services.

Image may differ from the actual product PSN5384996USEN, August, 2022.

Make the right purchase decision. Contact our presales specialists.

Find a partner



Mitel MiVoice 6920 IP Phone

58015114 REV03

RELEASE 1.2 USER GUIDE FOR MIVOICE BUSINESS

NOTICE

The information contained in this document is believed to be accurate in all respects but is not warranted by Mitel Networks[™] Corporation (MITEL[®]). The information is subject to change without notice and should not be construed in any way as a commitment by Mitel or any of its affiliates or subsidiaries. Mitel and its affiliates and subsidiaries assume no responsibility for any errors or omissions in this document. Revisions of this document or new editions of it may be issued to incorporate such changes.

No part of this document can be reproduced or transmitted in any form or by any means - electronic or mechanical - for any purpose without written permission from Mitel Networks Corporation.

Trademarks

The trademarks, service marks, logos and graphics (collectively "Trademarks") appearing on Mitel's Internet sites or in its publications are registered and unregistered trademarks of Mitel Networks Corporation (MNC) or its subsidiaries (collectively "Mitel") or others. Use of the Trademarks is prohibited without the express consent from Mitel. Please contact our legal department at legal@mitel.com for additional information. For a list of the worldwide Mitel Networks Corporation registered trademarks, please refer to the website: http://www.mitel.com/trademarks.

Mitel's Power Over Ethernet (PoE) Powered Device (PD) products are covered by one or more of the U.S. patents (and any foreign patent counterparts thereto) identified at Mitel's website: www.mitel.com/patents.

For more information on the PD patents that are licensed, please refer www.cmspatents.com.

Mitel MiVoice 6920 IP Phone User Guide

58015114 - June 2017

 ®,™ Trademark of Mitel Networks Corporation
 © Copyright 2017, Mitel Networks Corporation All rights reserved

Software License Agreement

Mitel, hereinafter known as "Seller", grants to Customer a personal, worldwide, non-transferable, non-sublicenseable and non-exclusive, restricted use license to use Software in object form solely with the Equipment for which the Software was intended. This Product may integrate programs, licensed to Mitel by third party Suppliers, for distribution under the terms of this agreement. These programs are confidential and proprietary, and are protected as such by copyright law as unpublished works and by international treaties to the fullest extent under the applicable law of the jurisdiction of the Customer. In addition, these confidential and proprietary programs are works conforming to the requirements of Section 401 of title 17 of the United States Code. Customer shall not disclose to any third party such confidential and proprietary programs and information and shall not export licensed Software to any country except in accordance with United States Export laws and restrictions.

Customer agrees to not reverse engineer, decompile, disassemble or display Software furnished in object code form. Customer shall not modify, copy, reproduce, distribute, transcribe, translate or reduce to electronic medium or machine readable form or language, derive source code without the express written consent of the Seller and its Suppliers, or disseminate or otherwise disclose the Software to third parties. All Software furnished hereunder (whether or not part of firmware), including all copies thereof, are and shall remain the property of Seller and its Suppliers and are subject to the terms and conditions of this agreement. All rights reserved.

Customer's use of this software shall be deemed to reflect Customer's agreement to abide by the terms and conditions contained herein. Removal or modification of trademarks, copyright notices, logos, etc., or the use of Software on any Equipment other than that for which it is intended, or any other material breach of this Agreement, shall automatically terminate this license. If this Agreement is terminated for breach, Customer shall immediately discontinue use and destroy or return to Seller all licensed software and other confidential or proprietary information of Seller. In no event shall Seller or its suppliers or licensors be liable for any damages whatsoever (including without limitation, damages for loss of business profits, business interruption, loss of business information, other pecuniary loss, or consequential damages) arising out of the use of or inability to use the software, even if Seller has been advised of the possibility of such damages.

Software License Agreement
Welcome
About this guide
Supporting documentation
Phone features
Installation and setup
Tips for your comfort and safety
Don't cradle the handset
Protect your hearing
Adjust the viewing angle
Phone keys
Key panel
Key description
Dialpad keys
Getting started
UI overview
Home screen
Call screen
UI navigation
Customizing your phone
Using the Settings menu
Status
Language
Call Forward
Audio
Ring Iones
Audio Faut
Disnlav 28
Config Teleworker 29
Diagnostics 29
Audio Diagnostics
Ping
TCP DUMP
DHCP Trace
Restart

Speed Call	. 37
Other Features	. 37
Phone Lock	. 37
Call Forward Always	. 39
	. 41
Account codes	. 43
Making calls	. 45
Dialing a number	. 45
Initiating a callback	. 45
Redialing a number	. 46
Receiving calls	. 47
Answering an incoming call	. 47
Manually forwarding an incoming call	. 47
Handling calls	. 49
Placing a call on hold	. 49
Muting the microphone	. 49
Transferring a call	. 50
Creating a multi-party conference call	. 51
Ending a conference and transferring remaining parties	. 53
Ending calls	. 53
Managing calls.	. 54
Contacts	. 54
General functions	. 54
Personal folder features	. 58
Corporate folder features	. 63
Call History	. 67
Deleting entries from the Call History application	. 68
Copying entries to the Personal folder in the Contacts application	. 71
Missed calls indicator	. 72
Voicemail	. 73
Visual voicemail	. 74
Enable voicemail application	. 74
Log into your voicemail	. 75
	. 76
	. 70 77
	. <i></i> 78
Save a voicemail	. 79

Send voicemail to a number
Additional features
Accessory support
Mitel M695 Programmable Key Module (PKM)86
Mitel Wireless LAN Adapter
Analog, DHSG/EHS, and USB headsets
_imited warranty
Exclusions
Warranty repair services
After warranty service

Welcome

Part of the new-generation desktop family, the Mitel MiVoice 6920 IP phone demonstrates Mitel's commitment to ease of use and enhanced user value, while delivering innovative features and applications. A full-feature enterprise-class telephone, the Mitel MiVoice 6920 IP phone provides a large color backlit graphics display with 6 programmable self-labeling keys allowing up to 18 functions and four intuitive state sensitive softkeys.

This applications phone also features superior sound quality with wideband audio, an enhanced speakerphone that utilizes advanced audio processing to achieve richer and clearer handsfree conversations, and native analog, DHSG/EHS, and USB headset functionality.

Accommodating today's high speed networks through dual Gigabit Ethernet ports, the Mitel MiVoice 6920 IP phone also offers plenty of support for Mitel and third-party peripherals and has an environmentally efficient PoE class 2 rating. The Mitel MiVoice 6920 IP phone is one of the most advanced desktop IP phones available on the market today and is ideal for enterprise executives, managers, and employees.

The 6920 IP phone is compatible with the MiVoice Business phone system.

About this guide

This guide explains how to use the basic features of your new Mitel MiVoice 6920 IP phone. Not all features listed are available by default. Your System Administrator has the ability to customize many features on this phone. Contact your System Administrator to find out which features and services are available to you on your phone system.

Supporting documentation

To access phone and system-specific documentation:

- 1. Log in to <u>Mitel Connect</u>.
- 2. In the left-hand menu, click Mitel OnLine.
- 3. Click Product Documentation under the Support section.
- 4. Select PDF Guides and Online Help under the End User Documents drop-down menu.
- 5. In the left-hand menu, navigate to **Telephones**.

Phone features

Feature highlights include:

- 3.5" QVGA (320x240) color TFT LCD display with brightness controls
- Built-in-two-port, 10/100/1000 Gigabit Ethernet switch
- USB 2.0 port (100mA maximum)
- Wideband handset
- Enhanced wideband, full-duplex speakerphone for handsfree calls

- Extensive support for peripherals and modules: USB, DHSG/EHS, and wired analog headsets, Mitel M695 Programmable Key Module, and Mitel Wireless LAN Adapter
- AC power adapter (sold separately)
- Six programmable, multi-function, self-labeling keys allowing up to 18 functions and four intuitive state-sensitive softkeys

Installation and setup

Please refer to the *Mitel MiVoice 6920 IP Phone Installation Guide* included with your phone for basic installation and physical setup information.

Tips for your comfort and safety

Don't cradle the handset

Prolonged use of the handset can lead to neck, shoulder, or back discomfort, especially if you cradle the handset between your ear and shoulder. If you use your phone a lot, you may find it more comfortable to use a headset. See "Accessory support" on page 86 for more information on headsets.

Protect your hearing

Your Mitel MiVoice 6920 IP phone has a control for adjusting the volume of the handset, headset, and speakerphone. Because continuous exposure to loud sounds can contribute to hearing loss, keep the volume at a moderate level.

Adjust the viewing angle

The stand for your Mitel MiVoice 6920 IP phone supports two viewing angles (30° and 60°). Adjust the stand to suit your viewing preference. For instructions on how to adjust the stand, see the *Mitel MiVoice 6920 IP Phone Installation Guide*.

Phone keys

Key panel

The following sections describe the various Mitel MiVoice 6920 IP phone key functions and how they can help you make and manage your calls and caller information.



a	Goodbye Key	1	LCD Screen
d	Hold Key	0	Dialpad
e	Message Waiting Indicator (MWI)	q	Handset
f	Mute Key	r	Contacts Key
g	Speaker/Headset Key	S	Redial Key
h	Navigation Keys/Select Button	t	Call History Key
i	Volume Control	u	Settings Key
j	State-Sensitive Softkeys	W	Speaker
k	Programmable Keys	У	Voicemail Key

Key description

The following table describes the keys on the Mitel MiVoice 6920 IP phone:

Кеу	Description
	Contacts key - Displays a list of your contacts. For more information, see "Contacts" on page 54.
₹	Call History key - Displays a list of your missed, outgoing, and answered calls. For more information, see "Call History" on page 67.
	Voicemail key - Provides access to your voicemail service (if configured). For more information, see "Voicemail" on page 73.
()	Settings key - Provides services and settings that allow you to customize your phone.
	Volume controls - Adjusts the volume for the ringer, handset, headset, and speakerphone. Press the volume control keys while the phone is ringing to adjust the ringer volume.
	device being used (handset, headset, or speaker). Goodbye key - Ends an active call. The Goodbye key also exits an open list
	(such as Call History) and menus (such as the Settings menu) without saving changes.
5	twice redials the last dialed number displayed on the Home screen. For more information, see "Redialing a number" on page 46.
	Hold key - Places an active call on hold. To retrieve a held call, press the applicable Line key. For more information, see "Placing a call on hold" on page 49.
	Mute key - Mutes the microphone so that your caller cannot hear you (the LED beside the key turns on when the microphone is on mute). For more information, see "Muting the microphone" on page 49.
	Speaker/Headset key - Transfers the active call to the speaker or headset, allowing handsfree use of the phone. For more information, see "Audio Path" on page 26.
	Navigation keys and select button - Multi-directional navigation keys that allow you to navigate through the phone's User Interface (UI). Pressing the center Select button sets options as well as performs actions such as dialing out from the Contacts or Call History.
	On the Home screen, the left and right navigation keys can be used to access the additional pages of programmable softkeys. For more information, see "UI navigation is performed using the navigation keys located to the right of the softkeys." on page 15.
	State-sensitive softkeys - Four state-sensitive softkeys that allow you to perform different functions during specific states (i.e. when the phone is an idle, connected, incoming, outgoing, or busy state).



Description

Programmable keys - Six programmable, multi-function self-labeling keys that allow you to use up to 18 specific functions.

For more information see "Configuring and using programmable keys" on page 36.

Dialpad keys

The Mitel MiVoice 6920 IP phone has a dialpad with digits from 0 through 9, a * key, and a # key. Keys 2 through 9 contain the letters of the alphabet. These dialpad keys can be used to dial a phone number to make a call (see "Making calls" on page 45) and to press keys requested by an Interactive Voice Response (IVR) system. The basic inputs for the Mitel MiVoice 6920 IP phone dialpad includes the following:

Dialpad Key	Description
0	Dials 0
1	Dials 1
2 ABC	Dials 2
	When entering text, this key enters A with one press, B with two presses, and C with three presses.
3 DEF	Dials 3
	When entering text, this key enters D with one press, E with two presses, and F with three presses.
4 GHI	Dials 4
	When entering text, this key enters G with one press, H with two presses, and I with three presses.
5 JKL	Dials 5
	When entering text, this key enters ${\bf J}$ with one press, ${\bf K}$ with two presses, and ${\bf L}$ with three presses.
6 MNO	Dials 6
	When entering text, this key enters ${\bf M}$ with one press, ${\bf N}$ with two presses, and ${\bf O}$ with three presses.
7 PQRS	Dials 7
	When entering text, this key enters P with one press, Q with two presses, R with three presses, and S with four presses.
8 TUV	Dials 8
	When entering text, this key enters ${\bf T}$ with one press, ${\bf U}$ with two presses, and ${\bf V}$ with three presses.
9 WXYZ	Dials 9
	When entering text, this key enters ${\bf W}$ with one press, ${\bf X}$ with two presses, ${\bf Y}$ with three presses, and ${\bf Z}$ with four presses.

The * is called the star key. The # is called the pound key, number sign, or hash key, depending on one's nationality or personal preference. These can be used for special functions such as accessing advanced services. The star key and pound key functions are dependent on your call server's feature availability. Contact your System Administrator for more information about available functions using these keys.

When entering characters in the various **Settings** menus or in different applications (for example, when adding a contact to your **Contacts** application), advanced characters can be entered using the dialpad keys.

Кеу	Uppercase Characters	Lowercase Characters
0	0 +	0 +
1	1 . : ; =_ , - ' & () \$!	1 . : ; =_ , - ' & () \$!
2	ABC2	abc2
3	DEF3	d e f 3
4	GHI4	ghi4
5	JKL5	j k l 5
6	M NO 6	m n o 6
7	PQRS7	pqrs7
8	T U V 8	t u v 8
9	W X Y Z 9	w x y z 9
*	* <space></space>	* <space></space>
#	#/\@	#/\@

The following tables identifies the characters that you can use and the order in which they are accessible.

Getting started

The Mitel MiVoice 6920 IP phone must be set up and be configured by the System Administrator prior to its first use.

UI overview

Home screen

When the phone has successfully booted up and connected to the network, the phone displays the **Home** screen. The **Home** screen shows the date and time along with your number and last dialed number. It is the default screen displayed when the phone is in an idle state.



Note: Your System Administrator should have configured your number and any other administrative features required. If your number does not display on the screen, contact your System Administrator.

Phone and network status indicators

There are icons that display on the phone to indicate the status of the phone and network. These icons display on the upper right corner of the **Home** screen.



The following table identifies and describes the phone and network status indicators on the **Home** screen.

lcon	Description
윪	Network Connected - The Network Connected icon displays on the Home screen when the phone is connected to the MiVoice Business call server.
율	Attempting Network Connection - The Attempting Network Connection icon displays on the Home screen when the phone is attempting to connect to the MiVoice Business call server.
윩	Network Disconnected - The Network Disconnected icon displays on the Home screen when the phone loses connection to the MiVoice Business call server.
	If this icon is displayed and you are using an AC adapter for power, check that the Ethernet cable is tightly connected to the phone and to the network device port. The phone should automatically detect when it is reconnected and attempt to connect to the network. If this does not resolve the issue, check with your System or Network Administrator for assistance.
<u>K</u>	Call Forward Always - The Call Forward Always icon displays on the Home screen when the Call Forward feature is enabled using the "Always" condition (as opposed to the "Busy" and "No Answer" conditions).
	For more information on the Call Forward feature, see "Call Forward" on page 22.
<mark>.≪</mark> 4	Missed Calls - The Missed Calls icon displays on the Home screen only when incoming calls have not been received. The number in the red badge indicates how many calls were missed. To access a list of your missed calls, press the <i>M</i> (Call History) key.
	For more information on missed calls, see "Missed calls indicator" on page 72.
9	Do Not Disturb - The Do Not Disturb (DND) icon displays on the Home screen when the DND feature is enabled. When DND is enabled, callers calling into the phone may hear a DND signal and a "Do Not Disturb" message is displayed on their phones, depending on the setting in the System Administrator Tool.
	For more information on the DND feature, see "Do Not Disturb" on page 41.
¢	Headset Connected - The Headset Connected icon displays on the Home screen when a headset (USB, DHSG/EHS, or analog wired) connection has been detected.
	For more information on headset functionality, see "Analog, DHSG/EHS, and USB headsets" on page 88.
ది	Hotdesk User - The Hotdesk User icon displays on the Home screen when you log in using your hotdesk account.
ഹ	Voicemail - The Voicemail icon displays on the Home screen only if you have new messages waiting. To access your voicemail messages, press the <i>(Voicemail)</i> key.
	For more information on voicemail functionality, see "Voicemail" on page 73.

Programmable keys, state-sensitive softkeys, and line keys

Six programmable keys (including the **Prime Line** key) and four state-sensitive softkeys are always displayed on the **Home** screen. The programmable keys allow you to configure up to 18 different features (such as speed calls, phone lock, DND, and so on).

Note: If enabled by your System Administrator, certain features can be programmed on the programmable keys by pressing and holding the respective programmable key. Others can only be programmed by your System Administrator through the MiVoice Business call server. See "Configuring and using programmable keys" on page 36 for information on how to program the programmable keys using the phone. Contact your System Administrator for details on what features are available to be programmed through the MiVoice Business call server.

The state-sensitive softkeys change depending on the state of the phone and are used to perform different functions when the phone is an idle, connected, outgoing, or busy state.

The **Prime Line** key is a visual indication of the status of your prime line. When your phone is in an idle state, your **(Prime Line**) icon is solid blue. Your System Administrator may also have configured additional line appearances on your phone. Line appearance keys are solid green when your phone is in an idle state.



If you have six or more programmable keys configured, programmable page indicators are displayed above the softkeys. These dots indicate the number of programmable key pages and its relative position. Press the left and right navigation keys to switch the programmable key pages.

Call screen

During an incoming or outgoing call and when on an active call, all the essential information regarding the call is displayed on screen, including the caller's avatar, name, number, and call duration timer. The context-sensitive softkeys also change allowing you access to more call handling features (such as **Transfer** and **Add User** during an active call).



Additionally, if the call is on the **Prime Line** key, the **Prime Line** key changes to reflect the status of the line as per the following table.

lcon	Description
L	(Blinking) Indicates the prime line is ringing due to an incoming call.
	Indicates the prime line is busy.
0	(Blinking) Indicates the call on the prime line is on hold.

If the call is on a different line, the applicable line appearance key change to reflect the status of the line as per the following table.

lcon	Description
	(Blinking) Indicates the applicable line is ringing due to an incoming call.
	Indicates the applicable line is busy.
m	(Blinking) Indicates the call on the applicable line is on hold.

UI navigation

UI navigation is performed using the navigation keys located to the right of the softkeys.



There are four directional keys (up, down, left, and right) on the navigation cluster that can be used to navigate through the various applications and settings menus. The center key acts as a **Select** button that is used to set options and perform actions such as dialing out from the **Contacts** or **Call History** applications.

Customizing your phone

You can customize your phone and view your phone status by pressing the (Settings) key, opening the Settings menu. This menu provides a single location to access all your phone settings. The following phone settings can be configured through the Settings menu:

lcon	Option
*	Status
\bigcirc	Language
00	Voicemail
<u>K</u>	Call Forward
Ø	Config Teleworker
-1-	Diagnostics
□	Audio Diagnostics
Ū	• Ping
Ū.	TCP DUMP
Ø	DHCP Trace
り	Audio
IJ	Ring Tones
IJ	Audio Path
Ū	Display
	Restart

Notes:

- **1.** The **Advanced** options are Administrator-level functions only, and are not accessible by the user. These options should only be set up and changed by your System Administrator.
- 2. The Log Issue softkey is used to generate a log file that can help Administrators debug issues. You should only press this softkey when requested by your System Administrator.

Using the Settings menu

1. Press the (Settings) key on the phone to enter the Settings menu.

Settings					
مە >	Kr S	🔸 tatus	ø	-h-)	>
Select	Advanced	l Log	Issue	Close	

- **2.** Use the left and right navigation keys to highlight an option and the up and down navigation keys to scroll through the sub options.
- 3. Press the Select button or Select softkey to select an option.
- 4. Use the navigation keys, dialpad keys, and softkeys that display for each setting to make any applicable changes.
- 5. Press the **Save** softkey to save your changes.
- 6. Press the **Coodbye**) key or the **Close** softkey at any time to exit without saving changes.
- Note: For more information about Advanced options, contact your System Administrator.

Status

The Status settings in the Settings menu allows you to view your phone's:

Network General Info:

General information regarding your phone's network settings such as IP address, subnet mask, gateway, MAC address, call server IP addresses, TFTP server IP address and port, IPA IP address, and DNS values.

- Network Persistent Data: The IP addresses for call servers 1 to 4.
- Network DHCP Info:

Information regarding your phone's Dynamic Host Configuration Protocol (DHCP) status and T1/T2 renewal times.

Network Status QoS:

Information regarding Quality of Service (QoS) features such as Virtual LAN (VLAN), Layer 2 Priority (L2P), and Differentiated Services Code Point (DSCP).

Port Info:

Speed and duplex values for both the LAN and PC ports.

Phone Info:

General phone-related information such as model number, hardware and call server revisions, main firmware version, and boot version.

Note: Press the **Copyright** softkey to view copyright information.

To view the Status menu:

- **1.** Press the (Settings) key on the phone to enter the Settings menu.
- 2. Press the Select button or Select softkey to enter the Status menu.
- 3. Use the up and down navigation keys to scroll the through the following status items:
 - Network General Info

Status	
Net. General I	IP Address
Net. Persist. D Net. DHCP Info	10.10.158.117
	Subnet Mask
	255.255.255.128
Net. QoS	Gateway
Port Info	10.10.158.1
Phone Info	
	Cancel

Network Persistent Data

Status	
Net. General I	Call Server1 IP Address
Net. Persist. D	
Net. DHCP Info	Call Server2 IP Address
Net. QoS	Call Server3 IP Address
Port Info	
Phone Info	
	Cancel

Network DHCP Info

Status	
Net. General I	DHCP Status
Net. Persist. D	DISABLED
Net. DHCP Info	T1 (Seconds)
Net. QoS	T2 (Seconds)
Port Info	0
Phone Info	
	Cancel

Network Static QoS

Status	
Net. General I	VLAN Id
Net. Persist. D	-1()
Net. DHCP Info	L2P Default
Net. QoS	L2P Voice
Port Info	6
Phone Info	
	Cancel

Port Info

•

Status	
Net. General I	LAN Port Speed
Net. Persist. D	100M LAN Port Duplex Full_Duplex PC Port Speed
Net DHCPInfo	
Net. QoS	
Port Info	No Link
Phone Info	
	Cancel

Phone Info

Status	
Net. General I	Model
Net. Persist. D	MINET_6920
Net DHCP Info	Hardware Rev.
	000
Net. QoS	Call Server Rev.
Port Info	14.0.1.22
Phone Info	
Copyright	Cancel

4. To view more information on the selected status item, press the right navigation key and press the down navigation key to scroll through the information.

Language

The Mitel MiVoice 6920 IP phone supports the following languages:

- English
- French
- German
- Spanish
- Spanish (Latin America)
- Portuguese
- Portuguese (Brazil)
- Dutch
- Italian
- Romanian
- Russian
- Swedish
- Polish

When you set the language on the phone, all on-screen text (in idle/call screens, setting menus, and applications) displays in the selected language.

- 1. All languages may not be available for selection as availability is dependent on the language files provided by the MiVoice Business call server. Contact your System Administrator for details on what languages are available in your specific environment.
- 2. Unofficially supported languages may be available for selection on your phone depending on the language files available on the MiVoice Business call server. If available, these languages can be used by selecting Local 1 or Local 2 in the Language menu.

To specify the language in which on screen text displays:

- 1. Press the (Settings) key on the phone to enter the Settings menu.
- 2. Navigate to the Language setting and press the Select button or Select softkey.

Language	
Language	English
	Français
	O Deutsch
	Español LA
	Português BR
Save	Cancel

- 3. Press the right navigation key to move to the language selection column.
- 4. Use the up and down navigation keys to choose the applicable screen language.
- 5. Press the **Save** softkey to save your changes.

Call Forward

Call Forward on the Mitel MiVoice 6920 IP phone allows incoming calls to be forwarded to another destination. You can enable your calls to be forwarded:

- When your phone is busy (individual settings for internal and external incoming calls).
- When you are not answering the phone (individual settings for internal and incoming calls).
- All the time.

You can enable different call forward modes independently. For example, you can set different phone numbers for Always, Busy Internal, Busy External, No Answer Internal, and No Answer External, and then turn them on/off individually.

- 1. A Call Forward Always programmable key can be created that allows you quickly enable Always mode without having to enter the Call Forward settings menu. For more information on the Call Forward Always programmable key, see "Call Forward Always" on page 39.
- **2.** If voicemail is enabled, call forwarding numbers may already be defined by your System Administrator.

To configure call forward settings:

- 1. Press the (Settings) key on the phone to enter the Settings menu.
- 2. Navigate to the Call Forward setting and press the Select button or Select softkey.

Call Forward					
Always					
Busy Internal	N				
Busy External		Number			
No Answer Internal		Number			
No Answer External		Number			
Save	Backspace	Copy to All?	Cancel		

- 3. Enter forwarding numbers using the dialpad keys for any of the following states:
 - Always:

Immediately forwards all incoming calls to the specified number.

Busy Internal:

Immediately forwards internal phone calls if your line is busy.

- Busy External: Immediately forwards external phone calls if your line is busy.
- **No Answer Internal:** Forwards internal phone calls after several rings if you don't answer.
- No Answer External: Forwards external phone calls after several rings if you don't answer.

- 1. Pressing the down navigation key moves the selection to the next input field. Pressing the right navigation key moves the selection to the corresponding checkbox.
- 2. Press the **Backspace** softkey to delete the last digit entered.
- 3. Pressing the **Copy to AII**? softkey copies the call forward number of the call forward mode in focus to every call forward mode. For example, if the cursor is on the **Busy Internal** field and the field has a phone number entered, pressing the **Copy to AII**? softkey assigns the same phone number to all the other modes as well.
- 4. Using the up and down navigation keys, move to the checkbox beside the respective call forward mode you want to enable and press the **Select** button to turn on the call forward mode.
- 5. Repeat steps 3 and 4 for all call forward modes you want to enable.

6. Press the Save softkey to save your changes.

Notes:

- 1. When **Call Forward Always** is enabled, it takes precedence over all other types of call forwarding.
- 2. If Call Forward Always is enabled, then the Call Forward Always icon is displayed on the Screen Saver.
- **3.** If both **Call Forward Always** and **DND** are enabled, then the **DND** icon is displayed on the Screen Saver.



- Call Forward Always Icon

Audio

The **Audio** setting on the Mitel MiVoice 6920 IP phone allows you to personalize ring tones, select the initial audio path used when pressing the () (**Speaker/Headset**) key, and enable or disable DHSG headset mode.

Ring Tones

There are a total of 20 (10 modern and 10 classic) distinct ring tones available for selection on the Mitel MiVoice 6920 IP phone. Two different ring tones can be configured for incoming calls from internal numbers and external numbers.

To configure ringtones for internal and external calls:

- 1. Press the (Settings) key on the phone to enter the Settings menu.
- 2. Navigate to the Audio > Ring Tones setting and press the Select button or Select softkey.

Ring Tones	
Internal Ring	Pronto
External Ring	Voyage
	Bloom
	Move
	Classic 1
Save	Cancel

- 3. With the Internal Ring setting highlighted press the right navigation key to move to Internal Ring selection column.
- **4.** Use the up and down navigation keys to scroll through the list and choose the applicable ring tone.
- 5. Press the left navigation key and then the down navigation key to highlight the **External Ring** setting.
- 6. With the External Ring setting highlighted press the right navigation key to move to External Ring selection column.
- **7.** Use the up and down navigation keys to scroll through the list and choose the applicable ring tone.
- 8. Press the Save softkey to save your changes.

Audio Path

_

The Mitel MiVoice 6920 IP phone allows you go "handsfree" when handling incoming and outgoing calls by supporting the use of a headset (USB, analog wired, or DHSG/EHS wireless) or through speakerphone mode. The **Audio Path** setting allows you to choose the initial audio path used when pressing the ()(Speaker/Headset) key. There are two audio path options that you can select:

Audio path	Description
Speaker	The default setting. When using the handset, pressing the ()/() (Speaker/Headset) key on the phone switches to the speakerphone. Pressing the ()/() (Speaker/Headset) key again switches to the headset (if connected). When using the speakerphone or headset, lift the handset to switch to the handset.
Headset	When using the handset, pressing the ()(Speaker/Headset) key on the phone switches to the headset. Pressing the ()(Speaker/Headset) key again switches to the speakerphone. When using the headset or speakerphone, lift the handset to switch to the handset.

To configure the audio path:

- 1. Press the (Settings) key to enter the Settings menu.
- 2. Navigate to the Audio > Audio Path setting and press the Select button or Select softkey.

Audio Path	Audio Path		
Headset			
Speaker			
Save	Cancel		

3. Use the up and down navigation keys to select the applicable audio path (Speaker or Headset).

Note: If you select the Headset option, ensure that only one headset is connected to the phone.

4. Press the Save softkey to save your changes.

Headset

The **Headset** setting allows you to enable DHSG mode when a DHSG/EHS headset is connected to the phone.

To enable/disable DHSG headset mode:

- 1. Press the (Settings) key to enter the Settings menu.
- 2. Navigate to the Audio > Headset setting and press the Select button or Select softkey.

Headset	
DHSG	Disabled
	Enabled
Save	Cancel

- 3. Press the right navigation key to move to enable/disable selection column.
- 4. Use the up and down navigation keys to choose the applicable setting.
- 5. Press the **Save** softkey to save your changes.

Display

The **Display** setting allows you to set the following on your phone:

Brightness Settings:

The **Brightness Level** setting on the Mitel MiVoice 6920 IP phone allows you to set the amount of light that illuminates the LCD display.

Screen Saver Settings:

The **Screen Saver Timer** setting allows you to configure the amount of time (from 1 to 90 minutes) the phone must be idle before the screen saver starts. When the specified amount of time expires, a screen saver displaying the date and time and the number of missed calls (if applicable) is displayed on screen.



The **Dimmer Level** setting defines how brightly you want the screen to be illuminated when the screen saver is displayed.

Note: Availability of the above display settings is dependent on how your account is configured on the MiVoice Business call server.

To configure the display settings:

- 1. Press the (Settings) key on the phone to enter the Settings menu.
- 2. Navigate to the **Display** setting and press the **Select** button or **Select** softkey.

Display		
Brightness Brightness Level	< L	evel 5
Screen Saver		
Screen Saver Timer	5	minutes
Dimmer Level	Level 1	
Save		Cancel

- **3.** Use the left and right navigation keys to increase or decrease the intensity of brightness on the LCD (valid values are from Levels 1 (darkest) to 5 (brightest).
- 4. Press the down navigation key to highlight Screen Saver Timer.

- **5.** Enter a value, in minutes, using the dialpad keys. You can set the timer from 1 to 90 minutes. The default is 5 minutes.
- 6. Press the down navigation key to highlight **Dimmer Level**.
- 7. Use the left and right navigation keys to increase or decrease the intensity of brightness on the LCD when the screen saver is displayed (valid values are from Levels 0 (LCD off) to 5 (brightest).
- 8. Press the **Save** softkey to save your changes.

Config Teleworker

The Mitel MiVoice 6920 IP phone can be configured to work in teleworker mode, which allows remote workers access to the same rich communications capabilities as in-office workers when using the IP phones.

The **Config Teleworker** menu allows you to enter the IP address of the server providing the teleworker configuration details.

To enter the IP address of the teleworker configuration server on the Mitel MiVoice 6920 IP phone:

- 1. Press the (Settings) key on the phone to enter the Settings menu.
- 2. Navigate to Config Teleworker using the navigation keys and press the Select softkey.

		0	
Teleworker S	ettings		
IP Address			
Sava	Packenaco		Cancol
Save	Васкѕрасе		Cancel

Mitel MiVoice 6920 Config Teleworker

3. Enter the IP address of the server providing the teleworker configuration details in the **IP Address** field using the dialpad keys.

Note: The **Backspace** softkey can be used to delete the last digit/character entered and the dot (".") softkey can be used to enter a dot where applicable.

4. Press the Save softkey to save your changes.

Diagnostics

The **Diagnostics** menu allows you to perform the following diagnostic routines:

- Audio Diagnostics
- Ping
- TCP DUMP
- DHCP Trace

Audio Diagnostics

The **Audio Diagnostics** sub-menu allows you to collect up to 5 minutes of audio log files that can help to debug audio issues on the Mitel MiVoice 6920 IP phone.

To capture audio diagnostic logs on the Mitel MiVoice 6920 IP phone:

- 1. Press the (Settings) key on the phone to enter the Settings menu.
- Navigate to Diagnostics > Audio Diagnostics using the navigation keys and press the Select softkey.

		U
Audio Diagr	nostics	
Timeout (1-5 Minutes)	1
Start	Backspace	Close

Mitel MiVoice 6920 Audio Diagnostics

In the Timeout input field, enter the amount of time (in minutes from 1 to 5) you would like to run the audio diagnostic tool for using the dialpad keys.
 The IP phone displays "Capturing..." and when the timeout elapses, "Collecting Logs..." is displayed. When all the logs have been collected, a "Complete..." message is displayed.

Note: Press the Stop softkey at any time to stop capturing the audio diagnostic logs.

To save the audio diagnostic logs to your PC using the Mitel Web UI:

- 1. Open your web browser and enter the phone's IP address or host name into the address bar.
- Enter your username and password and click OK/Login. The status window displays for the IP phone you are accessing.

Note: The default username is "root" and the default password is "73738".

3. Click on Diagnostics.

🛤 Mitel	
Tools Reboot Phone Upgrade Software Upload File Diagnostics Screenshot	Diagnostics Updated on 06/09/17 12:03:01 PM UTC by root Capture
© 2017 Mitel Networks Corporation Timeout [1>1440 Minutes] 2 Start/Stop Start Collect Logs	
	Collect/tar Logs Collect Erase Logs Erase Get Log Files Automatic Upload Enable Syslog Level TRACED • Save
	Copyright 2017 Mitel Networks Corporation

- 4. Click on the Download Audio Log Files link beside Get Log Files.
- 5. Save the file to the desired location on your computer.

Note: For more information on Mitel Web UI features, see the *Mitel MiVoice 6900 Series IP Phones Administrator Guide.*

Ping

The **Ping** sub-menu allows you to ping a host name or IP address directly from the Mitel MiVoice 6920 IP phone. This tool can be used to verify whether or not network connections between the Mitel MiVoice 6920 IP phone and other network endpoints are intact.

To use the ping tool on the Mitel MiVoice 6920 phone:

- **1.** Press the (Settings) key on the phone to enter the Settings menu.
- 2. Navigate to **Diagnostics > Ping** using the navigation keys and press the **Select** softkey.

Mitel MiVoice 6920 Ping

Pin	3				
Но	ost Name or IP	Addr			
Ba	ckspace		123	Þ	• 0

3. In the input field, enter the host name or IP address of the network endpoint you want to ping using the dialpad keys.

Note: The **Backspace** softkey can be used to delete the last digit/character entered, the dot (".") softkey can be used to enter a dot where applicable, and the abc/123 softkey can be used to switch the dialpad keys from alphabetic to numeric.

- 4. Press the Ping softkey to begin. The IP phone displays the number of packets sent, the number of packets received, and the Round-Trip Time (RTT) min/avg/max in milliseconds.
- 5. In the input field, enter the host name or IP address of the network endpoint you want to ping using the on-screen keyboard.
- Tap the Ping softkey to begin.
 The IP phone displays the number of packets sent, the number of packets received, and the Round-Trip Time (RTT) min/avg/max in milliseconds.

TCP DUMP

The **TCP DUMP** sub-menu allows you to capture TCP network packets for up to 1440 minutes (24 hours) that can in turn be used to help debug and troubleshoot various issues.

To capture TCP network packets on the Mitel MiVoice 6920 IP phone:

- 1. Press the (Settings) key on the phone to enter the Settings menu.
- 2. Navigate to Diagnostics > TCP DUMP using the navigation keys and press the Select softkey.

TCP DUMP		
Timeout (:	1-1440 Minutes)	2
Start	Backspace	Close

Mitel MiVoice 6920 TCP DUMP

3. In the **Timeout** input field, enter the amount of time (in minutes from 1 to 1440) you would like to capture TCP network packet data using the dialpad keys.

The IP phone displays "Capturing..." and when the timeout elapses, "Complete..." is displayed.

Note: Press the Stop softkey at any time to stop capturing the TCP DUMP logs.

To save the TCP network packet logs to your PC using the Mitel Web UI:

- 1. Open your web browser and enter the phone's IP address or host name into the address bar.
- Enter your username and password and click OK/Login. The status window displays for the IP phone you are accessing.

Note: The default username is "root" and the default password is "73738".

3. Click on Diagnostics.

🕅 Mitel		
Tools Reboot Phone Upgrade Software Upload File Diagnostics Screenshot	Diagnostics Updated on 06/09/17 12:03:0 Capture	1 PM UTC by root
© 2017 Mitel Networks Corporation	Timeout [1>1440 Minutes] 2 Start/Stop Start Collect Logs	
	Collect/tar Logs Erase Logs Get Log Files Automatic Upload Syslog Level	Collect Erase Enable TRACE0 • Save
	Copyright 2017 Mitel Netwo	rks Corporation

- 4. Click on the Download Log Files link beside Get Log Files.
- 5. Save the file to the desired location on your computer.

Note: For more information on Mitel Web UI features, see the *Mitel MiVoice 6900 Series IP Phones Administrator Guide*.

DHCP Trace

The **DHCP Trace** sub-menu allows you to perform a DHCP trace request. You can then review the parameters that the DHCP trace returns. After you have reviewed the trace information, the DHCP Lease is released.

To perform a DHCP trace on the on the Mitel MiVoice 6920 IP phone:

1. Press the (Settings) key on the phone to enter the Settings menu.

2. Navigate to **Diagnostics** > **DHCP Trace** using the navigation keys and press the **Select** softkey.

DHCP Trace	
mac_addr: 08:00:0f:9f:9c:ec	
ip addr: 10.30.100.105	
netmask: 255.255.255.0	
gateway: 10.30.100.1	
dns: 10.72.2.26	
	Cancel

Mitel MiVoice 6920 DHCP Trace

The following parameters and associated values are displayed:

- mac_addr: The MAC address of the IP phone.
- **ip addr:** The IP address of the IP phone.
- **netmask:** The IP address range local to the IP phone.
- gateway: The IP address of the network's gateway or default router IP address.
- **dns:** The IP address of the primary DNS server.
- icp: The IP address of the MiVB call server.
- tftp: The IP address of the TFTP server.
- http: The IP address of the HTTP server.
- ipa: The IP address of the IPA server.
- VLAN: The VLAN id of the IP phone.
- **I2p:** L2P priorities for default, voice, signaling, and other.
- dscp: DSCP values for default, voice, signaling, and other.

Restart

The **Restart** setting allows you to restart your phone. You may need to restart your phone if you have been asked to do so by your System Administrator should you experience any unexpected behavior.

To restart your phone:

- 1. Press the (Settings) key on the phone to enter the Settings menu.
- 2. Navigate to the **Restart** option and press the **Select** button or **Select** softkey.

Restart Ph		
	Restart Phone?	
	Yes No	
Restart		Cancel

3. When the prompt, **Restart Phone?** appears, navigate to the **Yes** button and press the **Select** button or press the **Restart** softkey.

If you do not wish to restart your phone, press **No** or the **Cancel** softkey.

Note: Your phone will be temporarily out of service during the restart process.

Configuring and using programmable keys

Your Mitel MiVoice 6920 IP phone has six programmable, multi-function, self-labeling keys allowing you to program and use up to 18 functions.



Notes:

- 1. If Mitel M695 Programmable Key Modules (PKMs) are attached to the phone, you can configure up to an additional 28 keys on each PKM.
- 2. The Mitel MiVoice 6920 IP phone allows up to three PKMs on each phone. For more information about PKMs, see "Mitel M695 Programmable Key Module (PKM)" on page 86.

If enabled by your System Administrator, you can program or edit the following functions on the programmable keys by pressing and holding the respective programmable key for approximately four seconds:

Function	Default label
Speed Call	N/A
Phone Lock	Phone Lock
Call Forward Always	Fwd Always
Do Not Disturb	DND
Account Code Non Verified	Acct Non-Ver
Account Code Verified	Acct Verified

Note: Programming keys using the press-and-hold method is dependent on your Class of Service settings defined for your account on the MiVoice Business call server.

Speed Call

The **Speed Call** key allows you to dial a specified number with one key press. **Speed Call** keys can be useful as they can be programmed to dial external numbers, dial directly to another person's extension or quickly access features that use feature access codes. You can also transfer calls to or create conference calls using your **Speed Call** keys in place of dialing out manually.

To program or edit a **Speed Call** key:

1. Press and hold the applicable programmable key for four seconds.

Programmable K	ey	
Speed Call Other Features	Label Name	
More	Private	
Save Ba	ckspace ABC 🕨	Cancel

2. In the Label Name field, enter a label to apply to the key.

Note: If necessary, use the **ABC** 4 softkey to specify uppercase letters or lowercase letters when entering the label and use the **Backspace** softkey to delete the last character/digit entered. To add a space, press the * (star) key twice.

- **3.** Press the down navigation key to move to the **Number** field and enter a number using the dialpad keys.
- 4. (Optional) Press the down navigation key to move to the Private checkbox and press the Select button to make the key a Private Speed Call key. When a Private Speed Call key is pressed, the call is considered private and caller ID information is not displayed in the phone's call history.
- 5. Press the Save softkey to save the information to the key you selected.

To make a speed call, press the key that you have programmed with Speed Call functionality.

Other Features

Phone Lock

By programming a **Phone Lock** key you can lock your phone to prevent the phone from being used or configured. **Phone Lock** prevents access to all the features on the phone, with the following exceptions:

- Unlocking the set via a user PIN.
- Hot Desk Login and Logout support.
- Emergency Call Notification support.

Phone Lock has no effect on incoming calls but restricts outgoing calls, with the following exceptions:

- · Calls to emergency trunk routes.
- · Local operators.

All hard keys on the device are disabled except for the dialpad and volume keys. All programmable keys and softkeys are disabled except for the **Phone Lock**, **Hot Desk**, and Emergency Notification feature access keys.

To program a **Phone Lock** key:

- 1. Press and hold the applicable programmable key for four seconds.
- 2. Press the left navigation key to move to the feature category column.
- 3. Press the down navigation key to highlight Other Features.
- 4. Press the right navigation key to move to the **Other Features** selection column.

Programmable Key	
Speed Call Other Features More	Phone Lock Call Forward Always Do Not Disturb Account Code Non Ver Account Code Verified
Save	Cancel

5. With Phone Lock highlighted, press the Save softkey.

To lock the phone:

1. Press the Phone Lock key.

A "Locked" prompt appears on screen and the **Phone Lock** key turns light blue indicating the phone is locked.

To unlock the phone:

- 1. Press the Phone Lock key again.
- 2. At the "Enter PIN:" prompt, enter your pin number and press the Enter softkey.

Call Forward Always

Programming a **Call Forward Always** key allows you to quickly enable and disable call forwarding for all calls on your phone even if you have other call forward modes enabled through your **Call Forward** settings menu.

For example, you can enable settings to forward calls to your voicemail when your phone is busy or unanswered. Then, when you leave your desk, you can quickly enable **Call Forward Always** to forward calls to another phone, such as your home phone or cell. After you return to your desk and disable **Call Forwarding Always**, the defaults for busy or unanswered conditions remain active.

To program a Call Forward Always key:

- 1. Press and hold the applicable programmable key for four seconds.
- 2. Press the left navigation key to move to the feature category column.
- 3. Press the down navigation key to highlight Other Features.
- 4. Press the right navigation key to move to the Other Features selection column.
- 5. Navigate to Call Forward Always using the down navigation key.

Programmable Key	
Speed Call	Phone Lock
Other Features	Call Forward Always
More	Do Not Disturb
	Account Code Non Ver
	Account Code Verified
Save	Cancel

6. With Call Forward Always highlighted, press the Save softkey.

To enable Call Forward Always:

- Press the key that you have programmed with Call Forward Always functionality. When Call Forward Always is enabled, a (Call Forward Always) icon is displayed in the status bar.
- 2. If a number is already programmed, press **Activate** to enable forwarding to the current number or

If a number has not been programmed, enter a new destination number and then press the **Activate** softkey.

Notes:

- 1. When **Call Forward Always** is enabled, it takes precedence over all other types of call forwarding.
- 2. If Call Forward Always is enabled, then the Call Forward Always icon is displayed on the Screen Saver.
- 3. If both **Call Forward Always** and **DND** are enabled, then the **DND** icon is displayed on the Screen Saver.



- · Call Forward Always Icon

To disable Call Forward Always:

- Press the key that you have programmed with Call Forward Always functionality. When Call Forward Always is disabled, a (Call Forward Always) icon is removed from the status bar.
- 2. Press the **Deactivate** softkey.

Note: When **Call Forward Always** is disabled, the default call forwarding settings are active.

Do Not Disturb

When you program and enable a **Do Not Disturb** key, callers may get a busy tone and an on-screen text message indicating that you do not wish to be disturbed, depending on the setting in the System Administrator Tool.

To program a **Do Not Disturb** key:

- 1. Press and hold the applicable programmable key for four seconds.
- 2. Press the left navigation key to move to the feature category column.
- 3. Press the down navigation key to highlight **Other Features**.
- 4. Press the right navigation key to move to the **Other Features** selection column.
- 5. Navigate to **Do Not Disturb** using the down navigation key.

Programmable Key	
Speed Call	Phone Lock
Other Features	Call Forward Always
More	 Do Not Disturb
	Account Code Non Ver
	Account Code Verified
Save	Cancel

6. With **Do Not Disturb** highlighted, press the **Save** softkey.

To enable **Do Not Disturb**:

 Press the key that you have programmed with **Do Not Disturb** functionality. When DND is enabled, a
 (DND) icon is displayed in the status bar and the **Do Not Disturb** key turns red.

Notes:

4. The DND icon is displayed on the Screen Saver.



To disable Do Not Disturb:

 Press the key that you have programmed with **Do Not Disturb** functionality. When DND is disabled, the (DND) icon is removed from the status bar and the **Do Not Disturb** key returns to its normal color.

Account codes

Accounts codes are an effective way of grouping calls together for accounting, auditing, and reporting purposes. When you tag a call with an account code, the code is picked up in the records generated by your call server. This allows System Administrators the ability to quickly identify groups of calls, which can then be used by different departments for organizational and reporting purposes.

The following are the two types of account codes that you can add to a call:

Account Code Non Verified:

A non-verified account code is one that has not been pre-programmed on your call server. Non-verified account codes can be used to group a set of calls that has not yet been verified and pre-programmed on your call server.

Account Code Verified:

A verified account code is one that has been pre-programmed on your call server by a System Administrator.

Account Code Non Verified

To program an Account Code Non Verified key:

- 1. Press and hold the applicable programmable key for four seconds.
- 2. Press the left navigation key to move to the feature category column.
- 3. Press the down navigation key to highlight Other Features.
- 4. Press the right navigation key to move to the **Other Features** selection column.
- 5. Navigate to Account Code Non Verified using the down navigation key.

Programmable Key	
Speed Call	Phone Lock
Other Features	Call Forward Always
More	📃 Do Not Disturb
	Account Code Non Ver
	Account Code Verified
Save	Cancel

6. With Account Code Non Verified highlighted, press the Save softkey.

To assign a non-verified account code before a call:

- 1. Press the key that you have programmed with Account Code Non Verified functionality.
- 2. Enter the digits (up to 12) for the account code using the dialpad keys.
- **3.** Press the # (hash) key.
- 4. Enter the number you wish to call.

To assign a non-verified account code during a call:

- 1. While on an active call, press the key that you have programmed with Account Code Non Verified functionality.
- 2. Enter the digits (up to 12) for the account code using the dialpad keys.
- 3. Press the # (hash) key.

Account Code Verified

To program an Account Code Verified key:

- 1. Press and hold the applicable programmable key for four seconds.
- 2. Press the left navigation key to move to the feature category column.
- 3. Press the down navigation key to highlight Other Features.
- 4. Press the right navigation key to move to the **Other Features** selection column.
- 5. Navigate to Account Code Verified using the down navigation key.

Programmable Key		
Speed Call	Phone Lock	
Other Features	Call Forward Always	
More	Do Not Disturb	
	Account Code Non Ver	
	Account Code Verified	
Save	Cancel	

6. With Account Code Verified highlighted, press the Save softkey.

To assign a verified account code before a call:

- 1. Press the key that you have programmed with Account Code Verified functionality.
- 2. Enter the digits (up to 12) for the account code using the dialpad keys.
- 3. Press the # (hash) key.
- 4. Enter the number you wish to call.

To assign a verified account code during a call:

- 1. While on an active call, press the applicable key that you have programmed with **Account Code Verified** functionality.
- 2. Enter the digits (up to 12) for the account code using the dialpad keys.
- 3. Press the # (hash) key.

Making calls

This section describes ways to make calls on your Mitel MiVoice 6920 IP phone, using your handset, speakerphone or headset.

Dialing a number

First, take the phone off-hook by:

- · Lifting the handset for handset operation, or
- Pressing the () (Speaker/Headset) key or Prime Line key for speaker or headset operation. Your phone must be configured to use the Speaker audio path for speaker operation or Headset audio path for headset operation.

At the dial tone, enter the number you wish to call. When your party picks up, a timer appears on your display that records the length of your call.

Alternatively, you can enter the number you wish to call without taking the phone off-hook. This method of dialing will use the speakerphone or headset audio path depending on the audio path setting you selected in the **Settings > Audio > Audio Path** menu. For more information, see the section "Audio Path" on page 26.

Note: When dialing, press the **Backspace** softkey to delete the last digit entered. To cancel the dialing process, press the **End Call** softkey or the **Call** (**Goodbye**) key.

If the dialed number matches with a contact stored in the **Personal**, **Mobile** or **Corporate** folder in the **Contacts** application, then the contact name is displayed. The contact name look up is in the following order:

- 1. Personal Contacts, where respective Personal Contacts name is displayed.
- 2. Mobile Contacts, where respective Mobile Contacts name is displayed.
- MiVoice Business System Contacts, where if the MiVoice Business System is configured with MiNET Phone Object Message Name, then the respective MiVoice Business System name is displayed.
- 4. Corporate Contacts, where respective Corporate Contacts name is displayed.

If there is no match, then the number returned by the MiVoice Business system is displayed.

If you are unable to make calls within certain area codes, check with your System Administrator for any toll restrictions placed on your extension that may restrict your access to long distance area codes or dialing prefixes.

Initiating a callback

The callback feature allows you to be notified when a busy phone becomes free or when a phone has been used after a no-answer condition was encountered.

To request a callback when the call you initiated is busy or has not answered:

1. Press the Call Me Back softkey.

2. Hang up.

When the party is available, your phone starts ringing, and the display shows the callback ringing indicator.

To answer a callback when you see the callback ringing indicator:

1. Lift the handset, or press the ()/() (Speaker/Headset) key or Prime Line key.

Redialing a number

You can redial the most recently dialed number that is displayed on the **Home** screen by pressing the **Redial** hard key twice or by pressing the **Redial** softkey.

Pressing the **Redial** hard key once accesses the **Outgoing** calls list in the **Call History** application. For more information on the **Call History** application, see "Call History" on page 67.

Receiving calls

When a call is ringing at your extension, the callers name, number, and avatar (if applicable) display on the **Call** screen. Additionally, the **Prime Line** key icon flashes indicating the incoming call.

If the incoming number matches with a contact stored in the **Personal**, **Mobile** or **Corporate** folder in the **Contacts** application, then the contact name is displayed. The contact name look up is in the following order:

- 1. Personal Contacts, where respective Personal Contacts name is displayed.
- 2. Mobile Contacts, where respective Mobile Contacts name is displayed.
- MiVoice Business System Contacts, where if the MiVoice Business System is configured with MiNET Phone Object Message Name, then the respective MiVoice Business System name is displayed.
- 4. Corporate Contacts, where respective Corporate Contacts name is displayed.

If there is no match, then the name returned by the MiVoice Business system is displayed.



Answering an incoming call

To answer an incoming call:

- Lift the handset for handset operation.
- For speakerphone operation, press the ()(Speaker/Headset) key, or the Prime Line softkey. Your phone must be configured to use the Speaker audio path.
- For headset operation, press the ()(Speaker/Headset) key or the answer button on your headset (if available). Your phone must be configured to use the Headset audio path.

Note: The audio path setting you selected in the **Settings > Audio > Audio Path** menu determines if the call goes to the speakerphone or headset. For more information, see the section "Audio Path" on page 26.

Manually forwarding an incoming call

If call forwarding numbers have been defined for the **Always** or **No Answer** states, press the **Forward Call** softkey to manually forward an incoming call to the defined number. A common way

of using this feature is to configure call forwarding to forward calls to your voicemail service. You can then can manually decide whether to answer an incoming call or send the call to voicemail.

- 1. The number defined for the **Always** state has precedence over the numbers defined for the **No Answer** states.
- 2. For more information on how to configure the **Call Forward** feature, see the section "Call Forward" on page 22.

Handling calls

When you are connected to a call, you can use hard keys on the phone perform various action such as placing a call on hold or muting the microphone. Softkeys can also be used to perform actions such as transferring calls or creating multi-party conference calls.

Placing a call on hold

You can place an active call on hold by pressing the [11] (Hold) key.

To place a call on hold (while on an active call):

1. Press the (Hold) key. Your applicable Line key flashes the ((Hold) icon indicating the call is on hold. To let your caller know that they are on hold, music plays softly (if this has been set up on your call server).



Note: After a prolonged period of time on hold, the held call may automatically switch to an incoming call to remind you that the call was on hold.

2. To retrieve a call on hold press the applicable Line key.

Muting the microphone

You can use the *(Mute)* key to mute the microphone on the handset, headset, or speakerphone. When you use the *(Mute)* key on your phone, you cannot be heard on an active call or on a conference.

To indicate a muted microphone, the LED beside the *(Mute)* key turns on. To switch mute on or off, press the *(Mute)* key.

Note: If you place a muted call on hold, the phone automatically takes the call off mute when you reconnect to the call.

Transferring a call

There are two types of call transfers that can be performed on the Mitel MiVoice 6920 IP phone:

Unattended Transfer:

An unattended transfer is when you transfer a call directly to another extension without consulting with the person receiving the call. To do this, complete the transfer immediately after you have entered the number. The call goes directly to the extension or outside line to which you transferred.

Attended transfer:

You also have the option to consult with the person to which you are transferring the call, before you complete the transfer. To do this, remain on the line until the receiving party answers the call. After consulting with the receiving party, you can either complete the transfer or cancel the transfer to go back to the original call.

To transfer a call:

1. Ensure you are on active call with the party you wish to transfer.



2. Press the **Transfer** softkey.

The active call is placed on hold.



3. Enter the transfer recipient's number.



4. Press the **Transfer** softkey or hang up the handset to complete an unattended call transfer or

Wait for an answer, consult, and then press the **Transfer** softkey or hang up the handset to complete an attended transfer.

Creating a multi-party conference call

The Mitel MiVoice 6920 IP phone supports multiples parties in a conference call.

To create a multi-party conference call:

1. Ensure you are on active call with one of the parties with whom you wish to create a conference.



2. Press the Add User softkey. The active call is placed on hold.



3. Enter the conference target's number.



4. Wait for an answer, consult, and then press the **Join Calls** softkey to create the three-way conference call.



Notes:

- 1. Pressing the **Trade Calls** softkey swaps the active call with the other party currently on hold.
- 2. Pressing the **Split** softkey splits the conference call by placing all callers on hold except for the initial caller.
- 5. Repeat steps 2 to 4 to add additional parties to the conference.

Note: If you want to create a conference call with two parties you are already in calls with (on different line appearances), press the **Add Held** softkey and press the applicable **Line** key with the held call.

Ending a conference and transferring remaining parties

The Mitel MiVoice 6920 IP phone allows a host to easily leave a conference call without disrupting the connection of the other remaining parties.

If you are the host of a conference call and want to leave the conference, press the **Leave Call** softkey on the phone. Your phone leaves the conference call but the remaining parties are still connected together.

Ending calls

To end a call, you first need to connect or reconnect to the call if not already connected (e.g. if your caller is on hold). With the call connected, place the handset on its cradle, press the **End Call** softkey, or press the **Call** softkey.

Managing calls

The Mitel MiVoice 6920 IP phone has several features that make it easier to manage calls. Three of the main call management features include the **Contacts** application, the **Call History** application, and **Voicemail**.

Contacts

The **Contacts** application is your personal phone book and directory, conveniently stored within your phone. The Mitel MiVoice 6920 IP phone supports a localized **Personal** contact folder as well as enhanced functionality allowing for interoperability with LDAP (**Corporate**) directories.



Note: Access to LDAP (Corporate) directories must be set up by your System Administrator.

General functions

Accessing and navigating the Contacts application

To access and navigate the Contacts application:

1. Press the 🔔 (Contacts) key.



2. You can view the list of contacts on the right for each respective contact folder by pressing the up and down navigation keys.

Note: If any of the folders are empty, a "Directory empty" message displays on screen.

 Press the right navigation key to move to the contact selection column. From this column, you can use the up and down navigation keys to scroll through the list of contacts.

Contacts	٩	Enter name	
Personal	5 And 457	erson, Colin 327	
Corporate (110 Cole 3452	e, David 231	
	Gah 9023	a, Annie	
	Johr 3435	nson, James	
	Ten: 1435	ner, Danny 535	
Dial	Delete	Add New	Close

4. From the contact selection column, press the right navigation key to display the contact card for the selected contact.

Contacts		Personal
CA Anderson, Colin	Phone Numbers	
	457327	Work
	6472307117	Mobile
Dial E	dit	Close

- 5. Press the up and down navigation keys to scroll through the list of numbers.
- 6. Press the (Goodbye) key, or Close softkey to exit.

Performing a search

Search functionality is applied to your **Personal** contact folder and your **Corporate** contact folder if your corporate directory is configured to be cached.

Note: If you do not see a number indicating the amount of contacts beside your **Corporate** folder, your corporate directory is configured to be non-cached. See "Performing a search on a non-cached Corporate contact folder" on page 63 if your phone is configured to use a non-cached corporate directory.

Pressing on any dialpad key initiates the search. After entering the first few letters of the contact's first name or last name and initiating the search, you can then quickly select a contact and perform the applicable action. Use the following procedure on the phone's UI to perform a search.

To perform a search:

1. Press the (Contacts) key.



2. Using the dialpad keys enter in the first letters of the contact's name and (if required) press the **Search** softkey.

Contacts	Q. Col
Personal 🛐	Anderson, Colin 457327
Corporate 110	Cole, David 345231
	Gaha, Annie 902323
	Johnson, James 343535
	Tenner, Danny 143535
Search Backs	space abc 🕨 Close
3. When matches are found, navigate to the contact entry and perform your desired action.



Dialing from the Contacts application

Dialing can be performed by manually navigating to the contact, selecting a phone number (if the contact has more than one) and then by picking up the handset, or by pressing the **Dial** softkey, **Select** button or **Select**

To dial a contact's phone number.

- 1. Press the 🔝 (Contacts) key.
- **2.** (Optional) Using the dialpad keys input the first letters of the contact's name to initialize the search function.
- 3. Use the navigation keys to highlight the contact you want to call.



- 4. (Optional) If the contact has multiple phone numbers defined, press the right navigation key to navigate to the contact card and use the up and down navigation keys to select the phone number you wish to call.
- Press the Dial softkey, Select button or (3/(3) (Speaker/Headset) key for speaker or headset operation or

Pickup the handset for handset operation.

Personal folder features

The **Personal** folder contains the contacts (up to 50) that have been created or copied directly to the phone using the phone UI.

From the **Personal** folder you can search for a personal contact, place a call to the selected personal contact, add new personal contacts, delete all personal contacts, delete individual personal contacts, or edit existing personal contacts. You can also view contact details (e.g. avatar, name, and numbers) by navigating to the contact card.

Adding a new contact to your Personal folder

To add a new contact to your **Personal** folder:

- **1.** Press the (Contacts) key.
- 2. Press the Add New softkey.

Contacts - Add New	Personal
Name	
Number	First Name
	Last Name
Save	Cancel

3. Press the right navigation key to move to the **First Name** input field and enter the first name of the personal contact using the dialpad keys.

Contacts - Ad	d New		Personal
Name			
Number	John	ame	
	Last Na	ame	
Save	Backspace	abc 🕨	Cancel

- 4. (Optional) Press the down navigation key to move the Last Name input field and enter the last name of the personal contact using the dialpad keys.
- 5. Press the left navigation key to return to the Name and Number category column.

6. Press the down navigation key to select the **Number** category.

Contacts - Add New	Persona	al
Name		
Number	Work	
	Mobile	b
Save	Cancel	

7. With **Number** highlighted, press the right navigation key to move to the number type tab.

Contacts - Add New		Personal
Name		
Number	Work	>
	Mobile	
Save		Cancel

8. Use the left and right navigation keys to switch the type of number to either Work, Home, or Mobile.

Contacts - Add New			Personal
Name			
Number	<	Home	>
	Mobile		
Save			Cancel

9. Press the down navigation key to move to the number input field and enter the number of the personal contact using the dialpad keys.

Contacts - Add New		Personal
Name		
Number	Home	
	5556700	
	Mobile	
Savo	Packapaco	Cancel
Save	васкорасе	Cancel

- **10.** (Optional) Press the down navigation key to move to the second number type tab field and repeat steps 8 and 9 to enter a secondary number.
- 11. Press the Save softkey to save the contact to your Personal folder.

Editing an existing contact in your Personal folder

To edit an existing contact in your **Personal** folder:

- **1.** Press the (Contacts) key.
- **2.** (Optional) Using the dialpad keys input the first letters of the contact's name to initialize the search function.
- 3. Use the navigation keys to highlight the contact you want to edit.



4. Press the right navigation key to view the contact card for the selected contact and press the **Edit** softkey.



- 5. Use the navigation keys to navigate through the various tabs and fields and edit any information using the dialpad keys.
- 6. Press the Save softkey to save your changes.

Deleting all contacts from the Personal folder

- **1.** Press the (Contacts) key.
- 2. Press the **Delete** softkey.

Contacts	Q Enter name	
Personal 🚳	Anderson, Colin 457327	
Corporate 110	Cole, David 345231	
	Gaha, Annie 902323	
	Johnson, James 343535	
	Tenner, Danny 143535	
Del	ete Add New Close	

A delete all confirmation message appears on screen.

Contacts Po	ersonal
Delete all contacts from Personal	?
Yes	No

3. Press the Yes softkey to delete all contacts from your Personal folder.

Note: Press the No softkey to cancel the delete process.

Deleting a contact from the Personal folder

To delete contacts from your Personal folder:

- **1.** Press the (Contacts) key.
- **2.** (Optional) If you want to delete an individual entry, using the dialpad keys input the first letters of the contact's name to initialize the search function.
- 3. Highlight the applicable contact and press the **Delete** softkey.



A delete confirmation message appears on screen.

Contacts	
Delete Anderson, Colin from Personal?	
Yes No	

4. Press the Yes softkey to delete the selected contact from your Personal folder.

Note: Press the No softkey to cancel the delete process.

Corporate folder features

The **Corporate folder** contains the contacts that are available through your corporate LDAP directory.

Notes:

- 1. Access to corporate directories must be set up by your System Administrator.
- **2.** Updates made to your corporate directory are reflected on your phone only at the re-synchronization time configured on the MiVoice Business phone system.

There are two types of corporate directories that can be configured:

1. Cached:

Contacts are cached on the phone and are viewable as a list in the **Corporate** contact folder upon entering the Contacts application. You can scroll up and down to view your list of corporate contacts. Performing a basic search applies to both the **Personal** and **Corporate** contacts folders.

2. Non-cached:

Contacts are not cached on the phone and therefore contacts are not available until you perform a lookup in the **Corporate** contact folder. Used for corporate directories with a large amount of contacts as caching a large amount of contacts could, at times, overload the IP phone.

You can perform all the same functions as the **Personal** folder from the **Corporate** folder except for adding, deleting, and editing entries. In addition to being able to search for a corporate contact, place a call to the selected corporate contact, and view a corporate contact's contact card, you can copy the corporate contact to your **Personal** folder.

If a call is placed from the **Corporate** folder matches with a contact stored in the **Personal** or **Mobile** in the **Contacts** application, then the contact name is displayed. The contact name look up is in the following order:

Note: See "Performing a search" on page 56 for details on how to perform a search on your cached **Corporate** contact folder.

Performing a search on a non-cached Corporate contact folder

To perform a search on a non-cached Corporate contact folder:

1. Press the (Contacts) key.



2. Press the down navigation key to select the Corporate folder.

Corporate	Q Lastname Firstname
Personal	
Corporate	
	Press "Search" key to search corporate contacts
Search	Close

- 3. Press the Search softkey to enter the search bar.
- 4. Using the dialpad keys enter in the first letters of the contact's last name and press the **Search** softkey.

Corporate	Q smi
Personal	
Corporate	
	Press "Search" key to search corporate contacts
Search Backs	pace abc 🕨 Close

Note: For non-cached corporate directories, searches are based on the last name of the contact.

5. If matches are found, navigate to the contact entry and perform your desired action.



Copying a Corporate contact to the Personal folder

To copy a contact in your Corporate folder to your Personal folder:

- **1.** Press the (Contacts) key.
- For non-cached corporate directories, perform a search as detailed in "Performing a search on a non-cached Corporate contact folder" on page 63 or

(Optional) For cached corporate directories, using the dialpad keys input the first letters of the contact's name to initialize the search function.

3. Use the navigation keys to highlight the corporate contact you want to copy to your **Personal** folder.



4. Press the right navigation key to view the corporate contact's contact card.

Contacts		Corporate
DC	Phone Numbers	
Smith Dave	73814	Work
Department Marketing Location Toronto		
Dial Add C	ontact	Close

5. Press the Add Contact softkey.

A copy confirmation message is displayed.

Call History

The **Call History** application is a stored log of all your calls. You can view details of all your outgoing and received calls as well as any calls that were unanswered or missed. Your telephone logs the type of call, information regarding the remote party (name, number, and avatar), and the time of the call.

You can view and delete entries in the **Call History** as well as copy selected entries to the your **Personal** folder in the **Contacts** application. You can also dial an entry directly from the **Call History** application.



The following table identifies and describes the various icons displayed in the **Call History** application:

lcon	Description
⊻	Indicates an incoming call that was not answered (missed).
2	Indicates an outgoing call (outgoing)
∠	Indicates an incoming call that was answered (received).

To access and use the Call History application:

1. Press the 🛛 (Call History) key.



- 2. You can view the list of call entries on the right for each respective type of call by pressing the up and down navigation keys.
- 3. Press the right navigation key to move to the call entry column. From this column, you can use the up and down navigation keys to scroll through the call entries.
- **4.** From the call entry column, press the right navigation key to display the individual call details for the selected call.



Note: The call details screen contains the date and time of the call, and the caller ID details.

 Press the Dial softkey, Select button or ()/() (Speaker/Headset) key for speaker or headset operation

or

Pickup the handset for handset operation.

6. Press the **Goodbye**) key, or **Close** softkey to exit.

Deleting entries from the Call History application

You can delete all entries in the **Call History** application regardless of call type, all entries within a call type folder, or individual entries.

To delete all entries from the Call History application:

1. Press the *(Call History)* key.

2. Use the navigation keys to highlight the My Phone call type folder.



- 3. Press the **Delete** softkey.
 - A delete all confirmation message appears on screen.

Call History	
Delete all entries in My Phon	e?
Delete	Cancel

4. Press the **Delete** softkey again to delete all entries in the **Call History** application.

Note: Press the Cancel softkey to cancel the delete process.

To delete all entries from a call type folder:

- 1. Press the 🛃 (Call History) key.
- 2. Use the navigation keys to highlight the applicable call type folder.



3. Press the Delete softkey.

A delete all confirmation message appears on screen.

Call History	
Delete all entries in Outgoing?	
Delete	Cancel

4. Press the **Delete** softkey again to delete all entries in the selected call type folder.

Note: Press the Cancel softkey to cancel the delete process.

To delete individual entries from a call type folder:

- 1. Press the 😹 (Call History) key.
- 2. Use the navigation keys to highlight the applicable call type folder.
- 3. Press the right navigation key to move to the call entry column.
- 4. Use up and down navigation keys to highlight the individual call entry.



5. Press the **Delete** softkey.

A delete confirmation message appears on screen.

Call History	
Delete Christiano Gabriani from My	Phone?
Delete	Cancel

6. Press the Delete softkey again to delete the individual call entry.

Note: Press the Cancel softkey to cancel the delete process.

Copying entries to the Personal folder in the Contacts application

To copy a Call History entry to the **Personal** folder in the **Contacts** application:

- 1. Press the 🛃 (Call History) key.
- 2. Use the navigation keys to highlight the applicable call type folder.
- 3. Press the right navigation key to move to the call entry column.
- 4. Use the navigation keys to highlight the call entry you want to copy to your **Personal** folder in your **Contacts** application.

Call History		
🛃 My Phone	Christiano Gabriani 10:17am 17 Mar 2017	2
💌 Missed	Dani Pedrosa 10:17am 17 Mar 2017	2
Outgoing	Livio Suppi 10:17am 17 Mar 2017	
	Livio Suppi 10:16am 17 Mar 2017	<u>~</u>
	Maverick Vinales 11:02am 14 Mar 2017	.⊷
Dial Add Co	ontact Delete C	lose

5. Press the Add Contact softkey. A copy confirmation message is displayed.



Missed calls indicator

The Mitel MiVoice 6920 IP phone has a missed calls icon located on the status bar that indicates the number of calls that went unanswered.



As the number of unanswered calls increment, the information associated with the calls are stored in the **Call History** application under the **Missed** call folder. When the user accesses the **Missed** call folder, the missed call indicator on the **Home** screen is removed.

Note: Missed calls are also indicated on the screensaver.

To view missed call details in Call History application:

- **1.** Press the *(Call History)* key.
- 2. Use the navigation keys to highlight the Missed call type folder.



- 3. Press the right navigation key to move to the call entry column.
- 4. Use up and down navigation keys to highlight the individual call entry.
- 5. Press the right navigation key to see more detailed information about the call entry.
- Press the (Goodbye) key, or Close softkey to exit.
 The missed calls indicator on the Home screen is cleared upon exit.

Voicemail

Note: Voicemail functionality must be configured by your System Administrator.

When voicemail functionality is enabled, the MWI LED on the Mitel MiVoice 6920 flashes red and the or (Voicemail) icon displays on the status bar indicating that voicemail messages are available. You can access your voicemail service by pressing the Voicemail key.



To access your voicemail:

1. Press the e (Voicemail) key.

Notes:

- 1. If the handset is offhook or if the phone is in speakerphone or headset mode, pressing the (Voicemail) key will call your message center directly.
- **2.** The call to your message center is placed regardless of whether or not a message is pending.
- 2. (Optional) If your phone is idle, press the **More** softkey to view the date and time of the last voicemail message received.

Note: Pressing the **Erase** key clears the current message indicators on the phone (i.e. MWI LED and last voicemail message received history). It does not delete the messages contained in your voicemail account.

- 3. If your phone is idle, press the **Call** softkey.
- **4.** When you are connected to your message center, follow the audio prompts to perform any applicable actions.

Note: For more detailed information on voicemail features and functionality, see the *Mitel MiVoice Business Voice Mail User Guide*.

Visual voicemail

Note: Visual voicemail functionality must be configured by your System Administrator.

Visual voicemail allows you to visually interact with your voicemail. Visual voicemail displays an accurate count of your voicemail messages on the status bar and the screen saver. The count on the screen saver is a total of missed calls and voicemail messages.

Notes:

- 1. If you have subscribed to Embedded Voicemail (EMEM), missed voicemail count is displayed only if the EMEM and the 6900 Series IP Phones are programmed on the same MiVoice Business System.
- 2. If you have subscribed to NuPoint visual voicemail, the message count is not supported.



Visual voicemail by default, selects **New** and displays new voicemail messages in Bold font. Urgent voicemail messages are indicated by a Red exclamation point.

Enable voicemail application

To enable voicemail application:

1. Press the (Settings) key on the phone to enter the Settings menu.



2. Navigate to Voicemail using the left navigation key and then press the Select softkey.

Settings			
<	Uoice	o 🔨	∡ >
Select	Advanced	Log Issue	Close

3. Press the Select button to select the key opens Voicemail App checkbox.



4. Press the Save softkey.

Note: For security reasons, the password can be enabled only from the user login screen.

Log into your voicemail

To log into your voicemail:

1. Press the (Voicemail) key.

2. In the **Passcode** field, enter the passcode.

Voicemail Use	er Login	
Mailbox Numb	per:	2506
Passcode:	(Remember passcode
Login	Backspace	Close

3. (Optional) Press the down navigation key and press the **Select** button to select the **Remember passcode** checkbox.

Note: If **Remember passcode** checkbox is selected, the phone stores your passcode and launches your voicemail without having to enter your passcode each time you log into your voicemail.

4. Press the Login softkey to log into your voicemail.

Log out of your voicemail

To log out of your voicemail:

1. Press the Logout softkey.



Note: After five minutes of inactivity, you are automatically logged out of your mailbox and the **Voicemail** application closes. When you close the **Voicemail** application, you are not automatically logged out of your mailbox.

Play a voicemail

To play a voicemail:

- 1. Log into your voicemail.
- 2. Use the up and down navigation keys to choose one of the following:
 - All Displays all voicemail messages.
 - New Displays new voicemail messages.

Recorded - Displays recorded voicemail messages.



3. Press the right navigation key and then use the up and down navigations keys to choose the voicemail message you want to play.



4. Press the Select button to play the voicemail.

Notes:

- If your voicemail system is on EMEM, then use **Replay**, **Backward**, **Forward**, and **Cancel** softkeys to control the audio of the voicemail.
- If your voicemail system is on NuPoint Unified Messaging, then pressing the **Select** button will pause or stop the voicemail audio. Use the **Cancel** softkey to cancel the operation.

Dial a voicemail

To dial a voicemail:

- **1.** Log into your voicemail.
- 2. Use the up and down navigation keys to choose one of the following:
 - All Displays all voicemail messages.
 - New Displays new voicemail messages.

• Recorded - Displays recorded voicemail messages.



3. Press the right navigation key and then use the up and down navigations keys to choose the voicemail you want to dial.

Voicemail and Recordings					
All	4	Vina 05:4	a lesM 4pm, Today (0	0:01) (\triangleright
New	4	Vina 05:4	alesM 3pm, Today (0	0:02)	!
Recorded		MarquezM 05:41pm, Today (00:06)			
		Vina 05:4	ilesM Opm, Today (O	0:04)	
Dial	Sen	d To	Delete	Clos	e

4. Press the **Dial** softkey to dial the number.

Delete a voicemail

To delete a voicemail:

- 1. Log into your voicemail.
- 2. Use the up and down navigation keys to choose one of the following:
 - All Displays all voicemail messages.
 - New Displays new voicemail messages.
 - Recorded Displays recorded voicemail messages.



3. Press the right navigation key and then use the up and down navigations keys to choose the voicemail message you want to delete.



- Press the Delete softkey.
 A delete confirmation message appears on screen.
- 5. Press the Delete softkey to delete the voicemail.

Save a voicemail

Visual voicemail automatically saves played voicemail messages in the All folder.

Send voicemail to a number

To send a voicemail to a number:

- 1. Log into your voicemail.
- 2. Use the up and down navigation keys to choose one of the following:
 - All Displays all voicemail messages.
 - New Displays new voicemail messages.
 - Recorded Displays recorded voicemail messages.



3. Press the right navigation key and then use the up and down navigations keys to choose the voicemail message you want to send.



- 4. Press the Send To softkey.
- 5. In the Number field, enter the number.

Voicemail Se	nd to	
То	Number	
Send	Backspace	Cancel

6. Press the **Send** softkey to send the voicemail to the number. A confirmation message appears on screen.

Additional features

The following features are available using feature keys only if your System Administrator has defined a programmable key or softkey with the respective functionality on your phone.

When configured on a programmable key, icons are displayed beside the label indicating type of functionality. When a generic ON/OFF icon or a custom ON/OFF icon (such as those displayed for DND, Make Busy, Phone Lock, Call Fwd Always, Night Service, and Record a Call) is displayed, the key acts as a toggle to enable and disable the respective feature. If no icon s displayed beside the key's label, the feature either requires user input or is used to convey information.

For detailed information on how to use any of the following features, please contact your System Administrator.

Feature	Description	Icons	Туре
Account Code NonVerified	Non-verified Account Codes let you enter codes on the SMDR record for billing and/or call management.		Non-Toggle
Account Code Verified	Verified Account Codes let you access features that are not normally available at a station. You enter Verified Account Codes before making a call to change the COS and COR at any station. When you hang up, the station returns to normal.		Non-Toggle
ACD	Automatic Call Distribution (ACD) functionality allows call centers to process incoming calls based on a number of customizable parameters.		Non-Toggle
Auto Answer	Auto-Answer lets you automatically answer calls that ring your Prime line.	OFF ON	Toggle ON/OFF
Call Fwd Always	Call Fwd Always is used to redirect all incoming calls to an alternate number.	<u></u>	Toggle ON/OFF
Call Fwd Busy	Call Fwd Busy is used to redirect both incoming internal and external calls to an alternate number when your phone is busy.	KK KK	Toggle ON/OFF
Call Fwd Busy External	Call Fwd Busy External is used to redirect incoming external calls to an alternate number when your phone is busy.	K	Toggle ON/OFF
Call Fwd Busy Internal	Call Fwd Busy Internal is used to redirect incoming internal calls to an alternate number when your phone is busy.	K	Toggle ON/OFF
Call Fwd No Ans. External	Call Fwd No Ans. External is used to redirect incoming external calls to an alternate number when you do not answer.		Toggle ON/OFF

Feature	Description	Icons	Туре
Call Fwd No Ans. Internal	Call Fwd No Ans. Internal is used to redirect incoming internal calls to an alternate number when you do not answer.	KK KK	Toggle ON/OFF
Call Fwd No Answer	Call Fwd No Answer is used to redirect both incoming internal and external calls to an alternate number when you do not answer.	KK KK	Toggle ON/OFF
Call Park and Call Park - Retrieve	Call Park allows you to place a call in a special hold state. You, or someone else, can then retrieve the call from another extension in the system by pressing the Call Park - Retrieve key.	OFF ON	Toggle ON/OFF
Call Pickup	Call Pickup lets you answer a call that is ringing at another station, including an attendant console.	OFF ON	Toggle ON/OFF
CDE Speedcall	This feature allows you to speed dial telephone numbers that the administrator has programmed into the system.		Non-Toggle
Direct Page	Direct Page allows you to page another telephone over its built-in speaker, even if the telephone is in use.	OFF ON	Toggle ON/OFF
Do Not Disturb	Do Not Disturb (DND) stops internal calls from ringing on your phone and returns busy tone to the caller. DND does not stop external calls from ringing nor does it affect your ability to make calls.	\bigcirc	Toggle ON/OFF
DSS/Busy Lamp	You can use the Busy Lamp Field feature to monitor the status of single-line sets, single-member lines, key system lines, multicall Broadcast Groups, and Hot Desk users.	Line States	Non-Toggle
Emergency Call	The Emergency Call Notification key allows you receive emergency call notifications. Pressing the key allows you to view the caller's directory number, special instructions that are programmed for the DN in the Comments field of the CESID Assignment form, and the date and time of the call.	OFF ON	Toggle ON/OFF
Group Listen	With Group Listen, you can carry on a conversation using the handset or headset while allowing others nearby to listen to the person at the far end over the handsfree speaker.	OFF ON	Toggle ON/OFF

Feature	Description	Icons	Туре
Group Park	Group Park allows you to place a call in a special hold state. You, or someone else, can then retrieve the call from another extension in the system. Group Park uses a single feature key to both park and retrieve calls. Any telephone programmed with the same Group Park key can retrieve the call.	OFF ON	Toggle ON/OFF
Group Presence	Group Presence allows members of call groups (Hunt, ACD, Ring, Personal Ring, and Multi-device User) to make themselves absent or present in a group. Only members who are present in a group are offered calls to that group.	OFF ON	Toggle ON/OFF
Handoff	The Handoff feature for Personal Ring Groups (PRGs) allows calls to be "Pushed" or "Pulled" between group member devices. A Push passes the call to the group, where it can be answered by any other present group device. A Pull takes an in-progress call away from another member.	OFF ON	Toggle ON/OFF
Key System	Key System Groups let multiple telephones share the same extension number. Incoming calls ring all of the idle stations, and the stations stop ringing when one member answers the call. Only one member of a Key System Group can use the line at one time.	Line States	Non-Toggle
Make Busy	The Make Busy feature ensures an ACD extension does not receive ACD calls when the agent is unavailable. The ACD extension remains in the Make Busy state until an agent deactivates it.	× ×	Toggle ON/OFF
Meet Me Answer	Meet-Me Answer lets you respond to a Group Page that you missed, even if you do not know the identity or location of the person that paged you.		Non-Toggle
Message Waiting Indication	Keys can be programmed as message waiting indicators which are associated with the mailboxes of other stations. The indicator flashes when a message is waiting in the associated mailbox. This allows you to look at the phone and recognize which mailboxes have messages.	OFF ON	Toggle ON/OFF

Feature	Description	Icons	Туре
Multicall	Multicall Groups let multiple telephones share the same extension number. Incoming calls ring all of the idle stations, and the stations stop ringing when one member answers the call. A Multicall group allows more than one group member to use the line at the same time.	Line States	Non-Toggle
Night Answer	Night Answer lets you send your calls to another location when the system is in Night Service.	OFF ON	Toggle ON/OFF
Night Service Indication	Night Service switches the system from day service to night service and vice versa. When the system is placed into night service (usually at the end of the workday) incoming external calls ring at an extension(s) selected by the Administrator.		Toggle ON/OFF
Paging	Paging lets you connect to loudspeaker or paging equipment to make an announcement. You can access individual paging zones or all paging zones simultaneously. A one-way connection is established for "All Zones" paging. A two-way connection is established for "Individual Zones" loudspeaker page.	OFF ON	Toggle ON/OFF
Personal Presence	Personal Presence lets you choose which of your personal answer points you want to receive your calls at.	OFF ON	Toggle ON/OFF
Phone Lock	The Phone Lock feature locks a set, preventing access to the majority of features, with the following exceptions: unlocking the set via a user PIN, Hot Desk Login and Logout support, and Emergency Call Notification support. Phone Lock has no effect on incoming calls but restricts outgoing calls, with the following exceptions: calls to emergency trunk routes and local operators.	8	Toggle ON/OFF
Privacy Release	Privacy Release allows members of a Key System Group to disable automatic call privacy protection to allow another member of the group to join the call. When a new member enters the call the original members receive a conference tone and the privacy feature is restored.	OFF ON	Toggle ON/OFF
Record a Call	Record-a-Call (RAC) allows extension users to record a two-party call when one party is connected to a trunk. The recorded conversation is stored in the users' voice mail mailbox.	0 🖲 🕕	Toggle ON/OFF (and Pause)

Feature	Description	Icons	Туре
Secretarial	A DSS key provisioned with the "secretarial" option allows you to perform a transfer to the associated device or directory number.	Line States	Non-Toggle
Single Line	The Single Line feature is used to set the prime line as a single line, whereby it is dedicated to one phone only.	Line States	Non-Toggle
Specific Group Alert	A Specific Group Alert key normally appears only on an ACD supervisor's telephone. Each Specific Group Alert key allows the supervisor to obtain queue status information for a specific agent skill group. A supervisor's telephone may have several Specific Group Alert keys.		Non-Toggle
Superkey	Superkey programming allows for easy access to telephone system features and feature-set customization.		Non-Toggle
Tag Call	Tag Call allows phone users to signal that they are the victim of a malicious call. This Malicious Call Indication is logged as an SMDR record.		Non-Toggle
User Speedcall	User Speedcall keys allow you to store external numbers under feature keys for faster dialing. You can press a Speed Call Key to dial a phone number or, during a call, to outpulse DTMF tones.		Non-Toggle
User Speedcall - Private	A private User Speedcall is a User Speedcall that is considered private, whereby Caller ID information is not displayed in the phone's call history.		Non-Toggle

Accessory support

Mitel M695 Programmable Key Module (PKM)

Up to three Mitel M695 PKMs can be attached to the right of the Mitel MiVoice 6920 IP phone providing you with 84 additional programmable keys (28 keys per PKM).



The Mitel M695 PKM features a 4.3" 480x272 pixel color backlit LCD display that provides a unified graphical user experience with the Mitel MiVoice 6920 IP phone.

The Mitel M695 PKM supports the same programmable key functions as the Mitel MiVoice 6920 IP phone. The programmable keys can only be programmed by your System Administrator.

Notes:

- The fixed keys on the Mitel M695 PKM are not supported on the Mitel MiVoice Business System.
- 2. Use a IEEE 802.3at compliant PoE L2 Switch, IEEE 802.2at Inline Power Injector, or an AC to DC Adapter to power the Mitel MiVoice 6920 IP phone.
- 3. For information how to install your Mitel M695 PKM, see the Mitel M695 PKM Installation Guide.

Mitel Wireless LAN Adapter

The Mitel Wireless LAN Adapter is a wireless bridge that can add wireless connectivity functionality to your Mitel MiVoice 6920 IP phone. It allows your Ethernet-enabled phone to join a secure, high-speed network. It is software transparent meaning no changes to drivers, management tools, or applications are needed.



Feature highlights include:

• Easily add robust, secure wireless capabilities to Ethernet devices:

The Mitel Wireless LAN Adapter is a high-performance wireless network adapter which enables any Ethernet device to become an 802.11a/b/g/n dual band wireless network device, giving you freedom to place it anywhere in your facility.

• Dual band IEEE 802.11a/b/g/n support:

The Mitel Wireless LAN Adapter is designed to communicate in the 2.4 GHz and 5 GHz bands. Radio interference in the commonly used 2.4 GHz band can be avoided by utilizing 5 GHz.

• Gigabit Ethernet support:

The wired LAN port supports 10/100/1000BASE-T (auto-recognition).

• Simple to setup and use:

The Mitel Wireless LAN Adapter is easy to set up using the enclosed network setup cable. No special drivers or software are required.

Enterprise security:

The Mitel Wireless LAN Adapter supports the following security functions:

- WEP (64 Bit/128 Bit)
- WPA-PSK (TKIP/AES)
- WPA2-PSK (AES)
- IEEE 802.1X EAP-PEAP, EAP-TLS, EAP-TTLS, EAP-FAST, EAP-LEAP

Note: For information how to install and set up your Mitel Wireless LAN Adapter, see the *Mitel Wireless LAN Adapter Setup Guide.*

Analog, DHSG/EHS, and USB headsets

The Mitel MiVoice 6920 IP phone accepts analog and DHSG/EHS headsets through the modular RJ22/RJ45 port on the back of the phone or USB headsets through the USB port.



Contact your telephone equipment retailer or distributor to purchase a compatible headset.

Notes:

- 1. The RJ22/R45 headset port is for headset use only. Plugging any other devices into this port may cause damage to the phone and will void your warranty.
- 2. Customers should read and observe all safety recommendations contained in headset operating guides when using any headset.
- **3.** Contact your System Administrator for questions regarding supported headsets or refer to the *MiVoice Business Release 8.0 Release Notes*.
- 4. Headsets connected to the Mitel MiVoice 6920 IP phone are limited in functionality to answering, ending, and muting calls. Additional headset features (such as rejecting calls) are not supported and will cause unexpected behavior (e.g. invoking the reject call function will answer instead of reject).
- 5. If both an analog (non-DHSG) headset and a USB headset are connected to the phone, the USB headset will take precedence over the analog headset (i.e. the USB headset will be functional while the analog headset will not be functional).
- 6. If both a DHSG headset and a USB headset are connected to the phone and DHSG is enabled, the DHSG headset will take precedence over the USB headset.

To make and receive calls using a headset:

- 1. Ensure that you have selected the Headset audio path (see "Audio Path" on page 26).
- 2. Turn the phone over and, depending on the type of headset, locate the headset jack marked ${\rm f}$ or the USB port.
 - a. If you have a USB headset, insert the USB plug into the USB port.
 - **b.** If you have a headset with an RJ9/RJ22 4-pin connector, insert the headset cord into the jack until it clicks into place.
 - **c.** If you have a DHSG/EHS headset with an RJ45 6-pin or 8-pin connector, remove the plastic headset jack adapter and then insert the cord into the jack until it clicks into place.

Note: See the *Mitel MiVoice* 6920 *IP Phone Installation Guide* for more information.

Limited warranty

Mitel warrants this product against defects and malfunctions in accordance with Mitel's authorized, written functional specification relating to such products during a one (1) year period from the date of original purchase ("Warranty Period"). If there is a defect or malfunction, Mitel shall, at its option, and as the exclusive remedy, either repair or replace the product at no charge, if returned within the Warranty Period. If replacement parts are used in making repairs, these parts may be refurbished, or may contain refurbished materials. If it is necessary to replace the product, it may be replaced with a refurbished product of the same design and color. If it should become necessary to repair or replace a defective or malfunctioning product under this warranty, the provisions of this warranty shall apply to the repaired or replaced product until the expiration of ninety (90) days from the date of pick up, or the date of shipment to you, of the repaired or replacement product, or until the end of the original Warranty Period, whichever is later. Proof of the original purchase date is to be provided with all products returned for warranty repairs.

Exclusions

Mitel does not warrant its products to be compatible with the equipment of any particular telephone company. This warranty does not extend to damage to products resulting from improper installation or operation, alteration, accident, neglect, abuse, misuse, fire or natural causes such as storms or floods, after the product is in your possession. Mitel will not accept liability for any damages and/or long distance charges, which result from unauthorized and/or unlawful use.

Mitel shall not be liable for any incidental or consequential damages, including, but not limited to, loss, damage or expense directly or indirectly arising from the customer's use of or inability to use this product, either separately or in combination with other equipment. This paragraph, however, shall not apply to consequential damages for injury to the person in the case of products used or bought for use primarily for personal, family or household purposes.

This warranty sets forth the entire liability and obligations of Mitel with respect to breach of warranty, and the warranties set forth or limited herein are the sole warranties and are in lieu of all other warranties, expressed or implied, including warranties or fitness for particular purpose and merchantability.

Warranty repair services

Should the product fail during the Warranty Period;

- In North America, please call 1-800-574-1611 for further information.
- Outside North America, contact your sales representative for return instructions.

You will be responsible for shipping charges, if any. When you return this product for warranty service, you must present proof of purchase.

After warranty service

Mitel offers ongoing repair and support for this product. This service provides repair or replacement of your Mitel product, at Mitel's option, for a fixed charge. You are responsible for all shipping charges. For further information and shipping instructions:

- In North America, contact our service information number: 1-800-574-1611.
- Outside North America, contact your sales representative.

Note: Repairs to this product may be made only by the manufacturer and its authorized agents, or by others who are legally authorized. This restriction applies during and after the Warranty Period. Unauthorized repair will void the warranty.



mitel.com

© Copyright 2017, Mitel Networks Corporation. All Rights Reserved. The Mitel word and logo are trademarks of Mitel Networks Corporation. Any reference to third party trademarks are for reference only and Mitel makes no representation of ownership of these marks.



Vertiv[™] Liebert® GXT5 UPS

750 - 3000VA 230V

Intelligent and Efficient UPS Protection for your Mission Critical Applications



Vertiv™ Liebert® GXT5 | 750 - 3000VA 230V UPS

Intelligent and Efficient UPS for Protection of Your Mission-Critical Applications

The Vertiv[™] Liebert[®] GXT5 UPS is an online double conversion UPS solution which offers premium power outage protection and continuous power conditioning in a compact and flexible rack/tower form factor.

The Vertiv Liebert GXT5 single phase UPS operates with high power efficiency, making it ideally suited to protect critical infrastructure in both centralized and edge network applications.

Scalable runtime options with matching external battery cabinets offer additional flexibility when extended uninterrupted power is required. Plus, the Liebert GXT5 provides battery health status and replacement date prediction for intelligent battery health management.

Vertiv Liebert GXT5 Features

Leading UPS Technology

- High output PF = 1.0
- Full color graphic LCD display with gravity sensing
- Individually controllable output power sockets
- External battery cabinets with auto-detection
- Battery health status and replacement date prediction
- Remote management, update and configuration
- Optimized thermal management and variable speed fan

The UPS system is easy to deploy and maintain due to its user-friendly LCD interface and remote management capabilities supported by the Vertiv RDU101 communications card which makes the Liebert GXT5 compatible with infrastructure management solutions such as LIFE Services, environmental sensors, Power Insight, and more.

With market leading efficiency and unity power factor operation, the Liebert GXT5 will meet your critical application needs. And you can rest assured that your business is protected with this Vertiv solution that includes a standard, two-year product warranty.



Efficient and Green Product

- High efficiency (up to 94%) in on-line mode
- Energy Star[®] 2.0 certified
- Even higher efficiency (up to 98%) in Active ECO mode
- Programmable sockets for power cycling hung equipment and optimum battery usage
- RoHS and REACH compliance

Solutions Wide

- Compact Rack / Tower design
- Broad range of services and extended warranty
- Easy to install, configure and operate
- Vertiv RDU101 network communications card with advanced features
- Compatibility with environmental sensors
- Integrated dry-contacts with selectable definition
- Free Trellis Power Insight management software
- Serial Port for out-of-band management with Avocent Serial Consoles
- Automatic internal bypass and additional external maintenance bypass as an option


Vertiv[™] Liebert[®] GXT5 Features



High power factor (1.0) More usable power enables more connected loads saving space and costs.



Efficiency (up to 98%) in Active ECO mode Superior protection with maximum efficiency.



Energy Star 2.0 certification. Higher efficiency means an optimized energy management and lower heat dissipation, thus providing energy savings and improved reliability.



User friendly interface provides insight to UPS status for easy installation, configuration and operation.



Compact rack/tower design Space-saving UPS provides rack space optimization and flexible



COLOR

LCD

installation.



Battery cabinets with auto-detection

Be confident your UPS is set up correctly to report available run time when used with external battery cabinets.



Product warranty

Comprehensive coverage through a standard two-year exchange warranty.

10

 $\mathbf{\mathbf{x}}$

VERTIV.

811

 \odot

VERTI

How You Benefit from the Liebert® GXT5 UPS

Designed for high availability



- The higher power factor (1.0) ensures the connection of more loads and IT equipment
- With **individually controlled output sockets**, you can manage power to individual devices without impacting other critical network equipment operation
- Minimum downtime of the device provided by **hot-swappable, user-replaceable battery** modules which can be changed during operation
- Automatic self-test

User-friendly installation and operation



- Easy to read gravity sensing graphical color display
- Intuitive user interface, local configuration and management
- Support for the new Vertiv suite of **remote management tools**
- (*Trellis™* Power Insight, RDU101 NIC and serial connectivity support)
- Autodetection of external battery cabinets enables faster deployment and accurate runtime information
- Remote UPS firmware upgrade capability ensures your UPS has the latest features and enhancements

Longer service life and runtime of the batteries



- Extended runtimes provided by the addition of external battery cabinets
- Improved battery care by temperature compensated battery charging
- Intelligent battery health management ensures a longer service life (optimized battery maintenance and replacement when needed)

Optimized energy and capacity management



- Active ECO operating mode with up to 98% efficiency
- Efficiency in online double conversion mode up to 94%
- Energy Star 2.0 certified
- 4 **individually programmable output sockets** can be use to extend runtime for the most critical loads and smart disconnection of the less critical ones

Seamless connectivity



- Four onboard, user-definable dry contact I/O for integration of support management systems
- Supports SNMP, web, and environmental sensors with the optional RDU101 Communications card
 - Serial connection for integration of Avocent ACS product or direct serial management and control of the UPS



Power Emergency:

Comprehensive Service Support for Critical Systems

- Five-year protection plan with 100% parts coverage and 24x7 emergency service
- **100% parts coverage** for UPS with advance swap within 8 working hours from ticket acknowledgement, which means a total of two business days maximum from customer's claim. This service includes also internal batteries up to the third year.
- 24x7 access to customer resolution center and online access to Customer Services Network

Liebert GXT5 Service Program Items

Power Emergency and Warranty Extensions

UPS	Power Emergency	Warranty Extension 1 year	Warranty Extension 3 years
GXT5-750IRT2UXLE GXT5-1000IRT2UXLE	RUPS-PE5-1kVA	PF1YR-MU-13	PF3YR-MU-13
GXT5-1500IRT2UXLE GXT5-2000IRT2UXLE	RUPS-PE5-2kVA	PF1YR-MU-14	PF3YR-MU-14
GXT5-3000IRT2UXLE	RUPS-PE5-3kVA	PF1YR-MU-15	PF3YR-MU-15



US AND CANADA

LATIN AMERICA

Manuf. and Assembly Locations 13	Manuf. and Assembly Locations 1
Service Centers 100+	Service Centers 20+
Service Field Engineers 850+	Service Field Engineers 240+
Technical Support/Response 120+	Technical Support/Response 20+
Customer Experience Centers/Labs 4	Customer Experience Centers/Labs 2

Key Benefits:

- Reduces the worry of critical equipment downtime
- Ensures rapid recovery in the event of failure (within 24-48 hours)

	POWER EMERGENCY
TECHNICAL SUPPORT HOTLINE	24/7
PARTS INCLUDED	\checkmark^*
RESPONSE TIME	✓ 8 working hours**
CONTRACT DURATION	5 YEARS

* Excluding batteries after the 3rd year.

** Shipment of the new unit within 8 working hours from ticket acknowledgement, which means a total of two business days maximum after customer's claim.

Power Emergency is available in the following EMEA countries: Austria, Belgium, France, Germany, Ireland, Luxemburg, Portugal, Sweden, Switzerland, The Netherlands, United Kingdom, Czech Republic, Croatia, Italy, Poland, Slovakia, Spain, and Turkey.

See specific scope of work for offering details.

OUR PURPOSE

We believe there is a better way to meet the world's accelerating demand for data - one driven by passion and innovation.

OUR PRESENCE

GLOBAL PRESENCE

Manuf. and Assembly Locations 28 Service Centers 250+ Service Field Engineers 2,650+ Technical Support/Response 300+ Customer Experience Centers/Labs 16

ASIA PACIFIC

EUROPE, MIDDLE EAST AND

Manuf. and Assembly Locations 9

Technical Support/Response 90+

Customer Experience Centers/Labs 5

Service Field Engineers 590+

AFRICA

Service Centers 70+

Manuf. and Assembly Locations 5 Service Centers 60+ Service Field Engineers 970+ Technical Support/Response 80+ Customer Experience Centers/Labs 5

Technical Specifications

	GXT5-750IRT2UXLE	GXT5-1000IRT2UXLE	GXT5-1500IRT2UXLE	GXT5-2000IRT2UXLE	GXT5-3000IRT2UXLE
Ratings (VA/W)	750VA/750W	1000VA/1000W	1500VA/1500W	2000VA/2000W	3000VA/3000W
Dimensions, m	m				
Unit (W x D x H)	430 x 400 x 85	430 x 400 x 85	430 x 470 x 85	430 x 470 x 85	430 x 540 x 85
Weight, kg					
Unit	21	21	21	21	28.20
Input AC Parar	neters				
Range	Typical 230V Range: 115-288V				
Frequency	40-70Hz; Auto Sensing				
Input Power Cord	IEC 60320 C14 input socket	IEC 60320 C14 input socket	IEC 60320 C14 input socket	IEC 60320 C20 input socket	IEC 60320 C20 input socket
Output AC Par	ameters				
Output Receptacles	IEC60320-C13 × 8	IEC60320-C13 × 8	IEC60320-C13 × 8	IEC60320-C13 × 8	IEC60320-C13 × 6 IEC60320-C19 × 1
Typical 230V User selectable output	200/208/220/230/240VAC (user-configurable); ±3%				
Waveform	Sine wave				
Utility (AC) Mode Overload	>200% for 250ms; 150- 200% for 2 seconds; 125- 150% for 50 seconds; 105-125% 60 seconds	>200% for 250ms; 150- 200% for 2 seconds; 125- 150% for 50 seconds; 105-125% 60 seconds	>200% for 250ms; 150- 200% for 2 seconds; 125- 150% for 50 seconds; 105-125% 60 seconds	>200% for 250ms; 150- 200% for 2 seconds; 125- 150% for 50 seconds; 105-125% 60 seconds	>200% for 250ms; 150- 200% for 2 seconds; 125- 150% for 50 seconds; 105-125% 60 seconds
Battery					
Туре	Valve-regulated, non-spillable, lead acid				
Backup Time (100% Load)	8.5 Minutes	6 Minutes	5 Minutes	3 Minutes	3 Minutes
Backup Time (50% Load)	21.5 Minutes	15 Minutes	13 Minutes	9 Minutes	9 Minutes
+1 External Battery Cabinet (100% Load)	41 Minutes	28.5 Minutes	25 Minutes	17 Minutes	17 Minutes
+1 External Battery Cabinet (50% Load)	88 Minutes	65 Minutes	59 Minutes	42 Minutes	43.5 Minutes
Environmental	Requirements				
Operating Temperature, °C	0 to 40 °C				
Storage Temperature, °C	-20 to 60 °C contain batteries will be from -15 to 40°C.	-20 to 60 °C contain batteries will be from -15 to 40°C.	-20 to 60 °C contain batteries will be from -15 to 40°C.	-20 to 60 °C contain batteries will be from -15 to 40°C.	-20 to 60 °C contain batteries will be from -15 to 40°C.
Relative Humidity	0% to 95%,				
Operating Elevation	3,000m	3,000m	3,000m	3,000m	3,000m
Storage Elevation	15 000m				
Audible Noise	<46dBA max at 1m front and sides < 43dBA at 1m rear	<46dBA max at 1m front and sides < 43dBA at 1m rear	<46dBA max at 1m front and sides < 43dBA at 1m rear	<48dBA max at 1m front and side <48dBA max at 1m rear	<48dBA max at 1m front and side <48dBA max at 1m rear
Agency					
Surge Immunity	ANSI C62.41 Category B				
Transportation	ISTA Procedure 1A				
Safety	EN 62040-1:2008+A1:2013				
Emissions	EN 62040-2:2006 EN 61000-3-2:2014 EN 61000-3-3:2013				
Environmental	WEEE and ROHS2 REACH				
Warranty					
Warranty	Std. 2 year; Opt. +1 or +3 year	Std. 2 year; Opt. +1 or +3 year	Std. 2 year; Opt. +1 or +3 year	Std. 2 year; Opt. +1 or +3 year	Std. 2 year; Opt. +1 or +3 year



Available Liebert GXT5 UPS Accessories

External battery cabinets and replacement battery kits

UPS	External battery cabinets	Replacement battery kits
GXT5-750IRT2UXLE	GXT5-EBC36VRT2UE	GXT5-36BATKIT
GXT5-1000IRT2UXLE	GXT5-EBC36VRT2UE	GXT5-36BATKIT
GXT5-1500IRT2UXLE	GXT5-EBC48VRT2UE	GXT5-48BATKIT
GXT5-2000IRT2UXLE	GXT5-EBC48VRT2UE	GXT5-48BATKIT
GXT5-3000IRT2UXLE	GTX5-EBC72VRT2UE	GXT5-72BATKIT

Network communications and environmental sensors

Network Communications	RDU101	Intellislot web card for SNMP and web management
	RELAYCARD-PG	Intellislot Relay Card
Environmental Sensors	SN-Z01	Integrated cable with single temperature sensor
SN-Z02 SN-Z03 SN-T SN-TH SN-2D	SN-Z02	Integrated cable with three temperature sensors
	SN-Z03	Integrated cable with three temperature and one humidity sensors
	SN-T	Modular with single temperature sensor
	SN-TH	Modular with single temperature and single humidity sensor
	SN-2D	Modular with two door contact inputs
	SN-3C	Modular with three dry contact inputs

Micro POD

LIDE	DOD	Output Receptacles / Protection		
0P5	POD	socket group A	socket group B	
GXT5-750IRT2UXLE GXT5-1000IRT2UXLE GXT5-1500IRT2UXLE	MP2-210K	4 x IEC 60320 C13 with 10 A breaker	4 x IEC 60320 C13 with 10 A breaker	
GXT5-2000IRT2UXLE	MP2-220K	4 x IEC 60320 C13 with 10 A breaker	4 x IEC 60320 C13 with 10 A breaker	
GXT5-3000IRT2UXLE	MP2-220L	4 x IEC 60320 C13 with 10 A breaker	2 x IEC 60320 C13 with 10 A breaker 1 x IEC 60320 C19 with 16 A breaker	

Installation

RMKIT18-32	All GXT5 battery modules as well as all GXT5 750 thru 3000VA models (racks with 18" - 32" deep support rails)
RS500	Flush-mount shelf for 19" wide telecom/relay racks
RS600	Center-mount shelf for 19" wide telecom/relay racks
RS700	Flush-mount shelf for 23" wide telecom/relay racks
RS800	Center-mount shelf for 23" wide telecom/relay racks
2UTELECOMRKIT	Center-mount brackets for 19" wide telecom/relay racks

Vertiv[™] VR Rack

Supports a wide variety of equipment and gives you the flexibility you need with easy installation



Reliably distributes power to the rack, supporting dynamic data center operations and DCIM



Prevents overheating of network switches by directing cool air to switch intakes, keeping hot exhaust air out

Vertiv ACS Console

Enables integrated remote monitoring, out-of-band management, and IoT connectivity

Vertiv KVM Switch

Enables single-point access for switching between multiple computers

Vertiv Rack Cooling

Provides energy-efficient cooling close to the IT equipment and UPS units

Vertiv[™] Liebert[®] GXT5

Helps protect mission-critical equipment from all power disturbances due to blackouts, brownouts, sags, surges or noise interference **Racks and enclosures:** Support a wide variety of equipment with the Vertiv VR Rack including servers, storage, switches, routers, PDUs, UPS units, console port servers, and KVM switches.

Rails and mounting hardware: Install equipment with a four-post rail kit and hardware for mounting in a 19- or 23-inch rack or choose a two-post telecom rack for front- or mid-chassis, wall, or Zero U configuration.

Rack mount PDUs: For basic or intelligent power distribution that helps prevent overloaded circuits in the data center, choose from products such as upgradeable PDUs, inline power meters, transfer switches, and monitoring sensors.

Extended battery modules: Enable scalable runtime for support during extended power outage situations by adding reliable power and protection to new or existing deployments.

Environmental sensor: Maintain knowledge of remote environments with temperature, humidity and leak detection, or monitor available dry contact sensors for security access control or smoke detection.



Vertiv.com | Vertiv Infrastructure Limited, George Curl Way, Southhampton, SO18 2RY, VAT Number: GB188146827

© 2019 Vertiv Group Corp. All rights reserved. Vertiv³⁴ and the Vertiv logo are trademarks or registered trademarks of Vertiv Group Corp. While every precaution has been taken to ensure accuracy and completeness herein, Vertiv Group Corp. assumes no responsibility, and disclaims all liability, for damages resulting from use of this information or for any errors or omissions. Specifications are subject to change without notice.



DS-2CD2163G2-I(S) 6 MP AcuSense Fixed Dome Network Camera





Empowered by deep learning algorithms, Hikvision AcuSense technology brings human and vehicle targets classification alarms to front- and back-end devices. The system focuses on human and vehicle targets, vastly improving alarm efficiency and effectiveness. High quality imaging with 6 MP resolution

HIKVISION

 Clear imaging against strong backlight due to 120 dB true WDR technology

HIKVISION

- Efficient H.265+ compression technology
- Focus on human and vehicle targets classification based on deep learning
- -S: Audio and alarm interface available
- Water and dust resistant (IP67) and vandal-resistant (IK10)





Specification

Camera	
Image Sensor	1/2.8" Progressive Scan CMOS
Min. Illumination	Color: 0.005 Lux @ (F1.6, AGC ON), B/W: 0 Lux with IR
Shutter Speed	1/3 s to 1/100,000 s
Slow Shutter	Yes
P/N	P/N
Wide Dynamic Range	120 dB
Day & Night	IR cut filter
Angle Adjustment	Pan: 0° to 355°, tilt: 0° to 75°, rotate: 0° to 355°
Lens	
Lens Type	Fixed focal length, 2.8 and 4 mm optional
Focal Longth & FOV	2.8 mm, horizontal FOV 107°, vertical FOV 57°, diagonal FOV 128°
Focal Length & FOV	4 mm, horizontal FOV 87°, vertical FOV 46°, diagonal FOV 105°
Lens Mount	M12
Iris Type	Fixed
Aperture	F1.6
DORI	
DOP	2.8 mm: D: 73 m, O: 29 m, R: 15 m, I: 7 m
DOKI	4 mm: D: 90 m, O: 35 m, R: 18 m, I: 9 m
Illuminator	
Supplement Light Type	IR
Smart Supplement Light	Yes
IR Wavelength	850 nm
IR Range	Up to 30 m
Video	
Max. Resolution	3200 × 1800
	50 Hz: 20 fps (3200 × 1800)
Main Stroom	25 fps (2688 × 1520, 1920 × 1080, 1280 × 720)
	60 Hz: 20 fps (3200 × 1800)
	30 fps (2688 × 1520, 1920 × 1080, 1280 × 720)
Sub Stroom	50 Hz: 25 fps (640 × 480, 640 × 360)
Sub Stream	60 Hz: 30 fps (640 × 480, 640 × 360)
	50 Hz: 10 fps (1920 × 1080, 1280 × 720, 640 × 480, 640 × 360)
Third Stream	60 Hz: 10 fps (1920 × 1080, 1280 × 720, 640 × 480, 640 × 360)
	*Third stream is supported under certain settings.
	Main stream: H.265/H.264/H.264+/H.265+
Video Compression	Sub-stream: H.265/H.264/MJPEG
video compression	Third stream: H.265/H.264
	*Third stream is supported under certain settings.
Video Bit Rate	32 Kbps to 16 Mbps
Н.264 Туре	Baseline Profile/Main Profile/High Profile
Н.265 Туре	Main Profile
H.264+	Main stream supports



H.265+	Main stream supports
Bit Rate Control	CBR/VBR
Scalable Video Coding (SVC)	H.264 and H.265 encoding
Region of Interest (ROI)	1 fixed region for main stream and sub-stream
Audio	
Audio Compression	-S: G.711ulaw/G.711alaw/G.722.1/G.726/MP2L2/PCM/MP3/AAC
Audio Dit Data	-S: 64 Kbps (G.711ulaw/G.711alaw)/16 Kbps (G.722.1)/16 Kbps (G.726)/16 Kbps to 64
Audio Bit Rate	Kbps (AAC)/32 to 192 Kbps (MP2L2)/8 to 320 Kbps (MP3)
Audio Sampling Rate	-S: 8 kHz/16 kHz/32 kHz/44.1 kHz/48 kHz
Environment Noise Filtering	-S: Yes
Network	
Durate sala	TCP/IP, ICMP, HTTP, HTTPS, FTP, DHCP, DNS, DDNS, RTP, RTSP, NTP, UPnP, SMTP,
Protocols	IGMP, 802.1X, QoS, IPv4, IPv6, UDP, Bonjour, SSL/TLS, PPPoE, SNMP
Simultaneous Live View	Up to 6 channels
API	Open Network Video Interface (Profile S, Profile G), ISAPI, SDK
User/Host	Up to 32 users. 3 user levels: administrator, operator and user
	Password protection, complicated password, HTTPS encryption, IP address filter,
Security	Security Audit Log, basic and digest authentication for HTTP/HTTPS, TLS 1.1/1.2, WSSE
	and digest authentication for Open Network Video Interface
Network Storage	NAS (NFS, SMB/CIFS), auto network replenishment (ANR)
Client	iVMS-4200, Hik-Connect, Hik-Central
	Plug-in required live view: IE 10, IE 11
Web Browser	Plug-in free live view : Chrome 57.0+, Firefox 52.0+
	Local service: Chrome 57.0+, Firefox 52.0+
Image	
Image Parameters Switch	Yes
	Rotate, mirror, privacy mask, saturation, brightness, contrast, sharpness, gain, white
Image Settings	balance adjustable by client software or web browser
Day/Night Switch	Day, Night, Auto, Schedule
Image Enhancement	BLC, HLC, 3D DNR
SNR	≥ 52 dB
Interface	
Ethernet Interface	1 RJ45 10 M/100 M self-adaptive Ethernet port
On-board Storage	Built-in memory card slot, support microSD/SDHC/SDXC card, up to 256 GB
	-S: 1 input (line in), max. input amplitude: 3.3 Vpp, input impedance:4.7 K Ω , interface
Audio	type: non-equilibrium; 1 output (line out), max.output amplitude: 3.3 Vpp, output
	impedance: 100 Ω , interface type: non-equilibrium,mono sound
Alarm	-S: 1 input, 1 output (max. 12 VDC, 30 mA)
Hardware Reset	Yes
Event	
Denie Frank	Motion detection (human and vehicle targets classification), video tampering alarm,
Basic Event	exception
Smart Event	Line crossing detection intrusion detection human and vehicle targets classification
	Line crossing detection, intrusion detection, numan and vehicle targets classification



Linkage Method	Upload to FTP/memory card/NAS, notify surveillance center, trigger recording, trigger
	capture, send email
Camora Matorial	Aluminum alloy body
	Bubble: plastic
Camera Dimension	Ø110.8 mm × 84.7 mm (Ø4.4" × 3.3")
Package Dimension	134 mm × 134 mm × 108 mm (5.3" × 5.3" × 4.3")
Camera Weight	Approx. 520 g (1.1 lb.)
With Package Weight	Approx. 720 g (1.6 lb.)
Storage Conditions	-30 °C to 60 °C (-22 °F to 140 °F). Humidity 95% or less (non-condensing)
Startup and Operating Conditions	-30 °C to 60 °C (-22 °F to 140 °F). Humidity 95% or less (non-condensing)
	33 languages
	English, Russian, Estonian, Bulgarian, Hungarian, Greek, German, Italian, Czech, Slovak,
Web Client Language	French, Polish, Dutch, Portuguese, Spanish, Romanian, Danish, Swedish, Norwegian,
	Finnish, Croatian, Slovenian, Serbian, Turkish, Korean, Traditional Chinese, Thai,
	Vietnamese, Japanese, Latvian, Lithuanian, Portuguese (Brazil), Ukrainian
General Function	Anti-flicker, heartbeat, password reset via email, pixel counter
Firmware Version	V5.5.113
Software Reset	Yes
Power Consumption and Current	12 VDC, 0.5 A, max. 6 W
Tower consumption and current	PoE (802.3af, 36 V to 57 V), 0.25 A to 0.15 A, max. 7.5 W
Power Supply	12 VDC ± 25%, reverse polarity protection
Tower Suppry	PoE: 802.3af, Class 3
Power Interface	Ø5.5 mm coaxial power plug
Approval	
	FCC (47 CFR Part 15, Subpart B); CE-EMC (EN 55032: 2015, EN 61000-3-2: 2014, EN
EMC	61000-3-3: 2013, EN 50130-4: 2011 +A1: 2014); RCM (AS/NZS CISPR 32: 2015); IC
	(ICES-003: Issue 6, 2016); KC (KN 32: 2015, KN 35: 2015)
Safety	UL (UL 60950-1); CB (IEC 60950-1:2005 + Am 1:2009 + Am 2:2013); CE-LVD (EN
Jaiety	60950-1:2005 + Am 1:2009 + Am 2:2013);
Environment	CE-RoHS (2011/65/EU); WEEE (2012/19/EU); Reach (Regulation (EC) No 1907/2006)
Protection	IP67 (IEC 60529-2013), IK10 (IEC 62262: 2002)

Available Model

DS-2CD2163G2-I (2.8/4 mm) DS-2CD2163G2-IS (2.8/4 mm) DS-2CD2163G2-IS (2.8 mm) (Black)



Dimension







 Accessory Optional 					
DS-1271ZJ-110 (Black) Pendant Mount	DS-1259ZJ(Black) Junction Box	DS-1275ZJ-SUS Vertical Pole Mount	DS-1250ZJ Rain Shade	DS-1272ZJ-110 (Black) Wall Mount	
DS-1280ZJ-DM18	DS-1272ZJ-110B	DS-1276ZJ-SUS	DS-1276ZJ-SUS	DS-1259ZJ	
Junction box	(Black)	Corner mount	(Black)	Inclined Ceiling	
			Corner Mount	Mount	
				O P	
DS-127271-110B	DS-127171-110	DS-127271-110	DS-1275ZJ-SUS	DS-1280ZJ-DM18	
Wall Mount	Pendant Mount	Wall Mount	(Black)	(Black)	
			Vertical Pole Mount	Junction Box	
e			6.6.6		
DS-1250ZJ(Black)					
Rain Shade					





Distributed by

Γ



Headquarters

No.555 Qianmo Road, Binjiang District, Hangzhou 310051, China T +86-571-8807-5998 overseasbusiness@hikvision.com

Hikvision USA T +1-909-895-0400 sales.usa@hikvision.com

Hikvision Australia T +61-2-8599-4233 salesau@hikvision.com

Hikvision India T +91-22-28469900 sales@pramahikvision.com

Hikvision Canada T +1-866-200-6690 sales.canada@hikvision.com

Hikvision Thailand T +662-275-9949 sales.thailand@hikvision.com Hikvision Europe T +31-23-5542770 sales.eu@hikvision.com

Hikvision Italy T +39-0438-6902 info.it@hikvision.com

Hikvision Brazil T +55 11 3318-0050 Latam.support@hikvision.com

Hikvision Turkey T +90 (216)521 7070- 7074 sales.tr@hikvision.com

Hikvision Malaysia T +601-7652-2413 sales.my@hikvision.com Hikvision UK & Ireland T +01628-902140 sales.uk@hikvision.com

Hikvision South Africa Tel: +27 (10) 0351172 sale.africa@hikvision.com

Hikvision France T +33(0)1-85-330-450 info.fr@hikvision.com

Hikvision Kazakhstan T +7-727-9730667 nikia.panfilov@hikvision.ru

Hikvision Vietnam T +84-974270888 sales.vt@hikvision.com Hikvision UAE T +971-4-4432090 salesme@hikvision.com

Hikvision Singapore T +65-6684-4718 sg@hikvision.com

Hikvision Spain T +34-91-737-16-55 info.es@hikvision.com

Hikvision Tashkent T +99-87-1238-9438 uzb@hikvision.ru

Hikvision Hong Kong T +852-2151-1761 info.hk@hikvision.com Hikvision Russia T +7-495-669-67-99 saleru@hikvision.com

Hikvision Korea T +82-(0)31-731-8817 sales.korea@hikvision.com

Hikvision Poland T +48-22-460-01-50 info.pl@hikvision.com

Hikvision Indonesia T +62-21-2933759 Sales.Indonesia@hikvision.com

Hikvision Colombia sales.colombia@hikvision.com

© Hikvision Digital Technology Co., Ltd. 2020 | Data subject to change without notice |

FlatPack2 | 2U Integrated





POWERFUL TECHNOLOGY



Описание продукта

Flatpack2 2U Integrated специально разработанная система для того, чтобы соответствовать требованиям высокой плотности мощности и более компактным размерам системных решений.

Flatpack2 2U Integrated — это компактная система питания постоянного тока включающая в состав блок контроля и управления (Smartpack), контакторы отключения нагрузки и батарей (LVLD,LVBD). Система может включать до 4 выпрямительных модулей Flatpack2.

Использование технологии переключения режимов резонансной технологии и высокочастотных переключений позволило уменьшить высоту и массу для достижения максимальной надежности.

Система поддерживает широкий диапазон входных напряжений (от 85 до 290VAC) и выдает плавный стартовый ток в момент включения питания.

Применение

Беспроводная, оптоволоконная и проводная связь

Современная связь требует эффективных и компактных систем постоянного тока. **FlatPack2** обладает очень высокой плотностью мощности, имеет экономичную конструкцию, способен работать с широким диапазоном входных напряжений и в жестких условиях эксплуатации, что делает незаменимым для любой телекоммуникационной области.

Широкополосный и сетевой доступ

Постоянное наращивание и модернизация инфраструктуры сетевых служб требует простых и гибких решений по увеличению мощности систем гарантированного электропитания. **FlatPack2**—ваш ключ к решению будущих задач.



Основные особенности

✓ Высокая эффективность при минимальном пространстве

Резонансная технология делает модуль ведущим по эффективности в индустрии телекоммуникаций и способствует крайне компактным размерам выпрямительного модуля.

✓ Цифровые контроллеры

Цифровая идеология управления позволяет получить превосходный контроль за состоянием выпрямительных модулей. Количество компонентов сокращено на 40% - для безотказной, долговечной, бесперебойной работы энергетической системы.

✓ Управление охлаждением

Воздушный поток, проходя через радиаторы, создает модулю наиболее приемлемую рабочую среду и отсутствие ограничений универсальности системных решений.

✓ Уникальный способ установки

Простая система установки модулей Plug&Play не требует затрат времени на установку или замену модулей.

🗸 Стандарты

Удовлетворяет CE, UL, NEBS



FlatPack2 | 2U Integrated

Технические характеристики

Вход

Напряжение Частота	1 фаза 230VAC 45-66 Гц
Максимальный ток	см. описание выпрямительного модуля Flatpack2 48VDC
Защита по входу	Плавный старт Варистор для защиты от переходных процессов Предохранители на входе (L&N) Отключение при напряжении более 290VAC

Контроль и управление

Модуль управления	см. описание модуля контроля и управления Smartpack		
Управление на месте	С помощью LCD экрана и кнопок управления на модуле либо с помощью ПК (ПО Powersuit)		
Удаленное управление	ПО Powersuit через модем или управление и мониторинг с помощью WebPower (Web интерфейс и SNMP- протокол)		
Реле аварийной сигнализации	6 реле		
Визуальные индикаторы	Зеленый LED: вкл. Красный LED: Срочная авария Желтый LED: Несрочная авария LCD – системная информация		
Цифровые входы	6 (для мониторинга внешнего оборудования)		
Измерители тока	Ток батарей Ток выпрямителей		
Аварийная сигнализация	 Отключение автомата нагрузки Отключение автомата батарей Срабатывание LVD Низкое выходное напряжение (задается 2 уровня) Высокое выходное напряжение (задается 2 уровня) Емкость батарей Температура Симметрия 		

Выход	
Напряжение	48 VDC
Максимальный ток	150A

Информация для заказа

241115.001	Flatpack2 48V Rectifier Module
M20402.000	Flatpack2 2U Integrated - 1ph 7,2kW 2x100A Battery Breakers, 150A
	LVBD, 6 I/O, 1.5m temp probe
900210-xxxxx	Flatpack2 2U Integrated – Sales configured system

МестонахождениеКомпРоссия и СНГООСEuropeEltekAmericasEltek

Компания ООО «Элтек» Eltek Energy, AS Eltek Energy, LLC Телефон +7 812 3321117 +47 32 20 32 00 +1 815 459 9100 Факс +7 812 3273577 +47 32 20 32 10 +1 815 459 9118

Распределение постоянного тока

Автоматы защиты сети (нагрузка)	6 (2-32A)
Автоматы защиты сети (батареи)	1 или 2 (максимум 100А каждый)
Программируемое LVD	Опционально: 1 LVBD (150A) Опционально: 1 LVLD (150A)

Соединения

Соединение батарей	Батарейный кабель, болты Мб.
Соединение нагрузочных МСВ	Отрицательное распределение с проводного коннектора
Присоединение аварийной сигнализации	Доступ с тыльной части проводного коннектора

Другие характеристики

Изоляция	3.0 kVAC – вход/выход 1.5 kVAC – вход/земля 0.5 kVDC – выход/земля
Рабочая температура	от -40 до +65 ^о С
Температура хранения	от -40 до +85 ^о С
Размер	482x370x89 мм (2U) (вхшхг)
Вес	5 кг без выпрямителей

Стандарты

Электрическая защита	IEC 60950-1 UL 60950
Электромагнитная совместимость	ETSI EN 300 386 V.1.3.1 (telecommunication network) EN 61000-6-2 (immunity, industry) EN 61000-6-3 (emission, light industry)
Экология	ETSI EN 300 019-2 ETSI EN 300 132-2



e-catalogue international version

1

CAT. N° 3 101 72



DAKER DK PLUS CONVERTIBLE UPS WITH BATTERY -SINGLE PHASE VFI - 3000 VA - 2700 W

Pack (number of units) Volume (dm³) 91,802 Weight (g) 29 681,00

Product characteristics

Convertible UPS with batteries

Single phase

- The stated backup time in minutes is estimated and may vary according to the load characteristics, operating conditions and environment
- Nominal power: : 3000 VA
- Active power: : 2700 W
- Backup time: : 7 min
- Weight: : 30 kg

General characteristics

DAKER DK PLUS - On-line double conversion UPS (rack/tower) - single phase and three phase VFI

The main parameters of the UPS, including the battery charge level and faults, are displayed on the LCD screen on the front panel

 The integrated communication software not only controls the UPS and its switch-off if there is a malfunction, and enables the user to test the main functions remotely, communicate via SNMP/Internet/network adaptor and access the functions of the UPS via the Internet, but can also send the user an SMS if specific events occur

The internal extension connector enables aWEB/SNMP card or a relay interface to be installed which provides insulated contacts for applications on industrial control panels or remote alarm panels

 If there is an electronic fault, overload, overheating or for scheduled maintenance operations, the automatic or manual (optional) bypass ensures continuity of the power supply for critical loads. A bypass switch is available for maintenance

Documentation

Catalogue pages & additional information

- Description page (p.707)
- Technical page (p.708)
- DAKER DK PLUS

Installation & technical data

■ <u>UPS_LGR_0105_GB_AA</u>

🔀 Video

■ <u>B2B-VIDEO-LG-310172</u>

X Installation instructions & related documents LE09705AD

- Documents & videos ■ <u>B2B-VIDEO-LG-310172</u>
- Environmental product profile LGRP-00427-V02

* All prices given are the manufacturer's list prices for professional customers, and are given for information only. They are not to be considered as a mandatory retail price - Basis : before-tax price list, May 2018

The information given in this e-catalogue (characteristics and dimensions) may be amended, and therefore does not represent an undertaking on our part. The e-catalogue is reserved for professionals only. You are also reminded that all products must be installed by a qualified professional in accordance with the installation rules.



MINI-LINK 6600 and MINI-LINK 6366 Release 1.17

Product Specification

Copyright

 \odot Ericsson AB 2015–2021. All rights reserved. No part of this document may be reproduced in any form without the written permission of the copyright owner.

Disclaimer

The contents of this document are subject to revision without notice due to continued progress in methodology, design and manufacturing. Ericsson shall have no liability for any error or damage of any kind resulting from the use of this document.

Contents

1	Introduction	1
1.1	Scope	3
2	Ethernet Characteristics	4
2.1	Traffic Capacity	4
2.2	Traffic Latency	13
2.3	Radio Link Bonding	70
2.4	Hierarchical Radio Link Bonding	71
3	Radio Link	73
3.1	Channel Separation to Frame ID Relation	73
3.2	Radio Frequencies	74
3.3	Carrier Aggregation	114
3.4	Transmitter Performance	116
3.5	Emission Designator	121
3.6	Receiver Performance	122
3.7	Supported Physical Modes	138
	Dhusiani Interferen	
4	Physical Interfaces	160
4 4.1	Ethernet Traffic Interfaces	160 160
4 4.1 4.2	Physical Interfaces Ethernet Traffic Interfaces PDH Interfaces	160 160 163
4 4.1 4.2 4.3	Physical Interfaces Ethernet Traffic Interfaces PDH Interfaces SDH Interfaces	160 160 163 164
4 4.1 4.2 4.3 4.4	Physical Interfaces Ethernet Traffic Interfaces PDH Interfaces SDH Interfaces PDH Traffic Connectors	160 160 163 164 164
4 4.1 4.2 4.3 4.4 4.5	Physical Interfaces Ethernet Traffic Interfaces PDH Interfaces SDH Interfaces PDH Traffic Connectors Local O&M	160 163 164 164 165
4 4.1 4.2 4.3 4.4 4.5 4.6	Physical Interfaces Ethernet Traffic Interfaces PDH Interfaces SDH Interfaces PDH Traffic Connectors Local O&M Antenna Interface	160 160 163 164 164 165 167
4 4.1 4.2 4.3 4.4 4.5 4.6 4.7	Physical Interfaces Ethernet Traffic Interfaces PDH Interfaces SDH Interfaces PDH Traffic Connectors Local O&M Antenna Interface Indoor-Outdoor Interconnection	160 163 164 164 165 167
 4 4.1 4.2 4.3 4.4 4.5 4.6 4.7 5 	Physical Interfaces Ethernet Traffic Interfaces PDH Interfaces SDH Interfaces PDH Traffic Connectors Local O&M Antenna Interface Indoor-Outdoor Interconnection	160 163 164 164 165 167 167 167
 4.1 4.2 4.3 4.4 4.5 4.6 4.7 5 5.1 	Physical InterfacesEthernet Traffic InterfacesPDH InterfacesSDH InterfacesPDH Traffic ConnectorsLocal O&MAntenna InterfaceIndoor-Outdoor InterconnectionPowerPower Supply	160 163 164 164 165 167 167 167 169
 4.1 4.2 4.3 4.4 4.5 4.6 4.7 5 5.1 5.2 	Physical InterfacesEthernet Traffic InterfacesPDH InterfacesSDH InterfacesPDH Traffic ConnectorsLocal O&MAntenna InterfaceIndoor-Outdoor InterconnectionPowerPower SupplyPower Line Disturbance	160 163 164 164 165 167 167 167 169 169
 4 4.1 4.2 4.3 4.4 4.5 4.6 4.7 5 5.1 5.2 5.3 	Physical InterfacesEthernet Traffic InterfacesPDH InterfacesSDH InterfacesPDH Traffic ConnectorsLocal O&MAntenna InterfaceIndoor-Outdoor InterconnectionPowerPower SupplyPower Line DisturbancePower Consumption	160 160 163 164 164 165 167 167 167 169 169 172
 4 4.1 4.2 4.3 4.4 4.5 4.6 4.7 5 5.1 5.2 5.3 6 	Physical InterfacesEthernet Traffic InterfacesPDH InterfacesSDH InterfacesPDH Traffic ConnectorsLocal O&MAntenna InterfaceIndoor-Outdoor InterconnectionPowerPower SupplyPower Line DisturbancePower ConsumptionEnvironment	160 160 163 164 164 165 167 167 167 169 169 172 172 172
 4 4.1 4.2 4.3 4.4 4.5 4.6 4.7 5 5.1 5.2 5.3 6 6.1 	Physical InterfacesEthernet Traffic InterfacesPDH InterfacesSDH InterfacesPDH Traffic ConnectorsLocal O&MAntenna InterfaceIndoor-Outdoor InterconnectionPowerPower SupplyPower Line DisturbancePower ConsumptionEnvironmentEnvironmental Conditions	160 160 163 164 164 165 167 167 167 169 169 172 172 172 172

7.1	Outdoor Units	183
7.2	Indoor Units	184
8	Standards	189
8.1	Radio Equipment Directive (RED)	189
8.2	ATEX Directive	189
8.3	RoHS Directive	189
8.4	Council Recommendation	189
8.5	RoHS	189
8.6	Normative References	190
8.7	Mechanics	190
8.8	Power	191
8.9	EMC	191
8.10	Safety/Health	191
8.11	Environmental and Climatic	194
8.12	PDH	195
8.13	SDH	196
8.14	Ethernet	197
8.15	Synchronization	199
8.16	Digital Radio Transmission	200
8.17	Operation & Maintenance Interfaces	205
8.18	Security	208
9	Definitions/Abbreviations	209



1 Introduction

MINI-LINK 6600 and MINI-LINK 6366 product family covers microwave nodes tailor made for different site needs spanning from simplest outdoor tail site to advanced split mounted aggregation site.

The following products are available:

- MINI-LINK 6651
- MINI-LINK 6691 with NPU 1002 or NPU1003 or NPU 1005
- MINI-LINK 6692 with NPU 1002 or NPU 1005
- MINI-LINK 6693 with NPU 1002 or NPU 1003 or NPU 1005
- MINI-LINK 6694 with NPU 1002 or NPU 1003 or NPU 1005
- MINI-LINK 6654 with PNM 1001 or PNM 1002
- MINI-LINK 6655 with PNM 1002
- MINI-LINK 6366
- MINI-LINK 6371

MINI-LINK 6651 is a compact 1RU node for tail sites.

MINI-LINK 6651/1 has a switch capacity of 14.5 Gbps and can connect one or two split mount radio links in one direction. It has 1 Gbps and 2.5 Gbps Ethernet interfaces on the front. It has forced cooling (fan unit).

MINI-LINK 6651/2 has a switch capacity of 14.5 Gbps and can connect one or two split mount radio links in two directions. It has 1 Gbps and 2.5 Gbps Ethernet interfaces on the front. It has forced cooling (fan unit).

MINI-LINK 6651/3 has a switch capacity of 14.5 Gbps and can connect one split mount radio link in one direction. It has 1 Gbps and 2.5 Gbps Ethernet interfaces on the front. It has convectional cooling (no fan).

MINI-LINK 6651/4 has a switch capacity of 47 Gbps and can connect one or two split mount radio links in two directions. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front. Hierarchical Radio Link Bonding is supported. It has forced cooling (fan unit).

In this document, the notation MINI LINK 6651 will be used for generic specifications and parameters. The explicit notation MINI-LINK 6651/1, MINI-LINK 6651/2, MINI-LINK 6651/3 and MINI-LINK 6651/4 will be used where there are differences in the product versions.

MINI-LINK 6691 with NPU 1002 is a small flexible 1RU node for small aggregation hubs up to 4 radio links in a maximum of 4 radio directions. It has a switch capacity of 88 Gbps, where 38 Gbps can be used in this enclosure. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front.

MINI-LINK 6691 with NPU 1003 is a small flexible 1RU node for small aggregation hubs up to 4 radio links in a maximum of 4 radio directions. It has a switch capacity of 45.5 Gbps, where 38 Gbps can be used in this enclosure. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front.

MINI-LINK 6691 with NPU 1005 is a small flexible 1RU node for small aggregation hubs up to 4 radio links in a maximum of 4 radio directions. It has a switch capacity of 65.5 Gbps, where 43 Gbps can be used in this enclosure. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front. Hierarchical Radio Link Bonding is supported.

MINI-LINK 6692 with NPU 1002 is a flexible 3RU node for large aggregation hubs up to 16 radio links in a maximum of 15 radio directions. It has a switch capacity of 88 Gbps, where 7.5 Gbps is reserved for future use. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front. MINI-LINK 6692 supports redundant NPUs and PFUs.

MINI-LINK 6692 with NPU 1005 is a flexible 3RU node for large aggregation hubs up to 16 radio links in a maximum of 11 radio directions. It has a switch capacity of 65.5 Gbps, where the full capacity can be used in this enclosure. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front. MINI-LINK 6692 supports redundant NPUs and PFUs. Hierarchical Radio Link Bonding is supported.

MINI-LINK 6693 with NPU 1002 is a flexible 1.5RU node for medium aggregation hubs up to 8 radio links in a maximum of 8 radio directions. It has a switch capacity of 88 Gbps, where 46.5 Gbps can be used in this enclosure and has 1 Gbps and 10 Gbps Ethernet interfaces on the front.

MINI-LINK 6693 with NPU 1003 is a flexible 1.5RU node for medium aggregation hubs up to 8 radio links in a maximum of 7 radio directions. It has a switch capacity of 45.5 Gbps, and the full capacity can be used in this enclosure. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front.

MINI-LINK 6693 with NPU 1005 is a flexible 1.5RU node for medium aggregation hubs up to 8 radio links in a maximum of 7 radio directions. It has a switch capacity of 65.5 Gbps, where 55.5 Gbps can be used in this enclosure. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front. Hierarchical Radio Link Bonding is supported.

MINI-LINK 6694 with NPU 1002 is a flexible 2RU node for medium aggregation hubs up to 8 radio links in a maximum of 8 radio directions. It has a switch capacity of 88 Gbps where 59.5 Gbps can be used in this enclosure. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front. MINI-LINK 6694 supports redundant NPUs and PFUs.

MINI-LINK 6694 with NPU 1003 is a flexible 2RU node for medium aggregation hubs up to 8 radio links in a maximum of 7 radio directions. It has a switch capacity of 45.5 Gbps and the full capacity can be used in this enclosure. It has 1



Gbps and 10 Gbps Ethernet interfaces on the front. MINI-LINK 6694 supports redundant PFUs. Redundant NPU with NPU 1003 is not possible.

MINI-LINK 6694 with NPU 1005 is a flexible 2RU node for medium aggregation hubs up to 8 radio links in a maximum of 7 radio directions. It has a switch capacity of 65.5 Gbps where 55.5 can be used in this enclosure. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front. MINI-LINK 6694 supports redundant NPUs and PFUs. Hierarchical Radio Link Bonding is supported.

MINI-LINK 6654 with PNM 1001 is a small flexible 1RU node for medium aggregation hubs up to 5 radio links in a maximum of 5 radio directions. It has a switch capacity of 36.5 Gbps, and the full capacity can be used in this enclosure. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front.

MINI-LINK 6654 with PNM 1002 is a small flexible 1RU node for medium aggregation hubs up to 6 radio links in a maximum of 6 radio directions. It has a switch capacity of 39 Gbps, and the full capacity can be used in this enclosure. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front.

MINI-LINK 6655 with PNM 1002 is a small flexible 1.5RU node for medium aggregation hubs up to 10 radio links in a maximum of 8 radio directions. It has a switch capacity of 44 Gbps, and the full capacity can be used in this enclosure. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front.

MINI-LINK 6366 is an all-outdoor compact node. It has a switch capacity of 9.5 Gbps and can connect one or two radio units. It has 1 Gbps and 2.5 Gbps Ethernet interfaces on the front. In this document, the notation MINI LINK 6366 will be used for generic specifications and parameters. The explicit notation MINI-LINK 6366/1 and MINI-LINK 6366/4 will be used where there are differences in the product versions.

MINI-LINK 6371 is an all-outdoor compact node. It has a switch capacity of 36 Gbps and can connect one or two radio units, in two directions. It has 1 Gbps and 10 Gbps Ethernet interfaces on the front. Hierarchical Radio Link Bonding is supported.

1.1 Scope

This product specification defines the basic conditions, characteristics and performance of the MINI-LINK 6600, MINI-LINK 6366, and MINI-LINK 6371 Network Elements.

For more information about supported features please see

47/221 02-HRA 901 20/11 Technical Description MINI-LINK 6600,

49/221 02-HRA 901 20/13 Technical Description MINI-LINK 6366, and

4/221 02-HRA 901 20/18 Technical Description MINI-LINK 6300.

2 Ethernet Characteristics

For more information about supported features please see

47/221 02-HRA 901 20/11 Technical Description MINI-LINK 6600 and 49/221 02-HRA 901 20/13 Technical Description MINI-LINK 6366.

For relations between Frame ID and corresponding Channel Separation support, see also Channel Separation to Frame ID Relation on page 73.

2.1 Traffic Capacity

Capacity characteristics for Ethernet depend on hardware configuration, selected physical mode and assigned Ethernet traffic capacity. Measured values are typical, not guaranteed.

Measurements are done according to RFC 2544.

2.1.1 Layer 1 Line Interface Capacity for ETSI

Data for standard Frame ID's (256, 257, 258, 259, 260, 261, and 303) can be found in revision BT of this document.

Physical m	node		Layer 1 Line Interface Capacity [Mbps]							
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]								
		Air Interface Capacity [Mbps]	64	128	256	512	1024	1280	1518	9216
	4QAM S	8	10.8	9.8	8.9	8.8	8.8	8.8	8.8	8.8
	4QAM	10	12.7	11.5	10.1	10.3	10.4	10.3	10.3	10.3
	16QAM S	18	22.6	19.9	18.4	18.2	17.9	17.6	17.6	17.7
	16QAM	21	26.4	23.2	20.4	21.3	21	21	20.9	20.7
7	32QAM	26	33.1	29.6	27.8	26.9	26.4	26.3	26.2	26.1
(356)	64QAM	33	41.6	37.1	34.9	33.7	33.1	33	33	32.7
	128QAM	39	49.2	44	41.3	40	39.2	39.1	39	38.7
	256QAM	45	56.8	50.8	47.6	46.1	45.3	45.1	45	44.7
	512QAM	51	64.4	57.5	54	52.3	51.3	51.2	51	50.6
	1024QAM	56	71.2	63.7	59.8	57.8	56.8	56.6	56.5	56

Table 1 Layer 1 Line Interface Capacity for ETSI



Physical n	node		Layer 1 Line Interface Capacity [Mbps]									
CS [MHz] (Frame ID)	Modulation	Frame size [bytes] Air Interface Capacity [Mbps]	64	128	256	512	1024	1280	1518	9216		
	4QAM S	19	23.3	20.8	19.5	18.9	18.6	18.5	18.4	18.3		
	4QAM	22	27.2	24.3	22.9	22.1	2.7	21.6	21.6	21.4		
	16QAM S	37	47	42	39.4	38.1	37.5	37.3	37.2	37		
	16QAM	43	54.3	49.1	46.1	44.6	43.8	43.6	43.5	43.2		
	32QAM	54	69	61.7	58	56	55	54.8	54.7	54.3		
13.75 14	64QAM	68	86.4	77.3	72.7	70.2	69	68.7	68.6	68		
(357)	128QAM	81	102.3	91.5	86	83	81.6	81.3	81.1	80.5		
	256QAM	93	118.1	105.6	99.2	95.9	94.2	93.9	93.7	92.9		
	512QAM	105	133.8	119.7	112.5	108.7	106.8	106.4	106.2	105.3		
	1024QAM	117	148.2	132.5	124.5	120.3	118.2	117.8	117.5	116.6		
	2048QAM	121	154	137.8	129.5	125.3	123	122.6	122.4	121.2		
	2048QAM L	127	161.9	144.8	136.1	131.5	129.2	128.7	128.5	127.4		
	4QAM S	37	47.1	42.1	39.5	38.3	37.6	37.5	37.4	37.1		
	4QAM	44	55	49.2	46.2	44.7	43.9	43.7	43.7	43.3		
	16QAM S	75	94.6	84.7	79.4	76.8	75.5	75.2	75.1	74.5		
	16QAM	87	110.4	98.8	92.9	89.7	88.1	87.8	87.6	86.9		
	32QAM	109	138.7	124.1	116.6	112.8	110.8	110.4	110.2	109.1		
	64QAM	137	173.8	155.5	146.1	141.2	138.6	138.2	137.9	136.8		
27.5	128QAM	162	205.4	183.8	172.8	167.1	164	163.4	163	161.7		
28	256QAM	186	237.2	212.2	199.4	192.9	189.5	188.8	188.4	186.7		
(358)	512QAM	211	268.9	240.5	226	218.6	214.8	214.1	213.6	211.6		
	1024QAM	234	297.5	266.2	250.1	242	237.6	236.9	236.4	234.2		
	2048QAM	243	309.4	276.7	260	251.6	247.1	246.3	245.8	243.5		
	2048QAM L	256	325.2	290.9	273.4	264.5	259.7	258.9	258.4	255.9		
	4096QAM	264	336	300.6	282.4	273.3	268.4	267.5	267	264.3		
	4096QAM L	276	351.9	314.8	295.8	286.1	281	280.1	279.5	276.9		
	8192QAM	284	361.1	323	303.5	293.7	288.7	287.7	287.1	284.4		

Physical mode			Layer 1 Line Interface Capacity [Mbps]									
CS [MHz] (Frame ID)	Modulation	Frame size [bytes] Air Interface Capacity [Mbps]	64	128	256	512	1024	1280	1518	9216		
	4QAM S	53	67.5	60.4	56.7	54.9	53.9	53.7	53.6	53.1		
	4QAM	62	78.8	70.5	66.2	64.1	62.9	62.7	62.6	62		
	16QAM S	107	135.4	121.1	113.8	110.1	108.1	107.8	107.6	106.5		
	16QAM	124	158	141.4	132.8	128.5	126.2	125.8	125.5	124.4		
	32QAM	156	198.3	177.4	166.7	161.3	158.4	157.9	157.5	156.1		
	64QAM	196	248.6	222.4	208.9	202.1	198.5	197.9	197.5	195.6		
	128QAM	231	293.8	262.9	247	239	234.7	233.9	233.5	231.3		
40	256QAM	267	339.1	303.4	285.1	275.8	270.9	270	269.4	266.9		
(337)	512QAM	302	384.4	343.9	323.1	312.6	307	306	305.4	302.5		
	1024QAM	335	425.4	380.4	357.6	345.9	339.8	338.6	338	334.8		
	2048QAM	348	441.8	395.2	371.3	359.3	352.9	351.7	351	347.7		
	2048QAM L	366	464.9	415.9	390.8	378.1	371.3	370.1	369.4	365.9		
	4096QAM	378	480.3	429.7	403.7	390.6	383.7	382.4	381.6	378		
	4096QAM L	395	499.9	450	422.8	409	401.7	400.4	399.6	395.8		
	8192QAM	406	516.6	462.1	434.3	420.2	413.1	411.7	410.8	406.6		
	4QAM S	75	95.7	85.6	80.4	77.8	76.4	76.2	76	75.3		
	4QAM	88	111.7	99.9	93.9	90.8	89.2	88.9	88.7	87.9		
	16QAM S	151	191.8	171.6	161.2	156	153.2	152.7	152.1	150.9		
	16QAM	176	223.8	199.7	188.1	182	178.6	177.9	177.6	176.1		
	32QAM	221	280.8	251.2	236	228.3	224.3	223.5	223.1	221		
	64QAM	277	351.9	314.8	295.8	286.2	281.1	280.2	279.6	277		
55 56	128QAM	327	416	372.1	349.7	338.3	332.3	331.2	330.5	327.4		
62.5 (360)	256QAM	377	480	429.4	403.5	390.4	383.4	382.2	381.4	377.8		
(300)	512QAM	427	544	486.6	457.3	442.4	434.5	433.1	432.2	428.1		
	1024QAM	473	601.8	538.4	505.9	489.5	480.7	479.2	478.2	473.7		
	2048QAM	492	625.7	559.8	526	508.9	499.8	498.2	497.2	492.5		
-	2048QAM L	517	657.7	588.4	552.9	534.9	525.3	523.6	522.6	517.6		
	4096QAM	534	679.5	607.9	571.2	552.6	542.8	541	539.9	534.8		
	4096QAM L	559	711.5	636.5	598.1	578.7	568.3	566.5	565.3	560		



Physical n	node		Layer 1 Line Interface Capacity [Mbps]								
CS [MHz] (Frame ID)	Modulation	Frame size [bytes] Air Interface Capacity [Mbps]	64	128	256	512	1024	1280	1518	9216	
	4QAM S	108	137.5	123	115.6	111.8	109.8	109.5	109.2	108.2	
	4QAM	126	160.5	143.6	134.9	130.5	128.2	127.8	127.5	126.3	
	16QAM S	217	275.4	246.4	231.5	224	220	219.3	218.8	216.7	
	16QAM	253	321.4	287.5	270.2	261.4	256.7	255.9	255.3	252.9	
	32QAM	317	403.1	360.7	338.9	327.9	322	321	320.3	317.3	
	64QAM	397	504.4	451.2	423.9	410.1	402.8	401.5	400.7	396.9	
80	128QAM	469	597.2	534.3	502	485.7	477	475.5	474.5	470.1	
(403)	256QAM	542	689.2	616.5	579.4	560.5	550.5	548.7	547.6	542.4	
	512QAM	614	781.1	698.8	656.7	635.3	623.9	621.9	620.6	614.8	
	1024QAM	679	863.4	772.2	725.8	702.1	689.6	687.3	685.9	679.5	
	2048QAM	706	898.6	803.6	755.4	730.8	717.8	715.5	714	707.2	
	2048QAM L	742	943.2	843.8	792.9	767.1	753.4	750.9	749.4	742.3	
	4096QAM	767	973.5	870.8	818.3	791.6	777.5	774.9	773.4	766.1	
	4096QAM L	803	1020.9	913.2	858.2	830.3	816.3	813.5	811.7	803.9	
	4QAM S	151	191.6	171.4	161	155.8	153	152.5	152.2	150.8	
	4QAM	176	223.6	200	187.9	181.8	178.6	178	177.6	175.9	
	16QAM S	302	383.5	343.1	322.4	311.9	306.4	305.4	304.7	301.8	
	16QAM	352	447.5	400.4	376.2	364	357.5	356.3	355.6	352.2	
	32QAM	441	561.3	502.2	471.9	456.5	448.4	446.9	446	441.8	
110	64QAM	553	703.5	629.4	591.4	572.1	561.9	560.1	559	553.7	
125	128QAM	654	831.5	743.8	699	676.2	664.2	662	660.7	654.4	
(361 404)	256QAM	754	956.7	855.9	804.3	778	764.2	761.7	760	753	
	512QAM	855	1085	966.5	908.2	878.7	862.9	860	858.5	850.3	
	1024QAM	946	1201	1068	1009	974.8	957.7	952.6	952.2	943.3	
	2048QAM	983	1248	1110	1048	1012	994.8	990.8	990.1	980.2	
	2048QAM L	1033	1311	1167	1102	1063	1043	1041	1040	1030	
	4096QAM	1068	1356	1206	1139	1100	1078	1076	1075	1065	

2.1.2 Layer 1 Line Interface Capacity ANSI

Data for standard Frame ID's (262, 263, 264, 265, 266, 267, and 268) can be found in revision BT of this document.

Physical n	node		Layer 1 Line Interface Capacity [Mbps]								
CS		Frame size [bytes]									
Frame ID	Modulation	Air Interface Capacity [Mbps]	64	128	256	512	1024	1280	1518	9216	
	4QAM S	13	16	14.1	13.2	12.8	12.6	12.5	12.5	12.4	
	4QAM	15	18.4	16.5	15.5	15	14.7	14.7	14.6	14.5	
	16QAM S	25	31.8	28.4	26.7	25.8	25.4	25.3	25.2	25	
	16QAM	29	37.1	33.2	31.2	30.2	29.7	29.6	29.5	29.2	
10	32QAM	37	46.7	41.7	39.2	37.9	37.3	37.1	37.1	36.7	
(362)	64QAM	46	58.5	52.4	49.2	47.6	46.7	46.6	46.5	46.1	
	128QAM	55	69.2	61.9	58.2	56.3	55.3	55.1	55	54.5	
	256QAM	63	79.9	71.5	67.2	65	63.8	63.6	63.5	62.9	
	512QAM	71	90.6	81.1	76.2	73.7	72.4	72.2	72	71.3	
	1024QAM	79	110.3	89.7	84.3	81.6	80.1	79.9	79.7	78.9	
	4QAM S	26	33.1	29.6	27.8	26.9	26.4	26.4	26.3	26.1	
	4QAM	31	38.7	34.6	32.5	31.5	30.9	30.8	30.7	30.4	
	16QAM S	53	66.6	59.6	56	54.2	53.2	53	52.9	52.4	
	16QAM	61	77.8	69.6	65.4	63.3	62.1	61.9	61.8	61.2	
	32QAM	77	97.7	87.4	82.1	79.4	78	77.8	77.6	76.9	
20	64QAM	97	122.5	109.6	103	99.6	97.8	97.5	97.3	96.4	
(363)	128QAM	114	144.8	129.6	121.7	117.8	115.7	115.3	115.1	114	
	256QAM	132	176.2	149.6	140.5	136	133.5	133.1	132.8	131.6	
-	512QAM	149	189.5	169.5	159.3	154.1	151.4	150.9	150.6	149.2	
	1024QAM	165	209.7	187.6	176.3	170.6	167.5	167	166.6	165.1	
	2048QAM	172	218.1	195.1	183.3	177.3	174.2	173.6	173.3	171.6	
	2048QAM L	180	229.3	205.1	192.7	186.4	183.1	182.5	182.1	180.4	

Table 2 Layer 1 Line Interface Capacity ANSI



Physical n	node		Layer 1 Line Interface Capacity [Mbps]										
CS [MHz] Frame ID	Modulation	Frame size [bytes] Air Interface Capacity	64	128	256	512	1024	1280	1518	9216			
		[Mbps]											
	4QAM S	37	47.1	42.1	39.5	38.3	37.6	37.5	37.4	37.1			
	4QAM	44	55	49.2	46.2	44.7	43.9	43.7	43.7	43.3			
	16QAM S	75	94.6	84.7	79.4	76.8	75.5	75.2	75.1	74.5			
	16QAM	87	110.4	98.8	92.9	89.7	88.1	87.8	87.6	86.9			
	32QAM	109	138.7	124.1	116.6	112.8	110.8	110.4	110.2	109.1			
	64QAM	137	173.8	155.5	146.1	141.2	138.6	138.2	137.9	136.8			
28	128QAM	162	205.4	183.8	172.8	167.1	164	163.4	163	161.7			
(371)	256QAM	186	237.2	212.2	199.4	192.9	189.5	188.8	188.4	186.7			
	512QAM	211	268.9	240.5	226	218.6	214.8	214.1	213.6	211.6			
	1024QAM	234	297.5	266.2	250.1	242	237.6	236.9	236.4	234.2			
	2048QAM	243	309.4	276.7	260	251.6	247.1	246.3	245.8	243.5			
	2048QAM L	256	325.2	290.9	273.4	264.5	259.7	258.9	258.4	255.9			
	4096QAM	264	336	300.6	282.4	273.3	268.4	267.5	267	264.3			
	4096QAM L	276	351.9	314.8	295.8	286.1	281	280.1	279.5	276.9			
	4QAM S	40	50.4	45.1	42.4	41	40.3	40.1	40.1	39.7			
	4QAM	47	58.9	52.7	49.5	47.9	47	46.9	46.8	46.4			
	16QAM S	80	101.3	90.6	85.1	82.4	80.9	80.6	80.5	79.7			
	16QAM	93	118.3	105.8	99.4	96.2	94.4	94.1	94	93.1			
	32QAM	117	148.4	132.8	124.8	120.7	118.6	118.2	117.9	116.8			
	64QAM	147	186.1	166.5	156.4	151.3	148.6	148.1	147.8	146.4			
70	128QAM	173	220	196.8	184.9	178.9	175.7	175.1	174.8	173.1			
(364)	256QAM	200	251	224.6	211	204.1	200.5	199.8	199.4	197.6			
	512QAM	226	287.8	257.5	241.9	234.1	229.9	229.1	228.7	226.5			
-	1024QAM	251	318.5	284.9	267.7	259	254.4	253.6	253.1	250.7			
	2048QAM	261	331.2	296.2	278.4	269.3	264.5	263.6	263.1	260.6			
	2048QAM L	274	348.1	311.4	292.6	283.1	278.1	277.1	276.6	274			
_	4096QAM	283	359.7	321.8	302.4	292.5	287.3	286.4	285.8	283.1			
	4096QAM L	296	376.6	336.9	316.6	306.3	300.8	299.9	299.3	296.4			
	8192QAM	304	387	346.1	325.3	314.7	309.4	308.3	307.7	304.7			

Physical mode			Layer 1 Line Interface Capacity [Mbps]									
CS [MHz] Frame ID	Modulation	Frame size [bytes] Air Interface Capacity [Mbps]	64	128	256	512	1024	1280	1518	9216		
	4QAM S	54	68	60.9	57.2	55.3	54.3	54.2	54	53.5		
	4QAM	63	79.4	71.1	66.8	64.6	63.4	63.2	63.1	62.5		
	16QAM S	108	136.5	122.1	114.7	111	109	108.7	108.4	107.4		
	16QAM	126	159.3	142.5	133.9	129.6	127.3	126.8	126.6	125.4		
	32QAM	157	199.9	178.8	168.1	162.5	159.7	159.2	158.8	157.3		
	64QAM	197	250.6	224.2	210.7	203.8	200.2	199.5	199.1	197.2		
	128QAM	233	296.2	265	249	240.9	236.6	235.9	235.4	233.2		
40 (365)	256QAM	269	341.9	305.9	287.4	278.1	273.1	272.2	271.6	269.1		
(303)	512QAM	305	387.6	346.7	325.8	315.2	309.6	308.5	307.9	305		
	1024QAM	337	428.9	383.6	360.5	348.8	342.5	341.4	340.7	337.5		
	2048QAM	351	445.9	398.9	374.8	362.6	356.1	355	354.3	350.9		
	2048QAM L	369	468.7	419.3	394	381.2	374.4	373.2	372.4	368.9		
	4096QAM	381	484.3	433.2	407.1	393.8	386.8	385.6	384.8	381.1		
	4096QAM L	399	504.4	451.2	423.9	410.1	402.8	401.5	400.7	396.9		
	8192QAM	409	520.9	465.9	437.9	423.6	416.5	415.1	414.2	410		
	4QAM S	68	85.5	76.5	71.8	69.6	68.3	68.1	68	67.3		
	4QAM	79	99.9	89.3	83.9	81.2	79.8	79.5	79.3	78.6		
	16QAM S	135	171.5	153.4	144.1	139.4	137	136.5	136.2	134.9		
	16QAM	158	200	179	168.2	162.7	159.8	159.3	159	157.5		
	32QAM	198	251	224.6	211	204.1	200.5	199.8	199.4	197.6		
	64QAM	248	314.7	281.5	264.5	255.9	251.3	250.5	250	247.7		
50	128QAM	293	372	332.8	312.7	302.5	297.1	296.1	295.5	292.7		
(366)	256QAM	338	429.2	384	360.8	349.1	342.9	341.7	341	337.8		
	512QAM	383	486.5	435.2	409	395.7	388.6	387.3	386.6	382.9		
	1024QAM	423	538.4	481.6	452.6	437.8	430	428.6	427.6	423.7		
	2048QAM	440	559.7	500.7	470.5	455.2	447.1	445.6	444.7	440.5		
	2048QAM L	463	588.4	526.3	494.6	478.5	470	468.4	467.5	463.1		
	4096QAM	478	607.9	543.8	511	494.4	485.6	484	483	478.4		
	4096QAM L	500	636.4	569.5	535.1	517.7	508.5	506.8	505.8	501		



Physical n	Physical mode			Layer 1 Line Interface Capacity [Mbps]									
CS [MH7]		Frame size [bytes]											
Frame ID	Modulation	Air Interface Capacity [Mbps]	64	128	256	512	1024	1280	1518	9216			
	4QAM S	75	95.7	85.6	80.4	77.8	76.4	76.2	76.0	75.3			
	4QAM	88	111.7	99.9	93.9	90.8	89.2	88.9	88.7	87.9			
	16QAM S	151	191.8	171.6	161.2	156	153.2	152.7	152.1	150.9			
	16QAM	176	223.8	199.7	188.1	182	178.6	177.9	177.6	176.1			
	32QAM	221	280.8	251.2	236	228.3	224.3	223.5	223.1	221			
	64QAM	277	351.9	314.8	295.8	286.2	281.1	280.2	279.6	277			
56	128QAM	327	416	372.1	349.7	338.3	332.3	331.2	330.5	327.4			
(373)	256QAM	377	480	429.4	403.5	390.4	383.4	382.2	381.4	377.8			
	512QAM	427	544	486.6	457.3	442.4	434.5	433.1	432.2	428.1			
	1024QAM	473	601.8	538.4	505.9	489.5	480.7	479.2	478.2	473.7			
	2048QAM	492	625.7	559.8	526	508.9	499.8	498.2	497.2	492.5			
	2048QAM L	517	657.7	588.4	552.9	534.9	525.3	523.6	522.6	517.6			
	4096QAM	534	679.5	607.9	571.2	552.6	542.8	541	539.9	534.8			
	4096QAM L	559	711.5	636.5	598.1	578.7	568.3	566.5	565.3	560			
	4QAM S	81	102.8	92	86.4	83.6	82.1	81.8	81.7	80.9			
	4QAM	95	120	107.4	100.9	97.6	95.9	95.5	95.4	94.5			
	16QAM S	162	206.1	184.4	173.2	167.6	164.6	164.1	163.7	162.2			
	16QAM	189	240.5	215.1	202.2	195.6	192.1	191.5	191.1	189.3			
	32QAM	237	301.7	269.9	253.6	245.4	241	240.2	239.7	237.5			
	64QAM	297	378.2	338.3	317.9	307.6	302.1	301.1	300.5	297.6			
60	128QAM	352	447	399.9	375.8	363.5	357.1	355.9	355.2	351.8			
(367)	256QAM	406	515.8	461.5	433.6	419.5	412	410.7	409.9	406			
	512QAM	460	584.7	523.1	491.5	475.5	467	465.5	464.6	460.2			
	1024QAM	509	647	578.8	543.9	526.2	516.8	515.1	514	509.2			
	2048QAM	529	672.7	601.8	565.4	547	537.3	535.5	534.5	529.4			
	2048QAM L	556	707.1	632.1	594.4	575	564.8	562.9	561.8	556.5			
	4096QAM	574	730.6	653.6	614.1	594.1	583.5	581.6	580.5	575			
	4096QAM L	601	759.2	679.1	638.1	617.4	606.4	604.4	603.2	597.5			

Physical mode			Layer 1 Line Interface Capacity [Mbps]									
CS [MHz]	Modulation	Frame size [bytes]	64	129	256	512	1024	1290	1519	0216		
Frame ID	Modulation	Air Interface Capacity [Mbps]	04	120	250	512	1024	1200	1219	9210		
	4QAM S	108	137.5	123	115.6	111.8	109.8	109.5	109.2	108.2		
	4QAM	126	160.5	143.6	134.9	130.5	128.2	127.8	127.5	126.3		
	16QAM S	217	275.4	246.4	231.5	224	220	219.3	218.8	216.7		
	16QAM	253	321.4	287.5	270.2	261.4	256.7	255.9	255.3	252.9		
	32QAM	317	403.1	360.7	338.9	327.9	322	321	320.3	317.3		
	64QAM	397	504.4	451.2	423.9	410.1	402.8	401.5	400.7	396.9		
80	128QAM	469	597.2	534.3	502	485.7	477	475.5	474.5	470.1		
(368)	256QAM	542	689.2	616.5	579.4	560.5	550.5	548.7	547.6	542.4		
	512QAM	614	781.1	698.8	656.7	635.3	623.9	621.9	620.6	614.8		
	1024QAM	679	863.4	772.2	725.8	702.1	689.6	687.3	685.9	679.5		
	2048QAM	706	898.6	803.6	755.4	730.8	717.8	715.5	714	707.2		
	2048QAM L	742	943.2	843.8	792.9	767.1	753.4	750.9	749.4	742.3		
	4096QAM	767	973.5	870.8	818.3	791.6	777.5	774.9	773.4	766.1		
	4096QAM L	803	1020	913.2	858.2	830.3	816.3	813.5	811.7	804		
	4QAM S	151	191.6	171.4	161	155.8	153	152.5	152.2	150.8		
	4QAM	176	223.6	200	187.9	181.8	178.6	178	177.6	175.9		
	16QAM S	302	383.5	343.1	322.4	311.9	306.4	305.4	304.7	301.8		
	16QAM	352	447.5	400.4	376.2	364	357.5	356.3	355.6	352.2		
	32QAM	441	561.3	502.2	471.9	456.5	448.4	446.9	446	441.8		
	64QAM	553	703.5	629.4	591.4	572.1	561.9	560.1	559	553.7		
112 (374)	128QAM	654	831.5	743.8	699	676.2	664.2	662	660.7	654.4		
	256QAM	754	956.7	855.9	804.3	778	764.2	761.7	760	753		
	512QAM	855	1085	966.5	908.2	878.7	862.9	860	858.5	850.3		
	1024QAM	946	1201	1068	1009	974.8	957.7	945.9	945.3	943.3		
	2048QAM	983	1248	1110	1048	1012	994.8	990.8	990.1	980.2		
	2048QAM L	1033	1311	1167	1102	1063	1043	1041	1040	1030		
	4096QAM	1068	1356	1206	1139	1100	1078	1076	1075	1065		

2.1.3 Layer 2 Line Interface Capacity

L2 = L1 x (packet size) / (packet size +20)



2.2 Traffic Latency

Latency characteristics for Ethernet depend on hardware configuration, selected physical mode and assigned Ethernet traffic capacity. Measured values are typical, not guaranteed.

Measurements are done according to RFC 2544.

2.2.1 Traffic Latency for ETSI

The table below shows latency values for low latency Frame ID's.

Values for standard Frame ID's (256, 257, 258, 259, 260, 261 and 303) can be found in revision BT of this document.

Physico	al mode			Latency [µs]	
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	766	750	840
	4QAM S	128	825	810	860
7 (356)		256	946	930	980
		512	1204	1170	1210
		1024	1639	1630	1670
		1280	1896	1860	1900
		1518	2115	2080	2120
		9216	9185	9160	9200
		64	752	730	920
		128	799	780	840
		256	896	880	930
7	404M	512	1101	1090	1120
(356)	4QAM	1024	1510	1480	1520
		1280	1716	1680	1720
		1518	1904	1880	1910
	-	9216	7917	7910	7940

Table 3Traffic Latency for ETSI

Physico	Physical mode		Latency [µs]					
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum			
		64	709	698	725			
		128	736	728	754			
		256	795	788	816			
7	1/0446	512	917	900	980			
(356)	I DUAM S	1024	1152	1140	1170			
		1280	1272	1260	1290			
		1518	1388	1370	1390			
		9216	4914	4900	4980			
	16QAM	64	698	690	713			
		128	722	716	738			
		256	776	770	790			
7 (356)		512	875	870	890			
		1024	1081	1070	1090			
		1280	1180	1170	1190			
		1518	1279	1273	1290			
		9216	4314	4290	4360			
		64	686	681	700			
		128	707	702	720			
		256	751	745	762			
7	720AM	512	832	827	840			
(356)	52QAM	1024	994	990	1002			
		1280	1076	1070	1080			
		1518	1157	1148	1163			
		9216	3580	3570	3630			
		64	680	673	688			
		128	694	690	705			
		256	730	720	740			
7	640AM	512	796	793	803			
(356)		1024	924	922	933			
		1280	992	980	1000			
		1518	1054	1051	1062			
	-	9216	3013	3000	3030			



Physico	ıl mode		Latency [µs]					
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum			
		64	673	669	682			
		128	686	683	696			
		256	716	713	724			
7	1200414	512	773	770	780			
(356)	128QAM	1024	886	881	891			
		1280	941	939	949			
		1518	995	992	1001			
		9216	2652	2640	2680			
		64	669	665	676			
	256QAM	128	681	678	689			
7		256	707	700	720			
		512	757	750	760			
(356)		1024	853	852	859			
		1280	902	900	920			
		1518	951	947	956			
		9216	2395	2380	2420			
		64	665	662	672			
		128	678	674	684			
		256	696	690	701			
7	E120AM	512	745	740	750			
(356)	SIZQAM	1024	829	828	836			
		1280	874	872	880			
		1518	917	915	922			
		9216	2198	2190	2220			
		64	664	661	670			
		128	674	671	680			
		256	695	692	701			
7	102404M	512	737	735	741			
(356)	τυζ4ŲΑΜ	1024	815	813	820			
		1280	854	852	859			
		1518	891	889	896			
		9216	2056	2051	2059			

Physico	al mode			Latency [µs]	
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	391	384	413
		128	423	410	480
		256	482	470	550
13.75	404145	512	597	580	660
(357)	4QAM S	1024	821	810	890
		1280	940	920	1000
		1518	1043	1030	1050
		9216	4482	4460	4540
	4QAM	64	391	370	460
		128	411	400	429
		256	460	450	510
13.75		512	559	550	610
14 (357)		1024	752	740	800
		1280	848	840	910
		1518	943	930	950
		9216	3869	3860	3930
		64	366	350	410
		128	379	373	402
		256	407	400	440
13.75	1/00145	512	466	460	500
(357)	I6QAM S	1024	581	570	610
		1280	642	630	670
		1518	699	690	704
		9216	2428	2420	2460
		64	358	353	382
		128	371	367	396
		256	398	390	420
13.75	160414	512	448	440	470
14 (357)		1024	548	540	570
		1280	600	590	620
		1518	648	642	655
		9216	2140	2130	2160


Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	353	349	378
		128	364	360	386
		256	384	380	400
13.75	720414	512	426	420	450
(357)	32QAM	Initiation Frame size [bytes] Average 04 353 128 364 256 384 256 384 128 364 256 384 1024 506 11280 549 1518 588 9216 1791 64 349 1280 549 1518 588 9216 1791 64 349 128 357 256 376 512 409 1024 475 1280 508 1518 539 9216 1510 64 346 1280 485 1518 510 9216 1342 64 355 1280 485 1518 510 9216 1342 64 355 128 <td>506</td> <td>505</td> <td>515</td>	506	505	515
	Physical modeFrame size (bytes)AuxS(MHz) (ame 1D)Modulation64313.75 14 (357)32QAM64313.75 	549	547	554	
		1518	588	585	593
		9216	1791	1780	1810
		64	349	Latency [µs] Minimum 349 360 380 420 505 547 585 1780 345 354 354 354 354 354 354 354	374
		128	357	354	373
	Hz] 5 (10) Modulation Frame size [bytes] 75 64 128 75 32QAM 1024 1024 1280 1518 9216 75 64QAM 128 75 64QAM 128 75 64QAM 1280 1024 1280 1518 9216 75 128QAM 64 1280 1518 9216 128 75 128QAM 128 75 128QAM 64 1280 1518 9216 1518 9216 1518 9216 1518 9216 75 128 128 75 128 128 75 512 512	256	376	374	380
13.75 14 (357)		512	409	407	424
		1024	475	472	493
		1280	508	506	528
		1518	539	536	544
		9216	1510	1508	1514
		64	346	343	367
		128	354	350	370
		256	369	Latency [µs] Minimum 349 360 380 420 505 547 585 1780 345 354 354 354 354 374 407 472 506 536 536 1508 343 350 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 343 350 368 350 368 350 368 350 368 350 368 350 368 350 368 350 368 350 368 350 368 350 368 350 368 350 368 350 368 350 350 368 350 350 350 368 350 350 350 350 350 350 350 350 350 350	374
13.75	1200414	512	398	397	414
(357)	128QAM	1024	454	452	467
		1280	485	482	490
		1518	510	508	515
		9216	1342	1337	1343
		64	355	340	370
		128	353	349	358
		256	365	360	370
13.75 14 (357) 13.75 14 (357) 13.75 14 (357)	2560 1 M	512	391	389	393
(357)	2JUQAM	1024	AverageLatency [µs]AverageMinimum355334936436038438042642042642058650554954758858535934535735435735440940747547250850653953653953635435035435044634335534035534013421337440439365360364343440439440439440439440439451482467464485482412181214	443	
		1280	467	464	469
		1518	490	488	492
		9216	1218	1214	1230

Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	354	Minimum 346 347 360 384 427 451 471 1120 346 345 362 379 419 442 458 1052 345 360 377 417 437 456 1027 344 356 376 413 4433 450 998	357
		128	349	347	355
		256	362	360	392
13.75	13.75 14 (357) 256 362 360 512QAM 512 385 384 1024 428 427 1280 454 451 1518 473 471 9216 1123 1120 13.75 64 352 346 1280 347 345 256 363 362 512 381 379 14(357) 1024QAM 512 381 379 18024QAM 1024 420 419 1280 443 442 1518 459 458 9216 1054 1052 13.75 14 350 345 1518 459 458 9216 1054 1052 13.75 1280 348 345 (357) 2048QAM 512 379 377 1280 439 437 1407 1280	388			
(357)	512QAM	1024	428	427	431
		1280	Average Minimum 64 354 346 1 128 349 347 1 256 362 360 1 512 385 384 1 1024 428 427 1 1280 454 451 1 1280 454 451 1 1280 454 451 1 1280 453 120 1 64 352 346 1 128 347 345 1 64 352 346 1 128 347 345 1 128 341 379 1 128 443 442 1 1280 443 442 1 1280 348 345 1 1280 348 345 1 1280 439 344 1 1280 439 <td>455</td>	455	
		1518		475	
		9216	1123	Kinimum 346 347 360 384 427 451 471 1120 346 345 362 379 419 442 458 1052 345 360 377 417 437 456 1027 344 356 376 413 4433 450	1134
		64	352	Minimum 346 347 360 384 427 451 471 1120 346 345 362 379 419 442 458 1052 345 360 377 417 437 456 1027 344 356 376 413 4433 356 376 413 458 360 377 417 437 456 1027 344 356 376 413 433 450 998	357
		128	347	345	355
		256	363	362	380
13.75	10040444	1280 454 451 1518 473 471 9216 1123 1120 9216 1123 1120 64 352 346 128 347 345 256 363 362 512 381 379 1024 420 419 1280 443 442 1518 459 458 9216 1054 1052 64 350 345 128 348 345 9216 1054 1052 128 348 345 256 361 360 512 379 377 1024 418 417 1280 439 437 1518 457 456	379	383	
(357)	1024QAM		419	423	
13.75 14 (357)		1280	443	442	445
		1518	459	458	461
		9216	1054	1052	1065
		64	350	345	354
		128	348	345	354
		256	361	Minimum 346 347 360 384 427 451 471 1120 346 345 362 379 419 442 458 1052 345 360 377 417 437 456 1027 344 356 376 413 433 450 998	378
13.75	20400414	512	379	377	383
(357)	2048QAM	1024	418	417	420
		1280	439	437	441
		1518	457	456	460
		9216	1030	1027	1035
		64	349	344	359
		128	346	344	351
		256	358	356	360
13.75 1024QAM 128 11.75 1024QAM 512 11.75 1024QAM 1024 11.8 1024 11280 11.8 9216 1128 11.3.75 64 1128 11.3.75 2048QAM 64 11.3.75 2048QAM 1024 11.3.75 2048QAM 1024 11.3.75 64 1128 11.3.75 2048QAM L 64 11.3.75 14 1280 11.3.75 14 2048QAM L 64 11.3.75 14 128 11.3.75 14 128 1128 11.3.75 14 128 1128 11.3.75 14 1280 1128 11.3.75 14 1024 1128 11.3.75 14 1024 11280 11.3.75 11.3.75 11.3.75 11.3.75 11.3.75 11.3.75 14 10.24 11.3.75	378	376	380		
(357)	2048QAM L	1024	415	413	417
		1280	435	433	437
		1518	452	450	454
		9216	1000	998	1010



Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	217	213	240
		128	232	228	255
		256	262	258	284
27.5	101115	512	322	318	345
(358)	4QAM S	nodeFrame size (bytes)AverageLaModulationAverageModulation642171281282321256262151232211024437112804981128029216229292162292112822511282251128225112822511282251128256212830311280440311280455112804551151850211518502115185021160AMS1280210160AMS1280351160AMS128208160AMS128208160AMS128208160AMS128208160AMS128208160AMS128208160AMS128208160AMS128208160AMS128208160AMS128208160AMS128208160AMS128208160AMS128208160AMS128208160AMS128256160AMS128208160AMS128256160AMS128160AMS	437	432	456
			492	518	
		1518	553	549	567
		9216	2292	2289	2312
		64	213	209	236
		128	225	222	249
	ModulationFrame size [bytes]AverageLattModulationAverageM128232112823212562621512322110244371128049811128253112804981128049811280498112802292112822511282251128225112825625211024403112804551111850211280210112812101160AMS512259160AMS512380151838011518380112802185112803511128035111280351115183801151838011518380115182082160AM21011512256222160AM12852160AM1282160AM1281160AM1282160AM1282160AM1282160AM1282160AM1282160AM1282	248	275		
27.5 28 (358)		512	303	300	327
		1024	403	398	424
		1280	455	450	477
		1518	502	498	519
		9216	1995	1992	2014
		64	202	200	226
		128	210	208	226
		256	Latency [µs] Average Minimum 217 213 232 228 262 258 322 318 437 432 437 432 498 492 2292 2289 213 209 2292 2289 213 209 225 222 252 248 303 300 403 398 455 450 502 498 1995 1992 202 200 202 200 210 208 2259 257 319 318 351 348 360 378 201 198 202 200 203 301 380 378 201 198 2020 220 203 <td>243</td>	243	
27.5	1604145	512	259	257	274
(358)	IOQAM S	1024	319	318	335
		1280	351	348	367
		1518	380	378	392
		9216	1285	1283	1301
		64	201	198	221
		128	208	205	232
		256	222	220	235
27.5	160AM	512	250	249	263
(358)	modeFrame size [bytes]Image: constant size [bytes]Lutered size [bytes]ModulationFrame size [bytes]AverageMini412823222128232321282562622210244374411280498441151855356921622922212822522128225241282252412822524128256331284033312804554415185024415185024415185022416QAM S642022416QAM S5122592416QAM S642013316QAM S642013416QAM S5122592416QAM S5122592416QAM S642013416QAM12803513516QAM1282082416QAM1282082416QAM1282082416QAM1282562222416QAM1283313516QAM1313313516QAM1313513516QAM1313513516QAM1283313516QAM131835635	301	316		
		1280	331	328	344
		1518	356	355	367
	9216	1141	1139	1155	

Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	198	196	220
		128	204	202	223
		256	224	210	240
27.5	720.444	512	239	238	265
(358)	52QAM	1024	282	281	293
		1280	rame size [bytes] Average Minimum 64 198 196 1 128 204 202 1 256 224 210 1 512 239 238 1 1024 282 281 1 1280 305 303 1 1280 305 303 1 1280 305 303 1 1280 305 303 1 1280 305 193 1 64 195 193 1 128 201 199 1 256 216 214 1 512 231 229 1 1024 266 265 1 1280 286 284 1 1518 303 301 1 256 221 207 1 128 199 198 1	316	
	ModulationFrame size [bytes]Average64198.128204.128204.512239.512239.1024282.1280305.1280305.1518326.9216967.128201.128201.128201.128201.128201.128201.128201.128216.1280286.1518303.9216829.1280AM1280AM1280AM1280AM1280AM1280AM1280AM1280AM1280AM1280AM1280AM1280AM1280AM1280AM1280AM1280AM128012801280128012801280128012801280	325	334		
		9216	967	Littericy (µs) Minimum 196 202 210 238 281 303 325 966 193 199 214 229 265 284 301 828 196 198 207 224 255 272 287 744 196 196 224 255 272 287 744 196 204 220 248 263 276 682	979
		64	195	193	211
		128	201	199	217
		256	216	214	237
27.5 28 64QA (358)	(40.1)4	512	231	229	252
(358)	64QAM	1024	266	265	274
(358)		1280	286	284	294
		1518	303	301	309
		9216	829	Latency [µs] Minimum 196 202 210 238 281 303 325 966 193 199 214 229 265 284 301 828 196 284 301 828 196 198 207 224 255 272 287 744 196 198 207 224 255 272 287 744 196 198 207 224 255 272 287 744 196 198 207 224 255 272 287 744 196 196 198 207 224 255 272 287 744 196 196 196 198 207 224 255 272 287 744 196 196 196 198 207 224 255 272 287 744 196 196 196 207 224 255 272 287 744 196 196 196 207 224 255 272 287 744 196 196 207 224 255 272 287 744 196 196 207 227 287 744 196 196 207 272 287 744 196 196 207 272 287 744 196 196 207 272 287 744 196 196 207 272 287 744 196 196 207 272 287 744 196 196 207 272 287 744 196 196 207 272 287 744 196 196 207 272 287 744 196 196 207 272 287 744 196 196 207 272 287 744 196 196 207 272 287 744 196 207 272 287 744 196 207 272 287 744 196 207 272 287 744 196 207 272 272 287 744 207 272 287 744 196 207 272 272 287 744 196 204 220 276 276 276 276 276 276 276 276	838
		64	202	196	220
		128	199	198	213
		256	221	ge Minimum 196 202 210 210 210 210 210 210 238 281 303 325 966 193 199 214 229 265 284 301 828 301 828 196 198 207 224 255 272 287 196 198 207 224 255 272 287 196 196 207 224 255 272 287 287 287 287 284 204 <td< td=""><td>226</td></td<>	226
27.5	1200 4 14	512	225	224	242
(358)	128QAM	1024	256	255	263
		1280	273	272	280
		1518	288	287	294
		9216	745	744	752
		64	205	196	217
		128	198	196	210
		256	216	204	222
27.5	28 (358) 32QAM 1024 282 281 1280 305 303 1518 326 325 9216 967 966 9216 967 966 9216 967 966 9216 967 966 9216 967 966 128 201 199 256 216 214 512 231 229 1024 266 265 1280 286 284 1518 303 301 9216 829 828 1518 1024 266 255 1280 128 199 198 27.5 28 1024 256 221 207 512 221 224 255 224 1024 256 255 1518 288 287 9216 745 744 265 196	220	236		
∠8 (358)	∠50ŲAM	1024	249	248	254
		1280	265	263	270
		1518	278	276	282
		9216	684	682	691



Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	204	199	213
		128	197	196	208
		256	214	203	219
27.5	5120AM	512	219	218	231
(358)	512QAM	1024	Frame size [bytes]InterpretainAverageMinimum6420419912819719625621420351221921810242442421280258256151827126915182712699216637636642021981282162155122162165122162151024239238112826426212802642629216603662642021981280216216128021620212802162161280216216128025119812802512161280261216128021126112812021981284202198128420221612842022161284211214128421121412842362351284249248128424924812842592581284259258128425925812842592581284259258128425925812842592581284	248	
		1280		262	
		1518	271	269	275
		9216	637	636	643
		64	202	198	211
		128	203	198	213
	1024QAM	256	211	202	215
27.5 28 (358)		512	216	215	228
		1024	239	238	243
		1280	253	251	256
		1518	264	262	267
		9216	603	Latency [µs] Minimum 199 196 203 218 242 256 269 636 198 262 269 636 198 202 215 238 251 262 602 198 198 202 215 237 250 261 589 197 198 202 215 237 250 261 589 197 198 201 248 258 575	608
		64	202	198	210
		128	202	198	212
		256	Average Minimum 64 204 199 128 197 196 256 214 203 512 219 218 1024 244 242 1280 258 256 1518 271 269 9216 637 636 64 202 198 128 211 202 9216 637 636 64 202 198 128 203 198 256 211 202 512 216 215 1024 239 238 1280 253 251 1518 264 262 9216 603 602 64 202 198 128 202 198 256 211 202 1024 238 237 1280 261 197	215	
27.5	20400414	512	216	215	227
(358)	2048QAM	1024	238	237	242
		1280	251	250	255
		1518	262	Latency [µs] Minimum 199 196 203 218 242 256 269 636 198 202 215 238 251 262 602 198 251 262 602 198 251 253 251 252 602 198 251 251 251 251 251 251 251 262 602 198 292 215 237 250 261 589 197 198 201 214 235 248 258	265
		9216	591	589	596
		64	201	197	209
		128	202	198	211
		256	211	201	214
27.5	204904141	512	215	214	225
(358)	2048QAM L	1024	236	235	240
	Physical mode MeliD)Frame size [bytes]Image: [bytes]Image: [bytes]MdulationFrame size [bytes]AverageImage: [bytes]128197Image: [bytes]128197Image: [bytes]5120AM64204Image: [bytes]5120AM512219Image: [bytes]5120AM1024244Image: [bytes]5120AM1024244Image: [bytes]512219Image: [bytes]Image: [bytes]512219Image: [bytes]Image: [bytes]7.564202Image: [bytes]512216Image: [bytes]Image: [bytes]7.51024QAM512216Image: [bytes]5122216211Image: [bytes]Image: [bytes]7.51024QAM64202Image: [bytes]7.52048QAM64202Image: [bytes]7.52048QAM128201Image: [bytes]7.52048QAM64201Image: [bytes]7.52048QAM64201Image: [bytes]7.52048QAM128202Image: [bytes]7.52048QAM128202Image: [bytes]7.52048QAM128202Image: [bytes]7.52048QAM128201Image: [bytes]7.52048QAM128202Image: [bytes]7.52048QAM128201Image: [bytes]7.52048QAM<	248	253		
		rationFrame size (bytes)AverageAtt1281971128197125621415122191102424411280258115182711921663711282031128203125621115122161128203112820311282031128203112820311282031128203115182641151826411518264112820211282011128201112820111282021128201112820211282011128202112820111282021128202112820111282021128211112821111282111128215112821111282111128211112821111282111128<	258	260	
		9216	576	575	581

Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
	Modulation Frame 4096QAM	64	201	197	208
		128	203	197	210
		256	210	201	213
27.5	400/0414	512	214	213	224
(358)	4096QAM	1024	235	234	238
		Frame size [bytes] Average Minimum 64 201 197 1 128 203 197 1 256 210 201 1 512 214 213 1 1024 235 234 1 1280 247 246 1 1280 247 246 1 1280 247 246 1 9216 566 565 1 64 200 197 1 256 208 202 1 128 204 197 1 1024 234 233 1 1280 246 245 1 1280 246 254 1 1280 265 198 1 1280 203 198 1 1280 214 213 1 1280 245 244 1	251		
	Modulation Frame size [bytes] Average Minimum 4096QAM 64 201 197 128 203 197 256 210 201 512 214 213 1024 235 234 1280 247 246 1518 258 256 9216 566 565 9216 566 565 9216 566 565 9216 208 202 512 213 212 1024 234 233 1280 246 245 1512 213 212 1024 234 233 1280 246 245 9216 554 553 1518 256 210 256 210 202 512 214 213 1280 245 244 1518 255 254<	260			
		9216	566	Lutericy [µs] Minimum 197 197 201 213 234 246 256 565 197 197 202 212 233 245 254 553 197 198 202 213 233 245 254 553 197 198 202 213 234 254 553 197 198 202 213 234 254 546 159 170 192 235 318 360 400 1628	571
		64	200	197	208
		128	204	197	209
		256	208	202	212
27.5 28 (358)	400(04041	512	213	212	223
	4090QAM L	1024	234	233	237
		1280	246	245	249
		1518	256	254	258
		9216	554	Latency [µs] Minimum 197 201 213 234 246 256 565 197 197 202 212 233 245 254 553 197 198 202 213 245 254 553 197 198 202 213 245 254 553 197 198 202 213 245 254 553 197 198 202 213 245 254 553 197 198 202 213 245 254 553 197 198 202 213 245 254 553 197 198 202 213 233 244 254 553 197 198 202 213 233 245 254 553 197 198 202 213 235 318 360 400 1628	559
		64	200	197	208
		128	203	198	210
		256	210	Minimum 197 201 213 234 246 256 565 197 201 234 246 256 565 197 202 212 233 245 254 553 197 198 202 213 235 197 198 202 213 235 318 360 400 1628	214
27.5	0102/0414	512	214	213	223
(358)	81926QAM	1024	234	233	237
		1280	245	244	249
		1518	255	254	257
		9216	548	546	552
		64	162	159	186
		128	173	170	195
		256	195	192	217
27.5 28 (358) 4096QAM L 512 213 212 1024 234 233 1280 246 245 1518 256 254 9216 554 553 9216 554 553 28 (358) 64 200 197 128 203 198 256 210 202 27.5 28 (358) 81926QAM 512 214 213 1024 234 233 198 256 210 202 512 214 213 1024 234 233 1024 234 233 128 255 254 9216 548 546 546 546 9216 548 546 159 192 40 162 159 192 192 40 512 234 235 1024 320 318 1280 363 360 <t< td=""><td>235</td><td>260</td></t<>	235	260			
(359)	4QAM 5	1024	320	318	343
		1280	363	360	385
		1518	403	400	420
		9216	1630	1628	1653



Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	159	156	183
		128	168	166	187
		256	187	185	206
40	10.11	512	225	223	243
(359)	4QAM	1024	296	293	314
		1280	333	330	352
		1518	367	365	383
		9216	1433	Latency [µs] Minimum 156 166 185 223 293 330 365 1432 150 156 168 168 168 192 236 259 281 936 259 281 936 152 259 281 936 152 154 152 154 165 152 154 165 152 245 245 245 245 245 245 264 8355 151 151 161 151 151 161 179 210 228 243	1454
		64	152	150	177
		128	158	156	178
40		256	171	168	195
	1/04/16	512	194	192	204
(359)	16QAM S	1024	238	236	248
		1280	261	259	271
		1518	283	281	290
		9216	939	936	948
		64	159	152	177
		128	156	154	174
		Frame size [bytes] Average Minimum 64 159 156 128 168 166 256 187 185 512 225 223 1024 296 293 1280 333 330 1518 367 365 9216 1433 1432 64 152 150 128 158 156 256 171 168 512 194 192 128 158 256 171 168 152 128 256 171 1024 238 236 1280 261 259 1518 283 281 9216 939 936 64 159 152 128 156 154 256 179 165 512 187 186 1024 226	191		
40	1604M	512	187	186	211
(359)	IOQAM	1024	226	225	235
		1280	247	245	256
		1518	265	Latency [µs]AverageMinimum15915616816618718522522329629333333036736514331432152150158156171168194192238236261259283281939936159152156154179165187186226225247245265264838835156151153151154179212210229228247243715714	272
		9216	838	835	847
		64	156	151	176
		128	153	151	167
		256	175	161	181
40	320414	512	180	179	198
(359)	JZQAM	nodeFrame size (bytes)AverageMinimumModulation6415915612816816612816816625618718551222522310242962931128367365921614331432921614331432160AM 51512158160AM 512802612561711681512194192160AM 51512194160AM 512802612561711681512194192160AM 5256171161256179160AM 51280261256179165160AM1512187160AM2562641518265264160AM1518265160AM1518265160AM1518265160AM1518265160AM1518265160AM1518265160AM1518265160AM1518265160AM1518265161151826516115182651611518265161151826516215182161631515161163152161164156151165	210	218	
		1280	229	228	236
		1518	247	243	250
		9216	715	714	724

Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	158	152	169
		128	151	150	167
		256	169	158	174
40	6404M	Frame size [bytes]AverageMinimum6415815212815115025616915851217417310242001991280215214151822822792166196186415615212815715212815715212820720651217016910241941931280207206151821821792165605596415415112802072061518218217921656055964154151128020119915182112109216517516641531501280201199151821121092165175161281551501281551501281551501281551501281551641024184183128019619515182052049216164184	188		
(359)	64QAM	1024	Production Minimum Average Minimum 158 152 3 151 150 5 169 158 2 174 173 4 200 199 0 215 214 8 228 227 6 619 618 156 152 3 157 152 6 169 618 2 170 169 4 194 193 0 207 206 8 218 217 6 560 559 154 151 3 157 151 5 163 155 2 168 167 4 189 187 0 201 199 8 211 210 6 517 516 5	205	
		Frome size (bytes)AverageMinimum6dultion641581521281511502561691585121741731024200199128021521415182282279216619618128015515212801561529216616156512170169128020720651217016912802072061518218217921656055912802072061518218217921656055915121631551512168167151821121092165175161518211210921651751612801551501280155150128015515012801551641280155164128015516412801561641280155164128015516412801561641512165164151216516415121651641512165164151322616115182052041518205 <td>220</td>	220		
			227	232	
		9216	619	618	626
		64	156	152	165
		128	157	152	167
		256	166	156	170
40	1200414	512	170	169	182
40 (359)	128QAM	1024	194	193	198
		1280	207	206	211
		1518	218	217	221
		9216	560	Latency [µs] Minimum 152 150 158 173 199 214 227 618 152 152 152 155 167 151 155 167 187 199 210 516 155 167 187 199 210 516 155 167 187 199 210 516 155 167 187 199 210 516 155 167 187 199 210 516 155 167 187 199 210 516 155 167 187 199 210 516 155 167 187 199 210 516 155 167 187 199 210 516 155 167 187 199 210 516 155 167 150 155 167 150 155 167 150 155 167 155 167 155 167 155 167 155 167 155 167 155 167 156 150 155 167 150 155 167 150 150 155 167 150 155 167 150 150 150 150 155 167 150 150 150 150 150 155 164 183 195 204 483	566
		64	154	151	162
		128	157	151	164
		256	163	Minimum 152 150 158 173 199 214 227 618 152 152 152 152 152 152 152 152 152 152 152 152 152 155 169 193 206 217 559 151 155 151 155 167 187 199 210 155 164 150 155 164 183 195 204 483	166
40	25404M	512	168	167	178
(359)	256QAM	1024	189	187	192
		1280	201	199	204
		1518	211	Latency [µs] Minimum 152 150 158 173 199 214 227 618 152 152 153 206 217 559 151 155 167 187 199 210 516 150 155 167 187 199 210 516 150 155 164 183 195 204	213
		9216	517	516	522
		64	153	150	159
		128	155	150	161
		256	161	155	166
40	40 (359)64QAM256169164QAM1024200110242001128021519216619192166191128157112801281571280AM641561280A10241941280A10241941280A10241941280A641541280A2071151821811280A64154128056011518218112801514112801514112802011128020111280201112802011128015182111280151811280151611280151611280151611280151611280151611280155112801651128019611280196112801961128019611280196112801961128019611280196112801961128019611280196112801	164	174		
(359)	512QAM	1024	184	183	187
		1280	196	195	199
		1518	205	204	207
		9216	484	483	489



Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	152	149	158
		128	156	149	159
		256	160	154	163
40	10240404	512	164	163	171
(359)	modeFrame size (bytes)AverageLatModulationAverageM128156M1281561601281641601024QAM512164102418211024182110241821128019211518201192164611128015512048QAM512164128019111280191112801911128019111280191112801551128015511281155112811551128115111280199115181981128015411280158112801581128015811280158112801581128015811280158112801891128018911280189112801891128018911280189112801891128018911280189112801891	181	184		
		1280	192	191	195
		1518	201	200	203
		9216	461	Latency [μs] Minimum 149 149 154 163 181 191 200 460 154 150 149 191 200 460 150 149 149 149 162 180 199 451 149 153	465
		64	153	150	158
40 (359)		128	155	149	159
	2048QAM	256	159	154	163
		512	164	162	171
(359)		1024	181	180	183
		1280	191	190	194
		1518	200	199	201
		9216	453	451	456
		64	152	149	157
		128	155	149	158
	Prome size [bytes]ModulationFrame size [bytes]641282565121024QAM128010241280151892166412802048QAM5121024128010241280102412801024128015189216641280151892161024102410241282048QAM641280151892161024102412801518921611286412801518921612810241281282561211024128015181211281282561211281282561221024128015181280151812801518128015181280151812801518128015181280151812801518128015181280151812801518128015181280151812801518129161518129216151812801518129216151812921615181292161518129216151812921	159	153	163	
40	204904.04	512	163	Latency [µs] Minimum 149 149 154 163 181 191 200 460 191 200 460 191 191 153 162 180 199 199 451 149 199 451 149 199 451 149 199 451 149 199 451 149 199 451 149 199 451 149 199 451 149 153 162 179 189 197 441 149 153 162 179 189 197 189 197 189 197 189 197 189 197 197 189 197 197 189 197 197 189 197 197 189 197 197 189 197 197 189 197 197 189 197 197 189 197 197 189 197 197 197 197 197 197 197 197 197 19	170
(359)	2046QAM L	1024	180	179	182
		1280	190	Latency [µs]AverageMinimum1521491561491561491601541641631821811921912012004614601531501551491591541641621811801911902001994534511551491541621811801911902001994534511551491561491571491581621631621541491551491541491551491591531631621791781581531631621791781891881971964436435	192
		1518	198		200
		9216	442	441	446
		64	153	149	157
		128	154	149	157
		256	158	153	162
40	40060414	512	163	162	169
(359)	ModulationFrame size [bytes]AverageMinimum1641521491281561491281561491281561491281561601641621631024QAM1621811280192191151820120092164614601280192191151820120092164614601281551492648QAM51216412801911901280191190128019119012801551491280155149128015514912801551491280155149128015915315121631621024180179128019819792164424414096QAM641531280154162128015816212801581621280159153128015816212801581621280158163128015816212801581621280158162158158162158158162158158153158<	181			
		1280	189	188	191
		1518	197	196	198
	9216	436	435	439	

Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	152	149	156
		128	154	149	157
		256	158	153	161
40	4006 O A M I	512	162	161	168
(359)	4096QAM L	1024	178	177	180
		1280	188	186	190
		1518	195	194	196
		9216	427	AverageMinimum1521491541491581531621611781771881861951944274261521491531481571521621611771761871861951944224211221191301271461441771752362342692662962941185118112011712712414113916816621921724724427226910351034	430
		64	152	149	155
		128	153	148	156
		256	157	152	161
40	010204M	512	162	161	168
(359)	8192QAM	1024	177	176	179
		1280	187	186	189
		1518	195	194	195
		9216	422	Latency [µs] Minimum 149 149 149 153 161 177 186 194 426 194 426 194 426 194 426 194 426 194 426 194 421 161 176 186 194 421 161 176 186 194 421 119 127 144 175 234 266 294 1181 117 124 139 166 217 244 269 1034	425
		64	122	119	144
		128	130	127	144
		256	Ibytes) Average Minimum 152 149 154 149 158 153 162 161 4 178 177 0 188 186 8 195 194 6 427 426 152 149 149 3 153 148 6 427 426 152 149 153 3 153 148 5 157 152 162 161 14 177 176 16 4 177 176 0 187 186 8 195 194 6 422 421 122 119 1 3 130 127 5 146 144 2 177 175 4 236 234 <td< td=""><td>149</td></td<>	149	
55 56	404145	512	177	175	187
62.5 (360)	4QAM S	1024	236	234	251
(000)		1280	269	266	283
		1518	296	294	299
		9216	1185	1181	1189
		64	120	117	146
		128	127	124	140
		256	141	139	153
55 56	406412812825651251210241024110241102411024110241024110241024111010241128102411281024112810241128010241128010241110102411	168	166	181	
62.5 (360)	4QAM	1024	219	217	232
(330)		1280	247	244	260
		1518	272	269	274
		9216	1035	1034	1041



Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	130	120	142
		128	119	117	127
		256	131	127	147
55 56	1/04/45	512	146	145	153
62.5 (360)	I6QAM S	Frame size (bytes) Average Minimum 64 130 120 1 128 119 117 1 256 131 127 1 512 146 145 1 1024 178 177 1 1024 178 177 1 1024 178 177 1 118 211 210 1 9216 684 683 1 64 126 119 1 128 118 116 1 256 131 124 1 1024 170 169 1 1024 170 169 1 1280 187 185 1 1280 187 185 1 1280 132 122 1 1280 132 135 1 1280 134 160 159 1	186		
		1280	196	Latency [µs]AverageMinimum130120119117131127146145178177196195211210684683126119118116131124142140170169187185199198613612123118132122136135160159174172185183527526119116128119132131152151164162173172459458	203
		IdationFrame size [bytes]Interface (bytes)AverageMinimum641301201128119117125613112715121461451102417817711280196195115182112101921668468316412611911280131124192166131124125613112411280187185112801871851128018718511280187185115181991981128013211811280132122115181321221128017417211280174172112801741721128017417211280174172112801281831128012812812812801741721128017417211280174172112801281281831280174172112801741721128012818311280128 <td>217</td>	217		
		9216	684	683	688
		64	126	Latency [µs] Minimum 120 117 127 145 177 195 210 683 119 116 124 140 169 185 198 612 185 198 612 118 198 612 118 118 122 135 159 172 135 159 172 183 526 116 116 116 117 151 151 162 172 458	138
		128	118	116	125
	55 16QAM 128 55 16QAM 512 1024 1024 1280 1280 1518 9216	131	124	143	
55 56 62.5 (360)		512	142	140	147
		1024	170	169	176
		1280	187	185	192
		1518	199	198	204
		9216	613	612	617
		64	123	118	131
		128	124	118	133
		IditionFrame size (bytes)AverageMinimum12813012012811911712811911725613112751214614510241781771280196195151821121092166846830921668468310241261191281181162561311245121421401024170169128018718515181991989216613612151819919892161321181280174172151818518392165275261280174172151818518392165275261281281281281281191518185183921652752612801641621518132131102415215112801641621518172151128016416215181731729216459458	136		
55	720414	512	136	135	141
62.5 (360)	JZQAM	1024	160	159	164
		1280	174	172	178
		1518	185	183	188
		9216	527	526	533
		64	119	116	127
		128	122	116	128
		256	128	119	129
55 56	640AM	512	132	131	136
62.5 (360)	modeFrame size [bytes]Latency [µs]ModulationFrame size [bytes]AverageMinimum1281301201281191171281191171281461451024178177128019619515121606841024178110128011811612801181161280118116128111811612821181161284118161128411816112841181611284187185151216111812841241181284124118128412411812841241181284124118128412411812841241181284124118128412411812841241181284124116128412812413141281241484119116151818518316241281281635128119164128128164128119164128116151812813116241281281635128131 <t< td=""><td>175</td></t<>	175			
		1280	164	162	167
		1518	173	172	176
		9216	459	458	464

Physico	Il mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	115	110	121
		128	121	116	124
		256	121	115	125
55 56	1200414	512	125	118	129
62.5 (360)	128QAM	1024	AverageMinimum64115110112812111612561211151512125118110241301281128014714611518158156192161661641641191151128120114125612411815121281261128120114125612411815121531521128015315211518161159112801201141256123118151212712611280159149112801561551921636536311280118114140128011811411280129124112801211171512129124112801481461128014814611280128138146128014814611280148146112801281381461280148146112801481461<	133	
(300)		1280		150	
		tionFrome size (bytes)AverageMinimum12811511011281211161256121115151212511811024130128112801471461151815815619216166164112812011412561241181102414414212561241181151212815211518161159115181611591151816115911518161159115181611591151816115911518161159115181561181151815615511518156155115181561553631512129124115121291241151212912411280148146112801481461128014814611280128138112801481461128014814611280148146112801481461 <t< td=""><td>160</td></t<>	160		
		9216	166	Latency [µs] Minimum 110 116 115 118 128 146 156 164 115 114 115 114 115 114 115 114 115 142 152 159 386 114 118 126 142 152 386 114 118 126 140 114 117 126 363 114 117 124 138 146 152 347	168
		64	119	115	122
		128	120	114	123
		256	124	118	128
55 56	25604M	512	128	126	135
56 62.5 (360)	256QAM	1024	144	142	146
(300)		1280	153	152	155
		1518	161	159	162
		9216	388	386	391
		64	119	114	122
		128	120	114	122
		Prome size loyresAverageMinimum641151101281211162561211155121251181024130128128014714615181581569216166164641191151281201145121281261281201145121281261024144142128015315215181611591518161159921638838664119114128012311851212712612811501491282150149151815636364117114128118155921636536364117114128118114128118114128118114128118114128118114128014814612801481461280148146	126		
55 56	5120AM	512	127	126	129
62.5 (360)	JIZQAM	1024	141	140	143
		1280	150	149	152
		1518	156	Participant Participant e Minimum 110 1 1116 1 1115 1 1115 1 118 1 118 1 116 1 118 1 164 1 114 1 114 1 115 1 114 1 115 1 114 1 115 1 116 1 118 1 159 3 386 1 114 1 114 1 114 1 114 1 1140 1 140 1 140 1 140 1 140 1 140 1 155 3 363 1	158
		9216	365	363	368
		64	117	114	120
		128	118	114	121
		256	121	117	125
55 56	112 112 116 128 121 116 1 256 121 115 1 1024 130 128 1 1024 130 128 1 1024 130 128 1 1024 130 128 1 1024 130 128 1 1024 130 128 1 1280 147 146 1 1518 156 166 164 1 1280 124 114 1 1 2560AM 64 119 114 1 2560AM 1512 128 126 1 1280 153 152 1 1 1280 153 152 1 1 1280 128 118 1 1 1280 150 149 1 1 1280 150 149<	131			
62.5 (360)	TOCHONIN	1024	139	138	141
(/		1280	148	146	149
		1518	154	152	155
		9216	349	347	351



Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	117	Latency [µs] Minimum 114 114 117 123 137 146 151 341 113 113 113 113 117 123 136 145 150 333 113 145 150 333 113 117 123 136 145 150 333 113 117 123 136 144 149 329 114 117 123 136 144 149 329 114 117 123 136 144 149 329 114 117 123 136 144 149 329 114 117 123 136 144 149 329 114 117 123 136 144 149 123 136 144 149 123 136 144 149 123 136 144 149 123 136 144 149 123 136 144 149 123 136 144 149 123 136 144 149 123 136 144 149 123 136 144 149 123 136 144 149 149 143 143 143 144 149 143 144 144 149 145 145 145 146 147 149 143 143 143 144 147 147 147 148 149 148 148 148 148 149 148 148 148 149 148 148 149 148 149 148 149 148 148 149 148 148 148 149 148 148 149 148 148 148 149 148 148 148 148 148 148 149 148 148 148 149 148 148 148 148 148 148 148 148	120
		128	118	114	121
		256	121	117	124
55 56	20400404	512	127	123	130
62.5 (360)	2048QAM	1024	Latency [µs] Average Minimum 64 117 114 128 118 114 256 121 117 512 127 123 1024 138 137 1280 147 146 1518 153 151 9216 342 341 64 116 113 128 117 113 9216 342 341 64 116 113 128 117 113 256 120 117 512 128 123 1024 138 136 1280 146 145 1518 152 150 9216 335 333 64 116 113 128 117 113 128 125 123 1024 137 136 1280	140	
		1280	147	Latency [µs] Minimum 114 114 114 117 123 137 146 151 341 113 113 113 113 113 117 123 136 145 145 150 333 113 113 113 113 113 113 11	148
		1518	153	151	154
		9216	342	341	345
		64	116	113	119
		128	117	113	120
		256	rame size [bytes]AverageMinimum64117114112811811412561211171512127123110241381371128014714611518531511921634234116411611311280120117112801281231128115315119216342341112811711311281161131128014614511280116113112811711311281211171128121117112815132911280145144115181201171128117114112811711411281271361128128123112812812311281281231128117144112812812311281281231128128123112812812311281281231128128143 </td <td>124</td>	124	
55 56	204004041	512		129	
62.5 (360)	2048QAM L	SQAM 512 127 123 1024 138 137 1280 147 146 1518 153 151 9216 342 341 9216 342 341 128 117 113 128 117 113 128 117 113 128 117 113 128 117 113 128 128 123 1024 138 136 1280 146 145 1518 152 150 9216 335 333 60AM 60A 116 113 128 117 113 128 117 113 160 145 144 1518 151 149 9216 331 329	139		
		1280	146	145	147
		1518	152	150	153
		9216	335	333	338
		64	116	113	119
		128	117	113	120
		Average Minimum 64 117 114 128 118 114 256 121 117 512 127 123 1024 138 137 1280 147 146 1518 153 151 9216 342 341 64 116 113 128 117 113 256 120 117 512 128 123 1024 138 136 128 117 113 256 120 117 512 128 123 1024 138 136 1518 152 150 9216 335 333 64 116 113 128 117 113 256 121 117 512 125 123 1024 137 136	124		
55	400604M	512	125	123	129
62.5 (360)	4090QAM	1024	137	136	138
		1280	145	144	146
		1518	151	149	152
		9216	331	329	333
		64	117	114	119
		128	117	114	120
		256	120	117	123
55	40060AM	512	128	123	129
62.5 (360)	4090QAM L	eFrame size [bytes]Interface [bytes]VodulationFrame size [bytes]AverageMinimum64117114112811811412848QAM512127123151212712311102413813711280147146115181551511921634234111281171131128117113112811711311281171131128128128136128014614511280146145112801461131128012111711280125123112801451441128014514411280145144112801171141128012811711412801451431128014514311280145143112801451431128014514311280145143112801451431128014514311280145143112801451431128	138		
		1280	145	143	146
		1518	150	149	151
		9216	325	323	327

Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	98	Minimum 96 102 110 135 181 204 225 874 98 100 110 135 181 204 225 874 98 100 110 131 169 190 208 771 97 95 102 116 141 155 166 527 97 97 97 101 113 135 148 158 478	119
		128	103	102	123
		AdulationFrame size (bytes)AverageMinimumAverageMinimum12810310212810310225612011051214513510241831811280206204151822722592168758746410598128101100256120110512133131102417116912801911901518210208921677377112801911901518210208921677377164103971289795256110102512118116102414214112801561551518168166921652952764101971281019712810197128101971281019712811110112911411312041361351205114148151815915812801491481518159158128014914815041541591280149148 </td <td>140</td>	140		
80	404146	512	145	Minimum 96 102 110 135 181 204 225 874 98 100 110 125 874 98 100 110 131 169 190 208 771 97 95 102 116 141 155 166 527 97 97 101 113 155 166 527 97 97 97 101 113 135 148 158 478	155
(403)	4QAM S	1024	183	181	193
		1280	206	age Minimum 3 96 3 102 0 110 5 135 3 181 6 204 7 225 5 874 5 98 1 100 0 110 3 131 1 169 1 190 0 208 3 771 3 97 7 95 0 102 8 116 2 141 6 155 8 166 9 527 1 97 1 101 4 113 6 135 9 148 9 158 9 478	216
		1518	Average Minimum 64 98 96 128 103 102 256 120 110 512 145 135 1024 183 181 1280 206 204 1518 227 225 9216 875 874 64 105 98 128 101 100 256 120 110 512 133 131 128 101 100 256 120 110 512 133 131 1024 171 169 128 210 208 9216 773 771 64 103 97 128 97 95 256 110 102 512 118 116 1024 142 141 128 168 166	235	
		9216	875	Latency [µs] Minimum 96 102 110 135 181 204 225 874 98 100 110 110 131 169 190 208 771 97 97 97 95 102 102 116 141 155 166 527 97 97 95 102 116 141 155 166 527 97 97 101 113 135 148 158 478	887
		64	105	98	123
		128	101	100	119
		256	120	110	136
80		512	Average Minimum 98 96 103 102 120 110 145 135 183 181 206 204 227 225 875 874 105 98 101 100 120 110 105 98 101 100 120 110 105 98 101 100 120 110 171 169 191 190 210 208 773 771 103 97 97 95 110 102 111 101 142 141 156 155 168 166 529 527 101 97 101 97 101 97 114	156	
(403)	4QAM	1024	171	Verage Minimum 98 96 103 102 120 110 145 135 183 181 206 204 227 225 875 874 105 98 101 100 120 110 133 131 171 169 191 190 210 208 773 771 103 97 97 95 110 102 118 116 142 141 156 155 168 166 529 527 101 97 111 101 114 113 136 135 149 148 159 158 479 478	179
(403)		1280	191	190	200
		1518	210	208	214
		9216	773	771	783
		64	103	97	112
		128	97	95	111
		256	110	Minimum 96 102 110 135 181 204 225 874 98 100 110 131 169 190 208 771 97 95 102 116 141 155 166 527 97 97 97 101 113 135 148 158 478	117
80	1/0446	512	118	116	130
(403)	IOQAM S	1024	142	141	146
		1280	156	155	160
		1518	168	166	171
		9216	529	527	535
		64	101	97	109
		128	101	97	111
		256	111	101	113
80	160414	512	64 98 96 28 103 102 156 120 110 112 145 135 024 183 181 280 206 204 518 227 225 216 875 874 64 105 98 128 101 100 126 120 110 518 227 225 216 875 874 64 105 98 128 101 100 120 1133 131 024 171 169 280 191 190 518 210 208 216 773 771 64 103 97 128 97 95 256 110 102 1312 118 116 024 142 141	125	
(403)		1024	136	135	140
		1280	149	148	153
		1518	159	158	161
		9216	479	478	484



Physico	l mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	98	95	105
		128	101	95	106
		256	106	100	111
80	720.004	512	111	110	119
(403)	52QAM	1024	Ibytes] Average Minimum 98 95 101 95 106 100 111 110 129 128 140 139 140 139 140 139 140 139 140 139 140 139 140 139 141 148 98 95 100 95 103 99 108 106 124 123 133 132 3 141 140 132 3 141 102 98 97 94 98 94 102 98 103 135 3 136 120 119 121 129 122 128 3 343 <	132	
		1280	140	139	143
		1518	149	148	151
		9216	419	Latency [µs] Minimum 95 95 100 110 128 139 148 418 95 95 95 95 95 99 106 123 132 140 371 140 371 140 371 94 94 94 94 94 94 94 94 94 94	424
		64	98	95	102
		128	100	95	102
80		256	103	99	107
	(404)4	512	108	106	114
(403)	64QAM	1024	124	123	126
		1280	133	132	135
		1518	141	140	142
		9216	372	371	375
		64	97	94	100
		128	98	94	101
		256	102	98	104
80	12804M	512	107	105	111
(403)		1024	120	119	122
		1280	129	128	131
		1518	136	verageMinimum9895101951061001111101291281401391494189895100951039910413213313214114037237197949894102981071051201191291281361353433429694979310097108104118117126125132321	137
		9216	343	342	346
		64	96	94	98
		128	97	93	100
		256	100	97	103
80	25604M	512	108	104	109
80 (403) 80 (403) 80 (403) 80 (403) 80 (403)		1024	118	117	119
	Physical mode CS [MHz] (Rame ID) Modulation 80 (403) 32QAM 80 (403) 64QAM 64QAM 128QAM 80 (403) 128QAM 80 (403) 128QAM	1280	126	125	127
		1518	132	131	133
		9216	322	321	324

Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	95	93	97
		128	96	94	98
		IduationFrame size [bytes]AverageMinimum6495936128969462569996651210710310241161151280124123151813113092163063059216989651210610212809169324QAM649592128969325698965121061021024115114102411511412801221211518128959216295294128959312895102128012112612181271261280121209151812712692162912896494921518127126921629128965121061021518127126921629128964949215181271269216291289649492151812712616494921518126102151812712615121051021513126	101		
80	5120AM	512	107	Minimum 93 94 96 103 115 123 130 305 92 93 96 123 124 125 92 93 96 102 114 121 127 294 93 96 102 113 127 294 93 93 93 93 93 93 93 93 93 94 92 93 92 93 96 102 113 120 113 120 113 120 125 </td <td>108</td>	108
(403)	512QAM	1024	116	115	117
		1280	Average Minimum 95 93 96 94 99 96 107 103 116 115 124 123 306 305 98 96 99 96 116 115 124 123 306 305 99 96 995 92 96 93 98 96 105 114 122 121 128 127 295 294 94 93 95 93 98 96 105 102 114 113 121 120 122 289 94 92 95 93 98 96 127 126 291 289 94 92	125	
		ionAverageMinimum649593112896941256999612569996110241161151128012412311518131130192163063051256989611281061021128106102112812629114412812212111024115114112801221211151812812719216295294112801211021128012112011280121120112802519311289593112895931128012112011280121126112801211021128012110211280120102112801201021	132		
		9216	306	305	308
		64	95	Minimum 93 94 96 103 115 123 130 305 92 93 96 102 114 121 127 294 93 96 102 114 121 127 294 93 96 102 113 120 123 120 124 93 96 102 113 120 123 93 96 102 113 120 1213 1220 123 124 125 284	96
		128	96	93	98
		256	98	96	100
80	10240414	512	NS NE 96 93 98 96 106 102 115 114 122 121 128 127 295 294 94 93 95 93 98 96	107	
(403)	1024QAM	1024		118	
(403)		1280	122	121	123
		1518	128	127	128
		9216	295	294	296
		64	94	93	96
		128	95	93	97
		256	98	96	100
80	20400404	512	105	102	106
(403)	2048QAM	1024	114	113	115
		1280	121	120	122
		1518	127	126	127
		9216	291	289	292
		64	94	92	95
		128	95	93	97
		256	98	96	99
80	20400404	512	105	959396949996107103116115124123131130306305959296939896106102115114122121128127295294949395939896105102114113121120125289949295939896105102114113121120126289949295939896105102114113120120124125286284	106
(403)	2048QAM L	1024	114	113	116
		1280	120	120	121
		1518	126	125	127
		9216	286	284	287



Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	94	Latency [µs] Minimum 92 93 95 102 112 112 119 124 281 92 93 93 95 102 112 112 119 124 277 80 102 112 119 124 277 80 80 80 80 80 80 80 80 80 80 80 80 80	95
		128	95	93	96
		256	97	95	99
80	400/0414	512	105	102	106
(403)	4096QAM	1024	Average Minimum 64 94 92 128 95 93 256 97 95 512 105 102 1024 113 112 1280 120 119 1512 105 102 1024 113 112 1280 120 119 1518 125 124 9216 282 281 64 94 92 128 95 93 256 97 95 512 104 102 1024 113 112 128 95 277 129 104 102 1210 124 113 1212 104 102 128 82 80 128 82 80 129 109 107 1280 159 157	116	
	80 (403) 4096QAM 51 102 102 110 102 921 921 921 921 921 921 921 921 921 921 921 921 12 12 100 102 110 102 112 4096QAM L 125 40AM S 102 51 102 51 102 51 102 51 103 51 104 102	1280	120	119	123
		1518	125	124	126
		9216	282	281	284
		64	94	92	95
		128	95	93	96
		256	97	Latency [µs] age Minimum 1 92 5 93 7 95 5 102 3 112 0 119 5 124 2 281 4 92 5 124 2 281 4 92 5 124 2 281 4 92 5 124 92 80 3 112 0 119 5 124 8 277 2 80 2 80 2 80 2 80 5 193 9 107 1 139 9 157 4 172 9 648 3 81 3 83 <	99
80	128 95 256 97 512 104 1024 113 1280 120 1518 125 9216 278	102	453		
80 (403)		1024	113	112	116
		1280	120	119	122
		1518	125	124	125
		9216	278	277	279
		64	92	80	105
		128	82	80	96
110		256	96	rage Minimum P4 92 P5 93 P7 95 05 102 13 112 20 119 25 124 82 281 P4 92 P5 93 P7 95 04 92 P5 93 P7 95 04 102 13 112 20 119 25 93 P7 95 04 102 13 112 20 119 25 124 78 277 P2 80 32 80 32 80 32 80 9 107 41 139 59 157 74 172 49 648	111
112	101115	512	109	107	116
(361	4QAM 5	1024	141	139	149
404)		1280	159	157	166
		1518	174	172	180
		9216	649	648	653
		64	88	81	101
		128	88	83	102
110		256	98	87	105
80 (403) 4096QAM 64 128 256 1024 1024 1280 1128 110 1518 9216 1128 64 1280 1518 9216 64 128 9216 1128 128 128 128 128 128 1024 1024 1280 1518 9216 1518 9216 1518 9216 110 128 112 125 (361 40AM S 110 1280 1518 9216 1518 9216 1518 9216 128 128 128 128 1212 128 125 1361 126 122 125 1024 128 128 128 128 1280	105	103	110		
(361	4QAM	1024	Latency [μs] Average Minimum 94 92 95 93 97 95 105 102 113 112 120 119 125 124 282 281 94 92 95 93 97 95 120 119 125 124 282 281 94 92 95 93 97 95 104 102 113 112 120 119 125 124 278 277 92 80 82 80 96 89 109 107 141 139 159 157 174 172 649 648 88 83 <tr td=""> 105</tr>	138	
404)		1280	149	147	154
		1518	162	160	167
		9216	575	574	578

Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	83	Latency [µs] Minimum 78 78 78 81 92 1111 122 130 399 78 78 80 90 107 107 107 107 107 107 107 10	89
		128	83	78	89
110		256	90	81	93
110	1/04146	512	93	92	96
(361	16QAM S	1024	112	111	115
404)		1280	123	122	126
		1518	132	130	134
		9216	400	399	404
		64	82	78	86
		128	82	78	87
110		256	87	80	90
110		512	92	90	94
(361	16QAM	1024	12 92 90 024 108 107 280 118 117 518 126 125 216 364 363	110	
(361 404)		1280	118	117	120
		1518	126	125	128
		9216	364	363	368
		64	80	77	83
		128	81	77	84
110		256	84	80	88
112	720414	512	90	87	95
(361	32QAM	1024	103	102	105
404)		1280	112	111	114
		1518	118	117	120
		9216	322	321	325
		64	78	75	81
		128	80	76	82
110 1110 1110 1110 1110 1110 1110 1110 1111	85				
112	640414	512	89	85	91
(361	04ŲAM	1024	99	98	100
404)		1280	107	106	108
		1518	112	111	114
		9216	288	287	290



Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	77	75	79
		128	78	75	80
110		256	81	78	83
112	1200404	IditionFrome size [bytes]Lotency [us]IditionAverageMinimumI647775I1287875I2568178I5128884I10249796I1180104103I1518109108I9216268266I9216268266I921626880751287775I1287775I1288683I128102100I128102100I128102100I1280102100I1280102100I12817675I128010099I128010099I1280104103I12807674I1280104103I1280104103I12807877I1287674I1287674I1287674I1287674I1287674I1287674I1287674I1287674I1287674I	89		
(361	128QAM	1024	97	96	100
404)		1280	104	103	105
		1518	Average Minimum 64 77 75 128 78 75 256 81 78 512 88 84 1024 97 96 1280 104 103 1518 109 108 9216 268 266 64 76 75 128 77 75 128 77 75 128 77 75 128 77 75 128 77 75 128 77 75 128 76 78 1280 102 100 1518 106 105 9216 253 251 64 76 74 128 76 75 256 79 77 512 86 84 1024 93 92 1280	110	
		9216	268	266	269
		64	76	75	78
		128	77	75	78
110		256	80	78	82
112	25/0414	512	86	83	88
(361	256QAM	1024	95	94	98
(361 404)		1280	102	100	103
		1518	106	105	107
		9216	253	251	254
		64	76	74	77
		128	76	75	78
110		256	79	Latency [µs] Minimum 75 75 78 84 96 103 108 266 75 75 75 75 78 83 94 100 105 251 74 100 105 251 74 75 77 84 92 99 103 240 74 74 75 77 84 92 99 103 240 74 74 77 82 91 98 102 231	81
112	5120AM	512	86	84	87
(361	512QAM	1024	93	92	96
404)		1280	100	99	103
		1518	104	103	105
		9216	241	240	242
		64	75	74	77
		128	76	74	77
110		256	78	77	80
112	110 512 88 1280AM 512 88 1024 97 1280 104 1280 104 1518 109 9216 268 64 76 128 77 256 80 112 256 128 77 256 80 112 256 128 77 256 80 112 256 112 86 128 76 1280 102 1518 106 9216 253 64 76 128 76 128 76 128 1024 93 1280 1024 93 1280 100 1518 104 9216 241 64 75 128 76	82	86		
(361	LØZ4QAM	1024	92	91	95
404)		1280	99	98	102
		1518	103	102	104
		9216	233	231	234

Physical mode			Latency [µs]			
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum	
		64	75	74	76	
		128	76	74	77	
		256	79	77	80	
110 112	20400414	512	Average Minimum 64 75 74 28 76 74 26 79 77 512 85 83 024 92 91 280 98 97 518 103 101 216 230 228 64 75 74 280 98 97 518 103 101 216 230 228 64 75 74 28 76 74 28 76 74 28 76 74 28 76 74 28 76 74 28 76 74 280 98 97 518 102 101 216 226 225 64 75 74 28 75 74 291 90	86		
125 (361)	2048QAM	1024	92	91	95	
(301)		1280	98	97	101	
		1518	103	101	103	
		9216	230	228	231	
		64	75	74	76	
	20/20 000	128	76	74	77	
		256	78	77	80	
110 112		512	85	83	85	
125 (361)	2048QAM L	1024	size [bytes] Average Minimum 64 75 74 1 128 76 74 1 256 79 77 1 512 85 83 1 1024 92 91 1 1280 98 97 1 1280 98 97 1 1280 98 97 1 1280 98 97 1 9216 230 228 1 64 75 74 1 128 76 74 1 256 78 77 1 128 76 2101 1 1280 98 97 1 1280 98 97 1 1280 98 97 1 1280 75 74 1 256 78 77 1 128 75	94		
(002)		1280	98	97	100	
		1518	102	101	102	
		9216	226	225	227	
		64	75	74	76	
		128	75	74	76	
		256	78	77	80	
110 112	400604M	512	84	82	85	
125 (361)	4070QAM	1024	91	90	94	
		1280	97	96	98	
		1518	101	100	102	
		9216	224	222	225	

2.2.2 Traffic Latency for ANSI

The table below shows the latency values for low latency Frame ID's.

Values for standard Frame ID's (262, 263, 264, 265, 266, 267 and 268) can be found in revision BT of this document.



Physico	al mode			Latency [µs]	
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	567	540	730
		128	609	580	730
		256	730	700	830
10	404145	512	936	900	1030
(362)	4QAM S	1024	1341	1260	1380
		1280	1512	1440	1560
		1518	1532	1490	1570
		9216	6665	6640	6700
		64	556	530	670
		128	584	570	690
		256	684	650	770
10	40414	512	859	830	930
(362)	4QAM	1024	1202	1170	1240
		1280	1363	1340	1400
		1518	1367	1340	1410
		9216	5757	5730	5790
		64	521	500	540
		128	541	520	590
		256	590	570	610
10	1604M 5	512	687	672	701
(362)	IOQAM 5	1024	876	840	920
		1280	967	930	990
		1518	1002	992	1020
		9216	3590	3570	3600
		64	514	500	580
		128	532	521	550
		256	570	558	587
10	1/04/4	512	651	630	700
(362)	τουΑΜ	1024	818	790	850
		1280	894	860	930
		1518	930	920	980
		9216	3148	3136	3163

Table 4 Traffic Latency for ANSI

Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	504	496	525
		128	519	511	540
		256	547	542	571
10	720414	512	Jize [bytes] Average Minimum 64 504 496 128 519 511 256 547 542 512 613 600 024 728 717 280 791 776 518 840 830 216 2619 2612 64 496 491 128 509 503 256 536 520 512 585 570 024 682 670 256 536 520 512 585 570 024 682 670 280 737 720 216 2201 2197 64 493 487 128 506 498 256 524 510 512 567 560 024 651 640 280 695	650	
(362)	52QAM	1024	728	717	746
		1280	791	Average Minimum 504 496 519 511 547 542 613 600 728 717 791 776 840 830 2619 2612 496 491 509 503 536 520 585 570 682 670 737 720 769 760 2201 2197 493 487 506 498 524 510 567 560 651 640 695 680 725 721 1948 1941 490 485 498 490 517 510 518 540 628 622 667 660 693 691 1761 1756	805
		Frame size [bytes] Average Minimum 64 504 496 1 128 519 511 1 256 547 542 1 1024 728 717 1 1024 728 717 1 1024 728 717 1 11024 728 717 1 11280 791 776 1 11280 791 776 1 11280 791 776 1 11280 791 2612 1 1128 509 503 1 1128 509 503 1 1128 506 520 1 11280 737 720 1 11280 769 760 1 11280 506 498 1 1128 506 498 1 1128 506 498 1 </td <td>870</td>	870		
		9216	2619	Latency [μs] Minimum 496 511 542 600 717 776 830 2612 491 503 520 570 670 720 760 2197 487 498 510 560 640 680 721 1941 485 490 510 560 640 680 721 1941 485 490 510 540 622 660 691 1756	2635
		64	496	491	520
		128	509	503	532
		256	536	520	560
10	(10.11)	1024 728 717 1280 791 776 1518 840 830 9216 2619 2612 64 496 491 128 509 503 256 536 520 512 585 570 1024 682 670 1280 737 720 1518 769 760 9216 2201 2197 64 493 487 128 506 498 256 524 510 512 567 560 1280 695 680	610		
(362)	64QAM	1024	1024 728 717 1280 791 776 1518 840 830 9216 2619 2612 64 496 491 128 509 503 256 536 520 512 585 570 1024 682 670 1024 682 670 1280 737 720 1518 769 760 9216 2201 2197 64 493 487 128 506 498 256 524 510 512 567 560 128 506 498 256 524 510 512 567 560 1024 651 640 1280 695 680 1280 695 680 1518 725 721 9216 1948	710	
(362)		1280	737	720	760
		1518	769	760	790
		9216	2201	2197	2214
		64	493	487	516
		128	506	498	526
		256	524	Editency (µs) 496 511 542 600 717 776 830 2612 491 503 520 570 670 720 760 2197 487 498 510 560 640 680 721 1941 485 490 510 542 660 640 680 721 1941 485 490 510 540 622 660 691 1756	570
10	1200414	512	567	560	590
(362)	128QAM	1024	651	640	670
		1280	695	680	720
		1518	725	721	741
		9216	1948	1941	1955
		64	490	485	514
		128	498	490	530
		256	517	510	560
10 (362) 128QAM 506 10 (362) 128QAM 512 567 1024 651 1280 695 1518 725 9216 1948 64 490 128 498 10 2560AM 512 554	540	570			
(362)		1024	628	622	642
		1280	667	660	680
		1518	693	691	707
		9216	1761	1756	1768



Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	488	483	512
		128	497	490	520
10 (362) 10 (362)		256	514	500	550
10	51204M	de ModulationFrame size [bytes]AverageMinimum64488483612849749061284974906512547542610246116046128064564261518669667692161619161561280494480615186696676921616191615612805994956128059949561281537532612826436696671280629623612161514151161216315300612163563406128072066961280720669612807206696128030029261280315302861280720669612807211504001280647550312806476263	553		
(362)	SIZQAM	1024	611	Latency [µs] Minimum 483 490 500 542 604 642 667 1615 480 495 532 592 623 650 1511 279 300 340 420 592 623 650 1511 279 300 340 420 590 680 740 3228 274 292 327 400 550 620 670 2798	615
		Frame size [bytes]AverageMinimum644884836128497490625651450065125475426102461160461280645642615186696676921616191615692165375326102459859261280629623612806296236128062962361280629623610245985921128062530011280625340692163151300112807206801128072832281128072832281128072832281128030029221281300292212823154001128430029221285339327512806476201128064762011280647620112806476201128064762011280647620112806476201128064762011	654		
		1518	669	667	681
		9216	1619	1615	1625
		64	487	Litericy [µs] Minimum 483 490 500 542 604 642 667 1615 482 480 495 532 592 623 650 1511 279 300 340 420 590 680 740 3228 274 292 327 400 550 620 670 2798	511
		128	494	480	520
		256	509	Latency [µs] Average Minimum 488 483 4 497 490 1 514 500 1 547 542 1 641 604 1 645 642 1 645 642 1 645 647 1 645 642 1 645 642 1 649 1615 1 1619 1615 1 487 482 1 537 532 1 537 532 1 629 623 1 629 623 1 1514 1511 1 287 300 1 720 680 1 626 590 1 720 680 1 720 680 1 7339 327 1	530
10	10240444	512	537		552
(362)	1024QAM	9216 1619 1615 64 487 482 128 494 480 256 509 495 512 537 532 1024 598 592 1280 629 623 1518 653 650 9216 1514 1511 64 287 279 128 315 300 256 356 340 512 452 420	592	613	
(562)		1280	629	623	642
		1518	653	650	663
		9216	1514	1511	1521
		64	287	279	308
		128	315	300	360
		256	356	Minimum 483 490 500 542 604 642 667 1615 480 495 532 592 623 650 1511 279 300 340 420 592 653 592 623 650 1511 279 300 340 420 592 3279 3200 590 680 740 3228 274 292 327 400 550 620 670 2798	410
20	404145	512	452	420	500
(363)	4QAM S	1024	626	590	670
		1280	720	680	760
		1518	764	740	810
		9216	3238	3228	3256
		64	283	274	303
		128	300	292	321
		256	339	327	356
20	404M	512	415	400	460
(363)	4QAM	1024	571	550	600
		1280	647	620	660
		1518	684	670	730
		9216	2804	2798	2822

Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	267	Minimum 261 270 290 330 423 460 503 1771 259 268 287 325 398 439 469 1563 256 260 280 310 368 399 427 1312 258 260 270 298 346 371 394 1113	279
		128	279	270	320
		256	Ibytes) Average Minimum 267 261 279 270 299 290 346 330 4 433 4 433 4 433 4 433 4 433 4 433 4 433 4 433 4 433 508 503 6 1775 1771 263 259 259 3 272 268 259 330 325 4 405 330 325 4 405 346 1563 258 256 267 260 258 266 267 260 263 280 2 315 310 427 6 1315 316	350	
20	1(04MC	512	346	Minimum 261 270 290 330 423 460 503 1771 259 268 287 325 398 439 469 1563 256 260 280 310 368 399 427 1312 258 260 270 298 346 371 394 1113	370
(363)	TOUAM 2	1024	433	423	442
		1280	480	Proge Minimum 67 261 79 270 99 290 46 330 33 423 80 460 08 503 775 1771 163 259 172 268 190 287 30 325 05 398 47 439 174 469 568 1563 258 256 267 260 183 280 315 310 371 368 104 399 130 427 315 1312 266 258 262 260 276 270 302 294 315 346 374 346	500
		1518	Average Minimum 267 261 279 270 299 290 346 330 433 423 480 460 508 503 1775 1771 263 259 272 268 290 287 330 325 330 325 447 439 447 439 447 469 1568 1563 258 256 267 260 263 280 315 310 315 310 371 368 404 399 430 427 1315 1312 266 258 262 260 276 270 302 298 351 346 376 371	524	
		9216	1775	1771	1784
		64	263	259	288
		128	272	268	293
	256 290	287	300		
20	1/01/1	512	Latency [µs] Average Minimum 267 261 1 279 270 1 279 270 1 299 290 1 346 330 1 433 423 1 480 460 1 508 503 1 1775 1771 1 263 259 1 272 268 1 272 268 1 272 268 1 274 469 1 330 325 1 405 398 1 447 439 1 447 469 1 568 1563 1 258 256 1 267 260 1 315 310 1 430 427 1 430 427 1	338	
(363)	16QAM	1024	405	Minimum 261 270 290 330 423 460 503 1771 259 268 287 325 398 439 469 1563 256 260 280 310 427 256 260 280 310 256 260 280 310 256 260 280 310 368 399 427 312 258 260 270 270 270 298 346 371 394 1113	415
(363)		1280	447	439	454
		1518	474	469	485
		9216	1568	1563	1574
		64	258	256	284
		128	267	260	290
		256	283	280	300
20	700.004	512	315	310	321
(363)	52QAM	1024	371	368	378
		1280	404	399	411
		1518	430	427	436
		9216	1315	1312	1321
		64	266	258	287
		128	262	260	287
		256	276	270	300
20	(404)	512	302	rerage Minimum 267 261 279 270 299 290 346 330 433 423 480 460 508 503 1775 1771 263 259 272 268 290 287 330 325 405 398 447 439 474 469 1568 1563 258 256 267 260 283 280 315 310 371 368 404 399 430 427 1315 1312 266 258 262 260 276 270 302 298 351 346 376 371 397 394	316
(363)	04QAM	1024	351	346	365
		1280	376	371	390
		1518	397	394	401
		9216	1115	1113	1120



Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	265	Latency [μs] Minimum 257 257 269 291 332 354 373 991 255 256 266 286 322 341 359 903 258 256 266 283 316 334 349 835 258 255 264 281 311 327 341	285
		128	259	257	281
		256	271	269	283
20	1200414	512	294	291	305
(363)	128QAM	1024	Latency [µs]AverageMinimum265257259257259257271269294291337332358354376373994991262255258256269266288286325322344341361359905903266258259256269266286283319316337334351349839835267258258255271264283281313311329327343341789786	348	
	ModulationFrame size [bytes]AverageModulation642651282562711280AM5122941280AM10243371280AM102433712803581280358128035812803581280358128035812803581280256269128034412803441280344128036112803511280351128035112803511280351128035112803511280351128035112812561280351128035112803511280351128035112803511280351	354	369		
		1518	376	373	379
		9216	994	991	997
		64	262	255	272
		128	258	256	276
		1518 376 373 9216 994 991 64 262 255 128 258 256 256 269 266 512 288 286 1024 325 322 1280 344 341 1518 361 359 9216 905 903 64 266 258 128 259 256	278		
20	25/0414	512	Average Minimum 64 265 257 128 259 257 256 271 269 512 294 291 1024 337 332 1280 358 354 1518 376 373 9216 994 991 64 262 255 128 258 256 256 269 266 512 288 286 1024 325 322 128 258 286 1024 325 322 128 259 266 512 288 286 1024 325 322 1280 344 341 1518 361 359 64 266 258 128 259 256 256 269 266 512 286 283	298	
20 (363)	256QAM	1024	325	322	335
		1280	344	341	355
		1518	361	359	364
		9216	905	903	908
		64	266	258	282
		128	259	256	275
		256	269	PrageMinimum6525759257712699429137332583547637394991622555825669266882862532244341613599590366258592566926686283513493513493523513493556725825825527126428328131331132932734334189786	284
20	5120AM	512	286	283	295
(363)	SIZQAM	1024	319	316	322
		1280	337	334	341
		1518	351	349	357
		9216	839	835	840
		64	267	258	280
		128	258	255	277
		256	271	264	285
20	102404M	512	283	281	291
(363)		1024	313	Latency [µs] Minimum 257 257 257 257 269 291 332 354 373 991 255 256 266 286 286 352 354 355 255 256 266 286 258 258 256 266 283 316 334 349 835 258 258 258 258 258 258 258 258 258 255 264 281 311 327 341 786	321
		1280	329	327	338
		1518	343	341	347
		9216	789	786	791

Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	266	258	279
		128	260	255	277
		256	Average Minimum 64 266 258 128 260 255 256 266 264 512 281 279 1024 311 309 1280 327 325 1518 340 338 9216 775 769 64 265 257 128 262 259 256 267 263 512 280 279 1024 309 307 128 262 259 256 267 263 512 280 279 1024 309 307 1280 324 322 1518 337 335 9216 750 747 64 217 213 1280 262 258 512 322 318 1024 437 432	283	
20	20/80AM	512	281	279	289
(363)	20400711	1024	311	309	318
		1280	266 258 260 255 266 264 281 279 311 309 327 325 3 340 38 340 5 773 265 257 265 257 265 257 265 257 262 259 267 263 280 279 4 309 307 367 267 263 280 279 4 309 307 335 6 750 747 213 3 232 322 318 4 437 437 432 0 498 492 228 553 549 6 2292 213 209 325 222 <td>335</td>	335	
		1518	340	Minimum 258 255 264 279 309 325 338 769 257 263 279 307 325 338 769 257 259 263 279 307 322 335 747 213 228 258 318 432 492 549 2289 209 209 209 222 248 300 398 450 498 1992	344
		9216	773	769	774
		64	265	257	277
		128	262	259	278
		256	267 263 280 279	282	
20	20480AM I	512	280	Pe Minimum P 258 2 255 2 264 2 279 2 309 2 309 3 309 3 338 1 325 3 338 1 257 2 257 2 257 2 257 2 259 2 263 2 279 2 307 2 307 2 335 1 228 2 213 2 228 2 318 2 492 2 209 2 209 2 209 2 300 3 300 3 300 3 300 3 300 3 <tr t=""></tr>	287
(363)	ZUHUQAME	1024	309	316	
(555)		1280	324	322	331
		1518	337	335	339
		9216	750	747	752
		64	217	213	240
		128	232	228	255
		256	262	Minimum 258 255 264 279 309 325 338 769 257 259 263 279 307 325 335 747 213 228 258 318 432 492 549 2289 209 222 248 300 398 450 498 1992	284
28	10AM S	512	322	318	345
(371)		1024	437	432	456
		1280	498	492	518
		1518	553	549	567
		9216	2292	2289	2312
		64	213	209	236
		128	225	222	249
		256	252	248	275
28	40AM	512	303	Latency [µs] Minimum 258 255 264 279 309 325 338 769 257 259 263 279 307 257 259 263 279 307 307 307 322 335 747 213 228 258 318 432 492 2492 2492 2492 318 432 492 289 209 209 209 300 398 450 498 1992	327
(371)	דערויו	1024	403	398	424
		1280	455	450	477
		1518	502	498	519
		9216	1995	1992	2014



Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	202	200	226
		128	210	208	226
	Promesize (prime in)ModulationPromesize (prime in)AverageMinimumS[Mh2] (prime in)ModulationAverageMinimumI1282012001128210200125622822512562282251102431931811024319318110243513481102435134811024351128312831024128512831102420110911024201109110243033011102430330111139113132811139114111391113911411139112802042012011281305303128320AH1284204201128021611411139112803053031112813262232381139412803053031128132632511139412803053031149412803053031151832612841011161112803053031179412803053031 </td <td>225</td> <td>243</td>	225	243		
28		274			
(371)	IOQAM S	1024	319	Latency [µs] Average Minimum 202 200 210 208 228 225 259 257 319 318 351 348 380 378 228 205 210 257 319 318 351 348 380 378 228 205 221 220 201 198 202 220 250 249 303 301 331 328 356 355 1141 1139 198 196 204 202 224 210 239 238 305 303 326 325 967 966 195 193 201 199 216 214 231	335
		Frame size (bytes)AverageMinimum6420220011282102082128210208125622822515122592571102431931811280351348115183803781921612851283110242011981128022222012562222201256222220110243033013128033132811280331328112803563551921611411139112802392381128030530311280305303115183263251151832632511518326325112802011931128020119311280201193112802011931128020119311280201193112802011931128020119311280201193112802011931128020119311280 </td <td>367</td>	367		
		1518	380	378	392
		9216	1285	1283	1301
		64	201	198	221
		128	208	205	232
		256	222	Latency [µs] Minimum 200 208 208 225 257 257 318 318 348 378 1283 1283 205 205 210 220 203 204 205 205 205 206 207 208 209 201 202 203 301 328 301 328 301 196 202 210 210 210 210 210 210 210 210 210 2210 2210 221 2301 2214<	235
28	160AM	512	250		263
(371)	IUQAM	1024	303		316
28 (371) 16QAM 11 12 12 12 12 12 12 12 12 12 12 12 12		1280	331	328	344
	1518	356	355	367	
		9216	1141	1139	1155
		64	198	196	220
		128	204	202	223
		256	224	210	240
28	320AM	512	239	238	265
(371)	JEQAM	1024	282	281	293
		1280	305	303	316
		1518	326	325	334
		9216	967	966	979
		64	195	193	211
		128	201	199	217
		256	216	214	237
28	640AM	512	231	229	252
(371)	UHUAIM	1024	266	265	274
		1280	286	284	294
		1518	303	301	309
		9216	829	828	838

Physico	ıl mode		Latency [µs]		-
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	202	196	220
	Physical mode Prame size CS [MHz] (frame ID) Modulation Prame size [1280,AM] [1280,AM] [1280,AM] [1280,AM] [28] (371) [1280,AM] [1280,AM] [1280,AM] [1280,AM] [28] (371) [292,23] [1280,AM] [1280,AM] [1280,AM] [1280,AM] [28] (371) [2560,AM] [201,23] [1280,AM] [1280,AM] [1280,AM] [28] (371) [2560,AM] [1280,AM] [1280,AM] [1280,AM] [1280,AM] [28] (371) [351,20,AM] [351,20,20,20,20,20,20,20,20,20,20,20,20,20,	128	199	198	213
		256	Average Minimum 202 196 199 198 221 207 225 224 256 255 273 272 288 287 205 196 198 196 205 196 205 196 198 196 205 196 205 196 205 196 205 204 205 204 204 204 205 263 204 204 205 263 206 265 207 196 210 218 204 199 197 196 219 218 244 242 258 256 261 269 637 636 203 198 20	226	
28	1290AM	512	225	Minimum 196 198 207 224 255 272 287 744 196 207 224 255 272 287 287 248 204 204 204 205 203 218 203 218 242 256 203 218 242 256 263 218 242 256 269 198 198 198 202 215 238 251 262 602	242
(371)	IZOQAM	1024	256	255	263
		1280	273	Minimum 196 198 207 224 255 272 287 744 196 197 287 744 196 204 205 272 287 744 196 204 205 248 263 276 682 199 196 203 218 242 256 263 199 196 203 218 242 256 269 636 198 198 202 215 238 251 262 602	280
		1518	288	287	294
		9216	745	744	752
		64	205	196	217
		128	198	196	210
		256	128 198 196 256 216 204 512 221 220 1024 249 248 1280 265 263 1518 278 276	222	
28	2540AM	512	221	220	236
(371)	ZJUQAM	1024	249	Average Minimum 202 196 199 198 221 207 225 224 256 255 273 272 288 287 745 744 205 196 198 196 216 204 221 220 249 248 265 263 278 276 684 682 204 199 197 196 214 203 215 256 265 263 278 276 684 682 204 199 197 196 214 203 215 256 271 269 637 636 202 198 211 202 263 198 21	254
28 (371)	28 (371) 256QAM 128 198 256QAM 512 221 1024 249 1024 249 1280 265 1518 278 9216 684 64 204 128 197 256 214 512 221	1280	265	263	270
		276	282		
		9216	684	682	691
		64	204	199	213
		128	197	196	208
		256	214	203	219
28	5120AM	512	219	218	231
(371)	JIZQAM	1024	244	242	248
		1280	258	256	262
		1518	271	269	275
		9216	637	636	643
		64	202	198	211
		128	203	198	213
		256	211	202	215
28	10240AM	512	216	215	228
(371)	1027QAN	1024	239	238	243
		1280	253	251	256
		1518	264	262	267
		9216	603	602	608



Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	202	198	210
		128	202	198	212
		256	211	202	215
28	28 (371) 2048QAM 256 211 202 512 216 215 216 1024 238 237 1280 251 250 1518 262 261 9216 591 589 9216 591 589 128 202 198 28 2048QAM L 512 215 214 128 202 198 201 197 28 2048QAM L 512 215 214 1024 236 235 214 1024 236 235 214 1024 236 235 214 1024 236 235 214 118 259 258 258 9216 576 575 575 128 203 197 256 210 201	227			
(371)	2040QAM	rtionFrame size (bytes)AverageMinimum420219821282021982128202198251221621521024238237212802512502128025125021280251259212802512592921659158921280202198225621120121280249248212802492482128024924821280249248212802492482128024924821280249248212802492482128024924821280211201124M1282831972128024724621280247246212802472462128020820234ML1282841971282841972128024423523412802442352341280244236265128024428019712802462665651280266266 </td <td>242</td>	242		
		1280	251	250	255
		1518	262	261	265
		9216	591	Latency [μs] Minimum 198 198 202 215 237 250 261 589 197 198 201 214 235 248 258 575 197 197 248 258 575 197 197 248 258 575 197 197 201 213 234 256 565 565 565 197 197 201 213 234 246 256 565 202 212 233 245 254 553	596
		64	201	197	209
		128	202	198	211
		256	211	Latency [µs]AverageMinimum20219812021981202198121120212162151251250126226115915891202198120119712152141215235125925812592581201197120319712142131215234120319712142131215256565200197120325612041971205254233213212225625422562542	214
28	20490 004	512	215		225
(371)	2048QAM L	1024	ize [bytes] Average Minimum 64 202 198 1 54 202 198 1 28 202 198 1 56 211 202 1 56 211 202 1 64 203 237 1 64 238 237 1 64 251 250 1 64 261 197 1 216 591 589 1 64 201 197 1 28 202 198 1 56 211 201 1 64 201 197 1 62 215 214 1 64 201 197 1 518 256 210 201 1 12 214 213 1 1 64 203 197 1 1<	240	
(371)		1280	249	248	253
		1518	259	258	260
		9216	576	575	581
		64	201	197	208
		128	203	197	210
		256	210	Lotency [µs] Minimum 198 198 202 215 237 250 261 250 261 589 197 197 198 201 214 235 248 258 575 197 197 201 213 248 258 575 197 197 201 213 234 246 256 565 197 197 201 213 234 245 256 565 197 197 201 213 234 245 256 565 197 197 201 213 234 245 256 565 197 197 201 213 234 245 256 565 197 197 201 213 234 245 256 565 197 197 201 213 234 245 256 565 197 256 256 256 256 256 256 256 256	213
28	490604M	512	214	213	224
(371)	4090QAM	1024	235	234	238
		1280	247	246	251
		1518	258	256	260
		9216	566	565	571
		64	200	197	208
		128	204	197	209
		256	208	202	212
28	1006041	512	213	212	223
(371)	4090QAM L	1024	234	233	237
		1280	246	245	249
		1518	256	254	258
	9216	554	553	559	

Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes] 64	Average	Minimum	Maximum
		64	204	Editeity (µs) Minimum 200 214 243 299 406 462 515 2144 197 209 406 462 515 2144 197 209 234 280 375 420 467 1867 188 196 211 241 299 329 356 1205 186 193 200 233 283 310 333 1071	226
		128	219	214	241
		256	size [bytes] Average Minimum 64 204 200 1 128 219 214 1 256 247 243 1 512 302 299 1 1024 411 406 1 1280 467 462 1 1280 467 462 1 1280 2147 2144 1 64 200 197 1 128 212 209 1 256 237 234 1 128 212 209 1 128 212 288 280 1 128 471 467 1 1 128 471 467 1 1 9216 1869 1867 1 1 128 198 196 1 1 128 198 196 1 1	266	
30	404145	512	302	Minimum 200 214 243 299 406 462 515 2144 197 209 234 237 280 375 420 467 188 196 211 241 299 329 356 1205 186 193 200 233 283 310 333 1071	322
(364)	4QAM S	1024	411	406	428
		1280	467	Latency [µs] Average Minimum 204 200 219 214 247 243 302 299 411 406 467 462 519 515 2147 2144 200 197 212 209 2137 234 288 280 378 375 427 420 471 467 1869 1867 190 188 198 196 214 211 244 241 300 299 331 329 258 356 1207 1205 188 186 195 193 213 200 235 233 213 200 235 233 310 310 335 <td>485</td>	485
		1518	519		534
		9216	2147		2166
		64	200	197	224
		128	212	209	236
		256	237	234	253
30	40414	256 237 234 512 288 280 1024 378 375 1280 427 420 1518 471 467	310		
(364)	4QAM	1024	378	375	397
(304)		1280	427	420	450
		1518	471	467	475
		9216	1869	1867	1874
		64	190	188	212
		128	198	196	212
		256	214	Editency (µs) Minimum 200 214 243 299 406 462 515 2144 197 209 234 280 375 420 467 1867 188 196 211 241 299 329 356 1205 186 193 200 233 283 310 333 1071	228
30	1/0004	512	244	241	258
(364)	I6QAM S	1024	300	299	315
		1280	331	329	345
		1518	258	356	360
		9216	1207	1205	1209
		64	188	186	200
		128	195	193	207
		256	213	200	240
30	160414	128 219 214 256 247 243 512 302 299 1024 411 406 1280 467 462 1518 519 515 9216 2147 2144 64 200 197 128 212 209 256 237 234 512 288 280 1024 378 375 1280 427 420 1518 471 467 9216 1869 1867 64 190 188 128 198 196 256 214 211 512 244 241 1024 300 299 1280 331 329 1518 258 356 9216 1207 1205 64 188 186 128 195 <t< td=""><td>240</td></t<>	240		
(364)		1024	285	283	290
		1280	311	310	317
		1518	335	333	337
		9216	1073	1071	1079



Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	196	189	214
		128	192	190	201
		256	211	202	228
30	720.414	512	225	224	234
(364)	Physical modePrame size (byte state)Hz] (128)1281281281024102410241024102410241024102410241024102410241151810241280AM1280AM1280AM1280AM1280AM1280AM1280AM12801518121012801518151216412801518151216412801518 <td>1024</td> <td>266</td> <td>265</td> <td>270</td>	1024	266	265	270
		1280	288	286	297
		1518	307	306	309
		9216	911	Latency [µs] Minimum 189 190 202 224 265 286 306 909 187 197 216 249 267 284 781 186 186 195 211 240 256 271 186 195 211 240 256 271 702 187 193 207 234 248 193 207 234 248 261 644	916
		64	192	187	207
		128	189	187	203
30 (364)		256	213	197	219
	(404)4	512	217	216	224
(364)	64QAM	1024	250	249	258
		1280	269	267	276
		1518	285	284	290
		9216	782	Latency [µs] Minimum 189 190 202 224 265 286 306 909 187 187 197 216 249 267 284 781 186 195 211 240 256 271 702 187 185 193 207 234 248 261	790
		64	189	186	207
		128	187	186	200
30 (364) 30 (364) 30 (364) 30 (364)		256	206	195	213
30	1200404	512	212	211	218
(364)	128QAM	1024	241	240	247
		1280	258	256	264
		1518	272	Minimum 189 190 202 224 265 286 306 909 187 197 216 249 267 284 197 267 284 781 186 195 211 240 256 271 702 187 240 256 271 702 271 702 256 271 702 187 193 207 234 248 261 261 644	274
		9216	703	702	296
		64	194	187	203
		128	186	185	205
		256	204	193	209
30		512	208	207	222
30 (364) 30 (364) 30 (364) 30 (364)	250QAM	1024	235	234	240
	hysicul modeFrame size [bytes]Image: Image: I	255			
		1518	262	261	266
		9216	646	644	652

Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	192	186	201
		128	192	188	203
		256	200	191	120
30	E120AM	64 192 186 128 192 188 256 200 191 512 206 205 1024 230 229 1280 243 242 1518 255 254 9216 602 601 64 190 186 128 191 187 256 200 190 512 208 198 1024 226 225 1280 238 237 1518 249 248 9216 570 569 1280 238 237 1518 249 248 9216 570 569 64 190 186 128 192 186 128 192 186 128 192 186 1280 237 236 1280 237 236	218		
(364)	SIZQAM	1024	230	Latency [µs] Minimum 186 188 191 205 229 242 254 601 186 187 190 198 225 237 248 569 186 186 190 225 237 248 569 186 186 186 186 186 186 186 186 186 186 186 186 185 185 185 186 189 201 222 234 244 544	234
		512 206 205 1024 230 229 1280 243 242 1518 255 254 9216 602 601 64 190 186 128 191 187 256 200 190 512 208 198 1024 226 225 1280 238 237 1518 249 248 9216 570 569 1280 238 237 1518 249 248 9216 570 569 64 190 186 128 192 186 128 192 186 128 192 203 202 1024 225 223 1 1280 237 236 1 1280 237 236 1 1280 237 236<	248		
		1518	255	254	258
		9216	602	Minimum 186 188 191 205 229 242 254 601 186 187 190 198 225 237 248 569 186 186 190 225 237 248 569 186 186 186 186 186 186 186 186 186 186 186 186 201 222 234 244 244	608
		64	190	186	198
		128	191	187	200
		256	200	190	203
30	102404M	512	208	198	211
30 (364)	1024QAM	1024	226	225	230
		1280	238	237	242
		1518	249	248	251
		9216	570	569	575
		64	190	186	198
		128	192	186	199
		256	199	Minimum 186 188 191 205 229 242 254 601 186 187 190 198 225 237 248 569 186 186 186 237 248 569 186 186 186 186 186 187 202 223 236 248 558 185 185 186 187 223 236 246 558 185 186 189 201 222 234 244 544	202
30	204904M	512	203	202	213
(364)	2048QAM	1024	225	223	228
		1280	237	236	240
		1518	247	246	249
		9216	559	558	564
		64	189	185	197
		128	191	186	198
		256	198	189	202
30	20490414	512	202	201	212
(364)	2040QAM L	1024	223	222	226
		1280	235	234	238
		1518	245	244	247
		9216	545	544	550



Physical mode			Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	189	185	196
		128	191	186	197
		256	197	189	202
30	400/0414	512	201	200	211
(364)	4096QAM	1024	Jutes Latency [µs] Average Minimum 189 185 191 186 191 186 201 200 222 221 233 232 243 242 189 185 189 185 192 186 193 185 199 185 199 185 199 185 191 186 192 186 193 201 201 200 221 219 232 231 241 240 255 523 188 185 191 185 192 200 193 230 231 230 231 230 231 230 240 239 518 517	225	
		1280	233	232	237
		1518	243	242	245
		9216	536	Latency [µs] Minimum 185 186 187 200 221 232 242 535 185 185 185 185 185 200 219 219 219 231 240 523 185 185 185 185 185 185 230 231 240 252 316 358 398 1628	541
		64	189	185	195
		128	192	186	197
		256	196	189	200
30	400/0414	128 192 186 256 196 189 512 201 200 1024 221 219 1280 232 231 1518 241 240	210		
30 (364)	4096QAM L	1024	221	219	223
		1280	232	231	235
		1518	241	240	243
		9216	525	523	529
		64	188	185	195
		128	191	185	196
		256	196	Minimum 185 186 189 200 221 232 242 535 185 185 185 185 185 185 185 186 189 200 219 231 240 523 185 185 185 185 185 185 185 185 185 185 185 185 185 185 185 185 189 191 230 231 158 169 191 235 316 358 398 1628	200
30	010204M	512	200	199	209
(364)	8192QAM	1024	220	218	222
		1280	231	230	321
		1518	240	239	242
		9216	518	517	522
		64	161	158	185
		128	172	169	194
		256	194	191	216
40	40446	512	237	235	258
(365)	4QAM 5	1024	319	316	340
		1280	362	358	383
		1518	401	398	404
		9216	1630	1628	1653

Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes] 64	Average	Minimum	Maximum
		64	158	Minimum 155 165 184 221 292 329 362 1421 149 155 160 192 235 280 930 152 153 164 185 223 244 263 830 151 160 178 209 227 242 710	182
		128	168	165	1186
		Frame size [bytes]AverageMinimumI64158155112816816512561861841512223221110242942921128033132911518365362192161423142116415114911281571551512193192112815736211281573561128237235110242372351128026025811518281330115181551531152172164112802252231128024624411518264263112801521511128115215111282152151112841521511128517116011280228227112802282271128022822711280243242112802282271128022822711280228227112802282271128	205		
40	40414	512	223	Minimum 155 165 184 221 292 329 362 1421 149 155 160 192 235 280 930 152 153 164 185 223 244 263 830 151 160 178 209 227 242 710	242
(365)	4QAM	1024	294	292	313
		1280	331	Latency [µs] nge Minimum 8 155 8 165 6 184 3 221 4 292 1 329 5 362 23 1421 1 149 7 155 3 160 3 192 7 235 0 258 1 280 1 280 1 280 1 280 1 280 1 280 1 280 1 280 1 280 1 280 1 280 1 280 1 280 1 164 6 185 5 223 6 244 263 151 2 <t< td=""><td>350</td></t<>	350
		1518	365		380
		9216	1423		1443
		64	151	149	176
		128	157	155	178
		256 173 160 512 193 192	190		
40	1/01110	512	158 155 168 165 186 184 223 221 294 292 331 329 365 362 1423 1421 157 155 173 160 193 192 237 235 260 258 281 280 931 930 158 152 155 153 172 164 186 185 225 223 246 244 264 263 832 830 155 151 155 151 152 151 155 151 152 151 153 151 154 160	203	
(365)	16QAM S	1024	237	165 184 221 292 329 362 1421 149 155 160 192 235 258 280 930 152 153 164 185 223 244 263 830 151 160	247
		1280	260	258	271
		1518	281	280	290
		9216	931	930	943
		64	158	152	176
		128	155	153	172
		256	172	Litency [µs] Minimum 155 165 184 221 292 329 362 1421 149 155 160 192 235 258 280 930 152 153 164 185 223 244 263 830 151 160 178 209 227 242 710	190
40	1/0114	512	186	185	209
(365)	16QAM	1024	225	223	233
		1280	246	244	254
		1518	264	263	271
		9216	832	830	841
		64	155	151	169
		128	152	151	167
		256	171	160	181
40	720 414	512	179	178	198
(365)	52QAM	1024	211	209	217
		1280	228	227	235
		1518	243	242	249
		9216	711	710	719



Physical mode			Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	158	151	168
		128	151	149	167
		256	168	157	173
40	(404)4	512	178	164	180
(365)	64QAM	1024	Latency [µs]AverageMinimum158151151149151149168157178164199198214213227226615614155151156152164155164155164155169168193192206205217216557556153150162154162154163150164153155150154513155150155150155150161153165164184183195194204203482481	205	
		1280	214	213	219
		1518	227	226	231
		9216	615	Latency [μs] Minimum 151 149 157 164 198 213 226 614 155 168 192 205 216 556 150 154 166 187 199 209 513 150 153 150 153 164 187 199 209 513 150 153 164 183 194 203	622
		64	155	151	164
		128	156	152	166
		256	164	155	169
40	1200404	128 151 149 256 168 157 512 178 164 1024 199 198 1280 214 213 1518 227 226 9216 615 614 9216 615 614 64 155 151 128 156 152 256 164 155 512 169 168 1024 193 192 1280 206 205 1518 217 216 9216 557 556 1512 169 168 1024 193 150 9216 557 556 1512 167 166 128 155 150 256 162 154 512 167 166 1024 188 187 1280 200 <td< td=""><td>181</td></td<>	181		
40 (365)	128QAM	1024	193	192	197
		1280	206	205	126
		1518	217	216	220
		9216	557	556	563
		64	153	150	161
		128	155	150	163
		256	162	Minimum 151 151 149 157 164 198 213 226 614 151 152 155 168 192 205 216 556 150 154 150 154 150 154 150 150 150 151 150 150 150 151 150 150 150 150 150 150 150 150 150 150 150 153 164 183 194 203 481	167
40	25404M	512	167	166	177
(365)	256QAM	1024	188	187	198
		1280	200	199	203
		1518	210	209	212
		9216	514	513	520
		64	153	150	160
		128	155	150	161
		256	161	153	165
40	5120AM	512	165	164	174
(365)		1024	184	Latency [μs] Minimum 151 149 157 164 198 213 213 151 151 157 164 198 213 213 151 152 151 152 155 168 192 205 216 556 150 150 150 150 150 150 150 150 150 150 150 150 150 150 150 150 150 153 164 183 194 203 481	187
		1280	195	194	198
		1518	204	203	207
		9216	482	481	486

Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	151	149	157
		128	155	149	158
		256	159	153	163
40	10240414	512	Average Minimum 64 151 149 128 155 149 256 159 153 512 163 162 1024 181 180 1280 192 191 1518 200 199 9216 459 457 64 152 149 128 154 149 256 159 153 64 152 149 128 154 149 256 159 153 512 163 162 1024 180 179 128 151 162 192 163 162 1024 180 179 1280 191 190 1518 199 188 9216 450 449 64 152 149 1280 189 188	171	
(365)	1024QAM	1024	181	Minimum 149 149 149 153 162 180 191 199 457 149 153 162 180 191 199 457 149 153 162 179 190 198 449 149 149 153 161 178 188 196 439 149 149 153 161 178 188 196 439 149 149 149 149 149 149 149 149 149 149 149 149 153 <	184
		1280	Average Minimum 64 151 149 128 155 149 256 159 153 512 163 162 1024 181 180 1280 192 191 1518 200 199 9216 459 457 64 152 149 128 164 149 256 159 153 512 163 162 128 154 149 256 159 153 512 163 162 1024 180 179 1280 191 190 1518 199 198 9216 450 449 64 152 149 128 153 153 512 162 161 1024 179 178 1280 189 188	194	
		InterpretationInterpretationulationFrame size (bytes)AverageMinimum6415114911281551491256159153151216316211024181180112801921911151820019919216459457164152149112801541491256159153151216316211280191190115181991981921645044911280191190115181991981921645044911280158153149128115314911280189188112801891881128018918811280158153149641521491128018918811280158153164152149112801581531641521491128015815316415214911280158153164152149112	202		
		9216	459	Latency [µs] Minimum 149 149 149 153 162 180 191 199 457 149 149 149 149 149 153 162 179 190 198 449 149 198 449 149 198 449 149 198 449 198 449 198 449 198 449 198 449 198 449 198 449 198 449 198 449 198 449 198 449 198 449 198 449 198 449 198 449 198 449 198 198 449 198 198 198 198 198 198 198 19	337
		64	152	149	157
		128	154	149	158
40 (365)		256	159	153	162
	20400414	512	163	162	170
	2048QAM	1024	180	179	183
		1280	191	190	193
		1518	199	198	201
		9216	450	449	454
		64	152	149	157
		128	153	149	158
		256	158	Minimum 149 149 149 153 162 180 191 199 457 149 153 162 179 149 149 149 149 149 149 149 153 162 179 190 198 449 149 149 149 149 149 149 149 153 161 178 188 196 439 149 149 149 149 149 149 149 149 149 149 149 149 153 <	162
40	204004041	512	162	161	169
(365)	2048QAM L	1024	179	178	181
		1280	189	188	192
		1518	197	196	198
		9216	440	439	444
		64	152	149	156
		128	154	149	157
		256	158	153	161
40	40 (365)2048QAM5121631621024180179128019119015181991989216450449921645044992161531492048QAML1281532048QAML512162102417917812801891881518197196921644043940512162151815414925615815315181971969216440439405121621611024179128154149256158153512162161102417917712801881871518196195	161	169		
(365)	4090QAM	1024	179	177	181
		1280	188	187	190
		1518	196	195	197
		9216	434	432	189


Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	152	148	155
		128	153	149	156
		256	157	152	161
40	400/0414	512	162	160	168
(365)	4090QAM L	1024	178	176	179
		1280	187	186	121
		1518	195	194	196
		9216	425	Latency [µs] age Minimum 2 148 3 149 3 149 3 149 7 152 2 160 8 176 92 160 18 176 17 186 95 194 95 194 95 194 95 194 95 194 92 148 92 148 92 148 92 149 97 152 91 160 97 176 96 185 94 193 90 150 91 138 90 292 91 1327 91 1327 91 127 92 140 93 140	428
		64	152	148	155
		128	152	149	156
		256	157	152	160
40	01020404	512	161	160	167
(365)	8192QAM	1024	177	176	179
		1280	186	185	188
		1518	194	193	195
		9216	420	Latency [µs] Minimum 148 149 152 160 176 186 194 424 148 149 194 424 148 149 152 160 176 160 176 185 193 419 152 160 176 185 193 419 129 138 150 193 419 129 138 150 190 256 292 322 1327 127 127 127 127 127 127 127 1	423
		64	133	129	158
		128	142	138	161
		256	160	150	200
50	404445	512	195	190	215
(366)	4QAM 5	1024	264	256	281
		1280	300	292	315
		1518	327	322	336
		9216	1331	1327	1337
		64	131	127	156
		128	138	130	160
		256	153	140	175
50	404M	512	184	180	201
(366)	4QAM	1024	242	237	252
CS [MHZ] (Frame ID) Modula 40 (365) 4096QA 40 (365) 8192Q 50 (366) 4QAN 50 (366) 4QAN		1280	272	267	283
		1518	298	Minimum 148 149 152 160 176 186 194 424 148 149 152 160 176 186 194 424 148 149 152 160 176 185 193 419 129 138 150 190 256 292 322 1327 127 130 140 180 237 267 294 1161	305
		9216	1163	1161	1168

Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	133	127	150
		128	129	127	147
	Physicul modeFrame size [b]Hz] (10)Modulation64128128160AM S128010241024102412801160AM S641181281160AM S641181281160AM S641181281160AM S12801160AM S12801160A	256	139	137	142
50	1604 M S	512	InterestingLatency [µs]AverageMinimumAverage6413312712812912712812913725613913751215815610241941921280215211151823022792167637616413612712812812525613913712812812515121531511024185183128020320015182162149216682678641331261281391315121471461280138186151820019892165825796413012512813112512813712951214214112801371295121421411280137129512142141128013712951214214112801371295121421411280137129512142141128013712951214214112801371295121421411280137 </td <td>168</td>	168	
(366)	TOQAM S	1024		204	
		Frome size (bytes)AverageMinimum6413312711281291271256139137151215815611024194192111802152111151823022719216763761164136127112812812812512812813111281282001128215151112821667811280203200112802166781128013012711281130127112811301271128113012711281130127112811391311128018818611280188186112801311251128013712911280137129112801371291128013712911280137129112801371291128013712911280137129112801371291128013712911280 <td< td=""><td>224</td></td<>	224		
	16QAM S 512 158 156 1024 194 192 1280 215 211 1518 230 227 9216 763 761 9216 763 127 128 128 125 128 128 125 128 128 125 16QAM 512 153 151 16QAM 1024 185 183 16QAM 64 133 200 16QAM 64 133 125 16QAM 64 133 126 1280 203 209 1518 1280 216 6182 678 128 130 127 256 139 131 512 147 146 1024 173 171 1280 188 186 186 186 1518 200 198 198 198 </td <td>227</td> <td>232</td>	227	232		
		9216	763	761	766
		64	136	127	143
		128	128	125	148
		256	139	137	156
50	1/04/4	512	139 137 153 151 185 183 203 200	161	
(366)	16QAM	1024	185	183	187
		1280	203	200	210
		1518	216	214	218
		9216	682	678	684
		64	133	126	144
		128	130	127	142
		256	139	Minimum 127 127 137 156 192 211 227 761 127 125 137 151 183 200 214 678 126 127 131 146 171 186 198 579 125 127 131 146 171 186 198 579 125 129 141 163 175 185 502	145
50	320AM	512	147	146	149
(366)	52QAM	1024	173	171	176
		1280	188	186	195
		1518	200	198	202
		9216	582	579	584
		64	130	125	138
		128	131	125	139
		256	137	129	140
50	50 (366) 32QAM 1280 203 50 (366) 9216 682 32QAM 64 133 128 130 256 139 512 147 1024 173 1280 188 1518 200 9216 582 64 130 1518 200 9216 582 64 130 1518 200 9216 582 64 130 128 131 256 137 50 (356) 64QAM 1024 164 1024 164 164	142	141	144	
(366)	U4QAM	1024	164	163	166
		1280	177	175	182
		1518	187	185	189
		9216	504	502	506



Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	128	125	135
		128	130	124	135
		256	136	127	139
50	1200 414	512	140	138	144
(366)	128QAM	1024	Latency [µs] Average Minimum 128 125 130 124 136 127 140 138 159 158 170 169 180 177 456 453 128 124 130 169 140 138 159 158 170 169 180 177 456 453 128 123 134 126 135 154 165 164 165 164 173 172 422 419 127 123 132 127 136 135 136 135 136 135 136 135 136 167 152 151 163 167 152	163	
	50 128QAM 50 256QAM (366) 256QAM 50 512QAM	1280	170	169	174
		1518	180	177	182
		9216	456	Latency [µs] Minimum 125 124 127 138 158 169 177 453 124 123 124 123 126 136 154 164 172 419 123 123 123 123 123 127 135 151 160 167 393 123 123 123 123 123 123 123 12	459
		64	128	124	133
		128	128	123	133
		64 128 124 128 128 123 256 134 126 512 138 136 1024 155 154 1280 165 164 1518 173 172	137		
50	25404M		146		
50 (366)	256QAM	1024	155	154	158
		1280	165	164	169
		1518	173	172	175
		9216	422	419	425
		64	127	123	130
		128	128	123	131
		256	Average Minimum 128 125 130 124 136 127 140 138 159 158 170 169 180 177 456 453 128 124 128 124 128 123 134 126 138 136 155 154 165 164 173 172 422 419 127 123 128 123 165 164 173 172 422 419 127 123 132 127 136 135 152 151 161 160 162 163 163 167 395 393 127 123 127 123 12	127	135
50	5120AM	512	136	Latency [μs] Minimum 125 124 127 138 158 169 177 453 124 123 126 136 154 164 172 419 123 123 127 135 151 160 167 393 123 123 123 124 125 126 135 151 160 167 393 123 124 125 125 126 133 148 157 164 374	143
(366)	SIZQAM	1024	152	151	156
		1280	161	160	164
		1518	168	167	170
		9216	395	393	398
		64	127	123	129
		128	127	123	131
		256	131	126	135
50	10240AM	512	135	133	141
(366)		Frame size (bytes)AverageMinimum641281251281301242561361275121401381024159158102415915811280170169151818017792164564531280128124128013812412811281241281281241281281241281281265121381361024155154102415515412801651641518173172921642241912812812312812812310241521511280161160151816816792163953936412712312812712312812712312812712312812712312813112651213513310241501481280158157151816516412801581571518165164128015815715181651641280158157151816516416	152		
		1280	158	Latency [µs] Minimum 125 124 127 138 158 169 177 453 124 127 138 158 169 177 453 124 123 126 136 154 164 172 419 123 124 123 124 172 419 123 124 125 127 135 151 160 167 393 123 123 123 123 124 125 123 124 125 126 133	161
		1518	165	164	167
	9216	376	374	379	

Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	126	123	129
		128	127	122	130
		256	130	126	134
50	20400414	512	135	133	141
(366)	2048QAM	1024	149	Latency [µs] je Minimum 123 122 126 133 126 133 148 156 163 368 122 122 123 148 156 163 163 368 122 122 126 133 127 126 133 147 155 161 155 161 122 122 122 122 122 122 125 161 155 161 155 161 155 161 155 161 155 161 155 161 155 161 155 161 154 122 125 132 145 154 159 159	151
		1280	Average Minimum 64 126 123 128 127 122 256 130 126 512 135 133 1024 149 148 1280 158 156 1518 164 163 9216 369 368 64 126 123 128 127 122 256 130 126 1518 164 163 9216 369 368 64 126 123 128 127 122 256 130 126 512 137 133 1024 148 147 1280 157 155 1518 163 161 9216 361 359 64 125 122 128 126 155 1518 162 161	160	
		1518	164	163	165
		9216	369	Litelity [µ3] Minimum 123 122 126 133 148 156 163 368 122 126 133 148 156 163 368 122 126 133 147 155 161 359 122 125 132 145 154 159 347 <td>372</td>	372
		64	126	123	128
		128	127	122	130
		256	130	126	133
50	50 (366) 2048QAM L 11 12 92	512	137	133	140
(366)		1024	148	147	150
		1280	157	155	159
		1518	163	161	164
		9216	361	359	364
		64	125	122	128
		128	126	122	129
		256	130	Minimum 123 122 126 133 148 156 163 368 122 123 148 156 163 368 122 122 125 161 359 122 122 122 123 147 155 161 359 122 122 122 122 122 122 122 122 122 122 122 122 122 122 122 122 122 125 132 145 154 159 347	133
50	400/0414	512	136	132	139
(366)	4096QAM	1024	147	146	149
		1280	156	155	158
		1518	162	161	163
		9216	356	354	358
		64	125	122	127
		128	126	122	129
		256	129	125	132
50	4006.04.04	512	136	132	138
(366)	4090QAM L	1024	147	145	149
		1280	155	154	157
		1518	161	159	162
		9216	348	347	351



Physico	ıl mode		Latency [µs]			
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum	
		64	122	119	144	
		128	130	127	144	
		256	146	144	149	
56	Comparing (Frame 10)ModulationFrame size [bytes]AverageMinimum[128139127128139127256146144512177175102423623411282692661518296294151829629415182961181118112811771281271242561411395651216816012216816112216816221616116221616116413021756131127561311275613112715182722699216103510345513112712813911725613112712801961951512146145160AMS162131160AMS162131151821121015182112101518211210151821121015182112101518211210151821121015182112101518211210151821121015182112101518211210	175	187			
(373)	4QAM S	1024	Latency [µs] Average Minimum 122 119 130 127 146 144 177 175 236 234 269 266 296 294 1185 1181 120 117 121 124 120 117 121 124 121 124 121 124 121 124 122 269 141 139 126 127 217 244 219 217 247 244 130 120 119 117 131 127 131 127 146 145 178 177 196 195 211 210 684 683 684 683 126	251		
		IndicionFrame size (bytes)AverageMinimumiulation6412211911281301271128130127125614614415121771751102423623411280269266115182962941921611851181112812712411281271241128127124112812712411281212171128024724411518272269112801035103411280119117125613112711280196195115182112101128019611911280131124160AH126119160AH12613112460AH12613112460AH12613112460AH12613112460AH12613112460AH12613112460AH12613112460AH12613112460AH12613112460AH12613112460AH <td< td=""><td>283</td></td<>	283			
		1518	296	294	299	
		9216	1185	Latency [μs] Minimum 119 127 144 175 234 266 294 1181 117 124 139 166 217 244 269 1034 120 117 127 145 177 195 210 683 119 116 124 185 198 612	1189	
		64	120	117	146	
		128	127	124	140	
		256	141	Latency [µs]AverageMinimum122119113012711301271146144117775123623412692661296118111201171127124114113911681661219217124724412722691130120113112711311271146145117817711781161126119113121011311241131124113112411311241131124113112411311241131124113112411311241131124113112411311241131124113919919813919811311241132149113312411461451157199198158199198159198 <td>153</td>	153	
56	404M	512	168		181	
(373)	4QAM	1024	219		232	
56 (373)		1280	247	244	260	
		1518	272	269	274	
		9216	1035	1034	1041	
		64	130	120	142	
		128	119	117	127	
		256	131	erage Minimum 122 119 130 127 146 144 177 175 236 234 269 266 296 294 185 1181 120 117 127 124 141 139 168 166 219 217 247 244 272 269 035 1034 130 120 119 117 131 127 146 145 178 177 196 195 211 210 584 683 126 119 118 116 131 124 142 140 170 169 187 185 199 198 613 612	147	
56	160AM S	512	146	145	153	
(373)	TOCAM 3	1024	178	177	186	
		1280	196	195	203	
		1518	211	210	217	
		9216	684	683	688	
		64	126	119	138	
		128	118	116	125	
		256	131	124	143	
56	1604M	512	142	140	147	
(373)	TOQAM	1024	170	Latency [μs] Minimum 119 127 144 175 234 266 294 1181 117 124 139 166 217 244 269 1034 120 117 127 244 269 1034 120 117 128 119 145 177 195 210 683 119 116 124 145 177 195 210 683 119 116 124 140 169 185 198 612	176	
		1280	187	185	192	
		1518	199	Latency [µs] Minimum 119 127 144 175 234 266 294 1181 117 124 139 166 217 244 269 1034 120 1034 120 117 127 145 177 195 210 683 119 116 124 140 169 185 198 612	204	
		9216	613	612	617	

Physico	ıl mode		Latency [µs]		_
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	123	118	131
		128	124	118	133
	Physicurode Frome size (bytes) Average Minimum 3) Modulation Average Minimum 128 123 118 1280 124 118 1280 124 118 1280 124 118 1280 132 122 110 1512 136 135 1280 174 172 1280 174 172 1518 185 183 1640 190 116 1518 185 183 1640 119 116 1640 119 116 1640 128 111 1280 164 162 1280A 164 116 1280A 116 116 1280A 128 121 116 1280A 128 128 128 1280A 128 128 128 1280A 1	136			
56	720414	AdulationFrame size [bytes]AverageMinimumI6412311811281241181128124118125613212211024160159111281741721128017417211280174172112801741601128017416019216527526112811281161128212811611284122116112841521511128415215111284115110112841211161128412111611284121116112841211161128412111611284128121116128412812611812841281261128412816416415181581561128412812812612841281281641518158156112841281281641518158156115212812812615315218153154128 <t< td=""><td>141</td></t<>	141		
(373)	52QAM	1024	160	159	164
		1280	174	Latency [µs] Prage Minimum 23 118 24 118 32 122 36 135 60 159 74 172 85 183 27 526 19 116 22 116 28 119 32 131 52 151 64 162 73 172 159 458 15 110 21 116 22 118 30 128 47 146 58 156 66 164 19 115 20 114 22 114 23 152 61 159 638 386	178
		1518	Average Minimum 123 118 124 118 132 122 136 135 4 160 159 0 174 172 8 185 183 6 527 526 119 116 3 6 527 526 119 116 3 6 527 526 119 116 3 6 527 526 119 116 3 128 119 1 132 131 4 152 151 0 164 162 3 115 110 3 121 116 3 121 116 3 121 116 3 121 116 3 123 124 118 14 130	188	
		9216	527	526	533
		64	119	116	127
		128	122	116	128
		256	128	119	129
56	(10114	512	AverageMinimum641231181281241182561321225121361351024160159128017417215181851839216527526641191161281221161281221165121321311281221161281221161281221161281221511281221511281521511281521511281151101281211161281211161281211161281211161281211161281211161281261281281261641518158156128120114128120114128124118128126124128126124128126124128126124128126124128126124128126124128126124128126124128126124128126124128126126128 <td>136</td>	136	
(373)	1280 1518 9216 64 128 64 128 512 64QAM 1024 1280 1518 9216 64 1024 1280 1518 9216 64 1280 1518 9216 64 1280 128 1280 1280 1280 1280	152	151	175	
(213)		1280	164	162	167
		1518	173	172	176
		9216	459	458	464
		64	115	110	121
		128	121	116	124
		256	121	Minimum 118 118 122 135 159 172 183 526 116 119 131 152 172 183 526 116 117 131 151 162 172 458 110 116 115 128 146 156 164 156 164 155 114 128 146 156 164 155 142 152 159 386	125
56	12004M	512	125	118	129
(373)	128QAM	1024	130	128	133
		1280	147	146	150
		1518	158	Minimum 118 118 122 135 159 172 183 526 116 117 183 526 116 117 183 526 116 117 131 151 162 172 458 110 116 115 118 128 146 156 164 115 118 128 146 156 164 155 164 156 164 157 158 159 386	160
		9216	166	164	168
		64	119	115	122
		128	120	114	123
		256	124	118	128
56	ModulationHuma bergers (Minimum)6412311812812411812812411825613212251213613510241601591280174172151818518392165275261280128116128012811912811221161282123131128016416215181731721280164162151817317292164594581280164116151817311612801211161280147146128014714615181581561280147146151815815612801281281280124118128012812612801281261280128126128012812612801281261280128126128012812612801281261280128126128012812612801281261280128126128012812612801281261280128126	135			
(373)	ZOUAM	1024	144	142	146
		1280	153	152	155
		1518	161	159	162
		9216	388	386	391



Physico	ıl mode		Latency [µs]			
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum	
		64	119	114	122	
		128	120	114	122	
Physical model CS [MHz] (Frame ID) 56 (373) 56 (373) 56 (373) 56 (373)		256	123	118	126	
56	Physicumode HtzjProme size [bytes]AverageMinimumHtzj e D0ModulationAverageMinimum1141141141281291141281201141281271265121271261024141140128015914912801591491518156155921636536392163653631024QAM1281181024QAM15121171121291241024QAM15181541024QAM15181541024QAM151811412801181141280118114128011811412801181141280118114128011813712801171131280116113128011611312801171131280116113128011711312801171131280117113128011611312801171131280117113128011611312801161131280116113128011611312801171131280116113 <t< td=""><td>129</td></t<>	129				
(373)	SIZQAM	1024	141	Latency [µs] rage Minimum 19 114 20 114 23 118 27 126 41 140 50 149 56 155 55 363 17 114 18 114 21 117 29 124 39 138 48 146 54 152 49 347 17 114 18 114 21 177 17 114 18 146 54 152 49 347 17 114 18 114 21 117 27 123 38 137 47 146 53 151 42 341 16 113	143	
		Frame size (bytes)AverageMinimum64119114128120114128120114256123118512127126102414114012801591491518156155921636536364117114128118114128118114128118114128118114128121117512129124102413913812801481461518154152921634934764117114128118114128121117512127123102413813712801471461518153151921634234112811711312812811713128117131281171413813612801461131280128123128112812813813614914615181521512128151315115141581515159151815615191591512128 <tr< td=""><td>152</td></tr<>	152			
		1518	156	155	158	
		9216	365	363	368	
		64	117	114	120	
		128	118	114	121	
		256	121	Latency [µs] Minimum 114 114 114 114 114 114 114 126 140 149 149 155 363 114 114 114 114 117 124 138 146 152 347 146 152 347 114 114 117 123 137 146 151 341 117 123 137 146 151 341 113 113 113 113 113 113 113 113 11	125	
56	102404M	512	129		131	
(373)	1024QAM	1024	Latency [µs]AverageMinimum119114120114121118127126141140150149156155365363117114118114121117129124139138148146154152349347117114118114118114154152349347117114118114118114121117123138138137147146153151342341116113117113120117128123138136146145152150335333	141		
(373)		1280	148	146	149	
		1518	154	152	155	
		9216	349	347	351	
		64	117	114	120	
		128	118	114	121	
		256	121	Littency [µs] Minimum 114 114 114 118 126 140 142 143 140 149 149 149 149 114 114 115 363 114 117 124 138 146 152 347 114 117 123 137 146 151 341 113 113 113 114 115 341 113 114 115 341 113 114 115 341 113 1145 150 333	124	
56	204804M	512	127	123	130	
(373)	2040QAM	1024	138	137	140	
		1280	147	146	148	
		1518	153	151	154	
		9216	342	341	345	
		64	116	113	119	
		128	117	113	120	
		256	120	117	124	
56	2049041	512	128	123	129	
(373)	2040QAM L	1024	138	Latency [µs] Minimum 114 114 114 114 114 114 114 114 114 115 140 140 140 140 140 140 140 140 140 140 155 363 114 117 124 138 146 152 347 114 117 123 137 146 151 341 113 113 113 113 113 113 1145 150 333	139	
		1280	146	145	147	
		1518	152	150	153	
		9216	335	333	338	

Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	116	113	119
		128	117	113	120
		256	121	117	124
56		512	125	123	129
(373)	4096QAM	eFrame size [bytes]AverageInternationalModulation641161128117112811712561211102413711280145115181511921633111281171128117112811711281171128117112812811281501128150112815911281591128125112812511281251128256139151828111282571151828111281201 <td>136</td> <td>138</td>	136	138	
		1280	145	144	146
		1518	151	149	152
		9216	331	Minimum 113 113 113 117 123 136 144 149 329 114 117 123 136 144 149 329 114 117 123 136 143 149 323 112 120 130 164 220 252 276 1124 116 117 130 156 204 230 253 985	333
		64	117	114	119
		128	117	114	120
		256	120	117	123
56		512	128	123	129
(373)	4096QAM L	1024	137	136	138
		1280	145	143	146
		1518	150	149	151
		9216	325	323	327
		64	115	112	140
		128	125	120	150
		256	139	Latency [µs] Minimum 113 113 117 123 136 144 149 329 114 149 329 114 117 123 136 143 143 143 143 149 323 136 143 143 149 323 136 143 143 149 323 136 143 149 323 136 143 149 323 112 120 130 164 220 252 276 1124 116 117 130 156 204 230 253 985	159
60	101110	512	169	164	180
(367)	4QAM S	1024	225	220	242
		1280	257	252	271
		1518	281	276	288
		9216	1127	1124	1132
		64	123	116	144
		128	120	117	145
		256	134	130	160
60		Average Minimu 64 116 113 128 117 113 256 121 117 512 125 123 1024 137 136 1280 145 144 1518 151 149 9216 331 329 64 117 114 128 117 114 128 117 114 256 120 117 512 128 123 1024 137 136 128 117 114 256 120 117 512 128 123 1024 137 136 1280 145 143 1518 150 149 9216 325 323 64 115 112 128 125 120 128 255 220	156	174	
(367)	4QAM	1024	209	204	223
		1280	236	230	248
		1518	256	253	264
		9216	988	985	991



Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	123	113	132
		128	114	111	127
		256	124	120	130
60		512	139	137	147
(367)	TOQAM 2	1024	169	167	177
		1280	186	184	194
		1518	200	198	202
		9216	654	Latency [µs] Minimum 1113 1111 120 137 167 184 198 651 112 114 118 133 159 174 187 584 111 115 128 159 174 187 584 111 112 115 128 150 163 159 174 187 584 111 112 115 128 150 163 159 163 163 173 501 110 110 110 110 110 110 110	656
		64	120	112	129
		128	118	114	132
60 (367)		256	126	118	133
	16QAM	512	135	133	141
		1024	161	159	168
		1280	176	174	183
		1518	188	187	190
		9216	585	584	588
		64	115	111	125
		128	117	112	126
	16QAM	256	119	115	130
60	720414	512	129	128	141
(367)	52QAM	1024	152	150	157
		1280	164	163	170
		1518	175	173	180
		9216	503	501	506
		64	114	110	121
		128	116	110	122
		256	120	113	125
60	6404M	512	125	124	129
)367)	04QAM	de ModulationFrame size (bytes)AverageMinimum6412311312811411112612412051213913710241691671280186184151820019892166546519216654651128011811415182061121280186184151213513316QAM51215516QAM6441201518188114128017611816QAM16115916QAM51213516QAM51213516QAM64161151818818716QAM51215816QAM64115110241611591280176174151818818715121291281641521591518164163151817517392165835911280116110128011611312801161161280164163151816416412801551441280155144128015514412801551441280155144128	143	148	
		1280	155	Latency [µs] rage Minimum 23 113 24 111 24 120 39 137 39 137 39 167 36 184 30 198 34 651 20 112 18 114 26 118 35 133 31 159 76 174 38 187 35 584 15 111 17 112 19 115 29 128 32 501 14 110 16 110 20 113 25 124 34 162 35 144	160
		1518	164	162	168
		9216	438	435	441

Physico	ıl mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	113	109	118
		128	114	109	118
		256	119	113	123
60	1200414	Frame size (bytes)AverageMinimum641131091128114109125611911315121231221102413913811280150148115123983961921639839612561171121151212312019216398396112801141091128114109112811410911281131201128014614411518153151112801461141128014611211280116112112801421411151212211911280142141115181491471151814914711518149147112801421411081280142141115181491471128014214111518149147115181491471151814914711518149147115181491411081518 <td>131</td>	131		
(367)	128QAM	1024	139	Average Minimum 113 109 114 109 119 113 123 122 139 138 150 148 157 156 398 396 113 109 114 109 1150 148 157 156 398 396 113 109 114 109 117 112 123 120 137 135 146 144 153 151 370 368 112 108 113 108 114 133 142 141 143 133 142 141 149 147 348 346 111 108 112 108 115 111 1	143
		Frame size [bytes] Average Minimum 64 113 109 1 128 114 109 1 256 119 113 1 512 123 122 1 1024 139 138 1 1280 150 148 1 1280 150 148 1 1280 150 148 1 1518 157 156 1 1280 114 109 1 128 114 109 1 128 117 112 1 1024 137 135 1 1280 146 144 1 1280 146 144 1 1280 146 141 1 1280 146 112 1 1280 142 141 1 1280 142 141 1	153		
		1518	157	156	160
		9216	398	396	401
		64	113	109	116
		128	114	109	117
		256 117 112 512 123 120	121		
60	25604M	512	123	98 396 13 109 14 109 17 112 23 120 37 135 46 144 53 151 70 368 12 108 13 108 16 112	128
(367)	256QAM	1024	137		140
		1280	146	144	148
		1518	153	151	155
		9216	370	368	373
		64	112	108	115
		128	113	108	116
		256	116	112	119
60	5120AM	512	122	119	126
(367)	SIZQAM	1024	134	133	137
		1280	142	141	145
		1518	149	147	151
		9216	348	346	351
		64	111	108	113
		128	112	108	114
		256	115	111	118
60	102/0AM	512	Average Minimum 1113 109 1114 109 1119 113 123 122 139 138 150 148 157 156 398 396 113 109 113 109 114 109 1157 156 398 396 113 109 114 109 117 112 123 120 137 135 146 144 153 151 370 368 112 108 113 108 114 133 115 111 122 119 134 133 142 141 149 147 149 147 149 147 111 108 <t< td=""><td>124</td></t<>	124	
(367)	TOCHQAIN	1024	132	131	134
		1280	140	139	142
		1518	146	145	148
		9216	332	331	335



Physico	al mode		Latency [µs]		
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum
		64	111	108	113
		128	111	108	114
		256	114	111	118
60		512	120	117	124
(367)	2048QAM	1024	131	130	133
		1280	139	138	141
		1518	145	144	147
		9216	326	Latency [µs] Minimum 108 108 111 117 130 138 144 325 107 117 117 130 137 111 117 107 107 111 117 129 137 143 318 108 107 117 129 137 143 318 108 107 111 117 129 137 142 314 107 110 1110 1128 136 136 141 308	329
		64	110	107	113
		128	111	107	113
		256	114	111	117
60	204004041	512	121	117	123
(367)	2048QAM L	1024	131	129	135
		1280	139	137	141
		1518	144	143	145
		9216	320	318	323
		64	110	108	112
		128	111	107	113
		256	114	111	117
60	4006.04M	512	119	117	123
(367)	4096QAM	1024	130	129	132
		1280	138	137	140
		1518	143	142	144
		9216	315	314	318
		64	110	107	112
		128	110	107	113
		256	113	110	116
60	400(04)41	512	119	116	122
(367)	4090QAM L	1024	130	128	134
(Frame ID) 60 (367) 60 (367) 60 (367) 60 (367)		1280	137	136	139
		64 128 128 128 512 1024 1280 1024 1280 11518 9216 64 128 64 128 9216 64 128 1024 128 128 128 1024 128 1024 1280 1518 9216 64 1280 1518 9216 1024 1280 1518 9216 1518 9216 1280 1518 9216 1024 1280 11024 1280 11024 1280 11024 1280 1518	142	141	143
		9216	310	308	312

Physical mode			Latency [µs]			
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum	
		64	105	96	125	
		128	101	98	123	
		256	113	110	122	
80	10000	512	136	133	149	
(368)	4QAM S	1024	Latency [µs] Average Minimum 105 96 101 98 113 110 136 133 180 176 203 199 223 219 872 870 103 95 103 95 103 95 103 95 103 95 103 95 103 95 103 127 168 164 109 100 103 95 205 204 168 164 100 95 100 95 101 95 102 101 103 150 104 112 151 150 151 150 163 161 163 161 100 93 <td>192</td>	192		
		rdulationFrame size [bytes]Average64105.128101.256113.512136.1024180.1280203.9216872.9216872.12898.12898.128109.128108.128188.1512130.1024168.1518205.9216768.9216768.9216768.128101.128101.128101.128163.128151.128163.128100.128100.128100.128100.128100.128100.128100.128100.128100.128100.128100.128100.128100.128100.128100.128100.129111.120111.120111.144132.	199	211		
		1518	223	219	231	
		9216	872	870	877	
		64	103	95	121	
		128	98	96	118	
		256	109	100	120	
80 (368)	10.114	512	130	127	140	
	4QAM	1024	168	164	178	
		1280	188	184	199	
		1518	205	204	213	
		9216	768	Lotency [µs] Minimum 96 98 110 133 176 199 219 870 95 96 100 127 164 184 204 767 95 95 95 95 95 95 95 95 95 95	772	
		64	100	95	110	
		128	101	95	111	
		256	109	Minimum 96 98 110 133 176 199 219 870 95 96 100 127 164 184 204 767 95 95 95 95 164 184 204 150 150 150 150 161 514 94 93 97 109 131 143 153 464	115	
80	1/04/46	512	114	112	120	
(368)	I6QAM S	1024	138	136	144	
		1280	151	150	158	
		1518	28 101 98 36 113 110 12 136 133 24 180 176 80 203 199 18 223 219 16 872 870 4 103 95 28 98 96 36 109 100 12 130 127 24 168 164 80 188 184 18 205 204 16 768 767 4 100 95 28 101 95 36 109 99 12 114 112 24 138 136 80 151 150 18 163 161 16 516 514 4 98 94 28 100 93 56 106 97 12 111 109 24 </td <td>168</td>	168		
		9216	516	514	520	
		64	98	94	107	
		128	100	93	107	
		256	106	97	111	
80	160414	512	111	109	116	
(368)	TOQAM	1024	132	131	137	
		1280	144	143	149	
		1518	154	153	157	
		9216	466	464	469	



Physical mode			Latency [µs]			
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum	
		64	98	93	103	
80		128	99	92	103	
		256	102	95	106	
	720414	512	107	106	116	
(368)	52QAM	1024	125	124	129	
		1280	136	134	139	
		1518	144	142	147	
		9216	404	402	408	
		64	98	94	100	
		128	99	94	102	
		256	103	98	106	
80	64QAM	512	107	105	113	
(368)		1024	122	121	123	
		1280	132	130	134	
		1518	138	137	139	
		9216	358	357	362	
		64	94	91	97	
		128	95	91	98	
	1000.004	256	98	94	101	
80		512	103	101	108	
(368)	IZOQAM	1024	116	115	119	
		1280	125	123	126	
		1518	131	130	133	
		9216	327	325	329	
		64	95	93	97	
		128	96	93	98	
		256	99	96	103	
80	25604M	512	105	102	108	
(368)		1024	116	115	118	
		1280	124	123	126	
		1518	130	129	131	
		9216	308	316	311	

Physical mode			Latency [µs]			
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum	
		64	95	92	96	
		128	95	93	98	
		256	98	96	101	
80	5120AM	512	105	102	107	
(368)	512QAM	1024	115	113	116	
		1280	122	121	123	
		1518	127	126	128	
		9216	292	291	294	
		64	94	92	95	
		128	95	92	96	
		256	98	95	100	
80	1024QAM	512	105	101	106	
(368)		1024	113	112	117	
		1280	120	119	122	
		1518	125	124	126	
		9216	281	279	283	
		64	94	92	95	
		128	94	93	96	
		256	98	95	100	
80		512	104	101	105	
(368)	2048QAM	1024	113	111	116	
		1280	120	119	121	
		1518	124	123	125	
		9216	276	275	277	
		64	91	89	92	
		128	91	89	93	
		256	94	92	96	
80	20490004	512	101	98	103	
(368)	2040ŲΑΙΜ L	1024	110	108	113	
		1280	116	115	117	
		1518	121	120	122	
		9216	269	267	270	



Physical mode			Latency [µs]			
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum	
		64	93	92	94	
		128	94	92	95	
		256	97	95	99	
80	10060AM	512	104	100	105	
(368)	4090QAM	1024	112	110	115	
		1280	119	118	120	
		1518	123	122	124	
		9216	268	267	270	
		64	94	92	95	
		128	95	93	96	
		256	97	95	99	
80	40960AM I	512	104	102	453	
(368)	4070QAM E	1024	113	112	116	
		1280	120	119	122	
		1518	125	124	125	
		9216	278	277	279	
	4QAM S	64	92	80	105	
		128	82	80	96	
		256	96	89	111	
112		512	109	107	116	
(374)		1024	141	139	149	
		1280	159	157	166	
		1518	174	172	180	
		9216	649	648	653	
		64	88	81	101	
		128	88	83	102	
		256	98	87	105	
112	40AM	512	105	103	110	
(374)	4QAM	1024	133	131	138	
		1280	149	147	154	
		1518	162	160	167	
		9216	575	574	578	

Physical mode			Latency [µs]			
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum	
		64	83	78	89	
112		128	83	78	89	
		256	90	81	93	
	1604 M S	512	93	92	96	
(374)	IOQAM S	1024	112	111	115	
		1280	123	122	126	
		1518	132	130	134	
		9216	400	399	404	
		64	82	78	86	
		128	82	78	87	
		256	87	80	90	
112	16QAM	512	92	90	94	
(374)		1024	108	107	110	
		1280	118	117	120	
		1518	126	125	128	
		9216	364	363	368	
	32QAM	64	80	77	83	
		128	81	77	84	
		256	84	80	88	
112		512	90	87	95	
(374)		1024	103	102	105	
		1280	112	111	114	
		1518	118	117	120	
		9216	322	321	325	
		64	78	75	81	
		128	80	76	82	
		256	82	79	85	
112	6400M	512	89	85	91	
(374)	04QAM	1024	99	98	100	
		1280	107	106	108	
		1518	112	111	114	
		9216	288	287	290	



Physical mode			Latency [µs]			
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum	
112		64	77	75	79	
		128	78	75	80	
		256	81	78	83	
	12004M	512	88	84	89	
(374)	IZOQAM	1024	97	96	100	
		1280	104	103	105	
		1518	109	108	110	
		9216	268	266	269	
		64	76	75	78	
		128	77	75	78	
		256	80	78	82	
112	256QAM	512	86	83	88	
(374)		1024	95	94	98	
		1280	102	100	103	
		1518	106	105	107	
		9216	253	251	254	
		64	76	74	77	
		128	76	75	78	
		256	79	77	81	
112	5120414	512	86	84	87	
(374)	SIZQAM	1024	93	92	96	
		1280	100	99	103	
		1518	104	103	105	
		9216	241	240	242	
		64	75	74	77	
		128	76	74	77	
		256	78	77	80	
112	102404M	512	85	82	86	
(374)	1024QAM	1024	92	91	95	
		1280	99	98	102	
		1518	103	102	104	
		9216	233	231	234	

Physical mode			Latency [µs]			
CS [MHz] (Frame ID)	Modulation	Frame size [bytes]	Average	Minimum	Maximum	
		64	75	74	76	
		128	76	74	77	
		256	79	77	80	
112	204904M	512	85	83	86	
(374)	2048QAM	1024	92	91	95	
		1280	98	97	101	
		1518	103	101	103	
		9216	230	228	231	
	2048QAM L	64	75	74	76	
		128	76	74	77	
		256	78	77	80	
112		512	85	83	85	
(374)		1024	92	91	94	
		1280	98	97	100	
		1518	102	101	102	
		9216	226	225	227	
		64	75	74	76	
		128	75	74	76	
		256	78	77	80	
112	4006 O A M	512	84	82	85	
(374)	4070QAM	1024	91	90	94	
		1280	97	96	98	
		1518	101	100	102	
		9216	224	222	225	

2.3 Radio Link Bonding

Radio Link Bonding enables transparent transport of Ethernet frames over a number of parallel Packet Links.



Table 5 Radio Link Bonding

Radio Link bonding	MMU 1002 MMU 1004	MMU 1001	MINI-LINK 6651 ⁽¹⁾	MINI-LINK 6366 MINI-LINK 6371	PNM 1002
Max number of bonded links	4	2	2	2	2
Max capacity for bonded links	2.5 Gbit/s	2.5 Gbit/s	2.5 Gbit/s	2.5 Gbit/s	2.5 Gbit/s

(1) Not applicable for MINI-LINK 6651/3

Protection and bonding do not add additional overhead, nor any latency compared to a 1+0 configuration.

The latency for a bonded hop will be determined by the slowest member in the bonding group.

2.4 Hierarchical Radio Link Bonding

With Hierarchical Radio Link Bonding (hRLB) there is a possibility to increase both the number of bonded links and the aggregated capacity for bonded links. hRLB is supported with NPU 1005 and MMU 1002 or MMU 1004. MINI-LINK 6651/4 and MINI-LINK 6371 also support hRLB.

Table 6	hRLB capacity	/
---------	---------------	---

	NPU 1005	MINI-LINK 6651/4 and MINI-LINK 6371
hRLB capacity in total	10 Gbps	17 Gbps
Maximum hRLB group capacity	10 Gbps	10 Gbps
Maximum number of hRLB groups	2	2
Maximum number of hRLB members in total, per group	4	2
Maximum number of carriers in total	16	6(1)

(1) Figure includes 2 AOD carriers per group. hRLB is only possible together with AOD.

2.4.1 Link Efficiency and Capacity

The hRLB capacity is 7.5 Mpps. Considering RFC2544 throughput test with 64, 128 bytes frames the 10Gbps capacity should not be expected.

If two hRLB groups are configured the hRLB capacity is 4 Mpps per group. Considering RFC2544 throughput test with 64, 128, 256 bytes frames the 10Gbps capacity should not be expected

The utilization of an hRLB link compared to an ordinary link depends on the Ethernet frame size, and it is around 99%. In a typical mobile backhaul or Internet traffic (IMIX 7/4/1) it is recommended to plan with the hRLB group interface capacity value.

2.4.2 Latency

The latency on an hRLB link depends on the latency of all its members, that is the latency of the configured Radio Links. It is recommended to plan with worst case latency which is the value for the link with highest latency and an additional delay. This additional delay is typically in the range of 30 µsec depending on the configuration.

In a steady state the hRLB algorithm ensures that traffic flows are distributed among the hRLB members in a way to optimize the average latency of the traffic flows. This means that the link with highest latency will not determine the latency of all flows.



3 Radio Link

For relations between Frame ID and corresponding Channel Separation support, see Channel Separation to Frame ID Relation on page 73.

3.1 Channel Separation to Frame ID Relation

The tables below show the relationship between Channel Separation and corresponding Frame ID.

Data for standard Frame ID's (256, 257, 258, 259, 260, 261, 262, 263, 264, 265, 266, 267, 268, 303, 1256, 1257, 1258, 1259, 1260, 1261, 1262, 1263, 1264, 1265, 1266, 1267, 1268, and 1303) can be found in revision BT of this document.

ETSI	Frame ID	Frame ID	Frame ID	Frame ID	Frame ID
Channel Separation	Single Carrier	Carrier Aggregation	XPIC	2x2 MIMO	4x4 MIMO
7 MHz	356		1356		
13.75 MHz	357		1357		
14 MHz	357		1357		
27.5 MHz	358	358	1358	2358	2358
28 MHz	358	358	1358	2358	2358
40 MHz	359	359	1359		
55 MHz	360	360	1360	2360	2360
56 MHz	360	360	1360	2360	2360
62.5 MHz	360		1360		
80 MHz	403		1403		
110 MHz	361	361	1361		
112 MHz	361	361	1361		
125 MHz	361		1361		
220 MHz ⁽¹⁾		404			
224 MHz ⁽²⁾		404			

Table 7 Channel Separation to Frame ID Relation - ETSI

(1) For CS 220 MHz, Carrier Aggregation with 2x110 MHz shall be configured and FFID 404 shall be used to fulfill CS 220 MHz requirements.

(2) For CS 224 MHz, Carrier Aggregation with 2x112 MHz shall be configured and FFID 404 shall be used to fulfill CS 224 MHz requirements.

ANSI	Frame ID	Frame ID	Frame ID	Frame ID	Frame ID
Channel Separation	Single Carrier	Carrier Aggregation	XPIC	2x2 MIMO	4x4 MIMO
10 MHz	362		1362		
20 MHz	363		1363		
28 MHz	371		1371		
30 MHz	364	364	1364	2364	2364
40 MHz	365	365	1365	2365	2365
50 MHz	366	366	1366		
56 MHz	373		1373		
60 MHz	367	367	1367		
80 MHz	368	368	1368		
112 MHz	374		1374		

Table 8 Channel Separation to Frame ID Relation - ANSI

3.2 Radio Frequencies

3.2.1 Radio Frequencies

Within the restrictions of channel arrangement and sub-band ranging imposed by the different Radio Unit the channel frequency can be selected from the O&M system.

Frequency band and sub range, as well as lowest, highest and selected center frequency is readable from the O&M system.

The operating center frequency can be set in steps of 0.005, 0.010, 0.025, or 0.250 MHz depending on sub-band, frequency band and Radio Unit.

For Radio Units with SW configurable duplex distance, the distance between transmitter and receiver shall always comply with regulatory requirements. The most common duplex distances can be found after each sub-band table.

The Radio Units are produced for frequency bands, sub ranges and channel plans according to the tables in the following chapters.

Note: For the MINI-LINK 6364 product family, a limitation of a maximum of 1000 frequency shifts applies for full performance and capacity fulfillment.

3.2.2 CS Dependence

Lower and upper frequencies in below tables state the edges for respective subband. The given frequency edge shall therefore be compensated for actual CS according below. (CS= 7, 14, 28, 40, 56, 80, and 112 MHz for ETSI and 5, 10, 20, 30, 40, 50, 60, and 80 MHz for ANSI)

Lowest user frequency = Tx or Rx Lower edge + CS/2

Highest user frequency = Tx or Rx Higher edge – CS/2

3.2.3 RAU2 X

3.2.3.1 RAU2 X 5 GHz Band

Table 9 RAU2 X 5 HP

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver frequency information			Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
11	300	4400.00	4570.00	170.00	4700.00	4870.00	170.00	ETSI
15	300	4700.00	4870.00	170.00	4400.00	4570.00	170.00	ETSI
12	300	4530.00	4700.00	170.00	4830.00	5000.00	170.00	ETSI
16	300	4830.00	5000.00	170.00	4530.00	4700.00	170.00	ETSI
A11		4400.00	4570.00	170.00	4700.00	4870.00	170.00	ETSI
A15		4700.00	4870.00	170.00	4400.00	4570.00	170.00	ETSI
A12		4530.00	4700.00	170.00	4830.00	5000.00	170.00	ETSI
A16		4830.00	5000.00	170.00	4530.00	4700.00	170.00	ETSI

Sub-band 1x is valid for CS 7, 14, 28, 40, and 56 MHz.

Sub-band A1x (Duplex 300 and 312 MHz) is valid for CS 7, 14, 28, 40, and 56 MHz.

3.2.3.2 RAU2 X 6 GHz Lower Band

Table 10 RAU2 X 6L [HP]

Sub-band / Duplex (MHz)		Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
11	252.04	5925.000	6020.250	95.25	6177.050	6272.300	95.25	ETSI/ANSI
15	252.04	6177.050	6272.300	95.25	5925.000	6020.250	95.25	ETSI/ANSI
12	252.04	6018.000	6110.250	92.25	6270.050	6362.300	92.25	ETSI/ANSI
16	252.04	6270.050	6362.300	92.25	6018.000	6110.250	92.25	ETSI/ANSI
13	252.04	6078.500	6173.250	94.75	6330.550	6425.300	94.75	ETSI/ANSI

Sub-band / D)uplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
17	252.04	6330.550	6425.300	94.75	6078.500	6173.250	94.75	ETSI/ANSI
14	252.04	5989.675	6048.975	59.30	6241.725	6301.025	59.30	ETSI/ANSI
18	252.04	6241.725	6301.025	59.30	5989.675	6048.975	59.30	ETSI/ANSI
A11		5925.000	6020.250	95.250	6177.050	6277.000	99.950	ETSI/ANSI
A15		6177.050	6277.000	99.950	5925.000	6020.250	95.250	ETSI/ANSI
A12		6018.000	6110.250	92.250	6270.050	6362.300	92.250	ETSI/ANSI
A16		6270.050	6362.300	92.250	6018.000	6110.250	92.250	ETSI/ANSI
A13		6078.500	6173.250	94.750	6330.550	6425.300	94.750	ETSI/ANSI
A17		6330.550	6425.300	94.750	6078.500	6173.250	94.750	ETSI/ANSI
A14		5989.675	6048.975	59.300	6241.725	6301.025	59.300	ETSI/ANSI
A18		6241.725	6301.025	59.300	5989.675	6048.975	59.300	ETSI/ANSI
A51		5925.000	6020.250	95.250	6177.050	6277.000	99.950	ANSI
A55		6177.050	6277.000	99.950	5925.000	6020.250	95.250	ANSI
A52		6011.000	6110.250	99.250	6270.050	6362.300	92.250	ANSI
A56		6270.050	6362.300	92.250	6011.000	6110.250	99.250	ANSI
A53		6078.500	6173.250	94.750	6330.550	6425.300	94.750	ANSI
A57		6330.550	6425.300	94.750	6078.500	6173.250	94.750	ANSI
A54		5989.675	6048.975	59.300	6241.725	6301.025	59.300	ANSI
A58		6241.725	6301.025	59.300	5989.675	6048.975	59.300	ANSI

Sub-band 1x is valid for CS 7, 14, 28/29.65, and 56/59.3 MHz.

Sub-band 1x is valid for CS 9.88/10, 20, 29.65/30, 40, 50, and 59.3/60 MHz.

Sub-band A1x (Duplex 240, 252.04, 260, and 266 MHz) is valid for CS 7, 14, 28/29.65, and 56/59.3 MHz.

Sub-band A1x (Duplex 252.04 MHz) are valid for CS 9.88/10, 20, 29.65/30, 40, 50, and 59.3/60 MHz.

Sub-band A5x (Duplex 252.04 MHz) are valid for CS 9.88/10, 20, 29.65/30, 40, 50, and 59.3/60 MHz.

3.2.3.3 RAU2 X 6 GHz Upper Band

Table 11 RAU2 X 6U [HP]

Sub-band / Duplex (MHz)		Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
21	340	6430.00	6565.00	135.00	6770.00	6905.00	135.00	ETSI/ANSI
25	340	6770.00	6905.00	135.00	6430.00	6565.00	135.00	ETSI/ANSI
22	340	6550.00	6685.00	135.00	6890.00	7025.00	135.00	ETSI/ANSI
26	340	6890.00	7025.00	135.00	6550.00	6685.00	135.00	ETSI/ANSI

Sub-ban	Sub-band / Duplex (MHz)		frequency infor	mation	Receiver free	quency informa	ition	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
23	340	6635.00	6765.00	130.00	6975.00	7105.00	130.00	ETSI/ANSI
27	340	6975.00	7105.00	130.00	6635.00	6765.00	130.00	ETSI/ANSI
24	340	6520.00	6600.00	80.00	6860.00	6940.00	80.00	ETSI/ANSI
28	340	6860.00	6940.00	80.00	6520.00	6600.00	80.00	ETSI/ANSI
A21		6430.000	6565.000	135.000	6770.000	6905.000	135.000	ETSI/ANSI
A25		6770.000	6905.000	135.000	6430.000	6565.000	135.000	ETSI/ANSI
A22		6550.000	6685.000	135.000	6890.000	7025.000	135.000	ETSI/ANSI
A26		6890.000	7025.000	135.000	6550.000	6685.000	135.000	ETSI/ANSI
A23		6635.000	6765.000	130.000	6975.000	7105.000	130.000	ETSI/ANSI
A27		6975.000	7105.000	130.000	6635.000	6765.000	130.000	ETSI/ANSI
A24		6520.000	6600.000	80.000	6860.000	6940.000	80.000	ETSI/ANSI
A28		6860.000	6940.000	80.000	6520.000	6600.000	80.000	ETSI/ANSI
31	160	6540.00	6602.50	62.50	6700.00	6762.50	62.50	ANSI
35	160	6700.00	6762.50	62.50	6540.00	6602.50	62.50	ANSI
32	160	6597.50	6657.50	60.00	6757.50	6817.50	60.00	ANSI
36	160	6757.50	6817.50	60.00	6597.50	6657.50	60.00	ANSI
33	160	6650.00	6710.00	60.00	6810.00	6870.00	60.00	ANSI
37	160	6810.00	6870.00	60.00	6650.00	6710.00	60.00	ANSI
A31		6540.000	6610.000	70.000	6700.000	6770.000	70.000	ANSI
A35		6700.000	6770.000	70.000	6540.000	6610.000	70.000	ANSI
A32		6597.500	6657.500	60.000	6757.500	6817.500	60.000	ANSI
A36		6757.500	6817.500	60.000	6597.500	6657.500	60.000	ANSI
A33		6640.000	6710.000	70.000	6800.000	6870.000	70.000	ANSI
A37		6800.000	6870.000	70.000	6640.000	6710.000	70.000	ANSI
A41		6523.750	6626.250	102.500	6773.750	6876.250	102.500	ANSI
A45		6773.750	6876.250	102.500	6523.750	6626.250	102.500	ANSI

Sub-band 2x is valid for 7, 14, 28/30, 40, and 56/60 MHz.

Sub-band 2x is valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band A2x (Duplex 340 MHz) is valid for CS 7, 14, 28/30, 40, and 56/60 MHz.

Sub-band A2x (Duplex 340 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band 3x is valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band A3x (Duplex 160 and 170 MHz) are valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.3.4 RAU2 X 7 GHz Band

Table 12 RAU2 X 7 [HP]

Sub-band /	Duplex (MHz)	Transmitter f	requency inform	mation	Receiver freq	uency information	tion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
01	196	7107.00	7163.00	56.00	7303.00	7359.00	56.00	ETSI
05	196	7303.00	7359.00	56.00	7107.00	7163.00	56.00	ETSI
02	196	7163.00	7219.00	56.00	7359.00	7415.00	56.00	ETSI
06	196	7359.00	7415.00	56.00	7163.00	7219.00	56.00	ETSI
03	196	7191.00	7247.00	56.00	7387.00	7443.00	56.00	ETSI
07	196	7387.00	7443.00	56.00	7191.00	7247.00	56.00	ETSI
11	154	7428.00	7484.00	56.00	7582.00	7638.00	56.00	ETSI
15	154	7582.00	7638.00	56.00	7428.00	7484.00	56.00	ETSI
12	154	7456.00	7512.00	56.00	7610.00	7666.00	56.00	ETSI
16	154	7610.00	7666.00	56.00	7456.00	7512.00	56.00	ETSI
13	154	7484.00	7540.00	56.00	7638.00	7694.00	56.00	ETSI
17	154	7638.00	7694.00	56.00	7484.00	7540.00	56.00	ETSI
14	154	7512.00	7568.00	56.00	7666.00	7722.00	56.00	ETSI
18	154	7666.00	7722.00	56.00	7512.00	7568.00	56.00	ETSI
21	245	7426.50	7513.75	87.25	7671.50	7758.75	87.25	ETSI
25	245	7671.50	7758.75	87.25	7426.50	7513.75	87.25	ETSI
22	245	7482.25	7569.75	87.50	7727.25	7814.75	87.50	ETSI
26	245	7727.25	7814.75	87.50	7482.25	7569.75	87.50	ETSI
23	245	7510.25	7597.75	87.50	7755.25	7842.75	87.50	ETSI
27	245	7755.25	7842.75	87.50	7510.25	7597.75	87.50	ETSI
24	245	7566.25	7653.75	87.50	7566.25	7653.75	87.50	ETSI
28	245	7811.25	7898.75	87.50	7566.25	7653.75	87.50	ETSI
31	161	7124.50	7184.00	59.50	7285.50	7345.00	59.50	ETSI
35	161	7285.50	7345.00	59.50	7124.50	7184.00	59.50	ETSI
32	161	7152.50	7212.00	59.50	7313.50	7373.00	59.50	ETSI
36	161	7313.50	7373.00	59.50	7152.50	7212.00	59.50	ETSI
33	161	7180.50	7240.00	59.50	7341.50	7401.00	59.50	ETSI
37	161	7341.50	7401.00	59.50	7180.50	7240.00	59.50	ETSI
34	161	7208.50	7268.00	59.50	7369.50	7429.00	59.50	ETSI
38	161	7369.50	7429.00	59.50	7208.50	7268.00	59.50	ETSI
41	161	7424.50	7484.00	59.50	7585.50	7645.00	59.50	ETSI
45	161	7585.50	7645.00	59.50	7424.50	7484.00	59.50	ETSI
42	161	7449.50	7512.00	62.50	7610.50	7673.00	62.50	ETSI
46	161	7610.50	7673.00	62.50	7449.50	7512.00	62.50	ETSI
43	161	7477.50	7540.00	62.50	7638.50	7701.00	62.50	ETSI
47	161	7638.50	7701.00	62.50	7477.50	7540.00	62.50	ETSI
44	161	7505.50	7568.00	62.50	7666.50	7729.00	62.50	ETSI
48	161	7666.50	7729.00	62.50	7505.50	7568.00	62.50	ETSI

Sub-band	d / Duplex (MHz)	Transmitter	frequency infor	mation	Receiver free	quency informa	tion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
51	168	7107.00	7163.00	56.00	7275.00	7331.00	56.00	ETSI
55	168	7275.00	7331.00	56.00	7107.00	7163.00	56.00	ETSI
52	168	7135.00	7191.00	56.00	7303.00	7359.00	56.00	ETSI
56	168	7303.00	7359.00	56.00	7135.00	7191.00	56.00	ETSI
53	168	7163.00	7219.00	56.00	7331.00	7387.00	56.00	ETSI
57	168	7331.00	7387.00	56.00	7163.00	7219.00	56.00	ETSI
54	168	7191.00	7247.00	56.00	7359.00	7415.00	56.00	ETSI
58	168	7359.00	7415.00	56.00	7191.00	7247.00	56.00	ETSI
61	168	7414.00	7470.00	56.00	7582.00	7638.00	56.00	ETSI
65	168	7582.00	7638.00	56.00	7414.00	7470.00	56.00	ETSI
62	168	7442.00	7498.00	56.00	7610.00	7666.00	56.00	ETSI
66	168	7610.00	7666.00	56.00	7442.00	7498.00	56.00	ETSI
63	168	7470.00	7526.00	56.00	7638.00	7694.00	56.00	ETSI
67	168	7638.00	7694.00	56.00	7470.00	7526.00	56.00	ETSI
64	168	7498.00	7554.00	56.00	7666.00	7722.00	56.00	ETSI
68	168	7666.00	7722.00	56.00	7498.00	7554.00	56.00	ETSI
71	154	7128.00	7184.00	56.00	7282.00	7338.00	56.00	ETSI
75	154	7282.00	7338.00	56.00	7128.00	7184.00	56.00	ETSI
72	154	7156.00	7212.00	56.00	7310.00	7366.00	56.00	ETSI
76	154	7310.00	7366.00	56.00	7156.00	7212.00	56.00	ETSI
73	154	7184.00	7240.00	56.00	7338.00	7394.00	56.00	ETSI
77	154	7338.00	7394.00	56.00	7184.00	7240.00	56.00	ETSI
74	154	7212.00	7268.00	56.00	7366.00	7422.00	56.00	ETSI
78	154	7366.00	7422.00	56.00	7212.00	7268.00	56.00	ETSI
81	161	7249.50	7309.00	59.50	7410.50	7470.00	59.50	ETSI
85	161	7410.50	7470.00	59.50	7249.50	7309.00	59.50	ETSI
82	161	7277.50	7337.00	59.50	7438.50	7498.00	59.50	ETSI
86	161	7438.50	7498.00	59.50	7277.50	7337.00	59.50	ETSI
83	161	7305.50	7365.00	59.50	7466.50	7526.00	59.50	ETSI
87	161	7466.50	7526.00	59.50	7305.50	7365.00	59.50	ETSI
84	161	7333.50	7393.00	59.50	7494.50	7554.00	59.50	ETSI
88	161	7494.50	7554.00	59.50	7333.50	7393.00	59.50	ETSI
91	168	7443.00	7499.00	56.00	7611.00	7667.00	56.00	ETSI
95	168	7611.00	7667.00	56.00	7443.00	7499.00	56.00	ETSI
92	168	7471.00	7527.00	56.00	7639.00	7695.00	56.00	ETSI
96	168	7639.00	7695.00	56.00	7471.00	7527.00	56.00	ETSI
93	168	7499.00	7555.00	56.00	7667.00	7723.00	56.00	ETSI
97	168	7667.00	7723.00	56.00	7499.00	7555.00	56.00	ETSI
94	168	7527.00	7583.00	56.00	7695.00	7751.00	56.00	ETSI
98	168	7695.00	7751.00	56.00	7527.00	7583.00	56.00	ETSI
101	175	7125.00	7170.00	45.00	7300.00	7345.00	45.00	ANSI
105	175	7300.00	7345.00	45.00	7125.00	7170.00	45.00	ANSI
102	175	7155.00	7195.00	40.00	7330.00	7370.00	40.00	ANSI

Sub-band / [Duplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
106	175	7330.00	7370.00	40.00	7155.00	7195.00	40.00	ANSI
103	175	7180.00	7225.00	45.00	7355.00	7400.00	45.00	ANSI
107	175	7355.00	7400.00	45.00	7180.00	7225.00	45.00	ANSI
104	175	7210.00	7250.00	40.00	7385.00	7425.00	40.00	ANSI
108	175	7385.00	7425.00	40.00	7210.00	7250.00	40.00	ANSI
111	150	7425.00	7480.00	55.00	7575.00	7630.00	55.00	ANSI
115	150	7575.00	7630.00	55.00	7425.00	7480.00	55.00	ANSI
112	150	7460.00	7515.00	55.00	7610.00	7665.00	55.00	ANSI
116	150	7610.00	7665.00	55.00	7460.00	7515.00	55.00	ANSI
113	150	7515.00	7555.00	40.00	7665.00	7705.00	40.00	ANSI
117	150	7665.00	7705.00	40.00	7515.00	7555.00	40.00	ANSI
114	150	7545.00	7575.00	30.00	7695.00	7725.00	30.00	ANSI
118	150	7695.00	7725.00	30.00	7545.00	7575.00	30.00	ANSI
132	269	7505.00	7533.00	28.00	7774.00	7802.00	28.00	ETSI
136	269	7774.00	7802.00	28.00	7505.00	7533.00	28.00	ETSI
A01		7107.000	7163.000	56.00	7303.000	7359.000	56.00	ETSI
A05		7303.000	7359.000	56.00	7107.000	7163.000	56.00	ETSI
A02		7163.000	7219.000	56.00	7359.000	7415.000	56.00	ETSI
A06		7359.000	7415.000	56.00	7163.000	7219.000	56.00	ETSI
A03		7191.000	7247.000	56.00	7387.000	7443.000	56.00	ETSI
A07		7387.000	7443.000	56.00	7191.000	7247.000	56.00	ETSI
A11		7414.000	7485.000	71.00	7575.000	7645.000	70.00	ETSI/ANSI
A15		7575.000	7645.000	70.00	7414.000	7485.000	71.00	ETSI/ANSI
A12		7442.000	7515.000	73.00	7610.000	7673.000	63.00	ETSI/ANSI
A16		7610.000	7673.000	63.00	7442.000	7515.000	73.00	ETSI/ANSI
A13		7470.000	7540.000	70.00	7638.000	7701.000	63.00	ETSI/ANSI
A17		7638.000	7701.000	63.00	7470.000	7540.000	70.00	ETSI/ANSI
A14		7498.000	7568.000	70.00	7665.000	7729.000	64.00	ETSI/ANSI
A18		7665.000	7729.000	64.00	7498.000	7568.000	70.00	ETSI/ANSI
A21		7426.500	7513.750	87.25	7671.500	7758.750	87.25	ETSI
A25		7671.500	7758.750	87.25	7426.500	7513.750	87.25	ETSI
A22		7482.250	7569.750	87.50	7727.250	7814.750	87.50	ETSI
A26		7727.250	7814.750	87.50	7482.250	7569.750	87.50	ETSI
A23		7510.250	7597.750	87.50	7755.250	7842.750	87.50	ETSI
A27		7755.250	7842.750	87.50	7510.250	7597.750	87.50	ETSI
A24		7566.250	7653.750	87.50	7811.250	7898.750	87.50	ETSI
A28		7811.250	7898.750	87.50	7566.250	7653.750	87.50	ETSI
A31		7107.000	7184.000	77.00	7275.000	7345.000	70.00	ETSI/ANSI
A35		7275.000	7345.000	70.00	7107.000	7184.000	77.00	ETSI/ANSI
A32		7135.000	7212.000	77.00	7303.000	7373.000	70.00	ETSI/ANSI
A36		7303.000	7373.000	70.00	7135.000	7212.000	77.00	ETSI/ANSI
A33		7163.000	7240.000	77.00	7331.000	7401.000	70.00	ETSI/ANSI
A37		7331.000	7401.000	70.00	7163.000	7240.000	77.00	ETSI/ANSI

Sub-band / D	uplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
A34		7191.000	7268.000	77.00	7359.000	7429.000	70.00	ETSI/ANSI
A38		7359.000	7429.000	70.00	7191.000	7268.000	77.00	ETSI/ANSI
A81		7249.500	7309.000	59.50	7410.500	7470.000	59.50	ETSI
A85		7410.500	7470.000	59.50	7249.500	7309.000	59.50	ETSI
A82		7277.500	7337.000	59.50	7438.500	7498.000	59.50	ETSI
A86		7438.500	7498.000	59.50	7277.500	7337.000	59.50	ETSI
A83		7305.500	7365.000	59.50	7466.500	7526.000	59.50	ETSI
A87		7466.500	7526.000	59.50	7305.500	7365.000	59.50	ETSI
A84		7333.500	7393.000	59.50	7494.500	7554.000	59.50	ETSI
A88		7494.500	7554.000	59.50	7333.500	7393.000	59.50	ETSI
A94		7527.000	7583.000	56.00	7695.000	7751.000	56.00	ETSI/ANSI
A98		7695.000	7751.000	56.00	7527.000	7583.000	56.00	ETSI/ANSI

Sub-band 0x, 1x, 2x, 3x, 4x, 5x, 6x, 7x, 8x, 9x, and 13x are valid for CS 7, 14, 28, and 56 MHz.

Sub-band 10x is valid for CS 10, 20, 30, and 40 MHz.

Sub-band 111, 112, 115, and 116 are valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band 114 and 118 are valid for CS 10, 20, and 30 MHz.

Sub-band A0x (Duplex 196 MHz), A2x (Duplex 245 and 269 MHz), and A8x (Duplex 161 MHz) are valid for CS 7, 14, 28, and 56 MHz.

Sub-band A1x (Duplex 154, 161, and 168 MHz), A3x (Duplex 154, 161, and 168 MHz), and A9x (Duplex 168 MHz) are valid for CS 7, 14, 28, and 56 MHz.

Sub-band A1x (Duplex 150 MHz), A3x (Duplex 175 MHz), and A9x (Duplex 150 MHz) are valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.3.5 RAU2 X 8 GHz Band

Sub-band / Duplex (MHz)		Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
11	311.32	7718.050	7807.000	88.950	8029.375	8118.325	88.950	ETSI
15	311.32	8029.375	8118.325	88.950	7718.050	7807.000	88.950	ETSI
12	311.32	7777.350	7866.300	88.950	8088.675	8177.625	88.950	ETSI
16	311.32	8088.675	8177.625	88.950	7777.350	7866.300	88.950	ETSI
13	311.32	7821.825	7910.775	88.950	8133.150	8222.100	88.950	ETSI
17	311.32	8133.150	8222.100	88.950	7821.825	7910.775	88.950	ETSI
14	311.32	7881.125	7970.075	88.950	8192.450	8281.400	88.950	ETSI

Table 13 RAU2 X 8 [HP]

Sub-band / [Duplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
18	311.32	8192.450	8281.400	88.950	7881.125	7970.075	88.950	ETSI
21	126	8282.50	8324.50	42.00	8408.50	8450.50	42.00	ETSI
25	126	8408.50	8450.50	42.00	8282.50	8324.50	42.00	ETSI
23	126	8324.50	8366.50	42.00	8450.50	8492.50	42.00	ETSI
27	126	8450.50	8492.50	42.00	8324.50	8366.50	42.00	ETSI
31	119	8286.00	8328.00	42.00	8405.00	8447.00	42.00	ETSI
35	119	8405.00	8447.00	42.00	8286.00	8328.00	42.00	ETSI
33	119	8328.00	8370.00	42.00	8447.00	8489.00	42.00	ETSI
37	119	8447.00	8489.00	42.00	8328.00	8370.00	42.00	ETSI
41	119	8279.00	8321.00	42.00	8398.00	8440.00	42.00	ETSI
45	119	8398.00	8440.00	42.00	8279.00	8321.00	42.00	ETSI
42	119	8307.00	8349.00	42.00	8426.00	8468.00	42.00	ETSI
46	119	8426.00	8468.00	42.00	8307.00	8349.00	42.00	ETSI
43	119	8335.00	8377.00	42.00	8454.00	8496.00	42.00	ETSI
47	119	8454.00	8496.00	42.00	8335.00	8377.00	42.00	ETSI
51	126	8279.00	8321.00	42.00	8405.00	8447.00	42.00	ETSI
55	126	8405.00	8447.00	42.00	8279.00	8321.00	42.00	ETSI
52	126	8307.00	8349.00	42.00	8433.00	8475.00	42.00	ETSI
56	126	8433.00	8475.00	42.00	8307.00	8349.00	42.00	ETSI
53	126	8328.00	8370.00	42.00	8454.00	8496.00	42.00	ETSI
57	126	8454.00	8496.00	42.00	8328.00	8370.00	42.00	ETSI
61	266	7905.00	8010.00	105.00	8171.00	8276.00	105.00	ETSI
65	266	8171.00	8276.00	105.00	7905.00	8010.00	105.00	ETSI
62	266	7968.00	8073.00	105.00	8234.00	8339.00	105.00	ETSI
66	266	8234.00	8339.00	105.00	7968.00	8073.00	105.00	ETSI
63	266	8031.00	8136.00	105.00	8297.00	8402.00	105.00	ETSI
67	266	8297.00	8402.00	105.00	8031.00	8136.00	105.00	ETSI
71	311.32	7718.050	7836.650	118.600	8029.375	8147.975	118.600	ETSI
75	311.32	8029.375	8147.975	118.600	7718.050	7836.650	118.600	ETSI
72	311.32	7777.350	7881.125	103.775	8088.675	8192.450	103.775	ETSI
76	311.32	8088.675	8192.450	103.775	7777.350	7881.125	103.775	ETSI
73	311.32	7851.475	7970.075	118.600	8162.800	8281.400	118.600	ETSI
77	311.32	8162.800	8281.400	118.600	7851.475	7970.075	118.600	ETSI
81	300	7725.00	7805.00	80.00	8025.00	8105.00	80.00	ANSI
85	300	8025.00	8105.00	80.00	7725.00	7805.00	80.00	ANSI
82	300	7785.00	7865.00	80.00	8085.00	8165.00	80.00	ANSI
86	300	8085.00	8165.00	80.00	7785.00	7865.00	80.00	ANSI
83	300	7835.00	7910.00	75.00	8135.00	8210.00	75.00	ANSI
87	300	8135.00	8210.00	75.00	7835.00	7910.00	75.00	ANSI
84	300	7895.00	7965.00	70.00	8195.00	8265.00	70.00	ANSI
88	300	8195.00	8265.00	70.00	7895.00	7965.00	70.00	ANSI
91	208	8050.00	8134.00	84.00	8258.00	8342.00	84.00	ETSI
95	208	8258.00	8342.00	84.00	8050.00	8134.00	84.00	ETSI

Sub-ban	d / Duplex (MHz)	Transmitter	frequency infor	mation	Receiver free	quency informo	ition	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	— Standard
92	208	8106.00	8190.00	84.00	8314.00	8398.00	84.00	ETSI
96	208	8314.00	8398.00	84.00	8106.00	8190.00	84.00	ETSI
93	208	8162.00	8246.00	84.00	8370.00	8454.00	84.00	ETSI
97	208	8370.00	8454.00	84.00	8162.00	8246.00	84.00	ETSI
101	310	7905.00	8017.00	112.00	8215.00	8327.00	112.00	ETSI
105	310	8215.00	8327.00	112.00	7905.00	8017.00	112.00	ETSI
102	310	7989.00	8101.00	112.00	8299.00	8411.00	112.00	ETSI
106	310	8299.00	8411.00	112.00	7989.00	8101.00	112.00	ETSI
103	310	8073.00	8185.00	112.00	8383.00	8495.00	112.00	ETSI
107	310	8383.00	8495.00	112.00	8073.00	8185.00	112.00	ETSI
111	310	7725.00	7837.00	112.00	8035.00	8147.00	112.00	ETSI
115	310	8035.00	8147.00	112.00	7725.00	7837.00	112.00	ETSI
112	310	7777.00	7881.00	104.00	8087.00	8191.00	104.00	ETSI
116	310	8087.00	8191.00	104.00	7777.00	7881.00	104.00	ETSI
113	310	7851.00	7965.00	114.00	8161.00	8275.00	114.00	ETSI
117	310	8161.00	8275.00	114.00	7851.00	7965.00	114.00	ETSI
121	148.5	7744.75	7793.75	49.00	7893.25	7942.25	49.00	ETSI
125	148.5	7893.25	7942.25	49.00	7744.75	7793.75	49.00	ETSI
122	148.5	7779.75	7828.75	49.00	7928.25	7977.25	49.00	ETSI
126	148.5	7928.25	7977.25	49.00	7779.75	7828.75	49.00	ETSI
131	148.25	7744.75	7793.75	49.00	7893.00	7942.00	49.00	ETSI
135	148.25	7893.00	7942.00	49.00	7744.75	7793.75	49.00	ETSI
132	148.25	7779.75	7828.75	49.00	7928.00	7977.00	49.00	ETSI
136	148.25	7928.00	7977.00	49.00	7779.75	7828.75	49.00	ETSI
A01		7744.750	7793.750	49.000	7893.000	7942.250	49.250	ETSI
A05		7893.000	7942.250	49.250	7744.750	7793.750	49.000	ETSI
A02		7779.750	7828.750	49.000	7928.000	7977.250	49.250	ETSI
A06		7928.000	7977.250	49.250	7779.750	7828.750	49.000	ETSI
A11		7718.050	7837.000	118.950	8025.000	8147.975	122.975	ETSI/ANSI
A15		8025.000	8147.975	122.975	7718.050	7837.000	118.950	ETSI/ANSI
A12		7777.000	7910.775	133.775	8085.000	8222.100	137.100	ETSI/ANSI
A16		8085.000	8222.100	137.100	7777.000	7910.775	133.775	ETSI/ANSI
A14		7835.000	7970.075	135.075	8135.000	8281.400	146.400	ETSI/ANSI
A18		8135.000	8281.400	146.400	7835.000	7970.075	135.075	ETSI/ANSI
A21		8279.000	8328.000	49.000	8405.000	8450.500	45.500	ETSI
A25		8405.000	8450.500	45.500	8279.000	8328.000	49.000	ETSI
A23		8324.500	8370.000	45.500	8447.000	8492.500	45.500	ETSI
A27		8447.000	8492.500	45.500	8324.500	8370.000	45.500	ETSI
A41		8279.000	8321.000	42.000	8398.000	8440.000	42.000	ETSI
A45		8398.000	8440.000	42.000	8279.000	8321.000	42.000	ETSI
A42		8307.000	8349.000	42.000	8426.000	8475.000	49.000	ETSI
A46		8426.000	8475.000	49.000	8307.000	8349.000	42.000	ETSI
A43		8328.000	8377.000	49.000	8454.000	8496.000	42.000	ETSI

Sub-band / Duplex (MHz)	Transmitter f	requency inforr	nation	Receiver freq	uency informat	tion	Telecom
	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
A47	8454.000	8496.000	42.000	8328.000	8377.000	49.000	ETSI
A61	7905.000	8017.000	112.000	8171.000	8327.000	156.000	ETSI
A65	8171.000	8327.000	156.000	7905.000	8017.000	112.000	ETSI
A62	7968.000	8073.000	105.000	8234.000	8339.000	105.000	ETSI
A66	8234.000	8339.000	105.000	7968.000	8073.000	105.000	ETSI
A63	7989.000	8136.000	147.000	8297.000	8411.000	114.000	ETSI
A67	8297.000	8411.000	114.000	7989.000	8136.000	147.000	ETSI
A64	8073.000	8185.000	112.000	8383.000	8495.000	112.000	ETSI
A68	8383.000	8495.000	112.000	8073.000	8185.000	112.000	ETSI
A91	8050.000	8134.000	84.000	8258.000	8342.000	84.000	ETSI
A95	8258.000	8342.000	84.000	8050.000	8134.000	84.000	ETSI
A92	8106.000	8190.000	84.000	8314.000	8398.000	84.000	ETSI
A96	8314.000	8398.000	84.000	8106.000	8190.000	84.000	ETSI
A93	8162.000	8246.000	84.000	8370.000	8454.000	84.000	ETSI
A97	8370.000	8454.000	84.000	8162.000	8246.000	84.000	ETSI

Sub-band 1x is valid for CS 7, 14, 28/29.65, and 56/59.3 MHz.

Sub-band 2x, 3x, 12x, and 13x are valid for 7, 14, and 28/29.65 MHz.

Sub-band 4x and 5x, are valid for CS 7, 14, and 28/29.65 MHz.

Sub-band 6x, 7x, 9x, 10x, and 11x are valid for CS 7, 14, 28/29.65, and 56/59.3 MHz.

Sub-band 8x is valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band A0x (Duplex 148.25 and 148.5 MHz), A2x and A4x (Duplex 119 and 126 MHz) are valid for CS 7, 14, and 28/29.65 MHz.

Sub-band A1x (Duplex 283.5, 310, and 311.32 MHz), A6x (Duplex 266 and 310 MHz), and A9x (Duplex 208 MHz) are valid for CS 7, 14, 28/29.65, and 56/59.3 MHz.

Sub-band A1x (Duplex 300 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.3.6 RAU2 X 10 GHz Band

Table 14	RAU2 X 10 [HP]
----------	----------------

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
11	65	10550.00	10570.00	20.00	10615.00	10635.00	20.00	ANSI
15	65	10615.00	10635.00	20.00	10550.00	10570.00	20.00	ANSI

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver frequency information			Telecom Standard
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	
12	65	10565.00	10585.00	20.00	10630.00	10650.00	20.00	ANSI
16	65	10630.00	10650.00	20.00	10565.00	10585.00	20.00	ANSI
13	65	10580.00	10600.00	20.00	10645.00	10665.00	20.00	ANSI
17	65	10645.00	10665.00	20.00	10580.00	10600.00	20.00	ANSI
14	65	10595.00	10615.00	20.00	10660.00	10680.00	20.00	ANSI
18	65	10660.00	10680.00	20.00	10595.00	10615.00	20.00	ANSI
21	91	10500.50	10528.50	28.00	10591.50	10619.50	28.00	ETSI
25	91	10591.50	10619.50	28.00	10500.50	10528.50	28.00	ETSI
22	91	10528.50	10556.50	28.00	10619.50	10647.50	28.00	ETSI
26	91	10619.50	10647.50	28.00	10528.50	10556.50	28.00	ETSI
23	91	10556.50	10584.50	28.00	10647.50	10675.50	28.00	ETSI
27	91	10647.50	10675.50	28.00	10556.50	10584.50	28.00	ETSI
31	350	10125.00	10300.00	175.00	10475.00	10650.00	175.00	ETSI
35	350	10475.00	10650.00	175.00	10125.00	10300.00	175.00	ETSI
41	168	10308.00	10420.00	112.00	10476.00	10588.00	112.00	ETSI
45	168	10476.00	10588.00	112.00	10308.00	10420.00	112.00	ETSI
51	350	10130.00	10186.00	56.00	10480.00	10536.00	56.00	ETSI
55	350	10480.00	10536.00	56.00	10130.00	10186.00	56.00	ETSI
52	350	10144.00	10200.00	56.00	10494.00	10550.00	56.00	ETSI
56	350	10494.00	10550.00	56.00	10144.00	10200.00	56.00	ETSI
71	350	10000.00	10189.00	189.00	10350.00	10539.00	189.00	ETSI
75	350	10350.00	10539.00	189.00	10000.00	10189.00	189.00	ETSI
72	350	10125.00	10330.00	205.00	10475.00	10680.00	205.00	ETSI
76	350	10475.00	10680.00	205.00	10125.00	10330.00	205.00	ETSI
A11		10550.00	10570.00	20.00	10615.00	10635.00	20.00	ANSI
A15		10615.00	10635.00	20.00	10550.00	10570.00	20.00	ANSI
A12		10565.00	10585.00	20.00	10630.00	10650.00	20.00	ANSI
A16		10630.00	10650.00	20.00	10565.00	10585.00	20.00	ANSI
A13		10580.00	10600.00	20.00	10645.00	10665.00	20.00	ANSI
A17		10645.00	10665.00	20.00	10580.00	10600.00	20.00	ANSI
A14		10595.00	10615.00	20.00	10660.00	10680.00	20.00	ANSI
A18		10660.00	10680.00	20.00	10595.00	10615.00	20.00	ANSI
A21		10500.50	10528.50	28.00	10591.50	10619.50	28.00	ETSI
A25		10591.50	10619.50	28.00	10500.50	10528.50	28.00	ETSI
A22		10528.50	10556.50	28.00	10619.50	10647.50	28.00	ETSI
A26		10619.50	10647.50	28.00	10528.50	10556.50	28.00	ETSI
A23		10556.50	10584.50	28.00	10647.50	10675.50	28.00	ETSI
A27		10647.50	10675.50	28.00	10556.50	10584.50	28.00	ETSI
A41		10308.00	10420.00	112.00	10476.00	10588.00	112.00	ETSI
A45		10476.00	10588.00	112.00	10308.00	10420.00	112.00	ETSI
A51		10130.00	10186.00	56.00	10480.00	10536.00	56.00	ETSI
A55		10480.00	10536.00	56.00	10130.00	10186.00	56.00	ETSI
A52		10144.00	10200.00	56.00	10494.00	10550.00	56.00	ETSI

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
A56		10494.00	10550.00	56.00	10144.00	10200.00	56.00	ETSI
A71		10000.00	10189.00	189.00	10350.00	10539.00	189.00	ETSI
A75		10350.00	10539.00	189.00	10000.00	10189.00	189.00	ETSI
A72		10125.00	10330.00	205.00	10475.00	10680.00	205.00	ETSI
A76		10475.00	10680.00	205.00	10125.00	10330.00	205.00	ETSI

Sub-band 1x is valid for CS 10 and 20 MHz.

Sub-band 2x is valid for 7, 14, and 28 MHz.

Sub-band 3x, 4x, and 5x are valid for 7, 14, 28, and 56 MHz.

Sub-band A1x (Duplex 65 MHz) is valid for CS 10 and 20 MHz.

Sub-band A2x (Duplex 91 MHz) is valid for CS 7, 14, and 28 MHz.

Sub-band A4x (Duplex 168 MHz), A5x and A7x (Duplex 350 MHz) are valid for CS 7, 14, 28, and 56 MHz.

3.2.3.7 RAU2 X 11 GHz Band

Sub-band / Duplex (MHz)		Transmitter f	requency inform	nation	Receiver frequency information			Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
11	530	10695.00	10855.00	160.00	11225.00	11385.00	160.00	ETSI
15	530	11225.00	11385.00	160.00	10695.00	10855.00	160.00	ETSI
12	530	10835.00	11035.00	200.00	11365.00	11565.00	200.00	ETSI
16	530	11365.00	11565.00	200.00	10835.00	11035.00	200.00	ETSI
13	530	11015.00	11175.00	160.00	11545.00	11705.00	160.00	ETSI
17	530	11545.00	11705.00	160.00	11015.00	11175.00	160.00	ETSI
21	490	10695.00	10875.00	180.00	11185.00	11365.00	180.00	ETSI/ANSI
25	490	11185.00	11365.00	180.00	10695.00	10875.00	180.00	ETSI/ANSI
22	490	10855.00	11055.00	200.00	11345.00	11545.00	200.00	ETSI/ANSI
26	490	11345.00	11545.00	200.00	10855.00	11055.00	200.00	ETSI/ANSI
23	490	11035.00	11195.00	160.00	11525.00	11685.00	160.00	ETSI/ANSI
27	490	11525.00	11685.00	160.00	11035.00	11195.00	160.00	ETSI/ANSI
A01		10695.00	10875.00	180.00	11185.00	11385.00	200.00	ETSI/ANSI
A05		11185.00	11385.00	200.00	10695.00	10875.00	180.00	ETSI/ANSI
A02		10835.00	11055.00	220.00	11345.00	11565.00	220.00	ETSI/ANSI
A06		11345.00	11565.00	220.00	10835.00	11055.00	220.00	ETSI/ANSI
A03		11015.00	11200.00	185.00	11525.00	11705.00	180.00	ETSI/ANSI
A07		11525.00	11705.00	180.00	11015.00	11200.00	185.00	ETSI/ANSI

Table 15 RAU2 X 11 [HP]

Sub-band 1x and 2x are valid for 7, 14, 28, 40, and 56 MHz.

Sub-band 2x is valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band A0x (Duplex 490 and 530 MHz) is valid for CS 7, 14, 28, 40, and 56 MHz.

Sub-band A0x (Duplex 490 and 500 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.3.8 RAU2 X 13 GHz Band

Table 16 RAU2 X 13 [HP]

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver frequency information			Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
11	266	12751.00	12835.00	84.00	13017.00	13101.00	84.00	ETSI/ANSI
15	266	13017.00	13101.00	84.00	12751.00	12835.00	84.00	ETSI/ANSI
12	266	12821.00	12905.00	84.00	13087.00	13171.00	84.00	ETSI/ANSI
16	266	13087.00	13171.00	84.00	12821.00	12905.00	84.00	ETSI/ANSI
13	266	12891.00	12975.00	84.00	13157.00	13241.00	84.00	ETSI/ANSI
17	266	13157.00	13241.00	84.00	12891.00	12975.00	84.00	ETSI/ANSI
14	266	12863.00	12919.00	56.00	13129.00	13185.00	56.00	ETSI/ANSI
18	266	13129.00	13185.00	56.00	12863.00	12919.00	56.00	ETSI/ANSI
42	266	12807.00	12891.00	84.00	13073.00	13157.00	84.00	ETSI/ANSI
46	266	13073.00	13157.00	84.00	12807.00	12891.00	84.00	ETSI/ANSI
A01		12751.00	12863.00	112.00	13017.00	13129.00	112.00	ETSI/ANSI
A05		13017.00	13129.00	112.00	12751.00	12863.00	112.00	ETSI/ANSI
A02		12807.00	12905.00	98.00	13073.00	13171.00	98.00	ETSI/ANSI
A06		13073.00	13171.00	98.00	12807.00	12905.00	98.00	ETSI/ANSI
A03		12863.00	12975.00	112.00	13129.00	13241.00	112.00	ETSI/ANSI
A07		13129.00	13241.00	112.00	12863.00	12975.00	112.00	ETSI/ANSI

Note: FCC part 101.147(p)(2) frequency range begins at 12700.00 MHz.

Sub-band 1x and 4x are valid for 7, 14, 28, and 56 MHz.

Sub-band 1x and 4x are valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band A0x (Duplex 266 MHz) is valid for 7, 14, 28, and 56 MHz.

Sub-band A0x (Duplex 225 and 266 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

For RAU2 X/Xu 13 [HP] CS=12.5 MHz and 25 MHz (10 MHz in a 12.5 MHz CS and 20 MHz in a 25 MHz CS) are supported.

3.2.3.9 RAU2 X 15 GHz Band

Table 17 RAU2 X 15 [HP]

Sub-band / Duplex (MHz)		Transmitter f	requency inform	nation	Receiver frequency information			Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
11	735	14499.25	14613.00	113.75	15234.25	15348.00	113.75	ETSI
18	735	15234.25	15348.00	113.75	14499.25	14613.00	113.75	ETSI
12	315	14604.25	14732.00	127.75	14919.25	15047.00	127.75	ETSI
15	315	14919.25	15047.00	127.75	14604.25	14732.00	127.75	ETSI
13	315	14709.25	14828.50	119.25	15024.25	15143.50	119.25	ETSI
16	315	15024.25	15143.50	119.25	14709.25	14828.50	119.25	ETSI
14	315	14814.25	14933.50	119.25	15129.25	15248.50	119.25	ETSI
17	315	15129.25	15248.50	119.25	14814.25	14933.50	119.25	ETSI
21	420	14495.75	14620.00	124.25	14915.75	15040.00	124.25	ETSI
25	420	14915.75	15040.00	124.25	14495.75	14620.00	124.25	ETSI
22	420	14607.75	14732.00	124.25	15027.75	15152.00	124.25	ETSI
26	420	15027.75	15152.00	124.25	14607.75	14732.00	124.25	ETSI
23	420	14719.75	14837.00	117.25	15139.75	15257.00	117.25	ETSI
27	420	15139.75	15257.00	117.25	14719.75	14837.00	117.25	ETSI
24	420	14817.75	14928.00	110.25	15237.75	15348.00	110.25	ETSI
28	420	15237.75	15348.00	110.25	14817.75	14928.00	110.25	ETSI
43	644	14495.75	14700.50	204.75	15139.75	15344.50	204.75	ETSI
49	644	15139.75	15344.50	204.75	14495.75	14700.50	204.75	ETSI
51	475	14700.00	14800.00	100.00	15175.00	15275.00	100.00	ANSI
53	475	15175.00	15275.00	100.00	14700.00	14800.00	100.00	ANSI
52	475	14775.00	14875.00	100.00	15250.00	15350.00	100.00	ANSI
54	475	15250.00	15350.00	100.00	14775.00	14875.00	100.00	ANSI
55	475	14500.00	14620.00	120.00	14975.00	15095.00	120.00	ANSI
57	475	14975.00	15095.00	120.00	14500.00	14620.00	120.00	ANSI
56	475	14600.00	14720.00	120.00	15075.00	15195.00	120.00	ANSI
58	475	15075.00	15195.00	120.00	14600.00	14720.00	120.00	ANSI
61	728	14495.75	14620.00	124.25	15223.75	15348.00	124.25	ETSI
68	728	15223.75	15348.00	124.25	14495.75	14620.00	124.25	ETSI
62	308	14614.75	14735.50	120.75	14922.75	15043.50	120.75	ETSI
65	308	14922.75	15043.50	120.75	14614.75	14735.50	120.75	ETSI
63	308	14698.75	14819.50	120.75	15006.75	15127.50	120.75	ETSI
66	308	15006.75	15127.50	120.75	14698.75	14819.50	120.75	ETSI
64	308	14810.75	14931.50	120.75	15118.75	15239.50	120.75	ETSI
67	308	15118.75	15239.50	120.75	14810.75	14931.50	120.75	ETSI
80	490	14397.75	14522.00	124.25	14887.75	15012.00	124.25	ETSI
85	490	14887.75	15012.00	124.25	14397.75	14522.00	124.25	ETSI
81	490	14481.75	14606.00	124.25	14971.75	15096.00	124.25	ETSI
86	490	14971.75	15096.00	124.25	14481.75	14606.00	124.25	ETSI
Sub-band / I	Duplex (MHz)	Transmitter f	requency inform	nation	Receiver free	juency informat	tion	Telecom
--------------	--------------	------------------	------------------	----------------	------------------	------------------	----------------	-----------
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
82	490	14565.75	14690.00	124.25	15055.75	15180.00	124.25	ETSI
87	490	15055.75	15180.00	124.25	14565.75	14690.00	124.25	ETSI
83	490	14649.75	14774.00	124.25	15139.75	15264.00	124.25	ETSI
88	490	15139.75	15264.00	124.25	14649.75	14774.00	124.25	ETSI
84	490	14733.75	14858.00	124.25	15223.75	15348.00	124.25	ETSI
89	490	15223.75	15348.00	124.25	14733.75	14858.00	124.25	ETSI
92	322	14627.00	14739.00	112.00	14949.00	15061.00	112.00	ETSI
95	322	14949.00	15061.00	112.00	14627.00	14739.00	112.00	ETSI
93	322	14711.00	14823.00	112.00	15033.00	15145.00	112.00	ETSI
96	322	15033.00	15145.00	112.00	14711.00	14823.00	112.00	ETSI
94	322	14795.00	14907.00	112.00	15117.00	15229.00	112.00	ETSI
97	322	15117.00	15229.00	112.00	14795.00	14907.00	112.00	ETSI
A01		14495.75	14774.00	278.25	15055.75	15348.00	292.25	ETSI
A05		15055.75	15348.00	292.25	14495.75	14774.00	278.25	ETSI
A02		14604.25	14739.00	134.75	14919.25	15061.00	141.75	ETSI
A06		14919.25	15061.00	141.75	14604.25	14739.00	134.75	ETSI
A03		14698.75	14828.50	129.75	15006.75	15145.00	138.25	ETSI
A07		15006.75	15145.00	138.25	14698.75	14828.50	129.75	ETSI
A04		14795.00	14933.50	138.50	15117.00	15248.50	131.50	ETSI
A08		15117.00	15248.50	131.50	14795.00	14933.50	138.50	ETSI
A11		14397.75	14660.00	262.25	14887.75	15130.00	242.25	ETSI/ANSI
A15		14887.75	15130.00	242.25	14397.75	14660.00	262.25	ETSI/ANSI
A12		14600.00	14732.00	132.00	15027.75	15195.00	167.25	ETSI/ANSI
A16		15027.75	15195.00	167.25	14600.00	14732.00	132.00	ETSI/ANSI
A13		14700.00	14928.00	228.00	15139.75	15350.00	210.25	ETSI/ANSI
A17		15139.75	15350.00	210.25	14700.00	14928.00	228.00	ETSI/ANSI

Sub-band 1x, 2x, 6x, 8x, and 9x are valid for CS 7, 14, 28, and 56 MHz.

Sub-band 5x are valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band A0x (Duplex 308, 315, 322, 490, 644, 728, and 735 MHz) and A1x (Duplex 420, 470, and 490 MHz) are valid for CS 7, 14, 28, and 56 MHz.

Sub-band A1x (Duplex 420, 470, and 490 MHz) are valid for CS 7, 14, 28, and 56 MHz.

Sub-band A1x (Duplex 475 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.3.10 RAU2 X 18 GHz Band

Table 18 RAU2 X 18 [HP]

Sub-band / [Duplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
11	1010	17706.50	18009.50	303.00	18716.50	19019.50	303.00	ETSI
15	1010	18716.50	19019.50	303.00	17706.50	18009.50	303.00	ETSI
12	1010	17933.50	18236.50	303.00	18943.50	19246.50	303.00	ETSI
16	1010	18943.50	19246.50	303.00	17933.50	18236.50	303.00	ETSI
13	1010	18153.50	18456.50	303.00	19163.50	19466.50	303.00	ETSI
17	1010	19163.50	19466.50	303.00	18153.50	18456.50	303.00	ETSI
14	1010	18373.50	18676.50	303.00	19383.50	19686.50	303.00	ETSI
18	1010	19383.50	19686.50	303.00	18373.50	18676.50	303.00	ETSI
21	340	18580.00	18670.00	90.00	18920.00	19010.00	90.00	ANSI
25	340	18920.00	19010.00	90.00	18580.00	18670.00	90.00	ANSI
22	340	18655.00	18745.00	90.00	18995.00	19085.00	90.00	ANSI
26	340	18995.00	19085.00	90.00	18655.00	18745.00	90.00	ANSI
23	340	18730.00	18830.00	100.00	19070.00	19170.00	100.00	ANSI
27	340	19070.00	19170.00	100.00	18730.00	18830.00	100.00	ANSI
24	340	18820.00	18920.00	100.00	19160.00	19260.00	100.00	ANSI
28	340	19160.00	19260.00	100.00	18820.00	18920.00	100.00	ANSI
31	1560	17700.00	18003.00	303.00	19260.00	19563.00	303.00	ANSI
35	1560	19260.00	19563.00	303.00	17700.00	18003.00	303.00	ANSI
32	1560	17837.00	18140.00	303.00	19397.00	19700.00	303.00	ANSI
36	1560	19397.00	19700.00	303.00	17837.00	18140.00	303.00	ANSI
41	1008	17720.50	18009.50	289.00	18728.50	19017.50	289.00	ETSI
45	1008	18728.50	19017.50	289.00	17720.50	18009.50	289.00	ETSI
42	1008	17935.50	18236.50	301.00	18943.50	19244.50	301.00	ETSI
46	1008	18943.50	19244.50	301.00	17935.50	18236.50	301.00	ETSI
43	1008	18155.50	18456.50	301.00	19163.50	19464.50	301.00	ETSI
47	1008	19163.50	19464.50	301.00	18155.50	18456.50	301.00	ETSI
44	1008	18375.50	18672.50	297.00	19383.50	19680.50	297.00	ETSI
48	1008	19383.50	19680.50	297.00	18375.50	18672.50	297.00	ETSI
71	1010	17706.50	18009.50	303.00	18716.50	19019.50	303.00	ETSI
75	1010	18716.50	19019.50	303.00	17706.50	18009.50	303.00	ETSI
72	1010	17933.50	18236.50	303.00	18943.50	19246.50	303.00	ETSI
76	1010	18943.50	19246.50	303.00	17933.50	18236.50	303.00	ETSI
73	1010	18153.50	18456.50	303.00	19163.50	19466.50	303.00	ETSI
77	1010	19163.50	19466.50	303.00	18153.50	18456.50	303.00	ETSI
74	1010	18373.50	18676.50	303.00	19383.50	19686.50	303.00	ETSI
78	1010	19383.50	19686.50	303.00	18373.50	18676.50	303.00	ETSI
A11		17706.50	18236.50	530.00	18600.00	19246.50	646.50	ANSI
A15		18600.00	19246.50	646.50	17706.50	18236.50	530.00	ANSI

Sub-band / Duplex (MHz)		Transmitter f	requency inform	nation	Receiver frequency information			Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
A13		18002.25	18676.50	674.25	19094.75	19686.50	591.75	ANSI
A17		19094.75	19686.50	591.75	18002.25	18676.50	674.25	ANSI
A21		18580.00	18670.00	90.00	18920.00	19010.00	90.00	ANSI
A25		18920.00	19010.00	90.00	18580.00	18670.00	90.00	ANSI
A22		18655.00	18745.00	90.00	18995.00	19085.00	90.00	ANSI
A26		18995.00	19085.00	90.00	18655.00	18745.00	90.00	ANSI
A23		18730.00	18830.00	100.00	19070.00	19170.00	100.00	ANSI
A27		19070.00	19170.00	100.00	18730.00	18830.00	100.00	ANSI
A24		18820.00	18920.00	100.00	19160.00	19260.00	100.00	ANSI
A28		19160.00	19260.00	100.00	18820.00	18920.00	100.00	ANSI
A31		17700.00	18140.00	440.00	19260.00	19700.00	440.00	ANSI
A35		19260.00	19700.00	440.00	17700.00	18140.00	440.00	ANSI

Sub-band 1x, 4x, and 7x are valid for CS 7, 13.75/14, 27.5/28, and 55/56 MHz.

Sub-band 2x and 3x are valid for CS 7, 13.75/14, 27.5/28, and 55/56 MHz.

Sub-band 2x and 3x are valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band A1x (Duplex 1008 and 1010 MHz), A2x (Duplex 340 MHz), and A3x (Duplex 1560 MHz) are valid for CS 7, 13.75/14, 27.5/28, and 55/56 MHz.

Sub-band A1x (Duplex 1160 MHz), A2x (Duplex 340 MHz), and A3x (Duplex 1560 MHz) are valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.3.11 RAU2 X 23 GHz Band

Table 19 RAU2 X 23 [HP]

Sub-band / Duplex (MHz)		Transmitt	er frequency in	formation	Receive	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
72	1008	22100.00	22400.00	300.00	23108.00	23408.00	300.00	ETSI
73	1008	23108.00	23408.00	300.00	22100.00	22400.00	300.00	ETSI
74	1008	22217.75	22540.25	322.50	23225.75	23548.25	322.50	ETSI
75	1008	23225.75	23548.25	322.50	22217.75	22540.25	322.50	ETSI
76	1008	22002.75	22316.25	313.50	23010.75	23324.25	313.50	ETSI
78	1008	23010.75	23324.25	313.50	22002.75	22316.25	313.50	ETSI
77	1008	22274.00	22590.75	316.75	23282.00	23598.75	316.75	ETSI
79	1008	23282.00	23598.75	316.75	22274.00	22590.75	316.75	ETSI
81	1200	21218.25	21523.25	305.00	22418.25	22723.25	305.00	ETSI/ANSI
85	1200	22418.25	22723.25	305.00	21218.25	21523.25	305.00	ETSI/ANSI

		Transmitt	ter frequency in	formation	Receive	r frequency info	ormation	Telesem
Sub-band / I	Duplex (MHz)	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
82	1200	21494.75	21824.75	330.00	22694.75	23024.75	330.00	ETSI/ANSI
86	1200	22694.75	23024.75	330.00	21494.75	21824.75	330.00	ETSI/ANSI
83	1200	21796.25	22145.25	349.00	22996.25	23345.25	349.00	ETSI/ANSI
87	1200	22996.25	23345.25	349.00	21796.25	22145.25	349.00	ETSI/ANSI
84	1200	22094.75	22400.00	305.25	23294.75	23600.00	305.25	ETSI/ANSI
88	1200	23294.75	23600.00	305.25	22094.75	22400.00	305.25	ETSI/ANSI
91	1232	21224.00	21532.25	308.25	22456.00	22764.25	308.25	ETSI/ANSI
95	1232	22456.00	22764.25	308.25	21224.00	21532.25	308.25	ETSI/ANSI
92	1232	21503.75	21812.25	308.50	22735.75	23044.25	308.50	ETSI/ANSI
96	1232	22735.75	23044.25	308.50	21503.75	21812.25	308.50	ETSI/ANSI
93	1232	21784.00	22092.25	308.25	23016.00	23324.25	308.25	ETSI/ANSI
97	1232	23016.00	23324.25	308.25	21784.00	22092.25	308.25	ETSI/ANSI
94	1232	22049.75	22363.25	313.50	23281.75	23595.25	313.50	ETSI/ANSI
98	1232	23281.75	23595.25	313.50	22049.75	22363.25	313.50	ETSI/ANSI
101	1050	21950.25	22263.75	313.50	23000.25	23313.75	313.50	ETSI
103	1050	23000.25	23313.75	313.50	21950.25	22263.75	313.50	ETSI
102	1050	22235.25	22538.25	303.00	23285.25	23588.25	303.00	ETSI
104	1050	23285.25	23588.25	303.00	22235.25	22538.25	303.00	ETSI
111	1200	21200.00	21523.25	323.25	22400.00	22723.25	323.25	ETSI
115	1200	22400.00	22723.25	323.25	21200.00	21523.25	323.25	ETSI
A01		21200.00	21824.75	624.75	22400.00	23044.25	644.25	ETSI/ANSI
A05		22400.00	23044.25	644.25	21200.00	21824.75	624.75	ETSI/ANSI
A02		21784.00	22600.00	816.00	22996.25	23600.00	603.75	ETSI/ANSI
A06		22996.25	23600.00	603.75	21784.00	22600.00	816.00	ETSI/ANSI

Sub-band 7x, 8x, 9x, 10x, and 11x are valid for CS 7, 14, 28, and 56 MHz.

Sub-band 8x and 9x are valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band A0x (Duplex 1008, 1050, 1200, and 1232 MHz) is valid for CS 7, 14, 28, and 56 MHz.

Sub-band A0x (Duplex 1200 and 1232 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.3.12 RAU2 X 24 GHz Band

Table 20 RAU2 X 24

		Transmitt	Transmitter frequency information			Receiver frequency information			
Sub-band / I	Duplex (MHz)	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard	
41	800	24250.00	24450.00	200.00	25050.00	25250.00	200.00	ANSI	
45	800	25050.00	25250.00	200.00	24250.00	24450.00	200.00	ANSI	
A41		24250.00	24450.00	200.00	25050,00	25250,00	200.00	ANSI	
A45		25050.00	25250.00	200.00	24250,00	24450,00	200.00	ANSI	

Sub-band 4x is valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band A4x (Duplex 800 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.3.13 RAU2 X 26 GHz Band

Table 21 RAU2 X 26

Sub-band / Duplex (MHz)		Transmitt	er frequency in	formation	Receive	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
51	1008	24549.00	24885.00	336.00	25557.00	25893.00	336.00	ETSI
55	1008	25557.00	25893.00	336.00	24549.00	24885.00	336.00	ETSI
52	1008	24829.00	25165.00	336.00	25837.00	26173.00	336.00	ETSI
56	1008	25837.00	26173.00	336.00	24829.00	25165.00	336.00	ETSI
53	1008	25109.00	25445.00	336.00	26117.00	26453.00	336.00	ETSI
57	1008	26117.00	26453.00	336.00	25109.00	25445.00	336.00	ETSI
A01		24549.00	25165.00	616.00	25557.00	26173.00	616.00	ETSI
A05		25557.00	26173.00	616.00	24549.00	25165.00	616.00	ETSI
A02		24883.25	25469.00	585.75	25891.25	26477.00	585.75	ETSI
A06		25891.25	26477.00	585.75	24883.25	25469.00	585.75	ETSI

Sub-band 5x is valid for CS 7, 14, 28, and 56 MHz.

Sub-band A0x (Duplex 1008 MHz) is valid for CS 7, 14, 28, and 56 MHz.

3.2.3.14 RAU2 X 28 GHz Band

Table 22 RAU2 X 28

Sub-band / D	uplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
31	420	27505.00	27701.00	196.00	27925.00	28121.00	196.00	ANSI
33	420	27925.00	28121.00	196.00	27505.00	27701.00	196.00	ANSI
32	420	27701.00	27925.00	224.00	28121.00	28345.00	224.00	ANSI
34	420	28121.00	28345.00	224.00	27701.00	27925.00	224.00	ANSI
41	1008	27548.50	27884.50	336.00	28556.50	28892.50	336.00	ETSI
45	1008	28556.50	28892.50	336.00	27548.50	27884.50	336.00	ETSI
42	1008	27828.50	28164.50	336.00	28836.50	29172.50	336.00	ETSI
46	1008	28836.50	29172.50	336.00	27828.50	28164.50	336.00	ETSI
43	1008	28108.50	28444.50	336.00	29116.50	29452.50	336.00	ETSI
47	1008	29116.50	29452.50	336.00	28108.50	28444.50	336.00	ETSI
51	450	27500.00	27700.00	200.00	27950.00	28150.00	200.00	ANSI
53	450	27950.00	28150.00	200.00	27500.00	27700.00	200.00	ANSI
52	450	27700.00	27900.00	200.00	28150.00	28350.00	200.00	ANSI
54	450	28150.00	28350.00	200.00	27700.00	27900.00	200.00	ANSI
A01		27548.50	27996.50	448.00	28556.50	29004.50	448.00	ETSI
A05		28556.50	29004.50	448.00	27548.50	27996.50	448.00	ETSI
A02		27996.50	28444.50	448.00	29004.50	29452.50	448.00	ETSI
A06		29004.50	29452.50	448.00	27996.50	28444.50	448.00	ETSI
A03		27828.50	28276.50	448.00	28836.50	29284.50	448.00	ETSI
A07		28836.50	29284.50	448.00	27828.50	28276.50	448.00	ETSI
A11		27500.00	27701.00	201.00	27925.00	28150.00	225.00	ANSI
A15		27925.00	28150.00	225.00	27500.00	27701.00	201.00	ANSI
A12		27700.00	27925.00	225.00	28121.00	28350.00	229.00	ANSI
A16		28121.00	28350.00	229.00	27700.00	27925.00	225.00	ANSI

Sub-band 3x and 5x are valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band 4x is valid for CS 7, 14, 28, and 56 MHz.

Sub-band A0x (Duplex 1008 MHz) is valid for CS 7, 14, 28, and 56 MHz.

Sub-band A1x (Duplex 420 and 450 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.3.15 RAU2 X 32 GHz Band

Table 23 RAU2 X 32

Sub-band / D	uplex (MHz)	Transmitter frequency information			Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
11	812	31815.00	32095.00	280.00	32627.00	32907.00	280.00	ETSI
15	812	32627.00	32907.00	280.00	31815.00	32095.00	280.00	ETSI
12	812	32053.00	32333.00	280.00	32865.00	33145.00	280.00	ETSI
16	812	32865.00	33145.00	280.00	32053.00	32333.00	280.00	ETSI
13	812	32291.00	32599.00	308.00	33103.00	33411.00	308.00	ETSI
17	812	33103.00	33411.00	308.00	32291.00	32599.00	308.00	ETSI
A11		31815.00	32207.00	392.00	32627.00	33019.00	392.00	ETSI
A15		32627.00	33019.00	392.00	31815.00	32207.00	392.00	ETSI
A12		32053.00	32333.00	280.00	32865.00	33145.00	280.00	ETSI
A16		32865.00	33145.00	280.00	32053.00	32333.00	280.00	ETSI
A13		32207.00	32599.00	392.00	33019.00	33411.00	392.00	ETSI
A17		33019.00	33411.00	392.00	32207.00	32599.00	392.00	ETSI

Sub-band 1x is valid for CS 7, 14, 28, and 56 MHz.

Sub-band A1x (Duplex 812 MHz) is valid for CS 7, 14, 28, and 56 MHz.

3.2.3.16 RAU2 X 38 GHz Band

Table 24 RAU2 X 38

Sub-band / D	Duplex (MHz)	Transmitter f	requency inforr	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
11	1260	37058.00	37339.75	281.75	38318.00	38599.75	281.75	ETSI
15	1260	38318.00	38599.75	281.75	37058.00	37339.75	281.75	ETSI
12	1260	37338.00	37619.75	281.75	38598.00	38879.75	281.75	ETSI
16	1260	38598.00	38879.75	281.75	37338.00	37619.75	281.75	ETSI
13	1260	37618.00	37899.75	281.75	38878.00	39159.75	281.75	ETSI
17	1260	38878.00	39159.75	281.75	37618.00	37899.75	281.75	ETSI
14	1260	37898.00	38179.75	281.75	39158.00	39439.75	281.75	ETSI
18	1260	39158.00	39439.75	281.75	37898.00	38179.75	281.75	ETSI
23	1260	37758.00	38039.75	281.75	39018.00	39299.75	281.75	ETSI
27	1260	39018.00	39299.75	281.75	37758.00	38039.75	281.75	ETSI
31	700	38600.00	38800.00	200.00	39300.00	39500.00	200.00	ANSI
35	700	39300.00	39500.00	200.00	38600.00	38800.00	200.00	ANSI
32	700	38770.00	38970.00	200.00	39470.00	39670.00	200.00	ANSI
36	700	39470.00	39670.00	200.00	38770.00	38970.00	200.00	ANSI
33	700	38930.00	39130.00	200.00	39630.00	39830.00	200.00	ANSI

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
37	700	39630.00	39830.00	200.00	38930.00	39130.00	200.00	ANSI
34	700	39100.00	39300.00	200.00	39800.00	40000.00	200.00	ANSI
38	700	39800.00	40000.00	200.00	39100.00	39300.00	200.00	ANSI
A11		37058.00	37619.75	561.75	38318.00	38879.75	561.75	ETSI
A15		38318.00	38879.75	561.75	37058.00	37619.75	561.75	ETSI
A12		37618.00	38179.75	561.75	38878.00	39439.75	561.75	ETSI
A16		38878.00	39439.75	561.75	37618.00	38179.75	561.75	ETSI
A31		38600.00	38800.00	200.00	39300.00	39500.00	200.00	ANSI
A35		39300.00	39500.00	200.00	38600.00	38800.00	200.00	ANSI
A32		38770.00	38970.00	200.00	39470.00	39670.00	200.00	ANSI
A36		39470.00	39670.00	200.00	38770.00	38970.00	200.00	ANSI
A33		38930.00	39130.00	200.00	39630.00	39830.00	200.00	ANSI
A37		39630.00	39830.00	200.00	38930.00	39130.00	200.00	ANSI
A34		39100.00	39300.00	200.00	39800.00	40000.00	200.00	ANSI
A38		39800.00	40000.00	200.00	39100.00	39300.00	200.00	ANSI

Sub-band 1x and 2x are valid for CS 7, 14, 28, and 56 MHz.

Sub-band 3x is valid for CS 10, 20, 30, 40, and 50 MHz for ANSI.

Sub-band A1x (Duplex 1260 MHz) is valid for CS 7, 14, 2, and 56 MHz.

Sub-band A3x (Duplex 700 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.3.17 RAU2 X 42 GHz Band

Table 25 RAU2 X 42

Sub-band / Duplex (MHz)		Transmitter f	requency inform	nation	Receiver freq	ion	Telecom Standard	
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
11	1500	40522.00	40830.00	308.00	42022.00	42330.00	308.00	ETSI
15	1500	42022.00	42330.00	308.00	40522.00	40830.00	308.00	ETSI
12	1500	40774.00	41054.00	280.00	42274.00	42554.00	280.00	ETSI
16	1500	42274.00	42554.00	280.00	40774.00	41054.00	280.00	ETSI
21	1500	40998.00	41278.00	280.00	42498.00	42778.00	280.00	ETSI
25	1500	42498.00	42778.00	280.00	40998.00	41278.00	280.00	ETSI
22	1500	41222.00	41502.00	280.00	42722.00	43002.00	280.00	ETSI
26	1500	42722.00	43002.00	280.00	41222.00	41502.00	280.00	ETSI
31	1500	41446.00	41726.00	280.00	42946.00	43226.00	280.00	ETSI
35	1500	42946.00	43226.00	280.00	41446.00	41726.00	280.00	ETSI
32	1500	41670.00	41964.00	294.00	43170.00	43464.00	294.00	ETSI
36	1500	43170.00	43464.00	294.00	41670.00	41964.00	294.00	ETSI
A01		40522.00	41054.00	532.00	42022.00	42554.00	532.00	ETSI

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
A05		42022.00	42554.00	532.00	40522.00	41054.00	532.00	ETSI
A02		40998.00	41502.00	504.00	42498.00	43002.00	504.00	ETSI
A06		42498.00	43002.00	504.00	40998.00	41502.00	504.00	ETSI
A03		41446.00	41964.00	518.00	42946.00	43464.00	518.00	ETSI
A07		42946.00	43464.00	518.00	41446.00	41964.00	518.00	ETSI

Sub-band 1x, 2x, and 3x are valid for CS 7, 14, 28, and 56 MHz.

Sub-band A0x (Duplex 1500 MHz) is valid for CS 7, 14, 28, and 56 MHz.

3.2.4 MINI-LINK 6363

3.2.4.1 MINI-LINK 6363 6 GHz Lower Band

Table 26 MINI-LINK 6363 6L

Sub-band / Duplex (MHz)		Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom Standard
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
11L		5925.000	6020.250	95.250	6177.050	6277.000	99.950	ETSI/ANSI
11H		6177.050	6277.000	99.950	5925.000	6020.250	95.250	ETSI/ANSI
12L		6011.000	6110.250	99.250	6270.050	6362.300	92.250	ETSI/ANSI
12H		6270.050	6362.300	92.250	6011.000	6110.250	99.250	ETSI/ANSI
13L		6078.500	6173.250	94.750	6330.550	6425.300	94.750	ETSI/ANSI
13H		6330.550	6425.300	94.750	6078.500	6173.250	94.750	ETSI/ANSI
14L		5989.325	6049.350	60.025	6241.350	6301.375	60.025	ETSI/ANSI
14H		6241.350	6301.375	60.025	5989.325	6049.350	60.025	ETSI/ANSI
51L		5925.000	6020.250	95.250	6177.050	6277.000	99.950	ANSI
51H		6177.050	6277.000	99.950	5925.000	6020.250	95.250	ANSI
52L		6011.000	6110.250	99.250	6270.050	6362.300	92.250	ANSI
52H		6270.050	6362.300	92.250	6011.000	6110.250	99.250	ANSI
53L		6078.500	6173.250	94.750	6330.550	6425.300	94.750	ANSI
53H		6330.550	6425.300	94.750	6078.500	6173.250	94.750	ANSI
54L		5989.330	6049.330	60.000	6241.370	6301.370	60.000	ANSI
54H		6241.370	6301.370	60.000	5989.330	6049.330	60.000	ANSI

Sub-band 1xL/H (Duplex 240, 252.04, 260, and 266 MHz) is valid for CS 7, 14, 28/29.65, and 56/59.3 MHz.

Sub-band 1xL/H (Duplex 252.04 MHz) are valid for CS 9.88/10, 20, 29.65/30, 40, 50, and 59.3/60 MHz.

Sub-band 5xL/H (Duplex 252.04 MHz) are valid for CS 9.88/10, 20, 29.65/30, 40, 50, and 59.3/60 MHz.

3.2.4.2 MINI-LINK 6363 6 GHz Upper Band

Table 27 MINI-LINK 6363 6U

Sub-band / Duplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
21L	6430.000	6565.000	135.000	6770.000	6905.000	135.000	ETSI/ANSI
21H	6770.000	6905.000	135.000	6430.000	6565.000	135.000	ETSI/ANSI
22L	6550.000	6685.000	135.000	6890.000	7025.000	135.000	ETSI/ANSI
22H	6890.000	7025.000	135.000	6550.000	6685.000	135.000	ETSI/ANSI
23L	6635.000	6765.000	130.000	6975.000	7105.000	130.000	ETSI/ANSI
23H	6975.000	7105.000	130.000	6635.000	6765.000	130.000	ETSI/ANSI
24L	6520.000	6600.000	80.000	6860.000	6940.000	80.000	ETSI/ANSI
24H	6860.000	6940.000	80.000	6520.000	6600.000	80.000	ETSI/ANSI
31L	6540.000	6610.000	70.000	6700.000	6770.000	70.000	ANSI
31H	6700.000	6770.000	70.000	6540.000	6610.000	70.000	ANSI
32L	6597.500	6657.500	60.000	6757.500	6817.500	60.000	ANSI
32H	6757.500	6817.500	60.000	6597.500	6657.500	60.000	ANSI
33L	6640.000	6710.000	70.000	6800.000	6870.000	70.000	ANSI
33H	6800.000	6870.000	70.000	6640.000	6710.000	70.000	ANSI

Sub-band 2xL/H (Duplex 340 MHz) is valid for CS 7, 14, 28/30, 40, and 56/60 MHz.

Sub-band 2xL/H (Duplex 340 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band 3xL/H (Duplex 160 and 170 MHz) are valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.4.3 MINI-LINK 6363 7 GHz Band

Table 28 MINI-LINK 6363 7

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
01L		7107.000	7163.000	56.00	7303.000	7359.000	56.00	ETSI
01H		7303.000	7359.000	56.00	7107.000	7163.000	56.00	ETSI
02L		7163.000	7219.000	56.00	7359.000	7415.000	56.00	ETSI
02H		7359.000	7415.000	56.00	7163.000	7219.000	56.00	ETSI
03L		7191.000	7247.000	56.00	7387.000	7443.000	56.00	ETSI
03H		7387.000	7443.000	56.00	7191.000	7247.000	56.00	ETSI

Sub-band / Duplex (MHz)	Transmitter	frequency infor	mation	Receiver free	quency informo	ition	Telecom Standard
	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
11L	7414.000	7485.000	71.00	7575.000	7645.000	70.00	ETSI/ANSI
11H	7575.000	7645.000	70.00	7414.000	7485.000	71.00	ETSI/ANSI
12L	7442.000	7515.000	73.00	7610.000	7673.000	63.00	ETSI/ANSI
12H	7610.000	7673.000	63.00	7442.000	7515.000	73.00	ETSI/ANSI
13L	7470.000	7540.000	70.00	7638.000	7701.000	63.00	ETSI/ANSI
13H	7638.000	7701.000	63.00	7470.000	7540.000	70.00	ETSI/ANSI
14L	7498.000	7568.000	70.00	7665.000	7729.000	64.00	ETSI/ANSI
14H	7665.000	7729.000	64.00	7498.000	7568.000	70.00	ETSI/ANSI
21L	7426.500	7513.750	87.250	7671.500	7758.750	87.250	ETSI
21H	7671.500	7758.750	87.250	7426.500	7513.750	87.250	ETSI
22L	7482.250	7569.750	87.500	7727.250	7814.750	87.500	ETSI
22H	7727.250	7814.750	87.500	7482.250	7569.750	87.500	ETSI
23L	7510.250	7597.750	87.500	7755.250	7842.750	87.500	ETSI
23H	7755.250	7842.750	87.500	7510.250	7597.750	87.500	ETSI
24L	7566.250	7653.750	87.500	7811.250	7898.750	87.500	ETSI
24H	7811.250	7898.750	87.500	7566.250	7653.750	87.500	ETSI
31L	7107.000	7184.000	77.00	7275.000	7345.000	70.00	ETSI/ANSI
31H	7275.000	7345.000	70.00	7107.000	7184.000	77.00	ETSI/ANSI
32L	7135.000	7212.000	77.00	7303.000	7373.000	70.00	ETSI/ANSI
32H	7303.000	7373.000	70.00	7135.000	7212.000	77.00	ETSI/ANSI
33L	7163.000	7240.000	77.00	7331.000	7401.000	70.00	ETSI/ANSI
33H	7331.000	7401.000	70.00	7163.000	7240.000	77.00	ETSI/ANSI
34L	7191.000	7268.000	77.00	7359.000	7429.000	70.00	ETSI/ANSI
34H	7359.000	7429.000	70.00	7191.000	7268.000	77.00	ETSI/ANSI
81L	7249.500	7309.000	59.50	7410.500	7470.000	59.50	ETSI
81H	7410.500	7470.000	59.50	7249.500	7309.000	59.50	ETSI
82L	7277.500	7337.000	59.50	7438.500	7498.000	59.50	ETSI
82H	7438.500	7498.000	59.50	7277.500	7337.000	59.50	ETSI
83L	7305.500	7365.000	59.50	7466.500	7526.000	59.50	ETSI
83H	7466.500	7526.000	59.50	7305.500	7365.000	59.50	ETSI
84L	7333.500	7393.000	59.50	7494.500	7554.000	59.50	ETSI
84H	7494.500	7554.000	59.50	7333.500	7393.000	59.50	ETSI
94L	7527.000	7583.000	56.00	7695.000	7751.000	56.00	ETSI/ANSI
94H	7695.000	7751.000	56.00	7527.000	7583.000	56.00	ETSI/ANSI

Sub-band 0xL/H (Duplex 196 MHz), 2xL/H (Duplex 245 and 269 MHz), 8xL/H (Duplex 161 MHz) are valid for CS 7, 14, 28, and 56 MHz.

Sub-band 1xL/H (Duplex 154, 161, and 168 MHz), 3xL/H (Duplex 154, 161, and 168 MHz), 9xL/H (Duplex 168 MHz) are valid for CS 7, 14, 28, and 56 MHz.

Sub-band 1xL/H (Duplex 150 MHz), 3xL/H (Duplex 175 MHz), and 9xL/H (Duplex 150 MHz) are valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.4.4 MINI-LINK 6363 8 GHz Band

Table 29 MINI-LINK 6363 8

Sub-band / Duplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
01L	7744.750	7793.750	49.000	7893.000	7942.250	49.250	ETSI
01H	7893.000	7942.250	49.250	7744.750	7793.750	49.000	ETSI
02L	7779.750	7828.750	49.000	7928.000	7977.250	49.250	ETSI
02H	7928.000	7977.250	49.250	7779.750	7828.750	49.000	ETSI
11L	7718.050	7837.250	119.200	8016.500	8147.975	131.475	ETSI/ANSI
11H	8016.500	8147.975	131.475	7718.050	7837.250	119.200	ETSI/ANSI
12L	7777.000	7910.775	133.775	8085.000	8222.100	137.100	ETSI/ANSI
12H	8085.000	8222.100	137.100	7777.000	7910.775	133.775	ETSI/ANSI
14L	7835.000	7970.075	135.075	8135.000	8281.400	146.400	ETSI/ANSI
14H	8135.000	8281.400	146.400	7835.000	7970.075	135.075	ETSI/ANSI
21L	8279.000	8328.000	49.000	8405.000	8450.500	45.500	ETSI
21H	8405.000	8450.500	45.500	8279.000	8328.000	49.000	ETSI
23L	8324.500	8370.000	45.500	8447.000	8492.500	45.500	ETSI
23H	8447.000	8492.500	45.500	8324.500	8370.000	45.500	ETSI
41L	8279.000	8321.000	42.000	8398.000	8440.000	42.000	ETSI
41H	8398.000	8440.000	42.000	8279.000	8321.000	42.000	ETSI
42L	8307.000	8349.000	42.000	8426.000	8475.000	49.000	ETSI
42H	8426.000	8475.000	49.000	8307.000	8349.000	42.000	ETSI
43L	8328.000	8377.000	49.000	8454.000	8496.000	42.000	ETSI
43H	8454.000	8496.000	42.000	8328.000	8377.000	49.000	ETSI
61L	7905.000	8017.000	112.000	8171.000	8327.000	156.000	ETSI
61H	8171.000	8327.000	156.000	7905.000	8017.000	112.000	ETSI
62L	7968.000	8073.000	105.000	8234.000	8339.000	105.000	ETSI
62H	8234.000	8339.000	105.000	7968.000	8073.000	105.000	ETSI
63L	7989.000	8136.000	147.000	8290.000	8411.000	121.000	ETSI
63H	8290.000	8411.000	121.000	7989.000	8136.000	147.000	ETSI
64L	8073.000	8185.000	112.000	8383.000	8495.000	112.000	ETSI
64H	8383.000	8495.000	112.000	8073.000	8185.000	112.000	ETSI
91L	8050.000	8134.000	84.000	8258.000	8342.000	84.000	ETSI
91H	8258.000	8342.000	84.000	8050.000	8134.000	84.000	ETSI
92L	8106.000	8190.000	84.000	8314.000	8398.000	84.000	ETSI
92H	8314.000	8398.000	84.000	8106.000	8190.000	84.000	ETSI
93L	8162.000	8246.000	84.000	8370.000	8454.000	84.000	ETSI
93H	8370.000	8454.000	84.000	8162.000	8246.000	84.000	ETSI

Sub-band 0xL/H (Duplex 148.25 and 148.5 MHz), 2xL/H and 4xL/H (Duplex 119 and 126 MHz) are valid for CS 7, 14, and 28/29.65 MHz.

Sub-band 1xL/H (Duplex 283.5, 310, and 311.32 MHz), 6xL/H (Duplex 266 and 310 MHz), and 9xL/H (Duplex 208 MHz) are valid for CS 7, 14, 28/29.65, and 56/59.3 MHz.

Sub-band 1xL/H (Duplex 300 and 310 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.4.5 MINI-LINK 6363 10 GHz Band

Sub-band / Duplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
21L	10500.50	10528.50	28.00	10591.50	10619.50	28.00	ETSI
21H	10591.50	10619.50	28.00	10500.50	10528.50	28.00	ETSI
22L	10528.50	10556.50	28.00	10619.50	10647.50	28.00	ETSI
22H	10619.50	10647.50	28.00	10528.50	10556.50	28.00	ETSI
23L	10556.50	10584.50	28.00	10647.50	10675.50	28.00	ETSI
23H	10647.50	10675.50	28.00	10556.50	10584.50	28.00	ETSI
41L	10308.00	10420.00	112.00	10476.00	10588.00	112.00	ETSI
41H	10476.00	10588.00	112.00	10308.00	10420.00	112.00	ETSI
71L	10000.00	10189.00	189.00	10350.00	10539.00	189.00	ETSI
71H	10350.00	10539.00	189.00	10000.00	10189.00	189.00	ETSI
72L	10125.00	10330.00	205.00	10475.00	10680.00	205.00	ETSI
72H	10475.00	10680.00	205.00	10125.00	10330.00	205.00	ETSI

Table 30 MINI-LINK 6363 10

Sub-band 2xL/H (Duplex 91 MHz) is valid for CS 7, 14, and 28 MHz.

Sub-band 4xL/H (Duplex 168 MHz) and 7xL/H (Duplex 350 MHz) are valid for CS 7, 14, 28, and 56 MHz.

3.2.4.6 MINI-LINK 6363 11 GHz Band

Table 31 MINI-LINK 6363 11

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
01L		10695.00	10877.00	182.00	11185.00	11385.00	200.00	ETSI/ANSI
01H		11185.00	11385.00	200.00	10695.00	10877.00	182.00	ETSI/ANSI
02L		10835.00	11055.00	220.00	11339.00	11575.00	236.00	ETSI/ANSI
02H		11339.00	11575.00	236.00	10835.00	11055.00	220.00	ETSI/ANSI
03L		11015.00	11200.00	185.00	11507.00	11705.00	198.00	ETSI/ANSI
03H		11507.00	11705.00	198.00	11015.00	11200.00	185.00	ETSI/ANSI

Sub-band 0xL/H (Duplex 490 and 530 MHz) is valid for CS 7, 14, 28, 40, and 56 MHz.

Sub-band 0xL/H (Duplex 490 and 500 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.4.7 MINI-LINK 6363 13 GHz Band

Table 32 MINI-LINK 6363 13; MINI-LINK 6363/2 13

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
01L		12751.00	12900.00	149.00	13017.00	13157.00	140.00	ETSI/ANSI
01H		13017.00	13157.00	140.00	12751.00	12900.00	149.00	ETSI/ANSI
03L		12863.00	12975.00	112.00	13129.00	13241.00	112.00	ETSI/ANSI
03H		13129.00	13241.00	112.00	12863.00	12975.00	112.00	ETSI/ANSI

Note: FCC part 101.147(p)(2) frequency range begins at 12700.00 MHz.

Sub-band 0xL/H (Duplex 266 MHz) is valid for 7, 14, 28, and 56 MHz.

Sub-band 0xL/H (Duplex 225 and 266 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.4.8 MINI-LINK 6363 15 GHz Band

Table 33 MINI-LINK 6363 15

Sub-band / Duplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	Telecom Standard		
	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
01L	14495.75	14774.00	278.25	15055.75	15348.00	292.25	ETSI
01H	15055.75	15348.00	292.25	14495.75	14774.00	278.25	ETSI
02L	14604.25	14739.00	134.75	14919.25	15061.00	141.75	ETSI
02H	14919.25	15061.00	141.75	14604.25	14739.00	134.75	ETSI
03L	14698.75	14828.50	129.75	15006.75	15145.00	138.25	ETSI
03H	15006.75	15145.00	138.25	14698.75	14828.50	129.75	ETSI
04L	14795.00	14933.50	138.50	15117.00	15248.50	131.50	ETSI
04H	15117.00	15248.50	131.50	14795.00	14933.50	138.50	ETSI
11L	14397.75	14660.00	262.25	14887.75	15130.00	242.25	ETSI/ANSI
11H	14887.75	15130.00	242.25	14397.75	14660.00	262.25	ETSI/ANSI
12L	14501.00	14732.00	231.00	14921.00	15195.00	274.00	ETSI/ANSI
12H	14921.00	15195.00	274.00	14501.00	14732.00	231.00	ETSI/ANSI
13L	14700.00	14928.00	228.00	15139.75	15350.00	210.25	ETSI/ANSI
13H	15139.75	15350.00	210.25	14700.00	14928.00	228.00	ETSI/ANSI

Sub-band 0xL/H (Duplex 308, 315, 322, 490, 644, 728, and 735 MHz) is valid for CS 7, 14, 28, and 56 MHz.

Sub-band 1xL/H (Duplex 420, 470, and 490 MHz) is valid for CS 7, 14, 28, and 56 MHz.

Sub-band 1xL/H (Duplex 475 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band / Duplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom Standard
	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
01L	14495.75	14774.00	278.25	15055.75	15348.00	292.25	ETSI
01H	15055.75	15348.00	292.25	14495.75	14774.00	278.25	ETSI
02L	14604.25	14739.00	134.75	14919.25	15061.00	141.75	ETSI
02H	14919.25	15061.00	141.75	14604.25	14739.00	134.75	ETSI
03L	14698.75	14828.50	129.75	15006.75	15145.00	138.25	ETSI
03H	15006.75	15145.00	138.25	14698.75	14828.50	129.75	ETSI
04L	14795.00	14933.50	138.50	15117.00	15248.50	131.50	ETSI
04H	15117.00	15248.50	131.50	14795.00	14933.50	138.50	ETSI
11L	14397.75	14660.00	262.25	14887.75	15130.00	242.25	ETSI/ANSI
11H	14887.75	15130.00	242.25	14397.75	14660.00	262.25	ETSI/ANSI
12L	14600.00	14732.00	132.00	15027.75	15195.00	167.25	ETSI/ANSI
12H	15027.75	15195.00	167.25	14600.00	14732.00	132.00	ETSI/ANSI
13L	14700.00	14928.00	228.00	15139.75	15350.00	210.25	ETSI/ANSI
13H	15139.75	15350.00	210.25	14700.00	14928.00	228.00	ETSI/ANSI

Table 34 MINI-LINK 6363/2 15

Sub-band 0xL/H (Duplex 308, 315, 322, 490, 644, 728, and 735 MHz) is valid for CS 7, 14, 28, and 56 MHz.

Sub-band 1xL/H (Duplex 420, 470, and 490 MHz) is valid for CS 7, 14, 28, and 56 MHz.

Sub-band 1xL/H (Duplex 475 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.4.9 MINI-LINK 6363 18 GHz Band

Table 35 MINI-LINK 6363 18; MINI-LINK 6363/2 18

Sub-band / D	Sub-band / Duplex (MHz)		requency inform	nation	Receiver frequency information			Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
11L		17706.50	18236.50	530.00	18600.00	19246.50	646.50	ETSI/ANSI
11H		18600.00	19246.50	646.50	17706.50	18236.50	530.00	ETSI/ANSI
13L		18002.25	18676.50	674.25	19094.75	19686.50	591.75	ETSI/ANSI
13H		19094.75	19686.50	591.75	18002.25	18676.50	674.25	ETSI/ANSI

Sub-band / Du	ıplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
21L		18580.00	18670.00	90.00	18920.00	19010.00	90.00	ETSI/ANSI
21H		18920.00	19010.00	90.00	18580.00	18670.00	90.00	ETSI/ANSI
22L		18655.00	18745.00	90.00	18995.00	19085.00	90.00	ETSI/ANSI
22H		18995.00	19085.00	90.00	18655.00	18745.00	90.00	ETSI/ANSI
23L		18730.00	18830.00	100.00	19070.00	19170.00	100.00	ETSI/ANSI
23H		19070.00	19170.00	100.00	18730.00	18830.00	100.00	ETSI/ANSI
24L		18820.00	18920.00	100.00	19160.00	19260.00	100.00	ETSI/ANSI
24H		19160.00	19260.00	100.00	18820.00	18920.00	100.00	ETSI/ANSI
31L		17700.00	18140.00	440.00	19260.00	19700.00	440.00	ETSI/ANSI
31H		19260.00	19700.00	440.00	17700.00	18140.00	440.00	ETSI/ANSI

Sub-band 1xL/H (Duplex 1008 and 1010 MHz), 2xL/H (Duplex 340 MHz), and 3xL/H (Duplex 1560 MHz) are valid for CS 7, 13.75/14, 27.5/28, 55/56, and 110/112 MHz.

Sub-band 1xL/H (Duplex 1160 MHz), 2xL/H (Duplex 340 MHz), and 3xL/H (Duplex 1560 MHz) are valid for CS 10, 20, 30, 40, 50, 60, and 80 MHz.

3.2.4.10 MINI-LINK 6363 23 GHz Band

Table 36 MINI-LINK 6363 23; MINI-LINK 6363/2 23

		Transmit	ter frequency in	formation	Receiver frequency information			Telecom
Sub-band / [Duplex (MHz)	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
01L		21200.00	21824.75	624.75	22400.00	23044.25	644.25	ETSI/ANSI
01H		22400.00	23044.25	644.25	21200.00	21824.75	624.75	ETSI/ANSI
02L		21784.00	22600.00	816.00	22996.25	23600.00	603.75	ETSI/ANSI
02H		22996.25	23600.00	603.75	21784.00	22600.00	816.00	ETSI/ANSI

Sub-band 0xL/H (Duplex 1008, 1050, 1200, and 1232 MHz) is valid for CS 7, 14, 28, 56, and 112 MHz.

Sub-band 0xL/H (Duplex 1200 and 1232 MHz) is valid for CS 10, 20, 30, 40, 50, 60, and 80 MHz.

3.2.4.11 MINI-LINK 6363 24 GHz Band

Table 37 MINI-LINK 6363 24

		Transmitt	er frequency in	formation	Receive	r frequency info	ormation	Telecom
Sub-band / [Duplex (MHz)	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
41L		24250.00	24450.00	200.00	25050.00	25250.00	200.00	ANSI
41H		25050.00	25250.00	200.00	24250.00	24450.00	200.00	ANSI

Sub-band 4xL/H (Duplex 800 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.4.12 MINI-LINK 6363 26 GHz Band

Table 38 MINI-LINK 6363 26

		Transmitt	er frequency in	formation	Receive	r frequency info	ormation	Telecom
Sub-band / [Duplex (MHz)	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
01L		24549.00	25165.00	616.00	25557.00	26173.00	616.00	ETSI
01H		25557.00	26173.00	616.00	24549.00	25165.00	616.00	ETSI
02L		24883.25	25473.00	589.75	25891.25	26481.00	589.75	ETSI
02H		25891.25	26481.00	589.75	24883.25	25473.00	589.75	ETSI

Sub-band 0xL/H (Duplex 1008 MHz) is valid for CS 7, 14, 28, 56, and 112 MHz.

3.2.4.13 MINI-LINK 6363 28 GHz Band

Table 39 MINI-LINK 6363 28

Sub-band / D	uplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
01L		27548.50	27996.50	448.00	28556.50	29004.50	448.00	ETSI
01H		28556.50	29004.50	448.00	27548.50	27996.50	448.00	ETSI
02L		27996.50	28444.50	448.00	29004.50	29452.50	448.00	ETSI
02H		29004.50	29452.50	448.00	27996.50	28444.50	448.00	ETSI
11L		27500.00	27701.00	201.00	27925.00	28150.00	225.00	ANSI
11H		27925.00	28150.00	225.00	27500.00	27701.00	201.00	ANSI
12L		27700.00	27925.00	225.00	28121.00	28350.00	229.00	ANSI
12H		28121.00	28350.00	229.00	27700.00	27925.00	225.00	ANSI

Sub-band 0xL/H (Duplex 1008 MHz) is valid for CS 7, 14, 28, 56, and 112 MHz.

Sub-band 1xL/H (Duplex 420 and 450 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.4.14 MINI-LINK 6363 32 GHz Band

Table 40 MINI-LINK 6363 32

Sub-band / D	uplex (MHz)	Transmitter fi	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
11L		31815.00	32207.00	392.00	32627.00	33019.00	392.00	ETSI/ANSI
11H		32627.00	33019.00	392.00	31815.00	32207.00	392.00	ETSI/ANSI
13L		32207.00	32599.00	392.00	33019.00	33411.00	392.00	ETSI/ANSI
13H		33019.00	33411.00	392.00	32207.00	32599.00	392.00	ETSI/ANSI

Sub-band 1xL/H (Duplex 812 MHz) is valid for CS 7, 14, 28, 56, and 112 MHz.

3.2.4.15 MINI-LINK 6363 38 GHz Band

Table 41 MINI-LINK 6363 38; MINI-LINK 6363/2 38

Sub-band / Duplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
11L	37058.00	37619.75	561.75	38318.00	38879.75	561.75	ETSI
11H	38318.00	38879.75	561.75	37058.00	37619.75	561.75	ETSI
12L	37618.00	38179.75	561.75	38878.00	39439.75	561.75	ETSI
12H	38878.00	39439.75	561.75	37618.00	38179.75	561.75	ETSI
31L	38600.00	38800.00	200.00	39300.00	39500.00	200.00	ANSI
31H	39300.00	39500.00	200.00	38600.00	38800.00	200.00	ANSI
32L	38770.00	38970.00	200.00	39470.00	39670.00	200.00	ANSI
32H	39470.00	39670.00	200.00	38770.00	38970.00	200.00	ANSI
33L	38930.00	39130.00	200.00	39630.00	39830.00	200.00	ANSI
33H	39630.00	39830.00	200.00	38930.00	39130.00	200.00	ANSI
34L	39100.00	39300.00	200.00	39800.00	40000.00	200.00	ANSI
34H	39800.00	40000.00	200.00	39100.00	39300.00	200.00	ANSI

Sub-band 1xL/H (Duplex 1260 MHz) is valid for CS 7, 14, 28, 56, and 112 MHz.

Sub-band 3xL/H (Duplex 700 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.4.16 MINI-LINK 6363 42 GHz Band

Table 42 MINI-LINK 6363 42

Sub-band / D	uplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
01L		40522.00	41054.00	532.00	42022.00	42554.00	532.00	ETSI
05H		42022.00	42554.00	532.00	40522.00	41054.00	532.00	ETSI
02L		40998.00	41502.00	504.00	42498.00	43002.00	504.00	ETSI
02H		42498.00	43002.00	504.00	40998.00	41502.00	504.00	ETSI
03L		41446.00	41964.00	518.00	42946.00	43464.00	518.00	ETSI
03H		42946.00	43464.00	518.00	41446.00	41964.00	518.00	ETSI

Sub-band 0xL/H (Duplex 1500 MHz) is valid for CS 7, 14, 28, 56, and 112 MHz.

3.2.4.17 MINI-LINK 6363 80 GHz Band

Table 43 MINI-LINK 6363 80

Sub-band / D	uplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
01L		71125.00	72250.00	1125.00	81125.00	82250.00	1125.00	ETSI/ANSI
01H		81125.00	82250.00	1125.00	71125.00	72250.00	1125.00	ETSI/ANSI
02L		72250.00	73500.00	1250.00	82250.00	83500.00	1250.00	ETSI/ANSI
02H		82250.00	83500.00	1250.00	72250.00	73500.00	1250.00	ETSI/ANSI
03L		73500.00	74750.00	1250.00	83500.00	84750.00	1250.00	ETSI/ANSI
03H		83500.00	84750.00	1250.00	73500.00	74750.00	1250.00	ETSI/ANSI
04L		74750.00	75875.00	1125.00	84750.00	85875.00	1125.00	ETSI/ANSI
04H		84750.00	85875.00	1125.00	74750.00	75875.00	1125.00	ETSI/ANSI

Sub-band 0xL/H is valid for CS 62.5 and 125 MHz (with CS 56, 60, 80, and 112 MHz physical modes).

3.2.5 MINI-LINK 6364

3.2.5.1 MINI-LINK 6364 13 GHz Band

Table 44 MINI-LINK 6364 13

Sub-band / D	uplex (MHz)	Transmitter f	requency inform	nation	Receiver frequency information			Telecom
	Tx Lo edge		Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
W01L	W01L 12751.00		12975.00	224.00	13017.00	13241.00	224.00	ETSI

Sub-band / Duplex (MHz) Transmit			requency inform	nation	Receiver frequency information			Telecom
-		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
W01H 13017.00		13241.00	224.00	12751.00	12975.00	224.00	ETSI	

Sub-band W01L/H (Duplex 266 MHz) is valid for 7, 14, 28, 56, and 112 MHz.

3.2.5.2 MINI-LINK 6364 15 GHz Band

Table 45 MINI-LINK 6364 15

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver freq	Telecom			
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard	
W01L		14397.75	14933.50	535.75	14865.00	15375.00	510.00	ETSI	
W01H		14865.00	15375.00	510.00	14397.75	14933.50	535.75	ETSI	

Sub-band W01L/H (Duplex 420, 470, 490, and 735 MHz) is valid for CS 7, 14, 28, 56, and 112 MHz.

3.2.5.3 MINI-LINK 6364 18 GHz Band

Table 46 MINI-LINK 6364 18

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver freq	Telecom			
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Stundard	
W01L		17700.00	18690.00	990.00	18590.00	19700.00	1110.00	ETSI	
W01H		18590.00	19700.00	1110.00	17700.00	18690.00	990.00	ETSI	

Sub-band W01L/H (Duplex 1010 and 1560 MHz) are valid for CS 7, 13.75/14, 27.5/28, 55/56, 110/112, and 220/224 MHz.

3.2.6 MINI-LINK 6365

3.2.6.1 MINI-LINK 6365 6 GHz Band

Table 47 MINI-LINK 6365 6

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
A41L		5925.000	6020.250	95.250	6175.000	6277.000	102.000	ETSI/ANSI
A41H		6175.000	6277.000	102.000	5925.000	6020.250	95.250	ETSI/ANSI
A42L		6005.000	6110.250	105.250	6255.000	6362.300	107.300	ETSI/ANSI

Sub-band / D	Ouplex (MHz)	Transmitter f	requency inforr	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
A42H		6255.000	6362.300	107.300	6005.000	6110.250	105.250	ETSI/ANSI
A43L		6078.500	6175.000	96.500	6330.550	6425.300	94.750	ETSI/ANSI
A43H		6330.550	6425.300	94.750	6078.500	6175.000	96.500	ETSI/ANSI
A44L		5983.000	6055.000	72.000	6241.350	6305.000	63.650	ETSI/ANSI
A44H		6241.350	6305.000	63.650	5983.000	6055.000	72.000	ETSI/ANSI
B21L		6430.000	6565.000	135.000	6770.000	6905.000	135.000	ETSI/ANSI
B21H		6770.000	6905.000	135.000	6430.000	6565.000	135.000	ETSI/ANSI
B22L		6540.000	6700.000	160.000	6880.000	7040.000	160.000	ETSI/ANSI
B22H		6880.000	7040.000	160.000	6540.000	6700.000	160.000	ETSI/ANSI
B23L		6620.000	6780.000	160.000	6960.000	7120.000	160.000	ETSI/ANSI
B23H		6960.000	7120.000	160.000	6620.000	6780.000	160.000	ETSI/ANSI
B24L		6500.000	6620.000	120.000	6840.000	6960.000	120.000	ETSI/ANSI
B24H		6840.000	6960.000	120.000	6500.000	6620.000	120.000	ETSI/ANSI
B31L		6540.000	6610.000	70.000	6700.000	6770.000	70.000	ANSI
B31H		6700.000	6770.000	70.000	6540.000	6610.000	70.000	ANSI
B32L		6597.500	6670.000	72.500	6740.000	6830.000	90.000	ANSI
B32H		6740.000	6830.000	90.000	6597.500	6670.000	72.500	ANSI
B33L		6640.000	6710.000	70.000	6800.000	6870.000	70.000	ANSI
B33H		6800.000	6870.000	70.000	6640.000	6710.000	70.000	ANSI

Sub-band A4xL/H (Duplex 240, 252.04, 260, and 266 MHz) is valid for CS 7, 14, 28/29.65, and 56/59.3 MHz.

Sub-band A4xL/H (Duplex 252.04 MHz) are valid for CS 9.88/10, 20, 29.65/30, 40, 50, and 59.3/60 MHz.

Sub-band B2xL/H (Duplex 340 MHz) is valid for CS 7, 14, 28/30, 40, 56/60 and 80 MHz.

Sub-band B2xL/H (Duplex 340 MHz) is valid for CS 10, 20, 30, 40, 50 and 60 MHz.

Sub-band B3xL/H (Duplex 160 and 170 MHz) are valid for CS 10, 20, 30, 40, 50 and 60 MHz.

3.2.6.2 MINI-LINK 6365 7/8 GHz Band

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver freq	Telecom			
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard	
A01L		7107.000	7219.000	112.000	7303.000	7415.000	112.000	ETSI	
A01H		7303.000	7415.000	112.000	7107.000	7219.000	112.000	ETSI	

Table 48 MINI-LINK 6365 7/8

Sub-band / Duplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
A03L	7163.000	7247.000	84.000	7359.000	7443.000	84.000	ETSI
A03H	7359.000	7443.000	84.000	7163.000	7247.000	84.000	ETSI
A11L	7414.000	7515.000	101.00	7575.000	7673.000	98.00	ETSI/ANSI
A11H	7575.000	7673.000	98.00	7414.000	7515.000	101.00	ETSI/ANSI
A13L	7470.000	7568.000	98.00	7635.000	7729.000	94.00	ETSI/ANSI
A13H	7635.000	7729.000	94.00	7470.000	7568.000	98.00	ETSI/ANSI
A21L	7426.500	7569.750	143.250	7671.500	7814.750	143.250	ETSI
A21H	7671.500	7814.750	143.250	7426.500	7569.750	143.250	ETSI
A23L	7510.250	7653.750	143.500	7755.250	7898.750	143.500	ETSI
A23H	7755.250	7898.750	143.500	7510.250	7653.750	143.500	ETSI
A31L	7107.000	7212.000	105.00	7275.000	7373.000	98.00	ETSI/ANSI
A31H	7275.000	7373.000	98.00	7107.000	7212.000	105.00	ETSI/ANSI
A33L	7163.000	7268.000	105.00	7331.000	7429.000	98.00	ETSI/ANSI
A33H	7331.000	7429.000	98.00	7163.000	7268.000	105.00	ETSI/ANSI
A81L	7249.50	7337.00	87.50	7410.50	7498.00	87.50	ETSI
A81H	7410.50	7498.00	87.50	7249.50	7337.00	87.50	ETSI
A83L	7305.50	7393.00	87.50	7466.50	7554.00	87.50	ETSI
A83H	7466.50	7554.00	87.50	7305.50	7393.00	87.50	ETSI
A94L	7527.000	7583.000	56.00	7695.000	7751.000	56.00	ETSI/ANSI
A94H	7695.000	7751.000	56.00	7527.000	7583.000	56.00	ETSI/ANSI
B01L	7744.750	7793.750	49.000	7893.000	7942.250	49.250	ETSI
B01H	7893.000	7942.250	49.250	7744.750	7793.750	49.000	ETSI
B02L	7779.750	7828.750	49.000	7928.000	7977.250	49.250	ETSI
B02H	7928.000	7977.250	49.250	7779.750	7828.750	49.000	ETSI
B11L	7718.050	7851.500	133.450	8016.500	8162.800	146.300	ETSI/ANSI
B11H	8016.500	8162.800	146.300	7718.050	7851.500	133.450	ETSI/ANSI
B12L	7777.000	7911.000	134.000	8073.000	8222.250	149.250	ETSI/ANSI
B12H	8073.000	8222.250	149.250	7777.000	7911.000	134.000	ETSI/ANSI
B14L	7835.000	7985.000	150.000	8128.500	8281.500	153.000	ETSI/ANSI
B14H	8128.500	8281.500	153.000	7835.000	7985.000	150.000	ETSI/ANSI
B21L	8279.000	8328.000	49.000	8398.000	8450.500	52.500	ETSI
B21H	8398.000	8450.500	52.500	8279.000	8328.000	49.000	ETSI
B23L	8324.500	8377.000	52.500	8447.000	8496.000	49.000	ETSI
B23H	8447.000	8496.000	49.000	8324.500	8377.000	52.500	ETSI
B42L	8307.000	8349.000	42.000	8426.000	8475.000	49.000	ETSI
B42H	8426.000	8475.000	49.000	8307.000	8349.000	42.000	ETSI
B61L	7905.000	8024.000	119.000	8171.000	8327.000	156.000	ETSI
B61H	8171.000	8327.000	156.000	7905.000	8024.000	119.000	ETSI
B62L	7961.000	8073.000	112.000	8234.000	8383.000	149.000	ETSI
B62H	8234.000	8383.000	149.000	7961.000	8073.000	112.000	ETSI
B63L	7989.000	8136.000	147.000	8290.000	8439.000	149.000	ETSI
B63H	8290.000	8439.000	149.000	7989.000	8136.000	147.000	ETSI
B64L	8073.000	8185.000	112.000	8383.000	8495.000	112.000	ETSI

Sub-band / Duplex (MHz)		Transmitter f	requency inform	nation	Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
B64H		8383.000	8495.000	112.000	8073.000	8185.000	112.000	ETSI
B91L		8050.000	8134.000	84.000	8258.000	8342.000	84.000	ETSI
B91H		8258.000	8342.000	84.000	8050.000	8134.000	84.000	ETSI
B92L		8106.000	8190.000	84.000	8314.000	8398.000	84.000	ETSI
B92H		8314.000	8398.000	84.000	8106.000	8190.000	84.000	ETSI
B93L		8162.000	8246.000	84.000	8370.000	8454.000	84.000	ETSI
B93H		8370.000	8454.000	84.000	8162.000	8246.000	84.000	ETSI

Sub-band A0xL/H (Duplex 196 MHz) and A2xL/H (Duplex 245 MHz) are valid for CS 7, 14, 28, 56, and 112 MHz.

Sub-band A1xL/H (Duplex 154 and 161 MHz) and A3xL/H and A8xL/H (Duplex 161 MHz) are valid for CS 7, 14, 28, and 56 MHz.

Sub-band A1xL/H (Duplex 150 MHz), A3xL/H (Duplex 175 MHz), and A9xL/H (Duplex 150 MHz) are valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band B2xL/H and B4xL/H (Duplex 119 and 126 MHz) are valid for CS 7, 14, and 28/29.65 MHz.

Sub-band B1xL/H (Duplex 283.5, 310, and 311.32 MHz) and B6xL/H (Duplex 266 and 310 MHz) are valid for CS 7, 14, 28/29.65, 56/59.3, and 112 MHz.

Sub-band B9xL/H (Duplex 208 MHz) is valid for CS 7, 14, 28/29.65, and 56/59.3 MHz.

Sub-band B1xL/H (Duplex 300 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

Sub-band B1xL/H (Duplex 310 MHz) is valid for CS 10, 20, and 40 MHz.

3.2.6.3 MINI-LINK 6365 10/11 GHz Band

Table 49 MINI-LINK 6365 10/11

Sub-band / Duplex (MHz)		Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
A21L		10500.50	10528.50	28.00	10591.50	10619.50	28.00	ETSI
A21H		10591.50	10619.50	28.00	10500.50	10528.50	28.00	ETSI
A22L		10528.50	10556.50	28.00	10619.50	10647.50	28.00	ETSI
A22H		10619.50	10647.50	28.00	10528.50	10556.50	28.00	ETSI
A23L		10556.50	10584.50	28.00	10647.50	10675.50	28.00	ETSI
A23H		10647.50	10675.50	28.00	10556.50	10584.50	28.00	ETSI
A41L		10308.00	10420.00	112.00	10476.00	10588.00	112.00	ETSI
A41H		10476.00	10588.00	112.00	10308.00	10420.00	112.00	ETSI
A71L		10000.00	10189.00	189.00	10350.00	10539.00	189.00	ETSI

Sub-band / Duplex (MHz)	Transmitter f	requency inforr	nation	Receiver freq	Telecom		
	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard
A71H	10350.00	10539.00	189.00	10000.00	10189.00	189.00	ETSI
A72L	10125.00	10330.00	205.00	10475.00	10680.00	205.00	ETSI
A72H	10475.00	10680.00	205.00	10125.00	10330.00	205.00	ETSI
B01L	10695.00	10905.00	210.00	11185.00	11425.00	240.00	ETSI/ANSI
B01H	11185.00	11425.00	240.00	10695.00	10905.00	210.00	ETSI/ANSI
B02L	10793.00	11075.00	282.00	11311.00	11605.00	294.00	ETSI/ANSI
B02H	11311.00	11605.00	294.00	10793.00	11075.00	282.00	ETSI/ANSI
B03L	10989.00	11200.00	211.00	11479.00	11705.00	226.00	ETSI/ANSI
B03H	11479.00	11705.00	226.00	10989.00	11200.00	211.00	ETSI/ANSI

Sub-band A2xL/H (Duplex 91 MHz) is valid for CS 7, 14, and 28 MHz.

Sub-band A4xL/H (Duplex 168 MHz) and 7xL/H (Duplex 350 MHz) are valid for CS 7, 14, 28, 56, and 112 MHz.

Sub-band B0xL/H (Duplex 490 and 530 MHz) is valid for CS 7, 14, 28, 40, 56, 80, and 112 MHz.

Sub-band B0xL/H (Duplex 520 MHz) is valid for CS 60 MHz.

Sub-band B0xL/H (Duplex 490 and 500 MHz) is valid for CS 10, 20, 30, 40, 60, and 80 MHz.

3.2.6.4 MINI-LINK 6365 13 GHz Band

Table 50	MINI-LINK	636513
	I ITLAT CTIALS	. 0000 10

Sub-band / Duplex (MHz)		Transmitter frequency information			Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
A01L		12751.00	12919.00	168.00	13017.00	13185.00	168.00	ETSI/ANSI
A01H		13017.00	13185.00	168.00	12751.00	12919.00	168.00	ETSI/ANSI
A03L		12863.00	12975.00	112.00	13125.00	13241.00	116.00	ETSI/ANSI
A03H		13125.00	13241.00	116.00	12863.00	12975.00	112.00	ETSI/ANSI

Note: FCC part 101.147(p)(2) frequency range begins at 12700.00 MHz.

Sub-band A0xL/H (Duplex 266 MHz) is valid for 7, 14, 28, 56, and 112 MHz.

Sub-band A0xL/H (Duplex 225 and 266 MHz) is valid for CS 10, 20, 30, 40, and 50 MHz.

3.2.6.5 MINI-LINK 6365 15 GHz Band

Table 51 MINI-LINK 6365 15

Sub-band / Dupl	lex (MHz)	Transmitter f	requency inforn	nation	Receiver freq	Telecom		
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
A01L		14495.75	14781.00	285.25	15055.75	15348.00	292.25	ETSI
A01H		15055.75	15348.00	292.25	14495.75	14781.00	285.25	ETSI
A11L		14397.75	14669.00	271.25	14865.00	15135.00	270.00	ETSI/ANSI
A11H		14865.00	15135.00	270.00	14397.75	14669.00	271.25	ETSI/ANSI
A12L		14501.00	14739.00	238.00	14919.25	15229.00	309.75	ETSI/ANSI
A12H		14919.25	15229.00	309.75	14501.00	14739.00	238.00	ETSI/ANSI
A13L		14700.00	14933.50	233.50	15117.00	15350.00	233.00	ETSI/ANSI
A13H		15117.00	15350.00	233.00	14700.00	14933.50	233.50	ETSI/ANSI
A21L		14698.75	14828.50	129.75	15006.75	15145.00	138.25	ETSI
A21H		15006.75	15145.00	138.25	14698.75	14828.50	129.75	ETSI

Sub-band A0xL/H and A1xL/H (Duplex 420, 490, and 728 MHz) is valid for CS 7, 14, 28, 56, and 112 MHz.

Sub-band A1xL/H (Duplex 475 MHz) is valid for CS 10, 20, 30, 40, 50, 60, and 80 MHz.

3.2.6.6 MINI-LINK 6365 18 GHz Band

Table 52 MINI-LINK 6365 18

Sub-band / Duplex (MHz)	Transmitter f	requency inform	nation	Receiver freq	uency informat	ion	Telecom	
	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara	
A11L	17700.00	18236.50	536.50	18590.00	19260.00	670.00	ETSI/ANSI	
A11H	18590.00	19260.00	670.00	17700.00	18236.50	536.50	ETSI/ANSI	
A13L	18002.25	18690.00	687.75	19081.25	19700.00	618.75	ETSI/ANSI	
A13H	19081.25	19700.00	618.75	18002.25	18690.00	687.75	ETSI/ANSI	
A21L	18580.00	18745.00	165.00	18920.00	19085.00	165.00	ETSI/ANSI	
A21H	18920.00	19085.00	165.00	18580.00	18745.00	165.00	ETSI/ANSI	
A23L	18730.00	18920.00	190.00	19070.00	19260.00	190.00	ETSI/ANSI	
A23H	19070.00	19260.00	190.00	18730.00	18920.00	190.00	ETSI/ANSI	
A31L	17700.00	18140.00	440.00	19260.00	19700.00	440.00	ETSI/ANSI	
A31H	19260.00	19700.00	440.00	17700.00	18140.00	440.00	ETSI/ANSI	

Sub-band A1xL/H (Duplex 1010 MHz) and A3xL/H (Duplex 1560 MHz) are valid for CS 7, 13.75/14, 27.5/28, 55/56, 110/112, and 220/224 MHz.

Sub-band A1xL/H (Duplex 1160 MHz) and A3xL/H (Duplex 1560MHz) are valid for CS 10, 20, 30, 40, 50, 60, and 80 MHz.

3.2.6.7 MINI-LINK 6365 23 GHz Band

Table 53 MINI-LINK 6365 23

		Transmitt	er frequency in	formation	Receive	Telecom			
Sub-band / [Duplex (MHz)	Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standard	
A01L		21200.00	21896.00	696.00	22400.00	23128.00	728.00	ETSI/ANSI	
A01H		22400.00	23128.00	728.00	21200.00	21896.00	696.00	ETSI/ANSI	
A02L		21784.00	22600.00	816.00	22996.00	23600.00	604.00	ETSI/ANSI	
A02H		22996.00	23600.00	604.00	21784.00	22600.00	816.00	ETSI/ANSI	

Sub-band A0xL/H (Duplex 1008, 1050, 1200, and 1232 MHz) is valid for CS 7, 14, 28, 56, 112, and 224 MHz.

Sub-band A0xL/H (Duplex 1200 and 1232 MHz) is valid for CS 10, 20, 30, 40, 50, 60, and 80 MHz.

3.2.6.8 MINI-LINK 6365 32 GHz Band

Table 54 MINI-LINK 6365 32

Sub-band / D	uplex (MHz)	Transmitter f	equency inform	nation	Receiver freq	ion	Telecom	
		Tx Lower edge	Tx Upper edge	Tx Bw (MHz)	Rx Lower edge	Rx Upper edge	Rx Bw (MHz)	Standara
A11L		31815.00	32319.00	504.00	32627.00	33131.00	504.00	ETSI/ANSI
A11H		32627.00	33131.00	504.00	31815.00	32319.00	504.00	ETSI/ANSI
A13L		32095.00	32599.00	504.00	32907.00	33411.00	504.00	ETSI/ANSI
A13H		32907.00	33411.00	504.00	32095.00	32599.00	504.00	ETSI/ANSI

Sub-band A1xL/H (Duplex 812 MHz) is valid for CS 7, 14, 28, 56, 112, and 224 MHz

3.3 Carrier Aggregation

Carrier Aggregation (CA) is supported by the following MMUs and radio units:

- MMU 1002
- MMU 1004
- MINI-LINK 6364
- MINI-LINK 6365

Detailed information about physical modes supported with CA can be found in Supported Physical Modes on page 138.

3.3.1 Maximum frequency separation

The figures below shows the maximum allowed frequency separation, in terms of empty channels, for the supported channels in Carrier Aggregation. Both channels must be within the radio sub-band.

IS	2x 28 MHz	≤2 empty gaps	
Ш	2x 56 MHz	Adjacent	

Figure 1 Maximum frequency separation for MINI-LINK 6364 13, 15

	2x 28 MHz	≤4 empty gaps	
ETSI	2x 56 MHz	≤1 empty gaps	
2	2x 112 MHz	Adjacent	

Figure 2 Maximum frequency separation for MINI-LINK 6364 18

	2x 28 MHz	≤4 empty gaps	
	2x 40 MHz	≤2 empty gaps	
ISI	2x 56 MHz	≤1 empty gaps	
"	2x 80 MHz	Adjacent	
	2x 112 MHz	Adjacent	
	2x 30 MHz	≤4 empty gaps	
	2x 40 MHz	≤2 empty gaps	
ISNN	2x 50 MHz	≤2 empty gaps	
A	2x 60 MHz	≤1 empty gaps	
	2x 80 MHz	Adjacent	

Figure 3 Maximum frequency separation for MINI-LINK 6365

3.3.2 Configuration limitations

- The channel separation (CS) must be the same on the two channels
- The duplex configuration must be the same on the two channels

 Both sides of the link (near end and far end) must be configured with Carrier Aggregation

3.4 Transmitter Performance

See Supported Physical Modes on page 138 for supported physical modes and traffic capacity.

3.4.1 Frequency Tolerance

The transmitter carrier frequency does not deviate more than ± 10 ppm from nominal frequency. During the first year of operation the deviation is less than ± 3 ppm.

3.4.2 Output Power Tolerance

Tuble 55 Output I ower Toleful		
Radio Type	Output Power Tolerance (dB)	Output Power Tolerance at Pmax (dB)
RAU2 X	±2	-1/+2
MINI-LINK 6363 6-42	±2	-1/+2
MINI-LINK 6363 80	±2.5	±2
MINI-LINK 6363/2	±2	±2
MINI-LINK 6364	±2	-1/+2
MINI-LINK 6365	±2	-1/+2
MINI-LINK 6364 CA mode	±2	±2
MINI-LINK 6365 CA mode	±2	±2

Table 55 Output Power Tolerance

3.4.3 Output Power for QAM, Static Modulation

MINI-LINK 6363, MINI-LINK 6364, and MINI-LINK 6365 Output power including high power license (+4 dB).

Note that 8192QAM is not supported as static modulation.

The output power is set in 1 dB steps in ranges:

Radio Type	Pmax (dBm) vers	us Modulo	ation (QAI	M)						P _{min}
	4	16	32	64	128	256	512	1024	2048	4096	(dBm)
RAU2 X 5 HP	30	30	30	30	30	29	29	28	27 ⁽¹⁾	26 <mark>(1)</mark>	-10
RAU2 X 6L	29	27	27	26	26	25	25	24	23 <mark>(1)</mark>	22 ⁽¹⁾	-10
RAU2 X 6L HP	30	30	30	30	30	29	29	28	27 <mark>(1)</mark>	26 ⁽¹⁾	-10
RAU2 X 6U	29	27	27	26	26	25	25	24	23 <mark>(1)</mark>	22 ⁽¹⁾	-10
RAU2 X 6U HP	30	30	30	30	30	29	29	28	27 ⁽¹⁾	26 ⁽¹⁾	-10
RAU2 X 7	29	27	27	26	26	25	25	24	23(1)	22 ⁽¹⁾	-10
RAU2 X 7 HP	30	30	30	30	30	29	29	28	27 <mark>(1)</mark>	26 ⁽¹⁾	-10
RAU2 X 8	29	27	27	26	26	25	25	24	23 <mark>(1)</mark>	22 ⁽¹⁾	-10
RAU2 X 8 HP	30	30	30	30	30	29	29	28	27 <mark>(1)</mark>	26 ⁽¹⁾	-10
RAU2 X 10 ⁽²⁾	28	26	26	25	25	24	24	23	22 <mark>(1)</mark>	21 ⁽¹⁾	-10
RAU2 X 10 HP ⁽²⁾	30	30	30	29	29	28	28	27	26 ⁽¹⁾	25 <mark>(1)</mark>	-10
RAU2 X 11	28	26	26	25	25	24	24	23	22 <mark>(1)</mark>	21 ⁽¹⁾	-10
RAU2 X 11 HP	30	30	30	29	29	28	28	27	26 ⁽¹⁾	25 <mark>(1)</mark>	-10
RAU2 X 13	24	22	22	21	21	20	20	19	18 <mark>(1)</mark>	17 ⁽¹⁾	-10
RAU2 X 13 HP	28	26	26	25	25	24	24	23	22 <mark>(1)</mark>	21 ⁽¹⁾	-10
RAU2 X 15	24	22	22	21	21	20	20	19	18 <mark>(1)</mark>	17 ⁽¹⁾	-10
RAU2 X 15 HP	28	26	26	25	25	24	24	23	22 <mark>(1)</mark>	21 ⁽¹⁾	-10
RAU2 X 18	21	19	19	18	18	17	17	16	15 <mark>(1)</mark>	14 ⁽¹⁾	-10
RAU2 X 18 HP	26	24	24	23	23	22	22	21	20(1)	19 ⁽¹⁾	-10
RAU2 X 23	21	19	19	18	18	17	17	16	15 ⁽¹⁾	14 ⁽¹⁾	-10
RAU2 X 23 HP	25	23	23	22	22	21	21	20	19 <mark>(1)</mark>	18 ⁽¹⁾	-10
RAU2 X 24	23	21	21	20	20	19	19	18	17 <mark>(1)</mark>	16 ⁽¹⁾	-10
RAU2 X 26	23	21	21	20	20	19	19	18	17 <mark>(1)</mark>	16 ⁽¹⁾	-10
RAU2 X 28	22	20	20	19	19	18	18	17	16 ⁽¹⁾	15 ⁽¹⁾	-10
RAU2 X 32	20	18	18	17	17	16	16	15	14 ⁽¹⁾	13 ⁽¹⁾	-10
RAU2 X 38	20	18	18	17	17	16	16	15	14 ⁽¹⁾	13 ⁽¹⁾	-10
RAU2 X 42	18	16	16	15	15	14	14	13	12 ⁽¹⁾	11 ⁽¹⁾	-10

Table 56 RAU2 X Output Power for QAM

(1) With RAU2 X R-state \geq R6A

(2) With duplex \leq 168 MHz; Pmax 1 dB lower

Table 57 MINI-LINK 6363 and MINI-LINK 6363/2 Output Power for QAM

Radio Type	Pmax (a	'max (dBm) versus Modulation (QAM)								P _{min}	
	4	16	32	64	128	256	512	1024	2048	4096	(dBm)
MINI-LINK 6363 6L	30	30	30	30	30	29	29	28	28	27	-10

Radio Type	Pmax (dBm) vers	us Modulo	tion (QAN	4)						P _{min}
	4	16	32	64	128	256	512	1024	2048	4096	(dBm)
MINI-LINK 6363 6U	30	30	30	30	30	29	29	28	28	27	-10
MINI-LINK 6363 7	30	30	30	30	30	29	29	28	28	27	-10
MINI-LINK 6363 8	30	30	30	29	29	28	28	27	27	27	-10
MINI-LINK 6363 10	30	30	30	29	29	28	28	27	27	26	-10
MINI-LINK 6363 11	30	30	30	29	29	28	28	27	27	26	-10
MINI-LINK 6363 13	27	26	26	25	25	24	24	23	23	22	-10
MINI-LINK 6363 15	27	26	26	25	25	24	24	23	23	22	-10
MINI-LINK 6363 18	25	24	24	23	23	22	22	21	21	20	-10
MINI-LINK 6363 23	24	23	23	22	22	21	21	20	20	19	-10
MINI-LINK 6363 24	25	24	24	23	23	22	22	21	21	20	-10
MINI-LINK 6363 26	25	24	24	23	23	22	22	21	21	20	-10
MINI-LINK 6363 28	23	22	22	21	21	20	20	19	19	18	-10
MINI-LINK 6363 32	22	21	21	20	20	19	19	18	18	17	-10
MINI-LINK 6363 38	22	21	21	20	20	19	19	18	18	17	-10
MINI-LINK 6363 42	18	17	17	16	16	15	15	14	14	13	-10
MINI-LINK 6363 80	17	16	16	15	15	14	14	13	13(1)	12 ⁽¹⁾	-10
MINI-LINK 6363/2 13	19	18	18	17	17	16	16	15	15	14	-10
MINI-LINK 6363/2 15	19	18	18	17	17	16	16	15	15	14	-10
MINI-LINK 6363/2 18	17	16	16	15	15	14	14	13	13	12	-10
MINI-LINK 6363/2 23	16	15	15	14	14	13	13	12	12	11	-10
MINI-LINK 6363/2 38	14	13	13	12	12	11	11	10	10	9	-10

(1) Modulations \geq 2048QAM is not supported. Specified values shall be used as Pmax reference in Admod.

Table 58 MINI-LINK 6364 Output Power for QAM

Radio Type	Pmax ((dBm) ve	rsus Mod	ulation (C	(AM)						P _{min}
	4	16	32	64	128	256	512	1024	2048	4096	(dBm)
MINI-LINK 6364 13	27	26	26	25	25	24	24	23	23	22	-10
MINI-LINK 6364 15	27	26	26	25	25	24	24	23	23	22	-10
MINI-LINK 6364 18 ⁽¹⁾	26	25	25	24	24	23	23	22	22	21	-10
MINI-LINK 6364 13 CA mode	21	20	20	19	19	18	18	17	17	16	+3
MINI-LINK 6364 15 CA mode	21	20	20	19	19	18	18	17	17	16	+3
MINI-LINK 6364 18 CA mode	19	18	18	18	18	17	17	16	16	15	0

(1) For CS 220 MHz, maximum output power is 3 dB lower and P_{min} is 3 dBm.

Table 59	MINI-LINK 6365 Output Power for QAM
----------	-------------------------------------

Radio Type	Pmax (Pmax (dBm) versus Modulation (QAM)						P _{min}			
	4	16	32	64	128	256	512	1024	2048	4096	(dBm)
MINI-LINK 6365 6	30	30	30	30	30	29	29	28	28	27	-10
MINI-LINK 6365 7/8	30	30	30	30	30	29	29	28	28	27	-10

Radio Type	Pmax	Pmax (dBm) versus Modulation (QAM)								P _{min}	
	4	16	32	64	128	256	512	1024	2048	4096	(dBm)
MINI-LINK 6365 10/11	30	30	30	29	29	28	28	27	27	26	-10
MINI-LINK 6365 13	27	26	26	25	25	24	24	23	23	22	-10
MINI-LINK 6365 15	27	26	26	25	25	24	24	23	23	22	-10
MINI-LINK 6365 18 ⁽¹⁾	26	25	25	24	24	23	23	22	22	21	-10
MINI-LINK 6365 23 ⁽²⁾	25	24	24	23	23	22	22	21	21	20	-10
MINI-LINK 6365 32 ⁽²⁾	22	21	21	20	20	19	19	18	18	17	-10
MINI-LINK 6365 6 CA mode	21	21	21	21	21	21	21	21	21	21	+3
MINI-LINK 6365 7/8 CA mode	21	21	21	21	21	21	21	21	21	21	+3
MINI-LINK 6365 10/11 CA mode	21	21	21	21	21	21	21	21	21	20	+3
MINI-LINK 6365 13 CA mode	21	20	20	19	19	18	18	17	17	16	+3
MINI-LINK 6365 15 CA mode	21	20	20	19	19	18	18	17	17	16	+3
MINI-LINK 6365 18 CA mode	19	19	19	18	18	17	17	16	16	15	0
MINI-LINK 6365 23 CA mode	19	18	18	17	17	16	16	15	15	14	-5
MINI-LINK 6365 32 CA mode	16	15	15	14	14	13	13	12	12	11	-5

(1) For CS 220 MHz, maximum output power is 3 dB lower and ${\rm P}_{\rm min}$ is 3 dBm.

(2) For CS 224 MHz, maximum output power is 3 dB lower and ${\rm P}_{min}$ is -2 dBm.

3.4.4 Output Power for QAM, Adaptive Modulation

3.4.4.1 ETSI

Maximum output power values (Pmax) are according to Output Power for QAM, Static Modulation on page 116 with the following limitations:

MINI-LINK 6363, MINI-LINK 6364, MINI-LINK 6365, MINI-LINK 6363/2, and RAU2 X Max output power is relative to 4096QAM (dB).

Table 60 C	Dutput Power fo	r QAM, Adaptive	Modulation -	ETSI
------------	-----------------	-----------------	--------------	------

RSEC	Pmax (dBm) re	P _{min}		
Modulation	2	4L	5LB/5HB	(dBm)
		4H	6LB/6HB	
			7B	
4QAM	5	4	3	-10
16QAM	4	4	3	-10
32QAM	4	4	3	-10
64QAM	3	3	3	-10
128QAM	3	3	3	-10
256QAM	2	2	2	-10

 \geq

RSEC	Pmax (dBm) re	P _{min}		
Modulation	2	4L	5LB/5HB	(dBm)
		4H	6LB/6HB	
			7B	
512QAM	1	1	1	-10
1024QAM	1	1	1	-10
2048QAM	0	0	0	-10
4096QAM	0/-1(1)	0/-1(1)	0/-1(1)	-10
8192QAM	-1	-1	-1	0

 Offset is -1 dB for MINI-LINK 6363 R-state ≥ R4A, MINI-LINK 6364, and MINI-LINK 6365 for frame ID 358 and 359. That is frame IDs with support for 8192QAM.

3.4.4.2 ANSI

Maximum output power values (Pmax) are according to Output Power for QAM, Static Modulation on page 116 with the following limitations:

MINI-LINK 6363, MINI-LINK 6365, MINI-LINK 6363/2 and RAU2 X/Xu Max output power is relative to 4096QAM (dB).

Modulation	Pmax (dBm) relative 4096QAM	Pmin (dBm)
4QAM	5	-10
16QAM	4	-10
32QAM	4	-10
64QAM	3	-10
128QAM	3	-10
256QAM	2	-10
512QAM	1	-10
1024QAM	1	-10
2048QAM	0	-10
4096QAM	0/-1(1)	-10
8192QAM	-1	0

Table 61 Output Power for QAM, Adaptive Modulation - ANSI

 Offset is -1 dB for MINI-LINK 6363 R-state ≥ R4A, MINI-LINK 6364, and MINI-LINK 6365 for frame ID 364 and 365. That is frame IDs with support for 8192QAM.



3.5 Emission Designator

3.5.1 Adaptive Modulation

Table 62 Adaptive Modulation - ETSI

Channel separation (MHz)	Emission Designator
7	6M30D7W
14/13.75	12M8D7W
28/27.5	25M6D7W
40	36M0D7W
56/55/62.5	51M5D7W
80	74M0D7W
112/110/125	103MD7W
224/220	212MD7W

Table 63 Adaptive Modulation - ANSI

Channel separation (MHz)	Emission Designator
10	8M75D7W
20	18M1D7W
28	25M6D7W
30	27M4D7W
40	36M8D7W
50	46M1D7W
56	51M5D7W
60	55M4D7W
80	74MD7W
112	103MD7W

3.5.2 Transmitter off

RF output power in Tx off mode: -50 dBm.

3.5.3 Transmitter Spurious Levels for QAM

Transmitter spurious levels are below:

- -50 dBm in the band 30 MHz-21.2 GHz
 - -50 dBm in any 100 kHz band from 30 MHz to 1 GHz
 - -50 dBm in any 1 MHz band from 1 GHz to 21.2 GHz
- -30 dBm in the band 21.2 110 GHz
 - -30 dBm in any 100 kHz band for channel separation ${\leq}7$ MHz, and in the range from ${\pm}$ 2.5 times of channel separation to ${\pm}$ 56 MHz
 - -30 dBm in any 1 MHz band in all other cases

The above excludes a frequency band \pm 2.5 times the channel separation from the nominal transmitter frequency.

3.6 Receiver Performance

See Supported Physical Modes on page 138 for supported physical modes, traffic capacity, and XPIC or MIMO support.

3.6.1 Threshold Definition

Guarantee	Guaranteed value for all temperatures.		
Planning	Ericsson recommendation for path planning.		

Ericsson recommendation for path planning.

Planning threshold = 1 dB below guarantee threshold.

Planning threshold is used by path planning programs.

- Typical Average value in room temperature.
- 3.6.2 Detection Performance for QAM, Adaptive Modulation

3.6.2.1 Detection Performance, ETSI

The receiver complies with detection performance listed below.

Data for standard Frame ID's (256, 257, 258, 259, 260, 261, 303, 1256, 1257, 1258, 1259, 1260, 1261, and 1303) can be found in revision BT of this document.

Table 64 Detection Performance, ETSI

The detection performance is applicable to the following radio types:

____ RAU2 X 5, 6L, 6U, 7, 8, 10 (Dupl. > 168MHz), 11, 13, 15

- ____ MINI-LINK 6363 6L, 6U, 7, 8, 10 (Dupl. > 168MHz), 11, 13, 15
- ____ MINI-LINK 6363/2 13, 15
- ____ MINI-LINK 6364 13, 15
- ____ MINI-LINK 6365 6, 7/8, 10/11, 13, 15

CS (Frame ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)
		Тур.
7	4QAM S	-95.5
1356)	4QAM	-94
	16QAM S	-89
	16QAM	-87.5
	32QAM	-84
	64QAM	-81.5
	128QAM	-78.5
	256QAM	-75.5
-	512QAM	-71
	1024QAM	-68
13.75	4QAM S	-92.5
14 (357	4QAM	-91
1357)	16QAM S	-86
-	16 QAM	-84.5
-	32QAM	-81
-	64QAM	-78.5
-	128QAM	-75.5
-	256QAM	-72.5
-	512QAM	-68
	1024QAM	-65
-	2048QAM	-63
-	2048QAM L	-61.5
27.5	4QAM S	-89.5
28	4QAM	-88
1358)	16QAM S	-83
-	16 QAM	-81.5
	32QAM	-78
	64QAM	-75.5
-	128QAM	-72.5
	256QAM	-69.5
	512QAM	-65
	1024QAM	-62
	2048QAM	-60
-	2048QAM L	-58.5
	4096QAM	-56.5
	4096QAM L	-55

CS (Frame ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)
		Тур.
27.5	4QAM S	-89
28 (2358)	4QAM	-87.5
	16QAM S	-82.5
	16 QAM	-81
	32QAM	-77.5
	64QAM	-75
	128QAM	-72
	256QAM	-69
	512QAM	-66
	512QAM L	-64.5
40	4QAM S	-88
(359)	4QAM	-86.5
	16QAM S	-81.5
	16 QAM	-80
	32QAM	-76.5
	64QAM	-74
	128QAM	-71
	256QAM	-68
	512QAM	-63.5
	1024QAM	-60.5
	2048QAM	-58.5
	2048QAM L	-57
	4096QAM	-55
	4096QAM L	-53.5
	8192 QAM	-50.5
55	4QAM S	-86.5
62.5	4QAM	-85
(360	16QAM S	-80
1300)	16 QAM	-78.5
	32QAM	-75
	64QAM	-72.5
	128QAM	-69.5
	256QAM	-66.5
	512QAM	-62
	1024QAM	-59
	2048 QAM	-57
	2048 QAM L	-55.5
	4096QAM	-53.5
	4096QAM L	-52
CS (Eramo ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)
------------------	------------	--------------------------------------
(Frame ID)		Тур.
55	4QAM S	-86
56 (2360)	4QAM	-84.5
	16QAM S	-79.5
	16 QAM	-78
	32QAM	-74.5
	64QAM	-72
	128QAM	-69
	256QAM	-66
	512QAM	-63
	512QAM L	-61.5
	1024QAM	-60
	1024QAM L	-58.5
80	4QAM S	-85
(403	4QAM	-83.5
1+00)	16QAM S	-78.5
	16 QAM	-77
	32QAM	-73.5
	64QAM	-71
	128QAM	-68
	256QAM	-65
	512QAM	-60.5
	1024QAM	-57.5
	2048QAM	-55.5
	2048QAM L	-54
	4096QAM	-52
	4096QAM L	-49.5
110	4QAM S	-83.5
112	4QAM	-82
(361	16QAM S	-77
1361)	16 QAM	-75.5
	32QAM	-72
	64QAM	-69.5
	128QAM	-66.5
	256QAM	-63.5
	512QAM	-59
	1024QAM	-56
	2048QAM	-54
	2048QAM L	-52.5
	4096QAM	-50.5

CS (Examp ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)
		Тур.
$220^{(1)}$	4QAM S	-80.5
(404)	4QAM	-79
	16QAM S	-74
	16 QAM	-72.5
	32QAM	-69
	64QAM	-66.5
	128QAM	-63.5
	256QAM	-60.5
	512QAM	-56
	1024QAM	-53

(1) Carrier Aggregation with 2x110 (Frame ID 404) is used in CS 220 but thresholds are related to total input power.

(2) Carrier Aggregation with 2x112 (Frame ID 404) is used in CS 224 but thresholds are related to total input power.

Table 65 Detection Performance, ETSI

The detection performance is applicable to the following radio types:

- ____ RAU2 X $10^{(1)}$ (Dupl. ≤ 168 MHz), 18, 23, 26
- MINI-LINK 6363 10⁽¹⁾ (Dupl. ≤ 168 MHz), 18, 23, 26
- ____ MINI-LINK 6363/2 18, 23
- ____ MINI-LINK 6364 18
- ____ MINI-LINK 6365 18, 23

CS (Errime ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)
		Тур.
All	All	Add 1 dB to figures in Table 64 above.

(1) Max CS = 28 MHz

Table 66 DetectionPerformance, ETSI

The detection performance is applicable to the following radio types:

- ____ RAU2 X 28, 32, 38
- ____ MINI-LINK 6363 28, 32, 38
- ____ MINI-LINK 6363/2 38

____ MINI-LINK 6365 32

CS (Errme ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)
		Тур.
All	All	Add 2 dB to figures in Table 64 above.

Table 67 Detection Performance, ETSI

The detection performance is applicable to the following radio types:

____ RAU2 X 42

____ MINI-LINK 6363 42

CS (Frame ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)
(nume ib)		Тур.
All	All	Add 3 dB to figures in Table 64 above.

Table 68 Detection Performance, ETSI

The detection performance is applicable to the following radio type:

____ MINI-LINK 6363 80

CS Modulation	Modulation	BER 10 ⁻⁶ threshold (dBm)
(Fighter 10)		Тур.
55 56 62.5 (360 1360)	All	Add 7 dB to figures in Table 64 above
110 112 125 (361 1361)	All	Add 7 dB to figures in Table 64 above

3.6.2.2 Detection Performance, ANSI

The receiver complies with detection performance listed below.

Data for standard Frame ID's (262, 263, 264, 265, 266, 267, 268, 1262, 1263, 1264, 1265, 1266, 1267, and 1268) can be found in revision BT of this document.

Table 69 Detection Performance, ANSI

The detection performance is applicable to the following radio types:

____ RAU2 X 5, 6L, 6U, 7, 8, 10 (Dupl. > 168MHz), 11, 13, 15

- ____ MINI-LINK 6363 6L, 6U, 7, 8, 10 (Dupl. > 168MHz), 11, 13, 15
- ____ MINI-LINK 6363/2 13, 15
- ____ MINI-LINK 6365 6, 7/8, 10/11, 13, 15

CS (Frame ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)
		Тур.
10	4QAM S	-94
(362)	4QAM	-92.5
	16QAM S	-87.5
	16QAM	-86
	32QAM	-82.5
	64QAM	-80
	128QAM	-77
	256QAM	-74
	512QAM	-69.5
	1024QAM	-66.5
20	4QAM S	-91
(363)	4QAM	-89.5
	16QAM S	-84.5
	16 QAM	-83
	32QAM	-79.5
	64QAM	-77
	128QAM	-74
	256QAM	-71
	512QAM	-66.5
	1024QAM	-63.5
	2048QAM	-61.5
	2048QAM L	-60
28	4QAM S	-89.5
1371)	4QAM	-88
	16QAM S	-83
	16 QAM	-81.5
	32QAM	-78
	64QAM	-75.5
	128QAM	-72.5
	256QAM	-69.5
	512QAM	-65
	1024QAM	-62
	2048QAM	-60
	2048QAM L	-58.5
	4096QAM	-56.5
	4096QAM L	-55

CS (Frame ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)
(Fighter 1D)		Тур.
30	4QAM S	-89.5
(364)	4QAM	-88
	16QAM S	-83
	16 QAM	-81.5
	32QAM	-78
	64QAM	-75.5
	128QAM	-72.5
	256QAM	-69.5
	512QAM	-65
	1024QAM	-62
	2048QAM	-60
	2048QAM L	-58.5
	4096QAM	-56.5
	4096QAM L	-55
	8192QAM	-52
30	4QAM S	-89
(2364)	4QAM	-87.5
	16QAM S	-82.5
	16 QAM	-81
	32QAM	-77.5
	64QAM	-75
	128QAM	-72
	256QAM	-69
	512QAM	-66
	512QAM L	-64.5
40	4QAM S	-88
(365)	4QAM	-86.5
	16QAM S	-81.5
	16 QAM	-80
	32QAM	-76.5
	64QAM	-74
	128QAM	-71
	256QAM	-68
	512QAM	-63.5
	1024QAM	-60.5
	2048QAM	-58.5
	2048QAM L	-57
	4096QAM	-55
	4096QAM L	-53.5
	8192QAM	-50.5

CS (Frame ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)
		Тур.
40	4QAM S	-87.5
(2365)	4QAM	-86
	16QAM S	-81
	16 QAM	-79.5
	32QAM	-76
	64QAM	-73.5
	128QAM	-70.5
	256QAM	-67.5
	512QAM	-64.5
	512QAM L	-63
50	4QAM S	-87
(366)	4QAM	-85.5
	16QAM S	-80.5
	16 QAM	-79
	32QAM	-75.5
	64QAM	-73
	128QAM	-70
	256QAM	-67
	512QAM	-62.5
	1024QAM	-59.5
	2048QAM	-57.5
	2048QAM L	-56
	4096QAM	-54
	4096QAM L	-52.5
56 (373	4QAM S	-86.5
1373)	4QAM	-85
	16QAM S	-80
	16 QAM	-78.5
	32QAM	-75
	64QAM	-72.5
	128QAM	-69.5
	256QAM	-66.5
	512QAM	-62
	1024QAM	-59
	2048 QAM	-57
	2048 QAM L	-55.5
	4096QAM	-53.5
	4096QAM L	-52

(Huine La) Typ. 60 (367) 1367) 4QAM S -86 14QAM -84.5 16QAM S -77.5 16 QAM -77.8 32QAM -74.5 32QAM -74.5 32QAM -74.5 32QAM -72 128QAM -60 128QAM -61.5 128QAM -66.5 128QAM -66.5 128QAM -65.5 2848QAM -55.5 2848QAM -55.5 4096QAM -55.5 16QAM S -77 1368) 40AM S -65.5 16QAM S -78.5 16QAM S -77.5 16QAM S -77.5 16QAM -77.5 128QAAM -68.5 120QAM -68.5 120AM -68.5 120AM -68.5 120AM -68.5 120AAM -55.5 120AAM -55.5	CS (Eramo ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)
60 (357) 40AM 5 -86 1357) 16 0AM -84,5 16 0AM 5 -79.5 - 16 0AM 7 -78 - 320AM - -74,5 - 640AM - -72 - 1280AM - -66 - 12120AM - -66.5 - 1220AM - -55 - 2840AM - -56.5 - 2840AM - -55 - 1569) 40AM 5 - 160AM 5 - - 160AM 5 - - 120AM - - <t< th=""><th>(Fighte 10)</th><th></th><th>Тур.</th></t<>	(Fighte 10)		Тур.
1367) 40AM 945 160AM 5 -79.5 160AM 4 778 320AM -74.5 640AM -72 1280AM -69 1280AM -66 1510AM -61.5 1280AM -56.5 28480AM L -55 160AM S -77 320AM -73.5 160AM -77 320AM -75.5 160AM -71 1280AM -68.5 250CAM -68 250CAM -55 1280AM -55.5 28480AM L -54 49960AM L -64 4020AM -65	60	4QAM S	-86
160AMS -785 16 0AM -78 320AM -745 64QAM -72 128QAM -69 250QAM -66 5120AM -615 182QAM -565 182QAM -565 28480AM -565 28480AM -555 40960AM -53 40960AM -53 40960AM -68 20480AM -885 20480AM -835 160AMS -785 160AMS -785 160AMS -77 320AM -735 64QAM -71 1280AM -66 512QAM -675 28480AM -555 64QAM -71 1280AM -555 64QAM -71 1280AM -555 28480AM -555 28480AM -555 28480AM -555 28480AM	(367)	4QAM	-84.5
16 QAM -78 32QAM -74.5 32QAM -74.5 40QAM -72 128QAM 60 256QAM -66 128QAM -56.5 128QAM -55.5 2848QAML -55.5 2848QAML -55.5 4996QAM -53.5 4996QAM -83.5 1668 4QAM -83.5 160AMS -78.5 160AMS -78.5 160AMS -77.5 160AM -77.5 202AM -65.5 160AM -77.5 202AM -75.5 40AM -71.1 1280AM -66 202AM -75.5 202AM -75.5 120AM -65 120AM -65 120AM -52 2040AML -54 2040AML -54 2040AM -52 2040AM -52		16QAM S	-79.5
320,4 -74,5 640,4M -72 1280,AM -69 2560,AM -66 5120,AM -56,5 2840,0AM -56,5 2840,0AM -56,5 2840,0AM -56,5 2840,0AM -55,5 4096,0AM -51,5 4096,0AM -51,5 4096,0AM -83,5 150,0B 40,AM S -85,5 160,0AM -77,7 320,AM -73,5 160,0AM -77,7 320,AM -73,5 160,0AM -77,7 320,AM -66,5 320,AM -61,5 160,0AM -77,5 160,0AM -77,5 160,0AM -75,5 1280,0AM -55,5 1280,0AM -55,5 1280,0AM -55,5 1280,0AM -52,5 1280,0AM -52,5 1280,0AM -52,5 1374) 40,0AM -72,5 <td></td> <td>16 QAM</td> <td>-78</td>		16 QAM	-78
64QAM -72 128QAM -69 128QAM -60 512QAM -61.5 1824QAM -58.5 2848QAM -55.5 2848QAM -55 4996QAM -55 4996QAM -55 4996QAM -55 4996QAM -55 4996QAM -85.5 100AMS -85.5 16QAMS -78.5 16QAM -77.5 16QAM -77.5 16QAM -77.5 16QAM -77.5 16QAM -77.5 120QAM -68 256QAM -66 122QAM -57.5 204BQAM -77.5 204BQAM -57.5 204BQAM -55.5 204BQAM -52.7 204BQAM -52.7 204BQAM -69.5 124QAM -52.5 204BQAM -72.5 204AQAM -72.5		32QAM	-74.5
1280,M -69 250,AM -66 10240,AM -58.5 20480,AM L -56.5 20480,AM L -55 20490,AM L -51 40960,AM L -51 40960,AM L -51 40960,AM L -51 40960,AM L -51 400,AM S -85 100,AM S -78.5 120,AM -71 220,AM -71 220,AM -75 220,AM -68 250,AM -55.5 2040,AM L -55.5 20440,AM L -55.5 20440,AM L -52 20440,AM L -52 20440,AM -52 20440,AM -52 20440,AM -52 20440,AM <td></td> <td>64QAM</td> <td>-72</td>		64QAM	-72
1260AM -66 5120AM -61.5 10240AM -58.5 20480AM L -55 20490AM L -51.5 20960AM L -51.5 40960AM L -51.5 80 (368) 1368) 40AM S -85 160AM S -78.5 160AM S -77.5 2040AM -77.5 2040AM -60.5 120AM -66 120AM -77.5 2040AM -77.5 2040AM -77.5 2040AM -65 120AM -66 120AM -65 120AM -65 120AM -55.5 2040AM L -52 2040AM L -52 2040AM S -52 2040AM S -63.5 120AM -64.5 120AM -52 40960AM L -52 1040AM S -77 160AM S -77 <td></td> <td>128QAM</td> <td>-69</td>		128QAM	-69
5120AM -61.5 1024QAM -56.5 2048QAM -56.5 2048QAM -55 4096QAM -51.5 4096QAM -51.5 4096QAM -51.5 4096QAM -51.5 4096QAM -63.5 150AM -78.5 16QAM S -78.5 16QAM -77.5 20QAM -66 250QAM -66 250QAM -65.5 120QAM -65.5 120QAM -55.5 204QAM -55.5 204QAM -55.5 2048QAM -52.5 2048QAM -62 16QAM -77 16QAM -75.5 204QAM -62.5 <		256QAM	-66
10240AM -58.5 20480AM -55.5 4096QAM -53.5 4096QAM -53.5 4096QAM -53.5 4096QAM -53.5 4096QAM -53.5 4096QAM -85.5 40AM -85.5 1508) 4QAM -85.5 16QAM -77.5 16QAM -77.5 32QAM -73.5 64QAM -68 250AM -68 250AM -68 250AM -66 5120AM -68 220AM -57.5 20480AM -57.5 20480AM -52 20480AM -52 20480AM -52 20480AM -49.5 112 40AM -88.2 1374) 40AM -89.2 160AM -77.7 160AM -77.5 320AM -66.5 204M -60.5		512QAM	-61.5
28480AM -56.5 28480AM L -55 284960AM L 51.5 40960AM L -51.5 40960AM L -85.5 40960AM L -85.5 1368) 4QAM S -85.5 160AM S -78.5 160AM S -78.5 64QAM -71.5 520AM -73.5 64QAM -71.5 520AM -68 250AM -68.5 1280AM -68.5 1280AM -68.5 1280AM -68.5 1220AM -68.5 1220AM -68.5 1220AM -57.5 28480AM L -54 40960AM L -49.5 112 40960AM L -49.5 1374) 40AM S -83.5 1374) 40AM S -77 160AM S -77 160AM S -77 160AM S -78.5 320AM -66.5 220AM </td <td></td> <td>1024QAM</td> <td>-58.5</td>		1024QAM	-58.5
20480AM L -55 40960AM -53 40960AM L -515 60 (368) 4QAM S -85 40AM S -855 160AM S -775 160AM S -775 320AM -771 128QAM -68 250QAM -675 64QAM -71 128QAM -68 250QAM -66 128QAM -68 250QAM -55 128QAM -55.5 2048QAM L -54 4096QAM L -54 4096QAM L -54 4096QAM L -49.5 112 (374) 4QAM S -83.5 1374) 4QAM -72.5 16QAM S -77 -16 1374) 4QAM -65.5 12QAM -65.5 -26 16QAM S -77 -16 16QAM -77.5 -20 16QAM -77.5 -50		2048QAM	-56.5
49960AM -53 49960AM L -51.5 40960AM S -85 40AM S -83.5 160AM S -77.8.5 160AM S -77.5 320AM -77.5 460AM -60.5 1280AM -65 5120AM -60.5 1240AM -57.5 20480AM L -55.5 20480AM L -54 40960AM L -49.5 112 404M S -83.5 1374) 40AM S -83.5 112 204M L -54 1374) 40AM S -83.5 1374) 40AM S -83.5 1374) 40AM S -83.5 1374) 40AM S -83.5 1374) 40AM S -60.5 1374) 40AM S -83.5 1374) 40AM S -82.5 1374) 40AM S -63.5 1374) 40AM S -64.5 140AM S -7		2048QAM L	-55
4096QAM L -51.5 80 (368) 4QAM S -85 1368) 4QAM S -83.5 16QAM S -77 32QAM -73.5 64QAM -71 32QAM -68 256QAM -66 256QAM -66 256QAM -65 128QAM -57.5 2048QAM L -57.5 2048QAM L -55.5 2048QAM L -54 4096QAM L -54 4096QAM L -54 4096QAM L -54 4096QAM L -49.5 1374) 4QAM S -83.5 1374) 4QAM S -83.5 1374) 4QAM S -772 64QAM -72 -640AM 1374) -56 -2204M 1374) -56 -2204M 140AM -60.5 -204 120QAM -72 -640AM -60.5 -204 -50.5 <		4096QAM	-53
80 (368) 40AM S -85 1368) 40AM -83.5 160AM S -78.5 160AM 3 -77.0 32QAM -73.5 64QAM -71 1280AM -68 2560AM -65 122QAM -55 1280AM -55.5 20480AM L -54 40960AM L -52 40960AM L -52 40960AM L -52 40960AM L -52 40960AM L -82 1374) 40AM S -83.5 1374) 40AM S -83.5 1374) 16 0AM S -77 1374) 2040M S -60.5 1280AM -60.5 -77 16 0AM S -77 -75 16 0AM S -77 -75 320AM -72 -640A 140AM -60.5 -77 1280AM -66.5 -65 1280AM -65 -6		4096QAM L	-51.5
40,M -83.5 160,AM S -78.5 160,AM S -77.3 32QAM -73.5 64QAM -71 1280,AM -68 2560,AM -65 512QAM -65.5 10240,AM -57.5 20480,AM L -54. 40960,AM L -49.5 112 40940,AM 1374) 40AM S 140,AM -82. 112 40,AM 1374) 40,AM 40,AM -82. 140,AM -82. 112 40,AM 1374) 40,AM 1374) 40,AM 40,AM -82. 160,AM -77. 160,AM -72. 640,AM -65.5 320,AM -72. 640,AM -65.5 320,AM -72. 640,AM -65.5 1280,AM -66.5 1280,AM -66.5	80	4QAM S	-85
160AM S -78.5 16 QAM -77 32QAM -73.5 64QAM -71 1280AM -68 256QAM -66 5120AM -66.5 1024QAM -57.5 2048QAM L -54 4096QAM L -52 4096QAM L -49.5 112 40AM S -83.5 1374) 40AM S -83.5 140AM -77 1374) 40AM S -77 16QAM -75.5 32QAM -75.5 32QAM -75.5 32QAM -76.5 1374) 16QAM S -77 16QAM -75.5 32QAM -72 64QAM -66.5 128QAM -66.5 128QAM -56 1024QAM -56 204QAM -56 204QAM -56 204QAM -52 2048QAML -52.5	1368)	4QAM	-83.5
16 QAM -77 32QAM -73.5 64QAM -71 128QAM -68 256QAM -60.5 112QAM -57.5 2048QAM -55.5 2048QAM -55.4 2049QAM -55.4 2049QAM -55.4 2049QAM -55.4 2049QAM -52 2049QAM -52 2049QAM -49.5 2049QAM -82 105/24 4096QAM -82 105/24 40AM S -77 16QAM -75.5 32QAM 1574 1374) -72 16QAM -69.5 -77 16QAM -69.5 -77 16QAM -66.5 -77 128QAM -66.5 -52 128QAM -66.5 -52 1024QAM -56 -2048QAM 1024QAM -56 -2048QAM 2048QAM -54 -2048QAM		16QAM S	-78.5
320AM -73.5 640AM -71 1280AM -68 2560AM -60.5 12820AM -60.5 12820AM -57.5 20480AM -55.5 20480AM -52 20480AM -52 40960AM -49.5 1102 -409.5 112 -404MS 40960AM -82 11374) -100AMS 112 -77 160AMS -77 160AMS -72 1280AM -66.5 320AM -66.5 120AM -66.5 320AM -72 160AMS -72 160AM -59 1280AM -66.5 1280AM -65.5 120AM -56 120AM -56 120AM -56 120AM -54 10240AM -54 10240AM -54 10240AM		16 QAM	-77
64QAM -71 128QAM -68 256QAM -60.5 1024QAM -57.5 2048QAM L -54 2096QAM L -49.5 4096QAM L -49.5 1374) 4QAM S -83.5 1374) 4QAM S -77 16QAM -75.5 -75.5 2048QAM L -49.5 -77 1374) 4QAM S -83.5 1374) 4QAM -82 112 (374 1374) 16QAM S -77 16QAM S -77 -71 16QAM S -72 -64QAM 16QAM -66.5 -65.5 204QAM -66.5 -65.5 128QAM -66.5 -65.5 128QAM -66.5 -65.5 128QAM -66.5 -65.5 128QAM -56 -65.5 2048QAM L -56 -65.5 2048QAM L -56 -65.5 2048QAM L -52.5 <		32QAM	-73.5
1280AM -68 2500AM -65 5120AM -60.5 10240AM -57.5 20480AM L -54 20400AM L -54 40960AM L -49.5 1020AM S -83.5 40960AM L -82 1574 40AM S -77 160AM S -77 160AM S -77 160AM S -72 640AM -69.5 120AM -63.5 120AM -56 120AM -56 120AM -56 120AM -54 20480AM L -52.5 104960AM L -50.5		64QAM	-71
250QAM -65 512QAM -605 1024QAM -57.5 2048QAM -55.5 2048QAM -54 4096QAM -52 4096QAM -49.5 112 4096QAM (574) -404MS 1374) 40AM S 40AM -82 16QAM S -77 16QAM S -772 16QAM -69.5 32QAM -66.5 256QAM -66.5 128QAM -66.5 128QAM -56 204QAM -56 204QAM -56 204QAM -56 204QAM -52		128QAM	-68
512QAM -60.5 1024QAM -57.5 2048QAM L -54 4096QAM L -52 4096QAM L -49.5 112 40AM S -83.5 157.4 4QAM S -82 16QAM S -77 -75.5 32QAM -72.5 -60.5 4QAM -60.5 -60.5 128QAM -66.5 -61.5 128QAM -66.5 -61.5 128QAM -66.5 -61.5 128QAM -56 -20.4 1024QAM -56 -20.4 2048QAM -54 -20.4 2048QAM -56 -20.4 2048QAM -54 -20.4 2048QAM -54 -20.4 2048QAM -52.5 -50.5		256QAM	-65
1024QAM -57.5 2048QAM -55.5 2048QAM L -54 4096QAM -52 4096QAM L -49.5 112 4QAM S -83.5 1374) 4QAM S -82 16QAM S -77 16 QAM 16QAM S -75.5 32QAM 12 52 52 16QAM S -77 16 QAM 16QAM S -77 16 QAM 16QAM S -72 64QAM 128QAM -66.5 128QAM 128QAM -66.5 128QAM 1024QAM -59 1024QAM 1024QAM -56 2048QAM L 2048QAM L -52.5 4096QAM 4096QAM L -50.5 1204		512QAM	-60.5
2048QAM -55.5 2048QAM L -54 4096QAM -52 4096QAM L -49.5 4096QAM S -83.5 1374) 4QAM S -82 16QAM S -77 16QAM S -75.5 32QAM -72 64QAM -69.5 128QAM -66.5 128QAM -66.5 128QAM -63.5 120AM -59 1024QAM -56 2048QAM L -54 2048QAM L -52.5 4096QAM L -50.5		1024QAM	-57.5
2048QAM L -54 4096QAM -52 4096QAM L -49.5 112 (374) 4QAM S -83.5 1374) 4QAM S -82 16QAM S -77 16QAM -75.5 32QAM -72 64QAM -69.5 128QAM -66.5 256QAM -59 1024QAM -59 1024QAM -54 2048QAM L -52.5 409AQAM -52.5		2048QAM	-55.5
4096QAM -52 4096QAM L -49.5 112 (374) 4QAM S -83.5 1374) 4QAM S -77 16QAM S -77 16QAM -75.5 32QAM -72 64QAM -69.5 128QAM -66.5 256QAM -59 1024QAM -56 2048QAM L -54 2048QAM L -52.5 4096QAM L -50.5		2048QAM L	-54
4096QAM L -49.5 112 (374) 4QAM S -83.5 1374) 4QAM S -82 16QAM S -77 16 QAM S -75.5 32QAM -72 64QAM -69.5 128QAM -66.5 256QAM -63.5 512QAM -59 1024QAM -56 2048QAM L -54 2048QAM L -52.5 4096QAM L -50.5		4096QAM	-52
112 (374) 4QAM S -83.5 4QAM -82 16QAM S -77 16QAM M -75.5 32QAM -72 64QAM -69.5 128QAM -66.5 256QAM -63.5 512QAM -59 1024QAM -56 2048QAM L -52.5 4096QAM -50.5		4096QAM L	-49.5
4QAM -82 16QAM S -77 16 QAM -75.5 32QAM -72 64QAM -69.5 128QAM -66.5 256QAM -63.5 512QAM -59 1024QAM -54 2048QAM L -52.5 4096QAM -50.5	112	4QAM S	-83.5
16QAM S -77 16 QAM -75.5 32QAM -72 64QAM -69.5 128QAM -66.5 256QAM -63.5 512QAM -59 1024QAM -56 2048QAM L -52.5 4096QAM -50.5	1374)	4QAM	-82
16 QAM -75.5 32QAM -72 64QAM -69.5 128QAM -66.5 256QAM -63.5 512QAM -59 1024QAM -56 2048QAM L -52.5 4096QAM -50.5		16QAM S	-77
32QAM -72 64QAM -69.5 128QAM -66.5 256QAM -63.5 512QAM -59 1024QAM -56 2048QAM L -52.5 4096QAM -50.5		16 QAM	-75.5
64QAM -69.5 128QAM -66.5 256QAM -63.5 512QAM -59 1024QAM -56 2048QAM L -52.5 4096QAM -50.5		32QAM	-72
128QAM -66.5 256QAM -63.5 512QAM -59 1024QAM -56 2048QAM -54 2048QAM L -52.5 4096QAM -50.5		64QAM	-69.5
256QAM -63.5 512QAM -59 1024QAM -56 2048QAM -54 2048QAM L -52.5 4096QAM -50.5		128QAM	-66.5
512QAM -59 1024QAM -56 2048QAM -54 2048QAM L -52.5 4096QAM -50.5		256QAM	-63.5
1024QAM -56 2048QAM -54 2048QAM L -52.5 4096QAM -50.5		512QAM	-59
2048QAM -54 2048QAM L -52.5 4096QAM -50.5		1024QAM	-56
2048QAM L -52.5 4096QAM -50.5		2048QAM	-54
4096QAM -50.5		2048QAM L	-52.5
		4096QAM	-50.5

Table 70 Detection Performance, ANSI

The detection performance is applicable to the following radio types:

- ____ RAU2 X $10^{(1)}$ (Dupl. ≤ 168 MHz), 18, 23, 24
- MINI-LINK 10⁽¹⁾(Dupl. ≤ 168 MHz), 18, 23, 26
- ____ MINI-LINK 6363/2 18, 23
- ____ MINI-LINK 6365 18, 23

CS (Erame ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)
(Hume ID)		Тур.
All	All	Add 1 dB to figures in Table 69 above

(1) Max CS=20 MHz

Table 71 Detection Performance, ANSI

The detection performance is applicable to the following radio types:

- ____ RAU2 X 28, 32, 38
- ____ MINI-LINK 6363 28, 32, 38
- ____ MINI-LINK 6363/2 38
- ____ MINI-LINK 6365 32

CS (Frame ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)		
		Тур.		
All	All	Add 2 dB to figures in Table 69 above		

Table 72 Detection Performance, ANSI

The detection performance is applicable to the following radio types:

- ____ RAU2 X 42
- ____ MINI-LINK 6363 42

CS (Frame ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)		
		Тур.		
All	All	Add 3 dB to figures in Table 69 above		

Table 73 Detection Performance, ANSI

The detection performance is applicable to the following radio type:

____ MINI-LINK 6363 80

CS (Frame ID)	Modulation	BER 10 ⁻⁶ threshold (dBm)		
(Hume ID)		Тур.		
60 (367)	All	Add 7 dB to figures in Table 69 above		
80 (368)				

3.6.3 Switching Level, Adaptive Modulation

Typical switching levels between different physical modes are at Residual BER thresholds and are based on a measurement of the SNIR in the received signal.

For path planning usage this RBER level for down switching can be approximated to 4 dB above the typical 10^{-6} BER threshold for the currently used modulation scheme and CS.

For Frame ID 2358, 2360, 2364 and 2365 the RBER level for down switching can be approximated to 6 dB above the typical 10^{-6} BER threshold for the currently used modulation scheme and CS.

3.6.4 Co-channel Interference for QAM

The limits of co-channel interference are as given in table below, giving C/I values for 1 dB and 3 dB increase of the 10^{-6} BER thresholds, specified in chapter Detection Performance for QAM, Adaptive Modulation on page 122.

Co-channel	C/I values for 1 dB & 3 dB		
Modulation	1dB	3dB	
4QAM S	13	9	
4QAM	14	10	
16QAM S	19	15	
16QAM	21	17	
32QAM	24	20	
64QAM	27	23	
128QAM	30	26	
256QAM	34	30	
512QAM	36.5	32.5	
1024QAM	40	36	
2048QAM	42	38	

Table 74 Limits of Co-channel Interference

Co-channel	C/I values for 1 dB & 3 dB				
Modulation	1dB	3dB			
2048QAM L	43	39			
4096QAM	45	41			
4096QAM L	48	44			
8192QAM	51	47			

3.6.5 Adjacent Channel Interference for QAM

3.6.5.1 ETSI

The limits of first adjacent-channel interference are as given in table below, giving C/I values for 1 dB and 3 dB increase of the 10^{-6} BER thresholds, specified in Detection Performance, ETSI on page 122.

Table 75 R/	AU2 X and	MINI-LINK	6363
-------------	-----------	-----------	------

Modulation	RSEC	C/I values for 1 dB and 3 dB			
		1 dB	3dB		
4QAM	2	-16	-20		
	4L/4H	-20	-24		
	5B/6B	-23	-27		
16QAM	2/4L/4H	-20	-24		
	5B/6B	-23	-27		
32QAM	2/4L/4H	-20	-24		
	5B/6B	-23	-27		
64QAM	2/4L/4H	-20	-24		
	5B/6B	-22	-26		
128QAM	All	-19	-23		
256QAM	All	-16	-20		
512QAM	All	-14	-14		
1024QAM	All	-10(1)	-10		
2048QAM	All	-7	-7		
4096QAM	All	1	1		
8192 QAM	All	1	1		

(1) Add 4 dB for MINI-LINK 6363 80GHz.

Commercial in confidence

Modulation	RSEC	C/I values for 1 dB	and 3 dB
		1 dB	3dB
4QAM	2	-16	-20
	4L/4H	-20	-24
	5B/6B	-23	-27
16QAM	2/4L/4H	-20	-24
	5B/6B	-23	-27
32QAM	2/4L/4H	-20	-24
	5B/6B	-23	-27
64QAM	2/4L/4H	-20	-24
	5B/6B	-22	-26
128QAM	All	-19	-23
256QAM	All	-16	-20
512QAM	All	-12	-16
1024QAM	All	-8	-12
2048QAM	All	-6	-10
4096QAM	All	-3	-7
8192QAM	All	0	-4

Table 76 MINI-LINK 6364 and MINI-LINK 6365

3.6.5.2 ANSI

The limits of first adjacent-channel interference are as given in table below, giving C/I values for 1 dB and 3 dB increase of the 10^{-6} BER thresholds, specified in Detection Performance, ANSI on page 127.

Table 77 RAU2 X and MINI-LINK 6363

Modulation	C/I values for 1 dB and 3	dB
	1 dB	3dB
4QAM	-23	-27
16QAM	-23	-27
32QAM	-23	-27
64QAM	-23	-27
128QAM	-19	-23
256QAM	-16	-20
512QAM	-14	-14

Modulation	C/I values for 1 dB and 3 dB					
	1 dB	3dB				
1024QAM	-10	-10				
2048QAM	-7	-7				
4096QAM	1	1				
8192QAM	1	1				

Table 78 MINI-LINK 6365

Modulation	C/I values for 1 dB and 3	dB	
	1 dB	3dB	
4QAM	-23	-27	
16QAM	-23	-27	
32QAM	-23	-27	
64QAM	-23	-27	
128QAM	-19	-23	
256QAM	-16	-20	
512QAM	-12	-16	
1024QAM	-8	-12	
2048QAM	-6	-10	
4096QAM	-3	-7	
8192QAM	0	-4	

3.6.6 CW Interference

For a receiver operating at the specified 10^{-6} threshold, the introduction of a CW interferer with C/I of -30 dB at any frequency up to 80 GHz, excluding a frequency 2.5 times the channel separation on either side of the wanted frequency, does not result in a BER greater than 10^{-5} .

3.6.7 Signature for QAM

Reference delay: 6.3 ns.

Minimum phase and non-minimum phase.

The table below states the Notch depth (dB) at BER 10⁻⁶, as specified in Detection Performance for QAM, Adaptive Modulation on page 122.

The Notch depth of BER 10^{-3} is 1 dB higher than the BER 10^{-6} value.

Table 79 ETSI/ANSI

BER 10-6 Notch depth (dB)													
Modulation	Signati	Signature width (MHz)											
	7.5	11	15	22	31	33	44	55	61	66	88	123	246
	Channe	el Separa	tion (MH:	z)					•				
	7	10	14	20	28	30	40	50	56	60	80	112	224
4QAM	40	40	40	40	40	40	40	40	40	40	40	40	40
16QAM	40	40	40	40	40	40	38	36	35	34	32	29	29
32QAM	40	40	40	38	36	34	32	30	29	28	26	23	23
64QAM	40	40	39	36	34	32	30	28	27	26	24	20	20
128QAM	40	40	37	34	31	30	28	26	25	24	22	18	18
256QAM	40	38	35	32	30	28	26	24	23	22	20	17	17
512QAM	40	37	34	31	29	27	25	23	22	21	19	16	16
1024QAM	39	36	33	30	27	26	24	22	21	20	18	15	15
2048QAM			32	29	26	25	23	21	20	19	17	14	14
4096QAM					25	23	21	19	18	17	15	13	13
8192QAM					21	20	18						

When the notch frequency is swept across the defined bandwidth with a rate up to 100 MHz/s, the notch depth for BER= 10^{-6} will not degrade by more than 1 dB with respect to the values listed in the table above.

3.6.8 Receiver Overload

Maximum input level where normal operation is guaranteed: -20 dBm

3.6.9 Receiver Resistibility

Maximum tolerable input power without permanent degradation is 0 dBm.

- 3.6.10 Received Signal Indication
- 3.6.10.1 For Path Acceptance

When measured in steady state condition, an RF-input level measure is given with an accuracy of:

- for RF-input levels -30 dBm to -60 dBm: ± 2 dB
- for RF-input levels -60 dBm to -80 dBm: ± 3 dB

3.6.10.2 For Antenna Alignment

This detector presents on an external standard voltmeter a voltage corresponding to RF-input level.

The RF input level referred to the antenna port in dBm can be calculated from the measured voltage using the following formula: Rfin(dBm) = 40^{*} measured_voltage(V) - 120

When measured at the Antenna Alignment Port, the accuracy of the measured voltage converted to RF-input level using the formula above shall be:

- for RF-input levels -30 dBm to -60 dBm: ± 2.5 dB
- for RF-input levels -60 dBm to -80 dBm: ± 3 dB

3.7 Supported Physical Modes

A physical mode is supported when both the radio unit and the MMU supports it. For readability purpose, the radio unit never show a higher physical mode than what is supported by the most capable MMU.

3.7.1 Radio Units

3.7.1.1 MINI-LINK RAU2 X HW

MINI-LINK RAU2 X HW supports physical modes according to the tables below. The physical modes supported are denoted by an X. See Radio Frequencies on page 74 for supported frequencies and channel separation for a specific RAU2 X.

Table 80	Supported Physical Modes ETSI

Telecom Standa	Telecom Standard: E 1 S1								
Products: RAU2 X 5-42 ≥ R6A									
CS	7	14/13.75	28/27.5	40	56/55				
Mod.	MHz	MHz	MHz	MHz	MHz				
4QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾				
4QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾				
16QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾				
16QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾				
32QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾				
64QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾				
128QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾				
256QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾				

Telecom Standard: ETSI								
Products: RAU2 X 5-42 ≥ R6A								
CS	7	14/13.75	28/27.5	40	56/55			
Mod.	MHz	MHz	MHz	MHz	MHz			
512QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
1024QAM	Х	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
2048QAM		X ⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾			
2048QAM L		x ⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾			
4096QAM			X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾			
4096QAM L			X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾			

(1) Supports XPIC

(2) Requires RAU SW CXP 901 2878 R5A or later

Table 81 Supported Physical Modes ETSI

Telecom Standard: ETSI								
Products: RAU2 X 5−11 < R6A, RAU2 X 13&15 ≥ R3A & < R6A, RAU2 X 18&23 ≥ R5A & < R6A								
CS	7	14/13.75	28/27.5	40	56/55			
Mod.	MHz	MHz	MHz	MHz	MHz			
4QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
4QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
16QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
16QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
32QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
64QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
128QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
256QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
512QAM	X	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
1024QAM		X ⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾			

(1) Supports XPIC

(2) Requires RAU SW CXP 901 2878 R-state R4D or later (RAU2 X/Xu 10/11/18/23 R-state R1A use RAU SW CXC 113 500 and hence cannot support 1024 QAM.)

Table 82	Supported	Physical	Modes	ETSI
----------	-----------	----------	-------	------

Telecom Standard: ETSI							
Products: RAU2 X 13&15 < R3A, RAU2 X 18&23 < R5A, RAU2 X 26-42 < R6A							
CS	7 14/13.75 28/27.5 40 56/55						
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ		
4QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾		
4QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾		

Telecom Standa	Telecom Standard: ETSI							
Products: RAU2 X 13&15 < R3A, RAU2 X 18&23 < R5A, RAU2 X 26-42 < R6A								
CS	7	14/13.75	28/27.5	40	56/55			
Mod.	MHz	MHz	MHz	MHz	MHz			
16QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
16QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
32QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
64QAM	X	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
128QAM	X	X	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
256QAM	X	X	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾			
512QAM	X	X	X	Х	X ⁽¹⁾			
1024QAM		X ⁽²⁾	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾			

(1) Supports XPIC

(2) Requires RAU SW CXP 901 2878 R-state R4D or later (RAU2 X/Xu 10/11/18/23 R-state R1A use RAU SW CXC 113 500 and hence cannot support 1024 QAM.)

Table 83	Supported Physica	l Modes ANSI
----------	-------------------	--------------

Telecom Standard: ANSI								
Products: RAU2 X 5-42 ≥ R6A								
CS	10	20	28	30	40	50	56	
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	
4QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
4QAM	X ⁽¹⁾	X ⁽¹⁾	χ(1)	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
16QAM S	X ⁽¹⁾	X ⁽¹⁾	χ(1)	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
16QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
32QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
64QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
128QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
256QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
512QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
1024QAM	Х	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
2048QAM		X ⁽²⁾	X ⁽¹⁾⁽²⁾					
2048QAM L		X ⁽²⁾	X ⁽¹⁾⁽²⁾					
4096QAM			X ⁽¹⁾⁽²⁾					
4096QAM L			X ⁽¹⁾⁽²⁾					

(1) Supports XPIC

(2) Requires RAU SW CXP 901 2878 R5A or later

Telecom Stand	Telecom Standard: ANSI								
Products: RAU	Products: RAU2 X 6L−11 < R6A, RAU2 X 13&15 ≥ R3A & < R6A, RAU2 X 18&23 ≥ R5A & < R6A								
CS	10	20	28	30	40	50	56		
Mod.	MHz	MHz	MHz	MHz	MHz	MHz	MHz		
4QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾		
4QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	χ(1)		
16QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	χ(1)		
16QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	χ(1)		
32QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	χ(1)		
64QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾		
128QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾		
256QAM	Х	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾		
512QAM	X	Х	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾		
1024QAM		X ⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	χ(1)(2)	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾		

Table 84 Supported Physical Modes ANSI

(1) Supports XPIC

(2) Requires RAU SW CXP 901 2878 R-state R4D or later (RAU2 X/Xu 10/11/18/23 R-state R1A use RAU SW CXC 113 500 and hence cannot support 1024 QAM.)

Table 85	Supported Physical Modes ANSI	
----------	-------------------------------	--

Telecom Standard: ANSI									
Products: RAU2 X 13 & 15 < R3A, RAU2 X 18 & 23 < R5A, RAU2 X 24 - 38 < R6A									
CS	10	20	28	30	40	50	56		
Mod.	MHz								
4QAM S	X ⁽¹⁾								
4QAM	X ⁽¹⁾	χ(1)	X ⁽¹⁾						
16QAM S	X ⁽¹⁾								
16QAM	X ⁽¹⁾								
32QAM	X ⁽¹⁾								
64QAM	X	X ⁽¹⁾							
128QAM	Х	Х	X ⁽¹⁾						
256QAM	Х	Х	X ⁽¹⁾						
512QAM	X	X	X	X	X	X ⁽¹⁾	X (1)		
1024QAM		χ ⁽²⁾	x ⁽²⁾						

(1) Supports XPIC

(2) Requires RAU SW CXP 901 2878 R-state R4D or later (RAU2 X/Xu 10/11/18/23 R-state R1A use RAU SW CXC 113 500 and hence cannot support 1024 QAM.)

3.7.1.2 MINI-LINK 6363, MINI-LINK 6363/2 HW

MINI-LINK 6363 and MINI-LINK 6363/2 HW supports physical modes according to the tables below. The physical modes supported are denoted by an X, and M in those cases where only MIMO is supported. See Radio Frequencies on page 74 for supported frequencies and channel separations for a specific frequency band.

Telecom Stand	lard: ETSI						
Products: MIN	I-LINK 6363	6L- 42 < R4A, MINI	-LINK 6363/2 1	3 – 38 ≥ R1A			
CS	7	14/13.75	28/27.5	40	56/ 55/62.5	80	112/ 110/125
Mod.	MHz	MHz	MHz	MHz	MHz	MHz	MHz
4QAM S	X ⁽¹⁾	X ⁽¹⁾	χ ⁽¹⁾ (2)	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
4QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
16QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
16QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
32QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
64QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
128QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
256QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
512QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
512QAM L			M ⁽²⁾		M ⁽²⁾		
1024QAM	Х	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
1024QAM L					M ⁽²⁾		
2048QAM		Х	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾
2048QAM L		X	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	χ(1)	X ⁽¹⁾
4096QAM			X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	Х	X
4096QAM L			X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X	
8192QAM							

Table 86 Supported Physical Modes ETSI

(1) Supports XPIC

(2) Supports MIMO

Table 87Supported Physical Modes ETSI

Telecom Stand	ard: ETSI										
Products: MINI-LINK 6363 10 - 26 ≥ R4A, MINI-LINK 6363 38 ≥ R4A											
CS	CS 7 14/13.75 28/27.5 40 56/55/62.5 80 112/110/125										
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ				
4QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾				
			(2)								

Telecom Stand	dard: ETSI						
Products: MIN	I-LINK 6363	10 – 26 ≥ R4A, MIN	VI-LINK 6363 38	l≥ R4A			
CS	7	14/13.75	28/27.5	40	56/ 55/62.5	80	112/110/125
Mod.	MHz	MHz	MHZ	MHZ	MHz	MHZ	MHZ
4QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
16QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
16QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
32QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
64QAM	χ(1)	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
128QAM	χ(1)	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
256QAM	χ(1)	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
512QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
512QAM L			M ⁽²⁾		M ⁽²⁾		
1024QAM	Х	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	X ⁽¹⁾
1024QAM L					M ⁽²⁾		
2048QAM		Х	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾
2048QAM L		Х	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾
4096QAM			X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾
4096QAM L			X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
8192QAM			X ⁽³⁾	X ⁽³⁾			

(1) Supports XPIC

(2) Supports MIMO

(3) Note that static modulation is not supported for 8192QAM

Table 88 Supported Physical Modes ETSI

Telecom Stand	ard: ETSI						
Products: MIN	I-LINK 6363 80 ≥	R1A					
CS	7	14/13.75	28/27.5	40	56/ 55/62.5	80	112/110/125
Mod.	MHz	MHZ	MHZ	MHz	MHZ	MHz	MHz
4QAM S					X ⁽¹⁾		X ⁽¹⁾
4QAM					X ⁽¹⁾		X ⁽¹⁾
16QAM S					X ⁽¹⁾		X ⁽¹⁾
16QAM					χ(1)		X ⁽¹⁾
32QAM					X ⁽¹⁾		X ⁽¹⁾
64QAM					X ⁽¹⁾		X ⁽¹⁾
128QAM					X ⁽¹⁾		X ⁽¹⁾
256QAM					X ⁽¹⁾		X ⁽¹⁾
512QAM					X ⁽¹⁾		X ⁽¹⁾
1024QAM					Х		Х

Telecom Stand	Telecom Standard: ETSI										
Products: MINI-LINK 6363 80 ≥ R1A											
CS	7 14/13.75 28/27.5 40 56/ 55/62.5 80 112/ 110/1										
Mod.	MHz	MHz	MHz	MHz	MHz	MHz	MHZ				
2048QAM											
2048QAM L											
4096QAM											
4096QAM L											
8192QAM											

(1) Supports XPIC

3

Table 89 Supported Physical Modes ANSI

Telecom Sta	ndard: AN	SI								
Products: MI	NI-LINK 6	5363 6L – 42	2 < R4A, MIN	II-LINK 6363	/2 13 – 38 ≥	: R1A				
CS	10	20	28	30	40	50	56	60	80	112
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ
4QAM S	χ(1)	X ⁽¹⁾	X ⁽¹⁾	χ ⁽¹⁾ (2)	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
4QAM	χ(1)	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
16QAM S	χ ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
16QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
32QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
64QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
128QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
256QAM	χ(1)	χ(1)	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾	χ ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾
512QAM	χ(1)	χ ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
512QAM L				M ⁽²⁾	M ⁽²⁾					
1024QAM	Х	χ(1)	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	χ(1)	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾
2048QAM		χ(1)	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	χ(1)	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾
2048QAM L		χ(1)	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	χ(1)	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾
4096QAM			X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	χ(1)	X ⁽¹⁾	Х	Х
4096QAM L			X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	χ(1)	X ⁽¹⁾	X	
8192QAM										

(1) Supports XPIC

(2) Supports MIMO

Telecom Sta	ndard: ANS	SI								
Products: MI	NI-LINK 6	5363 10 — 26	5 ≥ R4A, MIN	II-LINK 6363	38 ≥ R4A					
CS	10	20	28	30	40	50	56	60	80	112
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ
4QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	χ ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
4QAM	χ <mark>(1)</mark>	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
16QAM S	χ <mark>(1)</mark>	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
16QAM	χ <mark>(1)</mark>	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
32QAM	χ <mark>(1)</mark>	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
64QAM	χ <mark>(1)</mark>	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
128QAM	χ <mark>(1)</mark>	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
256QAM	χ <mark>(1)</mark>	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
512QAM	χ <mark>(1)</mark>	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾				
512QAM L				M ⁽²⁾	M ⁽²⁾					
1024QAM	Х	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾
2048QAM		X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾
2048QAM L		X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾
4096QAM			X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾
4096QAM L			X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
8192QAM				X ⁽³⁾	X ⁽³⁾					

Table 90 Supported Physical Modes ANSI

(1) Supports XPIC

(2) Supports MIMO

(3) Note that static modulation is not supported for 8192QAM

Table 91 Supported Physical Modes ANSI

Telecom Sta	ndard: ANSI										
Products MINI-LINK 6363 80 ≥ R1A											
CS	10	20	28	30	40	50	56	60	80	112	
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ	MHz	MHZ	MHZ	MHZ	MHZ	
4QAM S							X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
4QAM							X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
16QAM S							X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
16QAM							X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
32QAM							X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
64QAM							X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
128QAM							X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
256QAM							X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	

Telecom Star	ndard: ANS	51									
Products MINI-LINK 6363 80 ≥ R1A											
CS	10	20	28	30	40	50	56	60	80	112	
Mod.	MHz	MHz	MHz	MHz	MHz	MHz	MHz	MHz	MHz	MHz	
512QAM							X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
1024QAM							Х	Х	Х	Х	
2048QAM											
2048QAM L											
4096QAM											
4096QAM L											
8192QAM											

(1) Supports XPIC

3.7.1.3 MINI-LINK 6364 HW

MINI-LINK 6364 HW supports physical modes according to the tables below. The physical modes supported are denoted by an X, and M in those cases where only MIMO is supported. See <u>Radio Frequencies</u> on page 74 for supported frequencies and channel separations for a specific frequency band.

Table 92 Supported Physical Modes ETSI

Telecom Stand	ard: ETSI						
Products: MIN	I-LINK 6364 13	, 15 ≥ R1A ⁽¹⁾					
CS	7	14/13.75	28/27.5	40	56/55	80	112/110
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ
4QAM S	X ⁽²⁾	X ⁽²⁾	χ <mark>(2)</mark> (3)	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾
4QAM	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾⁽³⁾	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾
16QAM S	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾⁽³⁾	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾
16QAM	χ(2)	X ⁽²⁾	χ(2)(3)	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾
32QAM	χ(2)	X ⁽²⁾	χ ⁽²⁾⁽³⁾	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾
64QAM	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾
128QAM	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾
256QAM	X ⁽²⁾	X ⁽²⁾	X ^{(2) (3)}	X ⁽²⁾	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾
512QAM	χ(2)	X ⁽²⁾	χ(2) (3)	χ(2)	X ⁽²⁾⁽³⁾	χ(2)	X ⁽²⁾
512QAM L			M ⁽³⁾		M ⁽³⁾		
1024QAM	Х	X ⁽²⁾	X ⁽²⁾	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾
1024QAM L					M ⁽³⁾		
2048QAM		Х	χ(2)	χ(2)	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾
2048QAM L		Х	X ⁽²⁾	χ(2)	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾
4096QAM			X ⁽²⁾	χ(2)	X ⁽²⁾	χ(2)	X ⁽²⁾
4096QAM L			X ⁽²⁾	χ(2)	X ⁽²⁾	χ(2)	

Telecom Stande	ard: ETSI									
Products: MINI	Products: MINI-LINK 6364 13, $15 \ge R1A^{(1)}$									
CS	7 14/13.75 28/27.5 40 56/55 80 112/110									
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ			
8192QAM			X ⁽⁴⁾	X ⁽⁴⁾						

(1) The minimum modulation shall be set to 512 QAM or lower, for both static and adaptive modulation, when using MINI-LINK 6364 radio units. This is applicable for countries complying to ETSI EN 302 217.

(2) Supports XPIC

(3) Supports MIMO

(4) Note that static modulation is not supported for 8192QAM. See also (1).

Telecom Standard: ETSI									
Products: MI	NI-LINK 630	64 18 ≥ R1A ⁽¹⁾							
CS	7	14/13.75	28/27.5	40	56/55	80	112/110	220 ⁽²⁾	
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	
4QAM S	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾ (4)	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х	
4QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х	
16QAM S	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х	
16QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х	
32QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х	
64QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х	
128QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х	
256QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾ (4)	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х	
512QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾ (4)	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х	
512QAM L			M ⁽⁴⁾		M ⁽⁴⁾				
1024QAM	х	X ⁽³⁾	χ(3)	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х	
1024QAM L					M ⁽⁴⁾				
2048QAM		Х	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾		
2048QAM L		Х	χ(3)	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾		
4096QAM			χ(3)	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾		
4096QAM L			χ(3)	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾			
8192QAM			X ⁽⁵⁾	X ⁽⁵⁾					

Table 93Supported Physical Modes ETSI

(1) The minimum modulation shall be set to 512 QAM or lower, for both static and adaptive modulation, when using MINI-LINK 6364 radio units. This is applicable for countries complying to ETSI EN 302 217.

(2) For CS 220 MHz, Carrier Aggregation with 2x110 MHz is configured (Frame ID 404).

(3) Supports XPIC

(4) Supports MIMO

(5) Note that static modulation is not supported for 8192QAM. See also (1).

Telecom Standard: ETSI with Carrier Aggregation									
Products: MINI-LINK 6364 13, 15 \geq R1A ⁽¹⁾									
CS	28/27.5	40	56/55	80	112/110				
Mod.	MHz	MHz	MHz	MHz	MHz				
4QAM S	X		Х						
4QAM	X		Х						
16QAM S	X		Х						
16QAM	Х		Х						
32QAM	Х		Х						
64QAM	X		Х						
128QAM	X		Х						
256QAM	Х		Х						
512QAM	X		Х						
1024QAM	X		Х						
2048QAM	Х		Х						
2048QAM L	X								
4096QAM	X								
4096QAM L									
8192QAM									

Table 94 Supported Physical Modes ETSI with Carrier Aggregation

(1) The minimum modulation shall be set to 512 QAM or lower, for both static and adaptive modulation, when using MINI-LINK 6364 radio units. This is applicable for countries complying to ETSI EN 302 217.

Table 95 Supported Physical Modes ETSI with Carrier Aggregation

Telecom Standard: ETSI with Carrier Aggregation									
Products: MINI-LINK 6364 18 \ge R1A ⁽¹⁾									
CS	28/27.5	40	56/55	80	112/110				
Mod.	MHz	MHz	MHz	MHz	MHz				
4QAM S	Х		Х		X				
4QAM	Х		Х		X				
16QAM S	Х		Х		X				
16QAM	Х		Х		X				
32QAM	Х		Х		Х				
64QAM	Х		Х		X				
128QAM	Х		Х		X				
256QAM	Х		Х		X				
512QAM	Х		Х		X				
1024QAM	Х		Х		Х				
2048QAM	Х		Х						
2048QAM L	Х								
4096QAM	Х								
4096QAM L									

-

Telecom Standard: ETSI with Carrier Aggregation									
Products: MINI-LINK 6364 18 ≥ R1A ⁽¹⁾									
CS	28/27.5	40	56/55	80	112/110				
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ				
8192QAM									

(1) The minimum modulation shall be set to 512 QAM or lower, for both static and adaptive modulation, when using MINI-LINK 6364 radio units. This is applicable for countries complying to ETSI EN 302 217.

3.7.1.4 MINI-LINK 6365 HW

MINI-LINK 6365 HW supports physical modes according to the tables below. The physical modes supported are denoted by an X, and M in those cases where only MIMO is supported. See <u>Radio Frequencies</u> on page 74 for supported frequencies and channel separations for a specific frequency band.

Table 96 Supported Physical M

Telecom Standard: ETSI									
Products: MIN	I-LINK 6365 (6 - 15 ≥ R1A ⁽¹⁾							
CS	7	14/13.75	28/27.5	40	56/55	80	112/110		
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ		
4QAM S	X ⁽²⁾	X ⁽²⁾	χ ⁽²⁾ (3)	χ ⁽²⁾	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾		
4QAM	X ⁽²⁾	X ⁽²⁾	χ ⁽²⁾⁽³⁾	χ ⁽²⁾	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾		
16QAM S	X ⁽²⁾	X ⁽²⁾	χ ⁽²⁾⁽³⁾	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾		
16QAM	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾⁽³⁾	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾		
32QAM	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾⁽³⁾	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾		
64QAM	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾⁽³⁾	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾		
128QAM	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾⁽³⁾	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾		
256QAM	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾ (3)	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾		
512QAM	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾ (3)	χ(2)	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾		
512QAM L			M ⁽³⁾		M ⁽³⁾				
1024QAM	Х	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾⁽³⁾	X ⁽²⁾	X ⁽²⁾		
1024QAM L					M ⁽³⁾				
2048QAM		X	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾		
2048QAM L		X	X ⁽²⁾	χ(2)	X ⁽²⁾	X ⁽²⁾	X ⁽²⁾		
4096QAM			(2)	X ⁽²⁾	X ⁽²⁾	x ⁽²⁾	X ⁽²⁾		
4096QAM L			X ⁽²⁾	X ⁽²⁾	X ⁽²⁾	x ⁽²⁾			
8192QAM			X ⁽⁴⁾	X ⁽⁴⁾					

(1) The minimum modulation shall be set to 512 QAM or lower, for both static and adaptive modulation, when using MINI-LINK 6365 radio units. This is applicable for countries complying to ETSI EN 302 217.

(2) Supports XPIC

(3) Supports MIMO

(4) Note that static modulation is not supported for 8192QAM. See also (1).

Telecom Standard: ETSI												
Products: MI	Products: MINI-LINK 6365 18, 23, $32 \ge R1A^{(1)}$											
CS	7	14/13.75	28/27.5	40	56/55	80	112/110	224/ 220 ⁽²⁾				
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ				
4QAM S	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х				
4QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	χ ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х				
16QAM S	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х				
16QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х				
32QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х				
64QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х				
128QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х				
256QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾ (4)	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	χ(3)	Х				
512QAM	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾ (4)	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х				
512QAM L			M ⁽⁴⁾		M ⁽⁴⁾							
1024QAM	х	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾⁽⁴⁾	X ⁽³⁾	X ⁽³⁾	Х				
1024QAM L					M ⁽⁴⁾							
2048QAM		Х	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾					
2048QAM L		Х	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾					
4096QAM			X ⁽³⁾	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾					
4096QAM L			X ⁽³⁾	X ⁽³⁾	X ⁽³⁾	X ⁽³⁾						
8192QAM			X ⁽⁵⁾⁽⁶⁾	X ⁽⁵⁾⁽⁶⁾								

Table 97 Supported Physical Modes ETSI

(1) The minimum modulation shall be set to 512 QAM or lower, for both static and adaptive modulation, when using MINI-LINK 6365 radio units. This is applicable for countries complying to ETSI EN 302 217.

(2) For MINI-LINK 6365 23 and 32 at CS 224 MHz, Carrier Aggregation with 2x112 MHz is configured (Frame ID 404). For MINI-LINK 6365 18 and CS 220 MHz, Carrier Aggregation with 2x110 MHz is configured (Frame ID 404).

(3) Supports XPIC

(4) Supports MIMO

(5) Note that static modulation is not supported for 8192QAM. See also (1).

(6) Note that 8192QAM is currently not supported on MINI-LINK 6365 32. The HW is prepared.

Table 98 Supported Physical Modes ETSI with Carrier Aggregation

Telecom Standard: ETSI with Carrier Aggregation								
Products: MINI-LINK 6365 6, 7/8, 10/11, 13, 15, 18, 23, $32 \ge R1A^{(1)}$								
CS	28/27.5	40	56/55	80	112/110			
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ			
4QAM S	Х	Х	Х	Х	Х			

Telecom Standar	d: ETSI with Carrier /	Aggregation							
Products: MINI-LINK 6365 6, 7/8, 10/11, 13, 15, 18, 23, $32 \ge R1A^{(1)}$									
CS	28/27.5	40	56/ 55	80	112/110				
Mod.	MHz	MHz	MHz	MHz	MHz				
4QAM	Х	X	Х	X	Х				
16QAM S	Х	X	Х	X	Х				
16QAM	Х	X	Х	X	Х				
32QAM	X	Х	Х	X	Х				
64QAM	Х	X	Х	X	Х				
128QAM	Х	Х	Х	Х	Х				
256QAM	Х	Х	Х	X	Х				
512QAM	Х	Х	Х	Х	Х				
1024QAM	X	Х	Х	X	Х				
2048QAM	Х	Х	Х						
2048QAM L	Х	Х							
4096QAM	Х								
4096QAM L									
8192QAM									

(1) The minimum modulation shall be set to 512 QAM or lower, for both static and adaptive modulation, when using MINI-LINK 6365 radio units. This is applicable for countries complying to ETSI EN 302 217.

Table 99	Supported Ph	nysical Modes ANSI
----------	--------------	--------------------

Telecom Standard: ANSI											
Products MINI-LINK 6365 6 - 32 ≥ R1A											
CS	10	20	28	30	40	50	56	60	80	112	
Mod.	MHZ	MHZ	MHZ	MHZ	MHz	MHZ	MHZ	MHZ	MHZ	MHZ	
4QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾					
4QAM	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾					
16QAM S	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾					
16QAM	χ <mark>(1)</mark>	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾					
32QAM	χ ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾					
64QAM	χ ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾					
128QAM	χ ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾					
256QAM	χ <mark>(1)</mark>	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾					
512QAM	χ <mark>(1)</mark>	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾⁽²⁾	X ⁽¹⁾					
512QAM L				M ⁽²⁾	M ⁽²⁾						
1024QAM	Х	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
2048QAM		X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
2048QAM L		X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
4096QAM			X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	
4096QAM L			X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾	X ⁽¹⁾		

Telecom Standard: ANSI										
Products MINI-LINK 6365 6 - 32 ≥ R1A										
CS	10	20	28	30	40	50	56	60	80	112
Mod.	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ
8192QAM				X ^{(3) (4)}	X ⁽³⁾⁽⁴⁾					

(1) Supports XPIC

(2) Supports MIMO

(3) Note that static modulation is not supported for 8192QAM

(4) Note that 8192QAM is currently not supported on MINI-LINK 6365 32. The HW is prepared.

Table 100 Supported Physical Modes ANSI with Carrier Aggregation

Telecom Stand	lard: ANSI w	ith Carrier Aggı	regation					
Products MINI	I-LINK 6365	6, 7/8, 10/11, 1	L3, 15, 18, 23, 3	2 ≥ R1A				
CS	28	30	40	50	56	60	80	112
Mod.	MHz	MHz	MHz	MHz	MHz	MHz	MHz	MHz
4QAM S		Х	Х	Х		Х	Х	
4QAM		Х	Х	Х		Х	Х	
16QAM S		Х	Х	Х		Х	Х	
16QAM		Х	Х	Х		Х	Х	
32QAM		Х	Х	Х		Х	Х	
64QAM		Х	Х	Х		Х	Х	
128QAM		Х	Х	Х		Х	Х	
256QAM		Х	Х	Х		Х	Х	
512QAM		Х	Х	Х		Х	Х	
1024QAM		Х	Х	Х		Х	Х	
2048QAM		Х	Х	Х		Х		
2048QAM L		Х	Х					
4096QAM		Х						
4096QAM L								
8192QAM								

3.7.2 MMU

This chapter describes the traffic capacities of each MMU per carrier, depending on channel separation and modulation. Note that the support may require a certain SW.

Data for standard Frame ID's (256, 257, 258, 259, 260, 261, 262, 263, 264, 265, 266, 267, 268, and 303) can be found in revision BT of this document.

- Static physical modes are supported up to 4096QAM Light, by setting Adaptive Coding and Modulation (ACM) Max Capacity–Modulation equal to Min Capacity–Modulation.
- S= Strong coding and L= Light coding

- The figures in the tables related to maximum E1/DS1 are valid when TDM is used in a single direction. When TDM in 2 directions is enabled, the maximum number of E1/DS1 is limited to 40 per direction.
- **Note:** TDM is not supported by MINI-LINK 6366 and MINI-LINK 6371, therefore the maximum number of E1/DS1 in the following tables are not applicable to these network elements.

ETSI

Table 101 MMU 1001, PNM 1001, MINI-LINK 6651/1, MINI-LINK 6651/3 and MINI-LINK 6366

Modulation	CS	CS	CS	CS	CS	CS	CS
	7	13.75, 14	27.7, 28	40	55, 56, 62.5	80	110, 112, 125
	Frame ID 356	Frame ID 357	Frame ID 358	Frame ID 359	Frame ID 360	Frame ID 403	Frame ID 361
4QAM S	8 Mbps	19 Mbps	37 Mbps	53 Mbps	75 Mbps	108 Mbps	151 Mbps
	2XE1 ⁽¹⁾	6xE1 ⁽¹⁾	15xE1 ⁽¹⁾	23xE1 ⁽¹⁾	34xE1 ⁽¹⁾	49xE1 ⁽¹⁾	70xE1 ⁽¹⁾
4QAM	10 Mbps	22 Mbps	44 Mbps	62 Mbps	88 Mbps	126 Mbps	176 Mbps
	3XE1 ⁽¹⁾	8xE1 ⁽¹⁾	18xE1 ⁽¹⁾	28xE1 ⁽¹⁾	40xE1 ⁽¹⁾	58xE1 ⁽¹⁾	80xE1 ⁽¹⁾
16QAM S	18 Mbps	37 Mbps	75 Mbps	107 Mbps	151 Mbps	217 Mbps	302 Mbps
	6xE1 ⁽¹⁾	15xE1 ⁽¹⁾	34xE1 ⁽¹⁾	49xE1 ⁽¹⁾	70xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾
16QAM	21 Mbps	43 Mbps	87 Mbps	124 Mbps	176 Mbps	253 Mbps	352 Mbps
	7xE1 ⁽¹⁾	18xE1 ⁽¹⁾	33xE1 ⁽¹⁾	57xE1 ⁽¹⁾	71xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾
32QAM	26 Mbps	54 Mbps	109 Mbps	156 Mbps	221 Mbps	317 Mbps	441 Mbps
	10xE1 ⁽¹⁾	24xE1 ⁽¹⁾	50xE1 ⁽¹⁾	73xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾
64QAM	33 Mbps	68 Mbps	137 Mbps	196 Mbps	277 Mbps	397 Mbps	553 Mbps
	13xE1 ⁽¹⁾	30xE1 ⁽¹⁾	63xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾
128QAM	39 Mbps	81 Mbps	162 Mbps	231 Mbps	327 Mbps	470 Mbps	654 Mbps
	16xE1 ⁽¹⁾	36xE1 ⁽¹⁾	75xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾
256QAM	45 Mbps	93 Mbps	186 Mbps	267 Mbps	377 Mbps	542 Mbps	754 Mbps
	19xE1 ⁽¹⁾	42xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾
512QAM	51 Mbps	105 Mbps	211 Mbps	302 Mbps	427 Mbps	614 Mbps	855 Mbps
	22xE1 ⁽¹⁾	48xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾
1024QAM	56 Mbps	117 Mbps	234 Mbps	335 Mbps	473 Mbps	679 Mbps	946 Mbps
	25xE1	54xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾	80xE1 ⁽¹⁾
2048QAM	-	121 Mbps 56xE1	243 Mbps 80xE1 ⁽¹⁾	348 Mbps 80xE1 ⁽¹⁾	492 Mbps 80xE1 ⁽¹⁾	706 Mbps 80xE1 ⁽¹⁾	983 Mbps 80xE1 ⁽¹⁾
2048QAM L	-	127 Mbps 59xE1	256 Mbps 80xE1 ⁽¹⁾	366 Mbps 80xE1 ⁽¹⁾	517 Mbps 80xE1 ⁽¹⁾	742 Mbps 80xE1 ⁽¹⁾	1033 Mbps 80xE1 ⁽¹⁾
4096QAM	-	-	264 Mbps 80xE1 ⁽¹⁾	378 Mbps 80xE1 ⁽¹⁾	534 Mbps 80xE1 ⁽¹⁾	767 Mbps 80xE1	1068 Mbps 80xE1
4096QAM L	-	-	276 Mbps 80xE1 ⁽¹⁾	395 Mbps 80xE1 ⁽¹⁾	559 Mbps 80xE1 ⁽¹⁾	-	-
8192QAM	-	-	-	-	-	-	-

(1) Supports XPIC

Modulation	CS 7 Frame ID 356	CS 13.75, 14 Frame ID 357	CS 27.5, 28 Frame ID 358	CS 40 Frame ID 359	CS 55, 56, 62.5 Frame ID 360	CS 80 Frame ID 403	CS 110, 112, 125 Frame ID 361	CS 220, 224 ⁽¹⁾ (2) Frame ID 404
4QAM S	8 Mbps	19 Mbps	37 Mbps	53 Mbps	75 Mbps	108 Mbps	151 Mbps	302 Mbps
	2XE1 ⁽³⁾	6xE1 ⁽³⁾	15xE1 ⁽³⁾	23xE1 ⁽³⁾	34xE1 ⁽³⁾	49xE1 ⁽³⁾	70xE1 ⁽³⁾	70xE1
4QAM	10 Mbps	22 Mbps	44 Mbps	62 Mbps	88 Mbps	126 Mbps	176 Mbps	352 Mbps
	3XE1 ⁽³⁾	8xE1 ⁽³⁾	18xE1 ⁽³⁾	28xE1 ⁽³⁾	40xE1 ⁽³⁾	58xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1
16QAM S	18 Mbps	37 Mbps	75 Mbps	107 Mbps	151 Mbps	217 Mbps	302 Mbps	604 Mbps
	6xE1 ⁽³⁾	15xE1 ⁽³⁾	34xE1 ⁽³⁾	49xE1 ⁽³⁾	70xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1
16QAM	21 Mbps	43 Mbps	87 Mbps	124 Mbps	176 Mbps	253 Mbps	352 Mbps	704 Mbps
	7xE1 ⁽³⁾	18xE1 ⁽³⁾	33xE1 ⁽³⁾	57xE1 ⁽³⁾	71xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1
32QAM	26 Mbps	54 Mbps	109 Mbps	156 Mbps	221 Mbps	317 Mbps	441 Mbps	882 Mbps
	10xE1 ⁽³⁾	24xE1 ⁽³⁾	50xE1 ⁽³⁾	73xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1
64QAM	33 Mbps	68 Mbps	137 Mbps	196 Mbps	277 Mbps	397 Mbps	553 Mbps	1106 Mbps
	13xE1 ⁽³⁾	30xE1 ⁽³⁾	63xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1
128QAM	39 Mbps	81 Mbps	162 Mbps	231 Mbps	327 Mbps	470 Mbps	654 Mbps	1308 Mbps
	16xE1 ⁽³⁾	36xE1 ⁽³⁾	75xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1
256QAM	45 Mbps	93 Mbps	186 Mbps	267 Mbps	377 Mbps	542 Mbps	754 Mbps	1508 Mbps
	19xE1 ⁽³⁾	42xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1
512QAM	51 Mbps	105 Mbps	211 Mbps	302 Mbps	427 Mbps	614 Mbps	855 Mbps	1710 Mbps
	22xE1 ⁽³⁾	48xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1
1024QAM	56 Mbps	117 Mbps	234 Mbps	335 Mbps	473 Mbps	679 Mbps	946 Mbps	1892 Mbps
	25xE1	54xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1 ⁽³⁾	80xE1
2048QAM	-	121 Mbps 56xE1	243 Mbps 80xE1 ⁽³⁾	348 Mbps 80xE1 ⁽³⁾	492 Mbps 80xE1 ⁽³⁾	706 Mbps 80xE1 ⁽³⁾	983 Mbps 80xE1 ⁽³⁾	
2048QAM L	-	127 Mbps 59xE1	256 Mbps 80xE1 ⁽³⁾	366 Mbps 80xE1 ⁽³⁾	517 Mbps 80xE1 ⁽³⁾	742 Mbps 80xE1 ⁽³⁾	1033 Mbps 80xE1 ⁽³⁾	
4096QAM	-	-	264 Mbps 80xE1 ⁽³⁾	378 Mbps 80xE1 ⁽³⁾	534 Mbps 80xE1 ⁽³⁾	767 Mbps 80xE1 ⁽³⁾	1068 Mbps 80xE1 ⁽³⁾	
4096QAM L	-	-	276 Mbps 80xE1 ⁽³⁾	395 Mbps 80xE1 ⁽³⁾	559 Mbps 80xE1 ⁽³⁾	803 Mbps 80xE1 ⁽³⁾	-	
8192QAM	-	-	284 Mbps 80xE1	406 Mbps 80xE1	-	-	-	

Table 102 MMU 1004, MMU 1002, PNM 1002, MINI-LINK 6651/4, MINI-LINK 6651/2, and MINI-LINK 6371

(1) Only applicable for MMU 1004

(2) For CS 220 MHz and 224 MHz, Carrier Aggregation with 2x110 MHz or 2x112 MHz is configured.

(3) Supports XPIC

Table 103 MMU 1004, MMU 1002 MIMO

Modulation	CS 27.5, 28 Frame ID 2358	CS 55, 56 Frame ID 2360
4QAM S	36 Mbps 15xE1	73 Mbps 33xE1

Modulation	CS 27.5, 28 Frame ID 2358	CS 55, 56 Frame ID 2360
4QAM	42 Mbps 18xE1	86 Mbps 39xE1
16QAM S	72 Mbps 33xE1	146 Mbps 68xE1
16QAM	85 Mbps 38xE1	171 Mbps 80xE1
32QAM	109 Mbps 50xE1	219 Mbps 80xE1
64QAM	133 Mbps 62xE1	268 Mbps 80xE1
128QAM	157 Mbps 74xE1	317 Mbps 80xE1
256QAM	182 Mbps 80xE1	366 Mbps 80xE1
512QAM	194 Mbps 80xE1	390 Mbps 80xE1
512QAM L	206 Mbps 80xE1	415 Mbps 80xE1
1024QAM	-	435 Mbps 80xE1
1024QAM L	-	459 Mbps 80xE1

Table 104 MMU 1004, MMU 1002 Carrier Aggregation

Modulation	CS	CS	CS	CS	CS
	27.5, 28	40	55, 56	80	110, 112 ⁽¹⁾
	Frame ID 358	Frame ID 359	Frame ID 360	Frame ID 403	Frame ID 361
4QAM S	37 Mbps	53 Mbps	75 Mbps	108 Mbps	151 Mbps
	15xE1	23xE1	34xE1	49xE1	70xE1
4QAM	44 Mbps	62 Mbps	88 Mbps	126 Mbps	176 Mbps
	18xE1	28xE1	40xE1	58xE1	80xE1
16QAM S	75 Mbps	107 Mbps	151 Mbps	217 Mbps	302 Mbps
	34xE1	49xE1	70xE1	80xE1	80xE1
16QAM	87 Mbps	124 Mbps	176 Mbps	253 Mbps	352 Mbps
	33xE1	57xE1	71xE1	80xE1	80xE1
32QAM	109 Mbps	156 Mbps	221 Mbps	317 Mbps	441 Mbps
	50xE1	73xE1	80xE1	80xE1	80xE1
64QAM	137 Mbps	196 Mbps	277 Mbps	397 Mbps	553 Mbps
	63xE1	80xE1	80xE1	80xE1	80xE1
128QAM	162 Mbps	231 Mbps	327 Mbps	470 Mbps	654 Mbps
	75xE1	80xE1	80xE1	80xE1	80xE1
256QAM	186 Mbps	267 Mbps	377 Mbps	542 Mbps	754 Mbps
	80xE1	80xE1	80xE1	80xE1	80xE1
512QAM	211 Mbps	302 Mbps	427 Mbps	614 Mbps	855 Mbps
	80xE1	80xE1	80xE1	80xE1	80xE1
1024QAM	234 Mbps	335 Mbps	473 Mbps	679 Mbps	946 Mbps
	80xE1	80xE1	80xE1	80xE1	80xE1
2048QAM	243 Mbps 80xE1	348 Mbps 80xE1	492 Mbps 80xE1	-	-

Modulation	CS 27.5, 28 Frame ID 358	CS 40 Frame ID 359	CS 55, 56 Frame ID 360	CS 80 Frame ID 403	CS 110, 112 ⁽¹⁾ Frame ID 361
2048QAM L	256 Mbps 80xE1	366 Mbps 80xE1	-	-	-
4096QAM	264 Mbps 80xE1	-	-	-	-
4096QAM L	-	-	-	-	-
8192QAM	-	-	-	-	-

(1) Only applicable for MMU 1004

ANSI

Table 105 MMU 1001, PNM 1001, MINI-LINK 6651/1, MINI-LINK 6651/3 and MINI-LINK 6366

Modulation	CS 10 Frame ID 362	CS 20 Frame ID 363	CS 28 Frame ID 371	CS 30 Frame ID 364	CS 40 Frame ID 365	CS 50 Frame ID 366	CS 56 Frame ID 373	CS 60 Frame ID 367	CS 80 Frame ID 368	CS 112 Frame ID 374
4QAM S	13 Mbps 5xDS1 (1)	26 Mbps 14xDS1 (1)	37 Mbps 21xDS1 (1)	40 Mbps 23xDS1 (1)	54 Mbps 32xDS1 (1)	68 Mbps 40xDS1 (1)	75 Mbps 45xDS1 (1)	81 Mbps 49xDS1 (1)	108 Mbps 66xDS1 (1)	151 Mbps 80xDS1 (1)
4QAM	15 Mbps 7xDS1 (1)	31 Mbps 17xDS1 (1)	44 Mbps 25xDS1 (1)	47 Mbps 27xDS1 (1)	63 Mbps 37xDS1 (1)	79 Mbps 47xDS1 (1)	88 Mbps 54xDS1 (1)	95 Mbps 58xDS1 (1)	126 Mbps 78xDS1 (1)	177 Mbps 80xDS1 (1)
16QAM S	25 Mbps 13xDS1 (1)	53 Mbps 31xDS1 (1)	75 Mbps 45xDS1 (1)	80 Mbps 48xDS1 (1)	108 Mbps 66xDS1 (1)	135 Mbps 80xDS1 (1)	151 Mbps 80xDS1 (1)	162 Mbps 80xDS1 (1)	217 Mbps 80xDS1 (1)	302 Mbps 80xDS1 (1)
16QAM	29 Mbps 16xDS1 (1)	61 Mbps 36xDS1 (1)	87 Mbps 53xDS1 (1)	93 Mbps 57xDS1 (1)	126 Mbps 77xDS1 (1)	158 Mbps 80xDS1 (1)	176 Mbps 80xDS1 (1)	189 Mbps 80xDS1 (1)	253 Mbps 80xDS1 (1)	352 Mbps 80xDS1 (1)
32QAM	37 Mbps 21xDS1 (1)	77 Mbps 46xDS1 (1)	109 Mbps 67xDS1 (1)	117 Mbps 72xDS1 (1)	157 Mbps 80xDS1 (1)	198 Mbps 80xDS1 (1)	221 Mbps 80xDS1 (1)	237 Mbps 80xDS1 (1)	317 Mbps 80xDS1 (1)	441 Mbps 80xDS1 (1)
64QAM	46 Mbps 27xDS1 (1)	97 Mbps 59xDS1 (1)	137 Mbps 80xDS1 (1)	147 Mbps 80xDS1 (1)	197 Mbps 80xDS1 (1)	248 Mbps 80xDS1 (1)	277 Mbps 80xDS1 (1)	297 Mbps 80xDS1 (1)	397 Mbps 80xDS1 (1)	553 Mbps 80xDS1 (1)
128QAM	55 Mbps 32xDS1 (1)	114 Mbps 70xDS1 (1)	162 Mbps 80xDS1 (1)	173 Mbps 80xDS1 (1)	233 Mbps 80xDS1 (1)	293 Mbps 80xDS1 (1)	327 Mbps 80xDS1 (1)	352 Mbps 80xDS1 (1)	470 Mbps 80xDS1 (1)	654 Mbps 80xDS1 (1)
256QAM	63 Mbps 37xDS1 (1)	132 Mbps 80xDS1 (1)	186 Mbps 80xDS1 (1)	200 Mbps 80xDS1 (1)	269 Mbps 80xDS1 (1)	338 Mbps 80xDS1 (1)	377 Mbps 80xDS1 (1)	406 Mbps 80xDS1 (1)	542 Mbps 80xDS1 (1)	754 Mbps 80xDS1 (1)
512QAM	71 Mbps 43xDS1 (1)	149 Mbps 80xDS1 (1)	211 Mbps 80xDS1 (1)	226 Mbps 80xDS1 (1)	305 Mbps 80xDS1 (1)	383 Mbps 80xDS1 (1)	427 Mbps 80xDS1 (1)	460 Mbps 80xDS1 (1)	614 Mbps 80xDS1 (1)	855 Mbps 80xDS1 (1)

Modulation	CS 10 Frame ID 362	CS 20 Frame ID 363	CS 28 Frame ID 371	CS 30 Frame ID 364	CS 40 Frame ID 365	CS 50 Frame ID 366	CS 56 Frame ID 373	CS 60 Frame ID 367	CS 80 Frame ID 368	CS 112 Frame ID 374
1024QAM	79 Mbps 48xDS1	165 Mbps 80xDS1 (1)	234 Mbps 80xDS1 (1)	251 Mbps 80xDS1 (1)	337 Mbps 80xDS1 (1)	423 Mbps 80xDS1 (1)	473 Mbps 80xDS1 (1)	509 Mbps 80xDS1 (1)	679 Mbps 80xDS1 (1)	946 Mbps 80xDS1 (1)
2048QAM	-	172 Mbps 80xDS1 (1)	243 Mbps 80xDS1 (1)	260 Mbps 80xDS1 (1)	351 Mbps 80xDS1 (1)	440 Mbps 80xDS1 (1)	492 Mbps 80xDS1 (1)	529 Mbps 80xDS1 (1)	706 Mbps 80xDS1 (1)	983 Mbps 80xDS1 (1)
2048QAM L	-	180 Mbps 80xDS1 (1)	256 Mbps 80xDS1 (1)	274 Mbps 80xDS1 (1)	369 Mbps 80xDS1 (1)	463 Mbps 80xDS1 (1)	517 Mbps 80xDS1 (1)	556 Mbps 80xDS1 (1)	742 Mbps 80xDS1 (1)	1033 Mbps 80xDS1 (1)
4096QAM	-	-	264 Mbps 80xDS1	283 Mbps 80xDS1 (1)	381 Mbps 80xDS1 (1)	478 Mbps 80xDS1 (1)	534 Mbps 80xDS1 (1)	574 Mbps 80xDS1 (1)	767 Mbps 80xDS1	1068 Mbps 80xDS1
4096QAM L	-	-	276 Mbps 80xDS1	296 Mbps 80xDS1 (1)	399 Mbps 80xDS1 (1)	500 Mbps 80xDS1 (1)	559 Mbps 80xDS1 (1)	601 Mbps 80xDS1 (1)	-	-
8192QAM	-	-	-	-	-	-	-	-	-	-

(1) Supports XPIC

Table 106 MMU 1004, MMU 1002, PNM 1002, MINI-LINK 6651/4, MINI-LINK 6651/2, and MINI-LINK 6371

Modulation	CS 10 Frame ID 362	CS 20 Frame ID 363	CS 28 Frame ID 371	CS 30 Frame ID 364	CS 40 Frame ID 365	CS 50 Frame ID 366	CS 56 Frame ID 373	CS 60 Frame ID 367	CS 80 Frame ID 368	CS 112 Frame ID 374
4QAM S	13 Mbps 5xDS1 ⁽¹⁾	26 Mbps 14xDS1 (1)	37 Mbps 21xDS1 (1)	40 Mbps 23xDS1 (1)	54 Mbps 32xDS1 (1)	68 Mbps 40xDS1 (1)	75 Mbps 45xDS1 (1)	81 Mbps 49xDS1 (1)	108 Mbps 66xDS1 (1)	151 Mbps 80xDS1 (1)
4QAM	15 Mbps 7xDS1 (1)	31 Mbps 17xDS1 (1)	44 Mbps 25xDS1 (1)	47 Mbps 27xDS1 (1)	63 Mbps 37xDS1 (1)	79 Mbps 47xDS1 (1)	88 Mbps 54xDS1 (1)	95 Mbps 58xDS1 (1)	126 Mbps 78xDS1 (1)	177 Mbps 80xDS1 (1)
16QAM S	25 Mbps 13xDS1 (1)	53 Mbps 31xDS1 (1)	75 Mbps 45xDS1 (1)	80 Mbps 48xDS1 (1)	108 Mbps 66xDS1 (1)	135 Mbps 80xDS1 (1)	151 Mbps 80xDS1 (1)	162 Mbps 80xDS1 (1)	217 Mbps 80xDS1 (1)	302 Mbps 80xDS1 (1)
16QAM	29 Mbps 16xDS1 (1)	61 Mbps 36xDS1 (1)	87 Mbps 53xDS1 (1)	93 Mbps 57xDS1 (1)	126 Mbps 77xDS1 (1)	158 Mbps 80xDS1 (1)	176 Mbps 80xDS1 (1)	189 Mbps 80xDS1 (1)	253 Mbps 80xDS1 (1)	352 Mbps 80xDS1 (1)
32QAM	37 Mbps 21xDS1 (1)	77 Mbps 46xDS1 (1)	109 Mbps 67xDS1 (1)	117 Mbps 72xDS1 (1)	157 Mbps 80xDS1 (1)	198 Mbps 80xDS1 (1)	221 Mbps 80xDS1 (1)	237 Mbps 80xDS1 (1)	317 Mbps 80xDS1 (1)	441 Mbps 80xDS1 (1)
64QAM	46 Mbps 27xDS1 (1)	97 Mbps 59xDS1 (1)	137 Mbps 80xDS1 (1)	147 Mbps 80xDS1 (1)	197 Mbps 80xDS1 (1)	248 Mbps 80xDS1 (1)	277 Mbps 80xDS1 (1)	297 Mbps 80xDS1 (1)	397 Mbps 80xDS1 (1)	553 Mbps 80xDS1 (1)

Modulation	CS 10 Frame ID 362	CS 20 Frame ID 363	CS 28 Frame ID 371	CS 30 Frame ID 364	CS 40 Frame ID 365	CS 50 Frame ID 366	CS 56 Frame ID 373	CS 60 Frame ID 367	CS 80 Frame ID 368	CS 112 Frame ID 374
128QAM	55 Mbps 32xDS1 (1)	114 Mbps 70xDS1 (1)	162 Mbps 80xDS1 (1)	173 Mbps 80xDS1 (1)	233 Mbps 80xDS1 (1)	293 Mbps 80xDS1 (1)	327 Mbps 80xDS1 (1)	352 Mbps 80xDS1 (1)	470 Mbps 80xDS1 (1)	654 Mbps 80xDS1 (1)
256QAM	63 Mbps 37xDS1 (1)	132 Mbps 80xDS1 (1)	186 Mbps 80xDS1 (1)	200 Mbps 80xDS1 (1)	269 Mbps 80xDS1 (1)	338 Mbps 80xDS1 (1)	377 Mbps 80xDS1 (1)	406 Mbps 80xDS1 (1)	542 Mbps 80xDS1 (1)	754 Mbps 80xDS1 (1)
512QAM	71 Mbps 43xDS1 (1)	149 Mbps 80xDS1 (1)	211 Mbps 80xDS1 (1)	226 Mbps 80xDS1 (1)	305 Mbps 80xDS1 (1)	383 Mbps 80xDS1 (1)	427 Mbps 80xDS1 (1)	460 Mbps 80xDS1 (1)	614 Mbps 80xDS1 (1)	855 Mbps 80xDS1 (1)
1024QAM	79 Mbps 48xDS1	165 Mbps 80xDS1 (1)	234 Mbps 80xDS1 (1)	251 Mbps 80xDS1 (1)	337 Mbps 80xDS1 (1)	423 Mbps 80xDS1 (1)	473 Mbps 80xDS1 (1)	509 Mbps 80xDS1 (1)	679 Mbps 80xDS1 (1)	946 Mbps 80xDS1 (1)
2048QAM	-	172 Mbps 80xDS1 (1)	243 Mbps 80xDS1 (1)	260 Mbps 80xDS1 (1)	351 Mbps 80xDS1 (1)	440 Mbps 80xDS1 (1)	492 Mbps 80xDS1 (1)	529 Mbps 80xDS1 (1)	706 Mbps 80xDS1 (1)	983 Mbps 80xDS1 (1)
2048QAM L	-	180 Mbps 80xDS1 (1)	256 Mbps 80xDS1 (1)	274 Mbps 80xDS1 (1)	369 Mbps 80xDS1 (1)	463 Mbps 80xDS1 (1)	517 Mbps 80xDS1 (1)	556 Mbps 80xDS1 (1)	742 Mbps 80xDS1 (1)	1033 Mbps 80xDS1 (1)
4096QAM	-	-	264 Mbps 80xDS1	283 Mbps 80xDS1 (1)	381 Mbps 80xDS1 (1)	478 Mbps 80xDS1 (1)	534 Mbps 80xDS1 (1)	574 Mbps 80xDS1 (1)	767 Mbps 80xDS1 (1)	1068 Mbps 80xDS1 (1)
4096QAM L	-	-	276 Mbps 80xDS1	296 Mbps 80xDS1 (1)	399 Mbps 80xDS1 (1)	500 Mbps 80xDS1 (1)	559 Mbps 80xDS1 (1)	601 Mbps 80xDS1 (1)	803 Mbps 80xDS1 (1)	-
8192QAM	-	-	-	304 Mbps 80xDS1	409 Mbps 80xDS1	-	-	-	-	-

(1) Supports XPIC

Table 107 MMU 1004, MMU 1002 MIMO

Modulation	CS 30 Frame ID 2364	CS 40 Frame ID 2365	
4QAM S	39 Mbps 22xDS1	52 Mbps 31xDS1	
4QAM	46 Mbps 26xDS1	61 Mbps 36xDS1	
16QAM S	78 Mbps 47xDS1	105 Mbps 64xDS1	
16QAM	91 Mbps 55xDS1	122 Mbps 75xDS1	
32QAM	117 Mbps 72xDS1	157 Mbps 80xDS1	

Modulation	CS 30 Frame ID 2364	CS 40 Frame ID 2365	
64QAM	143 Mbps 80xDS1	192 Mbps 80xDS1	
128QAM	169 Mbps 80xDS1	227 Mbps 80xDS1	
256QAM	195 Mbps 80xDS1	262 Mbps 80xDS1	
512QAM	208 Mbps 80xDS1	279 Mbps 80xDS1	
512QAM L	221 Mbps 80xDS1	297 Mbps 80xDS1	

Table 108 MMU 1004, MMU 1002, Carrier Aggregation

Modulation	CS	CS	CS	CS	CS
	30	40	50	60 ⁽¹⁾	80 ⁽¹⁾
	Frame ID 364	Frame ID 365	Frame ID 366	Frame ID 367	Frame ID 368
4QAM S	40 Mbps	54 Mbps	68 Mbps	81 Mbps	108Mbps
	23xDS1	32xDS1	40xDS1	49xDS1	66xDS1
4QAM	47 Mbps	63 Mbps	79 Mbps	95 Mbps	126Mbps
	27xDS1	37xDS1	47xDS1	58xDS1	78xDS1
16QAM S	80 Mbps	108 Mbps	135 Mbps	162Mbps	217Mbps
	48xDS1	66xDS1	80xDS1	80xDS1	80xDS1
16QAM	93 Mbps	126 Mbps	158 Mbps	189Mbps	253Mbps
	57xDS1	77xDS1	80xDS1	80xDS1	80xDS1
32QAM	117 Mbps	157 Mbps	198 Mbps	237Mbps	317Mbps
	72xDS1	80xDS1	80xDS1	80xDS1	80xDS1
64QAM	147 Mbps	197 Mbps	248 Mbps	297Mbps	397Mbps
	80xDS1	80xDS1	80xDS1	80xDS1	80xDS1
128QAM	173 Mbps	233 Mbps	293 Mbps	352Mbps	470Mbps
	80xDS1	80xDS1	80xDS1	80xDS1	80xDS1
256QAM	200 Mbps	269 Mbps	338 Mbps	406Mbps	542Mbps
	80xDS1	80xDS1	80xDS1	80xDS1	80xDS1
512QAM	226 Mbps	305 Mbps	383 Mbps	460Mbps	614Mbps
	80xDS1	80xDS1	80xDS1	80xDS1	80xDS1
1024QAM	251 Mbps	337 Mbps	423 Mbps	509Mbps	679Mbps
	80xDS1	80xDS1	80xDS1	80xDS1	80xDS1
2048QAM	260 Mbps 80xDS1	351 Mbps 80xDS1	440 Mbps 80xDS1	529Mbps 80xDS1	-
2048QAM L	274 Mbps 80xDS1	369 Mbps 80xDS1	-	-	-
4096QAM	283 Mbps 80xDS1	-	-	-	-
4096QAM L	-	-	-	-	-
8192QAM	-	-	-	-	-

(1) Only applicable for MMU 1004

 \geq

4 Physical Interfaces

4.1 Ethernet Traffic Interfaces

4.1.1 Interface Types

10BASE-T IEEE802.3i ncl.. In 802.3-2005 IEEE802.3u ncl.. In 802.3-2005 100BASE-TX IEEE802.3ab incl. In 802.3-2005 1000BASE-T 1000BASE-SX IEEE 802.3z incl. In 802.3-2005 1000BASE-LX IEEE 802.3z incl. In 802.3-2005 IEEE 802.3z incl. In 802.3-2005 1000BASE-ZX 1000BASE-BX10 IEEE 802.3(58) incl. In 802.3-2005 1000BASE-X CWDM (1470-1610 nm) 10GBASE-LR IEEE 802.3ae (49) 10GBASE-ER IEEE 802.3ae (49) 10GBASE-ZR Single Mode 1550 nm

4.1.2 Interfaces per Application

Table 109 Interfaces per Application

Application	Connector type	Rate (Mbit/s)	# of ports
NPU 1002	Shielded RJ45 jack	10/100/1000	2
	SFP/SFP+	See Ethernet SFP on page 162.	3 (max 2 SFP+)
NPU 1003	Shielded RJ45 jack	10/100/1000	2
	SFP/SFP+	See Ethernet SFP on page 162.	3 (max 2 SFP+)
NPU 1005	Shielded RJ45 jack	10/100/1000	2
	SFP/SFP+	See Ethernet SFP on page 162.	3 (max 2 SFP+)
ETU 1001	Shielded RJ45 jack	10/100/1000	Up to 4 ⁽¹⁾
Application	Connector type	Rate (Mbit/s)	# of ports
----------------	-----------------------	-------------------------------	------------------------
ETU 1002	SFP	1000	Up to 4 ⁽²⁾
ETO 1002	SFP+	10G	1 ⁽³⁾
DNM 1001	Shielded RJ45 jack	10/100/1000	2
	SFP/SFP+ (≤ 2.5G)	See Ethernet SFP on page 162.	4 (max 2 SFP+)
DNM 1002	Shielded RJ45 jack	10/100/1000	2
PNM 1002	SFP/SFP+ (≤ 2.5G)	See Ethernet SFP on page 162.	4 (max 2 SFP+)
MINI-LINK	Shielded RJ45 jack	10/100/1000	2
6651/1	SFP/SFP+ (≤ 2.5G)	See Ethernet SFP on page 162.	4
MINI-LINK	Shielded RJ45 jack	10/100/1000	2
6651/2	SFP/SFP+ (≤ 2.5G)	See Ethernet SFP on page 162.	3
MINI-LINK	Shielded RJ45 jack	10/100/1000	2
6651/3	SFP/SFP+ (≤ 2.5G)	See Ethernet SFP on page 162.	4
MINI-LINK	Shielded RJ45 jack	10/100/1000	2
6651/4	SFP/SFP+	See Ethernet SFP on page 162.	4 (max 4 SFP+)
MINI-LINK 6366	Shielded RJ45 jack	10/100/1000	2
	SFP/SFP+ (≤ 2.5G)	See Ethernet SFP on page 162.	2
	Shielded RJ45 jack	10/100/1000	1
	SFP/SFP+	See Ethernet SFP on page 162.	3 (max 3 SFP+)

(1) Number of usable ports depends on the slot used, and the NPU type. .

(2) The ports can be configured for up to $4 \times 1G$ or $1 \times 10G$ Base-X, but not at the same time.

(3) 10G only available when the ETU 1002 is installed in specific slots. .

All interfaces support Auto negotiation and set-up.

4.1.3 Ethernet SFP

The "SFP" (Small Form-factor Pluggable) enables the customer to choose between optical and electrical interfaces. The SFPs are not locked to a certain brand.

Note: Non-Ericsson SFPs will be activated by the node if it is possible, but Ericsson can then not guarantee interoperability between the node and the SFP.

Interface	Capability		
Electrical, shielded RJ45 jack	10/100/1000BASE-T		
Optical, LC connector	1000BASE-SX Multi mode 850 nm		
Optical, LC connector	1000BASE-LX Single mode 1310 nm		
Optical, LC connector	1000BASE-LX Single/Multi mode 1310 nm		
Optical, LC connector	1000BASE-ZX Single mode 1550 nm		
Optical, LC connector	1000BASE-BX10-U Single mode Tx 1310 Rx 1490 nm		
Optical, LC connector	1000BASE-BX10-D Single mode Tx 1490 Rx 1310 nm		
Optical, LC connector	1000BASE-BX40-U Single mode Tx 1310 Rx 1490 nm		
Optical, LC connector	1000BASE-BX40-D Single mode Tx 1490 Rx 1310 nm		
Optical, LC connector	10GBASE-LR Single mode 1310nm		
Optical, LC connector	1000BASE-X CWDM Single mode 1470-1610 nm		
Optical, LC connector	10GBASE-ER Single Mode 1550 nm		
Optical, LC connector	10GBASE-ZR Single Mode 1550 nm		
Optical, LC connector	10GBASE-X DWDM Single mode 1528-1566 nm		
Optical, LC connector	10GBASE-BX20-U Single mode Tx 1270 Rx 1330 nm		

Table 110 Ethernet SFP Interface Capability

162

Interface	Capability
Optical, LC connector	10GBASE-BX20-D Single mode Tx 1330 Rx 1270 nm
Optical, LC connector	10GBASE-BX30-U Single mode Tx 1270 Rx 1330 nm
Optical, LC connector	10GBASE-BX30-D Single mode Tx 1330 Rx 1270 nm

4.1.3.1 Ethernet DAC Cable

The DAC (Direct Attach Copper) cable enables the customer to connect two collocated nodes at 10Gbps rate. DAC is a cost-effective alternative for Ethernet SFP+ modules and cables. The DACs are not locked to a certain brand.

Note: Non-Ericsson DACs will be activated by the node if it is possible but Ericsson can then not guarantee interoperability between the nodes.

Table 111Connector Type Capacity

Connector type	Rate	
SFP+	10Gbps	

4.2 PDH Interfaces

4.2.1 ETSI

E1 electrical interface is according to G.703 for 75 Ω and 120 $\Omega.$ (See ETSI on page 164.)

Output pulse mask in resistive load is according to ITU-T Rec. G.703.

4.2.2 ANSI

DS1 electrical interface is according to T1.403-1999 with line build out for 0-655 ft.

Output mask complies with T1.403-1999 Figure 2.

Line code is AMI/B8Z5.

4.3 SDH Interfaces

4.3.1 SDH SFP

The SFP (Small Form-factor Pluggable) enables the customer to choose between optical and electrical interfaces. The SFPs are not locked to a certain brand. Non-Ericsson SFPs will be activated by the node if it is possible, but Ericsson can then not guarantee interoperability between the node and the SFP.

The SDH SFP will be used as STM-1.

Table 112 SDH SFP Interface Capability

Interface	Capability		
Electrical, DIN 1.0/2.3 75 Ohm, female	S-1.E		
Optical, LC connector	S-1.1 1310nm		
Optical, LC connector	L-1.1 1310nm		
Optical, LC connector	L-1.2 1550nm		
Optical, LC connector	BiDi 1310/1550nm		
Optical, LC connector	BiDi 1550/1310nm		
Optical, LC connector	L-1/4/16.2C CWDM 1470-1610nm		

4.4 PDH Traffic Connectors

4.4.1 ETSI

For NPU 1003, PNM 1001, PNM 1002, LTU 1001 and LTU 1002, the E1 connector is a male SOFIX 24 pin connector with $4xE1 120 \Omega$ balanced and 75 Ω unbalanced short haul (6 dB) terminations. The impedance is selectable between 75 Ω and 120 Ω .

For MINI-LINK 6651, the E1 connector on is an RJ45 connector. Only 120 Ω impedance is supported.

4.4.2 ANSI

For NPU 1003, PNM 1001, PNM 1002, LTU 1001 and LTU 1002, the DS1 connector is a male SOFIX 24-pin connector, with 4xDS1 terminations.



4.5 Local O&M

4.5.1 Site LAN Port

One of the fixed Ethernet ports on NPU 1002, NPU 1003, NPU 1005, PNM 1001, PNM 1002, MINI-LINK 6651, MINI-LINK 6366, and MINI-LINK 6371 can be configured as a site LAN port.

- The Ethernet connection is a shielded 8 pin RJ45 modular jack (DTE).
- The interface supports 10/100/1000 Mbit/s full duplex Ethernet with autonegotiate
- The green LED on NPUs shows "Link up" connectivity.

4.5.2 USB Interface

There is a mini USB device port on NPU 1002, NPU 1003, NPU 1005, PNM 1001, PNM 1002, MINI-LINK 6651, MINI-LINK 6366, and MINI-LINK 6371.

- The interface is a 5 pin USB2 connector. (Mini-B USB connector)
- The bit rate is up to 480 Mbit

4.5.3 User I/O Interface

The User I/O interface includes a number of output and input signals.

- The output signals can be used to export alarms from the MINI-LINK 6600 to other equipment's supervision systems.
- The input signals can be used to transfer alarm and status information from on-site equipment to central management systems.

Board Type	# of input signals	# of output signals	Connector Type	
NPU 1002	6	3	Male 24 pin SOFIX connector	
NPU 1003	2	1	Male 24 pin SOFIX connector	
NPU 1005	6	3	Male 24 pin SOFIX connector	
PNM 1001	2	1	Male 24 pin SOFIX connector	

Table 113 Plug-in Units with User I/O Interface

Board Type	# of input signals	# of output signals	Connector Type	
PNM 1002	2	1	Male 24 pin SOFIX connector	

4.5.3.1 Input Signals

The input signals are opto-coupled and have a reference that is floating with respect to each other and station ground. All signals are positive to the reference and shall read as follows:

- logical "0" when the Voltage < 1.0 V DC
- logical "1" when the Voltage > 2.4 V DC

These signals can be connected directly to a "Normaly Open" (NO) or Normally Closed (NC) equipment like a relay or a mechanical switch, since they have week internal drivers (2.5 Volt driver with ~4 Kohm serial resistors).

An input Voltage less than 15 V DC will not damage the equipment.

4.5.3.2 Output Signals

The output signals are performed with relay settings.

The output signals consist of three pins:

- NCL (Normally Closed)
- RTN (ReTurN)
- NOP (Normally Open)

At start-up the default position is the following:

- NCL-RTN is "normally closed" (impedance < 1Ω)
- NOP-RTN is "normally open" (impedance > $1 M\Omega$)

The relays used to switch the settings have the contact rating:

- 100 V DC, 1 A
- 125 V AC, 1 A

The outputs are reset to default temporarily upon cold restart but are not affected by a warm restart of the board.



4.5.3.3 LED Indications

For Information about functionality and behavior please see CPI document, 35/1551-HRA 901 20/11 LED Descriptions.

4.6 Antenna Interface

Table 114	Antenna Interface
-----------	-------------------

Frequency band(s)	Waveguide interface, flange types
5 GHz	154 IEC-UDR 48
6 GHz Lower and Upper	154 IEC-UDR 70
7, 8, 7/8 GHz	154 IEC-UBR 84
10, 11, 10/11, 13 GHz	154 IEC-UBR 120
15 GHz	154 IEC-UBR 140
18, 23 GHz	154 IEC-UBR 220
24, 26, 28 GHz	154 IEC-UBR 260
32, 38 GHz	154 IEC-UBR 320
42 GHz	Ericsson proprietary flange (waveguide R500)
80 GHz	Integrated installation only (waveguide R740)

4.7 Indoor-Outdoor Interconnection

RAU connector type: N

MMU connector type: TNC

ETSI

Maximum cable length for performance as specified above:

	RAU2 X	MINI-LINK 6363			MINI-LINK 6364 MINI-LINK 6365	
Radio cable diameter	CS 7-56 MHz	CS 7-56 MHz	CS 80-112 MHz	Single Carrier CS 7-112 MHz	Carrier Aggregation CS 2x 28-56 MHz	Carrier Aggregation CS 2x 80-112 MHz
TZC 500 97 (7.6 mm)	100 m	100 m	85 m	100 m	100 m	85 m
TZC 500 32 (10 mm)	200 m	200 m	160 m	200 m	200 m	160 m
TZC 501 26 (16 mm)	400 m	400 m	300 m	400 m	400 m	350 m

Table 115 ETSI Maximum Cable Length

ANSI

Maximum cable length for performance as specified above:

	RAU2 X	MINI-LINK 6363			MINI-LINK 6365	
Radio cable diameter	CS 10-56 MHz	CS 10-60 MHz	CS 80-112 MHz	Single Carrier CS 10-112 MHz	Carrier Aggregation CS 2x 28-56 MHz	Carrier Aggregation CS 2x 60-112 MHz
TZC 500 95	656 ft	656 ft	524 ft	656 ft	656 ft	524 ft
(3/8 in)	(200 m)	(200 m)	(160 m)	(200 m)	(200 m)	(160 m)
TZC 500 80	1312 ft	1312 ft	984 ft	1312 ft	1312 ft	1148 ft
(1/2 in)	(400 m)	(400 m)	(300 m)	(400 m)	(400 m)	(350 m)

5 Power

5.1 Power Supply

MINI-LINK 6691 has two DC inputs located on the PFU 1101. MINI-LINK 6692 has one DC input located on each PFU 1201. MINI-LINK 6693 has two DC inputs located on the PFU 1301. MINI-LINK 6694 has one DC input located on each PFU 1601. MINI-LINK 6654 has two DC inputs located on PNM 1001/1002. MINI-LINK 6655 has two DC inputs located on PNM 1002. MINI-LINK 6651 has two DC inputs. MINI-LINK 6366 has one DC input. MINI-LINK 6371 has one DC input.

5.1.1 DC Supply Interface

- The power is floating, i.e. not connected to station ground.
- The PFU provides surge protection.

5.1.2 DC Supply Voltage

Service Voltage Ranges

Table 117 Service Voltage Ranges

Nominal Voltage	-48 VDC				
Product	6691, 6692, 6693 and 6694	6654 and 6655	6651	6366 and 6371	
Normal service voltage range	-58.8 V to -40 V	58.8 V to -40 V	-58.8 V to -40 V	-58.8 V to -38 V	
Abnormal service voltage range	-60 V to -58.8 V	-60 V to -58.8 V	-60 V to -58.8 V	-60 V to -58.8 V	
	-40 V to 0 V	-40 V to 0 V	-40 V to 0 V	-38 V to 0 V	
Typical startup voltage	-39 V to -43 V	-41 V to -43 V	-44 V to -46 V	-44 V to -46 V	
Typical shutdown voltage	-33.5 V to -37.5 V	-33.5 V to -37.5 V	-37.5 V to -39.5 V	-35.5 V to -37.5 V	

Normal Service Voltage Range

The system is able to run at full operation in this range.

Abnormal Service Voltage Range

No function is guaranteed, but it will not result in any permanent damage, when operated in this range.

The system will restore to full operation when returning to normal service voltage range.



5.1.2.1 Clarification of Input Voltage Behavior for MINI-LINK 6691, 6692, 6693 and 6694

Figure 4 Input Voltage Behavior for MINI-LINK 6691, 6692, 6693 and 6694

Figure 4 shows the Input voltage behavior for MINI-LINK 6691, 6692, 6693 and 6694. The different units within MINI-LINK 6691, 6692, 6693 and 6694 will be turned on and off at different voltages.

There is a hysteresis for all units to prevent power flapping (turning on/off repeatedly around a threshold).

5.1.2.2 Clarification of Input Voltage Behavior for MINI-LINK 6654 and 6655



Figure 5 Input Voltage Behavior for MINI-LINK 6654 and MINI-LINK 6655

Figure 5 shows the Input voltage behavior for MINI-LINK 6654 and 6655. PNM and APU's will be turned on and off at different voltages.

There is a hysteresis for all units to prevent power flapping (turning on/off repeatedly around a threshold).

5.1.3 Power Supply Protection

PFU 1101, PFU 1301, PNM 1001, PNM 1002, and MINI-LINK 6651 have an external redundant power supply.

The same is applicable for PFU 1201 and PFU 1601 but these PFUs also support a redundant PFU solution.

If the input voltages to the PFU are lost, an alarm is sent.

MINI-LINK 6366 and MINI-LINK 6371 does not support redundant power supply.

5.1.4 Soft Start (all Plug-in Units)

The soft start function limits inrush current at start up or in case of a short circuit accordingly to EN 300 132-2.

5.1.5 Power Connector

Table 118 Power Connector

PFU 1101	Male 2-pin bus bar
PFU 1201	Male 3-pin bus bar
PFU 1301	Male 2-pin bus bar
PFU 1601	Male 2-pin bus bar
PNM 1001	Male 2-pin bus bar
PNM 1002	Male 2-pin bus bar
MINI-LINK 6651	Male 2-pin bus bar
MINI-LINK 6366/1	2-pole power terminal with connection for shield
MINI-LINK 6366/4	Male 2-pin with shield, Octis type
MINI-LINK 6371	Male 2-pin with shield, Octis type

5.2 Power Line Disturbance

MINI-LINK 6691, 6692, 6693, 6694, PNM 1001, and PNM 1002 support Power Line Disturbance according to ATIS-0600315.2007 §5.5 (10 ms Fuse Blowing Transient) without restart.

APU, Application Plug-in Units are dimensioned to handle up to 0.1 ms transients.

5.3 Power Consumption

Power consumptions to be used for dimensioning of site power supply are 33% higher than the typical values. For OPEX calculation regarding power consumption the typical values give the best estimate.

5.3.1 Maximum Power Level for Fuse and Power Dimensioning

The supply voltage is specified in DC Supply Voltage on page 169.

The figures below are to be used for fuse and power dimensioning only.

For MINI-LINK 6200 installations other fuse values might be applicable. See 1/1301-HRA 901 20/14 MINI-LINK 6200 Product Specification.

Fuse Type Requirements

DC voltage of minimum 80 V, a breaking capacity of 5000 A or more, and one of the following types:

- Thermo-electromagnetic over-current release certified according to IEC 60947-2, according to curve C (IEC 60898).
- Hydraulic-magnetic over-current release according to Eaton/Heinemann curve 3 or equivalent.

Circuit Breaker and Cable Dimensioning

	CB and Cable Dimensioning		
Product	Max CB value	Cable	
MINI-LINK 6691	10 A	RPM 517 560 or NTM 20372 up to 20m	
MINI-LINK 6692	40 A	RPM 777 551 up to 10m	
MINI-LINK 6693	16 A	RPM 517 560 or NTM 20372 up to 10m	
MINI-LINK 6694	20 A	RPM 517 560 or NTM 20372 up to 10m	
MINI-LINK 6654	16 A	RPM 517 560 or NTM 20372 up to 10m	
MINI-LINK 6655	20 A	RPM 517 560 or NTM 20372 up to 10m	
MINI-LINK 6651	10 A	RPM 517 560 or NTM 20372 up to 20m	
MINI-LINK 6366	See Fuse and Power Dimensioning for Outdoor Products on page 173.		
MINI-LINK 6371	See Fuse and Power Dimensioning for Outdoor Products on page 173.		

Table 119 Circuit Breaker (CB) and Cable Dimensioning

5.3.1.1 Fuse and Power Dimensioning for Outdoor Products

Cable to be used:

— TFL 252 504/2 (2x4 mm²)

 \geq

NTM 203 72 (2x2.5 mm²)

Table 120Recommended Hydraulic Magnetic Circuit Breaker ValuesThe following circuit breaker values are applicable to TFL 252 504/2 (2x4 mm²).

Power Bating of	Length							
Product	10 m	20 m	50 m	70 m	100 m	120 m	150 m	180 m
50 W	4 A	4 A	4 A	4 A	4 A	4 A	4 A	Х
75 W	4 A	4 A	4 A	4 A	4 A	4 A	4 A ⁽¹⁾	Х
100 W	4 A	4 A	4 A	4 A	4 A ⁽¹⁾	4 A ⁽¹⁾	4 A ⁽¹⁾	Х
125 W	6 A	6 A	6 A	6 A	6 A ⁽¹⁾	Х	Х	Х
150 W	6 A	6 A	6 A	6 A ⁽¹⁾	6 A ⁽¹⁾	Х	Х	Х
200 W	10 A	10 A	10 A ⁽¹⁾	Х	Х	Х	Х	Х

(1) Acceptable from safety point of view but will lead to a higher voltage drop.

 Table 121
 Recommended Hydraulic Magnetic Circuit Breaker Values

The following circuit breaker values are applicable to NTM 203 72 ($2x2.5 \text{ mm}^2$).

Power Bating of	Length					
Product	10 m	20 m	50 m	70 m	100 m	120 m
50 W	4 A	4 A	4 A	4 A ⁽¹⁾	4 A ⁽¹⁾	Х
75 W	4A	4 A	4 A	4 A ⁽¹⁾	4 A ⁽¹⁾	Х
100 W	4 A	4 A	4 A	4 A ⁽¹⁾	4 A ⁽¹⁾	Х
125 W	6 A	6 A	6 A ⁽¹⁾	Х	Х	Х
150 W	6 A	6 A	6 A ⁽¹⁾	Х	Х	Х
200 W	10 A	10 A	Х	Х	Х	Х

(1) Acceptable from safety point of view but will lead to a higher voltage drop.

Table 122 Recommended Thermoelectric Magnetic Circuit Breaker

Power	Length					
product	10 m	20 m	50 m	70 m	100 m	120 m
50 W	4 A	4 A	4 A	4 A ⁽¹⁾	4 A ⁽¹⁾	Х
75 W	4 A	4 A	4 A	4 A ⁽¹⁾	4 A ⁽¹⁾	Х
100 W	6 A	6 A	6 A	Х	Х	Х
125 W	6 A	6 A	6A ⁽¹⁾	Х	Х	Х
150 W	10 A	10 A	Х	Х	Х	Х
200 W	10 A	10 A	Х	Х	Х	Х

The following circuit breaker values are applicable to NTM 203 72 (2x2.5 mm²).

(1) Acceptable from safety point of view but will lead to a higher voltage drop.

5.3.2 Power Consumption and Heat Dissipation

Preconditions: 25 °C ambient temperature, -48V DC supply

Table 123	Typical Power	Consumption and	Heat Dissipation
-----------	---------------	-----------------	------------------

Unit	Typical Power Consumption and Heat Dissipation
NPU 1002 ^{(1) (2)}	30 W
NPU 1003 ^{(1) (2)}	23 W
NPU 1005 ^{(1) (2)}	41 W
PNM 1001 ^{(1) (2) (3)}	44 W
PNM 1002 ^{(2) (3)}	48 W
PFU 1101 ^{(3) (4)}	2 W (4 W)
PFU 1201 ^{(3) (5)}	2 W (8 W)
PFU 1301 ^{(3) (6)}	2 W (5 W)
PFU 1601 ^{(3) (6)}	2 W (5 W)
FAU 1101 ⁽⁷⁾	6 W (20 W)
FAU 1201 ⁽⁷⁾	8 W (15 W)
FAU 1301 ⁽⁷⁾	2.5 W (10.5 W)
FAU 1401 ⁽⁷⁾	1.7 W (12 W)
FAU 1501 ⁽⁷⁾	2.5 W (10.5 W)

Unit	Typical Power Consumption and Heat Dissipation
FAU 1601 ⁽⁷⁾	2.5 W (10.5 W)
LTU 1001	5 W
LTU 1002 ⁽¹⁾	20 W
ETU 1001	7 W
ETU 1002 ⁽¹⁾	18 W
MINI-LINK 6651/1 ^{(1) (2) (7)}	40 W (57 W)
MINI-LINK 6651/2 ^{(1) (2) (7)}	40 W (57 W)
MINI-LINK 6651/3 ^{(1) (2)}	27 W
MINI-LINK 6651/4 ^{(1) (2) (7)}	63 W (80 W)
MINI-LINK 6366 ^{(1) (2)}	37 W
MINI-LINK 6371 ^{(1) (2)}	63 W

(1) Each SFP adds 1 W and each SFP+ adds 1.5 W to the consumption value.

(2) Additional power consumption per connected radio unit. See Typical MMU Power Consumption and Indoor Heat Dissipation on page 176. The specified value is applicable when no radio units are connected.

- (3) FAU DC loss included.
- (4) Enclosure equipped with 1 MMU 1002 or MMU 1004 and 2 radio units consuming 42 W each. Number within brackets represents PFU power consumption for enclosure equipped with 2 MMU 1002 or MMU 1004 and 4 radio units consuming 42 W each. Value within brackets includes fan-unit (FAU) running at full speed.
- (5) Enclosure equipped with 2 MMU 1002 or MMU 1004 and 4 radio units consuming 42W each. Number within brackets represents PFU power consumption for enclosure equipped with 8 MMU 1002 or MMU 1004 and 16 radio units consuming 42 W each. Value within brackets includes fan-unit (FAU) running at full speed.
- (6) Enclosure equipped with 2 MMU 1002 or MMU 1004 and 4 radio units consuming 42W each. Number within brackets represents PFU power consumption for enclosure equipped with 4 MMU 1002 or MMU 1004 and 8 radio units consuming 42 W each. Value within brackets includes fan-unit (FAU) running at full speed.
- (7) Nominal value at room temperature. Value within brackets is when the fan-unit (FAU) is running at full speed (high temperature).

5.3.3 Radio Terminal Power Consumption

5.3.3.1 Typical MMU Power Consumption and Indoor Heat Dissipation

Preconditions: 25 °C ambient temperature, -48V DC supply

Unit	Typical Power Consumption and Heat Dissipation
MMU 1001 ⁽¹⁾	21 W
MMU 1002 ⁽¹⁾	25 W
MMU 1004 ⁽¹⁾	25 W
Additional power consumption in MMU 1001, MMU 1002, MMU 1004, PNM 1001, PNM 1002, MINI-LINK 6651, MINI-LINK 6366 or MINI-LINK 6371 per connected radio unit	10% of the radio unit's power consumption

 Table 124
 Typical MMU Power Consumption and Indoor Heat Dissipation

(1) Applicable when no radio units are connected.

5.3.3.2 RAU2 X Power Consumption and Outdoor Heat Dissipation

Typical power consumption and heat dissipation (W) at different RAU output power levels.

DALLT	Nominal		Eco Mode	
КАО Туре	cons.	P _{max} ⁽¹⁾	P _{max-10dB}	P _{min}
RAU2 X 5 HP	42	37	23	15
RAU2 X 6L	30	27	24	17
RAU2 X 6L HP	42	37	28	15
RAU2 X 6U	28	25	22	15
RAU2 X 6U HP	42	37	28	15
RAU2 X 7	29	26	23	16
RAU2 X 7 HP	42	37	28	15
RAU2 X 8	30	27	24	16
RAU2 X 8 HP	42	42	26	16
RAU2 X 10	32	32	24	17
RAU2 X 10 HP	38	38	28	18
RAU2 X 11	27	27	20	16
RAU2 X 11 HP	39	39	28	17

Table 125 Typical Power Consumption and Outdoor Heat Dissipation of RAU2 X

DALLT	Nominal		Eco Mode	
RAU Type	cons.	P _{max} ⁽¹⁾	P _{max-10dB}	P _{min}
RAU2 X 13	23	20	19	15
RAU2 X 13 HP	26	22	20	15
RAU2 X 15	24	19	19	15
RAU2 X 15 HP	26	21	20	15
RAU2 X 18	22	20	19	16
RAU2 X 18 HP	27	24	24	16
RAU2 X 23	22	20	19	16
RAU2 X 23 HP	28	25	24	16
RAU2 X 24	25	22	18	16
RAU2 X 26	25	22	18	16
RAU2 X 28	26	23	19	17
RAU2 X 32	24	22	22	15
RAU2 X 38	27	24	24	20
RAU2 X 42	25	23	21	19

(1) Requires RAU2 X HW \geq R6A.

5.3.3.3 MINI-LINK 6363 Power Consumption and Outdoor Heat Dissipation

Typical power consumption and heat dissipation (W) at P_{max} (Including HP License), $P_{max-10dB}$ and P_{min}

Table 126 Typical Power Consumption and Outdoor Heat Dissipation of MINI-LINK 6363

Radio Type	P _{max}	P _{max-10dB}	P _{min}
MINI-LINK 6363 6L	33	27	17
MINI-LINK 6363 6U	33	27	17
MINI-LINK 6363 7	32	26	16
MINI-LINK 6363 8	32	26	16
MINI-LINK 6363 10	31	22	19
MINI-LINK 6363 11	31	24	19

Radio Type	P _{max}	P _{max-10dB}	P _{min}
MINI-LINK 6363 13	23	18	15
MINI-LINK 6363 15	25	20	16
MINI-LINK 6363 18	23	21	19
MINI-LINK 6363 23	23	21	19
MINI-LINK 6363 24	24	21	19
MINI-LINK 6363 26	24	21	19
MINI-LINK 6363 28	23	20	19
MINI-LINK 6363 32	23	20	19
MINI-LINK 6363 38	23	22	20
MINI-LINK 6363 42	23	22	20
MINI-LINK 6363 80	24	23	19

Typical power consumption and heat dissipation (W) at $\mathsf{P}_{max}, \, \mathsf{P}_{max-10dB}$ and $\mathsf{P}_{min}.$

Table 127 Typical Power Consumption and Outdoor Heat Dissipation of MINI-LINK 6363/2

Radio Type	P _{max}	P _{max-10dB}	P _{min}
MINI-LINK 6363/2 13	16	15	14
MINI-LINK 6363/2 15	16	15	14
MINI-LINK 6363/2 18	14	13	12
MINI-LINK 6363/2 23	14	13	12
MINI-LINK 6363/2 38	20	20	19

5.3.3.4 MINI-LINK 6364 Power Consumption and Outdoor Heat Dissipation

Typical power consumption and heat dissipation (W) at P_{max} (Including HP License), $P_{max-10dB}$ and P_{min} .

Table 128Typical Power Consumption and Outdoor Heat Dissipation ofMINI-LINK 6364

Radio Type	P _{max}	P _{max-10dB}	P _{min}
MINI-LINK 6364 13	26	22	18
MINI-LINK 6364 15	26	22	18

Radio Type	P _{max}	P _{max-10dB}	P _{min}
MINI-LINK 6364 18	25	22	20
MINI-LINK 6364 13 CA mode	26	22	21
MINI-LINK 6364 15 CA mode	26	22	21
MINI-LINK 6364 18 CA mode	27	23	22

5.3.3.5 MINI-LINK 6365 Power Consumption and Outdoor Heat Dissipation

Typical power consumption and heat dissipation (W) at P_{max} (Including HP License), $P_{max-10dB}$ and P_{min}

Table 129 Typical Power Consumption and Outdoor Heat Dissipation of MINI-LINK 6365

Radio Type	P _{max}	P _{max-10dB}	P _{min}
MINI-LINK 6365 6	35	29	19
MINI-LINK 6365 7/8	35	29	19
MINI-LINK 6365 10/11	35	29	19
MINI-LINK 6365 13	26	22	18
MINI-LINK 6365 15	26	22	18
MINI-LINK 6365 18	25	22	20
MINI-LINK 6365 23	25	22	20
MINI-LINK 6365 32	23	20	19
MINI-LINK 6365 6 CA mode	37	31	22
MINI-LINK 6365 7/8 CA mode	35	29	24
MINI-LINK 6365 10/11 CA mode	36	30	21
MINI-LINK 6365 13 CA mode	26	22	21
MINI-LINK 6365 15 CA mode	26	22	21
MINI-LINK 6365 18 CA mode	27	23	22
MINI-LINK 6365 23 CA mode	26	23	22
MINI-LINK 6365 32 CA mode	24	21	21

6	Environmen	t	
6.1	Environmental	Condit	ions
	Continuous Conditions	s (Norma l Environ functior	Operation) mental conditions in which all units are able to a as specified.
	Exceptional Condition	s (Safe Fu Environ operatic perform increase	unction) mental stress outside the limits for normal on in which all units continue to function, but ance or capacity may be reduced, e.g. slightly ed bit error rates.
6.1.1	Indoor Equipment		
	Ambient Temperature	Range:	
	Continuous Conditions	5 (Norma l -5 to +6	Operation) 0° C (+23 to +131° F)
	Exceptional Condition	s (Safe F i -25 to +	u nction – except for MINI-LINK 6651/3) 65° C (-13 to +140° F)
	Exceptional Condition	s (Safe F i -25 to +	u nction – for MINI-LINK 6651/3) 60° C (-13 to +131° F)
	Relative Humidity Ran	ige:	
	Relative Humidity Ran	i ge 5-95%	
6.1.2	Outdoor Equipment		
	Ambient Temperature	Range:	
	Continuous Conditions	s (Norma l -33 to +	Operation) 55° C (-27 to +131° F)
		Note:	Outdoor temperature is measured in an open unshielded area preferably close to the radio. Sensor has to be protected from the sun. Minimum distance to other objects is 50 cm.
	Exceptional Condition	s (Safe F i -45 to +	unction) 60° C (-49 to +140° F)

Note: Outdoor temperature is measured in an open unshielded area preferably close to the radio. Sensor has to be protected from the sun. Minimum distance to other objects is 50 cm.

Relative Humidity Range:

Relative Humidity Range

8-100 %

7 Mechanical Data

7.1 Outdoor Units

7.1.1 Weight (antennas not included)

Table 130 RAU2 X

Fraguancy	Typical		Max ⁽¹⁾	
Frequency	(kg)	(lb)	(kg)	(lb)
5–8 GHz	4.8	10.6	5.5	12.1
10-23 GHz	4.0	8.8	4.5	9.9
24-42 GHz	3.7	8.2	4.0	8.8

(1) Maximum weight varies due to different sub-bands' diplexers.

Table 131 MINI-LINK 6363, MINI-LINK 6363/2, MINI-LINK 6364, and MINI-LINK 6365

Frequency	Typical		Max ⁽¹⁾	
riequency	(kg)	(lb)	(kg)	(lb)
6 – 15 GHz	2.7	6.0	3.2	7.1
18 – 80 GHz	2.3	5.1	2.7	6.0

(1) Maximum weight varies due to different sub-bands' diplexers.

Table 132 MINI-LINK 6366/1 and MINI-LINK 6366/4

	(kg)	(lb)
Without mounting bracket	4.0	8.8
With mounting bracket for integrated mount	5.2	11.5
With two brackets for split-mount	4.2	9.3

Table 133 MINI-LINK 6371

	(kg)	(lb)
Without mounting bracket	5.6	12.2
With mounting bracket for integrated mount	6.7	14.8
With two brackets for split-mount	5.8	12.7
With mounting bracket for rail-mount	6.4	14.1

7.1.2 Nominal Dimensions

Table 134 Outdoor Unit Dimensions

Draduat	DxWxH		
	(mm)	(inch)	
RAU2 X	97 x 260 x 321	3 3/4 x 10 1/4 x 12 1/2	
MINI-LINK 6363, MINI-LINK 6363/2, MINI-LINK 6364, and MINI-LINK 6365	79 x 197 x 179	3 7/64 x 7 3/4 x 7 3/64	
MINI-LINK 6366/1 without mounting bracket	43 x 462 x 303	1 11/16 x 20 17/64 x 11 59/64	
MINI-LINK 6366/1 with mounting bracket for integrated mount	158 x 462 x 303	6 7/32 x 20 17/64 x 11 59/64	
MINI-LINK 6366/4 without mounting bracket	70 x 310 x 291	2 3/4 x 12 1/5 x 11 5/11	
MINI-LINK 6366/4 with mounting bracket for integrated mount	188 x 335 x 291	7 2/5 x 13 1/5 x 11 5/11	
MINI-LINK 6371 without mounting bracket	86 x 310 x 294	3 25/64 x 12 1/5 x 11 37/64	
MINI-LINK 6371 with mounting bracket for integrated mount	204 x 335 x 294	8 1/32 x 13 1/5 x 11 37/64	

7.2 Indoor Units

7.2.1 Weight

Table 135 Weight of units

	(kg)	(lb)
Enclosure 1101	2.9	6.4
Enclosure 1201	6.2	13.7
Enclosure 1301	3.6	7.9
Enclosure 1401	2.9	6.4
Enclosure 1501	3.5	7.7

2
2

	(kg)	(lb)
Enclosure 1601	4.8	10.6
PFU 1101	0.2	0.4
PFU 1201	0.2	0.4
PFU 1301	0.2	0.4
PFU 1401	0.2	0.4
FAU 1101	0.3	0.7
FAU 1201	0.6	1.3
FAU 1301	0.3	0.7
FAU 1401	0.2	0.4
FAU 1501	0.3	0.7
FAU 1601	0.4	0.9
FAU 1101	0.3	0.7
FAU 1201	0.6	1.3
FAU 1301	0.3	0.7
FAU 1401	0.2	0.4
FAU 1501	0.3	0.7
FAU 1601	0.4	0.9
NPU 1002	0.9	2.0
NPU 1003	0.7	1.5
NPU 1005	0.9	2.0
PNM 1001	1.0	2.2
PNM 1001	1.2	2.6
LTU 1001	0.4	0.9
LTU 1002	0.4	0.9
ETU 1001	0.4	0.9
ETU 1002	0.4	0.9
MMU 1001	0.6	1.4
MMU 1002	0.7	1.5
MMU 1004	0.7	1.5
APU Dummy unit	0.3	0.6
NPU Dummy unit	0.5	1.1
PFU Dummy unit	0.05	0.11
MINI-LINK 6651/1, /2 and /4 (incl. fan)	4.2	9.3

	(kg)	(lb)
MINI-LINK 6651/3	3.0	6.6

7.2.1.1 Full Configurations

Table 136 Weight of full configurations

	(kg)	(lb)
MINI-LINK 6691	5.6	12.3
(Incl. NPU 1002, PFU 1101, FAU 1101, 2 x MMU 1002)		
MINI-LINK 6692	13.5	29.8
(Incl. NPU 1002, PFU 1201, FAU 1201, 8 x MMU 1002)		
MINI-LINK 6693	7.3	16.1
(Incl. NPU 1003, PFU 1301, FAU 1301, 4 x MMU 1002)		
MINI-LINK 6694	9.5	20.9
(Incl. NPU 1002, PFU 1601, FAU 1601, 4 x MMU 1002)		
MINI-LINK 6654	5.5	12.1
(Incl. PNM 1001, FAU 1401, 2 x MMU 1002)		
MINI-LINK 6655	7.7	17.0
(Incl. PNM 1002, FAU 1501, 4 x MMU 1002)		

7.2.2 Nominal Dimensions

The sub-racks fit into standard IEC 297-3 19" cabinets or into cabinets following metric standard IEC 917-2-2.

Table 137	MINI-LINK 6691	and MINI-LINK	6654
-----------	----------------	---------------	------

	(mm)	(inch)
Depth (total; with connectors):	259	10 13/64
Depth (behind mounting brackets):	240	9 29/64
Width (total):	483	19 1/64
Width (without mounting brackets):	446	17 9/16
Height:	44	1 3/4

Table 138 MINI-LINK 6692

	(mm)	(inch)
Depth (total; with connectors):	259	10 13/64
Depth (behind mounting brackets):	240	9 29/64
Width (total):	483	19 1/64
Width (without mounting brackets):	446	17 9/16
Height:	132.7	5 7/32

Table 139 MINI-LINK 6693 and MINI-LINK 6655

	(mm)	(inch)
Depth (total; with connectors):	259	10 13/64
Depth (behind mounting brackets):	240	9 29/64
Width (total):	483	19 1/64
Width (without mounting brackets):	446	17 9/16
Height:	66	2 19/32

Table 140 MINI-LINK 6694

	(mm)	(inch)
Depth (total; with connectors):	259	10 13/64
Depth (behind mounting brackets):	240	9 29/64
Width (total):	483	19 1/64
Width (without mounting brackets):	446	17 9/16
Height:	89	3 1/2

Table 141 MINI-LINK 6651/1, /2 and /4

	(mm)	(inch)
Depth (total; with connectors):	261	10 9/32
Depth (behind mounting brackets):	240	9 29/64
Width (total):	483	19 1/64
Width (without mounting brackets):	444	17 31/64
Height:	44	13/4

Table 142 MINI-LINK 6651/3

	(mm)	(inch)
Depth (total; with connectors):	194	7 41/64
Depth (behind mounting brackets):	171	6 47/64
Width (total):	483	19 1/64
Width (without mounting brackets):	444	17 31/64
Height:	44	1 3/4

8 Standards

Compliance with relevant parts of the listed standards.

8.1 Radio Equipment Directive (RED)

Essential requirements of Directive 2014/53/EU (Radio Equipment Directive, RED)

8.2 ATEX Directive

Applicable for MINI-LINK 6363, MINI-LINK 6364, and MINI-LINK 6365.

Essential requirements of Directive 2014/34/EU (Equipment for potentially explosive atmospheres)

8.3 RoHS Directive

Directive 2011/65/EU	Of the European Parliament and of the Council of 8 June 2011 on the restriction of the use of certain hazardous
CE	substances in electrical and electronic equipment

8.4 Council Recommendation

Council Recommendation 1999/519/EC	Of 12 July 1999 on the limitation of exposure of the general public to electromagnetic fields (0 Hz to 300 GHz)
(10 W/m2 – Public)	

8.5 RoHS

EN 50581	Technical documentation for the assessment of electrical and electronic products with respect to the restriction of
	hazardous substances

8.6 Normative References

Code of Federal Regulations Title 21 Volume 8	Food and Drugs Chapter 1 – Food and Drugs Administration Department of Health and Human Services Subchapter J – Radiological Health
Code of Federal Regulations Title 47 Volume 1 & 5	Telecommunication Chapter I — Federal Communications Commission Chapter I
Interference- Causing Equipment Standards (ICES)	Spectrum Management and Telecommunication Official Publications, Standards, Interference-Causing Equipment Standards (ICES)
Radio Standards Specification (RSS)	Spectrum Management and Telecommunication Official Publications, Standards, Radio Equipment Standards, Radio Standards Specifications (RSS)
Spectrum Utilization Policy (SP)	Spectrum Management and Telecommunication Official Publications, Policies, Spectrum Utilization Policies (SP)
Standard Radio System Plan (SRSP)	Spectrum Management and Telecommunication Official Publications, Standards, Standard Radio System Plan (SRSP)
Telecommunica tions Regulation Circulars (TRC)	Spectrum Management and Telecommunication Official Publications, Information, Telecommunications Regulation Circulars (TRC)

8.7 Mechanics

IEC 60297-3-100	Mechanical structures for electronic equipment – Dimensions of mechanical structures of the 482.6 mm (19 in) series Part 3-100: Basic dimensions of front panels, sub racks, chassis, racks and cabinets
IEC 60297-3-101	Mechanical structures for electronic equipment – Dimensions of mechanical structures of the 482.6 mm (19 in) series Part 3-101: Sub-racks and associated plug-in units
IEC 60297-3-105	Mechanical structures for electronic equipment – Dimensions of mechanical structures of the 482.6 mm (19 in) series Part 3-105: Dimensions and design aspects for 1U high chassis
ETSI EN 300 119-3	Environmental Engineering: European telecommunication standard for equipment practice; Part 3: Engineering requirements for miscellaneous racks and cabinets

-

ETSI EN 300 119-4	Environmental Engineering: European telecommunication standard for equipment practice; Part 4: Engineering requirements for sub racks in miscellaneous racks and cabinets
IEC 60917-2-2	Modular order for the development of mechanical structures for electronic equipment practices – Part 2: Sectional specification – Interface co-ordination dimensions for the 25 mm equipment practice – Section 2: Detail specification – Dimensions for sub racks, chassis, back planes, front panels and plug-in units

8.8 Power

EN 300 132-2	Environment Engineering (EE); Power supply interface at the input to telecommunications equipment and atacom (ICT);	
	Part 2: Operated by -48 V direct current (dc)	
ATIS-06003 15	Voltage Levels for DC-powered Equipment Used in the Telecommunications Environment	

8.9 EMC

ETSI EN 301 489-4	Electromagnetic compatibility and Radio spectrum Matters (ERM); Electro-Magnetic Compatibility (EMC) standard for radio	
(CE DC — Table 6)	radio links and ancillary equipment	
ETSI EN 301 489-1	Electromagnetic compatibility and Radio spectrum Matters (ERM); Electro-Magnetic Compatibility (EMC) standard for radio	
(Emission – Class B Immunity – Level 2)	equipment and services; Part 1: Common technical requirement	
FCC 47 CFR part 15	Code of Federal Regulations Title 47: Telecommunication Part 15 – Radio frequency devices	
ICES-003	Information Technology Equipment (ITE) – Limits and methods of measurement	
GR-1089- CORE	Telcordia Technologies Generic Requirements Electromagnetic Compatibility (EMC) and Electrical Safety – Generic Criteria for Network Telecommunications Equipment	

8.10 Safety/Health

IEC/EN 60215	Safety requirements for radio
	transmitting equipment

(100 W/m2)		
IEC/EN 60 950-1	Information technology equipment – Safety, Part1: General requirements	
(Class III Equipment)		
IEC/EN 60 950-22	Information technology equipment – Safety – Part 22: Equipment installed outdoors	
IEC/EN 60 825-1	Safety of laser products – Part 1:	
(Class 1 Laser)	Equipment classification, requirements and user's guide	
IEC/EN 60529	Degrees of protection provided by	
— IP 66 – Outdoor MINI-LINK 6363	enciosures (IP Code)	
 — IP 55 – Outdoor RAU2 with Ingress Protection Cover 		
— IP 20 – Indoor		
IEC/EN 62311	Assessment of electronic and electrical	
(10 W/m2 – Public)	equipment related to human exposure restrictions for electromagnetic fields (0 Hz-300 GHz)	
EN 50385 (10 W/m2 – Public)	Product standard to demonstrate the compliance of radio base stations and fixed terminal stations for wireless telecommunication systems with the basic restrictions or the reference levels related to human exposure to radio frequency electromagnetic fields (110 MHz – 40 GHz) – General public	
ETSI TR 102 457	Transmission and Multiplexing I;	
— 10 W/m2 – Public	Study on the electromagnetic radiated field in fixed radio systems for	
— 50 W/m2 – Workers		
ICNIRP Health Physics (Table 7)	(Table 7) Guidelines for limiting exposure to time-varying electric, magnetic, and electromagnetic fields (Up to 300 GHz) s	
— 10 W/m2 – Public		
— 50 W/m2 – Workers		
ANSI/UL 60950-1	Information Technology Equipment –	
(Listed/Recognized)	Safety – Part 1: General Requirements	
CSA-C22.2 No. 60950-1	Information Technology Equipment – Safety – Part 1: General Requirements (Bi-National standard, with UL 60950-1)	
(Listed/Recognized)		

ANSI/UL 60950-22 (RAU2)	Information Technology Equipment – Safety – Part 22: Equipment to be Installed Outdoors
CSA-C22.2 No. 60950-22 (RAU2)	Information Technology Equipment – Safety – Part 22: Equipment to be Installed Outdoors (Bi-National standard, with UL 60950-22)
UL 50E	Enclosures for Electrical Equipment,
(Type 3R Enclosure)	Environmental Considerations
CSA-C22.2 No.94	Enclosures for Electrical Equipment,
(Type 3R Enclosure)	Environmental Considerations (Tri- National Standard, with NMX- J-235/2-ANCE-2007 and UL 50E)
FCC 21 CFR part 1040, § Sec. 1040.10	Code of Federal Regulations Title 21
(Class 1 Laser)	Volume 8 – Food and Drugs Chapter 1 – Food and Drugs Administration Department of Health and Human Services Subchapter J – Radiological Health – Performance Standards for Light-Emitting Products – Laser Products
REDR C1370	Radiation Emitting Devices Regulations
FCC OET Bulletin 65	Evaluating Compliance with FCC
— 10 W/m2 – Public	Radiofrequency Electromagnetic
— 100 W/m2 – Workers	rielus
IEEE Std C95.1	IEEE Standard for Safety Levels with
— 10 W/m2 – Public	Respect to Human Exposure to Radio Frequency Electromagnetic Fields, 3
— 100 W/m2 – Workers	
Safety Code 6	Limits of Human Exposure to radio
— 10 W/m2 – Public	trequency Electromagnetic Fields in the Frequency Range from 3 kHz to
— 50 W/m2 – Workers	

8.11 Environmental and Climatic

ETSI EN 300 132-2	Environment Engineering (EE); Power supply interface at the input to telecommunications equipment and atacom (ICT); Part 2: Operated by -48 V direct current (dc)
ETSI EN 300 019-1-1 (Class 1.2)	Environmental Engineering (EE); Environmental conditions and environmental tests for telecommunications equipment; Part 1-1: Classification of environmental conditions; Storage
ETSI EN 300 019-1-2 (Class 2.3)	Equipment Engineering (EE); Environmental conditions and environmental tests for telecommunications equipment; Part 1-2: Classification of environmental conditions; Transportation
ETSI EN 300 019-1-3 (Class 3.1E – Normal Operation, Class 3.3 – Safe Function)	Equipment Engineering (EE); Environmental conditions and environmental tests for telecommunications equipment; Part 1-3: Classification of environmental conditions; Stationary use at weather protected locations
ETSI EN 300 019-1-4 (Class 4.1 & 4.2H – Normal Operation, Class 4.1E & 4.2H – Safe Function)	Equipment Engineering (EE); Environmental conditions and environmental tests for telecommunications equipment; Part 1-4: Classification of environmental conditions; Stationary use at non- weather protected locations
ETSI EN 300 019-2-1	Environmental Engineering (EE); Environmental conditions and environmental tests for telecommunications equipment; Part 2-1: Specification of environmental tests; Storage
ETSI EN 300 019-2-2	Equipment Engineering (EE); Environmental conditions and environmental tests for telecommunications equipment; Part 2-2: Specification of environmental tests; Transportation

ETSI EN 300 019-2-3	Equipment Engineering (EE); Environmental conditions and environmental tests for telecommunications equipment; Part 2-3: Specification of environmental tests; Stationary use at weather protected locations
ETSI EN 300 019-2-4	Equipment Engineering (EE); Environmental conditions and environmental tests for telecommunications equipment; Part 2-4: Specification of environmental tests; Stationary use at non-weather protected locations
	Telcordia Technologies Generic Requirements
GR-63-CORE	NEBSTM Requirements: Physical Protection
FCC 47 CFR part 2	Code of Federal Regulations Title 47: Telecommunication Part 2 – Frequency allocations and radio treaty matters; general rules and regulations.
CENELEC EN 60079-0:2012	Explosive atmospheres - Part 8:
(MINI-LINK 6363, MINI-LINK 6364, and MINI-LINK 6365)	Equipment – General requirements
CENELEC EN 60079-7:2015	Explosive atmospheres — Part 7: Equipment protection by increased safety "e"
(MINI-LINK 6364 and MINI-LINK 6365)	
CENELEC EN 60079-15:2012	Explosive atmospheres – Part 15:
(MINI-LINK 6363)	protection "n"

8.12 PDH

ITU-T 0.151 10/92	Error performance measuring equipment operating at the primary rate and above
ITU-T G.703 10/98	Physical/electrical characteristics of hierarchical digital interfaces
ITU-T G.823 03/00	The control of jitter and wander within digital networks which are based on

	the 2048 Kbit/s hierarchy; Jitter and wander
ITU-T G.826 12/02	Error performance parameters and objectives for international, constant bit rate digital paths at or above the primary rate. (Definition of parameters)
ITU-T G.921	Digital sections based on the 2048 kbit/s hierarchy
ITU-T G.775 10/98	Loss of Signal (LOS), Alarm Indication Signal (AIS) and Remote Defect Indication (RDI) defect detection and clearance criteria for PDH signals
G.742-88	Second order digital multiplex operating at 8.448 Mbit/s.
G.751-88	Third order digital multiplex operating at 34.368 Mbit/s.
ANSI T1.403-1999	Network to customer installation interface-DS1 electrical interface.
ANSI T1.102-1993	Digital Hierarchy, electrical interfaces
ANSI T1.231-1997	Digital Hierarchy-Layer 1 in-Service Digital Transmission Performance Monitoring.
ANSI T1.404-1994	DS3 and Metallic Interface Specification
Bellcore GR-499 -CORE	Transport Systems Generic Requirements (TSGR).

8.13 SDH

ITU-T G.703 10/98	Physical/electrical characteristics of hierarchical digital interfaces
ITU-T G.707 12/03	Network Element interface for the synchronous digital hierarchy (SDH)
ITU-T G.783 02/04	Characteristics of synchronous digital hierarchy (SDH) equipment functional blocks
ITU-T G.826 12/02	Error performance parameters and objectives for international, constant bit rate digital paths at or above the primary rate.
ITU-T G.828 02/00	Error performance parameters and objectives for international, constant bit rate synchronous digital paths
--------------------	---
ITU-T G.841 10/98	Types and characteristics of SDH network protection architecture
ITU-T G.957 07/99	Optical interfaces for equipments and systems relating to the synchronous digital hierarchy
ITU-T M.2120 02/00	PDH path, section and transmission system and SDH path and multiplex section fault detection and localization procedures
ITU-T G.664	Optical safety procedures and requirements for optical transport systems
SFF-8472	Diagnostic Monitoring Interface for Optical Xcvrs
ETSI 301 167	Transmission and Multiplex II; Management of Synchronous Digital Hierarchy (SDH) transmission equipment; Fault management and performance monitoring; Functional description

8.14 Ethernet

Г

IEEE 802.3/802.3u/ 802.3ab	CSMA/CD Access Method and
(Ethernet Interface)	Filysical Layer Specifications
IEEE 802.3ac	Frame Extensions for Virtual Bridged Local Area Network (VLAN) Tagging on IEEE 802.3 Networks
IEEE 802.1ad	IEEE Standard for Local and metropolitan area networks Virtual Bridged Local Area Networks Amendment 4: Provider Bridges
IEEE 802.3ah	Media Access Control Parameters, Physical layers, and Management Parameters for Subscriber Access Networks
IEEE 802.3as-2006 (Frame size)	Carrier Sense Multiple Access with Collision Detection (CSMA/CD) Access Method and Physical Layer Specifications

IEEE 802.1	Working group for 802 LAN/MAN architecture, internetworking among 802 LANs, MANs and other wide area networks, 802 Security, 802 overall network management, and protocol layers above the MAC & LLC layers.
IEEE 802.1p (Priority)	Traffic Class Expediting and Dynamic Multicast Filtering (published in 802.1D-1998)
IEEE 802.1D	IEEE Standard for Local and metropolitan area networks: Media Access Control (MAC) Bridges
IEEE 802.1Q	IEEE Standard for Local and Metropolitan Area NetworksVirtual Bridged Local Area Networks
IEEE 802.3x (Flow control)	IEEE Standards for Local and Metropolitan Area Networks: Specification for 802.3 Full Duplex Operation
IEEE 802.1AX	IEEE Standard for Local and metropolitan area networks— Link Aggregation
IETF RFC 1242	Benchmarking Terminology for Network Interconnection Devices: Performance measurements
RFC 2309	Recommendations on Queue Management and Congestion Avoidance in the Internet
IETF RFC 2544	Benchmarking Methodology for Network Interconnect Devices: Measurement of latency for SDH
RFC 4188	Definitions of Managed Objects for Bridges
MEF 2	Requirements and Framework for Ethernet Service Protection in Metro Ethernet Networks
MEF 9	Abstract Test Suite for Ethernet Services at the UNI
MEF 10	Ethernet Services Attributes Phase I
MEF 14	Abstract Test Suite for Ethernet Services at the UNI
IEEE 802.1ag/ITU-T Y.1731	Ethernet Service OAM CFM
ITU-T Y.1731	Ethernet Service OAM PM

ITU-T G.781 07/99	Synchronization layer functions
ITU-T G.813 03/03	Timing characteristics of SDH equipment slave clocks (SEC)
ITU-T G.823 03/00	The control of jitter and wander within digital networks which are based on the 2048 Kbit/s hierarchy; Jitter and wander
ITU-T G.8261/Y.1361	Timing and synchronization aspects in packet network
ITU-T G.8262/Y.1362	Timing characteristics of a synchronous Ethernet equipment slave clock
ITU-T G.8264/Y.1364	Distribution of timing information through packet networks
G.8265.1	Precision time protocol telecom profile for frequency synchronization
ITU-T G.8273.2	Timing characteristics of telecom boundary clocks and telecom time slave clocks
ITU-T G.8271.1	Network limits for time synchronization in packet networks
ITU-T G.8275.1	Precision time protocol telecom profile for phase/time synchronization with full timing support from the network
IEEE 1588-2008	Standard for a Precision Clock Synchronization Protocol for Networked Measurement and Control Systems

8.16 Digital Radio Transmission

ETSI EN 302 217-2-2	Fixed Radio Systems;
(Radio equipment Class II)	Characteristics and requirements for point-to-point equipment and antennas;
	Part 2-2: Digital systems operating in frequency bands where frequency co- ordination is applied;
	Harmonized EN covering the essential requirements of article 3.2 of the R&TTE Directive
ETSI EN 302 217-2-1	Fixed Radio Systems;
	Characteristics and requirements for point-to-point equipment and antennas;
	Part 2-1: System-dependent requirements for digital systems operating in frequency bands where frequency coordination is applied
ETSI EN 302 217-1	Fixed Radio Systems;
	Characteristics and requirements for point-to-point equipment and antennas;
	Part 1: Overview and system independent common characteristics
ETSI EN 301 126-1	Fixed Radio Systems; Conformance testing; Part 1: Point-to-Point equipment – Definitions, general requirements and test procedures
ETSI EN 301 390	Fixed Radio Systems; Point-to-point and Multipoint Systems; Unwanted emissions in the spurious domain and receiver immunity limits at equipment/antenna port of Digital Fixed Radio Systems
CEPT ERC/DEC/(00) 07	Shared use of the band 17.7-19.7 GHz by the fixed service and Earth stations of the fixed-satellite service.
CEPT/ERC/REC 74-01E	Unwanted Emissions in the Spurious Domain

-
-

Rec. ITU-R F.1099	Radio-frequency channel arrangements for high- and medium- capacity digital fixed wireless systems in the upper 4 GHz (4 400-5 000 MHz) band
Rec. ITU-R F.383	Radio-frequency channel arrangements for high-capacity fixed wireless systems operating in the lower 6 GHz (5 925 to 6 425 MHz) band
Rec. ITU-R F.384	Radio-frequency channel arrangements for medium- and high- capacity digital fixed wireless systems operating in the 6 425-7 125 MHz band
Rec. ITU-R F.385	Radio-frequency channel arrangements for fixed wireless systems operating in the 7 110-7 900 MHz band
Rec. ITU-R F.386	Radio-frequency channel arrangements for fixed wireless systems operating in the 8 GHz (7 725 to 8 500 MHz) band
Rec. ITU-R F.747	Radio-frequency channel arrangements for fixed wireless systems operating in the 10.0-10.68 GHz band
Rec. ITU-R F.1568	Radio-frequency block arrangements for fixed wireless access systems in the range 10.15-10.3/10.5-10.65 GHz
Rec. ITU-R F.387	Radio-frequency channel arrangements for fixed wireless systems operating in the 10.7-11.7 GHz band
Rec. ITU-R F.497	Radio-frequency channel arrangements for fixed wireless systems operating in the 13 GHz (12.75-13.25 GHz) frequency band
Rec. ITU-R F.636	Radio-frequency channel arrangements for fixed wireless systems operating in the 14.4-15.35 GHz band
Rec. ITU-R F.595	Radio-frequency channel arrangements for fixed wireless systems operating in the 17.7-19.7 GHz frequency band

Rec. ITU-R F.637	Radio-frequency channel arrangements for fixed wireless systems operating in the 21.2-23.6 GHz band
Rec. ITU-R F.748	Radio-frequency arrangements for systems of the fixed service operating in the 25, 26 and 28 GHz bands
Rec. ITU-R F.1520	Radio-frequency arrangements for systems in the fixed service operating in the band 31.8-33.4 GHz
Rec. ITU-R F.749	Radio-frequency arrangements for systems of the fixed service operating in the 36-40.5 GHz band
Rec. ITU-R F.2005	Radio-frequency channel and block arrangements for fixed wireless systems operating in the 42 GHz (40.5 to 43.5 GHz) band
CEPT/ERC/REC 14-01 E	Radio-frequency channel arrangements for high capacity analogue and digital radio-relay systems operating in the band 5925 MHz – 6425 MHz
CEPT/ERC/REC 14-02 E	Radio-frequency channel arrangements for medium and high capacity analogue or high capacity digital radio-relay systems operating in the band 6425 MHz-7125 MHz
ECC/REC/(02)06	Preferred channel arrangements for digital Fixed Service Systems operating in the frequency range 7125-8500 MHz
CEPT/ERC/REC 12-05 E	Harmonized radio frequency channel arrangements for digital terrestrial fixed systems operating in the band 10.0 – 10.68 GHz
CEPT/ERC/REC 12-06 E	Preferred channel arrangements for fixed service systems operating in the frequency band 10.7 – 11.7 GHz
CEPT/ERC/REC 12-02 E	Harmonized radio frequency channel arrangements for analogue and digital terrestrial fixed systems operating in the bands 12.75 GHz to 13.25 GHz

CEPT/ERC/REC 12-07 E	Harmonized radio frequency channel arrangements for analogue and digital terrestrial fixed systems operating in the bands 14.5 – 14.62 GHz paired with 15.23 -15.35 GHz
CEPT/ERC/REC 12-03 E	Harmonized radio frequency channel arrangements for analogue and digital terrestrial fixed systems operating in the bands 17.7 GHz to 19.7 GHz
T/R 13-02	Preferred channel arrangements for fixed services in the range 22.0-29.5 GHz
ERC/REC/(01)02	Preferred channel arrangements for digital fixed service systems operating in the frequency band 31.8 – 33.4 GHz
T/R 12-01 E	Harmonized radio frequency channel arrangements for analogue and digital terrestrial fixed systems operating in the bands 37-39.5 GHz
ECC/REC/(01)04	Recommended guidelines for the accommodation and assignment of multimedia wireless systems (MWS) and point-to-point (P-P) fixed wireless systems in the frequency band 40.5 – 43.5 GHz
FCC 47 CFR part 2 (§ 2.902 Verification)	Code of Federal Regulations Title 47: Telecommunication Part 2 – Frequency allocations and radio treaty matters; general rules and regulations.
FCC 47 CFR part 101	Code of Federal Regulations Title 47: Telecommunication Part 101 – Fixed microwave services
RSS-191	Local Multipoint Communication Systems in the Band 25.35-28.35 GHz; Point-to-Point and Point-to-Multipoint Broadband Communication Systems in the Bands 24.25-24.45 GHz and 25.05-25.25 GHz; and Point-to- Multipoint Broadband Communications in the Band 38.6-40.0 GHz
SP Gen	General Information Related to Spectrum Utilization and Radio Systems Policies
SP 1-20 GHz	Revisions to Microwave Spectrum Utilization Policies in the Range of 1-20 GHz

SP 3-30 GHz	Revisions to Spectrum Utilization Policies in the 3-30 GHz Frequency Range and Further Consultation
SRSP-305.9	Technical Requirements for Fixed Line-of-Sight Radio Systems Operating in the Band 5925 – 6425 MHz
SRSP-306.4	Technical Requirements for Fixed Line-of-Sight Radio Systems Operating in the Band 6425 – 6930 MHz
SRSP-306.5	Technical Requirements for Radio Systems Operating in the Fixed Service and Providing Television Auxiliary Services in the Bands 6590 – 6770 MHz and 6930 – 7125 MHz
SRSP-307.1	Technical Requirements for Fixed Line-of-Sight Radio Systems Operating in the Band 7125 – 7725 MHz
SRSP-307.7	Technical Requirements for Fixed Line-of-Sight Radio Systems Operating in the Band 7725 – 8275 MHz
SRSP-310.5	Technical Requirements for Fixed Line-of-Sight Radio Systems Operating in the Band 10.55-10.68 GHz
SRSP-310.7	Technical Requirements for Fixed Line-of-Sight Radio Systems Operating in the Band 10.7 – 11.7 GHz
SRSP-314.5	Technical Requirements for Fixed Line-of-Sight Radio Systems Operating in the Band 14.5 – 15.35 GHz
SRSP-317.8	Technical Requirements for Fixed Line-of-Sight Radio Systems Operating in the Bands, 17.8-18.3 GHz and 19.3-19.7 GHz
SRSP-321.8	Technical Requirements for the Fixed Line-of-Sight Radio Systems Operating in the Bands 21.8 – 22.4 GHz and 23.0 – 23.6 GHz

SRSP-324.25	Technical requirements for Fixed Radio Systems Operating in the Bands 24.25 – 24.45 GHz and 25.05 – 25.25 GHz
SRSP-325.25	Technical Requirements for Fixed Radio Systems Operating in the Bands 25.25-26.5 GHz and 27.5-28.35 GHz
SRSP-338.6	Technical Requirements for Fixed Radio Systems Operating in the Band 38.6 – 40.0 GHz
TRC-43	Designation of Emissions (Including Necessary Bandwidth and Classification), Class of Station and Nature of Service

8.17 Operation & Maintenance Interfaces

8.17.1 DCN

IEEE 802.3	10 BASE-T Ethernet
IEEE 802.2	Local and metropolitan area networks-Specific requirements; Part 2: Logical Link Control
IETF RFC 768	User Datagram Protocol
IETF RFC 791	Internet protocol DARPA internet program protocol specification
IETF RFC 792	Internet control message protocol DARPA internet program protocol specification
IETF RFC 793	Transmission control protocol DARPA internet program protocol specification
IETF RFC 826	Address Resolution Protocol for SiteLAN and local access port
IETF RFC 854	Telnet protocol specification
IETF RFC 894	Transmission of IP datagrams over Ethernet Networks on SiteLAN Access and the local access port.
IETF RFC 951	Bootstrap protocol (bootp)
IETF RFC 959	File transfer protocol (ftp)
IETF RFC 1035	Domain names – implementation and specification
IETF RFC 1042	Transmission of IP Datagrams over IEEE 802 Networks on SiteLAN Access and the local access port.

IETF RFC 1144	PPP TCP/IP header compression	
IETF RFC 1305	Network Time Protocol (Version 3) Specification, Implementation and Analysis	
IETF RFC 1519	Classless Inter-Domain Routing (CIDR): an Address Assignment and Aggregation Strategy	
IETF RFC 1542	Clarifications and Extensions for the Bootstrap Protocol	
IETF RFC 1631	The IP Network Address Translator (NAT) for the local access port.	
IETF RFC 1700	Assigned Numbers (Telnet)	
IETF RFC 1812	Requirements for IP Version 4 Routers	
IETF RFC 2096	IP Forwarding Table MIB	
IETF RFC 2131	Dynamic Host Configuration Protocol for the Local Access Port.	
IETF RFC 2328	OSPF Version 2	
IETF RFC 2474	Definition of the Differentiated Services Field (DS Field) in the Ipv4 and Ipv6 Headers	
IETF RFC 2508	Compressing IP/UDP/RTP Headers for Low-Speed Serial Links	
IETF RFC 2509	IP Header Compression over PPP	
IETF RFC 2780	IANA Allocation Guidelines for values in the Internet Protocol and Related Headers	
IETF RFC 4251	The Secure Shell (SSH) Protocol Architecture	
IETF RFC 5426	Transmission of Syslog Messages over UDP	

8.17.2 SNMP

The list below shows all IETF RFCs, which the MINI-LINK 6600, MINI-LINK 6366, and MINI-LINK 6371 complies with in applicable parts. None of the IETF specified notification types are supported.

IETF RFC 1157	A Simple Network Management Protocol (SNMP)	
IETF RFC 1212	Concise MIB Definitions	
IETF RFC 1213	Management Information Base for Network Management of TCP/IP-based internets: MIB-II	
IETF RFC 1850	OSPF Version 2 Management Information Base	
IETF RFC 1901	Introduction to Community-Based SNMPv2	
IETF RFC 2011	SNMPv2 Management Information Base for the Internet Protocol using SMIv2	

IETF RFC 2012	SNMPv2 Management Information Base for the Transmission Control Protocol using SMIv2
IETF RFC 2013	SNMPv2 Management Information Base for the User Datagram Protocol using SMIv2
IETF RFC 2558	Definitions of Managed Objects for the SONET/SDH Interface Type
IETF RFC 2571	An Architecture for Describing SNMP Management Frameworks
IETF RFC 2578/STD58	Structure of Management Information Version 2 (SMIv2)
IETF RFC 2579/STD58	Textual Conventions for SMIv2
IETF RFC 2580/STD58	Conformance Statements for SMIv2
IETF RFC 2737	Entity MIB
IETF RFC 2863	The Interfaces Group MIB for SiteLAN
IETF RFC 2864 The Inverted Stack Table Extension to the Interfaces Group MIB	
IETF RFC 3410	Introduction and Applicability Statements for Internet-standard Network Management Framework
IETF RFC 3411/STD62	An architecture for Describing SNMP Management Frameworks
IETF RFC 3412/STD62	Message Processing and Dispatching for the SNMP
IETF RFC 3413/STD62	SNMP Applications
IETF RFC 3414/STD62	User-based Security Model (USM) for SNMPv3
IETF RFC 3415/STD62	View-based Access Control Model (VACM) for SNMP
IETF RFC 3416/STD62	Version 2 of the Protocol Operations for SNMP
IETF RFC 3417/STD62	Transport Mappings for SNMP
IETF RFC 3418/STD62	Management Information Base for SNMP
IETF RFC 3584	Co-existence Between Version 1, Version 2,and Version 3 of the Internet-Standard Network Management Framework
IETF RFC 3593	Textual Conventions for MIB Modules Using Performance History Based on 15 Minute Intervals
IETF RFC 4133	Entity MIB (Version 3)
IETF RFC 4188	Definition of Managed Objects for Bridges
IETF RFC 4001	Textual conventions for Internet network addresses
IETF RFC 4805	Definitions of Managed Objects for the DS1, J1, E1, DS2, and E2 Interface Types

8.18 Security

ITU-T X.800	Security Architecture for open system interconnection for CCIT applications
ITU-T X.805	Security Architecture for systems providing end-to-end communications
IETF RFC 2865	RADIUS
IETF draft-grant-tacacs-02.txtTACACS+ v 1.78	
IETF RFC 4251, RFC 4252, RFC 4253	SSH
IETF RFC 3414/STD62	User-based Security Model (USM) for SNMPv3
IETF RFC 3826	The Advanced Encryption Standard (AES) Cipher Algorithm in the SNMP User-based Security Model
IETF RFC 7630	HMAC-SHA-2 Authentication Protocols in the User-based Security Model (USM) for SNMPv3



9 Definitions/Abbreviations

ACAP	Adjacent Channel Alternate Polarization
ACCP	Adjacent Channel Co-Polarization
AIS	Alarm Indicating Signal
APU	Application Plug-in Unit
ATPC	Automatic Transmit Power Control
AU-n	Administrative Unit-n
BER	Bit Error Rate
BERT	Bit error Ratio Test
CBS	Committed Burst Size
CCDP	Co-Channel Dual-Polarization
CESoPSN	Structure-aware TDM Circuit Emulation Service over Packet Switched Network
CIR	Committed Information Rate
CW	Continuous Wave
DCN	Data Communication Network
DTE	Data Terminal Equipment
EBS	Excess Burst Size
ECID	Emulated Circuit Identifier
EIR	Excess Information Rate
EM	Electro Magnetic
EMC	Electro Magnetic Compatibility
ETU	Ethernet Termination Unit

FAU	Fan Unit
FTP	File Transfer Protocol
IEC	International Electro technical Commission
IETF	Internet Engineering Task Force
IME	Inverse Multiplexing
LAG	Link Aggregation Group
LAN	Local Area Network
LED	Light Emitting Diode
LTU	Line Termination Unit
MEF	Metro Ethernet Forum
MIB	Management Information Base
MMU	Modem Unit
MSP	Multiplexer Section Protection
NNI	Network-Network Interface
NPU	Node Processor Unit
OSPF	Open Shortest Path First
PEP	Provider Edge Port
PFU	Power Filtering Unit
PL	Packet Link
PPP	Point to Point Protocol
PRT	Product Ready for Tender
PSU	Power Supply Unit
PTP	Precision Timing Protocol

PW	Pseudowire
QAM	Quadrature Amplitude Modulation
QL	Quality Level
RADIUS	Remote Authentication Dial In User Service
RAU	Radio Unit
RBER	Residual Bit Error Rate
RFC	Request For Comments
RLB	Radio Link Bonding
RLP	Radio Link Protection
RSEC	Reference Spectral Efficiency Class
RTPC	Remote Transmit Power Control
SAToP	Structure-Agnostic TDM over Packet
SFP	Small Form-factor Pluggable
SFPe	SFP electrical
SFPo	SFP optical
SI	Single Interface
SMI	Structure of Management Information
SNCP	Sub-Network Connection Protection
SNMP	Simple Network Management Protocol
SPQ	Strict Priority Queuing
SSM	Synchronization Status Message
STM-n	Synchronous Transport Module-n
TACACS+	Terminal Access Control, Access Control Server

TBD	To Be Determined
тс	Traffic Class
TCP/IP	Transmission Control Protocol/Internet Protocol
TDM	Time Division Multiplexing
TOS	Type Of Service
TUG	Tributary Unit Group
U	Measure used in mechanical constructions (1U = 44.45 mm)
UDP	User Datagram Protocol
UNI	User-Network Interface
USM	User-based Security Model
VACM	View-based Access Model
VCAT	Virtual Concatenation
VC-n	Virtual Container
VCC	Virtual Channel Connection
VID	VLAN ID
VPC	Virtual Path Connection
WAN	Wide Area Network
WDPR	Weighted Deficit Round Robin
WFQ	Weighted Fair Queuing
WRED	Weighted Random Early Detection
XPIC	Cross-Polarization Interference Canceller



Technical Description MINI-LINK 6600 R1

DESCRIPTION

Copyright

 $\ensuremath{\mathbb{C}}$ Ericsson AB 2015–2021. All rights reserved. No part of this document may be reproduced in any form without the written permission of the copyright owner.

Disclaimer

The contents of this document are subject to revision without notice due to continued progress in methodology, design and manufacturing. Ericsson shall have no liability for any error or damage of any kind resulting from the use of this document.

Contents

1	Introduction	1
2 2.1 2.1.1 2.1.2 2.1.3	System Overview Indoor Part MINI-LINK 6651 Indoor Parts MINI-LINK 6654 and MINI-LINK 6655 Indoor Parts MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, and	1 2 4 5
2.2	MINI-LINK 6694 Indoor Parts Outdoor Part	6 8
3	Radio Link	10
3 .1	Hybrid Radio Link	10
3.2	Hitless Adaptive Coding and Modulation	12
3.3 3.3.1 3.3.2	Transmit Power Control RTPC Mode ATPC Mode	15 15 16
3.4	Dual Carrier Modem	17
3.5	Radio Link Bonding	17
3.6	Hierarchical Radio Link Bonding	19
3.7 3.7.1 3.7.2	Multi-Band Booster Multi-Band Booster with Policy-Based Forwarding Multi-Band Booster with Hierarchical Radio Link Bonding	19 20 21
3.8 3.8.1 3.8.2 3.8.3	Radio Link Equipment and Propagation Protection One Dual Carrier Modem Configurations Two Single or Dual Carrier Modems Configurations 1+1 RLP with Asymmetrical Power Splitters	22 24 25 27
3.9 3.9.1 3.9.2 3.9.3 3.9.4 3.9.5	Cross Polarization Interference Canceller (XPIC) XPIC Basics XPIC Recovery and Restore 2+0 RLB with XPIC Configuration Dual (1+0) with XPIC Configuration 2+0 RLB EQP with XPIC Configuration	28 28 33 34 35 36
3.10	Multiple-Input Multiple-Output (MIMO)	36
3.11	Carrier Aggregation (CA)	38
3.12	Multi-Layer Header Compression (MLHC)	38
3.13	AES Encryption Over-the-Hop	40
4	Ethernet Overview	41
4.1	Ethernet Services	41

5	Ethernet Bridging/Switching	42
5.1	L2 Connectivity Provisioning for Customers over a Provider Network	43
5.2	Port Roles	44
5.3	Supported Frame Sizes	45
5.4	Header Decoding	45
5.5	Multiple VLAN Configuration	45
6	Ethernet Security and Admission Control	46
6.1	Frame Admittance	46
6.2 6.2.1	L2CP Frames Processing L2CP Frames Processing on a Layer 1 Ethernet Service	46 51
6.3	White Lists	52
6.4	Ethernet Loop Detection	52
6.5	Storm Protection	53
6.6	Port Blocking	53
6.7	MAC Address Learning	53
7	Quality of Service	54
7.1 7.1.1 7.1.2 7.1.3 7.1.4 7.1.5	Classification and Tagging VID Handling C-VID Registration Table QinQ Priority Handling (Ingress) Priority Handling Layer 1	55 55 57 58 58 63
7.2	Policing/Color Marking (Ingress)	64
7.3	Queue Management	68
7.4	Scheduling (I-NNI and I-NNI (PN))	73
7.5	Congestion Handling Summary	76
8	Hierarchical Quality of Service	77
9	Forwarding	78
9.1	Unicast Forwarding	78
9.2	Multicast Forwarding	79
9.3	IGMP Snooping (IPv4) and MLD Snooping (IPv6)	79
10	Ethernet Protection	81
10.1	RSTP/MSTP	81
10.2 10.2.1 10.2.2	Ethernet Ring Protection (ERP) ERP Single Ring ERP Multiple Rings	81 82 83

10.3.1	LINK Aggregation Group (LAG) with LACP LAG/LACP 1+1 Protection of MINI-LINK 6352	84 86
10.4	Ethernet Layer 1 Protection Mechanisms	86
11	Ethernet Operation & Maintenance	87
11.1	Ethernet Performance Counters	87
11.2	Ethernet Link OAM	87
11.3	Ethernet Service OAM	89
11.4	Port Mirroring	93
12	L3 Services	94
12.1	L3 IP Router	95
12.2	L3VPNs	96
12.3	Supported Routing Protocols	96
12.4	MPLS Networking	98
12.4.1 12.4.2	MPLS-Based Solutions	98
12.4.3	RSVP-TE Fast Reroute	99
12.4.4	RSVP-TE Path Protection	100
12.4.5	Seamless MPLS	101
12 5	Supported IP Protocols	103
12.5		105
12.5 13	TDM	103
12.5 13 13.1	TDM Protection Mechanisms	103 104 104
12.5 13 13.1 13.1.1 13.1.2	TDM TDM Protection Mechanisms Overview	103 104 104 104
12.5 13 13.1 13.1.1 13.1.2 13.1.3	TDM TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH	103 104 104 106 108
12.5 13 13.1 13.1.1 13.1.2 13.1.3 13.2	TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH TDM over Packet using CES	103 104 104 106 108 110
12.5 13 13.1 13.1.1 13.1.2 13.1.3 13.2 13.2.1	TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH TDM over Packet using CES Synchronization of TDM Interface	103 104 104 104 106 108 110 111
12.5 13 13.1 13.1.1 13.1.2 13.1.3 13.2 13.2.1 14	TDM TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH TDM over Packet using CES Synchronization of TDM Interface Synchronization	103 104 104 106 108 110 111 111
12.5 13.1 13.1.1 13.1.2 13.1.3 13.2 13.2.1 14 14.1	TDM TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH TDM over Packet using CES Synchronization of TDM Interface Synchronization General Overview	103 104 104 106 108 110 111 111
12.5 13 13.1 13.1.1 13.1.2 13.1.3 13.2 13.2.1 14 14.1 14.2 14.2	TDM TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH TDM over Packet using CES Synchronization of TDM Interface Synchronization General Overview Frequency Synchronization Technologies Supported in Protocols	103 104 104 106 108 110 111 111 111 114
12.5 13.1 13.1.1 13.1.2 13.1.3 13.2 13.2.1 14.1 14.2 14.2.1 14.2.1 14.2.1	TDM TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH TDM over Packet using CES Synchronization of TDM Interface Synchronization General Overview Frequency Synchronization Technologies Sync over Radio Link Synchronous Ethernet	103 104 104 106 108 110 111 111 111 111 114 114 115
12.5 13 13.1 13.1.1 13.1.2 13.1.3 13.2 13.2.1 14 14.1 14.2 14.2.1 14.2.2 14.2.3	TDM TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH TDM over Packet using CES Synchronization of TDM Interface Synchronization General Overview Frequency Synchronization Technologies Sync over Radio Link Synchronous Ethernet Sync over SDH	103 104 104 106 108 110 111 111 111 111 114 114 115 115
12.5 13 13.1 13.1.1 13.1.2 13.1.3 13.2 13.2.1 14 14.1 14.2 14.2.1 14.2.1 14.2.2 14.2.3 14.2.4 14.2.5	TDM TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH TDM over Packet using CES Synchronization of TDM Interface Synchronization General Overview Frequency Synchronization Technologies Sync over Radio Link Synchronous Ethernet Sync over SDH Sync over PDH Sync over PDH	103 104 104 106 108 110 111 111 111 111 114 114 115 115 116 116
12.5 13 13.1 13.1.1 13.1.2 13.1.3 13.2 13.2.1 14 14.1 14.2 14.2.1 14.2.1 14.2.2 14.2.3 14.2.4 14.2.5 14.2.6	TDM TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH TDM over Packet using CES Synchronization of TDM Interface Synchronization General Overview Frequency Synchronization Technologies Sync over Radio Link Synchronous Ethernet Sync over SDH Sync over 2 MHz/2048 kbps Sync over PTP (G.8265.1 Mode)	103 104 104 106 108 110 111 111 111 111 114 115 115 116 116 117
12.5 13 13.1 13.1.1 13.1.2 13.1.3 13.2 13.2.1 14 14.1 14.2 14.2.1 14.2.1 14.2.2 14.2.3 14.2.4 14.2.5 14.2.6 14.3	TDM TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH TDM over Packet using CES Synchronization of TDM Interface Synchronization General Overview Frequency Synchronization Technologies Sync over Radio Link Synchronous Ethernet Sync over SDH Sync over PDH Sync over PTP (G.8265.1 Mode) Time Synchronization Technologies	103 104 104 104 106 108 110 111 111 111 111 114 115 115 116 116 117 118
12.5 13 13.1 13.1.1 13.1.2 13.1.3 13.2 13.2.1 14 14.1 14.2 14.2.1 14.2.1 14.2.2 14.2.3 14.2.4 14.2.5 14.2.6 14.3 14.3.1	TDM TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH TDM over Packet using CES Synchronization of TDM Interface Synchronization General Overview Frequency Synchronization Technologies Sync over Radio Link Synchronous Ethernet Sync over SDH Sync over PDH Sync over PTP (G.8265.1 Mode) Time Synchronization Technologies Default PTP Profile	103 104 104 104 106 108 110 111 111 111 111 114 115 115 116 116 116 117 118 119
12.5 13.1 13.1.1 13.1.2 13.1.3 13.2 13.2.1 14.1 14.2 14.2.1 14.2.2 14.2.3 14.2.4 14.2.5 14.2.5 14.2.6 14.3 14.3.1 14.3.2 14.3.5	TDM TDM Protection Mechanisms Overview Network Layer Protection MSP 1+1 for SDH TDM over Packet using CES Synchronization of TDM Interface Synchronization General Overview Frequency Synchronization Technologies Sync over Radio Link Synchronous Ethernet Sync over SDH Sync over SDH Sync over PTP (G.8265.1 Mode) Time Synchronization Technologies Default PTP Profile ITU-T G.8275.1 Profile Time Error Generation	103 104 104 106 108 110 111 111 111 111 114 115 115 116 116 116 117 118 119 120 120

14.4	Precision Time Protocol	121
15	Hardware	123
$15.1 \\ 15.1.1 \\ 15.1.2 \\ 15.1.3 \\ 15.1.4 \\ 15.1.5 \\ 15.1.6$	MINI-LINK 6651 DC Supply Voltage Fan Unit Traffic Capacities for MINI-LINK 6651 Hardware Handling Ethernet Switch Ethernet LAN Ports	123 130 130 131 134 135 135
15.2	MINI-LINK 6654	136
15.2.1	PNM 1001	138
15.2.2	PNM 1002	143
15.2.3	Enclosure 1401	149
15.2.4	Input Voltage Behavior	149
15.3	MINI-LINK 6655	150
15.3.1	Enclosure 1501	151
15.3.2	Input Voltage Behavior	151
15.4	MINI-LINK 6691	152
15.4.1	PFU 1101	153
15.4.2	FAU 1101	154
15.4.3	Backplane Architecture	154
15.4.4	Hardware Handling	155
15.4.5	Input Voltage Behavior	156
15.5	MINI-LINK 6692	157
15.5.1	PFU 1201	158
15.5.2	FAU 1201	159
15.5.3	Backplane Architecture	161
15.5.4	Hardware Handling	162
15.5.5	NPU Protection	163
15.5.6	Input Voltage Behavior	163
15.6	MINI-LINK 6693	164
15.6.1	PFU 1301	165
15.6.2	FAU 1301	166
15.6.3	Backplane Architecture	167
15.6.4	Hardware Handling	168
15.6.5	Input Voltage Behavior	169
15.7	MINI-LINK 6694	170
15.7.1	PFU 1601	170
15.7.2	FAU 1601	171
15.7.3	Backplane Architecture	173
15.7.4	NPU Protection	174
15.7.5	Hardware Handling	174
15.7.6	Input Voltage Behavior	175
15.8	NPU 1002	176
15.8.1	NPU 1002 Front Interfaces	178
15.8.2	Ethernet Switch	179
15.8.3	Ethernet LAN Ports	180

15.9 15.9.1 15.9.2 15.9.3	NPU 1003 NPU 1003 Front Interfaces Ethernet Switch Ethernet LAN Ports	181 182 184 184
15.10 15.10.1 15.10.2 15.10.3	NPU 1005 Front Interfaces Ethernet Switch Ethernet LAN Ports	185 187 188 188
15.11	MMU 1001	189
15.12	MMU 1002	195
15.13	MMU 1004	203
15.14	LTU 1001	211
15.15	LTU 1002	211
15.16	ETU 1001	213
15.17	ETU 1002	214
15.18 15.18.1 15.18.2	Radio Units Overview External Interfaces	214 214 217
15.18.3	Radio Modulations	217
15.19 15.19.1	Antennas MINI-LINK 6363, MINI-LINK 6364 and MINI-LINK 6365 Integrated Installation	225 226
15.19.2 15.19.3	RAU2 X Integrated Installation MINI-LINK 6363, MINI-LINK 6364, MINI-LINK 6365, and RAU2 X Separate Installation	228 230
16	Management	230
161	5	
16.1.1 16.1.2 16.1.3 16.1.4	DCN IP Services DCN Interfaces IP Addressing IP Router	230 231 231 234 235
16.1.1 16.1.2 16.1.3 16.1.4 16.2	DCN IP Services DCN Interfaces IP Addressing IP Router Link Layer Discovery Protocol (LLDP)	230 231 231 234 235 236
16.1.1 16.1.2 16.1.3 16.1.4 16.2 16.3 16.3.1 16.3.2 16.3.3 16.3.4 16.3.5	DCN IP Services DCN Interfaces IP Addressing IP Router Link Layer Discovery Protocol (LLDP) Management Tools and Interfaces MINI-LINK Node GUI SNMP Command Line Interfaces Syslog NETCONF	230 231 234 235 236 236 236 236 237 237 237
16.1.1 16.1.2 16.1.3 16.1.4 16.2 16.3 16.3.1 16.3.2 16.3.3 16.3.4 16.3.5 16.4 16.4.1 16.4.1	DCN IP Services DCN Interfaces IP Addressing IP Router Link Layer Discovery Protocol (LLDP) Management Tools and Interfaces MINI-LINK Node GUI SNMP Command Line Interfaces Syslog NETCONF Configuration Handling Configuration File Aligning the Node with the Network Management System	230 231 234 235 236 236 236 236 237 237 237 237 237 237 238
$\begin{array}{c} 16.1\\ 16.1.1\\ 16.1.2\\ 16.1.3\\ 16.1.4\\ 16.2\\ 16.3\\ 16.3.1\\ 16.3.2\\ 16.3.3\\ 16.3.4\\ 16.3.5\\ 16.4\\ 16.4.1\\ 16.4.2\\ 16.5\\ 16.5.1\\ \end{array}$	DCN IP Services DCN Interfaces IP Addressing IP Router Link Layer Discovery Protocol (LLDP) Management Tools and Interfaces MINI-LINK Node GUI SNMP Command Line Interfaces Syslog NETCONF Configuration Handling Configuration File Aligning the Node with the Network Management System Software Management Uploading Local Logs to a Remote FTP Server	230 231 234 235 236 236 236 236 237 237 237 237 237 237 237 237 238 238 238

	16.6	License Handling	239		
	16.7 16.7.1 16.7.2 16.7.3 16.7.4	Fault Management Alarm Handling Ethernet Link OAM Ethernet Service OAM Loops	240 240 241 241 241		
	16.8	Spectrum Diagnostics Scan for Potential Interferer	246		
	16.9 16.9.1 16.9.2 16.9.3 16.9.4 16.9.5 16.9.6 16.9.7 16.9.8 16.9.9	Performance Measurements Bulk PM Radio Link Performance Measurements Ethernet PM Counters PDH Bit Error Testing Ethernet Service OAM IP Performance Monitoring with TWAMP Reflector Light PTP Performance Monitoring Power Measurement PM	246 247 248 249 249 250 251 251 252		
	$\begin{array}{c} 16.10\\ 16.10.1\\ 16.10.2\\ 16.10.3\\ 16.10.4\\ 16.10.5\\ 16.10.6\\ 16.10.7\\ 16.10.8\\ 16.10.9\\ 16.10.10\\ 16.10.11\\ 16.10.12\\ 16.10.13\\ \end{array}$	Node Security Authentication, Authorization, and Accounting NTP Authentication SFTP File Integrity Violation Security Audit Logging Monitoring Local Users and Active Sessions Syslog for Security Events Disable Local O&M Ports Notification on Logon Password Cracking Defense: Setting Brute Force Threshold Firewall Legal Notice Access Control Lists	252 254 259 259 260 260 260 261 261 261 262 262 262 262		
	17	Accessories	263		
	17.1	SFP	263		
	17.2	SFP+	263		
	17.3	Tunable DWDM SFP+	264		
	17.4	Direct-Attach Copper Cable	264		
Reference List 26					



1 Introduction

MINI-LINK 6600 consists of small flexible nodes for high capacity, with advanced Ethernet functionality and designed for an efficient migration path for customers going to an all-IP network.

The MINI-LINK 6600 product family:

- MINI-LINK 6651 (Compact Node)
- MINI-LINK 6654 with PNM 1001 or PNM 1002
- MINI-LINK 6655 with PNM 1002
- MINI-LINK 6691 with NPU 1002, NPU 1003, or NPU 1005
- MINI-LINK 6692 with NPU 1002 or NPU 1005
- MINI-LINK 6693 with NPU 1002, NPU 1003, or NPU 1005
- MINI-LINK 6694 with NPU 1002, NPU 1003, or NPU 1005

MINI-LINK 6600 provides a switch capacity of up to 88 Gbps and can connect up to 16 split mounted radio links with a maximum of 15 radio directions due to being equipped with dual carrier modems. 1 Gbps, 2.5 Gbps, and 10 Gbps interfaces, fiber rings and E-band links can also be connected to the node. The node supports radio links with high modulation of 8192 QAM.

MINI-LINK 6600 offers both compact nodes and nodes using plug-in units, which make it easy to customize configurations and make future upgrades. The full range of MINI-LINK outdoor units can easily be combined in many different ways: traditional frequencies (6–42 GHz), V-band 60 GHz, E-band 70/80 GHz, single and dual carrier, Coax and Ethernet interface.

Different packet migration strategies are supported by MINI-LINK 6600. For cost efficient migration MINI-LINK 6600 are hop compatible with MINI-LINK TN R6 equipped with MMU4 A. Upgrading a site to MINI-LINK 6600, the radio unit, antenna and outdoor cabling can be reused.

2 System Overview

From a functional and configuration perspective, a Network Element (NE) based on MINI-LINK 6600 can be divided into the following parts:

Basic NodeThe Basic Node holds the system platform providing traffic
and system control.Specific plug-in units provide Ethernet and TDM traffic
interfaces for connection to other network equipment, for
example radio base stations.It also includes indoor mechanical housing, power
distribution, and cooling.Radio TerminalsA Radio Terminal provides microwave transmission and
consists of an outdoor radio connected to an indoor modem
carrier, that is, a split mount solution.MINI-LINK 6600 provides single or dual carrier modem,
and can be connected to two outdoor radios to create two
separate Radio Terminals.



For more information, see Section 3 on page 9.

Figure 1 Basic Node and Radio Terminals

2.1 Indoor Part

All MINI-LINK 6600 products fit into standard 19" or metric racks.

	MINI -LINK 6651/1 Compact Node	MINI -LINK 6651/2 Compact Node	MINI -LINK 6651/3 Compact Node	MINI -LINK 6651/4 Compact Node	MINI-LI NK 6654	MINI-LI NK 6655	MINI-LI NK 6691	MINI-LI NK 6692	MINI-LI NK 6693	MINI-LI NK 6694
Height	1U	1U	1U	1U	1U	1.5U	1U	3U	1.5U	2U
Traffic interface	2×FE/1G RJ45	2×FE/1G RJ45	2×FE/1G RJ45	2×1G RJ45	2xFE/1G RJ45	2xFE/1G RJ45	2×FE/1G RJ45	2×FE/1G RJ45	2×FE/1G RJ45	2×FE/1G RJ45
5	4×1G/2. 5G SFP+	3×1G/2. 5G SFP+	4×1G/2. 5G SFP+	4×1G/10 G SFP+	2x1G SFP	2x1G SFP	1×FE/1G SFP	1×FE/1G SFP	1×FE/1G SFP	1×FE/1G SFP
	8×E1	8×E1	8×E1	8×E1	2x1G/10 G SFP+	2x1G/10 G SFP+	2×1G/10 G SFP+	2×1G/10 G SFP+	2×1G/10 G SFP+	2×1G/10 G SFP+
					4xE1/DS 1	4xE1/DS 1				
Switch capacity	14.5 Gbps	14.5 Gbps	14.5 Gbps	47 Gbps	36.5 Gbps (PNM 1001)	44 Gbps	38 Gbps (NPU 1002 or NPU 1003)	88 Gbps ⁽ NPU 1002)	46.5 Gbps (NPU 1002)	59.5 Gbps (NPU 1002)
					44 Gbps (PNM 1002)		43 Gbps (NPU 1005)	65.5 Gbps (NPU 1005)	45.5 Gbps (NPU 1003)	45.5 Gbps (NPU 1003)
									55.5 Gbps (NPU 1005)	55.5 Gbps (NPU 1005)
Combo solution (PFU, NPU, and MMU)	_	_	_	_	PNM 1001 or PNM 1002	PNM 1002	_	_	_	_
Node Process or Unit (NPU)	_	_	_	_	_	_	NPU 1002, NPU 1003 or NPU 1005	NPU 1002 or NPU 1005 ⁽²⁾	NPU 1002, NPU 1003 or NPU 1005	NPU 1002, NPU 1003 or NPU 1005 ⁽²⁾
Dual Carrier Modem Unit (MMU)	_	_	_	_	MMU 1002 or MMU 1004	MMU 1002 or MMU 1004	MMU 1002 or MMU 1004	MMU 1002 or MMU 1004	MMU 1002 or MMU 1004	MMU 1002 or MMU 1004
Single Carrier Modem Unit (MMU)	-	-	_	-	MMU 1001	MMU 1001	MMU 1001	MMU 1001	MMU 1001	MMU 1001
Line Termina tion Unit (LTU) for E1/DS1	_	_	_	_	LTU 1001	LTU 1001	LTU 1001	LTU 1001	LTU 1001	LTU 1001
Line Termina tion Unit (LTU) for E1/DS 1 and STM-1	-	-	-	-	LTU 1002	LTU 1002	LTU 1002	LTU 1002	LTU 1002	LTU 1002

Table 1 MINI-LINK 6600 Comparison

	MINI -LINK 6651/1 Compact Node	MINI -LINK 6651/2 Compact Node	MINI -LINK 6651/3 Compact Node	MINI -LINK 6651/4 Compact Node	MINI-LI NK 6654	MINI-LI NK 6655	MINI-LI NK 6691	MINI-LI NK 6692	MINI-LI NK 6693	MINI-LI NK 6694
Ethernet Termina tion Unit (ETU)	-	_	-	-	ETU 1001	ETU 1001	ETU 1001	ETU 1001	ETU 1001	ETU 1001
Ethernet Termina tion Unit (ETU)	-	_	_	-	ETU 1002	ETU 1002	ETU 1002	ETU 1002	ETU 1002	ETU 1002

Table 1 MINI-LINK 6600 Comparison

(1) 7.5 Gbps is reserved for future use.

(2) NPU protection can be used with two NPU 1002 or two NPU 1005.

The indoor part also includes cables and installation accessories.

The interconnection between the outdoor part (radios and antennas) and the indoor part is one coaxial cable per carrier interface. This cable carries full duplex traffic, DC supply voltage, and management data.

2.1.1 MINI-LINK 6651 Indoor Parts

MINI-LINK 6651 is a compact node suitable for tail sites.

It has a replaceable Fan Unit that provides cooling for the indoor part.

There are four variants, MINI-LINK 6651/1 (Figure 2), MINI-LINK 6651/2 (Figure 3), MINI-LINK 6651/3 (Figure 4), and MINI-LINK 6651/4 (Figure 5).

MINI-LINK 6651/1



Figure 2 MINI-LINK 6651/1







Figure 4 MINI-LINK 6651/3



Figure 5 MINI-LINK 6651/4

2.1.2 MINI-LINK 6654 and MINI-LINK 6655 Indoor Parts

The indoor part consists of an Enclosure with plug-in units interconnected through a backplane. One plug-in unit occupies one slot in the Enclosure.

Enclosure The Enclosure houses the plug-in units and provides backplane interconnection of traffic, power, and control signals. It fits into standard 19" or metric racks.

PFU, NPU, and MMU (PNM) Unit				
	The PNM handles the control functions of the system. It also provides traffic and management interfaces.			
Modem Unit (MMU)				
	The MMU constitutes the indoor part of a Radio Terminal. It determines the traffic capacity and modulation scheme			
Fan Unit (FAU)	The FAU provides cooling for the indoor part.			
Line Termination Unit	(LTU)			
	The LTU provides E1/DS1 interfaces. LTU 1002 also provides an STM-1 interface.			
Ethernet Termination	Unit (ETU)			
	The ETU provides Ethernet interfaces.			

2.1.3 MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, and MINI-LINK 6694 Indoor Parts

The indoor part consists of an Enclosure with plug-in units interconnected through a backplane. One plug-in unit occupies one slot in the Enclosure.

Enclosure	The Enclosure houses the plug-in units and provides backplane interconnection of traffic, power, and control signals. It fits into standard 19" or metric racks.
Node Processor Unit (NPU)	The NPU handles the control functions of the system. It also provides traffic and management interfaces.
Modem Unit (MMU)	The MMU constitutes the indoor part of a Radio Terminal. It determines the traffic capacity and modulation scheme.
Power Filter Unit (PFU)	The PFU filters the external power and distributes the internal power to the plug-in units through the backplane.
Fan Unit (FAU)	The FAU provides cooling for the indoor part.
Line Termination Unit (LTU)	The LTU provides E1/DS1 interfaces. LTU 1002 also provides an STM-1 interface.
Ethernet Termination Unit (ETU)	The ETU provides Ethernet interfaces.



Figure 6 MINI-LINK 6691



Figure 7 MINI-LINK 6692



Figure 8 MINI-LINK 6693



Figure 9 MINI-LINK 6694

2.2 Outdoor Part

The outdoor part is supplied for various frequency bands, including the E-band (80 GHz). It consists of an antenna, a radio, and associated installation hardware. The MINI-LINK 6363, MINI-LINK 6363/2, MINI-LINK 6364, MINI-LINK 6365, and RAU2 X radios can be connected to a MINI-LINK 6600 modem.

The radio and the antenna are easily installed on a wide range of support structures. The radio is fitted directly to the antenna as standard, integrated installation. The radio and the antenna can also be fitted separately and connected by a flexible waveguide. In all cases, the antenna is easily aligned and the radio can be disconnected and replaced without affecting the antenna alignment.

The MINI-LINK 6363, MINI-LINK 6363/2, MINI-LINK 6364, MINI-LINK 6365, and RAU2 X radios are described in more detail in Section 15.18 on page 214.

The antennas are described in more detail in Section 15.19 on page 225.



Figure 10 MINI-LINK 6363, MINI-LINK 6364, or MINI-LINK 6365 and Antenna



Figure 11 RAU2 X and Antennas in Different Installation Alternatives

3 Radio Link

MINI-LINK 6600 provides radio links for microwave transmission up to 1067 Mbps, operating within the 5 to 80 GHz frequency bands, using 4–8192 QAM modulation schemes.

MINI-LINK 6600 supports Ethernet (packet link), PDH, and a mix thereof in a Hybrid Radio Link sending Ethernet and PDH traffic simultaneously.

MINI-LINK 6600 can be configured as unprotected (1+0), 1+1 Radio Link Protection, or 2+0 Radio Link Bonding, and supports Cross Polarization Interference Canceller (XPIC), Adaptive Coding and Modulation (ACM). MINI-LINK 6600, except from MINI-LINK 6651, also supports Multiple Input Multiple Output (MIMO). Dual 1+0 is available in MINI-LINK 6600, except with MINI-LINK 6651/1 or MINI-LINK 6651/3.

3.1 Hybrid Radio Link

A Hybrid Radio Link is a Radio Link optimized for maximum throughput of Ethernet and PDH traffic.

Note: MINI-LINK 6651 can only send PDH traffic with E1s, that is, only for ETSI.

Ethernet and PDH traffic are sent simultaneously over the Hybrid Radio Link, see Figure 12.



Figure 12 Traffic over a Hybrid Radio Link

The Ethernet traffic is sent over a dedicated physical link. When Ethernet is sent over a Hybrid Radio Link, the capacity range is from 0 Mbps to total link capacity depending on the amount of configured PDH Traffic.

A Hybrid Radio Link supports flat multiplexing of PDH traffic, which enables control of the number of E1s/DS1s to be transported. PDH traffic is normally transported in sets of $4 \times E1/DS1$ and $16 \times E1/DS1$. With flat multiplexing it is possible to set an exact number of E1s/DS1s to be transported, for example, $7 \times E1/DS1$ or $23 \times E1/DS1$. This allows optimized usage of bandwidth since all E1s/DS1s fitting in the bandwidth also can be transported. Due to the use of one mux layer for all E1s/DS1s, instead of one mux layer for each set of $4 \times E1/DS1$ and $16 \times E1/DS1$, the PDH overhead is decreased. A minimum of the total link capacity is used for overhead and PDH traffic and all remaining capacity can be used for Ethernet.

The ratio between Ethernet and PDH traffic is configurable and is set with E1/DS1 granularity, see Figure 13.





Example (ETSI)

- Total link capacity: 154 Mbps
- PDH traffic capacity: 22×E1 (45 Mbps)
- Ethernet capacity: 154 Mbps 45 Mbps = 109 Mbps

Example (ANSI)

- Total link capacity: 158 Mbps
- PDH traffic capacity: 22×DS1 (34 Mbps)
- Ethernet capacity: 158 Mbps 34 Mbps = 124 Mbps

In Hybrid Radio Links, Ethernet capacity range from 0 to maximum link capacity, while PDH capacity range from 0 to 80×E1/DS1. The PDH capacity is limited by the backplane capacity of the modem.

MMU 1002/1004 and MINI-LINK 6651 are hop compatible with MMU4 A in MINI-LINK TN R6. See MINI-LINK 6600 R1 Compatibility, Reference [1], for details regarding hop compatibility.

3.2 Hitless Adaptive Coding and Modulation

Hitless Adaptive Coding and Modulation (ACM) enables automatic switching between different physical modes, depending on radio channel conditions. Hitless ACM makes it possible to increase the available capacity over the same frequency channel during periods of normal propagation conditions.

Code rate and modulation, and thereby capacity, are high during normal radio channel conditions and lower during less favorable channel conditions, for example, when affected by rain or snow. Physical mode switches are hitless, that is, error free. In situations where traffic interruption normally would occur, it is possible to maintain parts of the traffic by switching to a lower physical mode, using Hitless ACM.

Figure 14 shows how the capacity changes when the received input signal crosses the receiver threshold for each physical mode order.




When using only Adaptive Modulation, the steps in Figure 14 only differ in terms of modulation. When using ACM, the steps can differ in both coding and modulation, which increases the number of possible steps.

In order to handle channel variations, the channel conditions are continuously monitored on the Rx side by measurement of Signal to Noise and Interference Ratio (SNIR). When the receiver, based on this data, detects that channel conditions imply a change to the next higher or lower physical mode, a message is sent to the transmitter on the other side requesting a higher or lower physical mode. Upon receipt of such request the transmitter starts transmitting with the new physical mode. Each direction is independent. At demodulation the receiver follows the physical mode as a slave.

The physical mode can also be configured with the maximum physical mode equal to the minimum physical mode, and thereby achieving a mode comparable to static mode, where the physical mode remains unchanged.

Hitless ACM is compatible with Automatic Transmit Power Control (ATPC), which works in a closed loop in any configured physical mode. If the modulation goes down, for example, due to interference, then the output power does not go to the maximum level.

Buffering

2

ACM can influence the design of the buffer dimensioning. In case packet aging is not used, the maximum delay variation time will increase due to that the buffer is configured in bytes and that data will travel at a slower speed during lower physical mode steps. When packet aging is enabled, the maximum delay variation time will be kept regardless of physical mode level. This will also ensure that there is no old data in lower priority queues when physical mode is increased after a fading situation.

ACM can influence the position of the narrowest congestion point in the network, with too small buffers this can have a strong negative impact on utilization and end user TCP performance. To ensure high link utilization and high TCP performance, buffers for TCP traffic should be dimensioned in the area above average Round Trip Time (RTT), which is typically in the area of 100–200 ms.

RSEC (ETSI only)

Reference Spectrum Efficiency Class (RSEC) defines the spectral mask, that is, which SEC that is used as a reference. The maximum output power is dependent on the RSEC used. The configurable RSEC depends on the configured minimum physical mode, which is illustrated in Table 2.

Configured Minimum Physical Mode	Configurable RSEC
4 QAM	2, 4L, 4H, 5LB, 5HB, 6LB, 6HB, 7
16 QAM	4L, 4H, 5LB, 5HB, 6LB, 6HB, 7
32 QAM	4H, 5LB, 5HB, 6LB, 6HB, 7
64 QAM	5LB, 5HB, 6LB, 6HB, 7
128 QAM	5HB, 6LB, 6HB, 7
256 QAM	6LB, 6HB, 7
512 QAM	6НВ, 7
1024 QAM	7
2048 QAM	7
4096 QAM	7
8192 QAM	7

Table 2 RSEC (ETSI)

When installing a new link or when upgrading an existing link to ACM, it is important to consider spectrum license requirements. The link configuration must be set with respect to the RSEC (referred to as Reference Mode in ETSI EN 302 217-1 and ETSI EN 302 217-2-2) and output power as specified in the spectrum license. Consult the relevant authority for information on the national radio transmission specifications and spectrum license conditions.

3.3 Transmit Power Control

The radio transmit power can be controlled in Remote Transmit Power Control (RTPC) mode, selectable from the management system, including setting of associated parameters. In Automatic Transmit Power Control (ATPC) mode, the transmit power can be increased rapidly during fading conditions, which allows the transmitter to operate at less than the maximum power during normal path conditions. The normally low transmit power allows more efficient use of the available spectrum, while the high transmit power can be used as input to path reliability calculations, such as fading margin and carrier-to-interference ratio.

Traditionally, the radio unit consumes the same amount of power independent of the output power used in the air interface. By using MINI-LINK ECO Mode, the radio units adjust the power requirements according to the used output power in the radio interface, which varies according to the actual link conditions and output margin design.



The transmitter can be turned on or off from the management system.

Figure 15 Transmit Power Control

3.3.1 RTPC Mode

In RTPC mode, the transmit power (P_{out}) ranges from a minimum level ($P_{fix min}$) to a maximum level (P_{max}). The desired value (P_{set}) can be set in 1 dBm increments.

3.3.2 ATPC Mode

ATPC is used to automatically adjust the transmit power (P_{out}) in order to maintain the received input level at the far-end terminal at a target value. The received input level is compared with the target value, and then a deviation is calculated and sent to the near-end terminal to be used as input for possible adjustment of the transmit power. ATPC varies the transmit power between a selected maximum level ($P_{ATPC max}$) and a selected minimum level ($P_{ATPC min}$).

The highest possible value for $P_{ATPC\ max}$ is P_{max} . The lowest possible value for $P_{ATPC\ min}$ is $P_{fix\ min}$. Therefore, the maximum adjustment range varies between these two values.

It is possible to enable a fallback mode for the ATPC, so that the transmitted power is decreased to a user settable level ($P_{ATPC \ Fallback}$) if it has been stuck at $P_{ATPC \ max}$ for too long. The ATPC fallback can be enabled using CLI or MINI-LINK Node GUI. When fallback mode is entered, an alarm is raised and is not cleared as long as the system is in ATPC mode.

 $P_{ATPC \ Fallback}$ can be set between $P_{ATPC \ max}$ and $P_{ATPC \ min}$.

The fallback timer can be set between 1 and 60 minutes.

It is recommended to set the target to a value 5 dB above the RBER-threshold for the highest configured modulation.

3.3.2.1 Traffic Aware Power Save (TAPS)

TAPS (Traffic Aware Power Save) is a power saving functionality available for all MINI-LINK 6600 products, when used with MINI-LINK 6363, MINI-LINK 6364, or MINI-LINK 6365 radio units. It is aimed at keeping the system power use as low as possible while still handling the actual traffic rate.

When TAPS is enabled, power consumption is automatically adapted based on actual capacity need by minimizing the transmitted power. Every 6 milliseconds, link utilization is measured by comparing the actual traffic rate with the possible traffic rate. If the system is working at the maximum possible rate configured, ATPC is enabled and no output power regulation is done by TAPS. On the other hand, if the system is not working at the maximum possible rate configured, TAPS is enabled and the output power is regulated to minimize the transmit power in relation to the actual traffic rate. If the system is running at the lowest configured rate, the power is slowly increased to avoid link loss. Based on the increases and decreases in near-end transmit power, higher or lower profile is requested by the far-end NE, which results in a higher or lower traffic rate, respectively.

Note: TAPS is a licensed feature.

ATPC has to be configured in order to enable TAPS.

3.4 Dual Carrier Modem

With the modem MMU 1002/1004 in MINI-LINK 6600, and the compact nodes MINI-LINK 6651/1, MINI-LINK 6651/2, and MINI-LINK 6651/4, dual carrier modems are introduced.

A dual carrier modem can be configured in different ways:

— Dual Directions

When configured as a 1+0 dual directions link, each carrier has its own connection to a switchport and both carriers can optionally be configured as a Hybrid radio link, that is combining TDM and Ethernet traffic on the same radio path. In MINI-LINK 6651/1 dual directions is not available.

— Bonded Single Direction

When configured as a 2+0 Radio Link Bonding (RLB) single direction link, the capacity of the two carriers are combined and only one switchport is used. In this configuration the combined capacity of the two can optionally be configured as a Hybrid Radio Link.

When this is configured with one MMU 1002/1004 or one MINI-LINK 6651, RAU and antenna protection is achieved.

When this is configured with two MMU 1002/1004s, MMU, RAU, and antenna protection is achieved, that is, Equipment Protection (EQP).

— Protected Single Direction

When configured as a 1+1 Radio Link Protection (RLP) single direction link, both carriers are used for Radio Link path protection, and a single switchport is used.

When this is configured with one MMU 1002/1004 or one MINI-LINK 6651, RAU and antenna protection is achieved.

When this is configured with two MMU 1002/1004, MMU, RAU, and antenna protection is achieved, that is, EQP.

- **Note:** EQP is not applicable for MINI-LINK 6651.
- MMU 1002/1004 is described in detail in Section 15.12 on page 195.
- **Note:** When having two MMU 1002/1004, 2+2 and 4+0 modes are also available, see Section 3.8.2 on page 24.

3.5 Radio Link Bonding

Radio Link Bonding (RLB) requires a license.

RLB uses multiple radio paths in a single direction to increase capacity. With RLB, a single port on the Ethernet switch is connected to an MMU/modem. The MMU/modem has an internal function to distribute the incoming Ethernet and PDH traffic between the available packet links.

The PDH traffic is prioritized and the remaining capacity is used for the packet links.





Figure 16 MINI-LINK 6691, 6692, 6693, and 6694, Hybrid Radio Link with RLB, Frequency Diversity



Figure 17 MINI-LINK 6651/1, 6651/2, and 6651/4 Hybrid Radio Link with RLB, Frequency Diversity

3.6 Hierarchical Radio Link Bonding

Hierarchical Radio Link Bonding (RLB) requires a license.

hRLB is a Layer 1 (L1) technology which is used for bonding separately configured Radio Link Terminals (RLTs) into one hRLB interface to achieve a higher radio link capacity. It can also be used for bonding an E-Band radio link with a traditional RLT.

hRLB requires a dedicated resource. In case of NPU1005, RLTs are configured separately on the MMU and the hRLB interface is configured on the NPU. MINI-LINK 6651/4 is a stand-alone product that supports hRLB.

Different configurations are available for the MINI-LINK 6600 family. For details on supported configurations, see MINI-LINK 6600 R1 Compatibility.

Bonding an E-Band radio link with a traditional RLT creates a Multi-Band Booster scenario and requires a MINI-LINK 6352. The MINI-LINK 6352 can connect to the NPU 1005 directly using a LAN interface or through an ETU 1002. For details on Multi-Band Booster supported hRLB configurations, see MINI-LINK 6600 R1 Compatibility.

3.7 Multi-Band Booster

Multi-Band Booster increases the performance of microwave backhaul, and is a tool that can increase network capacity up to tenfold. It supports flexible bonding of different carriers and frequency band combinations. Multi-Band Booster provides more efficient use of diverse backhaul spectrum assets, using higher frequencies over much wider geographical areas.

The main principle for Multi-Band Booster is to use two radio links in different frequency regions, one low frequency region and one high frequency region. Usually, the low frequency region carries also the TDM traffic and the high frequency region carries only Ethernet.

Figure 18 gives examples of multi-band possibilities.

 \geq





The low frequency band is named Ch1, and the high frequency band is named Ch2.

Ch1 – normally the link already in operation – has basic capacity and availability in the range of [99.99 ; 99.999]%. Since Ch1 gives high availability, TDM is transported on Ch1.

Ch2 – the added boosting link – adds high capacity in the range of [99.5; 99.95]% availability. This availability target is applicable for the capacity boost needed in non-fading conditions. Since Ch2 gives high capacity, Ethernet is transported on Ch2.

3.7.1 Multi-Band Booster with Policy-Based Forwarding

Policy-Based Forwarding is an enhancement to Multi-Band Booster by L1 packet bonding, which enables scheduling of the traffic with priority fields. This enables the user to select higher-priority traffic to be forwarded on trusted, high-availability links with a lower bandwidth and to forward the remaining traffic on higher-bandwidth links with lower availability. Available priority decisions are VLAN PCP, IP DSCP, and MPLS Traffic Class priority values. Forwarding decision can also be made using VLAN IDs in protected and unprotected mode. Protected mode is when the configured link fails, the VLAN is forwarded on the remaining link. In unprotected mode it is not moved to the remaining link. By default PCP values 5 through 7 are mapped to the master interface and 0 through 4 are mapped to the member interface.

Based on the actual usage of a local WAN interface, VLANs can be dynamically allocated to increase local link utilization.

Dynamic VLAN Mapping

VLANs forwarded using Policy-Based Forwarding can be measured using automatic VLAN allocation in the L1 bonding configuration page. This enables collection of bandwidth utilization data of the VLANs configured and also the current utilization of the master interface. If the utilization of the master interface is below 60%, then whole VLANs are mapped using best fit algorithm to reach 60% utilization. If the link utilization goes above 80%, then some of the dynamically mapped traffic is removed from the master interface to go below the 80% threshold. After a VLAN has been moved back, then the PBF rules are used to make forwarding decisions.

Dynamic PCP Mapping

When using Dynamic Priority Code Point (PCP) Mapping, the bandwidth utilization of the traditional link is measured automatically and traffic is moved to protect high priority traffic and enhance bandwidth utilization. If enabled, bandwidth utilization is measured and PCP mappings are assigned automatically, overriding the manual configuration. Bandwidth measurements are done on the PCP-level and the existing priority values (by default, PCP 5–7 for high-priority traffic and PCP 0–4 for low-priority traffic) are used.

Dynamic allocation can be either PCP-based or VLAN-based. So when Dynamic PCP is used, the dynamic VLAN must be disabled beforehand.

If Dynamic PCP Mapping is enabled, after a 30 seconds measurement period, a mapping action is done based on the gathered utilization data. If the measured utilization of the traditional link is below 60%, traffic with the highest PCP values is moved from the E-Band link to the traditional link until 80% of utilization is achieved. If the measured utilization of the tradition of the traditional link exceeds 80%, traffic with the lowest PCP values is moved from the traditional link to the E-Band link until the utilization of the traditional link to the E-Band link until the utilization of the traditional link drops to 80%. When the utilization of the traditional link exceeds 80%, or if more than 30 seconds have passed since the last mapping action, the 30 seconds measurement period is not needed.

3.7.2 Multi-Band Booster with Hierarchical Radio Link Bonding

The Multi-Band Booster function can be enhanced by using Hierarchical Radio Link Bonding (hRLB). Multi-Band Booster can bond two radio links that support different frequencies to boost the capacity of the complete link. hRLB can combine E-Band high frequency links with traditional standard bands.

Bonding an E-Band radio link with a traditional RLT creates a Multi-Band Booster scenario and requires a MINI-LINK 6352.

The MINI-LINK 6352 can connect to the NPU 1005 directly using a LAN interface or through an ETU 1002. It is supported in the same direction or in two directions.

Alternatively, the MINI-LINK 6352 can connect to a MINI-LINK 6651/4. The MINI-LINK 6651/4 bonds the traffic of a traditional and an E-Band radio link using MINI-LINK 6352. It is supported in the same direction or in two directions.

The Multi-Band Booster related configurations are described in the Configuring Multi-Band Booster with Hierarchical Radio Link Bonding (hRLB) document in the **Operation and Maintenance** folder.

The supported hRLB configurations are listed in MINI-LINK 6600 R1 Compatibility.

3.8 Radio Link Equipment and Propagation Protection

For Ethernet traffic, Ericsson recommend 2+0 Radio Link Bonding (RLB) with graceful degradation meaning that all available bandwidth is used and if one link experience traffic disturbance the capacity goes down to what the corresponding 1+0 configuration can handle. By using 2+0 RLB configurations, all hardware and spectrum is utilized to its maximum in normal case. It is important to note that it requires some configuration on QoS to decide which traffic gets priority in case of link failure.

For TDM traffic (E1/DS1 or STM-1) that needs high availability (that is, high capacity links or critical links for other parts of the network), 1+1 Radio Link Protection (RLP) configurations are recommended when using a limited spectrum, and 2+0 RLB configurations are recommended when fully utilizing spectrum.

The drawback with 1+1 RLP configurations is that they always have a higher cost than 1+0, by using redundant hardware or diversity installation.

The drawback with 2+0 RLB configurations is that they always consume more bandwidth than 1+0, since the basic idea is to transport traffic over more than one path.



Figure 19 Radio Link Protection Modes

How to choose the optimal configuration is dependent of availability requirements, performance needed by the specific hop situation (hop length, rain zones, fading situation, installation cost, site rent) as well as obtaining a cost-efficient solution.

When planning and configuring this, it is important to consider all means of link utilization and protection, for example RLB (see Section 3.5 on page 17) and Cross Polarization Interference Canceller (XPIC) (see Section 3.9 on page 28).

There are three major protection methods:

- Equipment Protection
 - Tx Equipment Protection
 - Rx Equipment Protection
- Radio Path Protection with Space Diversity
- Radio Path Protection with Frequency Diversity

Equipment Protection may be used on radio links where requirements on availability are high or repair time is assumed to affect the continuation of service.

Radio Path (propagation) Protection may be used on radio links where fading due to meteorological or ground conditions makes it difficult to meet the required transmission quality.

3.8.1 One Dual Carrier Modem Configurations

A Radio Terminal with one dual carrier modem (MINI-LINK 6691/6692/6693/6 694/6654/6655 with one MMU 1002/1004, MINI-LINK 6651/1, MINI-LINK 6651/2, or MINI-LINK 6651/4) can be configured for the following:

- 1+1 Radio Link Protection (RLP)
- 2+0 Radio Link Bonding (RLB)

3.8.1.1 1+1 Radio Link Protection

1+1 RLP gives the RAU and the antenna equipment protection.

1+1 RLP can be with or without space diversity. When using two antennas, they can be placed for space diversity with a mutual distance where the impact of selective fading (multipath) is reduced. With one (or two co-located) antenna, space diversity cannot be achieved.

Note: Space diversity is a counter measure against multipath, and is most efficient for frequencies up to 15 GHz.

1+1 RLP uses hot standby. One transmitter is working while the other one, tuned to the same frequency, is in standby. It is not transmitting but ready to transmit if the active transmitter malfunctions. Both RAUs receive signals.

Radio Path Protection

This protection case involves a Diversity selection in each MMU, providing hitless and error free traffic selection in case of radio channel degradation.

The Diversity selection works autonomously and is controlled by the protection logic in the active MMU Rx.

The Diversity selection reacts on quality measures from the two traffic streams. The selection is performed hitless.

3.8.1.2 2+0 Radio Link Bonding

2+0 RLB gives the RAU and the antenna graceful degradation.

2+0 RLB can use one or two frequencies. When using two different frequency channels (ACCP) the impact of selective fading (multipath) is reduced. When using one frequency channel (CCDP) XPIC must be used to achieve full performance.

Note: Frequency diversity is a counter measure against multipath, and is most efficient for frequencies up to 15 GHz.

2+0 RLB is most spectrum efficient when used together with XPIC (see Section 3.5 on page 17).

3.8.2 Two Single or Dual Carrier Modems Configurations

A Radio Terminal with two dual carrier modems (MINI-LINK 6691/6692/6693/6694/6654/6655 with two MMU 1002/1004) can be configured for the following:

- 1+1 Radio Link Protection (RLP) with Equipment Protection (EQP)
- 2+0 Radio Link Bonding (RLB) with EQP
- 4+0 RLB with EQP
- 2+2 RLP with EQP

A Radio Terminal with two single carrier modems (MINI-LINK 6691/6692/6693/6694/6654/6655 with two MMU 1001) can be configured for the following:

- 1+1 Radio Link Protection (RLP) with Equipment Protection (EQP)
- 2+0 Radio Link Bonding (RLB) with EQP

To achieve EQP, the two modems must be positioned in adjacent slots with 2-BPI connection as specified in the installation documents.

MMU Equipment Protection

This protection case involves the Ethernet or TDM traffic switching in the backplane. Tx and Rx traffic is switched from active MMU to standby MMU in case of radio link malfunction.

The NPU monitors the alarms generated in the RAU and MMU, and also controls the switches appropriately.

3.8.2.1 1+1 RLP with EQP

1+1 RLP with EQP gives the MMU, RAU, and antenna equipment protection.

1+1 RLP with EQP can be with or without space diversity. When using two antennas, they can be placed for space diversity with a mutual distance where the impact of selective fading (multipath) is reduced. With one (or two co-located) antenna, space diversity cannot be achieved.

Note: Space diversity is a counter measure against multipath, and is most efficient for frequencies up to 15 GHz.

1+1 RLP with EQP uses hot standby. One transmitter is working while the other one, tuned to the same frequency, is in standby. It is not transmitting but ready to transmit if the active transmitter malfunctions. Both RAUs receive signals.

Radio Path Protection

This protection case involves a Diversity selection in each MMU, providing hitless and error free traffic selection in case of radio channel degradation.

The Diversity selection works autonomously and is controlled by the protection logic in the active MMU Rx.

The Diversity selection reacts on quality measures from the two traffic streams. The selection is performed hitless.

3.8.2.2 2+0 RLB with EQP

2+0 RLB with EQP gives the MMU equipment protection and the RAU and the antenna graceful degradation.

2+0 RLB with EQP can use one or two frequencies. When using two different frequency channels (ACCP) the impact of selective fading (multipath) is reduced. When using one frequency channel (CCDP), XPIC must be used to achieve full performance.

Note: Frequency diversity is a counter measure against multipath, and is most efficient for frequencies up to 15 GHz.

2+0 RLB with EQP is most spectrum efficient when used together with XPIC (see Section 3.5 on page 17).

It is possible to set up a second 2+0 RLB with EQP configuration. Using two dual carrier MMUs, dual Radio Link Bonding can be achieved.

3.8.2.3 4+0 RLB with EQP

4+0 RLB with EQP gives the MMU equipment protection and the RAU and the antenna graceful degradation.

4+0 RLB with EQP with 2×MMU 1002/1004s is an efficient way to achieve high capacity with graceful degradation as well as EQP.

4+0 RLB EQP can use one, two, or four frequencies. When using four different frequency channels (ACCP) the impact of selective fading (multipath) is reduced. When using two frequency channel (CCDP), XPIC must be used to achieve full performance. When using one frequency channel (spatial CCDP), Multiple-Input Multiple-Output (MIMO) must be used to achieve full performance.

Note: Frequency diversity is a counter measure against multipath, and is most efficient for frequencies up to 15 GHz.

4+0 RLB with EQP is more spectrum-efficient when used together with XPIC (see Section 3.5 on page 17).

4+0 RLB with EQP is the most spectrum-efficient when used together with MIMO (see Section 3.10 on page 36).

4+0 RLB EQP with MIMO uses the corresponding XPIC licenses.

3.8.2.4 2+2 RLP with EQP

2+2 RLP with EQP gives the MMU, RAU, and antenna equipment protection.

2+2 RLP with EQP can use one or two frequencies. When using two different frequency channels (ACCP) the impact of selective fading (multipath) is reduced. When using one frequency channel (CCDP), XPIC must be used to achieve full performance.

2+2 RLP with EQP can be with or without space diversity. When using two antennas, they can be placed for space diversity with a mutual distance where the impact of selective fading (multipath) is reduced. With one (or two co-located) antenna, space diversity cannot be achieved.

Note: Space diversity is a counter measure against multipath, and is most efficient for frequencies up to 15 GHz.

2+2 RLP with EQP is most spectrum efficient when used together with XPIC (see Section 3.5 on page 17).

3.8.3 1+1 RLP with Asymmetrical Power Splitters

In a 1+1 protection hop with asymmetrical radio paths, revertive switching assures that a switch is made from a path with higher attenuation back to the path with lower attenuation when the path with lower attenuation has been restored. A switch in a 1+1 protection hop could be triggered by, for example, a system restart.

Revertive switching is applicable for 1+1 RLP with asymmetrical power splitters.

Revertive Switching

When asymmetrical power splitters (-1 dB and -7 dB) are used in master/slave 1+1 RLP configurations, the active radio normally uses the -1 dB path of the splitter. The active radio remains in this path until a fault on the radio occurs or a manual switch of active radio is done.

If system service is performed, it is possible that the active radio is via the -7 dB path after a manual functional test of the system has been performed. In this scenario the system gain is decreased by 12 dB (6 dB on each side of the hop).

The same scenario occurs at a DC power off/on if the -7 dB path is chosen for the radio connected to the MMU in the lowest slot, since this MMU is always active at startup. If this setup is done at both near and far end, the total system gain loss is 12 dB (6 dB on each side of the hop).

Revertive Tx Switching makes it possible to manually set the preferred txRadio (txRadio1 or txRadio2) and thereby avoid a decrease in system gain.

3.9 Cross Polarization Interference Canceller (XPIC)

Cross Polarization Interference Canceller (XPIC) is used to transmit two radio carriers across the same frequency, increasing the link utilization. XPIC is not a form of protection.

In an XPIC configuration, there is a risk that both receivers lock to the same transmitter. To prevent this, the Carrier-ID is used to uniquely identify each carrier. A far-end receiver can only lock to a near-end transmitter with the same Carrier-ID. Therefore, the Carrier-ID must be set to different values for the horizontal and the vertical polarization, and have identical settings for the same polarization on both sides of the hop according to Figure 20.



Figure 20 Carrier-ID for XPIC

It is recommended to use one dual-polarized antenna instead of two single-polarized antennas.

3.9.1 XPIC Basics

Cross polarization techniques are used in microwave networks to increase the capacity by transmitting data with two different polarizations over the same frequency band: vertical and horizontal polarization.

On each side of the hop, there are two radios installed per antenna; one radio for each polarization. The two different data streams from these radios are combined and transmitted from one common antenna on the near-end side and received through one common antenna on the far-end side. The received data on the far-end side is then separated again based on the two polarizations, so that one radio receives the vertically polarized signals and the other radio receives the horizontally polarized signals.

Cross Polarization Interference Canceller (XPIC) is an algorithm used to reduce the Cross Polar Interference (XPI) that occurs between the two polarizations when transmitted over the hop. The XPI of a system is represented by the XPI value (measured in dB), which is the sum of all possible factors that can cause XPI. For example, the following factors can cause cross polar interference:

- The channel conditions for each polarization.
- The alignment and rotation of the installed antennas.
- Variations and differences in the output power for each polarization.
- Cross Polar Discrimination (XPD), which is the ability of each antenna to isolate the two separate polarizations. XPD is relative to the phase angle (ϕ) between the polarizations.
 - **Note:** The XPD value is a characteristic of the antenna and does not change with the antenna alignment or antenna rotation.



Figure 21 Phase Angle (ϕ) Between the Vertical and Horizontal Polarizations

The XPI value can be adjusted by rotating one of the antennas. This changes the relative phase angle ($\psi)$ between them.

2



Figure 22 Cross Polar Interference - Mechanical Factors ($\phi)$ and Antenna Rotation ($\psi)$

Accepted Levels of Cross Polar Interference In the System

The rage of accepted XPI values for achieving a good radio link performance can be seen in Figure 23. The recommended XPI range for best radio link performance is 24–30 dB. Due to different channel conditions, the XPI value can have natural variations. This is considered in the XPI range recommendation.

Note: For E-band 80 GHz installations using MINI-LINK 6363 radios, the corresponding values are 20–30 dB (AGC 0.5–0.75 V).



Figure 23 Recommended Range of Cross Polar Interference Levels

Signal to Noise Ratio

XPI means that one polarization leaks onto the other, so that the receiver of the vertically polarized signals also receives levels of the horizontally polarized signals, and the other way around. The undesired signals are referred to as either internal co-channel interference or noise contribution.

Signal to Noise Ratio (SNR) represents the quality of the signal. SNR compares the level of desired signals to the level of undesired signals (the level of total noise).

In an ideal situation where there is no XPI and the isolation is 100% between the polarizations, the SNR is, in theory, the difference between the Received Signal Level (RSL) and the thermal noise floor.



Figure 24 Ideal Situation with 100% Isolation Between the Polarizations

In a system with a high level of XPI, the desired cross polar signals get suppressed, resulting in a low SNR. This leads to the XPIC function receiving a weaker cross polar signal with a higher noise contribution in the interference cancellation process.

If the XPI value is near or equal to the SNR Main value, the strength of the cross polar signal becomes below the noise floor value and the XPIC function can no longer identify the cross polar signals. This can lead to impaired performance.



Figure 25 High XPI Values Resulting in Low SNR Values

In a system with a low level of XPI, the desired cross polar signals get less suppressed, resulting in a higher SNR. The XPIC function receives a stronger cross polar signal with less noise contribution in the cancellation process. This condition results in the best performance.



Figure 26 Lower XPI Values Resulting in Higer SNR Values

3.9.2 XPIC Recovery and Restore

When a faulty (horizontal or vertical) receiver of one Carrier Termination (CT) prevents the received signal to be sent from one polarization to the other, then XPIC cannot recover the original signal. This means that both the horizontal and the vertical links are unavailable in the direction toward the faulty receiver, even if only one of the receivers is faulty.

XPIC Recovery

XPIC Recovery is the corrective procedure to recover the data traffic on the faultless channel. XPIC Recovery must be enabled by the user.

XPIC Recovery is triggered either by user actions or by faults, and consists of the following:

- Switching off the two transmitters of the faulty CT.
- Disabling the four XPIC cancellers, both near-end and far-end.

XPIC Restore

XPIC Restore is the procedure to re-configure XPIC on both CTs. XPIC Restore consists of re-enabling all XPIC cancellers and turning all transmitters on. Thanks to this, both cross-polar links (horizontal and vertical) are fully recovered and the traffic is error free.

Considering the example above, once the recovery procedure is triggered, both receivers stop generating XPI. As a result, the traffic over the faultless radio channel is error free. After the fault has been repaired, the restore is triggered, either manually or automatically.

3.9.3 2+0 RLB with XPIC Configuration

The 2+0 RLB with XPIC Radio Link configuration consists of the following equipment on one side of the hop:

 One 1002/1004 (MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, MINI-LINK 6694, MINI-LINK 6654, and MINI-LINK 6655)

or

One MINI-LINK 6651/1, one MINI-LINK 6651/2, or MINI-LINK 6651/4

- Two radios
- One integrated dual-polarized antenna

It is possible to use Adaptive Modulation in this configuration.







Figure 28 2+0 RLB with XPIC Configuration with MINI-LINK 6651/1, MINI-LINK 6651/2, or MINI-LINK 6651/4

Figure 27 and Figure 28 show only one direction of transmission (from left to right), but the transmission is symmetrical in both directions.

No XPIC cross-cable is needed, since the connection is done internally.

3.9.4 Dual (1+0) with XPIC Configuration

The dual (1+0) with XPIC Radio Link configuration consists of the following equipment on one side of the hop:

 One MMU 1002/1004 (MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, MINI-LINK 6694, MINI-LINK 6654, and MINI-LINK 6655)

or

One MINI-LINK 6651/2 or one MINI-LINK 6651/4

- Two radios
- One integrated dual-polarized antenna

It is possible to use Adaptive Modulation in this configuration.



Figure 29 Dual (1+0) with XPIC Configuration with MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, MINI-LINK 6694, MINI-LINK 6654, and MINI-LINK 6655



Figure 30 Dual (1+0) with XPIC Configuration with MINI-LINK 6651/2 or MINI-LINK 6651/4

Figure 29 and Figure 30 show only one direction of transmission (from left to right), but the transmission is symmetrical in both directions.

No XPIC cross-cable is needed, since the connection is done internally.

3.9.5 2+0 RLB EQP with XPIC Configuration

The 2+0 RLB EQP with XPIC Radio Link configuration consists of the following equipment on one side of the hop:

Two MMU 1002/1004s (MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, MINI-LINK 6694, MINI-LINK 6654, and MINI-LINK 6655)

or

Two MMU 1001s (MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, MINI-LINK 6694, MINI-LINK 6654, and MINI-LINK 6655)

- Two radios
- One integrated dual-polarized antenna
- XPIC cable

It is possible to use Adaptive Modulation in this configuration.



Figure 31 2+0 RLB EQP with XPIC Configuration with MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, MINI-LINK 6694, MINI-LINK 6654, and MINI-LINK 6655

Figure 31 shows only one direction of transmission (from left to right), but the transmission is symmetrical in both directions.

It is possible to set up a second 2+0 RLB EQP with XPIC configuration. Using two MMU 1002/1004s, dual Radio Link Bonding with Equipment Protection and XPIC can be achieved. When configuring two 2+0 RLB EQP links using two adjacent MMU boards, XPIC can be configured on both RLT1 and RLT 2.

3.10 Multiple-Input Multiple-Output (MIMO)

Multiple-Input Multiple-Output (MIMO) is a method for increasing the capacity of a radio link using multiple transmission and receiving antennas, to exploit multiple propagation.

2

In MINI-LINK 6600 products, the support of MIMO increases the spectral efficiency by utilizing the available spectrum with multiple carriers in the same frequency channel.

 $4x4\ \text{MIMO}$ provides up to four times more capacity and uses the combination of the following modes:

- 2x2 spatial MIMO with antenna separation
- 2x2 polarization MIMO with Cross Polarization Interference Canceller (XPIC)



Figure 32 4x4 MIMO

2x2 MIMO provides up to two times more capacity and uses the following mode:

- 2x2 spatial MIMO with antenna separation





The basis of a fully functional MIMO system is to exchange received baseband information between all involved carriers, for cancellation of the internal co-channel interference. All MMUs must be active in receiving direction, Rx, in order to achieve transmission over the hop. This will limit the EQP function.

When using EQP in a MIMO configuration, switching due to faults in TX direction will lead to graceful degradation and switching due to faults in RX direction may lead to traffic loss.

3.11 Carrier Aggregation (CA)

Carrier Aggregation (CA) aggregates existing channels to wider channels in a cost-effective way. CA has the following advantages:

- Doubled capacity with one radio
- No Integrated Power Splitter (IPS) is required

If a MINI-LINK 6600 product is equipped with MMU 1002 or MMU 1004 dual carrier modems, then it supports CA when connected to a MINI-LINK 6364 or a MINI-LINK 6365 radio.

Both carriers from the modem are placed on the same intermediate frequency (IF) port and cable, using different IF frequencies. The radios filter out each carrier and mix them to a radio frequency (RF) within a radio sub-band.

The CA support is available for a 2+0 RLB scenario only in MINI-LINK 6600 shorthaul products—that is, MINI-LINK 6691, 6692, 6693, 6694, 6654, and 6655.



Figure 34 Carrier Aggregation (CA)

Note: On the MMU, port RAU 1 must be used to apply CA. The second port, RAU 2, must be disconnected.

CA is allowed only with symmetrical Channel Separation (CS) on both channels, and symmetrical configuration on both near end and far end.

CA is controlled by license.

3.12 Multi-Layer Header Compression (MLHC)

Multi-Layer Header Compression (MLHC) improves the efficiency of transportation of Ethernet traffic over Radio Links by compressing Ethernet/MPLS (Layer 2), IP (Layer 3), and UDP (Layer 4) headers. MLHC increases the maximum throughput at WAN interface by reducing the size of the Ethernet frames as they are transmitted over the Radio Link. The size is not configurable as it only depends on the type of header of the Ethernet frame. See Figure 35.

Supported headers:

Ethernet



- Untagged frames
- Customer Bridge
- Provider Bridge
- MPLS
 - Up to three labels per frame
- IP
 - IPv4
 - IPv6
- UDP

MLHC compresses frames with combinations of the above listed headers. Also, compression of L2VPN and L3VPN are supported.

JA	+ -	ray 1	/ -	11 1/4/15				
					VUTIEA			
et Prov	/ider Bridge, A	Iternativ	/e 2					
SA	S-tag/C-tag C	C-tag T	-/L	MPLS	MPLS	MPLS		
	L							
V IP o	ver MPLS ove	r Ethern	et					
			- / 1					
SA	S-tag/C-tag (-tag I	/L	MPLS	MPLS		IPv4/IPv6 Header	
V: Ethe	ernet over MPL	_S over	Ethe	ernet				
SA	S-tag/C-tag i	C-tag T	7/L	MPLS	MPLS	MPLS	Content	
	±			ì		L J		
	DA	SA	S-tag	g/C-tag	C-tag	T/L	IPv4/IPv6 Header	UDP
	•			+				'
V. Ethe	arnet PW over	MPLS	wor	Ethern	≏t			
v. ∟uic								
SA	S-tag/C-tag C	C-tag T	7/L	MPLS	MPLS		CW Content	
				+				,
	DA	SA	S-tag	g/C-tag	C-tag	T/L	IPv4/IPv6 Header	UDP
				+		-		
	et Prov SA N: IP o SA N: Ethe SA	et Provider Bridge, A SA S-tag/C-tag C N: IP over MPLS ove SA S-tag/C-tag C N: Ethernet over MPL SA S-tag/C-tag C DA N: Ethernet PW over SA S-tag/C-tag C	et Provider Bridge, Alternativ SA S-tag/C-tag C-tag 1 N: IP over MPLS over Ethern SA S-tag/C-tag C-tag 1 N: Ethernet over MPLS over SA S-tag/C-tag C-tag 1 DA SA N: Ethernet PW over MPLS over SA S-tag/C-tag C-tag 1	et Provider Bridge, Alternative 2 SA S-tag/C-tag C-tag T/L N: IP over MPLS over Ethernet SA S-tag/C-tag C-tag T/L N: Ethernet over MPLS over Ether SA S-tag/C-tag C-tag T/L DA SA S-tag N: Ethernet PW over MPLS over SA S-tag/C-tag C-tag T/L	et Provider Bridge, Alternative 2 SA S-tag/C-tag C-tag T/L MPLS N: IP over MPLS over Ethernet SA S-tag/C-tag C-tag T/L MPLS N: Ethernet over MPLS over Ethernet SA S-tag/C-tag C-tag T/L MPLS DA SA S-tag/C-tag N: Ethernet PW over MPLS over Ethern SA S-tag/C-tag C-tag T/L MPLS	et Provider Bridge, Alternative 2 SA S-tag/C-tag C-tag T/L MPLS MPLS N: IP over MPLS over Ethernet SA S-tag/C-tag C-tag T/L MPLS MPLS N: Ethernet over MPLS over Ethernet SA S-tag/C-tag C-tag T/L MPLS MPLS DA SA S-tag/C-tag C-tag N: Ethernet PW over MPLS over Ethernet SA S-tag/C-tag C-tag T/L MPLS MPLS	et Provider Bridge, Alternative 2 SA S-tag/C-tag C-tag T/L MPLS MPLS MPLS N: IP over MPLS over Ethernet SA S-tag/C-tag C-tag T/L MPLS MPLS MPLS N: Ethernet over MPLS over Ethernet SA S-tag/C-tag C-tag T/L MPLS MPLS MPLS DA SA S-tag/C-tag C-tag T/L N: Ethernet PW over MPLS over Ethernet SA S-tag/C-tag C-tag T/L MPLS MPLS MPLS DA SA S-tag/C-tag T/L MPLS MPLS MPLS	et Provider Bridge, Alternative 2 SA S-tag/C-tag C-tag T/L MPLS MPLS MPLS N: IP over MPLS over Ethernet SA S-tag/C-tag C-tag T/L MPLS MPLS MPLS IPv4/IPv6 Header N: Ethernet over MPLS over Ethernet SA S-tag/C-tag C-tag T/L MPLS MPLS MPLS Content DA SA S-tag/C-tag C-tag T/L IPv4/IPv6 Header N: Ethernet PW over MPLS over Ethernet SA S-tag/C-tag C-tag T/L MPLS MPLS MPLS Content DA SA S-tag/C-tag C-tag T/L IPv4/IPv6 Header

Figure 35 Compression of Headers in Different Layers

Note: 1588 frames are not compressed.

At each side of the Radio Link there is an MLHC compression pair (one compressor and one decompressor), implementing the header compression functionality. The



compressor at one side of the Packet Radio Link sends compressed traffic frames to the corresponding decompressor at the other side of the hop. See Figure 36.

Figure 36 Compression and Decompression Functionality

3.13 AES Encryption Over-the-Hop

AES encryption over-the-hop is a radio link integrity feature. The encryption is to disguise if a radio link is used or not. The encryption is compliant to AES-256-CTR.

Note: Encryption is only possible on MMU 1002/1004 configured as 1+0 or 2+0 (on one modem).

Every carrier termination (CT) has two separate session keys for communication over the radio link:

- one for Near End \Rightarrow Far End
- one for Far End \Rightarrow Near End

Every CT can therefore be encrypted independently of each other. The Tx and Rx are also encrypted independently of each other.

AES Encryption over-the-hop is controlled by license. The feature is disabled and hidden until the proper licenses are installed on the node.



4 Ethernet Overview

MINI-LINK 6600 is targeting multiple applications and network environments with the embedded Ethernet capabilities. MINI-LINK 6600 therefore provides flexibility and supports a large number of Ethernet services and features.

For details on the required HW for specific functions, see MINI-LINK 6600 R1 Compatibility, Reference [1].

4.1 Ethernet Services

Ethernet services according to MEF (Metro Ethernet Forum) specifications are supported. The following MEF standards are supported: MEF 2, MEF 9, MEF 10, and MEF 14.

Figure 37 shows a basic model for Ethernet services. The Ethernet service is provided by Metro Ethernet Network (MEN) provider. The Customer Edge (CE) and MEN exchange service frames across the User Network Interface (UNI).



Figure 37 Ethernet Service Model

Based on Ethernet Virtual Connections (EVCs), the following service types are supported:

- Point-to-Point EVC:
 - Ethernet Private Line (EPL) service
 - Ethernet Virtual Private Line (EVPL) service
- Multipoint-to-Multipoint EVC:

- Ethernet Private LAN (EPLAN) service
- Ethernet Virtual Private LAN (EVPLAN) service

5 Ethernet Bridging/Switching

MINI-LINK 6600 has an embedded Ethernet bridge that can be configured for customer mode (IEEE802.1Q-2005) or provider mode (IEEE802.1ad-2005). MINI-LINK 6600 also supports a MAC bridge/switch according to IEEE802.1D-2004.

The following features are supported:

- Store and forward switching
- VLAN ID (VID) and priority tagging
- Port-based VLAN (up to 1,000 VLANs in the range of 1-4094)
- MAC address table (32,000 entries)
- Self learning with aging of MAC addresses
- Static L2 Unicast routes
- Static L2 Multicast routes
- Head of line blocking prevention (HOLB)
- Rapid Spanning Tree Protocol (RSTP)
- Multiple Spanning Tree Protocol (MSTP)
- Link Aggregation Group (LAG)
- Link Aggregation Control Protocol (LACP)
- SNMP-based management interface, for example, DCN over VLAN
- QinQ
- Many-to-one/All-to-one Mapping for Provider Mode and for QinQ
- Internet Group Management Protocol (IGMP) snooping
- Multicast Listener Discovery (MLD) snooping
- Multiple VLAN configuration

|

5.1 L2 Connectivity Provisioning for Customers over a Provider Network

The provider network functionality enables transport of customer Ethernet traffic over an L2 network. Upon entrance to a service provider network, a Service VLAN tag (S-tag) is added to the Ethernet frame. The S-tag contains the information required to provide a virtual separation of traffic from different customer domains. Once the frame has been transported through the service provider network, the S-tag is removed from the Ethernet frame at the service provider network egress point. In the service provider network it is treated as an ordinary VLAN Ethernet frame.

MINI-LINK 6600 supports the following two types of L2 connectivity provisioning:

- QinQ
- IEEE 802.1ad

For QinQ, the S-tag is similar to the Customer VLAN tag (C-tag) and includes the following:

- Tag Protocol Identifier (0×8100)
- Priority bits (for priority handling in the provider network)
- Drop Eligible Indicator (DEI) bit (for support of color marking).
- S-VLAN ID

QinQ enhances scalability by removing the 1000 VLANs limitation from Customer Mode (IEEE 802.1Q). In this case, the spanning tree of the customer network is proposed to be split into several spanning tree domains. The interconnection between the customer and provider allowed to be a single L2 link and special care of the loop-free L2 topology has to be taken.

For IEEE 802.1ad, the S-tag is similar to the C-tag and includes the following:

- Tag Protocol Identifier (0×88a8)
- Priority bits (for priority handling in the provider network)
- Drop Eligible Indicator (DEI) bit (for support of color marking).
- S-VLAN ID

Provider Bridge enhances scalability by removing the 1000 VLANs limitation from Customer Mode (IEEE 802.1Q) as well as the size limitation of the spanning tree since the number of hops in the service provider network is not a part of the calculation by the customer spanning tree.

MINI-LINK 6600 implements a Provider Edge Bridge which comprises a single S-VLAN component and one or more C-VLAN components.

 Customer Edge Port (CEP): A C-VLAN component Port on a Provider Edge Bridge that is connected to customer owned equipment and receives and transmits frames for a single customer

CEP is an external physical interface. It is a UNI interface and is named CE-UNI in MINI-LINK 6600.

— Customer Network Port (CNP): An S-VLAN component within a Provider Edge Bridge that receives and transmits frame for a single customer.

A CNP can be an external physical interface, or a virtual internal interface when connected to a Provider Edge Port (PEP). The CNP is a UNI interface, named CN-UNI in MINI-LINK 6600.

 Provider Network Port (PNP): An S-VLAN component Port on a Provider Bridge that transmits and receives frames for multiple customers.

A PNP is an external physical interface. It is an NNI interface, named I-NNI (PN) in MINI-LINK 6600.

 Provider Edge Port (PEP): A C-VLAN component Port within a Provider Edge Bridge that connects to a Customer Network Port and receives and transmits frames for a single customer.

The PEP is a virtual internal interface and the port parameters are set to fixed values and can be read from the management system.



Figure 38 Provider Bridge

5.2 Port Roles

Depending on bridge configuration (customer mode or provider mode) different switch port roles are available in MINI-LINK 6600.



UNI and I-NNI port roles are available in customer mode, while Customer Edge UNI (CE-UNI), Customer Network UNI (CN-UNI) and Provider Network NNI (I-NNI (PN)) are available in provider mode.

The UNI ports in customer mode and the CE-UNI and CN-UNI ports in provider mode are the interface between the customer and operator network.

The I-NNI and I-NNI (PN) ports are interfaces used within a transport network. The I-NNI ports in customer mode accept only C-tagged frames while the I-NNI (PN) ports in provider mode only accept S-tagged frames.

For I-NNI ports the TPID can be changed from the default value to the C-VLAN TPID 0x8100.

5.3 Supported Frame Sizes

Ethernet frames of different sizes are common in real networks and the frame size is often related to the application. A real-time application, for example, voice and video, that requires minimum latency typically uses small Ethernet frames, while a data application with non real-time characteristics uses the maximum frame size. MINI-LINK 6600 handles both small and large Ethernet frames as well as a mix.

All ports support 64 to 9,216 byte frame size.

5.4 Header Decoding

MINI-LINK 6600 supports header decoding for C-VLANs in customer mode and S-VLANs in provider mode. The EtherType value, also referred to as Tag Protocol Identifier (TPID), for C-VLANs is 0×8100. For S-VLANs the EtherType value is 0×88a8.

The EtherType value is unique for the entire network in customer and provider mode, respectively.

5.5 Multiple VLAN Configuration

VLAN operations on a range of VLANs are possible, as batch operations are supported.

2

6 Ethernet Security and Admission Control

MINI-LINK 6600 supports multiple security and admission control mechanisms to enable the operator to control the traffic flow into the network individually per port.

The security and admission control features are only available on a port configured as an L2 connection.

6.1 Frame Admittance

At the network edge it is possible, with the frame admittance feature, to select which frame types to admit and block. The following frame type options are available:

- Admit Only VLAN Tagged Frames
- Admit Only Untagged and Priority Tagged Frames
- Admit All Frames

The frame admittance settings for bridge ports are specified below.

Customer Mode

- UNI ports: configurable frame types with default value Admit All Frames
- I-NNI ports: non-configurable frame types set to Admit Only VLAN Tagged Frames

Provider Mode

— CE-UNI ports: non-configurable frame types set to Admit All Frames

Note: Tagged frames with EtherType other than 0×8100 are discarded.

- CN-UNI ports: configurable frame types with default value Admit All Frames
- I-NNI (PN) ports: non-configurable frame types set to Admit Only VLAN Tagged Frames

6.2 L2CP Frames Processing

MINI-LINK 6600 supports the following filtering behavior of Layer 2 control protocols:

Discard (IEEE 802.1ad/MEF 10.1)

- Peer (MEF 10.1)
- Forward Unchanged
- Tunneling

L2CP tunneling makes it possible to traverse the incoming Layer 2 protocol frames through the network. Tunneling can only be configured on one interface per network element. When using more tunnels in the same physical Layer 2 segment, different tunnel MAC addresses need to be used per tunnel.

The options above are configurable on UNI, CE-UNI, and CN-UNI ports. On I-NNI and I-NNI (PN) ports, the behavior depends on the protocol running in the node.

For all L2CP listed in the tables below, the default behavior on all port types is discard. This is in accordance with IEEE 802.1ad. The alternative behavior for the protocols supported in MINI-LINK 6600 is presented in the tables below.

MAC Address	Application	Tunnelable	Optional Configurable Behavior
01-80-C2-00-00-00	Customer Bridge Protocol Data Units (BPDUs)	Yes	Peer if enabled
01-80-C2-00-00-01	MAC Pause	No	None
01-80-C2-00-00-02 subtype 0x01	Link Aggregation Control Protocol (LACP)	Yes	Peer if enabled, Forward
01-80-C2-00-00-02 subtype 0x02	Link Aggregation Marker Protocol (LAMP)	Yes	Peer if enabled, Forward
01-80-C2-00-00-02 subtype 0x03	Link OAM	Yes	Peer if enabled, Forward
01-80-C2-00-00-02 subtype 0x04–0x09	Future slow protocols	Yes	None
01-80-C2-00-00-02 subtype 0x0A	Ethernet Synchronizati on Messaging Channel (ESMC)	No	Peer if enabled ⁽¹⁾
01-80-C2-00-00-03	802.1X authentication	Yes	None
01-80-C2-00-00-0D	Provider GVRP	Yes	None
01-80-C2-00-00-0E	Link Layer Discovery Protocol (LLDP)	Yes	Peer if enabled ⁽²⁾
01-80-C2-00-00-04	Future standard bridge -04	Yes	None
01-80-C2-00-00-05	Future standard bridge -05	Yes	None
01-80-C2-00-00-06	Future standard bridge -06	Yes	None
01-80-C2-00-00-07	Metro Ethernet Forum ELMI	Yes	None
01-80-C2-00-00-09	Future standard bridge -09	Yes	None

Table 3 Filtering Behavior of L2CP Protocols on UNI

3

MAC Address	Application	Tunnelable	Optional Configurable Behavior
01-80-C2-00-00-0A	Future standard bridge -0A	Yes	None
01-80-C2-00-00-0B	Future standard bridge -0B	Yes	None
01-80-C2-00-00-0C	Future standard bridge -0C	Yes	None
01-80-C2-00-00-10	All Bridges Protocol	Yes	None
01-80-C2-00-00-20	GARP Multicast Registration Protocol (GMRP)	Yes	None, Forward
01-80-C2-00-00-21	GARP VLAN Registration Protocol (GVRP)	Yes	None, Forward
01-80-C2-00-00-2x (x=2F)	Generic Attribute Registration Protocol (GARP)	Yes	None, Forward
01-80-C2-00-00-08	Provider Bridge Protocol Data Units (BPDUs)	Yes	None
01-80-C2-00-00-0F	Future standard bridge -0F	Yes	None

(1) Not supported if the interface is an hRLB group interface.

(2) : When LLDP is enabled on a port it is peered regardless of the configured L2CP behavior. The configured L2CP behavior is applied when LLDP is disabled.

Table 4	Filtering Behavior of L2CP Protocols on I-NNI
---------	---

MAC Address	Application	Alternative Behavior ⁽¹⁾
01-80-C2-00-00-00	Bridge Group Address (BPDUs)	Peer if enabled
01-80-C2-00-00-02 subtype 0x01	Link Aggregation Control Protocol (LACP)	Peer if enabled
01-80-C2-00-00-02 subtype 0x02	Link Aggregation Marker Protocol (LAMP)	Forward unchanged
01-80-C2-00-00-02 subtype 0x03	Link OAM	Peer if enabled
01-80-C2-00-00-02 subtype 0x0A	Ethernet Synchronization Messaging Channel (ESMC)	Peer if enabled
01-80-C2-00-00-0E	Link Layer Discovery Protocol (LLDP)	Peer if enabled
01-80-C2-00-00-0x (x=1,3F)	All other C-VLAN component reserved group addresses	None

(1) Not configurable

Table 5 Filtering Behavior of L2CP Protocols on CE-UNI

MAC Address	Application	Tunnelable	Optional Configurable Behavior
01-80-C2-00-00-00	Customer Bridge Protocol Data Units (BPDUs)	Yes	Forward
01-80-C2-00-00-01	MAC Pause	No	None


MAC Address	Application	Tunnelable	Optional Configurable Behavior
01-80-C2-00-00-02 subtype 0x01	Link Aggregation Control Protocol (LACP)	Yes	Peer if enabled, Forward
01-80-C2-00-00-02 subtype 0x02	Link Aggregation Marker Protocol (LAMP)	Yes	Peer if enabled, Forward
01-80-C2-00-00-02 subtype 0x03	Link OAM	Yes	Peer if enabled, Forward
01-80-C2-00-00-02 subtype 0x04–0x09	Future slow protocols	Yes	None
01-80-C2-00-00-02 subtype 0x0A	Ethernet Synchronizati on Messaging Channel (ESMC)	No	Peer if enabled ⁽¹⁾
01-80-C2-00-00-03	802.1X authentication	Yes	None
01-80-C2-00-00-0D	Provider GVRP	Yes	None
01-80-C2-00-00-0E	Link Layer Discovery Protocol (LLDP)	Yes	Peer if enabled ⁽²⁾
01-80-C2-00-00-04	Future standard bridge -04	Yes	None
01-80-C2-00-00-05	Future standard bridge -05	Yes	None
01-80-C2-00-00-06	Future standard bridge -06	Yes	None
01-80-C2-00-00-07	Metro Ethernet Forum ELMI	Yes	None
01-80-C2-00-00-09	Future standard bridge -09	Yes	None
01-80-C2-00-00-0A	Future standard bridge -0A	Yes	None
01-80-C2-00-00-0B	Future standard bridge -0B	Yes	None
01-80-C2-00-00-0C	Future standard bridge -0C	Yes	None
01-80-C2-00-00-10	All Bridges Protocol	Yes	None
01-80-C2-00-00-20	GARP Multicast Registration Protocol (GMRP)	Yes	None, Forward
01-80-C2-00-00-21	GARP VLAN Registration Protocol (GVRP)	Yes	None, Forward
01-80-C2-00-00-2x (x=2F)	Generic Attribute Registration Protocol (GARP)	Yes	None, Forward
01-80-C2-00-00-08	Provider Bridge Protocol Data Units (BPDUs)	Yes	None
01-80-C2-00-00-0F	Future standard bridge -0F	Yes	None

(1) Not supported if the interface is an hRLB group interface.

(2) : When LLDP is enabled on a port it is peered regardless of the configured L2CP behavior. The configured L2CP behavior is applied when LLDP is disabled.

MAC Address	Application	Tunnelable	Optional Configurable Behavior
01-80-C2-00-00-00	Customer Bridge Protocol Data Units (BPDUs)	Yes	Forward
01-80-C2-00-00-01	MAC Pause	No	None
01-80-C2-00-00-02 subtype 0x01	Link Aggregation Control Protocol (LACP)	Yes	Peer if enabled, Forward
01-80-C2-00-00-02 subtype 0x02	Link Aggregation Marker Protocol (LAMP)	Yes	Peer if enabled, Forward
01-80-C2-00-00-02 subtype 0x03	Link OAM	Yes	Peer if enabled, Forward
01-80-C2-00-00-02 subtype 0x04–0x09	Future slow protocols	Yes	None
01-80-C2-00-00-02 subtype 0x0A	Ethernet Synchronizati on Messaging Channel (ESMC)	No	Peer if enabled ⁽¹⁾
01-80-C2-00-00-03	802.1X authentication	Yes	None
01-80-C2-00-00-0D	Provider GVRP	Yes	None
01-80-C2-00-00-0E	Link Layer Discovery Protocol (LLDP)	Yes	Peer if enabled ⁽²⁾
01-80-C2-00-00-04	Future standard bridge -04	Yes	None
01-80-C2-00-00-05	Future standard bridge -05	Yes	None
01-80-C2-00-00-06	Future standard bridge -06	Yes	None
01-80-C2-00-00-07	Metro Ethernet Forum ELMI	Yes	Nonev
01-80-C2-00-00-09	Future standard bridge -09	Yes	None
01-80-C2-00-00-0A	Future standard bridge -0A	Yes	None
01-80-C2-00-00-0B	Future standard bridge -0B	Yes	None
01-80-C2-00-00-0C	Future standard bridge -0C	Yes	None
01-80-C2-00-00-10	All Bridges Protocol	Yes	None
01-80-C2-00-00-20	GARP Multicast Registration Protocol (GMRP)	Yes	None, Forward
01-80-C2-00-00-21	GARP VLAN Registration Protocol (GVRP)	Yes	None, Forward
01-80-C2-00-00-2x (x=2F)	Generic Attribute Registration Protocol (GARP)	Yes	None, Forward

Table 6 Filtering Behavior of L2CP Protocols on CN-UNI



MAC Address	Application	Tunnelable	Optional Configurable Behavior
01-80-C2-00-00-08	Provider Bridge Protocol Data Units (BPDUs)	Yes	Peer if enabled
01-80-C2-00-00-0F	Future standard bridge -0F	Yes	None

(1) Not supported if the interface is an hRLB group interface.

(2) : When LLDP is enabled on a port it is peered regardless of the configured L2CP behavior. The configured L2CP behavior is applied when LLDP is disabled.

Table 7	Filtering	Behavior	of L2CP	Protocols on	I-NNI (PN)
	i neering	Denavior		1 10100013 011	T 1 1

MAC Address	Application	Alternative Behavior ⁽¹⁾
01-80-C2-00-00-02 subtype 0x01	Link Aggregation Control Protocol (LACP)	Peer if enabled
01-80-C2-00-00-02 subtype 0x02	Link Aggregation Marker Protocol (LAMP)	Forward unchanged
01-80-C2-00-00-02 subtype 0x03	Link OAM	Peer if enabled
01-80-C2-00-00-02 subtype 0x0A	Ethernet Synchronization Messaging Channel (ESMC)	Peer if enabled
01-80-C2-00-00-08	Provider Bridge Group Address	Peer if enabled
01-80-C2-00-00-0E	Link Layer Discovery Protocol (LLDP)	Peer if enabled
01-80-C2-00-00-0x (1,37,9A)	All other S-VLAN component reserved group addresses	None

(1) Not configurable

6.2.1 L2CP Frames Processing on a Layer 1 Ethernet Service

On an L1 Ethernet Service, MINI-LINK 6600 supports the following filtering behavior of L2 control protocols:

- Discard (IEEE 802.1Q/IEEE 802.3/MEF 6.1)
- Peer (MEF 6.1)
- Forward Unchanged

The default and alternative behavior for the protocols supported in MINI-LINK 6600 are presented in Table 8.

Table 8	Filtering	Behavior of	of Reserved	Group	MAC Addresses.

MAC Address	Application	Default Behavior	Optional Configurable Behavior
01-80-C2-00-00-00	Bridge Group Address (BPDUs)	Forward unchang ed	None
01-80-C2-00-00-01	MAC Pause	Discard	None
01-80-C2-00-00-02 subtype 0x01	Link Aggregation Control Protocol (LACP)	Discard	Forward unchanged

MAC Address	Application	Default Behavior	Optional Configurat	ole Behavior
01-80-C2-00-00-02 subtype 0x02	Link Aggregation Marker Protocol (LAMP)	Discard	Forward unchanged	
01-80-C2-00-00-02 subtype 0x03	Link OAM	Discard	Peer if enabled	Forward unchange d
01-80-C2-00-00-02 subtype 0x04–0x09	All other slow Protocols multicast address	Forward unchang ed	None	
01-80-C2-00-00-02 subtype 0x0A	Ethernet Synchron ization Messaging Channel (ESMC)	Discard	Peer if enabled ⁽¹⁾	
01-80-C2-00-00-0 E	Link Layer Discovery Protocol (LLDP)	Forward unchang ed	Peer if enabled ⁽²⁾	Discard
01-80-C2-00-00-0x (x=3F)	All other reserved group addresses	Forward unchang ed	None	

(1) Not supported if the interface is an hRLB group interface.

(2) When LLDP is enabled on a port it is peered regardless of the configured L2CP behavior. The configured L2CP behavior is applied when LLDP is disabled.

6.3 White Lists

Individual white lists can be created per port to specify which source MAC addresses MINI-LINK 6600 can accept on a port. Frames with a source MAC address not on the list will be discarded.

6.4 Ethernet Loop Detection

Ethernet loop detection is used to detect Layer 2 Ethernet loops in the network by configuring Ethernet loop alarms.

There are two use cases for Ethernet loops:

- Ethernet loop

The NE itself is part of an Ethernet loop.

— Ethernet external loop

The NE itself is not part of an Ethernet loop, but is connected to a Layer 2 domain where an Ethernet loop occurs.

There are also two use cases regarding provider and customer relations:

Provider-only loops

Can be detected when NNI ports are used.

Provider-customer loops



The customer has redundant interconnection towards the provider network. Can be detected when both NNI and UNI ports are used.

In case a provider has redundant interconnection towards the customer, it is important to detect loops that can occur over the redundant connection. In such a case, configure also the UNI (CE-UNI or CN-UNI) ports for Ethernet loop detection. When Ethernet loop detection is configured on UNI (CE-UNI or CN-UNI) ports, the customer network receives traffic from outside of the customer network. To receive traffic from outside of the customer network, an agreement is needed between the network provider and the customer.

Note: Loop detection can be restricted to search for loops in one or a few VLANs. In this case, only these VLANs will be tested for loops.

It is recommended to use Ethernet loop detection together with storm protection to avoid flooding of the test frames, see Section 6.5 on page 53.

6.5 Storm Protection

Broadcast storm protection is a protection mechanism, which prevents that other parts or areas of the network are affected by traffic flooding from unicast, broadcast, or multicast traffic at a very high data rate.

Filters can be activated per port to prevent flooding of unwanted and hostile traffic. Individual filters are used for broadcast, multicast and destination lookup failure traffic. The filters are specified as frames per second. When the limit is reached, additional frames will be discarded until the frame rate is below the specified threshold.

Storm protection can be configured on all Ethernet bridge ports in provider bridge mode and customer bridge mode.

6.6 Port Blocking

The port blocking feature prevents forwarding of frames from a given ingress port to one or more egress ports.

6.7 MAC Address Learning

If MAC address learning is activated on the node it is possible, from one interface, to flood the learning table. A flooded learning table prevents learning on all other ports. By limiting the MAC addresses per port it is possible to prevent MAC address flooding.

MAC address learning can be disabled on a per port basis or on a per VLAN basis.

 \geq

7 Quality of Service

Quality of Service (QoS) of a network deals with the ability of the network to provide transport services suitable for the applications using the network itself. The Quality of Service support in MINI-LINK 6600 is ensured by classifying packets and handling their transmission through the network according to the classification results of each packet.

The following three main parameters affect the quality of the transmission:

- Packet loss: packets lost in the network.
- Delay: the time it takes for a packet to traverse through the network from sender to receiver.
- Jitter: variation in the delay within the same traffic flow.

The different service types (video, voice, data) have different QoS requirements.

In order to meet the different QoS requirements, QoS implementation is based on priority assignment. This supports a number of different priorities in accordance with the Ethernet or IP priority scheme. Ethernet priority scheme conforms to the IEEE 802.1Q specification and supports up to 8 priority levels usually served in Strict Priority. IP priority scheme relies on a higher number of priority levels served according to their per-hop-behavior defined for DiffServ.

Both priority schemes have traffic classification where each packet is associated to a traffic class (in a small or large set), instead of an individual flow. There is not a single flow, but a set of indistinct flows with the same characteristics. Each system in the network can be configured to differentiate traffic based on the traffic class. Each traffic class can be managed differently to separate the high priority traffic from the low priority traffic.

Different features contribute to provide QoS and allow the system to resolve bottlenecks and to guarantee higher priority traffic with respect to lower. The features are listed below.

- Policing: the policer manages traffic matching a certain level of service (usually a bandwidth value). The feature is applied to make sure a subscriber is using the allowed amount of resources in a network. A policer usually performs a configurable action on non-compliant traffic, for example, discard or de-prioritize.
- Queue Management: queue management operates at queue level. It drops or marks frames entering the queue before the queue is full. Typically, queue management operates with a probabilistic approach, dropping or marking frames depending on the queue usage (or its evaluation) and a drop-probability.

 Traffic Scheduler: the traffic scheduler typically operates at port level on the queues associated to that port. It manages the forwarding of different packets, based on their service class, using queue management and various scheduling algorithms. The scheduler ensures that frame delivery corresponds to the QoS parameters for each flow.

An overview of the execution order for different QoS functions is illustrated in Figure 39.



Figure 39 QoS Execution Order

Frames can be dropped at the ingress or egress side by the different QoS functions described in this section.

Note: IEEE 802.3x-based congestion handling mechanism does not take into account the usage of different priority levels. The reception of a PAUSE message triggers a network element to halt all Ethernet frames. This means that IEEE 802.3x-based congestion handling is not suitable for networks configured for different priority levels and transmission of real-time traffic.

7.1 Classification and Tagging

7.1.1 VID Handling

Frames entering UNIs, CN-UNIs, and CE-UNIs are discarded or accepted depending on the information in the frame header and on the parameters set on

the port. Before entering the bridge, the frame header can be modified by adding or changing the VLAN tag. The VID is often added to the VLAN tag together with a priority.

VLAN modifications are done if the frame complies with the following conditions:

- The source MAC address of the incoming frame does not belong to the reserved group MAC address
- The format of the incoming frame is supported by the acceptable frame type configured on the port
- The port receiving the frame belongs to the member set of the VLAN carried by the frame (when applicable)

If these conditions are met, the VLAN modifications below are performed.

Customer Mode

A UNI is a C-tag aware interface. Incoming C-tagged frames are accepted. Untagged and priority tagged frames are assigned the Port VID (PVID). The PVID is added as a C-tag.

An I-NNI is C-tag aware interface. C-tagged frames are accepted. Non C-tagged frames are always discarded.

Provider Mode

A CE-UNI is a C-tag aware interface. Incoming C-tagged frames having the C-VID registered in the C-VID registration table are tagged with the corresponding S-VID.

Incoming C-tagged frames not registered in the C-VID registration table are discarded. Untagged and priority tagged frames are assigned the S-VLAN corresponding to the Port VID (PVID) in the C-VID registration table. Untagged frames can also be assigned the PVID as a C-tag. Any tagged frame with EtherType different than 0×8100 is discarded.

Outgoing frames on a CE-UNI port can be filtered on S-VLANs or C-VLAN–S-VLAN pairs registered in the C-VID registration table.

As an external physical interface, a CN-UNI is S-tag aware and C-tag unaware. S-tagged frames are accepted. Incoming C-tagged frames are handled as untagged frames and are assigned the Port VID (PVID). The PVID is added as an S-tag. Untagged and priority tagged frames are assigned the Port VID (PVID).

An I-NNI (PN) is an S-tag aware interface. S-tagged frames are accepted. Non S-tagged frames are always discarded.

7.1.2 C-VID Registration Table

CE-UNI ports have a mapping table for C-VLAN and S-VLAN, which is an explicit C-VID registration table that provides mapping between C-VIDs and S-VIDs. The table has an entry for each of the allowed C-VLANs on the CE-UNI port.

When a frame enters a CE-UNI port and carries one of the allowed C-VLANs for the port, the S-VLAN that corresponds to the frame C-VLAN in the table is added to the frame.

In Provider Mode, it is possible to configure a CE-UNI port with characteristics comparable to a CN-UNI port. By configuring a default mapping rule into the C-VID registration table of a port, all C-VLANs can be mapped to an S-VLAN similar to how a CN-UNI port operates. If needed, it is possible to define exceptions from the all-to-one mapping by adding non-default entries to the C-VID registration table.

When untagged or priority-tagged frames enter the CE-UNI port, the entry corresponding to the port default PVID parameter is used to determine which S-VLAN that is added to the frame. If double tagging on untagged frames is enabled, the PVID is also added to untagged frames as a C-VLAN. If the PVID parameter is set to 0, incoming untagged and priority tagged frames are dropped (this is the default behavior).

Note: The default mapping rule does not apply to untagged and priority tagged frames.

In order to achieve the "default" tagging behavior for untagged and priority tagged frames, a non-default entry has to be created in the C-VID registration table, between any arbitrary C-VLAN ID and the S-VLAN ID also used for the default mapping rule, and the PVID of the port has to be set to this arbitrary C-VLAN ID.

Example configuration

PVID=999

C-VLAN ID	S-VLAN ID	Effect
1	1	S-VLAN ID 1 is added to incoming frames with C-VLAN 1 $$
2	1	S-VLAN ID 1 is added to incoming frames with C-VLAN 2
3	1	S-VLAN ID 1 is added to incoming frames with C-VLAN 3
0	2	Default mapping rule: S-VLAN ID 2 is added to incoming frames with any C-VLAN ID except 1, 2, 3 and 999
999	2	S-VLAN ID 2 is added to incoming frames with C-VLAN ID 999
		S-VLAN ID 2 is added to incoming untagged and priority tagged frames. If double tagging on untagged frames is enabled, C-VLAN ID 999 is also added to the untagged frames.

7.1.3 QinQ

When operating in customer mode, it is possible to encapsulate IEEE 802.1Q VLANs in other IEEE 802.1Q VLANs (both having TPID 0x8100), thus enabling the construction of Provider Ethernet networks without equipment supporting IEEE 802.1ad.

The configuration is possible on UNI ports, using the QinQ Termination Table. This table defines a mapping between Customer Edge VLAN IDs (Ce VLAN ID) and Provider Edge VLAN IDs (Pe VLAN ID).

When a frame enters a UNI port with QinQ Termination configured, and carries a VLAN ID defined as Ce VLAN ID in the QinQ Termination Table of the port, an outer IEEE 802.1Q VLAN tag is added to it with the corresponding Pe VLAN ID (this is called selective, or one-to-one mapping).

It is also possible to define a "default" (many-to-one/all-to-one) mapping in the QinQ Termination Table, by creating a mapping entry with Ce VLAN ID 0. If a many-to-one entry is configured in the QinQ Termination Table of the port, an outer IEEE 802.1Q VLAN tag is attached to ingressing frames which carry a VLAN ID not defined as a Ce VLAN ID in the QinQ Termination Table.

If QinQ Termination is configured on a UNI port, the outer VLAN tag is removed (stripped) from outgoing frames if the carried VLAN ID is registered as Pe VLAN ID in the QinQ Termination Table of the port (either for one-to-one or many-to-one mapping).

Ingressing frames with certain C-VLAN IDs can be admitted "transparently", without the addition of a second VLAN tag, even if "default" (many-to-one) mapping is configured in the QinQ Termination Table. This is done by configuring the port as member of these VLANs but not using these VLAN IDs either as Ce VLAN IDs or Pe VLAN IDs when configuring QinQ Termination. Egressing frames carrying these VLAN IDs will be transmitted without removing the (outer) VLAN tag.

Note: If a certain VLAN ID is configured as a Pe VLAN ID on a port (either with a one-to-one or many-to-one QinQ Termination mapping), it is not possible to "transparently" admit frames with that VLAN ID. If neither "default" (many-to-one) QinQ Termination mapping, nor a one-to-one QinQ Termination mapping with the certain VLAN ID as Ce VLAN ID is configured, incoming frames with the certain VLAN ID are dropped.

7.1.4 Priority Handling (Ingress)

The main objective of the priority classification/tagging process is to establish a frames relative priority to offer appropriate propagation in the network and internally in MINI-LINK 6600.

Definitions

Trusted Port



In MINI-LINK 6600 there are different criteria that come into play when a frame is priority tagged. First of all the operator must decide whether the connected user domain is trusted or not. For a trusted domain the operator can reuse the priority information in the frame, set by the client. For an untrusted port the default network priority value is used.

User/Customer and Network Domains

The actual priority of a frame in an end to end Ethernet connection can be different based on where in the network it is. A network can be logically split up in a network domain and one or more user/customer domains. All nodes in an area have a similar congestion behavior. In the user/customer domain, the priority settings of an Ethernet frame are based on the individual user/customer definitions and referred to as user/customer priority. In the network domain the network priority settings, defined by the operator, are used.

User/Customer Priority

The priority setting used in the user/customer domain and set by the user/customer. The user/customer priority information can be retrieved from the user domain frames in one of the following ways:

- 1. From the PCP bits in the C-VLAN Ethernet header (valid for UNI in customer mode; CE-UNI in provider mode; CN-UNI in provider mode, only if the acceptable frame type of the CN-UNI port is set to Admit Only Untagged and Priority Tagged Frames).
- 2. From the PCP bits in the S-VLAN Ethernet header (valid for CN-UNI in provider mode).
- 3. From the IPv4 DSCP bits in the IP header.
- 4. From the IPv6 DSCP bits in the IP header.
- 5. In Layer 2 mode, the option DSCP IPv4 and IPv6 makes it possible to use trusting on IPv4 and IPv6 DSCP at the same time.
- 6. From the EXP bits in the MPLS tag.
- 7. From both the EXP bits in the MPLS tag and the DSCP bits in the IP header. Primarily the classification in this case is based on MPLS EXP, but in case of non-MPLS traffic, the IP DSCP value is used for prioritization.

Network Priority

The network priority is the Ethernet frame's priority setting used in the operator/network domain. The different network priority values are defined by the operator.

The network priority of an Ethernet frame is typically a representation of the traffic type and the application generating the Ethernet flow, for example, voice call.

 \geq

The traffic types and associated priority values are standardized in IEEE 802.1Q and IEEE 802.1D. Each traffic type is associated with a priority value to indicate the relative importance of that traffic type. Highest number equals highest priority.

In a network, it is important that only one set of priority definitions is used, for example, IEEE 802.1D. Otherwise, the handling of Ethernet frames and the mapping to egress queues can differ between network elements. The outcome of this scenario is a non predictable behavior for the different traffic types, for example, voice.

Priority Code Point (PCP)

The Priority Code Point is the three bits in the Ethernet header that is used to carry the priority value and coloring information for the Ethernet frame. The priority information is mandatory while coloring information is only applied when the Ethernet flow is policed.

Priority Tagging

When a frame enters the network domain (UNI), the user/customer priority settings are either mapped to the operator network priority definitions (trusted interface) or to the default priority setting (untrusted interface). The main purpose of the priority tagging process is to mark a frame with appropriate priority to be used in the network domain, that is, network priority. Priority tagging is only available on a port configured as a Layer 2 connection.

Figure 40 illustrates how the user/customer priority is mapped to the appropriate network priority at the ingress side of MINI-LINK 6600 as a part of the frame tagging process. The network priority value is stored in the PCP bits in the Ethernet header (the possible user/customer priority alternatives are specified earlier in this chapter). For untagged frames and untrusted interfaces the default network priority is used. The network priority value is used as the frame's priority in the CoS/congestion handling mechanisms internally in MINI-LINK 6600. On the egress side of MINI-LINK 6600 the PCP value is encoded from the network priority value and stored in the Ethernet frame's PCP bits. The Ethernet frame is mapped to the appropriate priority/TC buffer.



Figure 40 Priority Tagging at Network Edge for Trusted Interface

Note: The network priority and PCP value is a 1:1 mapping unless the PCP bits are used to carry coloring information. See Section 7.2 on page 64, for more information.

The flowchart in Figure 41 is used in the tagging process to establish network priority at the network edge.



Figure 41 Priority Tagging Flowchart

Untagged & DA match test if the frame is untagged and have a destination MAC address belonging to the reserved group MAC addresses. BPDU, Link OAM, MAC Pause all belong to this category.

On a trusted port, the EXP/DSCP priority information in the MPLS/IP header or PCP bits in the Ethernet header is used to establish the network priority. The translation is done according to a mapping table customized by the operator.

Priority Classification

Priority classification is done internally in the network domain to decode the network priority value from the PCP bits. The network priority value is used as the priority of the frame in the CoS/congestion handling mechanisms internally in MINI-LINK 6600. On the egress side of MINI-LINK 6600 the PCP value is encoded



from the network priority value and stored in the PCP bits in the Ethernet frame. The Ethernet frame is put in the appropriate priority/TC buffer, see Figure 42.

Figure 42 Priority Classification on Internal Network Interfaces

Note: The network priority and PCP value is a 1:1 mapping unless the PCP bits are used to carry coloring information. See Section 7.2 on page 64, for more information.

7.1.5 Priority Handling Layer 1

The contents and concepts described in Section 7.1.4 on page 58 also apply to L1 Ethernet Services.

In the case of a trusted Layer 1 port, the user/customer priority information can be retrieved from the user domain frames in one of the following ways:

- 1. From the PCP bits in the C-VLAN Ethernet header.
- 2. From the PCP bits in the S-VLAN Ethernet header.
- 3. From the IPv4 DSCP bits in the IP header.
- 4. From the IPv6 DSCP bits in the IP header.
- 5. With the option DSCP IPv4 and IPv6 it is possible to use trusting on IPv4 and IPv6 DSCP at the same time.
- 6. From the EXP bits in the MPLS tag.

 \geq

7. From both the EXP bits in the MPLS tag and the DSCP bits in the IP header. Primarily the classification in this case is based on MPLS EXP, but in case of non-MPLS traffic, the IP DSCP value is used for prioritization.

Header Decoding

L1 Ethernet Service is transparent to all kinds of traffic: untagged traffic; any kind of tagged traffic, including tagged traffic with VLAN ID compliant to IEEE 802.1Q or IEEE 802.1ad.

7.2 Policing/Color Marking (Ingress)

At the network edge the policing rate enforcement feature controls the flow into the network. The policing feature can be used to make sure the traffic from an external customer does not violate the specified input rate and burst size. The policing feature in MINI-LINK 6600 is compliant to the ingress Bandwidth Profile functionality specified in MEF 10.1.

In MINI-LINK 6600, policing can be activated either per port or per port per user/customer priority or per port per VLAN. This means that policing rate enforcement can be activated for the consolidated input flow, independent of the traffic types , or individually for each traffic type in the input flow based on either priority or VLAN. The policing feature is only available on a port configured as a Layer 2 connection.

As described later in this chapter, the user/customer priority can be obtained in different ways, for example, DSCP. To select the appropriate Bandwidth Profile (policer) for a flow, the corresponding user/customer priority value is mapped to a user priority group. The user priority group is pointing to the appropriate Bandwidth Profile, see Figure 43.

In case of VLAN based policing, the Bandwidth Profile assignment can be done in a similar way. VLAN IDs can be mapped to a VLAN group, and the Bandwidth Profile can be assigned to the VLAN group.

For policing per port, the port number is directly associated with the Bandwidth Profile.



Figure 43 Policing Data Model

The output of the policing function in MINI-LINK 6600 is typically a colored or dropped frame, see Figure 44. The criteria for the deciding whether the frame is colored (green or yellow) or dropped (red) are based on the compliance with the parameters specified below.



Figure 44 Policing Function in MINI-LINK

By using the policing feature the input flow can be controlled with the following parameters:

 Committed Information Rate (CIR): The CIR value is typically a guaranteed end to end throughput rate, that is, all frames within the CIR value from

the external client are propagated transparently with no frame loss. CIR is specified as bps.

- Committed Burst Size (CBS): The CBS value indicates the guaranteed burst length, that is, all frames within the CBS burst length from the external client are propagated transparently with no frame loss. CBS is specified in number of bytes
- Excess Information Rate (EIR): The EIR value is a throughput rate above the CIR committed input rate. Frame rates above the CIR but below EIR are accepted in to the network but with no guarantees. EIR is specified as bps.
- Excess Burst Size (EBS): The EBS value indicates the excess burst length. Burst lengths above the CBS but below the EBS size are accepted in to the network but with no guarantees. EBS is specified in number of bytes.
- Coupling Flag (CF): The CF value controls the volume of the frames that are colored yellow. In MINI-LINK 6600 the CF flag is set to 0, that is, the long term average bit rate of frames declared yellow is bounded by EIR.
- Color mode (CM): The CM value decides whether existing color information in a frame should be used in the policing algorithm or not. In MINI-LINK 6600 the CM is set to Color Blind, that is, existing color information is disregarded in the policing algorithm.

The coloring process uses the above parameters in the following manner to identify the appropriate color for a frame:

- Green frames: Input Rate<CIR and Burst Size<CBS
- Yellow frames: CIR<Input Rate<EIR+CIR and CBS<Burst Size<EBS
- Red frames: Input Rate>EIR or Burst Size>EBS



A graphical presentation of the coloring criteria is illustrated in Figure 45.

Figure 45 Coloring of Frames in the Policing Function

The color information in a Customer bridge network is transported in the PCP bits in the Q-tag. If policing/coloring is used for a port, the number of priority/TC

queues for that port is reduced, that is, some PCP bit values are used for coloring information. For each priority level set up with policing the number of priority/TC queues is reduced by one.

The color information or drop eligibility in a provider network can be either explicitly or implicitly encoded in the S-VLAN tag. Drop eligibility can be encoded with the priority in the PCP field of the S-tag as specified for the C-tag, or it can be encoded explicitly by using the Drop Eligible Indicator (DEI) bit in the S-tag.

The IEEE802.1ad standard defines that PCP allows encoding of five, six, seven, or eight distinct priorities, with a drop eligibility on three, two, one, or zero of those priorities respectively, see Figure 46.



Note: Color dropping is not applicable for the two highest priorities, 6 and 7.

Figure 46 PCP Encoding

8p0d	8 priorities, 0 priorities with color dropping
7p1d	7 priorities, 1 priority with color dropping
6p2d	6 priorities, 2 priorities with color dropping
5p3d	5 priorities, 3 priorities with color dropping
8p8d	Only available in Provider mode. Eight priorities, the DEI bit in the S-VLAN tag is used to retrieve or indicate the drop eligibility of frames.

In Customer mode, coloring is performed at I-NNIs, while the UNIs always operate in 8p0d PCP selection as long as the PCP bits in the C-VLAN tag are trusted.

In Provider mode, coloring is performed at I-NNI (PN)s and CN-UNIs. On I-NNI (PN)s it obeys the selection. CN-UNIs have the following behavior:

 If the PCP value of the incoming frame differs from 8p8d and the PCP bits in the S-VLAN tag are trusted, the PCP value is set to 8p0d.

— If the PCP value of the incoming frame is 8p8d and the PCP bits in the S-VLAN tag are trusted, the PCP value is set to 8p8d.

The CE-UNI always operates in 8p0d PCP selection, if the PCP bits in the C-VLAN tag are trusted.

Coloring can be configured for maximum three priority levels, that is, 5p3d. This means that three out of five priority levels can be color marked. It is possible however to activate policing for the priority levels with no coloring support. These frames will be policed according to CIR/CBS and either dropped or passed through but with no color information applied to the frame.

For a customer interface protected with LAG, policing cannot be activated.

7.3 Queue Management

In a multi service network there is a need to differentiate the processing of the different flows in the congestion points. Typically a voice flow would require low latency/jitter and thus higher priority than a best effort data flow with no latency requirements.

Differentiation of flows is done based on the network priority value in the header part of each Ethernet frame. The network priority value to be used in the network domain is typically set as a part of the tagging process at the network edge.

The priority value of a frame is used to provide different Classes of Service (CoS) and is a representation of the end user application, for example, voice, best effort data, and so on. The priority value of an Ethernet frame indicates what type of behavior it expects from the network and how it shall be prioritized with respect to the other Ethernet frames in an interim network element.

An alternative to the mechanisms described in this chapter is to overprovision the network, that is, the capacity is sufficient to guarantee a congestion free network.

Traffic Class (TC) Mapping

MINI-LINK 6600 supports eight TCs or priority queues for each egress Ethernet port.

The Ethernet frames are mapped to the different TCs based on the network priority value in the PCP bits in the Ethernet header (see Figure 40). The mapping can be done according to IEEE 802.1D, see Figure 47, IEEE 802.1Q, see Figure 48, or it can be customized.



Figure 47 IEEE 802.1D Mapping



Figure 48 IEEE 802.1Q Mapping

Shared Packet Buffering

Ethernet interfaces share the available memory, so multiple queues compete for the same buffering resource. A buffer management algorithm is used to efficiently manage the existing buffer resources in fair manner.

A discard threshold for the queues is dynamically calculated based on the size of the remaining buffer space. In a non-congested or relatively uncongested state the remaining buffer space is large, so the queue discard thresholds are set to a higher value which allows better burst absorption. On the other hand, if the remaining buffer space is very small, then the switch is facing severe congestion, therefore queue discard threshold is lower.

A frame is enqueued in the corresponding queue, if the queue length is smaller than the calculated discard threshold. Otherwise the frame is dropped. In congested state the frames attempting to enter the longer queues are more likely to be dropped than frames entering shorter queues since the queue length of the longer queues would be higher than the threshold. Consequently this fair buffer management algorithm aggressively drops frames belonging to longer queues, thereby freeing up resources for queues that are relatively uncongested, see Figure 49.



Figure 49 Adaptive Dynamic Queue Threshold

The recommended way of limiting queue size is either by configuring a user-defined queue size limit for tail-drop or by using WRED.

Deep Buffering

Deep Buffering is either enabled or disabled on an Ethernet interface and used to provide enough buffering capacity for handling traffic bursts. All radio link (WAN) interfaces are assigned to deep buffer, as a fading event might cause congestion. In case of LAN interfaces, it is platform dependent whether Deep Buffering is enabled or disabled by default, but it can be enabled by the operator

|

Time Stamp-Based Dropping/Aging

A timestamp is applied to the frame when it enters the TC queue buffer. Before exiting the TC buffer (that is, served by the scheduler) the timestamp is checked towards the aging timer. If the timestamp value exceeds limit for aging time from when the frame was stored in the queue, the frame is discarded. The aging time can be adjusted from the management interface based on the PCP priority value of the incoming frame.

Note: Aged frames consume egress bandwidth and can cause the transmitted traffic rate to drop below line rate.

WRED

To give a better understanding of WRED, some basics of the TCP protocol is described first.

The nature of a TCP connection is an increasing bandwidth until frames are lost. When the TCP protocol recognizes frame loss, the throughput is reduced by 50%. TCP then repeats the process by increasing the throughput until frames are lost. The throughput pattern is often referred to as saw tooth shape, see Figure 50.



Figure 50 Saw Tooth Shaped TCP Throughput

When multiple TCP flows experience congestion in the same port, all TCP flows loose packets at the same time (tail drop). This means that all TCP flows simultaneously reduce throughput by 50%. The result is a link with low average throughput. With WRED activated, frames are randomly discarded at a configurable buffer filling threshold. This means that tail drop can be avoided and only a limited number of the TCP flows will experience frame loss and reduce the throughput. Thus, WRED can increase the average throughput by minimizing the throughput variations, see Figure 51.

WRED starts discarding frames when the buffer filling has reached a certain configurable threshold. If the buffer filling continues to increase the drop probability increases, see Figure 51. The buffer filling is calculated as an average value, estimated over a certain period of time. This is a more accurate way of measuring buffer than for instance instant measurements.



Figure 51 WRED Thresholds

WRED can be enabled on ports with deep buffer.

If Color-aware WRED is used to regulate a queue, then different WRED profiles can be applied for green and yellow traffic.

Note: WRED requires buffering, therefore it can only be set on interfaces with enabled deep-buffering.

Tail Drop

The tail drop discard mechanism is used when a TC buffer reaches the dynamically calculated discard threshold. In this case, the tail drop feature ensures that new frames are dropped at the entry point of the TC queue.

Tail drop is used to handle discarding of frames during long bursts that would completely fill up the available buffer space.

Tail drop queue user-defined size limitation is only effective when deep buffering enabled on the port.

Frame Discard Summary

Figure 52 illustrates how the different discard mechanisms are organized in MINI-LINK 6600.



Figure 52 Discard Mechanisms in MINI-LINK

7.4 Scheduling (I-NNI and I-NNI (PN))

When an Ethernet port is congested the egress priority/TC buffers are filled up with Ethernet frames. To empty the buffers a scheduler must serve the buffers according to a predefined algorithm. MINI-LINK 6600 supports both Strict Priority (SP) and weighted scheduling algorithms (for example Deficit Weighted Round Robin (DWRR)) to empty the egress priority/TC buffers.

The SP algorithm is typically used for delay sensitive traffic like voice, video and sync packets. With SP, Ethernet frames in the buffer with the highest priority are always scheduled first, for example, all Ethernet frames with priority level 7 are scheduled before Ethernet frames with priority level 6.





The weighted scheduling algorithm is used when low latency is not crucial, but it is important to prevent starvation of buffers. With SP, buffers with lower priority will never get served if high priority queues are constantly being filled up. Weighted scheduler, however, will make sure that all queues are served according to the predefined weight ratio. The accuracy of the allocation is roughly 1%, where constant bit rate traffic is more accurate than burst traffic. Weighted scheduler is typically used for buffers with non delay sensitive data traffic.



Figure 54 Weighted Scheduling

In a live network the SP and weighted scheduling algorithms are typically combined. In MINI-LINK 6600 the SP and weighted algorithms can be applied to different buffers as described in the example below. The size of the SP and weighted areas (x and y values) can differ but the SP/Weighted organization is fixed. The scheduler will serve the high priority SP area first and empty the



buffers according to the SP algorithm. When the high priority SP area is empty the scheduler starts serving the weighted buffers according to individual weight.

Figure 55 SP and Weighted Scheduling Combined in the Same Port

The SP and weighted schedulers can be configured individually per port from CLI and SNMP (not MINI-LINK Node GUI). However, to simplify the configuration effort the following predefined templates are supported. These templates can be selected individually from CLI, SNMP and MINI-LINK Node GUI,.



Figure 56 Scheduling Templates

7.5 Congestion Handling Summary

Congestion Handling in MINI-LINK can be summarized as follows:

- For trusted user domains, priority information is mapped from user/customer domain to network domain at the UNI.
- For untrusted user domains, default priority value is used as network priority in the network domain.
- Internally in the network domain the network priority settings are stored in the L2 PCP bits in the L2 header.
- Ethernet frames are mapped to egress TC buffer according to IEEE802.1D/Q or custom.
- In the egress TC buffer, frames are serviced according to a SP scheduler or weighted scheduler (DWRR), or both.
- Frames are dropped in the egress TC buffer due to tail dropping, color dropping, or WRED.
- WRED and color dropping are implemented for interfaces with deep buffer.



Figure 57 Congestion Handling in MINI-LINK

8 Hierarchical Quality of Service

Hierarchical QoS (HQoS) provides an additional level of traffic differentiation based on VLANs, therefore it enables that service quality can be ensured in a shared network scenario when multiple service providers share the same network, and also it helps to manage multiple RAN technologies separated by VLANs.

The traditional 2-level queue-port hierarchy can be extended to up to 4-level hierarchy, that is queue-node-node-port. The nodes can be configured to form the hierarchy according to the required use-case.

QoS features for HQoS are:

- VLAN based shaping: A maximum bandwidth can be set for a VLAN or a group of VLANs, that is, for an operator or an RBS.
- VLAN based scheduling: Both strict priority and deficit weighted round-robin are supported.



Figure 58 Principles of Hierarchical Quality of Service

9 Forwarding

9.1 Unicast Forwarding

MINI-LINK 6600 supports up to 1000 VLANs and storage and learning of 16K MAC addresses. It also supports an aging function, which discards MAC addresses that have been unused for a given time.

Each VLAN has individual learning, which in customer mode is performed on C-VLAN (EtherType 0×8100) and in provider mode performed on S-VLAN (EtherType 0×88a8). Each VLAN has a Filtering Database with dynamic and static entries.

Dynamic entries are automatically added and updated by the MAC learning process. Static entries are added manually and prevent flooding in case frames arrive for a destination address from which no Ethernet frames have arrived yet.

9.2 Multicast Forwarding

MINI-LINK 6600 supports a Layer 2 (MAC level) and a Layer 3 (IP level) multicast forwarding table.

- The L2 multicast table contains static entries. Each entry associates a (MAC address, VLAN) pair with a set of egress ports on that VLAN.
- The L3 multicast table is maintained dynamically by Internet Group Management Protocol (IGMP) and Multicast Listener Discovery (MLD) snooping. Each entry associates an IP multicast group (*, G, VLAN) or channel (S, G, VLAN) with a set of egress ports on that VLAN.
- **Note:** The table lookup is based on the outer VLAN tag of the frame, that is, the C-VLAN in customer mode and the S-VLAN in provider mode.

IPv4 and IPv6 multicast packets are first looked up in the L3 multicast forwarding table by their source, destination (group) IP address, and VLAN ID. If a matching entry is found, the packet is forwarded to the ports in the entry, otherwise the packet is looked up in the L2 multicast forwarding table by its destination MAC address and VLAN Id. If no matching entry is found, the packet is classified as an Unregistered Multicast frame.

Non-IP Ethernet multicast frames are looked up by their destination MAC address and VLAN Id in the L2 multicast forwarding table. If a matching entry is found the frame is forwarded to the ports in the entry, otherwise it is classified as Unregistered Multicast.

Unregistered Multicast frames are either flooded to the VLAN or discarded. The handling is configurable per VLAN.

9.3

IGMP Snooping (IPv4) and MLD Snooping (IPv6)

IGMP for IPv4 and MLD for IPv6 are the protocols used by IPv4 and IPv6 hosts to report their IP multicast group memberships to neighboring multicast routers.

By default, a Layer 2 switch will flood multicast traffic to all ports in the VLAN (broadcast). The broadcast traffic can cause unnecessary bandwidth consumption in the network and extra processing load on host devices.

IGMP&MLD snooping uses the mechanism by which a Layer 2 device actively listens to Layer 3 IGMP&MLD messages between multicast hosts and routers. By parsing the messages passing through the switch, the Layer 2 switch learns the member status of multicast groups, and then builds a multicast table with this learned knowledge.

IGMP&MLD snooping switches will only forward multicast traffic to the ports where interested hosts are attached, thereby using link capacities in an efficient way.

MINI-LINK 6600 supports IGMP and MLD snooping for IGMPv2/v3 and MLDv1/v2 messages. Both Source-Specific Multicast (SSM) and Any-Source Multicast (ASM) service models are supported.

In MINI-LINK 6600, IGMP and MLD snooping is configured independently per VLAN instance. IGMP and MLD snooping can be simultaneously enabled on a VLAN. In customer mode, IGMP and MLD snooping can be activated on single C-tagged VLANs, while in provider mode both single S-tagged or S+C double-tagged VLANs are supported. IGMP and MLD snooping is disabled on all VLANs by default.

Note: In provider mode only the SVID is used in forwarding lookup, thus all CVIDs in an SVID share a common forwarding state. It is recommended to use only one CVID per SVID for multicast to avoid unwanted multicast traffic flooded to CEs.

The switch detects multicast router ports automatically based on the reception of IGMP and MLD general query and PIMv2 Hello messages.

When MINI-LINK 6600 detects an STP topology change and is the root bridge on the affected VLAN, a Query solicitation message is broadcasted to the Querier router to facilitate fast convergence of multicast forwarding state.

By default MINI-LINK 6600 operates in transparent snooping mode. In this mode the switch forwards all IGMP and MLD messages between hosts and routers transparently, and processes them to maintain the L3 multicast forwarding table.

The following is valid for IGMP and MLD snooping:

Proxy Reporting

Proxy reporting extends the basic snooping functionality aiming to minimize the IGMP and MLD protocol processing load on upstream multicast routers and IGMP and MLD snooping switches.

When Proxy reporting is enabled, IGMP and MLD join reports for existing IGMP and MLD states are suppressed. Only the first IGMP and MLD join message is propagated towards routers. In addition, IGMP and MLD leave messages are forwarded from the hosts to the IGMP and MLD routers only when the last member leaves a group.

— Immediate-leave

MINI-LINK 6600 supports Immediate-leave on individual VLAN ports. Upon receiving an IGMP and MLD leave message, IGMP and MLD snooping immediately removes the port from the forwarding table entry for that multicast group without sending out an IGMP and MLD group-specific query and waiting for its timeout.

Note: Immediate-leave should only be enabled on VLAN ports to which a single host is attached. Otherwise, some hosts can be inadvertently dropped.

Ethernet Switch Limits

The L3 multicast forwarding table can accommodate up to 1000 multicast groups.

10 Ethernet Protection

MINI-LINK 6600 offers multiple redundancy features that fulfill different needs and operate both on plug-in unit and connection level.

10.1 RSTP/MSTP

Rapid Spanning Tree Protocol (RSTP) and Multiple Spanning Tree Protocol (MSTP) help build logical loop-free topology for Ethernet networks. The basic function is to prevent bridge loops and the broadcast radiation that results from them.

RSTP

The RSTP offers redundancy and rerouting of traffic if a node is out of service. The Ethernet transport network is typically divided in multiple RSTP domains. Within one RSTP domain a reconfiguration of the active Ethernet connections typically takes 50–100 ms without service impact.

MSTP

MSTP supports multiple concurrent topologies and spanning trees mapping VLANs into spanning trees. It uses basic RSTP mechanisms for topology calculation and rapid port state transitions.

MSTP configures a separate Spanning Tree for each VLAN group and blocks the links that are redundant within each Spanning Tree. This enables load balancing of network traffic across redundant links.

MSTP is backward compatible with RSTP and STP and the same performance requirements are applicable for MSTP as for RSTP.

10.2 Ethernet Ring Protection (ERP)

Ethernet Ring Protection (ERP) switching and Ethernet Linear Protection Switching (LPS) are supported according to ITU-T standards G.8031, G.8032v1, and G.8032v2.

10.2.1 ERP Single Ring

A single Ethernet ring consists of at least two NEs that are connected to each other in a ring configuration. One of the links is designated as the Ring Protection Link (RPL) which protects the entire ring.

To avoid loops, the RPL is inactive during normal operation. See Figure 59.

If one of the other links in the ring fails, the RPL is activated and the traffic is rerouted to avoid disturbances. See Figure 60.

When the failed link is restored, the RPL becomes inactive again.



Figure 59 Single Ring, Normal Operation



Figure 60 Single Ring, Link Failure



10.2.2 ERP Multiple Rings

Multiple Ethernet rings consist of NEs that are connected to each other in a configuration with two or more rings, one main ring and one or more sub-rings. In each ring, one of the links is designated as the Ring Protection Link (RPL) which protects that ring.

To avoid loops, the RPL is inactive during normal operation. See Figure 61.

If one of the other links in the ring fails, the corresponding RPL is activated and the traffic is rerouted to avoid disturbances. See Figure 62.



When the failed link is restored, the RPL becomes inactive again.

Figure 61 Multiple Rings, Normal Operation



10.3 Link Aggregation Group (LAG) with LACP

This section presents the protection mechanisms available on Layer 2 connections.

Link Aggregation Group

MINI-LINK 6600 supports Link Aggregation Group (LAG) with Link Aggregation Control Protocol (LACP), which aggregates several external Ethernet links into one logical link and provides line protection.

A LAG supports a maximum of four links, but in case of NPU 1002 or NPU 1005 the number of links increase so that a LAG supports a maximum of eight links.

LACP has three operational modes:

— Static

No LACP PDU communication. LAG hashing is governed by the port operational and administrative states or optionally by LinkOAM.

Active

LAG hashing is governed by the LACP state machine using the information inferred from LACP PDU communication.

— 1+1

Al LAG-LACP operation applicable for 2-port LAGs, when LACP allows only one active port out of the 2 LAG member ports.

When a new LAG is created, the default LACP operation mode is Static.

Link Aggregation Control configures and controls the Link Aggregation sublayer using static or dynamic information, exchanged between the connected NEs by means of LACP. LACP enables dynamic configuration of LAG.

LACP is used by the nodes on either end of a LAG to automatically monitor and maintain adjoining LAG links between them.

On a regular basis, L2 Link Aggregation devices with running LACP advertise their capabilities and status on potential aggregation. The peering devices compare their own configuration and the configuration of their partner and states with each other, associated with each link of the LAG. Each peer can then decide what action to take in terms of setting a link traffic carrying status. Once in a steady state, LAGs are maintained by regular exchange of information between peers.

If a link in a LAG fails, traffic is redirected from the faulty link to the remaining links in the LAG using graceful degradation. The switch from active to passive path takes less than 50 ms for a bidirectional link failure and does not impact the traffic. A faulty link affects the total link capacity and traffic with low priority may be discarded to ensure that traffic with high priority is sent.


When a 2-port LAG is configured in LACP 1+1 mode, the LAG member ports form an active/standby pair. Generic Ethernet traffic is transmitted and received through the active port. LACP PDUs are exchanged through both ports.

Whenever there is a connectivity fault in the link connected to the former active port, then the former standby port is activated and Ethernet traffic is redirected to this port. When the link of the higher priority link is recovered, it regains the active role if revertive switching is enabled.

Whenever there is a connectivity fault in the link connected to the standby port, the Ethernet traffic through the active port remains undisturbed.

In MINI-LINK 6600, the LAG master port is assigned with the highest LACP port priority.

Which of the 2 LAG member ports becomes active is determined by negotiating the LACP port priorities between the two LACP hosts.

Note: The near end LACP host is configured with LACP 1+1, while the far-end host is configured with LACP Active, then the LAG will operated in active/standby manner and the active port will be arbitrated between the two LACP hosts according to the LACP standard.

Network rerouting according to Ethernet Ring Protection (ERP) and Rapid/Multiple Spanning Tree Protocol (RSTP/MSTP) is not triggered unless all physical links in an LAG fails.

ERP and RSTP/MSTP activates a redundant link in case of link failure and protects the network from infinite loops.

To improve the switch-over time for unidirectional faults, LAG should be combined with Link OAM. Link OAM will take down the link if a unidirectional fault is detected.

A Hashing Algorithm is used to identify the different traffic flows and assign them to the different ports, belonging to the same LAG.

Hashing algorithm is configurable and is based on different L2 or L3/L4 traffic parameters, according to Table 9.

MINI-LINK Node GUI Parameter	CLI Parameter	Hashing Algorithm
srcMac	mac-sa	Source MAC address, VLAN ID, Ethertype, and source bridge port ID.
dstMac	mac-da	Destination MAC address, VLAN ID, Ethertype, and source bridge port ID.

Table 9 Hashing Algorithms

MINI-LINK Node GUI Parameter	CLI Parameter	Hashing Algorithm
srcDstMac	mac-sa-da	Source MAC address, destination MAC address, VLAN ID, Ethertype, and source bridge port ID.
srcIp	ip-src	Source IP address and source TCP/UDP port.
dstIp	ip-dst	Destination IP address and destination TCP/UDP port.
srcDstIp	ip-src-dst	Source IP address, destination IP address, source TCP/UDP port, and destination TCP/UDP port.

The hashing algorithm selection allows an effective traffic load balancing between the different links belonging to an LAG.

10.3.1 LAG/LACP 1+1 Protection of MINI-LINK 6352

LAG/LACP 1+1 protection can be used in combination with a redundant E-Band link pair of two MINI-LINK 6352 according to Figure 63, so that the connectivity is protected against both E-band link faults and MINI-LINK 6352 equipment.



Figure 63 LAG/LACP 1+1 Protection of MINI-LINK 6352

For the details of the protection mechanism, please refer to the CPI library of MINI-LINK 6352.

10.4 Ethernet Layer 1 Protection Mechanisms

The Link Loss Forwarding (LLF) feature improves the protection switching time for Layer 1 connections by propagating connectivity loss information to both connection end points. LLF is not supported with ETU interfaces.

LLF can be split in two scenarios, broken WAN links or broken LAN links.

WAN Link Broken

When the terminating end points of the point to point connection detects that the WAN connection is lost, the Ethernet ports on both sides will be taken down (that is, disable line carrier).

LAN Link Broken

When a terminating end point of the point to point connection detects that the LAN connection is lost, the corresponding WAN port will be taken down. On the remote end, the disconnected WAN link is detected and the corresponding LAN port will be taken down.

11 Ethernet Operation & Maintenance

MINI-LINK 6600 have support for a number of Ethernet application related O&M functionalities.

11.1 Ethernet Performance Counters

For information on the performance counters supported for LAN/WAN Ethernet ports in MINI-LINK 6600, see Section 16.9.3 on page 248.

11.2 Ethernet Link OAM

MINI-LINK 6600 supports Ethernet in the First Mile (EFM) or Link OAM according to IEEE802.3ah for all LAN ports. Link OAM is a diagnostic tool that can be used in both online and offline configurations.

Note: Ethernet Link OAM is only supported for LAN interfaces (Layer 1 Connection and Layer 2).

The three main Ethernet Link OAM services are as follows:

— Failure Notification

Notification of an Ethernet link failure to or from far end for an NE in operation.

The following three types of failures are supervised:

• Link fault (RDI)

The **Link fault (RDI)** alarm is generated when a failure in a physical layer has occurred in the receiving direction.

• Dying gasp

The **Dying gasp** alarm is generated when a plug-in unit is about to restart or is going to operational state **Down**. This occurs when an unrecoverable failure has occurred.

Note: Only supported in receiving direction (an event is raised).

Critical event

The **Critical event** is generated when an unspecified critical event has occurred.

Note: Only supported in receiving direction (an event is raised).

— Link Monitoring

Link monitoring is used for event notification on errored frames at both near and far end and is used on NEs in operation. The notifications are based on a threshold crossing within a specific time window.

The following events are reported:

• Errored Symbol Period Event

Generated when the number of symbol errors exceeds a threshold in a given time window, which is defined by a number of symbols.

• Errored Frame Event

Generated when the number of errored frames exceeds a threshold in a given window, which is defined by a period of time.

• Errored Period Event

Generated when the number of errored frames exceeds a threshold in a given window, which is defined by a number of frames.

• Errored Frame Seconds Summary Event

Generated when the number of errored frame seconds exceeds a threshold in a given time period. An errored frame second is defined as a 1 second interval with one or more frame errors.

— Remote Loopback

Link OAM remote loopback can be used for fault localization and link performance testing on LAN interfaces. Statistics from both near end and far end NE can be requested and compared at any time while the far end NE is in O&M remote loopback mode. The requests can be sent before, during, or after loopback frames have been sent to the far end NE. The loopback frames in the O&M sublayer can be analyzed to determine which frames are being dropped due to link errors.

In the offline set up all traffic is halted for the affected ports. A loop is configured manually and a traffic generator sends test frames as indicated in Figure 64. PM measurements in the receiving end are compared with the submitted traffic from the test generator to detect errors. The manually configured loop can be set on either side of the connection.



Figure 64 Link OAM Offline Configuration

As an online diagnostic tool, Link OAM is constantly monitoring the link status and detects unidirectional and bidirectional faults, see Figure 65. When a link fault is detected, for example, LOC, RDI is transmitted in the Tx direction. On the receiving side an alarm is triggered.



Figure 65 Link OAM Online Configuration

11.3 Ethernet Service OAM

Ethernet Service OAM is used to manage networks comprising of multiple LANs. It provides performance monitoring, according to ITU-T Y.1731 and Y.1730, and supports fault management on Ethernet links, according to IEEE 802.1ag.

Ethernet Service OAM can be used in both Customer mode and Provider mode, if Ethernet Service OAM PDUs are C- or S-VLAN tagged, and can be used in LAG and RSTP/MSTP scenarios.

The four main Ethernet Service OAM functions are as follows:

— Continuity Check Monitoring

Continuity Check Monitoring detects service interruption between MEPs. Continuity Check Messages (CCMs) are sent from one MEP to another, enabling MEPs to locate other MEPs. CCM confirmation can also be requested by an MEP from a linked MEP, to ensure that the CCMs are sent and received without fault.

The CCM intervals can be set at 3.3 ms, 10 ms, 100 ms, 1 s, 10 s, 1 min, or 10 min.

— Remote Defect Indication

An MEP uses Remote Defect Indication (RDI) to communicate with linked MEPs that a fault has occurred, usually that CCM confirmation were not received. The RDI is an indication that a fault has occurred either at the far-end MEP or between the two MEPs.

Loopback

Loopback is a troubleshooting tool that verifies the connectivity of a MEP with linked MEPs and linked Maintenance Intermediate Points (MIPs).

— Linktrace

Linktrace is a bidirectional continuity check used for fault localization. When a Linktrace Message (LTM) is sent to a destination MEP or MIP, a Linktrace Reply (LTR) is expected from all the intermediate MIPs along the path to the destination and from the destination MEP or MIP itself. Missing or misordered LTRs point out the location of a fault in an efficient way.

Ethernet Service OAM supports the following PM functions:

- Frame Delay
- Frame Delay Variation Measurement
- Frame Loss Measurement (ETH-LM)
- Ethernet Bandwidth Notification (ETH-BN)

A Maintenance Domain (MD) is defined as a network or sub-network, at the Ethernet level, within which OAM frames are exchanged. An MD determines the span of an OAM flow, across network administrative boundaries. There are the following three types of MDs:

- Network Operator MD
- Service Provider MD
- Customer MD

MDs are hierarchal and as such, MDs of the same type do not overlap each other, for example, two MDs of the same level do not overlap each other. However, different MD types from different levels may overlap, for example, a Customer MD may overlap multiple Service Provider MDs, but the Customer MD cannot overlap another Customer MD.

An MD consists of the following components:

- Maintenance Entity (ME) An OAM entity that requires management.
- Maintenance Association (MA) A group of MEs that belong to the same service inside a common MD. An IEEE concept.
- MD Level A way of distinguishing which MEs belong to the same MD. All MEs belonging to the same MD share the same MD Level. An IEEE concept.
- ME Group (MEG) A group of MEs that belong to the same service inside a common MD. An ITU-T concept.
- MEG Level (MEL) A way of distinguishing which MEs belong to the same MD. All MEs belonging to the same MD share the same MEG Level. An ITU-T concept.
- Maintenance End Point (MEP) An OAM reference point that can initiate and terminate OAM frames, and that reacts to diagnostic OAM frames.
- Maintenance Intermediate Point (MIP) An OAM reference point that reacts to diagnostic OAM frames initiated by MEPs.



Figure 66 Ethernet Service OAM Network Overview

ETH-BN is supported according to ITU-T G.8013/Y.1731 and Y.1730.

ETH-BN reports bandwidth changes of a monitored WAN interface with Bandwidth Notification Message (BNM) frames. When a bandwidth change is detected, ETH-BN reports it after a configured hold time. BNM frames contain the nominal (licensed) bandwidth and the actual bandwidth of the link.

MINI-LINK implementation of ETH-BN relies on a MEP to be configured on the node. It is used to determine on which interfaces the BNM frames should be sent. The MEP could be on a different interface, together with the up/down status these are used to determine the outgoing interfaces. Up and down MEPs are defined in IEEE 802.1ag.

A down MEP sends BNM frames out on only one interface. See Figure 67.

An up MEP sends BNM frames towards the switch rather than the egress direction of the interface. In this way the VLAN switching can send the BNM frames towards multiple outgoing interfaces. See Figure 68.



Figure 67 ETH-BN with Down MEP



Figure 68 ETH-BN with Up MEP

By default, BNM frames are always sent, even when the link is at full speed or drops to zero. This behavior can be changed to sending only one BNM frame or no BNM frames when the link is restored to full speed, and the same configuration is possible for link failure.

Note: It is not recommended to turn off sending BNM frames for both full link speed and link failure.

In this case, once the ETH-BN stops sending BNM frames, the receiving router cannot determine whether the link is restored to full speed, or it failed.

11.4 Port Mirroring

Port mirroring can be used for packet traffic crossing in MINI-LINK 6600. The port mirroring feature sends a copy of all packets seen on one LAN/WAN switch port to a LAN test port, see Figure 69. The test port is named Mirror to Port (MTP).

Note: Deep buffering must be disabled for the MTP ports.

This is commonly used for network appliances that require monitoring of network traffic, such as an intrusion detection system and troubleshooting.



Figure 69 Port Mirroring – Mirroring a WAN Interface to a LAN Port

It is possible to enable ingress and egress port mirroring separately.

If the MTP VLAN tag is configured, both ingress and egress mirrored traffic is always tagged at the MTP with the MTP VLAN tag as outer tag.

12 L3 Services

The NE provides Layer 3 Virtual Private Network (L3VPN) for IPv4 and IPv6 services over an IPv4-based MPLS transport network.

Table 10 lists the features that are possible to configure for Layer 3 solutions.

Business Application		Routing Options	Services
L3VPN	L3VPN BGP/MPLS	CE-PE routing options:	QoS
	Layer 3 VPN (L3 PE router)	OSPFv2	Route filters
		OSPFv3	LAG or ECMP
	P router	Static	BFD
		eBGP	MPLS OAM
		Core routing options:	
		iBGP	
		MPLS	
		LSP: LDP, RSVP-TE	
		IGP: OSPFv2, IS-IS	
Layer 3 IP Router		Core routing options:	QoS
		BGP	Route filters
		IGP: OSPFv2, OSPFv3,	LAG or ECMP
		15-15	BFD

Table 10	Configurable Features	for Layer 3 Sol	utions

12.1 L3 IP Router

The NE supports standard IP routing that forwards packets to their final destination using intermediate nodes. Each node looks up the destination IP address and forwards the packet toward the destination through routes collected in a routing table.

On the NE, route information is collected from the different routing protocols in the RIB on the controller card, which calculates the best routes and downloads them to the FIB stored on the NE. The RIB process collects routes to directly attached devices, configured static IP routes, and routes learned dynamically from OSPF and BGP.

When a network event causes routes to go down or become unavailable, routers distribute routing update messages that are propagated across networks, causing a recalculation of optimal routes. Routing algorithms that converge slowly can cause routing loops or network outages.

12.2 L3VPNs

The NE provides the following Layer 3 services and solutions:

- End-to-end Layer 3 connection over an IP/Multiprotocol Label Switching (MPLS) core network
- Business Virtual Private Networks (VPNs), such as Border Gateway Protocol (BGP)/MPLS L3VPNs.
- Core routing solutions, such as P router, in an IP/MPLS core network
- Multiple virtual routing instances via Virtual Routing Forwarding (VRF)
- The NE can provide Layer 2 Ethernet transport services and Layer 3 unicast routing on the same NE without separate Layer 2 and Layer 3 devices.

Figure 70 illustrates MINI-LINK 6600 in a Layer 3 network with VPNs.



Figure 70 L3VPNs

12.3 Supported Routing Protocols

The NE supports the following routing protocols:

- Static Routing
- OSPF

OSPF is an Interior Gateway Protocol (IGP) that uses Link-State Advertisements (LSAs) to inform other routers of the state of the sender links. In a link-state routing protocol, each NE distributes information about its interfaces and neighbor relationships. The collection of the link states forms a database that describes the Autonomous System (AS) topology. As OSPF routers accumulate link-state information, they use the Shortest Path First (SPF) algorithm to calculate the shortest path to each node, which forms the basis for developing routing information for that AS.

The following summarizes the OSPF features:

- An NE can be a part of a non-stub area, stub area or totally stub area.
- An NE can act as an Internal Router (IR), as an Area Border Router (ABR), or as an Autonomous System Border Router (ASBR).
- Virtual links are supported, which is useful when an area needs to be split in two parts.
- Link summarization is supported, which is used in the ABR to minimize the routing information distributed to the backbone and/or other areas.
- IS-IS

Intermediate System to Intermediate System (IS-IS) is a routing protocol designed to move information efficiently within a computer network, a group of physically connected computers or similar devices. It accomplishes this by determining the best route for data through a packet-switched network.

IS-IS enables the separation of an IS-IS domain into separate areas, improving the scalability of the IS-IS network. IS-IS level 2 and level 1-2 node functionality is required on the borders of the areas.

A multi-area IS-IS network can be used for MPLS transport with inter-area (Label Switched Paths) (LSPs) either by using LDP (RFC 5283) or by using RSVP-TE (RFC 5151).

– BGP

BGP, an EGP based on distance-vector algorithms, uses TCP as its transport protocol. BGP operates between two BGP nodes, called BGP speakers. After a TCP connection is established, the two BGP speakers exchange dynamic routing information over the connection. The exchange of messages is a BGP session between BGP peers.

- Multiprotocol internal BGP (MP-iBGP) is used to exchange routing information and service label allocations between VPN endpoints in PE nodes in a L3VPN network.
- External BGP (eBGP) is run by routers that belong to different autonomous systems and exchange BGP updates. It can for example be used to exchange routing information between PE and CE nodes.

— Routing Policies

Routing policies allow network administrators to enforce various routing policy decisions on incoming, outgoing, and redistributed routes. The tools used to configure routing policies include BGP AS path lists, BGP community lists, IP-prefix lists, and route maps with match and set conditions.

12.4 MPLS Networking

The NE supports MPLS to forward packets through a network efficiently.

In a conventional IP Network, routers forward packets through the network from one NE to the next, with each NE making an independent forwarding decision by analyzing the packet header. Packet processing often causes considerable forwarding delay. With MPLS, the complete analysis of the packet header is performed only once when it enters an MPLS-enabled network.

12.4.1 MPLS-Based Solutions

The NE supports L3VPN solutions using MPLS networks in which customer connectivity among multiple remote sites is deployed across a shared central infrastructure and still provides the same access or security as a private network.

— BGP/MPLS VPNs

Layer 3 BGP/MPLS VPNs are a collection of policies that control connectivity among a set of sites. A customer site is connected to the service provider network, often called a backbone, by one or more ports. The service provider associates each port with a VPN context.

A BGP/MPLS VPN allows you to implement a wide range of policies. For example, within a VPN, you can allow every site to have a direct route to every other site (full mesh), or you can restrict certain pairs of sites from having direct routes to each other (partial mesh).

12.4.2 Label Distribution

To communicate labels and their meanings among LSRs, MPLS uses a label distribution protocol (LDP or RSVP-TE), which enable dynamic label allocation and distribution in an MPLS network.

- LDP: If an LSR is enabled with LDP, it can establish LSPs to other LSRs in the network. LDP creates label bindings by assigning labels to the connected routes. Then, LDP advertises these bindings to the neighbors. When LDP learns about a label binding from a neighbor, it advertises these label bindings to its own neighbors as well. When an LSR advertises a label binding for a route, the LSR is advertising the availability of an LSP to the destination of that route. LDP can learn several LSPs from different neighbors for the same route. LDP must be configured with an IGP, for example, OSPF. LDP only assigns a label to those routes which are selected by the underlying IGP. In a multi-area network, to eliminate the need for leaking "/32" loopback addresses, the LDP matches the longest IGP prefix according to RFC 5283.
- RSVP-TE: The LSRs can establish LSPs to other LSRs using RSVP-TE. The RSVP-TE allows the establishment of MPLS LSPs, considering the network constraint parameters, such as link coloring or available bandwidth. Unlike LDP, where LSP paths always follow the IGP topology, the RSVP-TE allows the definition of explicit paths defined by the user. To do so, use one of the following methods:
 - Fully specify the entire path, using strict hops.
 - Set the IS-IS Level 1-2 routers as loose hops, then let the nodes do the CSPF calculation for the remaining segments in the partial Explicit Route Object (ERO) of the LSP from the other IS-IS area (RFC 5151).

12.4.3 RSVP-TE Fast Reroute

RSVP-TE based MPLS networks can offer restoration times in the range of several seconds using reconvergence of the protocols in case of a link or node failure. To improve resiliency times to 50 ms or below, the Fast Reroute solution can be applied.

Fast Reroute (FRR) provides local protection, that is, FRR is used to protect local link or node failures. To protect a link failure, any router - except for the egress node - can set up a backup (bypass/ detour) LSP avoiding the protected link to its next-hop router. On the other hand, to protect a node failure, a router has to set up a backup LSP to its next-next-hop router.



Figure 71 RSVP-TE Fast Reroute

The routers monitor the next hops by using Bidirectional Forwarding Detection (BFD), and immediately reroute to the backup LSP in case the BFD session is broken. Using BFD timers of 10 ms will result in detection times of 30 ms and rerouting times below 50 ms.

There are two schemes in RSVP-TE FRR, namely one-to-one backup and facility backup:

- In one-to-one backup, one backup LSP has to be established for every protected LSP. Specifically, if there are N protected LSPs, N backup LSPs are needed.
- In facility backup, one backup LSP can be used to protect all LSPs using the same outgoing interface/link; this requires pushing another (third) label to the packets.

Apparently, facility backup scales better than one-to-one backup, but the implementation is more complex due to the use of the extra label. In ML66, the one-to-one backup is supported.

Note: In ML66, the 50ms switchover time criterion is not guaranteed on managed node restart.

The backup LSP takes the traffic until the routing protocols reconverge and set up a new primary LSP (also with new backup LSPs). From that point, the new LSP takes over the traffic, while the old LSP and its backup LSPs are torn down.

12.4.4 RSVP-TE Path Protection

RSVP-TE Path Protection enables establishing primary and backup LSPs. In case of a failure along the primary LSP, the head-end will switch over to the backup LSP with a switchover time less than a second.

The primary and backup LSPs are manually established end-to-end between the head-end and tail-end nodes, using TE techniques such as explicit hops or administrative groups. One LSP is configured as primary and the other as secondary. In case of the failure of the primary LSP, the secondary LSP takes over the traffic, shown in Figure 72.



Figure 72 LSP Switchover

BFD over LSP (RFC5884) can be used for end-to-end connectivity monitoring of the primary LSP, enabling a switchover below 50ms.

This feature does not require any specific licenses.

12.4.5 BGP Alternate Path Routing

BGP Alternate Path Routing (APR) enables dual-homing of a Cell Site Provider Edge (CS-PE) node to two Switch Site Provider Edge (SS-PE) nodes, see Figure 73. The CS-PE has two routes to a destination and computes a primary and an alternate route to the two SS-PEs. The CS-PE downloads both the primary and alternate routes to the MPLS FTN Table. The alternate BGP remote nexthop is used only if the primary BGP remote nexthop is not reachable. By running multi-hop BFD to monitor the states of the BGP neighbors, the cell site PE can react very quickly to the failure of the primary SS-PE node and switch over to the alternate SS-PE node. Depending on the BFD timer settings, recovery time can be as low as a few 100 milliseconds.



Figure 73 BGP Alternate Path Routing

12.4.6 Seamless MPLS

Seamless MPLS offers a solution for extending MPLS across Autonomous Systems. An additional label is added to the MPLS header that is used for label-switching the traffic at AS Border Routers. BGP Labeled Unicast (BGP-LU) is used to distribute label allocations from the AS borders towards the routers in the connected ASs.





12.5 Supported IP Protocols

The NE supports the following IP service protocols:

— ARP

The ARP implementation is consistent with RFC 826, An Ethernet Address Resolution Protocol, also called Converting Network Protocol Addresses to 48.bit Ethernet Address for Transmission on Ethernet Hardware. In addition, the NE provides a configurable ARP entry-age timer and the option to delete expired dynamic ARP entries automatically.

— BFD

The NE supports RFC 5880, BFD. BFD is a simple Hello protocol that is similar to the detection components of some routing protocols. A pair of routers periodically transmits BFD packets over each path between the two routers. If a system stops receiving BFD packets after a predefined time interval, a component in the bidirectional path to the neighboring NE is assumed to have failed. A path is declared to be operational only when two-way communication has been established between the systems. To establish BFD sessions, configure one or more BFD clients on the same interface as BFD. BFD clients are routing protocols, which use BFD events to detect link failures; for example, BFD clients can be BGP, OSPF, and other applications.

— NDP

The NDP implementation is based on RFC 4861.

— NTP

The NE supports versions 1, 2, and 3 of the Network Time Protocol (NTP). On the NE, NTP operates only in client mode. A remote NTP server can synchronize the NE, but the NE cannot synchronize remote clients.

- **Note:** Before using NTP, the NE must be configured with the IP address of one or multiple NTP servers.
- DHCP

DHCP dynamically leases IP address information to host clients. The NE provides the following DHCP support.

• DHCPv4 relay server

The NE acts as an intermediary between an external DHCPv4 server and the client. The NE forwards requests from the client to the DHCPv4 server and relays the responses from the server back to the client.

• DHCPv6 relay server (only for DCN)

The NE relays or passes client requests for IPv6 addresses and prefixes to an external server (third-party equipment). The external server then provides the IPv6 addresses or prefixes to the client.

13 TDM

MINI-LINK 6600 provides an E1/DS1 traffic routing function that facilitates the handling of PDH traffic aggregation. This function enables interconnection of E1/DS1 traffic connections going through the NE.

Plug-in units connect $n \times E1/DS1$ to the backplane, where the traffic is cross-connected to another plug-in unit. The E1s/DS1s are unstructured with independent timing.

The traffic routing function is configured from MINI-LINK Node GUI, locally or remotely.

Note: For MINI-LINK 6651, only E1s for ETSI are supported. No support is available for DS1s (ANSI) or SDH/SONET in MINI-LINK 6651.

MINI-LINK 6600 provides STM-1 interfaces via LTU 1002 for SDH.

13.1 TDM Protection Mechanisms

This section describes the protection mechanisms provided by the Basic Node. Protection of the radio link is described in Radio Link.

13.1.1 Overview

To ensure high availability, MINI-LINK 6600 provides protection mechanisms on various layers in the transmission network as illustrated in Figure 75.

- Network layer protection using the 1+1 SNCP mechanism provides protection for the sub-network connection **a-b** in Figure 75. Network layer protection uses only signal failure as switching criterion.
- Physical link layer protection using MSP 1+1 indicated by the link c between two adjacent NEs 1 and 2 in Figure 75. Physical link layer protection uses both signal failure and signal degradation as switching criteria.
- By routing the protected traffic in parallel through different physical units, equipment protection can also be achieved. An example using two plug-in units is shown for the NEs 1 and 2 in Figure 75.



Figure 75 The NE Provides High Availability Through Various Protection Mechanisms

Network layer and physical link layer protection share the following characteristics:

Permanently Bridged	Identical traffic is transmitted on the active and the passive physical link/connection.
Uni-directional	Only the affected direction is switched to protection. The equipment terminating the physical link/connection in either end will select which line to be active independently.
Non-revertive	No switch back to the original link/connection is performed after recovery from failure. The original active link/connection is used as passive link/connection after the protection is reestablished.
1+1	One active link/connection and one passive (standby) link/connection.
Automatic/Manual switching mode	In automatic mode, the switching is done based on signal failure or signal degradation. Switching can also be initiated from the management system provided that the passive link/connection is free from alarms.
	In manual mode, the switching is only initiated from the management system, regardless of the state of the links/connections.

13.1.2 Network Layer Protection

13.1.2.1 1+1 E1/DS1 SNCP

1+1 E1/DS1 Sub-Network Connection Protection (1+1 E1/DS1 SNCP) is a protection mechanism used for network protection on E1/DS1 level, between two NEs. It is based on the simple principle that one E1/DS1 is transmitted on two separate E1/DS1 connections.

The switching is performed at the receiving end where the two connections are terminated. It switches automatically between the two incoming E1s/DS1s in order to use the better of the two. The decision to switch is based on signal failure of the signal received (LOS or AIS).

At each end of the protected E1/DS1 connection, two E1/DS1 connections must be configured to form a 1+1 E1/DS1 SNCP group.

An operator may also control the switch manually.

The connections may pass through other equipment in between, provided that AIS is propagated end-to-end.

The 1+1 E1/DS1 SNCP function is independent of the 1+1 radio protection and the MSP 1+1.



Figure 76 1+1 E1/DS1 SNCP Principle

Performance data is collected and fault management is provided for unprotected as well as protected VC/VT interfaces (that is the 1+1 E1/DS1 SNCP group). This gives accurate information on the availability of network connections.

13.1.2.2 Ring Protection for PDH and SDH



Figure 77 Network Topologies

The 1+1 SNCP mechanism described in the previous section can be used to create protected ring structures in the microwave network. In a ring topology, all nodes are connected so that two nodes always have two paths between them.

A connection entering a ring at one point and exiting at another point can therefore be protected with a 1+1 SNCP group configured at each end of the connection. The traffic is transmitted in both directions of the ring and the traffic is received from two directions at the termination point.

In this solution, the ring network can tolerate one failure without losing transmission. When the failure reoccurs, the affected connections are switched in the other direction.

In a MINI-LINK network, these ring structures can be built using PDH Radio Terminals with capacities of up to 80×2 Mbps, and using SDH Radio Terminals with LTU 1002 with capacities up to 63×2 Mbps.

Capacity is distributed from a common feeder node to the ring nodes where it is dropped off to star or tree structures as shown in Figure 78.

As an example, consider the nodes **A** and **E** in Figure 78. To protect the connection from **A** to **E** the two alternative connections from **A** to **E** must be defined as a 1+1 SNCP group at **A** and as a 1+1 SNCP group at **E**.

Similarly, to protect the connection from **A** to **C**, the two alternative connections between **A** and **C** must also be configured as two 1+1 SNCP groups at **A** and **C**.



Figure 78 Example of Ring Protection with 1+1 SNCP

The 1+1 SNCP function can be used to build protection in more complex topologies than rings, using the same principle.

13.1.3 MSP 1+1 for SDH

Note: SDH is only available for ETSI, and not for ANSI (SONET).

The LTU 1002 STM-1 interface supports Multiplexer Section Protection (MSP) 1+1. This SDH protection mechanism provides both link protection and equipment protection. Its main purpose is to provide maximum protection at the interface between the microwave network and the optical network.

MSP 1+1 requires two LTU 1002 plug-in units configured to work in an MSP 1+1 pair, delivering only one set of 63×E1 to the backplane at a time as illustrated in Figure 79. The unit intercommunication is done over the BPI bus.





The switching is done automatically if the following is detected:

- Signal Failure (SF): LOS, LOF, MS-AIS, or RS-TIM
- Signal Degradation (SD) based on MS-BIP Errors (BIP-24)
- Local equipment failure

The operator can also initiate the switching manually.

The switch logic for MSP 1+1 is handled by the Device Processor of the unit.



Figure 80 MSP 1+1 Principle for LTU 1002

13.2 TDM over Packet using CES

TDM traffic can be transported over Ethernet by using the Circuit Emulation Service (CES).

Note: CES is configurable in a MINI-LINK 6600 node that has an LTU 1002 installed or in a MINI-LINK 6651/4 node.

MINI-LINK supports both TDM and packet data transportation, with TDM being more traditional and packet-only networks slowly replacing old TDM or hybrid networks. CES enables a packet-only network to support TDM data transportation, meaning a hybrid network is not required. CES converts traditional TDM traffic at a node to Ethernet traffic and the traffic can then be sent over a packet network.

The MINI-LINK 6600, equipped with an LTU 1002, or the MINI-LINK 6651/4 converts the TDM traffic into a packet format and then it is sent over a Native Ethernet network.

CES adds an additional overhead to all data, due to the specific requirements of packet data.

The following TDM encapsulation mode is provided:

Structure Agnostic Model — The complete E1/DS1 is encapsulated into a pseudowire.

Each TDM interface is transferred in a separate pseudowire. The pseudowires can be transported directly over Ethernet using MEF8 protocol headers. The underlying Ethernet network can operate in Customer Mode or Provider Mode. The customer VLAN tags are set per pseudowire.

LTU 1002 provides CES services with different options depending on the selected LTU 1002 profile. For details on the available board profiles, see Section 15.15 on page 211.

13.2.1 Synchronization of TDM Interface

One of the basic tasks of the CES function is to reconstruct the timing on the transported TDM service.

MINI-LINK 6600 supports the following methods to synchronize the TDM transmit clock at one end of the packet network, to the TDM receive clock at the other end of the packet network:

- Loop Timing Method The TDM transmit clock is derived from the received TDM signal. It is required that the TDM network on both sides have access to the same network clock.
- Adaptive Method TDM transmit clock is derived based on the packet arrival rate from the packet network.
- Network Synchronization Method TDM transmit clock is derived from the SEC in network frequency synchronization mode.

The loop timing and adaptive methods are useful in scenarios when the TDM service has independent timing.

14 Synchronization

14.1 General Overview

Network synchronization is an important supporting function in telecommunication networks, since proper operation of different transport and radio technologies requires certain type and level of synchronization of the network equipment.



Different synchronization types are described in Figure 81.

Figure 81 Different Synchronization Types

If there is no alignment between the time instances of two system clocks, then the two clocks are not synchronized (Figure 81, top left). If the periodicity (or rate of change) of the clock instances in two systems are equal, then the two clocks are frequency synchronized (Figure 81, top right). Phase synchronization means that time instances of the two remote system clocks occur at exactly the same time, but absolute time is not known by the systems, so their significant instances may be unaligned (Figure 81, bottom left). In case absolute time information is assigned to each time instance of the clocks and they are aligned, then the two clocks are time synchronized (Figure 81, bottom right).

Though phase and time synchronization do not exactly mean the same, they are usually used interchangeably in synchronization terminology and it always means time synchronization in telecommunication networks.

Frequency synchronization is required by TDM-based transport networks and FDD-based radio technologies (GSM, WCDMA, LTE-FDD, and so on), while time synchronization is required by some advanced radio technologies (LTE-TDD, eMBMS, and so on).

In MINI-LINK, frequency synchronization is handled by the Netsync function, that contains a Synchronous Equipment Timing Generator (SETG). The SETG can be frequency synchronized to any physical interface that can carry synchronization. The recovered clock is handed over to the Frequency synchronization function as frequency reference interface. Out of all potential synchronization reference interfaces, four interfaces can be selected to be connected to the synchronization source selection logic.

Time synchronization is provided by the PTP described in the IEEE1588-2008 standard. It can be used in G.8275.1 profile mode, and it synchronizes the Real Time Clock (RTC) of the node to an external time reference through exchange of PTP packets supported by HW timestamping. Time synchronization function relies on stable clock frequency that is provided using the Netsync function based on physical layer frequency synchronization signals. Therefore it is also prerequisite to enable and lock the Netsync function to a Primary Reference Clock (PRC)/Primary Reference Source (PRS) traceable synchronization signal before time synchronization is activated.

Since the PTP function communicates over standard traffic interfaces it is a prerequisite to create the traffic configuration of the interfaces that are intended for PTP usage.

Figure 82 shows the node operation when PTP is configured in G.8275.1 profile and time recovery is supported by the frequency synchronization function.





14.2 Frequency Synchronization Technologies

MINI-LINK 6600 supports the following frequency synchronization functions:

- Sync over Radio Link
- Synchronous Ethernet
- Sync over SDH (only for ETSI)
- Sync over PDH
- Sync over 2 MHz/2048 kbps
- Sync over PTP (G.8265.1 Mode)

14.2.1 Sync over Radio Link

Quality Level (QL) is communicated over radio link side channel in order to support QL based reference selection.

14.2.2 Synchronous Ethernet

Synchronous Ethernet is supported on the various interfaces as shown in Table 11.

Table 11SyncE Support on Ethernet Interfaces

	Interface					
Unit	LAN RJ45	LAN RJ45	LAN SFPo	LAN SFPo	LAN SFPo	LAN SFPe
	100 Mbps	1 Gbps	1 Gbps	2.5 Gbps	10 Gbps	
NPU 1002	Yes	Yes	Yes	No	Yes	No
NPU 1003	Yes	Yes	Yes	No	Yes	No
NPU 1005	Yes	Yes	Yes	No	Yes	No
PNM 1001	Yes	Yes	Yes	No	Yes	No
PNM 1002	Yes	Yes	Yes	No	Yes	No
ETU 1001	Yes	Yes	-	-	-	_
ETU 1002	_	-	Yes	-	Yes	_
MINI-LINK 6651/1, 6651/2, or 6651/3	Yes	Yes	Yes	Yes	-	No
MINI-LINK 6651/4	Yes	Yes	Yes	_	Yes	No

Synchronous Ethernet enables propagation of frequency synchronization clock over Ethernet interfaces. It also supports QL communication using Ethernet Synchronization Status Message (ESMC) messages according to ITU-T G.8264. The ESMC handling can be disabled for interfaces in order to provide interoperability to other equipment that do not support the ESMC protocol.

Note: Disabling ESMC handling is not recommended and such usage is under the responsibility of the operator.

The terms "ESMC" and "Synchronization Status Message (SSM)" are used interchangeably in the Synchronous Ethernet context in MINI-LINK documentation.

14.2.3 Sync over SDH

Note: Sync over SDH is not available for MINI-LINK 6651.

Note: LTU 1002 is required for MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, or MINI-LINK 6694 with NPU 1002, NPU 1003, or NPU 1005.

Frequency synchronization over SDH requires termination of the STM-1 in an LTU 1002.

Quality Level (QL) is communicated over SDH interfaces using Synchronization Status Message (SSM) in order to support QL based reference selection.

It is not supported to use a PDH carried inside an STM-1 as a sync source. The recommendation is to use the STM-1 as sync source instead.

14.2.4 Sync over PDH

- Note: LTU 1001 or LTU 1002 is required for MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, or MINI-LINK 6694 with NPU 1002 or NPU 1005.
- Note: Sync over PDH is not available for MINI-LINK 6651.

Any E1/DS1 interface can be used as frequency reference source.

E1s/DS1s are cross-connected transparently, that is the clock frequency on the outgoing E1/DS1 interface will always follow the clock frequency of its relevant E1/DS1 input.

QL forwarding on PDH interfaces is not supported. It is configurable to Squelch (AIS) outgoing PDH interfaces when frequency synchronization fails.

It is not supported to use a PDH carried in an STM-1 as a sync source. The recommendation is to use the STM-1 as sync source instead.

14.2.5 Sync over 2 MHz/2048 kbps

Sync over 2 MHz/2048 kbps is supported on the various interfaces as shown in Table 12.

Table 12 Sync over 2 MHz/2048 kbps Support

Unit	2 MHz/2048 kbps
NPU 1002	Yes
NPU 1003	Yes
NPU 1005	Yes
PNM 1001	Yes
PNM 1002	Yes
MINI-LINK 6651	No

The synchronization clock signal can be connected to external units by connecting the external units directly to an arbitrary 2 MHz/2048 kbps port. This can be used in the peripheral parts of the network where an RBS is connected to a MINI-LINK 6600.

QL forwarding on 2 MHz/2048 kbps interface is not supported. It is configurable to Squelch (AIS) outgoing signal when frequency synchronization fails.

14.2.6 Sync over PTP (G.8265.1 Mode)

On NPU 1002, NPU 1003, NPU 1005, PNM 1001, PNM 1002, and MINI-LINK 6651, the PTP protocol can be used to provide frequency synchronization of multiple slaves to a grandmaster according to the ITU-T G.8265.1 profile using UDP/IPv4 unicast transport. This profile does not require any PTP support from the network between the master and the slave. This means that transport nodes along the PTP packet path do not need to be aware of or to support the PTP protocol. However, when setting up a network for packet-based frequency synchronization following the G.8265.1 profile, the 1588 packets must be given the highest prioritization in the QoS hierarchy to minimize the packet delay variation of the PTP packets.

If the NE is configured as a frequency grandmaster, the frequency of the SETG clock is distributed.

If the NE is configured as a packet slave, the PTP recovered frequency is a candidate for the frequency synchronization source selection. In this profile, the delay request-response mechanism in two-way message exchange is supported. Peer-to-Peer operation is not supported.

The NE sends PTP event messages in one-step mode, while both one-step and two-step messages are accepted and processed properly. The only allowed encapsulation in this profile is IP unicast.

IP host functionality of the PTP entity is handled by the port-based L3 subinterfaces in the NE. The exact routing of the G.8265.1 PTP messages is determined by the L3 routing stack. The VLAN encapsulation is also determined by the L3 subinterface configuration.

Note: Only IPv4 is supported.

Only clock type Ordinary Clock (OC) is supported. Slave mode consists of a Single Slave-only OC (SOOC) instance.

To maximize the performance of the frequency synchronization, it is beneficial to forward the frequency by microwave link function, SyncE, PDH/SDH, or 2MHz inside the transport domain and hand the frequency over by G.8265.1 profile only at the edges of the transport domain, as it is shown in Figure 83. For this reason, G.8265.1 OC is supported only through LAN ports. On microwave links, the standard Netsync frequency forwarding is supported.



Figure 83 G.8265.1 Use Case

14.3 Time Synchronization Technologies

The time synchronization device is the Precision Time Protocol (PTP) entity that can receive packet based synchronization signal from any PTP capable interface, and can distribute the time synchronization information over any PTP capable interface or 1PPS/ToD output.

Note: The 1PPS/ToD interface is available on NPU 1002, NPU 1003, NPU 1005, PNM 1001, and PNM 1002.

To synchronize time via PTP, MINI-LINK 6600 requires a PRC/PRS-traceable frequency from the Frequency Synchronization function that is recovered from an L1 (non-PTP) source, for example Synchronous Ethernet or Sync over Radio Link, as described in the previous chapter.

When setting up a network for time synchronization, it is essential to configure PTP support throughout the synchronization path. In addition the PTP packets shall be given the highest prioritization in the QoS hierarchy in order to avoid PTP packet loss on congested links.

IEEE1588-2008 standard provides the general framework for packet-based time synchronization, including Boundary Clock (BC), Ordinary Clock (OC), and Transparent Clock (TC). However, in order to meet the functional and performance requirements of a certain application, further requirements defined in a so called

"profile" need to be supported as well. Applicable performance requirements are defined in ITU-T G.8273.2 for telecom devices, and in G.8271.1 for end-to-end network limits.

14.3.1 Default PTP Profile

PTP can be used to provide time synchronization according to the IEEE1588-2008 Annex J using Ethernet multicast transport. This profile assumes full network timing support from the network. This means that all transport nodes along the time synchronization chain from the grandmaster to the end application must support PTP.

In this profile, MINI-LINK 6600 supports the delay request-response mechanism in two-way message exchange. Peer-to-Peer operation is not supported.

MINI-LINK 6600 sends PTP event messages in one-step mode, while both one-step two-step messages are accepted and processed properly.

Default encapsulation for MINI-LINK 6600 is Ethernet multicast. MINI-LINK 6600 always sends packets with the forwardable multicast address, while both the forwardable (01-1B-19-00-00-00) and the non-forwardable (01-80-C2-00-00-0E) multicast addresses are accepted. Ethernet unicast can also be configured.

MINI-LINK 6600 supports PTP packet exchange in Customer and Provider Bridge modes.

The following VLAN tagging options are supported for the default PTP profile:

- No VLAN tag
- Single VLAN tag (Customer mode, or Provider mode with S-tag only)
- Double VLAN tags (Provider mode with both S-tag and C-tag)

The following clock types are supported for the default PTP profile:

- Boundary Clock (BC)
- Ordinary Clock (OC)
- Transparent Clock (TC)

A PRC/PRS-traceable frequency from the Frequency Synchronization function that is recovered from an L1 (non-PTP) source is required in all operation modes.

In TC mode the node does not terminate the PTP protocol, just compensates for the residence time within the node of each PTP packet. The node clock is not synchronized in time, but is required to be synchronized in frequency via a physical layer technology.

14.3.2 ITU-T G.8275.1 Profile

PTP can be used to provide time synchronization according to the ITU-T G.8275.1 profile. This profile assumes full network timing support from the network. This means that all transport nodes along the time synchronization chain from the grandmaster to the end application must support PTP in G.8275.1 profile.

In this profile, MINI-LINK 6600 supports the delay request-response mechanism in two-way message exchange. Peer-to-Peer operation is not allowed in this profile.

MINI-LINK 6600 sends PTP event messages in one-step mode, while both one-step and two-step messages are accepted and processed properly.

The only allowed encapsulation in this profile is Ethernet multicast. MINI-LINK 6600 sends by default packets with the forwardable address, and non-forwardable packet sending can also be configured. Both the forwardable (01-1B-19-00-00-00) and the non-forwardable (01-80-C2-00-00-0E) multicast addresses are accepted.

MINI-LINK 6600 supports PTP packet exchange in Customer and Provider Bridge modes.

The following VLAN tagging options are supported for the ITU-T G.8275.1 profile:

- No VLAN tag
- Single VLAN tag (Customer mode, or Provider mode with S-tag only)
- Double VLAN tags (Provider mode with both S-tag and C-tag)

The following clock types are supported for the ITU-T G.8275.1 profile:

- Telecom Boundary Clock (T-BC)
- Telecom Time Slave Clock (T-TSC)
- Telecom Transparent Clock (T-TC)

A PRC/PRS-traceable frequency from the Frequency Synchronization function that is recovered from an L1 (non-PTP) source is required in all operation modes.

14.3.3 Time Error Generation

Timing characteristics requirements for Telecom Boundary Clocks and Telecom Slave Clocks are defined in G.8273.2 standard for networks with full timing support. The standard defines limits for maximum absolute time error (max|TE|), permissible range of constant time error (cTE), and dynamic time error (dTE).
[ns]	T-BC Class B (G.8273.2) ⁽¹⁾	1G LAN ⁽²⁾⁽³⁾⁽⁴⁾	10G LAN ⁽²⁾⁽³⁾	Radio Link ⁽⁵⁾
max TE	70	70	70	70
Range of cTE	±20	±20	±20	±20
dTE	40	40	40	40

Table 13Performance Categories of the Recovered Clock

(1) ITU-T G.8273.2 is supported, except for clause 7.2.

(2) Timestamping in FE is not supported.

(3) SFP interfaces are validated only with officially supported SFP types. Performance is not guaranteed with 3rd party SFP types that is not on the official Ericsson list.

(4) While 1588 timestamping is supported on any type of SFPs, Synchronous Ethernet – which is prerequisite for phase synchronization – is not supported on electrical SFPs. However, 1588 phase synchronization in general can work, in case network synchronization is provided to the node over any other interface.

(5) For supported HW, see the compatibility document in the Planning folder

14.4 Precision Time Protocol

This chapter summarizes the general aspects of packet based synchronization based on the IEEE1588-2008 protocol.

IEEE Std 1588[™]-2008 is a standard for packet-based network synchronization that is suitable both for frequency and time synchronization. It describes hierarchical master-slave architecture for clock distribution. The title of the standard is Precision Time Protocol (PTP), and so the terms IEEE1588 and PTP are used interchangeably to refer to the standard and the protocol.

The standard defines different clock types, that are embedded into network elements and provides synchronization function to the devices. These devices maintain various data sets that describe clock device and clock signal properties. The clock devices exchange these data sets information that allows each device to identify peer clocks and autonomously decide master-slave hierarchy. When master-slave relationships are established, clocks start to periodically exchange timing messages that are used to maintain synchronization of the device clocks.

The protocol has the following operational phases in the normal execution of the protocol:

Network configuration

Transport network and IEEE1588 clocks need to be configured according to required application in order to provide connectivity between clock devices.

- Establishing master-slave hierarchy

Clocks start to exchange clock quality information, run the reference selection algorithm locally and automatically decide on synchronization topology

— Synchronizing the clocks

Clocks start to exchange time event messages and perform clock synchronization.

If there is a change in the network topology, clock states, or performance, the master-slave hierarchy is automatically re-established, which may lead to re-synchronization of clocks.

IEEE1588 standard defines the framework for packet-based network synchronization in general and does not restrict the usage for any application, as well as does not define any performance requirements that would be needed for certain applications. Instead it defines the concept of Profile, which – within the framework of the standard – allows organizations to specify certain parameter subsets and operational requirements that are needed to achieve proper performance of certain applications. A PTP profile is a set of required options, prohibited options, and the ranges and defaults of configurable attributes.

ITU-T defined several standards that provide detailed requirements for packet-based network synchronization for telecommunication networks including definition of terminology, network-level requirements, clock definitions with performance levels, and profiles.

In order to support IEEE1588-2008 packet-based synchronization in any supported profile, MINI-LINK 6600 requires at least an NPU (acting as clock device) and time stamping capable interfaces such as LAN and WAN ports.

If time recovery is active, MINI-LINK 6600 can jump to phase/time holdover state in order to maintain the accurate time information. The prerequisite of phase holdover is that MINI-LINK 6600 previously was in phase locked state and the underlying frequency synchronization is stable. The period of holdover time is configurable from 0 to infinite.

MINI-LINK can monitor the performance of PTP, for more information, see Section 16.9.8 on page 251.

In MINI-LINK 6600, all Ethernet interfaces support time stamping except LAN RJ45 ports running at 10 Mbps or 100 Mbps.

Table 14 summarizes the 1PPS/ToD interface availability.

	1PPS/ToD input	1PPS/ToD output
NPU 1002	HW prepared	Yes
NPU 1003	HW prepared	Yes
NPU 1005	HW prepared	Yes
PNM 1001	HW prepared	Yes

Table 14 1PPS/ToD Support

Table 14 1PPS/ToD Support

	1PPS/ToD input	1PPS/ToD output
PNM 1002	HW prepared	Yes
MINI-LINK 6651	HW prepared	HW prepared

15 Hardware

15.1 MINI-LINK 6651

MINI-LINK 6651 is a compact node with a dual carrier modem, suitable for tail sites.

It has a replaceable Fan Unit (FAU) that provides cooling for the indoor part.

There are four variants, MINI-LINK 6651/1 (Figure 84), MINI-LINK 6651/2 (Figure 85), MINI-LINK 6651/3 (Figure 86), and MINI-LINK 6651/4 (Figure 87).

MINI-LINK 6651/3 is single carrier (only 1+0 is possible) and has no fan unit.



Figure 84 MINI-LINK 6651/1



Figure 85 MINI-LINK 6651/2



Figure 86 MINI-LINK 6651/3



Figure 87 MINI-LINK 6651/4

MINI-LINK 6651 can be fitted in a standard 19" or metric rack. The height of a MINI-LINK 6651 is 1U.

MINI-LINK 6651 supports Single WAN 1+0. MINI-LINK 6651/1, MINI-LINK 6651/2, and MINI-LINK 6651/4 have a dual carrier modem and also support 2+0 Radio Link Bonding (RLB) and 1+1 Radio Link Protection (RLP). MINI-LINK 6651/2 and MINI-LINK 6651/4 also support Dual WAN 2x(1+0).

MINI-LINK 6651 can be configured in different ways:

— Single Direction 1+0

Supported by all MINI-LINK 6651 variants (MINI-LINK 6651/1, MINI-LINK 6651/2, MINI-LINK 6651/3, and MINI-LINK 6651/4).

— Bonded Single Direction

When configured as a bonded single direction link, the capacity of the two carriers are combined and only one switchport is used.

Supported by MINI-LINK 6651/1, MINI-LINK 6651/2, and MINI-LINK 6651/4.

— Protected Single Direction

When configured as 1+1 Radio Link Protection, both carriers are used for Radio Link path protection, and a single switchport is used.

Supported by MINI-LINK 6651/1, MINI-LINK 6651/2, and MINI-LINK 6651/4.

- Dual Directions

When configured as a 1+0 dual directions link, each carrier has its own connection to a switchport and both carriers can optionally be configured as a Hybrid radio link, that is combining TDM and Ethernet traffic on the same radio path.

Supported by MINI-LINK 6651/2 and MINI-LINK 6651/4.

MINI-LINK 6651 has a high-capacity PDH/Ethernet modem with support for Adaptive Coding and Modulation (ACM), see Section 3.2 on page 12, and modulation schemes up to 8192 QAM.

MINI-LINK 6651/1, MINI-LINK 6651/2, and MINI-LINK 6651/4 supports Radio Link Bonding (RLB), see Section 3.5 on page 17 and XPIC (including XPIC Recovery and Restore), see Section 3.9 on page 28.

MINI-LINK 6651 provides physical layer Sync over Radio Link.

MINI-LINK 6651 carries PDH traffic and the Radio Link capacity is split between packet and circuit transport is configured with PDH capacity settings.

MINI-LINK 6651 supports control of the ratio between Ethernet and PDH traffic sent over Hybrid Radio Links. The Ethernet part of the aggregated capacity is set with E1 granularity.

MINI-LINK 6651/4 supports CES, enabling E1s being transported over packet. The following board profiles can be set for MINI-LINK 6651/4:

PDH only	Provides $8\times E1s$ through the front LAN port and $16\times E1s$ over the radio.

CES only Provides 8×CES E1s transported over packet.

MINI-LINK 6651/2 and MINI-LINK 6651/4 supports Ethernet and TDM in two directions. MINI-LINK 6651/1 and MINI-LINK 6651/3 only supports traffic (Ethernet and TDM) in one direction.

MINI-LINK 6651 supports the following Ethernet-related features:

- Store and forward MAC (IEEE802.1D-2004) and VLAN-based switching (IEEE802.1Q-2005 and IEEE802.1ad-2005).
- Port-based VLAN (up to 1,000 VLANs in the range of 1-4094).
- MAC address table (32,000 entries) with learning and aging.
- IPv4 and IPv6
- Static unicast/multicast routes.
- Jumbo frames for LAN and Packet Link WAN ports.
- Multiple internal Layer 1 and Layer 2 Ethernet connections.
- MSTP/RSTP/LAG protection mechanism.
- Link Aggregation Control Protocol (LACP).
- Link Loss Forwarding for internal Layer 1 connections.
- Policing/color marking per ingress port with CIR/EIR and CBS/EBS.
- Policing/color marking per ingress port per QoS priority with CIR/EIR and CBS/EBS.
- Policing/color marking per ingress port per VLAN with CIR/EIR and CBS/EBS.
- Traffic shaping per egress port with CIR and CBS.
- Ethernet loop detection.
- Broadcast/multicast/destination lookup failure storm protection.
- Frame admission control.
- Classification/tagging of priority and VLAN ID.
- Congestion handling with eight priority queues (TC) per port.
 - Frame discard: WRED, color dropping, aging and tail drop.
 - TC scheduling: SP, WDRR.

- In-band management transport in Ethernet flow (DCN over VLAN).
- Management/diagnostics.
 - PM counters.
 - Link OAM.
 - Ethernet Service OAM.
 - Port mirroring.

MINI-LINK 6651/1 provides the interfaces presented in Figure 88. MINI-LINK 6651/2 provides the interfaces presented in Figure 89. MINI-LINK 6651/3 provides the interfaces presented in Figure 90. MINI-LINK 6651/4 provides the interfaces presented in Page 128.



Figure 88 MINI-LINK 6651/1 Front Interfaces



Figure 89 MINI-LINK 6651/2 Front Interfaces





Figure 91 MINI-LINK 6651/4 Front Interfaces

PDH Traffic Interfaces

The front panel provides $8 \times E1$ interfaces ($4 \times (2 \times E1)$) configured to 120 Ohm.

Note: Only E1s for ETSI are supported. No support is available for DS1s (ANSI) in MINI-LINK 6651.

Ethernet Traffic Interfaces

The front panel at MINI-LINK 6651/1 and MINI-LINK 6651/3 provides two fixed electrical 10/100/1000 BASE-T Ethernet interfaces and four 1000/2500BASE-T/X SFP+ interfaces. One of the fixed Ethernet ports can be configured as a site LAN port.

The front panel at MINI-LINK 6651/2 provides two fixed electrical 10/100/1000 BASE-T Ethernet interfaces and three 1000/2500BASE-T/X SFP+ interfaces. One of the fixed Ethernet ports can be configured as a site LAN port.

The front panel at MINI-LINK 6651/4 provides two fixed electrical 10/100/1000 BASE-T Ethernet interfaces and four 1000/10GBASE SFP/SFP+ interfaces. One of the fixed Ethernet ports can be configured as a site LAN port.

All 1000/2500BASE-T/X SFP+ interfaces are both electrical and optical.

The SFP/SFP+ cages can be equipped with Ericsson approved SFP/SFP+ modules, see MINI-LINK 6600 R1 Compatibility, Reference [1].

For the basic offering for MINI-LINK 6651, two Ethernet ports are enabled at the front. To get access to more ports, licenses are needed.

The high speed connections and the LAN ports enable MINI-LINK 6651 to aggregate traffic from multiple directions as well as dropping Ethernet traffic locally at the site.

MINI-LINK 6651 supports Packet Link WAN connections.

User Input/Output

Table 15 MINI-LINK 6651/4 User Input and Output Ports

Compact Node	Number of Input Signals	Number of Output Signals	Connector Type
MINI-LINK 6651/4	Тwo	One	RJ-45 connector

O&M Interface

A USB interface for an alternative MINI-LINK Node GUI connection is available.

RMM Interface

A SIM card slot is available for a Removable Memory Module (RMM).

Radio Interface

The front panel on MINI-LINK 6651/1, 6651/2, and 6651/4 provides two 60 V RAU connectors.

The front panel on MINI-LINK 6651/3 provides one 60 V RAU connector.

Note: The coaxial interface on modems and radios is equipped with built-in gas discharge tubes for lightning protection.

Adding components to this interface (for example, extra lightning protection) can impact function and performance and should only be considered after consulting with Ericsson.

Power

MINI-LINK 6651 provides two -48V DC connectors for power redundancy. Power redundancy is possible by using two external power sources, each of them connected to one of the two DC connectors on the MINI-LINK 6651 unit, through a dedicated power cable.

Radio Cables

The radio cables connecting the Radio with MINI-LINK 6651 are available in the following three diameters:

— Ø 7,6 mm, with lengths up to 100 m.

This cable can be directly connected to the MINI-LINK 6651.

- Ø 10 mm, with lengths between 100 and 200 m.
- Ø 16 mm, with lengths between 200 and 400 m.

15.1.1 DC Supply Voltage

Normal service voltage range

The system is able to run at full operation in this range.

Abnormal service voltage range

No function is guaranteed, but no permanent damage will occur when operated in this range.

The system is restored to full operation when returning to normal service voltage range.

Nominal voltage	-48 VDC
Normal service voltage range	-58.8 V to -40 V
Abnormal service voltage range	-60 V to -58.8 V
	-40 V to 0 V
Typical startup voltage	-44 V to -46 V
Typical shut-down voltage	-37.5 V to -39.5 V

Table 16 DC Supply Voltage Ranges

15.1.2 Fan Unit

Note: MINI-LINK 6651/3 has no fan unit.

Air-cooling is provided by the Fan Unit, see Figure 92.



Figure 92 Fan Unit

The Fan Unit has an automatic fan speed control and houses four internal fans.





The air enters at the front of the MINI-LINK 6651 and exits at the back.

15.1.3 Traffic Capacities for MINI-LINK 6651

MINI-LINK 6651 is used for the traffic capacities specified in Table 17 and Table 18 (ETSI), or in Table 19 (ANSI).

Note: For ANSI, only Ethernet traffic is supported.

			Co	apacity in Mb	ps		
Madulatia			Cho	annel Bandwi	dth		
n	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz
4 QAM STRONG	8	18	37	53	75	108	150
4 QAM	10	21	43	62	87	126	175
16 QAM STRONG	17	37	74	106	150	216	301
16 QAM	20	43	87	124	175	252	351

Table 17Aggregated PDH and Ethernet Capacity for MINI-LINK 6651 Adaptive
Physical Modes (ETSI)

			(Capacity in M	bps			
Madulatia	Channel Bandwidth							
n	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz	
32 QAM	26	54	109	156	220	317	441	
64 QAM	32	68	136	195	276	397	553	
128 QAM	38	80	161	231	326	469	653	
256 QAM	44	92	186	266	377	541	754	
512 QAM	47	99	198	284	402	577	804	
512 QAM LIGHT	50	105	211	302	427	614	854	
1024 QAM	53 ⁽¹⁾	110	221	316	447	643	895	
1024 QAM LIGHT	56 ⁽¹⁾	116	233	334	472	679	945	
2048 QAM	-	121 ⁽¹⁾	243	347	491	706	983	
2048 QAM LIGHT	-	127 ⁽¹⁾	256	365	516	742	1033	
4096 QAM	-	-	264	377	533	767 ⁽²⁾	1067 ⁽²⁾	
4096 QAM LIGHT	-	-	276	395	558	803 ⁽²⁾⁽³⁾	-	
8192 QAM	-	-	283	405	-	-	-	

(1) XPIC not supported.

(2) XPIC not supported on MINI-LINK 6651/1 and MINI-LINK 6651/3.

(3) Only supported by MINI-LINK 6651/2 and MINI-LINK 6651/4.

(4) 8192 QAM is not supported as static modulation. It is recommended to set radio output power to 0 dBm or higher.

Table 18	Maximum PDH Capacities for MINI-LINK 6651 Adaptive Physical
	Modes (ETSI)

		Capacity in E1s ⁽¹⁾					
			Cho	annel Bandwi	dth		
Modulation	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz
4 QAM STRONG	2×E1	6×E1	15×E1	23×E1	34×E1	40×E1	40×E1
4 QAM	3×E1	8×E1	18×E1	28×E1	40×E1	40×E1	40×E1
16 QAM STRONG	6×E1	15×E1	34×E1	40×E1	40×E1	40×E1	40×E1
16 QAM	7×E1	18×E1	39×E1	40×E1	40×E1	40×E1	40×E1
32 QAM	10×E1	24×E1	40×E1	40×E1	40×E1	40×E1	40×E1
64 QAM	13×E1	30×E1	40×E1	40×E1	40×E1	40×E1	40×E1
128 QAM	16×E1	36×E1	40×E1	40×E1	40×E1	40×E1	40×E1
256 QAM	19×E1	40×E1	40×E1	40×E1	40×E1	40×E1	40×E1
512 QAM	21×E1	40×E1	40×E1	40×E1	40×E1	40×E1	40×E1

 \leq

		Capacity in E1s ⁽¹⁾					
			Cho	annel Bandwi	idth		
Modulation	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz
512 QAM LIGHT	22×E1	40×E1	40×E1	40×E1	40×E1	40×E1	40×E1
1024 QAM	23×E1 ⁽²⁾	40×E1	40×E1	40×E1	40×E1	40×E1	40×E1
1024 QAM LIGHT	25×E1 ⁽²⁾	40×E1	40×E1	40×E1	40×E1	40×E1	40×E1
2048 QAM	-	40×E1 ⁽²⁾	40×E1	40×E1	40×E1	40×E1	40×E1
2048 QAM LIGHT	-	40×E1 ⁽²⁾	40×E1	40×E1	40×E1	40×E1	40×E1
4096 QAM	-	-	40×E1	40×E1	40×E1	40×E1	40×E1
4096 QAM LIGHT	-	-	40×E1	40×E1	40×E1	40×E1 ⁽³⁾	-
8192 QAM ⁽³⁾	-	-	40×E1 ⁽⁴⁾	40×E1 ⁽⁴⁾	-	-	-

(1) The table shows the capacity for MINI-LINK 6651/2. The capacities for MINI-LINK 6651/1 and 6651/3 are limited to 8×E1. The capacity for MINI-LINK6651/4 is limited to 16xE1.
(2) XPIC not supported.

(3) Only supported by MINI-LINK 6651/2 and MINI-LINK 6651/4.

(4) 8192 QAM is not supported as minimum modulation.

Table 19	Ethernet Capacity for MINI-LINK 6651 Adaptive Physical Modes
	(ANSI)

		Capacity in Mbps									
		Channel Bandwidth									
Modul ation	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz	
4 QAM STRO NG	12	26	37	39	53	67	75	81	108	150	
4 QAM	14	30	43	46	62	78	87	94	126	175	
16 QAM STRO NG	25	52	74	79	107	135	150	162	216	301	
16 QAM	29	61	87	93	125	157	175	189	252	351	
32 QAM	36	77	109	116	157	197	220	237	317	441	
64 QAM	46	96	136	146	197	247	276	297	397	553	
128 QAM	54	114	161	173	233	292	326	351	469	653	
256 QAM	63	131	186	199	268	337	377	405	541	754	
512 QAM	67	140	198	213	286	360	402	432	577	804	

					Capacity	in Mbps					
		Channel Bandwidth									
Modul ation	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz	
512 QAM LIGHT	71	149	211	226	304	382	427	459	614	854	
1024 QAM	74 ⁽¹⁾	156	221	237	319	400	447	481	643	895	
1024 QAM LIGHT	79 ⁽¹⁾	165	233	250	337	423	472	508	679	945	
2048 QAM	-	171	243	260	350	440	491	528	706	983	
2048 QAM LIGHT	-	180	256	273	368	462	516	555	742	1033	
4096 QAM	-	-	264	282	380	477	533	574	767 ⁽²⁾	1067 (2)	
4096 QAM LIGHT	-	_	276	296	398	500	558	601	803 ⁽²⁾	-	
8192 OAM (1)(3)(4)	_	-	-	304	409	-	-	-	-	_	

(1) XPIC not supported.

(2) XPIC not supported by MINI-LINK 6651/1 and MINI-LINK 6651/3.

(3) Only supported by MINI-LINK 6651/2 and MINI-LINK 6651/4.

(4) 8192 QAM is not supported as static modulation. It is recommended to set radio output power to 0 dBm or higher.

15.1.4 Hardware Handling

The system offers several functions for easy operation and maintenance.

- The configuration file is stored non-volatile on the RMM, and can also be backed up and restored using a local or central FTP server. The RMM storage thus enables replacement without using an FTP server.
- When a radio is replaced, no new setup has to be performed.
- Various restarts can be ordered from the management system. A cold restart can be initiated for an NE, and this type of restart disturbs the traffic. A warm restart restarts the control system and does not affect the traffic, which is possible due to the separated control and traffic system.
- MINI-LINK 6651 is equipped with temperature sensors. Overheated parts that exceed limit thresholds, are put in out of service by the control system. This is to avoid hardware failures in case of over-temperature, for example, due to a fan failure or an ambient temperature that is too high. The MINI-LINK 6651 is automatically returned to normal operation when the temperature is below the high threshold level.

There are two thresholds:

• The high temperature threshold:

The MINI-LINK 6651 raises a temperature alarm (critical). The MINI-LINK 6651 is still in full operation.

• The excessive temperature threshold:

The MINI-LINK 6651 shuts down (out of service).

- Access to inventory data like software and hardware product number, serial number, and version. User defined asset identification is supported, enabling tracking of hardware.
- Ethernet SFPs can be replaced with minimal user intervention, without de-configuring port properties.

15.1.5 Ethernet Switch

The Ethernet switch functionality is shown in Table 20.

Modes	Port roles	Comment	
Provider Bridge mode	CE-UNI	CEP, including default mapping rule	
	CN-UNI	CNP	
	I-NNI	_	
Customer Bridge mode	UNI	Including QinQ termination, default mapping rule	
	NNI	_	

Table 20Ethernet Switch Functionality

The switch is a managed VLAN switch (IEEE 802.1Q and IEEE 802.1D) and supports Provider mode (IEEE 802.1ad) and Customer mode (IEEE 802.1Q) switching.

The switch supports Jumbo frames. The Ethernet site LAN ports on MINI-LINK 6651 have interfaces that support auto-negotiation 10/100/1000/2500 Mbps and 10Gbps speed. The interfaces are physical RJ-45 or SFP connectors.

15.1.6 Ethernet LAN Ports

The Ethernet LAN ports are described in Table 21.

Table 21	Ethernet LAN Ports
----------	--------------------

Functionality	Description
Interface types	10/100/1000/2500 Mbps
	2.5/10 Gbps
	Auto negotiation or manual setting (IEEE802.3-2005).
Duplex mode	Full duplex.
	Auto negotiation or manual setting (IEEE802.3-2005).
Connectors/cables	RJ-45 with MDI/MDIX support for all electrical i/f.
	Single/multi-mode fiber with LC connector for the optical i/f.
SFP modules	SFP/SFP+ plug-in modules are supported with 1 Gbps, 2.5 Gbps, and 10 Gbps optical and 100/1000 Mbps electrical interfaces.
	For more information regarding SFP/SFP+, see MINI-LINK 6600 R1 Compatibility, Reference [1].
LED indicators	Each Ethernet interface has one green LED for network link indication.
Standard frame sizes	Up to 2000 bytes when used as an external interface (IEEE802.3as-2006).
	Up to 2048 bytes when used as an internal interface.
Jumbo frames	9,216 byte frame size for data applications.
Buffer capacity	Ethernet ports share a 180 MB buffering capacity. The buffer capacity sharing among the traffic queues is based on a fair adaptive dynamic algorithm.
Flow control	MINI-LINK 6651 supports the IEEE 802.3x standard to handle temporary congestion on ports configured as Layer 2 connections (connected to the switch). MINI-LINK 6651 only supports asymmetrical flow control. When a congestion threshold is reached in an NE downstream, the NE generates a PAUSE signal and MINI-LINK 6651 temporarily halts the transmission of Ethernet frames, but MINI-LINK 6651 cannot generate PAUSE signals to control the transmission process on connected equipment.

15.2 MINI-LINK 6654

MINI-LINK 6654 supports Ethernet (packet link), PDH, and a mix thereof in a Hybrid Radio Link sending Ethernet and PDH traffic simultaneously.

It consists of a rack, Enclosure 1401, that must be equipped with a fan, FAU 1401, and a PNM board, either PNM 1001 or PNM 1002.



MINI-LINK 6654 can also be equipped with up to two MMU 1001/1002, two LTU 1001/1002, two ETU 1001/1002, or a combination thereof, see Figure 94.

Unused slots in MINI-LINK 6654 must be equipped with Dummy Units, as this is essential to maintain the cooling within the magazines and to meet EMC requirements.

MINI-LINK 6654 can be fitted in a standard 19" or metric rack. The height of a MINI-LINK 6654 is 1U, see Figure 95.



Figure 95 Front interfaces on the MINI-LINK 6654

The following slots are paired in MINI-LINK 6654:

Slot 1 and Slot 3

Figure 96 shows the paired slots in MINI-LINK 6654.



Figure 96 Interconnected slots in MINI-LINK 6654

Figure 94 MINI-LINK 6654

15.2.1 PNM 1001

PNM 1001 is a board that is combo solution for PFU (Power-Filtering-Unit), NPU (Node-Processor-Unit) and MMU (Modem Unit), see Figure 97.



Figure 97 PNM 1001

The Modem part of PNM 1001 is a single carrier modem supporting single WAN (1+0). For modem capability please refer to MMU 1001 performance in 1+0.

PDH Traffic Interfaces

4×E1/DS1

User I/O (two users in, one user out)

Ethernet Traffic Interfaces

2×Electrical (10M/100M/1G)

2×SFP (1G)

2×SFP+ (1G/10G)

Sync Interfaces

1×2 MHz

1×1PPS+ToD

O&M Interface

1×USB interface for an alternative MINI-LINK Node GUI connection is available.

Radio Interface

1×60 V RAU connector.

Note: The coaxial interface on modems and radios is equipped with built-in gas discharge tubes for lightning protection.

Adding components to this interface (for example, extra lightning protection) can impact function and performance and should only be considered after consulting with Ericsson.

Power

 2×-48 V DC connectors for power redundancy.

Figure 98 shows the interfaces on the PNM 1001.



Figure 98 Front Interfaces on the PNM 1001

15.2.1.1 User Input/Output

The User Input ports can be used to connect user alarms to MINI-LINK Node GUI. Applications like fire alarms, burglar alarms, and low power indicators are easily implemented using these input ports. The User Input ports can be configured to be normally opened or closed.

The User Output ports can be used to export summary alarms of the accumulated severity in the NE to the supervision system of other equipment. The User Output ports can be controlled by the operator or triggered by one or several alarm severities.

Table 22	PNM 1001	User Input and	Output Ports
----------	----------	----------------	--------------

Plug-In Unit	Number of Input Signals	Number of Output Signals	Connector Type
PNM 1001	Тwo	One	Male 24 pin SOFIX connector

15.2.1.2 Traffic Capacities for PNM 1001

PNM 1001 is used for the traffic capacities specified in Table 23 and Table 24 (ETSI), or Table 25 and Table 26 (ANSI).

	Capacity in Mbps									
	Channel Bandwidth									
n	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz			
4 QAM STRONG	8	19	37	53	75	108	151			
4 QAM	10	21	44	62	88	126	176			
16 QAM STRONG	18	37	75	107	151	217	302			
16 QAM	21	43	87	124	176	253	352			
32 QAM	26	54	109	156	221	317	441			
64 QAM	33	68	137	196	277	397	553			
128 QAM	39	81	162	231	327	470	654			
256 QAM	45	93	186	267	377	542	754			
512 QAM	48	99	199	285	402	578	804			
512 QAM LIGHT	51	105	211	302	427	614	855			
1024 QAM	53 ⁽¹⁾	110	221	317	448	643	895			
1024 QAM LIGHT	56 ⁽¹⁾	117	234	335	473	679	946			
2048 QAM	-	121 ⁽¹⁾	243	348	492	706	983			
2048 QAM LIGHT	-	127 ⁽¹⁾	256	366	517	742	1033			
4096 QAM	-	-	264	378	534	767 ⁽¹⁾	1068 ⁽¹⁾			
4096 QAM LIGHT	-	_	276	395	559	_	-			
8192 QAM	-	-	-	-	-	-	-			

Table 23Aggregated PDH and Ethernet Capacity for PNM1001 Adaptive
Physical Modes (ETSI)

(1) XPIC feature not supported by HW.

Table 24 Maximum PDH Capacities for PNM 1001 Adaptive Physical Modes (ETSI)

	Capacity in E1s									
		Channel Bandwidth								
Modulation	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz			
4 QAM STRONG	2×E1	6×E1	15×E1	23×E1	34×E1	49×E1	70×E1			
4 QAM	3×E1	8×E1	18×E1	28×E1	40×E1	58×E1	80×E1			
16 QAM STRONG	6×E1	15×E1	34×E1	49×E1	70×E1	80×E1	80×E1			
16 QAM	7×E1	18×E1	33×E1	57×E1	71×E1	80×E1	80×E1			
32 QAM	10×E1	24×E1	50×E1	73×E1	80×E1	80×E1	80×E1			
64 QAM	13×E1	30×E1	63×E1	80×E1	80×E1	80×E1	80×E1			

			C	Capacity in E1	Ls		
			Ch	annel Bandwi	idth		
Modulation	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz
128 QAM	16×E1	36×E1	75×E1	80×E1	80×E1	80×E1	80×E1
256 QAM	19×E1	42×E1	80×E1	80×E1	80×E1	80×E1	80×E1
512 QAM	21×E1	45×E1	80×E1	80×E1	80×E1	80×E1	80×E1
512 QAM LIGHT	22×E1	48×E1	80×E1	80×E1	80×E1	80×E1	80×E1
1024 QAM	23×E1 ⁽¹⁾	51×E1	80×E1	80×E1	80×E1	80×E1	80×E1
1024 QAM LIGHT	25×E1 ⁽¹⁾	54×E1	80×E1	80×E1	80×E1	80×E1	80×E1
2048 QAM	-	56×E1 ⁽¹⁾	80×E1	80×E1	80×E1	80×E1	80×E1
2048 QAM LIGHT	-	59×E1 ⁽¹⁾	80×E1	80×E1	80×E1	80×E1	80×E1
4096 QAM	-	-	80×E1	80×E1	80×E1	80×E1 ⁽¹⁾	80×E1 ⁽¹⁾
4096 QAM LIGHT	-	-	80×E1	80×E1	80×E1	-	-
8192 QAM	-	-	-	-	-	-	-

(1) XPIC feature not supported by HW.

Table 25Aggregated PDH and Ethernet Capacity for PNM 1001 Adaptive
Physical Modes (ANSI)

		Capacity in Mbps									
		Channel Bandwidth									
Modul ation	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz	
4 QAM STRO NG	13	26	37	40	54	68	75	81	108	150	
4 QAM	15	31	43	47	63	79	87	95	126	175	
16 QAM STRO NG	25	53	74	80	108	135	150	162	217	301	
16 QAM	29	61	87	93	126	158	175	189	253	351	
32 QAM	37	77	109	117	157	198	220	237	317	441	
64 QAM	46	97	136	147	197	248	276	297	397	553	
128 QAM	55	114	161	173	233	293	326	352	470	653	
256 QAM	63	132	186	200	269	338	377	406	542	754	
512 QAM	67	140	198	213	287	360	402	432	578	804	

		Capacity in Mbps									
		Channel Bandwidth									
Modul ation	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz	
512 QAM LIGHT	71	149	211	226	305	383	427	460	614	854	
1024 QAM	75 ⁽¹⁾	156	221	237	319	401	447	482	643	895	
1024 QAM LIGHT	79 ⁽¹⁾	165	233	251	337	423	472	509	679	946	
2048 QAM	-	172	243	260	351	440	491	529	706	983	
2048 QAM LIGHT	-	180	256	274	369	463	516	556	742	1033	
4096 QAM	-	-	264	283	381	478	533	574	767 ⁽¹⁾	1067 (1)	
4096 QAM LIGHT	-	-	276	296	398	500	558	601	-	_	
8192 QAM	-	-	-	-	-	-	-	-	-	-	

(1) XPIC feature not supported by HW.

Table 26	Maximum PDH Capacities for PNM 1001 Adaptive Physical Modes
	(ANSI)

	Capacity in DS1s												
		Channel Bandwidth											
Modula	10	20	28	30	40	50	56MH	60	80	112			
tion	MHz	MHz	MHz	MHz	MHz	MHz	z	MHz	MHz	MHz			
4 QAM STRON G	5×DS1	14×DS 1	21×DS 1	23×DS 1	32×DS 1	40×DS 1	45×DS 1	49×DS 1	66×DS 1	80×DS 1			
4 QAM	7×DS1	17×DS 1	25×DS 1	27×DS 1	37×DS 1	47×DS 1	54×DS 1	58×DS 1	78×DS 1	80×DS 1			
16 QAM STRON G	13×DS 1	31×DS 1	45×DS 1	48×DS 1	66×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1			
16 QAM	16×DS	36×DS	53×DS	57×DS	77×DS	80×DS	80×DS	80×DS	80×DS	80×DS			
	1	1	1	1	1	1	1	1	1	1			
32 QAM	21×DS	46×DS	67×DS	72×DS	80×DS	80×DS	80×DS	80×DS	80×DS	80×DS			
	1	1	1	1	1	1	1	1	1	1			
64 QAM	27×DS	59×DS	80×DS										
	1	1	1	1	1	1	1	1	1	1			
128	32×DS	70×DS	80×DS										
QAM	1	1	1	1	1	1	1	1	1	1			
256	37×DS	80×DS	80×DS	80×DS	80×DS	80×DS	80×DS	80×DS	80×DS	80×DS			
QAM	1	1	1	1	1	1	1	1	1	1			
512	40×DS	80×DS	80×DS	80×DS	80×DS	80×DS	80×DS	80×DS	80×DS	80×DS			
QAM	1	1	1	1	1	1	1	1	1	1			

					Capacity	in DS1s						
		Channel Bandwidth										
Modula tion	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56MH z	60 MHz	80 MHz	112 MHz		
512 QAM LIGHT	43×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
1024 QAM	45×DS 1 ⁽¹⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
1024 QAM LIGHT	48×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
2048 QAM	-	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
2048 QAM LIGHT	-	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
4096 QAM	-	-	80×DS 1 ⁽¹⁾	80×DS 1 ⁽¹⁾								
4096 QAM LIGHT	-	-	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	-	-		
8192 QAM	_	_	_	_	_	_	_	_	_	_		

(1) XPIC feature not supported by HW.

15.2.2 PNM 1002

PNM 1002 is a board that is combo solution for PFU (Power-Filtering-Unit), NPU (Node-Processor-Unit) and MMU (Modem Unit), see Figure 99.



Figure 99 PNM 1002

The Modem part of PNM 1002 is a Dual carrier modem supporting dual WAN 1+0/2+0/1+1 and Dual WAN $2\times(1+0)$. For modem capability please refer to MMU 1002 performance on a single modem.

PDH Traffic Interfaces

4×E1/DS1

 \geq

User I/O (two users in, one user out)

Ethernet Traffic Interfaces

2×10M/100M/1G (Electrical)

2×1G SFP

2×1G/10GSFP+

Sync Interfaces

1×2 MHz

1×1PPS+ToD

O&M Interface

1×USB interface for an alternative MINI-LINK Node GUI connection is available.

Radio Interfaces

2×60 V RAU connector.

Note: The coaxial interface on modems and radios is equipped with built-in gas discharge tubes for lightning protection.

Adding components to this interface (for example, extra lightning protection) can impact function and performance and should only be considered after consulting with Ericsson.

Power

2×-48V DC connectors for power redundancy.

Figure 100 shows the interfaces on the PNM 1002.



Figure 100 Front Interfaces of PNM 1002

15.2.2.1 User Input/Output

The User Input ports can be used to connect user alarms to MINI-LINK Node GUI. Applications like fire alarms, burglar alarms, and low power indicators are easily

implemented using these input ports. The User Input ports can be configured to be normally opened or closed.

The User Output ports can be used to export summary alarms of the accumulated severity in the NE to the supervision system of other equipment. The User Output ports can be controlled by the operator or triggered by one or several alarm severities.

Plug-In Unit	Number of Input Signals	Number of Output Signals	Connector Type
PNM 1002	Тwo	One	Male 24 pin SOFIX connector

Table 27 PNM 1002 User Input and Output Ports

15.2.2.2 Traffic Capacities for PNM 1002

Table 28	Aggregated PDH and Ethernet Capacity for PNM 1002 Adaptive
	Physical Modes (ETSI)

	Capacity in Mbps											
Madulatia			Cho	annel Bandwi	dth							
n	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz					
4 QAM STRONG	8	18	37/36 ⁽¹⁾	53	75/73 ⁽¹⁾	108	150					
4 QAM	10	21	43/42 ⁽¹⁾	62	87/86 ⁽¹⁾	126	175					
16 QAM STRONG	17	37	74/72 ⁽¹⁾	106	150/146	216	301					
16 QAM	20	43	87/85 ⁽¹⁾	124	175/171	252	351					
32 QAM	26	54	109/109	156	220/219	317	441					
64 QAM	32	68	136/133	195	276/268	397	553					
128 QAM	38	80	161/157	231	326/317	469	653					
256 QAM	44	92	186/182	266	377/366	541	754					
512 QAM	47	99	198/194	284	402/390	577	804					
512 QAM LIGHT	50	105	211/206	302	427/415	614	854					
1024 QAM	53 ⁽²⁾	110	221	316	447/435	643	895					
1024 QAM LIGHT	56 ⁽²⁾	116	233	334	472/459	679	945					
2048 QAM	-	121 ⁽²⁾	243	347	491	706	983					
2048 QAM LIGHT	-	127 ⁽²⁾	256	365	516	742	1033					

		Capacity in Mbps											
Madulatia	Channel Bandwidth												
n	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz						
4096 QAM	-	-	264	377	533	767	1067						
4096 QAM LIGHT	-	-	276	395	558	803	-						
8192 QAM ⁽²⁾⁽³⁾	-	-	283	405	-	-	-						

(1) MIMO supported. Smaller capacity applies for MIMO configuration.

(2) XPIC not supported.

(3) 8192 QAM is not supported as static modulation. It is recommended to set radio output power to 0 dBm or higher.

Table 29 Maximum PDH Capacities for PNM 1002 Adaptive Physical Modes (ETSI)

	Capacity in E1s ⁽¹⁾												
			Cho	annel Bandwi	dth								
Modulation	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz						
4 QAM STRONG	2×E1	6×E1	15×E1/15 ×E1 ⁽²⁾	23×E1	34×E1/33 ×E1 ⁽²⁾	47×E1	70×E1						
4 QAM	3×E1	8×E1	18×E1/18 ×E1	28×E1	40×E1/39 ×E1	56×E1	80×E1						
16 QAM STRONG	6×E1	15×E1	34×E1/33 ×E1 ⁽²⁾	49×E1	70×E1/68 ×E1 ⁽²⁾	80×E1	80×E1						
16 QAM	7×E1	18×E1	39×E1/38 ×E1	57×E1	80×E1/80 ×E1	80×E1	80×E1						
32 QAM	10×E1	24×E1	50×E1/50 ×E1	73×E1	80×E1/80 ×E1	80×E1	80×E1						
64 QAM	13×E1	30×E1	63×E1/62 ×E1 ⁽²⁾	80×E1	80×E1/80 ×E1 ⁽²⁾	80×E1	80×E1						
128 QAM	16×E1	36×E1	75×E1/74 ×E1	80×E1	80×E1/80 ×E1	80×E1	80×E1						
256 QAM	19×E1	42×E1	80×E1/80 ×E1	80×E1	80×E1/80 ×E1	80×E1	80×E1						
512 QAM	21×E1	45×E1	80×E1/80 ×E1	80×E1	80×E1/80 ×E1	80×E1	80×E1						
512 QAM LIGHT	22×E1	48×E1	80×E1/80 ×E1 ⁽²⁾	80×E1	80×E1/80 ×E1 ⁽²⁾	80×E1	80×E1						
1024 QAM	23×E1 ⁽³⁾	51×E1	80×E1	80×E1	80×E1/80 ×E1	80×E1	80×E1						
1024 QAM LIGHT	25×E1 ⁽³⁾	54×E1	80×E1	80×E1	80×E1/80 ×E1	80×E1	80×E1						
2048 QAM	-	56×E1 ⁽³⁾	80×E1	80×E1	80×E1	80×E1	80×E1						
2048 QAM LIGHT	_	59×E1 ⁽³⁾	80×E1	80×E1	80×E1	80×E1	80×E1						
4096 QAM	-	-	80×E1	80×E1	80×E1	80×E1	80×E1						

		Capacity in E1s ⁽¹⁾											
	Channel Bandwidth												
Modulation	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz						
4096 QAM LIGHT	-	-	80×E1	80×E1	80×E1	80×E1	80×E1						
8192 QAM ⁽³⁾	-	-	80×E1 ⁽⁴⁾	80×E1 ⁽⁴⁾	-	-	-						

(1) The table shows the maximum capacity values for one direction. When the NE is configured for two directions, the capacity decreases to half of these values.

(2) MIMO supported. Smaller capacity applies for MIMO configuration.

(3) XPIC not supported.

(4) 8192 QAM is not supported as a minimum modulation.

Table 30Aggregated PDH and Ethernet Capacity for PNM 1002 Adaptive
Physical Modes (ANSI)

	Capacity in Mbps												
				(Channel E	andwidth	I						
Modul ation	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz			
4 QAM STRO NG	12	26	37	39/39 ,0 ⁽¹⁾	53/52 ,4 ⁽¹⁾	67	75	81	108	150			
4 QAM	14	30	43	46/45 ,8 ⁽¹⁾	62/61 ,5 ⁽¹⁾	78	87	94	126	175			
16 QAM STRO NG	25	52	74	79/78 ,0 ⁽¹⁾	107/1 04,7 ⁽¹⁾	135	150	162	216	301			
16 QAM	29	61	87	93/91 ,0 ⁽¹⁾	125/1 22,2 ⁽¹⁾	157	175	189	252	351			
32 QAM	36	77	109	116/1 17,0 ⁽¹⁾	157/1 57,1 ⁽¹⁾	197	220	237	317	441			
64 QAM	46	96	136	146/1 43,0 ⁽¹⁾	197/1 92,0 ⁽¹⁾	247	276	297	397	553			
128 QAM	54	114	161	173/1 69,0 ⁽¹⁾	233/2 26,9 ⁽¹⁾	292	326	351	469	653			
256 QAM	63	131	186	199/1 95,0 ⁽¹⁾	268/2 61,9 ⁽¹⁾	337	377	405	541	754			
512 QAM	67	140	198	213/2 08,0 ⁽¹⁾	286/2 79,3 ⁽¹⁾	360	402	432	577	804			
512 QAM LIGHT	71	149	211	226/2 21,0 ^(I)	304/2 96,8 ^(I)	382	427	459	614	854			
1024 QAM	74 ⁽²⁾	156	221	237	319	400	447	481	643	895			
1024 QAM LIGHT	79 ⁽²⁾	165	233	250	337	423	472	508	679	945			
2048 QAM		171	243	260	350	440	491	528	706	983			

		Capacity in Mbps										
		Channel Bandwidth										
Modul ation	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz		
2048 QAM LIGHT	-	180	256	273	368	462	516	555	742	1033		
4096 QAM	-	-	264	282	380	477	533	574	767 ⁽²⁾	1067 (2)		
4096 QAM LIGHT	-	-	276	296	398	500	558	601	803	-		
8192 QAM (2)(3)	-	-	-	304	409	-	_	_	_	_		

(1) MIMO supported. The second capacity applies for MIMO configuration.

(2) XPIC not supported.

(3) 8192 QAM is not supported as static modulation. It is recommended to set radio output power to 0 dBm or higher.

Table 31 Maximum PDH Capacities for PNM 1002 Adaptive Physical Modes (ANSI)

		Capacity in DS1s ⁽¹⁾											
		Channel Bandwidth											
Modula tion	10 MHz	20 MHz	28MH z	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz			
4 QAM STRON G	5×DS1	14×DS 1	21×DS 1	23×D S1/22 ×DS1	32×D S1/31 ×DS1	40×DS 1	45×DS 1	49×DS 1	66×DS 1	80×DS 1			
4 QAM	7×DS1	17×DS 1	25×DS 1	27×D S1/26 ×DS1 (2)	37×D S1/36 ×DS1 (2)	47×DS 1	54×DS 1	58×DS 1	78×DS 1	80×DS 1			
16 QAM STRON G	13×DS 1	31×DS 1	45×DS 1	48×D S1/47 ×DS1 (2)	66×D S1/64 ×DS1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1			
16 QAM	16×DS 1	36×DS 1	53×DS 1	57×D S1/55 ×DS1 (2)	77×D S1/75 ×DS1 (2)	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1			
32 QAM	21×DS 1	46×DS 1	67×DS 1	72×D S1/72 ×DS1	80×D S1/80 ×DS1 (2)	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1			
64 QAM	27×DS 1	59×DS 1	80×DS 1	80×D S1/80 ×DS1	80×D S1/80 ×DS1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1			
128 QAM	32×DS 1	70×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)	80×D S1/80 ×DS1 (2)	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1			

					Capacity	in DS1s ⁽¹⁾)			
					Channel I	Bandwidtl	ו			
Modula tion	10 MHz	20 MHz	28MH z	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz
256 QAM	37×DS 1	80×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)	80×D S1/80 ×DS1 (2)	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
512 QAM	40×DS 1	80×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)	80×D S1/80 ×DS1 (2)	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
512 QAM LIGHT	43×DS 1	80×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)	80×D S1/80 ×DS1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
1024 QAM	45×DS 1 ⁽³⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
1024 QAM LIGHT	48×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
2048 QAM	-	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
2048 QAM LIGHT	-	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
4096 QAM	-	-	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1 ⁽³⁾
4096 QAM LIGHT	-	-	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	-	-
8192 QAM ⁽²⁾	-	-	-	80×DS 1 ⁽⁴⁾	80×DS 1 ⁽⁴⁾	-	-	-	-	-

(1) The table shows the maximum capacity values for one direction. When the NE is configured for two directions, the capacity decreases to half of these values.

(2) MIMO supported. Smaller capacity applies for MIMO configuration.

(3) XPIC not supported.

(4) 8192 QAM is not supported as minimum modulation.

15.2.3 Enclosure 1401

Enclosure 1401 supports both Ethernet and PDH traffic simultaneously.

The Ethernet capacity is $2 \times 2.5G$ for each APU slot.

15.2.4 Input Voltage Behavior

Figure 101 shows the input voltage behavior for MINI-LINK 6654. The different units turn on and off at different voltages.

To prevent the units from turning on and off repeatedly around a threshold (so called power flapping), there is a hysteresis for all units.



Figure 101 Start and Stop Voltages

15.3 MINI-LINK 6655

MINI-LINK 6655 supports Ethernet (packet link), PDH, and a mix thereof in a Hybrid Radio Link sending Ethernet and PDH traffic simultaneously.

It consists of a rack, Enclosure 1501, that must be equipped with a fan, FAU 1501, and a PNM board, PNM 1002. For the description of PNM 1002, see Section 15.2.2 on page 143.

MINI-LINK 6655 can also be equipped with up to four MMU 1001/1002, four LTU 1001/1002, four ETU 1001/1002, or a combination thereof.

Unused slots in MINI-LINK 6655 must be equipped with Dummy Units, as this is essential to maintain the cooling within the magazines and to meet EMC requirements.

MINI-LINK 6655 can be fitted in a standard 19" or metric rack. The height of a MINI-LINK 6655 is 1.5U, see Figure 102.



Figure 102 Front interfaces on the MINI-LINK 6655

The following slots are paired in MINI-LINK 6655:

- Slot 1 and Slot 2
- Slot 4 and Slot 5

15.3.1 Enclosure 1501

Enclosure 1501 supports both Ethernet and PDH traffic simultaneously.

The MMUs, LTUs and ETUs are placed in the Application Plug-in Unit (APU) slots, that is, slots 1–2 and 4–5. The available traffic capacity is dependent on the board types and their position in the enclosure. Traffic capacities are shown in Table 32.

	Table 32	Enclosure	1501	Ethernet	Capacities
--	----------	-----------	------	----------	------------

Slot	Ethernet Capacity with PNM 1002 [Gbps]
5	1×2.5
4	1×2.5
2	2×2.5
1	2×2.5

15.3.2 Input Voltage Behavior

Figure 103 shows the input voltage behavior for MINI-LINK 6655. The different units turn on and off at different voltages.

To prevent the units from turning on and off repeatedly around a threshold (so called power flapping), there is a hysteresis for all units.



Figure 103 Start and Stop Voltages

15.4 MINI-LINK 6691

MINI-LINK 6691 is suitable for end site and repeater site applications as well as medium-sized hub sites or prioritized small-sites.



Figure 104 MINI-LINK 6691

MINI-LINK 6691 consists of a rack, Enclosure 1101, that must be equipped with a fan, FAU 1101, and power supply, PFU 1101.

MINI-LINK 6691 can also be equipped with one NPU 1002/1003/1005 and two MMU 1001/1002, or with one NPU 1002/1003/1005, one MMU 1001/1002, and one LTU 1001/1002. LTU 1001 provides E1/DS1 interfaces for PDH and LTU 1002 provides E1/DS1 and STM-1 interfaces for PDH and SDH. MINI-LINK 6691 can also be equipped with ETU 1001/1002, which provides additional Ethernet interfaces.

Unused slots in MINI-LINK 6691 must be equipped with Dummy Units, as this is essential to maintain the cooling within the magazines and to meet EMC requirements.

MINI-LINK 6691 can be fitted in a standard 19" or metric rack. The height of a MINI-LINK 6691 is 1U.

15.4.1 PFU 1101

Power handling and filtering is performed by the Power Filter Unit (PFU) 1101, see Figure 105.



Figure 105 PFU 1101

To achieve redundant power, two power sources must be connected.

15.4.1.1 System Monitoring

The PFU supports the monitoring of the node power consumption by measuring the input current and voltage on the incoming -48V.

The following data can be read:

- The current and historical values of input voltage and the power consumption of the node.
- The accumulated and persistently stored energy consumption of the node.

15.4.2 FAU 1101

Air-cooling is provided by the FAU 1101, see Figure 106.



Figure 106 FAU 1101





Figure 107 Cooling Airflow in MINI-LINK 6691

The air enters at the right side of the enclosure and exits at the left side of the enclosure.

15.4.3 Backplane Architecture

MINI-LINK 6691 with NPU 1002/1003/1005 provides high-speed connections of up to 10 Gbps between the NPUs, MMUs, ETUs and LTUs. The available traffic capacity is dependent on the board types and their position in the enclosure.

Slot Ethernet capacity with NPU 1002 [Gbps]

3 1×10 or 4×2.5

1 2×2.5

Slot	Ethernet capacity with and NPU 1003 [Gbps]
3	1×10 or 4×2.5
1	2×2.5
Slot	Ethernet Capacity with NPU 1005 [Gbps]
Slot 3	Ethernet Capacity with NPU 1005 [Gbps] 1x10 or 4x2.5

The following pair of slots are interconnected (2-BPI) in the MINI-LINK 6691 backplane, and can be used for Modem Equipment Protection (EQP):

Slots 1 and 3



Figure 108 Interconnected Slots in MINI-LINK 6691

15.4.4 Hardware Handling

The system offers several functions for easy operation and maintenance.

- Plug-in units can be inserted while the NE is in operation. This enables adding of new Radio Terminals or other plug-in units without disturbing existing traffic.
- Plug-in units other than the NPU can be removed while the NE is in operation.
- When replacing a faulty plug-in unit, the new plug-in unit automatically inherits the configuration of the old plug-in unit.
- The configuration file is stored non-volatile both on the RMM and in a flash memory on the NPU, and can also be backed up and restored using a local or central FTP server. The RMM storage thus enables NPU replacement without using an FTP server.
- The backplane in all subracks has a digital serial number which is also stored on the RMM of the NPU. When inserting an NPU, for example, as a replacement, the serial numbers are compared on power up.
- Various restarts can be ordered from the management system. A cold restart can be initiated for an NE or single plug-in unit, and this type of restart disturbs the traffic. A warm restart is only available for the whole NE, and this

restarts the control system and does not affect the traffic, which is possible due to the separated control and traffic system.

— All plug-in units are equipped with temperature sensors. Overheated boards that exceed limit thresholds, are put in out of service by the control system. This is to avoid hardware failures in case of over-temperature, for example, due to a fan failure or an ambient temperature that is too high. The plug-in unit is automatically returned to normal operation when the temperature is below the high threshold level.

There are two thresholds:

• The high temperature threshold:

The NPU raises a temperature alarm (critical). The NPU is still in full operation.

All plug-in units except NPUs raise a temperature alarm (minor). The plug-in unit is still in full operation.

• The excessive temperature threshold:

The NPU shuts down the entire NPU (out of service).

All plug-in-units except NPUs raise a temperature alarm (critical) and shut down the entire plug-in unit (out of service).

- Access to inventory data like software and hardware product number, serial number, and version. User defined asset identification is supported, enabling tracking of hardware.
- Slot availability and capacity can be monitored by using MINI-LINK Node GUI, or CLI. In this way, choosing a slot for additional plug-in units is easier.
- Ethernet SFPs can be replaced with minimal user intervention, without de-configuring port properties.

15.4.5 Input Voltage Behavior

Figure 109 shows the input voltage behavior for MINI-LINK 6691. The different units turn on and off at different voltages.

To prevent the units from turning on and off repeatedly around a threshold (so called power flapping), there is a hysteresis for all units.


Figure 109 Start and Stop Voltages

15.5 MINI-LINK 6692

MINI-LINK 6692 is suitable for end site and repeater site applications as well as medium-sized hub sites or prioritized small-sites.



Figure 110 MINI-LINK 6692

MINI-LINK 6692 consists of a rack, Enclosure 1201, that must be equipped with a fan, FAU 1201, and power supply, PFU 1201.

MINI-LINK 6692 can also be equipped with one or two NPU 1002/NPU 1005, and up to eight MMU 1001/1002, LTU 1001/1002, or a combination thereof. It is not possible to combine NPU 1002 and NPU 1005 in the same magazine. LTU 1001 provides E1/DS1 interfaces for PDH and LTU 1002 provides E1 and STM-1 interfaces for PDH and SDH. MINI-LINK 6692 can also be equipped with ETU 1001/1002, which provides additional Ethernet interfaces.

To achieve NPU protection, two NPU 1002 or two NPU 1005 must be used. See Section 15.5.5 on page 163.

Unused slots in MINI-LINK 6692 must be equipped with Dummy Units, as this is essential to maintain the cooling within the magazines and to meet EMC requirements.

MINI-LINK 6692 can be fitted in a standard 19" or metric rack. The height of a MINI-LINK 6692 is 3U.

15.5.1 PFU 1201

Power handling and filtering is performed by the Power Filter Unit (PFU) 1201, see Figure 111.



Figure 111 PFU 1201

To achieve redundant power, two PFU 1201 are needed, and two power sources, one connected to each PFU 1201.

15.5.1.1 System Monitoring

The PFU supports the monitoring of the node power consumption by measuring the input current and voltage on the incoming -48V.

The following data can be read:

- The current and historical values of input voltage and the power consumption of the node.
- The accumulated and persistently stored energy consumption of the node.

15.5.2 FAU 1201

Air-cooling is provided by the FAU 1201, see Figure 112.









Figure 113 Cooling Airflow in MINI-LINK 6692

The air enters at the right side of the enclosure and exits at the left side of the enclosure.

MINI-LINK 6692 provides high-speed connections of up to 10 Gbps between the NPUs, MMUs, ETUs and LTUs. The available traffic capacity is dependent on the board types and their position in the enclosure.

The MMUs, LTUs, and ETUs are placed in the Application Plug-in Unit (APU) slots, that is, slots 1-2, 5-8, and 11-12.

Slot	Ethernet capacity with NPU 1002 [Gbps]	Slot	Ethernet capacity with NPU 1002 [Gbps]
6	1×2.5 + 1×1	12	1×2.5 + 1×1
5	2×2.5	11	2×2.5
2	1×10G (4 lanes) or 4×2.5	8	1×10G (4 lanes) or 4×2.5
1	2×2.5	7	1×2.5
Slot	Ethernet capacity with	Slot	Ethernet capacity with
Slot	Ethernet capacity with NPU 1005 [Gbps]	Slot	Ethernet capacity with NPU 1005 [Gbps]
Slot 6	Ethernet capacity with NPU 1005 [Gbps] 1×2.5	Slot 12	Ethernet capacity with NPU 1005 [Gbps] 1×2.5
Slot 6 5	Ethernet capacity with NPU 1005 [Gbps] 1×2.5 1×10G (4 lanes) or 4×2.5	Slot 12 11	Ethernet capacity with NPU 1005 [Gbps] 1×2.5 1×2.5
Slot 6 5 2	Ethernet capacity with NPU 1005 [Gbps] 1×2.5 1×10G (4 lanes) or 4×2.5 1×10G (4 lanes) or 4×2.5	Slot 12 11 8	Ethernet capacity with NPU 1005 [Gbps] 1×2.5 1×2.5 1×2.5

The following pairs of slots are interconnected (2-BPI) in the MINI-LINK 6692 backplane, and can be used for Modem Equipment Protection (EQP):

- Slots 1 and 2
- Slots 5 and 6
- Slots 7 and 8
- Slots 11 and 12



Figure 114 Interconnected Slots in MINI-LINK 6692

15.5.4 Hardware Handling

The system offers several functions for easy operation and maintenance.

- Plug-in units can be inserted while the NE is in operation. This enables adding of new Radio Terminals or other plug-in units without disturbing existing traffic.
- Plug-in units other than the NPU can be removed while the NE is in operation.
- When replacing a faulty plug-in unit, the new plug-in unit automatically inherits the configuration of the old plug-in unit.
- The configuration file is stored non-volatile both on the RMM and in a flash memory on the NPU, and can also be backed up and restored using a local or central FTP server. The RMM storage thus enables NPU replacement without using an FTP server.
- The backplane in all subracks has a digital serial number which is also stored on the RMM of the NPU. When inserting an NPU, for example, as a replacement, the serial numbers are compared on power up.
- Various restarts can be ordered from the management system. A cold restart can be initiated for an NE or single plug-in unit, and this type of restart disturbs the traffic. A warm restart is only available for the whole NE, and this restarts the control system and does not affect the traffic, which is possible due to the separated control and traffic system.
- All plug-in units are equipped with temperature sensors. Overheated boards that exceed limit thresholds are put in out of service by the control system. This is to avoid hardware failures in case of over-temperature, for example, due to a fan failure or an ambient temperature that is too high. The plug-in unit is automatically returned to normal operation when the temperature is below the high threshold level.

There are two thresholds:

• The high temperature threshold:

The NPU raises a temperature alarm (critical). The NPU is still in full operation.

All plug-in units except NPUs raise a temperature alarm (minor). The plug-in unit is still in full operation.

• The excessive temperature threshold:

The NPU shuts down the entire NPU (out of service).

All plug-in-units except NPUs raise a temperature alarm (critical) and shut down the entire plug-in unit (out of service).

- Access to inventory data like software and hardware product number, serial number, and version. User defined asset identification is supported, enabling tracking of hardware.
- Slot availability and capacity can be monitored by using MINI-LINK Node GUI, or CLI. In this way, choosing the correct slot when adding additional plug-in units is made easier.
- Ethernet SFPs can be replaced with minimal user intervention, without de-configuring port properties.

15.5.5 NPU Protection

When two NPU 1002/NPU 1005 are inserted into MINI-LINK 6692, NPU Protection can provide board, port, and full functional redundancy for the NPU, if configured and licensed properly. It is not possible to combine NPU 1002 and NPU 1005 in the same magazine.

In case of a hardware or software fault in the master NPU, an automatic failover takes place, and the slave NPU becomes the master by inheriting all configuration. All user traffic and DCN access recovers after the failover.

When having two NPU 1002, the front LAN ports of the slave NPU can be extended and configured to be used for user traffic. This way the extended ports could have different configuration than the corresponding port of the master NPU and used in different setup. After fail or switchover the configuration remains on each port, but the ports of the new slave NPU will be handled as extended.

Note: The front LAN ports on the slave NPU cannot be extended on NPU 1005.

15.5.6 Input Voltage Behavior

Figure 115 shows the input voltage behavior for MINI-LINK 6692. The different units turn on and off at different voltages.

To prevent the units from turning on and off repeatedly around a threshold (so called power flapping), there is a hysteresis for all units.



Figure 115 Start and Stop Voltages

15.6 MINI-LINK 6693

MINI-LINK 6693 is suitable for end site and repeater site applications as well as medium-sized hub sites or prioritized small-sites.



Figure 116 MINI-LINK 6693

MINI-LINK 6693 consists of a rack, Enclosure 1301, that must be equipped with a fan, FAU 1301, and power supply, PFU 1301.

MINI-LINK 6693 can also be equipped with one NPU 1002/1003/1005, and up to four MMU 1001/1002, LTU 1001/1002, or a combination thereof. LTU 1001 provides E1/DS1 interfaces for PDH and LTU 1002 provides E1/DS1 and STM-1 interfaces for PDH and SDH. MINI-LINK 6693 can also be equipped with ETU 1001/1002, which provides additional Ethernet interfaces.

Unused slots in MINI-LINK 6693 must be equipped with Dummy Units, as this is essential to maintain the cooling within the magazines and to meet EMC requirements.

MINI-LINK 6693 can be fitted in a standard 19" or metric rack. The height of a MINI-LINK 6693 is 1.5U.

15.6.1 PFU 1301

Power handling and filtering is performed by the Power Filter Unit (PFU) 1301, see Figure 117.



Figure 117 PFU 1301

To achieve redundant power, two power sources must be connected.

15.6.1.1 System Monitoring

The PFU supports the monitoring of the node power consumption by measuring the input current and voltage on the incoming -48V.

The following data can be read:

- The current and historical values of input voltage and the power consumption of the node.
- The accumulated and persistently stored energy consumption of the node.

15.6.2 FAU 1301

Air-cooling is provided by the FAU 1301, see Figure 118.



Figure 118 FAU 1301

FAU 1301 has an automatic fan speed control and houses two internal fans.



Figure 119 Cooling Airflow in MINI-LINK 6693

The air enters at the right side of the enclosure and exits at the left side of the enclosure.

MINI-LINK 6693 provides high-speed connections of up to 10 Gbps between the NPUs, MMUs, ETUs and LTUs. The available traffic capacity is dependent on the board types and their position in the enclosure.

The MMUs, LTUs, and ETUs are placed in the Application Plug-in Unit (APU) slots, that is, slots 1-2 and 4-5.

Slot	Ethernet capacity with NPU 1002 [Gbps]
5	1×2.5 + 1×1
4	2×2.5
2	1×10 or 4×2.5
1	2×2.5
Slot	Ethernet capacity with NPU 1003 [Gbps]
5	2×2.5
4	1×2.5
2	1×10 or 4×2.5
1	2×2.5
Slot	Ethernet capacity with NPU 1005 [Gbps]
5	1×2.5
4	1×10 or 4×2.5
2	1×10 or 4×2.5
1	1×10 or 4×2.5

The following pairs of slots are interconnected (2-BPI) in the MINI-LINK 6693 backplane, and can be used for Modem Equipment Protection (EQP):

- Slots 1 and 2
- Slots 4 and 5



Figure 120 Interconnected Slots in MINI-LINK 6693

15.6.4 Hardware Handling

The system offers several functions for easy operation and maintenance.

- Plug-in units can be inserted while the NE is in operation. This enables adding of new Radio Terminals or other plug-in units without disturbing existing traffic.
- Plug-in units other than the NPU can be removed while the NE is in operation.
- When replacing a faulty plug-in unit, the new plug-in unit automatically inherits the configuration of the old plug-in unit.
- The configuration file is stored non-volatile both on the RMM and in a flash memory on the NPU, and can also be backed up and restored using a local or central FTP server. The RMM storage thus enables NPU replacement without using an FTP server.
- The backplane in all subracks has a digital serial number which is also stored on the RMM of the NPU. When inserting an NPU, for example, as a replacement, the serial numbers are compared on power up.
- Various restarts can be ordered from the management system. A cold restart can be initiated for an NE or single plug-in unit, and this type of restart disturbs the traffic. A warm restart is only available for the whole NE, and this restarts the control system and does not affect the traffic, which is possible due to the separated control and traffic system.
- All plug-in units are equipped with temperature sensors. Overheated boards that exceed limit thresholds are put in out of service by the control system. This is to avoid hardware failures in case of over-temperature, for example, due to a fan failure or an ambient temperature that is too high. The plug-in unit is automatically returned to normal operation when the temperature is below the high threshold level.

There are two thresholds:

• The high temperature threshold:

The NPU raises a temperature alarm (critical). The NPU is still in full operation.

All plug-in units except NPUs raise a temperature alarm (minor). The plug-in unit is still in full operation.

• The excessive temperature threshold:

The NPU shuts down the entire NPU (out of service).

All plug-in-units except NPUs raise a temperature alarm (critical) and shut down the entire plug-in unit (out of service).

- Access to inventory data like software and hardware product number, serial number, and version. User defined asset identification is supported, enabling tracking of hardware.
- Slot availability and capacity can be monitored by using MINI-LINK Node GUI, or CLI. In this way, choosing the correct slot when adding additional plug-in units is made easier.
- Ethernet SFPs can be replaced with minimal user intervention, without de-configuring port properties.

15.6.5 Input Voltage Behavior

Figure 121 shows the input voltage behavior for MINI-LINK 6693. The different units turn on and off at different voltages.

To prevent the units from turning on and off repeatedly around a threshold (so called power flapping), there is a hysteresis for all units.



Figure 121 Start and Stop Voltages

15.7 MINI-LINK 6694

MINI-LINK 6694 is suitable for end site and repeater site applications as well as medium-sized hub sites or prioritized small-sites.



Figure 122 MINI-LINK 6694

MINI-LINK 6694 consists of a rack, Enclosure 1601, that must be equipped with a fan, FAU 1601, and power supply, PFU 1601.

MINI-LINK 6694 can also be equipped with one or two NPU 1002/NPU 1005 or one NPU1003 and up to four MMU 1001/1002, LTU 1001/1002, or a combination thereof. It is not possible to combine NPU 1002 and NPU 1005 in the same magazine. LTU 1001 provides E1/DS1 interfaces for PDH, and LTU 1002 provides E1/DS1 and STM-1 interfaces for PDH and SDH. MINI-LINK 6694 can also be equipped with ETU 1001/1002, which provides additional Ethernet interfaces.

To achieve NPU protection, two NPU 1002 or two NPU 1005 must be used. See Section 15.7.4 on page 173.

Unused slots in MINI-LINK 6694 must be equipped with Dummy Units, as this is essential to maintain the cooling within the magazines and to meet EMC requirements.

MINI-LINK 6694 can be fitted in a standard 19" or metric rack. The height of MINI-LINK 6694 is 2U.

15.7.1 PFU 1601

Power handling and filtering is performed by the Power Filter Unit (PFU) 1601, see Figure 123.



Figure 123 PFU 1601

To achieve redundant power, two PFU 1601 are needed, and two power sources, one connected to each PFU 1601.

15.7.1.1 System Monitoring

The PFU supports the monitoring of the node power consumption by measuring the input current and voltage on the incoming -48V.

The following data can be read:

- The current and historical values of input voltage and the power consumption of the node.
- The accumulated and persistently stored energy consumption of the node.

15.7.2 FAU 1601

Air-cooling is provided by FAU 1601, see Figure 124.





FAU 1601 has an automatic fan speed control and houses two internal fans.



Figure 125 Cooling Airflow in MINI-LINK 6694

The air enters at the right side of the enclosure and exits at the left side of the enclosure.

MINI-LINK 6694 provides high-speed connections of up to 10 Gbps between the NPUs, MMUs, ETUs and LTUs. The available traffic capacity depends on the board types and their positions in the enclosure.

The MMUs, ETUs, and LTUs are placed in the Application Plug-in Unit (APU) slots, that is, slots 1-2 and 5-6.

Slot	Ethernet Capacity with NPU 1002 [Gbps]
6	1×2.5 + 1×1
5	2×2.5
2	1×10 or 4×2.5
1	2×2.5
Clat	
Slot	Ethernet Capacity with NPU 1003 [Gbps]
6	2×2.5
5	1×2.5
2	1×10 or 4×2.5
1	2×2.5
Slot	Ethernet Capacity with NPU 1005 [Gbps]
6	1×2.5
5	1×10 or 4×2.5
2	1×10 or 4×2.5
1	1×10 or 4×2.5

The following pairs of slots are interconnected (2-BPI) in the MINI-LINK 6694 backplane and can be used for Modem Equipment Protection (EQP):

- Slots 1 and 2
- Slots 5 and 6



Figure 126 Interconnected Slots in MINI-LINK 6694

15.7.4 NPU Protection

When two NPU 1002/NPU 1005 are inserted into MINI-LINK 6694, NPU Protection can provide board, port, and full functional redundancy for the NPU, if configured and licensed properly. It is not possible to combine NPU 1002 and NPU 1005 in the same magazine.

In case of a hardware or software fault in the Master NPU, an automatic failover takes place, and the Slave NPU becomes the Master by inheriting all configurations. All user traffic and DCN access recovers after the failover.

When having two NPU 1002, the front LAN ports of the Slave NPU can be extended and configured to be used for user traffic. This way, the extended ports can have different configurations than the corresponding port of the Master NPU and can be used in different setups. After fail- or switchover, the configuration remains on each port, but the ports of the new Slave NPU are handled as extended.

Note: The front LAN ports on the slave NPU cannot be extended on NPU 1005.

15.7.5 Hardware Handling

The system offers several functions for easy operation and maintenance.

- Plug-in units can be inserted while the NE is in operation. This allows for adding new Radio Terminals or other plug-in units without disturbing the existing traffic.
- Plug-in units other than the NPU can be removed while the NE is in operation.
- When replacing a faulty plug-in unit, the new plug-in unit automatically inherits the configuration of the old plug-in unit.
- The configuration file is stored non-volatile both on the RMM and in a flash memory on the NPU and can also be backed up and restored using a local or central FTP server. The RMM storage thus enables NPU replacement without using an FTP server.
- The backplane in all subracks has a digital serial number which is also stored on the RMM of the NPU. When inserting an NPU, for example, as a replacement, the serial numbers are compared on power-up.
- Various restarts can be ordered from the management system. A cold restart can be initiated for an NE or a single plug-in unit, and it disturbs the traffic. A warm restart is only available for the whole NE, and it restarts the control system and does not affect the traffic due to the separated control and traffic systems.
- All plug-in units are equipped with temperature sensors. Overheated boards that exceed threshold limits are put in out-of-service by the control system. This is to avoid hardware failures in case of over-heating, for example, due to a fan failure or an ambient temperature that is too high. The plug-in unit is automatically returned to normal operation when the temperature is below the high threshold limit.

There are two thresholds:

• The high temperature threshold:

The NPU raises a temperature alarm (critical). The NPU is still in full operation.

All plug-in units except NPUs raise a temperature alarm (minor). The plug-in unit is still in full operation.

• The excessive temperature threshold:

The NPU shuts down the entire NPU (out of service).

All plug-in-units except NPUs raise a temperature alarm (critical) and shut down the entire plug-in unit (out of service).

- Access to inventory data like software and hardware product numbers, serial number, and version. User-defined asset identification is supported, enabling hardware tracking.
- Slot availability and capacity can be monitored by using MINI-LINK Node GUI or the CLI. In this way, choosing the correct slot when adding additional plug-in units is made easier.
- Ethernet SFPs can be replaced with minimal user intervention, without de-configuring port properties.

15.7.6 Input Voltage Behavior

Figure 127 shows the input voltage behavior for MINI-LINK 6694. The different units turn on and off at different voltages.

To prevent the units from turning on and off repeatedly around a threshold (so called power flapping), there is a hysteresis for all units.



Figure 127 Start and Stop Voltages

15.8 NPU 1002

NPU 1002 is a full size processor board that fits in MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, and MINI-LINK 6694.

NPU 1002 contains an embedded Ethernet switch which makes it an ideal component at a medium or small Ethernet aggregation site. It also supports Synchronous Ethernet.



Figure 128 NPU 1002

The NPU 1002 supports the following Ethernet-related features:

- Store and forward MAC (IEEE802.1D-2004) and VLAN-based switching (IEEE802.1Q-2005 and IEEE802.1ad-2005).
- Port-based VLAN (up to 1,000 VLANs in the range of 1-4094).
- MAC address table (32,000 entries) with learning and aging.
- IPv4 and IPv6
- Static unicast/multicast routes.
- Jumbo frames for LAN and Packet Link WAN ports.
- Multiple internal Layer 1 and Layer 2 Ethernet connections.
- MSTP/RSTP/LAG protection mechanism.
- Link Aggregation Control Protocol (LACP).
- Link Loss Forwarding for internal Layer 1 connections.
- Policing/color marking per ingress port with CIR/EIR and CBS/EBS.
- Policing/color marking per ingress port per QoS priority with CIR/EIR and CBS/EBS.
- Policing/color marking per ingress port per VLAN with CIR/EIR and CBS/EBS.
- Traffic shaping per egress port with CIR and CBS.
- 2 MHz sync input/output.
- Ethernet loop detection.
- Broadcast/multicast/destination lookup failure storm protection.
- Frame admission control.
- Classification/tagging of priority and VLAN ID.

- Congestion handling with eight priority queues (TC) per port.
 - Frame discard: WRED, color dropping, aging and tail drop.
 - TC scheduling: SP, WDRR.
- In-band management transport in Ethernet flow (DCN over VLAN).
- Management/diagnostics.
 - PM counters.
 - Link OAM.
 - Ethernet Service OAM.
 - Port mirroring.

15.8.1 NPU 1002 Front Interfaces

- **O&M** Mini USB port for Local Management.
- SYNC: 1PPS + ToD HW prepared.
- SYNC: 2 MHz 2 MHz/2048 Kbps sync input/output.
- TN 4 & TN 5 2 x RJ 45 10Mbps/100Mbps/1Gbps Ethernet ports. All LAN ports are SyncE capable.
- TN $6 1 \times SFP$ 1Gbps Ethernet port. SyncE capable except with SFPe.
- TN 7 & TN 8 2 x SFP/SFP+ 1Gbps or 10Gbps Ethernet ports. All LAN ports are SyncE capable except with SFPe.
- USER I/O 6 user in, 3 user out.



Figure 129 Front Interfaces on the NPU 1002

NPU 1002 has two fixed electrical LAN 10/100/1000BASE-T Ethernet interfaces, one SFP cage supporting 1 Gbps, and two SFP+ cages supporting up to 10 Gbps on the front. One of the fixed Ethernet ports can be configured as a site LAN port.

The SFP/SFP+ cages can be equipped with Ericsson approved SFP/SFP+ modules, see MINI-LINK 6600 R1 Compatibility, Reference [1].

The User I/O interface provides six User Input ports and three User Output ports.

For the basic offering for NPU 1002, two Ethernet ports are enabled at the front. To get access to more ports, licenses are needed. Licenses are also needed for the 10G ports.

NPU 1002 supports Packet Link WAN connections.

The high speed connections and the LAN ports enable NPU 1002 to aggregate traffic from multiple directions as well as dropping Ethernet traffic locally at the site.

15.8.1.1 User Input/Output

The User Input ports can be used to connect user alarms to MINI-LINK Node GUI. Applications like fire alarms, burglar alarms, and low power indicators are easily implemented using these input ports. The User Input ports can be configured to be normally opened or closed.

The User Output ports can be used to export summary alarms of the accumulated severity in the NE to the supervision system of other equipment. The User Output ports can be controlled by the operator or triggered by one or several alarm severities.

Plug-In Unit	Number of Input Signals	Number of Output Signals	Connector Type
NPU 1002	Six	Three	Male 24 pin SOFIX connector

Table 33 NPU 1002 User Input and Output Ports

15.8.2 Ethernet Switch

The Ethernet switch functionality is shown in Table 34.

Table 34 🛛 I	Ethernet	Switch	Functionality
--------------	----------	--------	---------------

Modes	Port roles	Comment
Provider Bridge mode	CE-UNI	CEP, including default mapping rule
	CN-UNI	CNP
	I-NNI	_

Modes	Port roles	Comment
Customer Bridge mode	UNI	Including QinQ termination, default mapping rule
	NNI	_

The switch is a managed VLAN switch (IEEE 802.1Q and IEEE 802.1D) and supports Provider mode (IEEE 802.1ad) and Customer mode (IEEE 802.1Q) switching.

The switch supports Jumbo frames. The Ethernet site LAN ports on NPU 1002 have interfaces that support auto-negotiation 10/100/1000 Mbps speed. The interfaces are physical RJ-45 connectors.

15.8.3 Ethernet LAN Ports

The Ethernet LAN ports are described in Table 35.

Functionality	Description
Interface types	10/100/1000 Mbps
	10 Gbps
	Auto negotiation or manual setting (IEEE802.3-2005).
Duplex mode	Full duplex.
	Auto negotiation or manual setting (IEEE802.3-2005).
Connectors/cables	RJ-45 with MDI/MDIX support for all electrical i/f.
	Single/multi-mode fiber with LC connector for the optical i/f.
SFP modules	SFP/SFP+ plug-in modules are supported with 1 Gbps and 10 Gbps optical and 100/1000 Mbps electrical interfaces.
	For more information regarding SFP/SFP+, see MINI-LINK 6600 R1 Compatibility, Reference [1].
LED indicators	Each Ethernet interface has one green LED for network link indication.
Standard frame sizes	Up to 2000 bytes when used as an external interface (IEEE802.3as-2006).
	Up to 2048 bytes when used as an internal interface.
Jumbo frames	9,216 byte frame size for data applications.

Table 35 Ethernet LAN Ports

Table 35 Ethernet LAN Ports

Functionality	Description
Buffer capacity	Ethernet ports share a 720 MB buffering capacity. The buffer capacity sharing among the traffic queues is based on a fair adaptive dynamic algorithm.
Flow control	NPU 1002 supports the IEEE 802.3x standard to handle temporary congestion on ports configured as Layer 2 connections (connected to the switch). NPU 1002 only supports asymmetrical flow control. When a congestion threshold is reached in an NE downstream, the NE generates a PAUSE signal and NPU 1002 temporarily halts the transmission of Ethernet frames, but NPU 1002 cannot generate PAUSE signals to control the transmission process on connected equipment.

15.9 NPU 1003

NPU 1003 is a processor board that fits in MINI-LINK 6691 MINI-LINK 6693, and MINI-LINK 6694.

NPU 1003 contains an embedded Ethernet switch which makes it an ideal component at a medium or small Ethernet aggregation site. It also supports Synchronous Ethernet.



Figure 130 NPU 1003

The NPU 1003 supports the following Ethernet-related features:

- Store and forward MAC (IEEE802.1D-2004) and VLAN-based switching (IEEE802.1Q-2005 and IEEE802.1ad-2005).
- Port-based VLAN (up to 1,000 VLANs in the range of 1–4094).
- MAC address table (32,000 entries) with learning and aging.
- IPv4 and IPv6
- Static unicast/multicast routes.

- Jumbo frames for LAN and Packet Link WAN ports.
- Multiple internal Layer 1 and Layer 2 Ethernet connections.
- MSTP/RSTP/LAG protection mechanism.
- Link Aggregation Control Protocol (LACP).
- Link Loss Forwarding for internal Layer 1 connections.
- Policing/color marking per ingress port with CIR/EIR and CBS/EBS.
- Policing/color marking per ingress port per QoS priority with CIR/EIR and CBS/EBS.
- Policing/color marking per ingress port per VLAN with CIR/EIR and CBS/EBS.
- Traffic shaping per egress port with CIR and CBS.
- 2 MHz/2048 Kbps sync input/output.
- Ethernet loop detection.
- Broadcast/multicast/destination lookup failure storm protection.
- Frame admission control.
- Classification/tagging of priority and VLAN ID.
- Congestion handling with eight priority queues (TC) per port.
 - Frame discard: WRED, color dropping, aging and tail drop.
 - TC scheduling: SP, WDRR.
- In-band management transport in Ethernet flow (DCN over VLAN).
- Management/diagnostics.
 - PM counters.
 - Link OAM.
 - Ethernet Service OAM.
 - Port mirroring.

15.9.1 NPU 1003 Front Interfaces

- **O&M** Mini USB port for Local Management.
- SYNC: 1PPS + ToD HW prepared.
- SYNC: 2 MHz 2 MHz/2048 Kbps sync input/output.

- TN 4 & TN 5 2 x RJ 45 10Mbps/100Mbps/1Gbps Ethernet ports. All LAN ports are SyncE capable.
- TN $6 1 \times SFP$ 1Gbps Ethernet port. SyncE capable except with SFPe.
- TN 7 & TN 8 2 x SFP/SFP+ 1Gbps or 10Gbps Ethernet ports. All LAN ports are SyncE capable except with SFPe.
- TN 9 4 E1/DS1 and User I/O (2 user in, 1 user out). The E1 interface can be used as sync source.



Figure 131 Front Interfaces on the NPU 1003

NPU 1003 has two fixed electrical LAN 10/100/1000BASE-T Ethernet interfaces, one SFP cage supporting 1Gbps, and two SFP+ cages supporting up to 10 Gbps on the front. One of the fixed Ethernet ports can be configured as a site LAN port. The NPU 1003 also has shared E1/DS1 and User I/O port.

The SFP/SFP+ cages can be equipped with Ericsson approved SFP/SFP+ modules, see MINI-LINK 6600 R1 Compatibility, Reference [1].

The User I/O interface provides two User Input ports and one User Output port.

For the basic offering for NPU 1003, two Ethernet ports are enabled at the front. To get access to more ports, licenses are needed. Licenses are also needed for the 10G ports.

NPU 1003 supports Packet Link WAN connections.

The high speed connections and the LAN ports enable NPU 1003 to aggregate traffic from multiple directions as well as dropping Ethernet traffic locally at the site.

15.9.1.1 User Input/Output

The User Input ports can be used to connect user alarms to MINI-LINK Node GUI. Applications like fire alarms, burglar alarms, and low power indicators are easily implemented using these input ports. The User Input ports can be configured to be normally opened or closed.

The User Output ports can be used to export summary alarms of the accumulated severity in the NE to the supervision system of other equipment. The User Output

ports can be controlled by the operator or triggered by one or several alarm severities.

Table 36 NPU 1003 User Input and Output Ports

Plug-In Unit	Number of Input Signals	Number of Output Signals	Connector Type
NPU 1003	Тwo	One	Male 24 pin SOFIX connector

15.9.2 Ethernet Switch

The Ethernet switch functionality is shown in Table 34.

Table 57 Ethernet Switch Functionality	Table 37	Ethernet Switch Functionality
--	----------	-------------------------------

Modes	Port roles	Comment
Provider Bridge mode	CE-UNI	CEP, including default mapping rule
	CN-UNI	CNP
	I-NNI	_
Customer Bridge mode	UNI	Including QinQ termination, default mapping rule
	NNI	_

The switch is a managed VLAN switch (IEEE 802.1Q and IEEE 802.1D) and supports Provider mode (IEEE 802.1ad) and Customer mode (IEEE 802.1Q) switching.

The switch supports Jumbo frames. The Ethernet site LAN ports on NPU 1003 have interfaces that support auto-negotiation 10/100/1000 Mbps speed. The interfaces are physical RJ-45 connectors.

15.9.3 Ethernet LAN Ports

The Ethernet LAN ports are described in Table 35.

Table 38 Ethernet LAN Ports

Functionality	Description
Interface types	10/100/1000 Mbps
	10 Gbps
	Auto negotiation or manual setting (IEEE802.3-2005).

Functionality	Description
Duplex mode	Full duplex.
	Auto negotiation or manual setting (IEEE802.3-2005).
Connectors/cables	RJ-45 with MDI/MDIX support for all electrical i/f.
	Single/multi-mode fiber with LC connector for the optical i/f.
SFP modules	SFP/SFP+ plug-in modules are supported with 1 Gbps and 10 Gbps optical and 100/1000 Mbps electrical interfaces.
	For more information regarding SFP/SFP+, see MINI-LINK 6600 R1 Compatibility, Reference [1].
LED indicators	Each Ethernet interface has one green LED for network link indication.
Standard frame sizes	Up to 2000 bytes when used as an external interface (IEEE802.3as-2006).
	Up to 2048 bytes when used as an internal interface.
Jumbo frames	9,216 byte frame size for data applications.
Buffer capacity	Ethernet ports share a 180 MB buffering capacity. The buffer capacity sharing among the traffic queues is based on a fair adaptive dynamic algorithm.
Flow control	NPU 1003 supports the IEEE 802.3x standard to handle temporary congestion on ports configured as Layer 2 connections (connected to the switch). NPU 1003 only supports asymmetrical flow control. When a congestion threshold is reached in an NE downstream, the NE generates a PAUSE signal and NPU 1003 temporarily halts the transmission of Ethernet frames, but NPU 1003 cannot generate PAUSE signals to control the transmission process on connected equipment.

Table 38 Ethernet LAN Ports

15.10 NPU 1005

NPU 1005 is a full size processor board that fits in MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, and MINI-LINK 6694.

NPU 1005 contains an embedded Ethernet switch which makes it an ideal component at a medium or small Ethernet aggregation site.





The NPU 1005 supports the following Ethernet-related features:

- Synchronous Ethernet
- Hierarchical Radio Link Bonding (hRLB)
- Store and forward MAC (IEEE802.1D-2004) and VLAN-based switching (IEEE802.1Q-2005 and IEEE802.1ad-2005).
- Port-based VLAN (up to 1,000 VLANs in the range of 1-4094).
- MAC address table (32,000 entries) with learning and aging.
- IPv4 and IPv6
- Static unicast/multicast routes.
- Jumbo frames for LAN and Packet Link WAN ports.
- Multiple internal Layer 1 and Layer 2 Ethernet connections.
- MSTP/RSTP/LAG protection mechanism.
- Link Aggregation Control Protocol (LACP).
- Link Loss Forwarding for internal Layer 1 connections.
- Policing/color marking per ingress port with CIR/EIR and CBS/EBS.
- Policing/color marking per ingress QoS TC with CIR/EIR and CBS/EBS.
- Traffic shaping per egress port with CIR and CBS.
- 2 MHz sync input/output.
- Ethernet loop detection.
- Broadcast/multicast/destination lookup failure storm protection.
- Frame admission control.
- Classification/tagging of priority and VLAN ID.

- Congestion handling with eight priority queues (TC) per port.
 - Frame discard: WRED, color dropping, aging and tail drop.
 - TC scheduling: SP, WDRR.
- In-band management transport in Ethernet flow (DCN over VLAN).
- Management/diagnostics.
 - PM counters.
 - Link OAM.
 - Ethernet Service OAM.
 - Port mirroring.

15.10.1 Front Interfaces

O&M	Mini USB port for Local management.
SYNC: 1PPS + ToD	HW prepared.
SYNC: 2 MHz	2 MHz/2048 Kbps sync input/output.
TN 4 & TN 5	2 x RJ 45 10Mbps/100Mbps/1Gbps Ethernet ports. All LAN ports are SyncE capable.
TN 6	1 x SFP 1Gbps Ethernet port. SyncE capable, except when it is used with SFPe.
TN 7 & TN 8	2 x SFP/SFP+ 1Gbps or 10Gbps Ethernet ports. All LAN ports are SyncE capable, except when they are used with SFPe.
USER I/O	6 user in, 3 user out.

91 100 [10-10]00000-00-00-00

Figure 133 Front Interfaces on the NPU 1005

NPU 1005 has two fixed electrical LAN 10/100/1000BASE-T Ethernet interfaces, one SFP cage, and two SFP+ cages on the front. One of the fixed Ethernet ports can be configured as a site LAN port.

The SFP/SFP+ cages can be equipped with Ericsson approved SFP/SFP+ modules, see MINI-LINK 6600 R1 Compatibility, Reference [1].

The User I/O interface provides six User Input ports and three User Output ports.

For the basic offering for NPU 1005, two Ethernet ports are enabled at the front. To get access to more ports, licenses are needed.

NPU 1005 supports Packet Link WAN connections.

The high speed connections and the LAN ports enable NPU 1005 to aggregate traffic from multiple directions as well as dropping Ethernet traffic locally at the site.

15.10.2 Ethernet Switch

The Ethernet switch functionality is shown in Table 34.

Modes	Port roles	Comment	
Provider Bridge mode	CE-UNI	CEP, including default mapping rule	
	CN-UNI	CNP	
	I-NNI	_	
Customer Bridge mode	UNI	Including QinQ termination, default mapping rule	
	NNI	-	

Table 39 Ethernet Switch Functionality

The switch is a managed VLAN switch (IEEE 802.1Q and IEEE 802.1D) and supports Provider mode (IEEE 802.1ad) and Customer mode (IEEE 802.1Q) switching.

The switch supports Jumbo frames. The Ethernet site LAN ports on NPU 1005 have interfaces that support auto-negotiation 10/100/1000 Mbps speed. The interfaces are physical RJ-45 connectors.

15.10.3 Ethernet LAN Ports

The Ethernet LAN ports are described in Table 35.

Functionality	Description
Interface types	10/100/1000 Mbps
	10 Gbps
	Auto negotiation or manual setting (IEEE802.3-2005).

Table 40Ethernet LAN Ports

 \leq

Functionality	Description
Duplex mode	Full duplex.
	Auto negotiation or manual setting (IEEE802.3-2005).
Connectors/cables	RJ-45 with MDI/MDIX support for all electrical i/f.
	Single/multi-mode fiber with LC connector for the optical i/f.
SFP modules	SFP/SFP+ plug-in modules are supported with 1 Gbps and 10 Gbps optical and 100/1000 Mbps electrical interfaces.
	For more information regarding SFP/SFP+, see MINI-LINK 6600 R1 Compatibility, Reference [1].
LED indicators	Each Ethernet interface has one green LED for network link indication.
Standard frame sizes	Up to 2000 bytes when used as an external interface (IEEE802.3as-2006).
	Up to 2048 bytes when used as an internal interface.
Jumbo frames	9,216 byte frame size for data applications.
Buffer capacity	Ethernet ports share a 720 MB buffering capacity. The buffer capacity sharing among the traffic queues is based on a fair adaptive dynamic algorithm.
Flow control	NPU 1005 supports the IEEE 802.3x standard to handle temporary congestion on ports configured as Layer 2 connections (connected to the switch). NPU 1005 only supports asymmetrical flow control. When a congestion threshold is reached in an NE downstream, the NE generates a PAUSE signal and NPU 1005 temporarily halts the transmission of Ethernet frames, but NPU 1005 cannot generate PAUSE signals to control the transmission process on connected equipment.

	Table 40	Ethernet LAN Ports
--	----------	--------------------

15.11 MMU 1001

MMU 1001 is a single carrier modem.

On one modem, the following is supported:

— Single WAN 1+0

On two modems, the following is supported:

- 2+0 RLB with Equipment Protection (EQP)
- 1+1 RLP with EQP





Note: The coaxial interface on modem and radios is equipped with built-in gas discharge tubes for lightning protection.

Adding components to this interface (for example, extra lightning protection) can impact function and performance and should only be considered after consulting with Ericsson.

Table 46 shows the characteristics of the WAN ports.

Ethernet WAN ports	Description
Interface types	1×Packet Link radio interface.
Frame size	An Ethernet over Packet Link WAN port is regarded as an internal MINI-LINK 6600 interface and supports up to 9,216 byte frames (jumbo frames).
Throughput	MMU 1001 supports simultaneous PDH and Ethernet transport over the same radio hop. The available bandwidth is split between the PDH and Ethernet traffic.
	The following maximum bit rate over the radio hop is supported:
	• ETSI: 1067 Mbps in a 112 MHz channel
	• ANSI: 767 Mbps in an 80 MHz channel
	See Table 42 and Table 43 (ETSI), orTable 44 and Table 45 (ANSI) for more information on capacity for MMU 1001.

Ta	ble	41
i u	DIC	TT

Ethernet WAN ports	Description
Latency	 The end to end latency for an Ethernet over Packet Link connection will differ based on different parameters, for example, link speed. Typical end to end latency contribution between two Ethernet ports on two nodes connected with Ethernet over packet Link with 558 Mbps link capacity is: Ø.23 ms for 64 byte frame size
	 0.29 ms for 1518 byte frame size
Buffer capacity	All egress WAN ports are assigned to the centralized deep buffer. The available buffer space is shared between the traffic classes (priority queues) to provide an effective burst absorption capability in a fair manner.
	The buffer size of traffic class queues can be regulated by configuring a user defined queue-size limit for tail drop or using WRED.

MMU 1001 is a high-capacity PDH/Ethernet modem with Adaptive Coding and Modulation (ACM), see Section 3.2 on page 12, HW support for XPIC (including XPIC Recovery and Restore), see Section 3.9 on page 28, and modulation schemes up to 4096 QAM.

MMU 1001 has HW support for Radio Link Bonding (RLB). For more information, see Section 3.5 on page 17.

MMU 1001 provides physical layer Sync over Radio Link.

MMU 1001 enables Ethernet transport over a radio hop, that is, a Packet Link. MMU 1001 uses the high speed connections to communicate with the Ethernet switch on the NPU boards.

MMU 1001 carries PDH traffic and have an interface to the TDM bus. The capacity split between packet and circuit transport is configured individually per MMU 1001.

MMU 1001 supports control of the ratio between Ethernet and PDH traffic sent over Hybrid Radio Links. The Ethernet part of the aggregated capacity is set with E1/DS1 granularity.

MMU 1001 has a backplane capacity of up to 2.5 Gbps.

MMU 1001 is used for the traffic capacities specified in Table 42 and Table 43 (ETSI), orTable 44 and Table 45 (ANSI).

	Capacity in Mbps						
Madulatia	Channel Bandwidth						
n	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz
4 QAM STRONG	8	18	37	53	75	108	150
4 QAM	10	21	43	62	87	126	175
16 QAM STRONG	17	37	74	106	150	216	301
16 QAM	20	43	87	124	175	252	351
32 QAM	26	54	109	156	220	317	441
64 QAM	32	68	136	195	276	397	553
128 QAM	38	80	161	231	326	469	653
256 QAM	44	92	186	266	377	541	754
512 QAM	47	99	198	284	402	577	804
512 QAM LIGHT	50	105	211	302	427	614	854
1024 QAM	53 ⁽¹⁾	110	221	316	447	643	895
1024 QAM LIGHT	56 ⁽¹⁾	116	233	334	472	679	945
2048 QAM	-	121 ⁽¹⁾	243	347	491	706	983
2048 QAM LIGHT	-	127 ⁽¹⁾	256	365	516	742	1033
4096 QAM	-	-	264	377	533	767 ⁽¹⁾	1067(1)
4096 QAM LIGHT	_	_	276	395	558	_	_

Table 42Aggregated PDH and Ethernet Capacity for MMU 1001 Adaptive
Physical Modes (ETSI)

(1) XPIC feature not supported by HW.

Table 43 Maximum PDH Capacities for MMU 1001 Adaptive Physical Modes (ETSI)

	Capacity in E1s										
	Channel Bandwidth										
Modulation	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz				
4 QAM STRONG	3×E1	7×E1	16×E1	23×E1	34×E1	47×E1	70×E1				
4 QAM	3×E1	8×E1	16×E1	27×E1	34×E1	56×E1	80×E1				
16 QAM STRONG	7×E1	16×E1	34×E1	49×E1	71×E1	80×E1	80×E1				
16 QAM	7×E1	19×E1	34×E1	57×E1	71×E1	80×E1	80×E1				
32 QAM	11×E1	24×E1	51×E1	73×E1	80×E1	80×E1	80×E1				
64 QAM	14×E1	31×E1	64×E1	80×E1	80×E1	80×E1	80×E1				
128 QAM	17×E1	37×E1	76×E1	80×E1	80×E1	80×E1	80×E1				
		Capacity in E1s									
-------------------	----------------------	----------------------	--------	--------	--------	----------------------	----------------------	--	--	--	--
	Channel Bandwidth										
Modulation	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz				
256 QAM	19×E1	43×E1	80×E1	80×E1	80×E1	80×E1	80×E1				
512 QAM	21×E1	46×E1	80×E1	80×E1	80×E1	80×E1	80×E1				
512 QAM LIGHT	21×E1	49×E1	80×E1	80×E1	80×E1	80×E1	80×E1				
1024 QAM	24×E1 ⁽¹⁾	51×E1	80×E1	80×E1	80×E1	80×E1	80×E1				
1024 QAM LIGHT	24×E1 ⁽¹⁾	54×E1	80×E1	80×E1	80×E1	80×E1	80×E1				
2048 QAM	_	57×E1 ⁽¹⁾	80×E1	80×E1	80×E1	80×E1	80×E1				
2048 QAM LIGHT	-	60×E1 ⁽¹⁾	80×E1	80×E1	80×E1	80×E1	80×E1				
4096 QAM	-	-	80×E1	80×E1	80×E1	80×E1 ⁽¹⁾	80×E1 ⁽¹⁾				
4096 QAM LIGHT	-	-	80×E1	80×E1	80×E1	-	-				

(1) XPIC feature not supported by HW.

Table 44Aggregated PDH and Ethernet Capacity for MMU 1001 Adaptive
Physical Modes (ANSI)

					Capacity	in Mbps				
					Channel B	Bandwidth				
Modul ation	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz
4 QAM STRO NG	12	26	37	39	53	67	75	81	108	150
4 QAM	14	30	43	46	62	78	87	94	126	175
16 QAM STRO NG	25	52	74	79	107	135	150	162	216	301
16 QAM	29	61	87	93	125	157	175	189	252	351
32 QAM	36	77	109	116	157	197	220	237	317	441
64 QAM	46	96	136	146	197	247	276	297	397	553
128 QAM	54	114	161	173	233	292	326	351	469	653
256 QAM	63	131	186	199	268	337	377	405	541	754
512 QAM	67	140	198	213	286	360	402	432	577	804
512 QAM LIGHT	71	149	211	226	304	382	427	459	614	854
1024 QAM	74 ⁽¹⁾	156	221	237	319	400	447	481	643	895

		Capacity in Mbps											
		Channel Bandwidth											
Modul ation	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz			
1024 QAM LIGHT	79 ⁽¹⁾	165	233	250	337	423	472	508	679	945			
2048 QAM	-	171	243	260	350	440	491	528	706	983			
2048 QAM LIGHT	-	180	256	273	368	462	516	555	742	1033			
4096 QAM	-	-	264	282	380	477	533	574	767 ⁽¹⁾	1067			
4096 QAM LIGHT	_	_	276	296	398	500	558	601	-	_			

(1) XPIC feature not supported by HW.

Table 45 Maximum PDH Capacities for MMU 1001 Adaptive Physical Modes (ANSI)

					Capacity	in DS1s				
					Channel E	Bandwidth	ı			
Modula	10	20	28	30	40	50	56	60	80	80
tion	MHz	MHz	MHz	MHz	MHz	MHz	MHz	MHz	MHz	MHz
4 QAM STRON G	5×DS1	14×DS 1	21×DS 1	23×DS 1	32×DS 1	40×DS 1	45×DS 1	49×DS 1	66×DS 1	80×DS 1
4 QAM	7×DS1	17×DS 1	25×DS 1	27×DS 1	37×DS 1	47×DS 1	54×DS 1	58×DS 1	78×DS 1	80×DS 1
16 QAM STRON G	13×DS 1	31×DS 1	45×DS 1	48×DS 1	66×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
16 QAM	16×DS	36×DS	53×DS	57×DS	77×DS	80×DS	80×DS	80×DS	80×DS	80×DS
	1	1	1	1	1	1	1	1	1	1
32 QAM	21×DS	46×DS	67×DS	72×DS	80×DS	80×DS	80×DS	80×DS	80×DS	80×DS
	1	1	1	1	1	1	1	1	1	1
64 QAM	27×DS	59×DS	80×DS							
	1	1	1	1	1	1	1	1	1	1
128	32×DS	70×DS	80×DS							
QAM	1	1	1	1	1	1	1	1	1	1
256	37×DS	80×DS								
QAM	1	1	1	1	1	1	1	1	1	1
512	40×DS	80×DS								
QAM	1	1	1	1	1	1	1	1	1	1
512 QAM LIGHT	43×DS 1	80×DS 1								
1024	45×DS	80×DS								
QAM	1 ⁽¹⁾	1	1	1	1	1	1	1	1	1

	Capacity in DS1s											
		Channel Bandwidth										
Modula tion	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	80 MHz		
1024 QAM LIGHT	48×DS 1 ⁽¹⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
2048 QAM	-	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
2048 QAM LIGHT	_	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
4096 QAM	-	-	80×DS 1 ⁽¹⁾	80×DS 1 ⁽¹⁾								
4096 QAM LIGHT	-	-	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	-	-		

(1) XPIC feature not supported by HW.

It is possible to configure all Adaptive Coding and Modulation (ACM) physical modes as static physical modes, except for 4096 QAM, by setting **Max Capacity – Modulation** and **Min Capacity – Modulation** to the same value, using MINI-LINK Node GUI. The ACM physical modes configured as static can only be used in hops configured with MMU 1001 on both sides.

Ethernet WAN Buffers

The WAN port buffers in MINI-LINK 6600 have been designed to handle burst and congestion in order to provide a high link utilization and throughput for high-speed data traffic.

Since extensive buffering has a negative impact on frame delay variation, it is important to have the possibility to regulate buffer/queue size for different traffic classes independently. This can be achieved by using WRED or packet aging.

This means that queues configured to handle delay variation sensitive traffic such as synchronization traffic, shall be regulated accordingly.

In contrast, for traffic queues for less delay variation sensitive traffic the TCP/IP has a congestion avoidance mechanism that is based on buffer utilization. In order to provide a high link utilization and high TCP throughput, queues configured to handle this type of traffic needs to be in the area of hundreds of milliseconds at the smallest congestion point.

15.12 MMU 1002

MMU 1002 is a dual carrier modem.

With one modem, the following is supported:

- Single WAN 1+0
- Single WAN 2+0 Radio Link Bonding (RLB)
- Dual WAN 2×(1+0)
- Single WAN 1+1 Radio Link Protection (RLP)

With two modems, the following is supported:

- 1+1 RLP with Equipment Protection (EQP)
- 2+0 RLB with EQP
- 2 x (2+0) RLB with EQP
- 4+0 RLB with EQP
- 2+2 RLP with EQP





Note: The coaxial interface on modem and radios is equipped with built-in gas discharge tubes for lightning protection.

Adding components to this interface (for example, extra lightning protection) can impact function and performance and should only be considered after consulting with Ericsson.

Table 46 shows the characteristics of the WAN ports.

Table 46

Ethernet WAN ports	Description
Interface types	2×Packet Link radio interface.

Ethernet WAN ports	Description
Frame size	An Ethernet over Packet Link WAN port is regarded as an internal MINI-LINK 6600 interface and supports up to 9,216 byte frames (jumbo frames).
Throughput	MMU 1002 supports simultaneous PDH and Ethernet transport over the same radio hop. The available bandwidth is split between the PDH and Ethernet traffic.
	The following maximum bit rate over the radio hop is supported:
	• ETSI: 1067 Mbps in a 112 MHz channel
	• ANSI: 803 Mbps in an 80 MHz channel
	See Table 47 and Table 48 (ETSI), or Table 49 and Table 50 (ANSI) for more information on capacity for MMU 1002.
Latency	The end to end latency for an Ethernet over Packet Link connection will differ based on different parameters, for example, link speed. Typical end to end latency contribution between two Ethernet ports on two nodes connected with Ethernet over packet Link with 558 Mbps link capacity is:
	• 0.23 ms for 64 byte frame size
	• 0.29 ms for 1518 byte frame size
Buffer capacity	All egress WAN ports are assigned to the centralized deep buffer. The available buffer space is shared between the traffic classes (priority queues) to provide an effective burst absorption capability in a fair manner.
	The buffer size of traffic class queues can be regulated by configuring a user defined queue-size limit for tail drop or using WRED.

MMU 1002 is a high-capacity PDH/Ethernet modem with support for XPIC (including XPIC Recovery and Restore), see Section 3.9 on page 28, Adaptive Coding and Modulation (ACM), see Section 3.2 on page 12, and modulation schemes up to 4096 QAM.

MMU 1002 supports Radio Link Bonding (RLB), see Section 3.5 on page 17.

MMU 1002 provides physical layer Sync over Radio Link.

 \geq

MMU 1002 enables Ethernet transport over a radio hop, that is, a Packet Link. MMU 1002 uses the high speed connections to communicate with the Ethernet switch on the NPU boards.

MMU 1002 carries PDH traffic and have an interface to the TDM bus. The capacity split between packet and circuit transport is configured individually per MMU 1002.

MMU 1002 supports control of the ratio between Ethernet and PDH traffic sent over Hybrid Radio Links. The Ethernet part of the aggregated capacity is set with E1/DS1 granularity.

MMU 1002 has a backplane capacity of up to 2×2.5 Gbps.

MMU 1002 is used for the traffic capacities specified in Table 47 and Table 48 (ETSI), or Table 49 and Table 50 (ANSI).

Using two MMU 1002s, a 4 x 4 MIMO configuration with two separated cross-polarization antennas can provide very high spectral efficiency.

MMU 1002 supports Carrier Aggregation, when connected to a MINI-LINK 6364 or a MINI-LINK 6365 radio, see Section 3.11 on page 37.

		Capacity in Mbps										
Madulatia			Ch	annel Bandwi	idth							
n	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz					
4 QAM STRONG	8	18	37/36 ⁽¹⁾⁽²⁾	53 ⁽²⁾	75/73 ⁽¹⁾⁽²⁾	108	150					
4 QAM	10	21	43/42 ⁽¹⁾⁽²⁾	62 ⁽²⁾	87/86 ⁽¹⁾⁽²⁾	126	175					
16 QAM STRONG	17	37	74/72 ⁽¹⁾⁽²⁾	106 ⁽²⁾	150/146	216	301					
16 QAM	20	43	87/85 ⁽¹⁾⁽²⁾	124 ⁽²⁾	175/171	252	351					
32 QAM	26	54	109/109 (1)(2)	156 ⁽²⁾	220/219	317	441					
64 QAM	32	68	136/133	195 ⁽²⁾	276/268	397	553					
128 QAM	38	80	161/157	231 ⁽²⁾	326/317	469	653					
256 QAM	44	92	186/182 (1)(2)	266 ⁽²⁾	377/366	541	754					
512 QAM	47	99	198/194 (1)(2)	284 ⁽²⁾	402/390	577	804					
512 QAM LIGHT	50	105	2 <u>11</u> /206	302 ⁽²⁾	427/415	614	854					
1024 QAM	53 ⁽³⁾	110	221 ⁽²⁾	316 ⁽²⁾	447/435	643	895					

Table 47 Aggregated PDH and Ethernet Capacity for MMU 1002 Adaptive Physical Modes (ETSI)

		Capacity in Mbps									
	Channel Bandwidth										
n	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz				
1024 QAM LIGHT	56 ⁽³⁾	116	233 ⁽²⁾	334 ⁽²⁾	472/459	679	945				
2048 QAM	-	121 ⁽³⁾	243 ⁽²⁾	347 ⁽²⁾	491 ⁽²⁾	706	983				
2048 QAM LIGHT	-	127 ⁽³⁾	256 ⁽²⁾	365 ⁽²⁾	516	742	1033				
4096 QAM	-	-	264 ⁽²⁾	377	533	767	1067				
4096 QAM LIGHT	-	-	276	395	558	803	-				
8192 QAM ⁽³⁾⁽⁴⁾	-	-	283	405	-	-	-				

(1) MIMO supported. Smaller capacity applies for MIMO configuration.

(2) Carrier Aggregation supported.

(3) XPIC not supported.

(4) 8192 QAM is not supported as static modulation. It is recommended to set radio output power to 0 dBm or higher.

Table 48	Maximum PDH Capacities for MMU 1002 Adaptive Physical Mo	des
	(ETSI)	

	Capacity in E1s ⁽¹⁾										
			Cho	annel Bandwi	dth						
Modulation	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz				
4 QAM STRONG	2×E1	6×E1	15×E1/15 ×E1 ⁽²⁾⁽³⁾	23×E1 ⁽³⁾	34×E1/33 ×E1 ⁽²⁾⁽³⁾	47×E1	70×E1				
4 QAM	3×E1	8×E1	18×E1/18 ×E1 ⁽²⁾⁽³⁾	28×E1 ⁽³⁾	40×E1/39 ×E1 ⁽²⁾⁽³⁾	56×E1	80×E1				
16 QAM STRONG	6×E1	15×E1	34×E1/33 ×E1 ⁽²⁾⁽³⁾	49×E1 ⁽³⁾	70×E1/68 ×E1	80×E1	80×E1				
16 QAM	7×E1	18×E1	39×E1/38 ×E1 ⁽²⁾⁽³⁾	57×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1	80×E1				
32 QAM	10×E1	24×E1	50×E1/50 ×E1 ⁽²⁾⁽³⁾	73×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1	80×E1				
64 QAM	13×E1	30×E1	63×E1/62 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1	80×E1				
128 QAM	16×E1	36×E1	75×E1/74 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1	80×E1				
256 QAM	19×E1	42×E1	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1	80×E1				
512 QAM	21×E1	45×E1	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1	80×E1				
512 QAM LIGHT	22×E1	48×E1	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1	80×E1				
1024 QAM	23×E1 ⁽⁴⁾	51×E1	80×E1 ⁽³⁾	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1	80×E1				
1024 QAM LIGHT	25×E1 ⁽⁴⁾	54×E1	80×E1 ⁽³⁾	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1	80×E1				

	Capacity in E1s ⁽¹⁾									
	Channel Bandwidth									
Modulation	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz			
2048 QAM	-	56×E1 ⁽⁴⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾	80×E1	80×E1			
2048 QAM LIGHT	-	59×E1 ⁽⁴⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾	80×E1	80×E1	80×E1			
4096 QAM	-	-	80×E1 ⁽³⁾	80×E1	80×E1	80×E1	80×E1			
4096 QAM LIGHT	-	-	80×E1	80×E1	80×E1	80×E1	80×E1			
8192 QAM ⁽⁴⁾	-	_	80×E1 ⁽⁵⁾	80×E1 ⁽⁵⁾	-	-	_			

(1) The table shows the maximum capacity values for one direction. When the NE is configured for two directions, the capacity decreases to half of these values.

(2) MIMO supported. Smaller capacity applies for MIMO configuration.

(3) Carrier Aggregation supported.

(4) XPIC not supported.

(5) 8192 QAM is not supported as a minimum modulation.

Table 49Aggregated PDH and Ethernet Capacity for MMU 1002 Adaptive
Physical Modes (ANSI)

		Capacity in Mbps											
					Channel E	Bandwidth	ı						
Modul ation	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz			
4 QAM STRO NG	12	26	37	39/39 ,0 ⁽¹⁾⁽²⁾	53/52 ,4 ⁽¹⁾⁽²⁾	67 ⁽²⁾	75	81	108	150			
4 QAM	14	30	43	46/45 ,8 ⁽¹⁾⁽²⁾	62/6 <u>1</u> ,5 ⁽¹⁾⁽²⁾	78 ⁽²⁾	87	94	126	175			
16 QAM STRO NG	25	52	74	79/78 ,0 ⁽¹⁾⁽²⁾	107/1 04,7 ^(I)	135 ⁽²⁾	150	162	216	301			
16 QAM	29	61	87	93/9 <u>1</u> ,0 ⁽¹⁾⁽²⁾	125/1 22,2 ⁽¹⁾	157 ⁽²⁾	175	189	252	351			
32 QAM	36	77	109	116/1 17,0 ⁽¹⁾	157/1 57,1 ^(I)	197 ⁽²⁾	220	237	317	441			
64 QAM	46	96	136	146/1 43,0 ^(T)	197/1 92,0 ^(I)	247 ⁽²⁾	276	297	397	553			
128 QAM	54	114	161	173/1 69,0	233/2 26,9	292 ⁽²⁾	326	351	469	653			
256 QAM	63	131	186	199/1 95,0 ⁽¹⁾	268/2 61,9 ⁽¹⁾	337 ⁽²⁾	377	405	541	754			
512 QAM	67	140	198	213/2 08,0 ^(T)	286/2 79,3 ^(II)	360 ⁽²⁾	402	432	577	804			

					Canadita	in Mhaa						
					Capacity	in Mops						
		Channel Bandwidth										
Modul ation	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz		
512 QAM LIGHT	71	149	211	226/2 21,0 ⁽¹⁾	304/2 96,8 ⁽¹⁾	382 ⁽²⁾	427	459	614	854		
1024 QAM	74 ⁽³⁾	156	221	237 ⁽²⁾	319 ⁽²⁾	400 ⁽²⁾	447	481	643	895		
1024 QAM LIGHT	79 ⁽³⁾	165	233	250 ⁽²⁾	337 ⁽²⁾	423 ⁽²⁾	472	508	679	945		
2048 QAM	-	171	243	260 ⁽²⁾	350 ⁽²⁾	440 ⁽²⁾	491	528	706	983		
2048 QAM LIGHT	_	180	256	273 ⁽²⁾	368 ⁽²⁾	462	516	555	742	1033		
4096 QAM	-	-	264	282 ⁽²⁾	380	477	533	574	767 ⁽³⁾	1067 3		
4096 QAM LIGHT	_	_	276	296	398	500	558	601	803	_		
8192 OAM (3)(4)	-	-	-	304	409	-	-	-	-	-		

(1) MIMO supported. The second capacity applies for MIMO configuration.

(2) Carrier Aggregation supported.

(3) XPIC not supported.

(4) 8192 QAM is not supported as static modulation. It is recommended to set radio output power to 0 dBm or higher.

Table 50 Maximum PDH Capacities for MMU 1002 Adaptive Physical Modes (ANSI)

					Capacity	in DS1s ⁽¹⁾							
		Channel Bandwidth											
Modula tion	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz			
4 QAM STRON G	5×DS1	14×DS 1	21×DS 1	23×D S1/22 ×DS1 (2)(3)	32×D S1/31 ×DS1 (2)(3)	40×DS 1 ⁽³⁾	45×DS 1	49×DS 1	66×DS 1	80×DS 1			
4 QAM	7×DS1	17×DS 1	25×DS 1	27×D S1/26 ×DS1 (2)(3)	37×D S1/36 ×DS1 (2)(3)	47×DS 1 ⁽³⁾	54×DS 1	58×DS 1	78×DS 1	80×DS 1			
16 QAM STRON G	13×DS 1	31×DS 1	45×DS 1	48×D S1/47 ×DS1 (2)(3)	66×D S1/64 ×DS1 (2)(3)	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1			
16 QAM	16×DS 1	36×DS 1	53×DS 1	57×D S1/55 ×DS1 (2)(3)	77×D S1/75 ×DS1 (2)(3)	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1			

					Capacity	in DS1s ⁽¹⁾)					
		Channel Bandwidth										
Modula tion	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz		
32 QAM	21×DS 1	46×DS 1	67×DS 1	72×D S1/72 ×DS1 (2)(3)	80×D S1/80 ×DS1 (2)(3)	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
64 QAM	27×DS 1	59×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)(3)	80×D S1/80 ×DS1 (2)(3)	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
128 QAM	32×DS 1	70×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)(3)	80×D S1/80 ×DS1 (2)(3)	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
256 QAM	37×DS 1	80×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)(3)	80×D S1/80 ×DS1 (2)(3)	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
512 QAM	40×DS 1	80×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)(3)	80×D S1/80 ×DS1 (2)(3)	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
512 QAM LIGHT	43×DS 1	80×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)(3)	80×D S1/80 ×DS1 (2)(3)	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
1024 QAM	45×DS 1 ⁽⁴⁾	80×DS 1	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
1024 QAM LIGHT	48×DS 1 ⁽⁴⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
2048 QAM	_	80×DS 1	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
2048 QAM LIGHT	_	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1		
4096 QAM	-	-	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1 ⁽⁴⁾	80×DS 1 ⁽⁴⁾		
4096 QAM LIGHT	-	-	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	-		
8192 ₍₄₎ QAM ⁽⁴⁾	-	-	-	80×DS 1 ⁽⁵⁾	80×DS 1	-	-	-	-	-		

(1) The table shows the maximum capacity values for one direction. When the NE is configured for two directions, the capacity decreases to half of these values.

(2) MIMO supported. Smaller capacity applies for MIMO configuration.

(3) Carrier Aggregation supported.

(4) XPIC not supported.

(5) 8192 QAM is not supported as minimum modulation.

It is possible to configure all Adaptive Coding and Modulation (ACM) physical modes as static physical modes, except for 4096 QAM, by setting **Max Capacity – Modulation** and **Min Capacity – Modulation** to the same value, using MINI-LINK Node GUI. The ACM physical modes configured as static can only be used in hops configured with MMU 1002 on both sides.

Ethernet WAN Buffers

The WAN port buffers in MINI-LINK 6600 have been designed to handle burst and congestion in order to provide a high link utilization and throughput for high-speed data traffic.

Since extensive buffering has a negative impact on frame delay variation, it is important to have the possibility to regulate buffer/queue size for different traffic classes independently. This can be achieved by using WRED or packet aging.

This means that queues configured to handle delay variation sensitive traffic such as synchronization traffic, shall be regulated accordingly.

In contrast, for traffic queues for less delay variation sensitive traffic the TCP/IP has a congestion avoidance mechanism that is based on buffer utilization. In order to provide a high link utilization and high TCP throughput, queues configured to handle this type of traffic needs to be in the area of hundreds of milliseconds at the smallest congestion point.

15.13 MMU 1004

MMU 1004 is a dual carrier modem.

With one modem, the following is supported:

- Single WAN 1+0
- Single WAN 2+0 Radio Link Bonding (RLB)
- Dual WAN 2×(1+0)
- Single WAN 1+1 Radio Link Protection (RLP)

With two modems, the following is supported:

- 1+1 RLP with Equipment Protection (EQP)
- 2+0 RLB with EQP
- 2 x (2+0) RLB with EQP
- 4+0 RLB with EQP
- 2+2 RLP with EQP



Figure 136 MMU 1004

Note: The coaxial interface on modem and radios is equipped with built-in gas discharge tubes for lightning protection.

Adding components to this interface (for example, extra lightning protection) can impact function and performance and should only be considered after consulting with Ericsson.

Table 51 shows the characteristics of the WAN ports.

Ethernet WAN ports	Description
Interface types	2×Packet Link radio interface.
Frame size	An Ethernet over Packet Link WAN port is regarded as an internal MINI-LINK 6600 interface and supports up to 9,216 byte frames (jumbo frames).
Throughput	MMU 1004 supports simultaneous PDH and Ethernet transport over the same radio hop. The available bandwidth is split between the PDH and Ethernet traffic.
	The following maximum bit rate over the radio hop is supported:
	• ETSI: 1067 Mbps in a 112 MHz channel
	• ANSI: 803 Mbps in an 80 MHz channel
	See Table 47 and Table 48 (ETSI), or Table 49 and Table 50 (ANSI) for more information on capacity for MMU 1004.

Table 51	
----------	--

Ethernet WAN ports	Description
Latency	The end to end latency for an Ethernet over Packet Link connection will differ based on different parameters, for example, link speed. Typical end to end latency contribution between two Ethernet ports on two nodes connected with Ethernet over packet Link with 558 Mbps link capacity is: • 0.23 ms for 64 byte frame size • 0.29 ms for 1518 byte frame size
Buffer capacity	All egress WAN ports are assigned to the centralized deep buffer. The available buffer space is shared between the traffic classes (priority queues) to provide an effective burst absorption capability in a fair manner. The buffer size of traffic class queues can be regulated by configuring a user defined queue-size limit for tail drop or using WRED.

MMU 1004 is a high-capacity PDH/Ethernet modem with support for XPIC (including XPIC Recovery and Restore), see Section 3.9 on page 28, Adaptive Coding and Modulation (ACM), see Section 3.2 on page 12, and modulation schemes up to 4096 QAM.

MMU 1004 supports Radio Link Bonding (RLB), see Section 3.5 on page 17.

MMU 1004 provides physical layer Sync over Radio Link.

MMU 1004 enables Ethernet transport over a radio hop, that is, a Packet Link. MMU 1004 uses the high speed connections to communicate with the Ethernet switch on the NPU boards.

MMU 1004 carries PDH traffic and have an interface to the TDM bus. The capacity split between packet and circuit transport is configured individually per MMU 1004.

MMU 1004 supports control of the ratio between Ethernet and PDH traffic sent over Hybrid Radio Links. The Ethernet part of the aggregated capacity is set with E1/DS1 granularity.

MMU 1004 has a backplane capacity of up to 2×2.5 Gbps.

MMU 1004 is used for the traffic capacities specified in Table 52 and Table 53 (ETSI), or Table 54 and Table 55 (ANSI).

Using two MMU 1004s, a 4 x 4 MIMO configuration with two separated cross-polarization antennas can provide very high spectral efficiency.

MMU 1004 supports Carrier Aggregation, when connected to a MINI-LINK 6364 or MINI-LINK 6365 radio, with up to maximum 2 x 112 MHz, see Section 3.11 on page 37.

			C	apacity in Mb	ps		
Modulatio			Che	annel Bandwi	idth		
n	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz
4 QAM STRONG	8	18	37/36 ⁽¹⁾⁽²⁾	53 ⁽²⁾	75/73 ⁽¹⁾⁽²⁾	108 ⁽²⁾	150 ⁽²⁾
4 QAM	10	21	43/42 ⁽¹⁾⁽²⁾	62 ⁽²⁾	87/86 ⁽¹⁾⁽²⁾	126 ⁽²⁾	175 ⁽²⁾
16 QAM STRONG	17	37	74/72 ⁽¹⁾⁽²⁾	106 ⁽²⁾	150/146	216 ⁽²⁾	301 ⁽²⁾
16 QAM	20	43	87/85 ⁽¹⁾⁽²⁾	124 ⁽²⁾	175/171 (1)(2)	252 ⁽²⁾	351 ⁽²⁾
32 QAM	26	54	109/109 (1)(2)	156 ⁽²⁾	220/219	317 ⁽²⁾	441 ⁽²⁾
64 QAM	32	68	136/133	195 ⁽²⁾	276/268	397 ⁽²⁾	553 ⁽²⁾
128 QAM	38	80	161/157	231 ⁽²⁾	326/317	469 ⁽²⁾	653 ⁽²⁾
256 QAM	44	92	186/182	266 ⁽²⁾	377/366	541 ⁽²⁾	754 ⁽²⁾
512 QAM	47	99	198/194	284 ⁽²⁾	402/390	577 ⁽²⁾	804 ⁽²⁾
512 QAM LIGHT	50	105	211/206	302 ⁽²⁾	427/415	614 ⁽²⁾	854 ⁽²⁾
1024 QAM	53 ⁽³⁾	110	221 ⁽²⁾	316 ⁽²⁾	447/435	643 ⁽²⁾	895 ⁽²⁾
1024 QAM LIGHT	56 ⁽³⁾	116	233 ⁽²⁾	334 ⁽²⁾	472/459	679	945
2048 QAM	-	121 ⁽³⁾	243 ⁽²⁾	347 ⁽²⁾	491 ⁽²⁾	706	983
2048 QAM LIGHT	-	127 ⁽³⁾	256 ⁽²⁾	365 ⁽²⁾	516	742	1033
4096 QAM	-	-	264 ⁽²⁾	377	533	767	1067
4096 QAM LIGHT	_	_	276	395	558	803	-
8192 QAM ⁽³⁾⁽⁴⁾	-	-	283	405	-	_	-

Table 52Aggregated PDH and Ethernet Capacity for MMU 1004 Adaptive
Physical Modes (ETSI)

(1) MIMO supported. Smaller capacity applies for MIMO configuration.

(2) Carrier Aggregation supported.

(3) XPIC not supported.

(4) 8192 QAM is not supported as static modulation. It is recommended to set radio output power to 0 dBm or higher.

	Capacity in E1s ⁽¹⁾											
			Cho	annel Bandwi	idth							
Modulation	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	80 MHz	112 MHz					
4 QAM STRONG	2×E1	6×E1	15×E1/15 ×E1	23×E1 ⁽³⁾	34×E1/33 ×E1 ⁽²⁾⁽³⁾	47×E1 ⁽³⁾	70×E1 ⁽³⁾					
4 QAM	3×E1	8×E1	18×E1/18 ×E1 ⁽²⁾⁽³⁾	28×E1 ⁽³⁾	40×E1/39 ×E1 ⁽²⁾⁽³⁾	56×E1 ⁽³⁾	80×E1 ⁽³⁾					
16 QAM STRONG	6×E1	15×E1	34×E1/33 ×E1	49×E1 ⁽³⁾	70×E1/68 ×E1	80×E1 ⁽³⁾	80×E1 ⁽³⁾					
16 QAM	7×E1	18×E1	39×E1/38 ×E1 ⁽²⁾⁽³⁾	57×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾					
32 QAM	10×E1	24×E1	50×E1/50 ×E1	73×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾					
64 QAM	13×E1	30×E1	63×E1/62 ×E1	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾					
128 QAM	16×E1	36×E1	75×E1/74 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾					
256 QAM	19×E1	42×E1	80×E1/80 ×E1	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾					
512 QAM	21×E1	45×E1	80×E1/80 ×E1	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾					
512 QAM LIGHT	22×E1	48×E1	80×E1/80 ×E1	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾					
1024 QAM	23×E1 ⁽⁴⁾	51×E1	80×E1 ⁽³⁾	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾					
1024 QAM LIGHT	25×E1 ⁽⁴⁾	54×E1	80×E1 ⁽³⁾	80×E1 ⁽³⁾	80×E1/80 ×E1 ⁽²⁾⁽³⁾	80×E1	80×E1					
2048 QAM	-	56×E1 ⁽⁴⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾	80×E1	80×E1					
2048 QAM LIGHT	-	59×E1 ⁽⁴⁾	80×E1 ⁽³⁾	80×E1 ⁽³⁾	80×E1	80×E1	80×E1					
4096 QAM	-	-	80×E1 ⁽³⁾	80×E1	80×E1	80×E1	80×E1					
4096 QAM LIGHT	-	-	80×E1	80×E1	80×E1	80×E1	80×E1					
8192 QAM ⁽⁴⁾	-	-	80×E1 ⁽⁵⁾	80×E1 ⁽⁵⁾	-	-	-					

Table 53 Maximum PDH Capacities for MMU 1004 Adaptive Physical Modes (ETSI)

(1) The table shows the maximum capacity values for one direction. When the NE is configured for two directions, the capacity decreases to half of these values.

(2) MIMO supported. Smaller capacity applies for MIMO configuration.

(3) Carrier Aggregation supported.

(4) XPIC not supported.

(5) 8192 QAM is not supported as minimum modulation.

					Capacity	in Mbps				
					Channel E	Bandwidtł	ı			
Modul ation	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz
4 QAM STRO NG	12	26	37	39/39 ,0 ⁽¹⁾⁽²⁾	53/ <u>52</u> ,4 ⁽¹⁾⁽²⁾	67 ⁽²⁾	75	81 ⁽²⁾	108 ⁽²⁾	150
4 QAM	14	30	43	46/45 ,8 ⁽¹⁾⁽²⁾	62/61 ,5 ⁽¹⁾⁽²⁾	78 ⁽²⁾	87	94 ⁽²⁾	126 ⁽²⁾	175
16 QAM STRO NG	25	52	74	79/78 ,0 ⁽¹⁾⁽²⁾	107/1 04,7 ⁽¹⁾	135 ⁽²⁾	150	162 ⁽²⁾	216 ⁽²⁾	301
16 QAM	29	61	87	93/9 <u>1</u> ,0 ⁽¹⁾⁽²⁾	125/1 22,2 ⁽¹⁾	157 ⁽²⁾	175	189 ⁽²⁾	252 ⁽²⁾	351
32 QAM	36	77	109	116/1 17,0 ⁽¹⁾	157/1 57,1 ⁽¹⁾	197 ⁽²⁾	220	237 ⁽²⁾	317 ⁽²⁾	441
64 QAM	46	96	136	146/1 43,0 ⁽¹⁾	197/1 92,0 ⁽¹⁾	247 ⁽²⁾	276	297 ⁽²⁾	397 ⁽²⁾	553
128 QAM	54	114	161	173/1 69,0 ⁽¹⁾	233/2 26,9 ⁽¹⁾	292 ⁽²⁾	326	351 ⁽²⁾	469 ⁽²⁾	653
256 QAM	63	131	186	199/1 95,0 ⁽¹⁾	268/2 61,9 ⁽¹⁾	337 ⁽²⁾	377	405 ⁽²⁾	541 ⁽²⁾	754
512 QAM	67	140	198	213/2 08,0 ⁽¹⁾	286/2 79,3 ⁽¹⁾	360 ⁽²⁾	402	432 ⁽²⁾	577 ⁽²⁾	804
512 QAM LIGHT	71	149	211	226/2 21,0 ⁽¹⁾	304/2 96,8 ⁽¹⁾	382 ⁽²⁾	427	459 ⁽²⁾	614 ⁽²⁾	854
1024 QAM	74 ⁽³⁾	156	221	237 ⁽²⁾	319 ⁽²⁾	400 ⁽²⁾	447	481 ⁽²⁾	643 ⁽²⁾	895
1024 QAM LIGHT	79 ⁽³⁾	165	233	250 ⁽²⁾	337 ⁽²⁾	423 ⁽²⁾	472	508 ⁽²⁾	679	945
2048 QAM	-	171	243	260 ⁽²⁾	350 ⁽²⁾	440 ⁽²⁾	491	528 ⁽²⁾	706	983
2048 QAM LIGHT	_	180	256	273 ⁽²⁾	368 ⁽²⁾	462	516	555	742	1033
4096 QAM	-	-	264	282 ⁽²⁾	380	477	533	574	767 ⁽³⁾	1067

Table 54Aggregated PDH and Ethernet Capacity for MMU 1004 Adaptive
Physical Modes (ANSI)

 \leq

					Capacity	in Mbps					
		Channel Bandwidth									
Modul ation	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz	
4096 QAM LIGHT	-	-	276	296	398	500	558	601	803	-	
8192 QAM 3/4)	-	-	-	304	409	-	-	-	-	-	

(1) MIMO supported. The second capacity applies for MIMO configuration.

(2) Carrier Aggregation supported.

(3) XPIC not supported.

(4) 8192 QAM is not supported as static modulation. It is recommended to set radio output power to 0 dBm or higher.

Table 55 Maximum PDH Capacities for MMU 1004 Adaptive Physical Modes (ANSI)

					Capacity	in DS1s ⁽¹⁾				
					Channel E	Bandwidth	ı		-	
Modula tion	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz
4 QAM STRON G	5×DS1	14×DS 1	21×DS 1	23×D S1/22 ×DS1 (2)(3)	32×D S1/31 ×DS1 (2)(3)	40×DS 1	45×DS 1	49×DS 1 ⁽³⁾	66×DS 1 ⁽³⁾	80×DS 1
4 QAM	7×DS1	17×DS 1	25×DS 1	27×D S1/26 ×DS1 (2)(3)	37×D S1/36 ×DS1 (2)(3)	47×DS 1 ⁽³⁾	54×DS 1	58×DS 1	78×DS 1	80×DS 1
16 QAM STRON G	13×DS 1	31×DS 1	45×DS 1	48×D S1/47 ×DS1 (2)(3)	66×D S1/64 ×DS1 (2)(3)	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
16 QAM	16×DS 1	36×DS 1	53×DS 1	57×D S1/55 ×DS1 (2)(3)	77×D S1/75 ×DS1 (2)(3)	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
32 QAM	21×DS 1	46×DS 1	67×DS 1	72×D S1/72 ×DS1 (2)(3)	80×D S1/80 ×DS1 (2)(3)	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
64 QAM	27×DS 1	59×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)(3)	80×D S1/80 ×DS1 (2)(3)	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1 ⁽³⁾	80×DS 1
128 QAM	32×DS 1	70×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)(3)	80×D S1/80 ×DS1 (2)(3)	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
256 QAM	37×DS 1	80×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)(3)	80×D S1/80 ×DS1 (2)(3)	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1 ⁽³⁾	80×DS 1

		Capacity in DS1s ⁽¹⁾								
		Channel Bandwidth								
Modula tion	10 MHz	20 MHz	28 MHz	30 MHz	40 MHz	50 MHz	56 MHz	60 MHz	80 MHz	112 MHz
512 QAM	40×DS 1	80×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)(3)	80×D S1/80 ×DS1 (2)(3)	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1 ⁽³⁾	80×DS 1
512 QAM LIGHT	43×DS 1	80×DS 1	80×DS 1	80×D S1/80 ×DS1 (2)(3)	80×D S1/80 ×DS1 (2)(3)	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
1024 QAM	45×DS 1 ⁽⁴⁾	80×DS 1	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1 ⁽³⁾	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1 ⁽³⁾	80×DS 1
1024 QAM LIGHT	48×DS 1 ⁽⁴⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
2048 QAM	-	80×DS 1	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1 ⁽³⁾	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1
2048 QAM LIGHT	-	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1
4096 QAM	-	-	80×DS 1	80×DS 1 ⁽³⁾	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1 ⁽⁴⁾	80×DS 1 ⁽⁴⁾
4096 QAM LIGHT	-	-	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	80×DS 1	-
8192 ₍₄₎ QAM ⁽⁴⁾	-	_	-	80×DS 1 ⁽⁵⁾	80×DS 1 ⁽⁵⁾	_	_	-	_	_

(1) The table shows the maximum capacity values for one direction. When the NE is configured for two directions, the capacity decreases to half of these values.

(2) MIMO supported. Smaller capacity applies for MIMO configuration.

(3) Carrier Aggregation supported.

(4) XPIC not supported.

(5) 8192 QAM is not supported as a minimum modulation.

It is possible to configure all Adaptive Coding and Modulation (ACM) physical modes as static physical modes, except for 4096 QAM, by setting **Max Capacity – Modulation** and **Min Capacity – Modulation** to the same value, using MINI-LINK Node GUI. The ACM physical modes configured as static can only be used in hops configured with MMU 1004 on both sides.

Ethernet WAN Buffers

The WAN port buffers in MINI-LINK 6600 have been designed to handle burst and congestion in order to provide a high link utilization and throughput for high-speed data traffic.

Since extensive buffering has a negative impact on frame delay variation, it is important to have the possibility to regulate buffer/queue size for different traffic classes independently. This can be achieved by using WRED or packet aging.

This means that queues configured to handle delay variation sensitive traffic such as synchronization traffic, shall be regulated accordingly.

In contrast, for traffic queues for less delay variation sensitive traffic the TCP/IP has a congestion avoidance mechanism that is based on buffer utilization. In order to provide a high link utilization and high TCP throughput, queues configured to handle this type of traffic needs to be in the area of hundreds of milliseconds at the smallest congestion point.

15.14 LTU 1001

LTU 1001 is a Line Termination Unit (LTU) with 16 E1/DS1 ports for PDH.

LTU 1001 can be used in MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, MINI-LINK 6694, MINI-LINK 6654, and MINI-LINK 6655.

Table 56 LTU 1001 Interfaces

Interfaces	16×E1/DS1
	• ETSI: 75 Ω or 120 Ω G.703
	• ANSI: 100 Ω Balanced G.703



Front Interfaces on the LTU 1001 Figure 137

15.15 LTU 1002

LTU 1002 is a Line Termination Unit (LTU) with 16 E1/DS1 ports for PDH and an STM-1 port with up to 63×E1 for SDH. The 16 E1 ports and the STM-1 port with up to 63×E1s can all be used simultaneously. The actual number of supported ports depends on the selected board profile.

Note: SDH on LTU 1002 is only available for ETSI, and not for ANSI.

The following board profiles can be set for LTU1002:

PDH and SDH	Provides 16×E1 front interfaces and one STM-1 front interface for up to 63×E1s.
PDH and CES	Provides 16×E1/DS1 front interfaces and 16×CES E1/DS1s through the backplane.
CES only	Provides 32×CES E1/DS1s through the backplane.
SDH48 and CES32	Provides one STM-1 front interface for up to 48×E1s, and 32×CES E1s through the backplane.
SDH63 and CES16	Provides STM-1 termination up to 63xE1s combined with 16xE1s circuit emulation towards the backplane.

PDH and SDH48 and CES16

Provides a combination of STM-1 (48xE1) and 16x physical E1 termination together with 16xE1s circuit emulation towards the backplane.

LTU 1002 can be used in MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, MINI-LINK 6694, MINI-LINK 6654, and MINI-LINK 6655.

The SDH functionality consists of front termination, terminal multiplexer, MSP protection and the possibility to send DCN traffic on DCC. The Terminal Multiplexer terminates one STM-1 with 63×E1 mapped asynchronously into 63×VC-12.

At aggregation nodes LTU 1002 acts as an interface between the optical domain and the microwave domain by providing an effective optical northbound interface using one STM-1 connection instead of n×E1 interfaces.

In ring configurations two LTU 1002 can be connected "back-to-back" to allow local add/drop of up to 63×E1. Using two LTU 1002 in a pair provides Multiplexer Section Protection (MSP) 1+1.

LTU 1002 has support for CES, enabling E1/DS1s being transported over packet.

Interfaces	1xSDH SFP (supporting STM-1 for up to 63×E1)
	16xE1/DS1
	• ETSI: 75 Ω or 120 Ω G.703
	• ANSI:100 Ω Balanced G.703

Table 57 LTU 1002 Interfaces



Figure 138 Front Interfaces on the LTU 1002

15.16 ETU 1001

ETU 1001 is an Ethernet Termination Unit (ETU) that supports up to four Ethernet ports. The actual number of ports depends upon the NPU type and which APU slot that is used.



Figure 139 ETU 1001

ETU 1001 can be seen as a port extension to the NPU, where basically the same Ethernet L2 and L3 functionality as on the NPU is supported on the ETU.

ETU 1001 supports SyncE and 1588 Timestamping.

Figure 140 shows the available interfaces of the ETU 1001.





15.17 ETU 1002

ETU 1002 is an ETU that supports up to four Ethernet ports. The actual number of supported ports depends upon the NPU type, the selected board profile, and which APU slot is used.

ETU 1002 can be seen as a port extension to the NPU or PNM, where basically the same Ethernet L2 and L3 functionality as on the NPU is supported on the ETU.

ETU 1002 has four SFP cages supporting 4x1G Ethernet, and one SFP+ cage supporting 10G Ethernet, on the front.

1G and 10G ports cannot be used simultaneously.

The following board profiles can be set for ETU 1002:

4X1G Provides four 1G front interfaces.

1X10G Provides one 10G front interface.

The 10G port can be used only in MINI-LINK 6691, MINI-LINK 6692, MINI-LINK 6693, and MINI-LINK 6694.

ETU 1002 supports SyncE and 1588 Timestamping on the 1GE and 10GE interfaces. When using the 10GE interface, ETU 1002 needs to be placed in a 10GE capable slot of the enclosure.

Note: SyncE is not supported on the SFPe interface.



Figure 141 Front Interfaces of ETU 1002

15.18 Radio Units

15.18.1 Overview

The basic function of the radio is to generate and receive the RF signal and convert it to/from the signal format in the radio cable, connecting the radio and the MMU. It can be combined with a wide range of antennas in integrated or separate installation. The radio connects to the antenna at the waveguide interface. Disconnection and replacement of the radio can be done without affecting the antenna alignment.

There are several types of Radio Units that can be used together with MINI-LINK 6600:

- MINI-LINK 6363 (including support for E-band 80 GHz), see Figure 142. MINI-LINK 6363 comes in two versions: MINI-LINK 6363, optimized for the highest system gain, and MINI-LINK 6363/2, optimized for the lowest power consumption.
- MINI-LINK 6364, see Figure 142, is a sub-band free radio optimized for highest output power and has support for Carrier Aggregation 2x28 MHz to 112 MHz when connected to MMU 1002 or MMU 1004.
- MINI-LINK 6365, see Figure 142, is optimized for highest system gain and has support for Carrier Aggregation from 2x28 MHz to 2x112 MHz when connected to MMU 1002 or MMU 1004.
- RAU2 X, see Figure 143.

DC power to the radio is supplied from the MMU through the radio cable.

The radio has a weather proof casing painted white (MINI-LINK 6363, MINI-LINK 6364, and MINI-LINK 6365) or light gray (RAU2 X), containing for example microwave electronics and a diplexer. It has mechanical features for easy installation on integrated antennas. It also has an optional handle (MINI-LINK 6363, MINI-LINK 6364, and MINI-LINK 6365) or an integral handle (RAU2 X) for simplified handling during installation.

The radio is independent of traffic capacity. The operating frequency is determined by the radio only and is set in advance at the factory and configured on site using MINI-LINK Node GUI. Frequency channel arrangements are available according to ITU-R and ETSI/FCC recommendations.

Note: The coaxial interface on radios and modems is equipped with built in gas discharge tubes, for lightning protection.

Adding components to this interface (for example, extra lightning protection) can impact function and performance and should only be considered after consulting with Ericsson.



Figure 142 MINI-LINK 6363, MINI-LINK 6364, or MINI-LINK 6365 Mechanical Design



Figure 143 RAU2 X Mechanical Design

15.18.2 External Interfaces



Figure 144 External Interfaces on the MINI-LINK 6363, MINI-LINK 6364, or MINI-LINK 6365



Figure 145 External Interfaces on the RAU2 X

Item Description

- 1 Radio cable connection to the MMU, 50 Ω N-type connector. The connector interface is equipped with gas discharge tubes for lightning protection.
- 2 Protective ground point for connection to mast ground.
- **3** Test port for antenna alignment.
- 4 Red LED: Unit alarm. Green LED: Power on.

15.18.3 Radio Modulations

The radio hardware can generate and receive RF signal in single carrier mode.

The tables below show the supported modulations for the radio hardware in single carrier mode and the modulations that also support XPIC and MIMO.

All modulations that support single carrier mode are marked with x. These modulations also support XPIC unless otherwise stated.

MINI-LINK 6363 and MINI-LINK 6363/2 (ETSI)

Table 58	Supported Modulations for All Frequency Bands (Excluding E-band
	80 GHz) with an R-state < R4A

	Channel Bandwidth						
Modula tion	7 MHz	14/1 3,75 MHz	28/27 ,5 MHz	40 MHz	56/55 MHz	80 MHz	112/11 0 MHz
4 QAM	х	х	x ⁽¹⁾	х	x ⁽¹⁾	х	х
8 QAM	х	х	x ⁽¹⁾	х	x ⁽¹⁾	х	х
16 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	х
32 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	х
64 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	х
128 QAM	х	x	x ⁽¹⁾	х	x ⁽¹⁾	x	х
256 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	х
512 QAM	х	x	x ⁽¹⁾	х	x ⁽¹⁾	x	х
1024 QAM	x ⁽²⁾	x	x	x	x ⁽¹⁾	x	x
2048 QAM	_	x ⁽²⁾	x	x	x	x	x
4096 QAM	_	_	x	x	x	x	x

(1) MIMO supported.

(2) XPIC not supported.

MINI-LINK 6363 (ETSI)

Table 59 Supported Modulations for All Frequency Bands (Excluding E-band 80 GHz) with an R-state \geq R4A

Modula			Chan	nel Bandı	width		
tion	7 MHz	14/1 3,75 MHz	28/27 ,5 MHz	40 MHz	56/55 MHz	80 MHz	112/11 0 MHz

4 QAM	х	х	x ⁽¹⁾	х	x ⁽¹⁾	х	х
8 QAM	х	х	x ⁽¹⁾	х	x ⁽¹⁾	х	х
16 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	x
32 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	x
64 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	x
128 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	x
256 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	x
512 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	x
1024 QAM	x ⁽²⁾	x	x	x	x ⁽¹⁾	x	x
2048 QAM	-	x ⁽²⁾	x	x	x	x	x
4096 QAM	_	_	x	x	x	x	x
8192 QAM ⁽²⁾	_	_	x	x	_	_	_

(1) MIMO supported.

(2) XPIC not supported.

ported Modu	lations for	E-band 80	GHz with an	R-state ≥ R1A
	ported Modu	ported Modulations for	ported Modulations for E-band 80	ported Modulations for E-band 80 GHz with an

	Channel Bandwidth		
Modulation	56/62,5 MHz	112/125 MHz	
4 QAM	х	х	
8 QAM	х	х	
16 QAM	x	х	
32 QAM	х	х	
64 QAM	х	х	
128 QAM	х	х	
256 QAM	х	х	
512 QAM	х	х	
1024 QAM	x ⁽¹⁾	x ⁽¹⁾	

2048 QAM	_	_
4096 QAM	-	_

(1) XPIC not supported.

MINI-LINK 6363 and MINI-LINK 6363/2 (ANSI)

Table 61	Supported Modulations for All Frequency Bands (Excluding E-band
	80 GHz) with an R-state < R4A

Modula	Channel Bandwidth								
tion	10 MHz	20 MHz	30 MHz	40 MHz	50 MHz	60 MHz	80 MHz		
4 QAM	х	х	x ⁽¹⁾	x ⁽¹⁾	х	х	х		
8 QAM	х	х	x ⁽¹⁾	x ⁽¹⁾	х	х	х		
16 QAM	x	x	x ⁽¹⁾	x ⁽¹⁾	x	x	x		
32 QAM	x	x	x ⁽¹⁾	x ⁽¹⁾	x	x	x		
64 QAM	x	x	x ⁽¹⁾	x ⁽¹⁾	x	x	x		
128 QAM	x	x	x ⁽¹⁾	x ⁽¹⁾	x	x	x		
256 QAM	x	x	x ⁽¹⁾	x ⁽¹⁾	x	x	x		
512 QAM	x	x	x ⁽¹⁾	x ⁽¹⁾	x	x	x		
1024 QAM	x ⁽²⁾	х	х	х	х	х	х		
2048 QAM	-	x ⁽²⁾	x	x	x	x	x		
4096 QAM	_	_	x	x	x	x	x ⁽²⁾		

(1) MIMO supported.

(2) XPIC not supported.

MINI-LINK 6363 (ANSI)

Table 62 Supported Modulations for All Frequency Bands (Excluding E-band 80 GHz) with an R-state \geq R4A

Modula tion	Channel Bandwidth							
	10 MHz	20 MHz	30 MHz	40 MHz	50 MHz	60 MHz	80 MHz	
4 QAM	х	х	x ⁽¹⁾	x ⁽¹⁾	х	х	х	

8 QAM	х	х	x ⁽¹⁾	x ⁽¹⁾	х	х	х
16 QAM	x	x	x ⁽¹⁾	x ⁽¹⁾	x	х	х
32 QAM	x	x	x ⁽¹⁾	x ⁽¹⁾	x	х	х
64 QAM	x	x	x ⁽¹⁾	x ⁽¹⁾	x	х	х
128 QAM	x	x	x ⁽¹⁾	x ⁽¹⁾	x	х	х
256 QAM	x	х	x ⁽¹⁾	x ⁽¹⁾	х	х	х
512 QAM	x	x	x ⁽¹⁾	x ⁽¹⁾	x	х	х
1024 QAM	x ⁽²⁾	x	x	x	x	х	х
2048 QAM	-	x ⁽²⁾	x	х	x	х	х
4096 QAM	-	-	x	x	x	x	х
8192 QAM ⁽²⁾	-	-	х	х	-	_	_

(1) MIMO supported.

(2) XPIC not supported.

Table 63	Supported Modulations for E-band 80 GHz with an R-state \geq R1A	
----------	--	--

	Channel Bandwidth					
Modulation	60/62,5 MHz	80/125 MHz				
4 QAM	x	х				
8 QAM	x	x				
16 QAM	x	x				
32 QAM	x	x				
64 QAM	x	x				
128 QAM	x	x				
256 QAM	x	x				
512 QAM	x	х				
1024 QAM	x ⁽¹⁾	x ⁽¹⁾				
2048 QAM	_	_				
4096 QAM	_	_				

(1) XPIC not supported.

					-					
Modula	Channel Bandwidth									
tion	7 MHz	14/1 3,75 MHz	28/27 ,5 MHz	40 MHz	56/55 MHz	80 MHz	112/11 0 MHz			
4 QAM	х	х	x ⁽¹⁾	х	x ⁽¹⁾	х	х			
8 QAM	х	х	x ⁽¹⁾	х	x ⁽¹⁾	х	х			
16 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	x			
32 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	x			
64 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	x			
128 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	x			
256 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	х	x			
512 QAM	x	x	x ⁽¹⁾	x	x ⁽¹⁾	x	x			
1024 QAM	x ⁽²⁾	x	x	x	x ⁽¹⁾	x	x			
2048 QAM		x	x	x	x	x	x			
4096 QAM			x	x	x	x	x			
8192 QAM ⁽³⁾			x ⁽²⁾	x ⁽²⁾						

MINI-LINK 6364 Supported Modulations (ETSI)

Table 64 Supported Modulations for All Frequency Bands with an R-state \geq R1A

(1) MIMO supported.

(2) XPIC not supported.

(3) Static modulation is not supported.

MINI-LINK 6365 Supported Modulations (ETSI)

Table 65	Supported Modul	ations for All	Frequency	Bands with	an R-state≥	R1A
----------	-----------------	----------------	-----------	------------	-------------	-----

Modula	Channel Bandwidth							
tion	7 MHz	14/1 3,75 MHz	28/27 ,5 MHz	40 MHz	56/55 MHz	80 MHz	112/11 0 MHz	
4 QAM	x	x	x ⁽¹⁾⁽²⁾	x	x ⁽¹⁾⁽²⁾	x	x ⁽²⁾	

8 QAM	х	х	x ⁽¹⁾⁽²⁾	х	x ⁽¹⁾⁽²⁾	х	x ⁽²⁾
16 QAM	x	x	x ⁽¹⁾⁽²⁾	x	x ⁽¹⁾⁽²⁾	x	x ⁽²⁾
32 QAM	x	x	x ⁽¹⁾⁽²⁾	x	x ⁽¹⁾⁽²⁾	x	x ⁽²⁾
64 QAM	x	x	x ⁽¹⁾⁽²⁾	x	x ⁽¹⁾⁽²⁾	x	x ⁽²⁾
128 QAM	x	x	x ⁽¹⁾⁽²⁾	x	x ⁽¹⁾⁽²⁾	x	x ⁽²⁾
256 QAM	x	x	x ⁽¹⁾⁽²⁾	x	x ⁽¹⁾⁽²⁾	x	x ⁽²⁾
512 QAM	x	x	x ⁽¹⁾⁽²⁾	х	x ⁽¹⁾⁽²⁾	х	x ⁽²⁾
1024 QAM	x ⁽³⁾	x	x ⁽²⁾	x	x ⁽¹⁾⁽²⁾	x	x ⁽²⁾
2048 QAM	-	x ⁽³⁾	x ⁽²⁾	x	x ⁽²⁾	x	х
4096 QAM	-	-	x ⁽²⁾	x	x	x	x
8192 OAM ⁽³⁾	-	_	x	x	-	_	_

(1) MIMO supported.

(2) Carrier Aggregation supported.

(3) XPIC not supported.

(4) Supported frequencies: 7–23 GHz

MINI-LINK 6365 Supported Modulations (ANSI)

Table 66	Supported	Modulations f	or All Frequenc	y Bands with an	R -state \geq R1A
				1	

Modula	Channel Bandwidth							
tion	10 MHz	20 MHz	30 MHz	40 MHz	50 MHz	60 MHz	80 MHz	
4 QAM	х	х	x ⁽¹⁾⁽²⁾	x ⁽¹⁾	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾	
8 QAM	х	х	x ⁽¹⁾⁽²⁾	x ⁽¹⁾⁽²⁾	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾	
16 QAM	х	х	x ⁽¹⁾⁽²⁾	x ⁽¹⁾⁽²⁾	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾	
32 QAM	х	х	x ⁽¹⁾⁽²⁾	x ⁽¹⁾⁽²⁾	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾	
64 QAM	х	x	x ⁽¹⁾⁽²⁾	x ⁽¹⁾⁽²⁾	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾	

128 QAM	x	x	x ⁽¹⁾⁽²⁾	x ⁽¹⁾⁽²⁾	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾
256 QAM	x	x	x ⁽¹⁾⁽²⁾	x ⁽¹⁾⁽²⁾	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾
512 QAM	x	x	x ⁽¹⁾⁽²⁾	x ⁽¹⁾⁽²⁾	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾
1024 QAM	x ⁽³⁾	x	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾
2048 QAM	-	x ⁽³⁾	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾	x ⁽²⁾	x
4096 QAM	-	_	x ⁽²⁾	x	x	x	x
8192 OAM ⁽³⁾	-	_	x	x	-	_	_

(1) MIMO supported.

(2) Carrier Aggregation supported.

(3) XPIC not supported.

(4) Supported frequencies: 8–23 GHz

RAU2 X Supported Modulations (ETSI)

Table 67	Supported Modulations in Single Carrier Mode for RAU2 X R-state \geq
	R6A

Modulatio	Channel Bandwidth					
n	7 MHz	14 MHz	28 MHz	40 MHz	56 MHz	
4 QAM	х	х	х	х	х	
16 QAM	х	х	х	х	х	
32 QAM	х	х	х	х	х	
64 QAM	х	х	х	х	х	
128 QAM	х	х	х	х	х	
256 QAM	х	х	х	х	х	
512 QAM	х	х	х	х	х	
1024 QAM	х	х	х	х	х	
2048 QAM	-	х	х	х	х	
4096 QAM	-	-	х	х	х	

RAU2 X Supported Modulations (ANSI)

Modulati	Channel Bandwidth						
on	10 MHz	20 MHz	30 MHz	40 MHz	50 MHz	60 MHz	
4 QAM	х	х	х	х	x	х	
16 QAM	х	х	х	х	x	х	
32 QAM	х	х	х	х	x	х	
64 QAM	х	х	х	х	x	х	
128 QAM	х	х	х	х	x	х	
256 QAM	х	х	х	х	x	х	
512 QAM	х	х	х	х	x	х	
1024 QAM	x	x	x	x	x	x	
2048 QAM	_	x	x	x	x	x	
4096 QAM	_	_	x	x	x	x	

Table 68 Supported Modulations in Single Carrier Mode for RAU2 X R-state \geq R6A

15.19 Antennas

All antennas are "compact", that is the design is compact with a low profile. The antennas are made of aluminum and painted light gray. All antennas have a standardized waveguide interface. The feed can be adjusted for vertical or horizontal polarization.

All antennas are delivered with a mounting kit.

All high performance antennas have an integrated radome.

There are two sorts of antennas available:

- Modular antennas
- Antennas with fixed radio interface

Modular antennas range from 0.3 m - 1.8 m in diameter. The antenna is divided into two modules: the antenna module and the interface module. The interface module is available as Single polarized or as Dual polarized (OMT), and also with interfaces for ANT0 (separate installation), ANT2 (integrated installation for RAU2 X), and ANT3 (integrated installation for MINI-LINK 6363, MINI-LINK 6364, and MINI-LINK 6365). This makes it possible to upgrade a modular \geq

antenna from single to dual polarization and from one radio type to another, for capacity increase when needed and without having to re-align the antenna.

It is not possible to change the frequency band of the antenna by replacing the interface, since also other parts in the antenna are frequency dependent.

The interface module is included when ordering a modular antenna, and is also possible to order as a separate item.

Antennas with fixed radio interfaces range from 0.3 m - 3.7 m in diameter, in single and dual polarized versions, and are available with ANT0 or ANT2 interface.

15.19.1 MINI-LINK 6363, MINI-LINK 6364 and MINI-LINK 6365 Integrated Installation

For a 2+0 configuration, the MINI-LINK 6363, MINI-LINK 6364, or MINI-LINK 6365 is fitted directly to the rear of the antenna in an integrated installation. Single polarized antennas up to 1.8 m in diameter are normally fitted integrated with the MINI-LINK 6363, MINI-LINK 6364 or MINI-LINK 6365.

For a 1+1 configuration the MINI-LINK 6363, MINI-LINK 6364, or MINI-LINK 6365 can be fitted directly to an Integrated Power Splitter (IPS), see Figure 146.

An asymmetrical power splitter is mainly used for 1+1 hardware protection only. The IPS provides one main channel with low attenuation and one standby channel with higher attenuation.

A symmetrical power splitter is mainly used for 1+1 or 2+0 hardware protection and frequency diversity. The IPS provides equal attenuation in both channels.



Figure 146 MINI-LINK 6363, MINI-LINK 6364, or MINI-LINK 6365 Fitted to an Integrated Power Splitter

The 0.3 m and 0.6 m integrated dual polarized antennas are used with two MINI-LINK 6363s, two MINI-LINK 6364s, or two MINI-LINK 6365s.

In an integrated installation, ANT3 or IPS3 can be used with the MINI-LINK 6363, MINI-LINK 6364, or MINI-LINK 6365.

Using an antenna adapter plate also enables antennas of type ANT2 or IPS2 to be connected to the MINI-LINK 6363, MINI-LINK 6364, or MINI-LINK 6365 radio. The antenna adapter plate is illustrated in Figure 147.



Figure 147 The Antenna Adapter Plate for MINI-LINK 6363, MINI-LINK 6364, or MINI-LINK 6365

15.19.2 RAU2 X Integrated Installation

For a 1+0 configuration, the RAU2 X is fitted directly to the rear of the antenna in integrated installation. Single polarized antennas up to 1.8 m in diameter are normally fitted integrated with the RAU2 X.



Figure 148 0.2 m, 0.3 m, and 0.6 m Compact Antennas Integrated with RAU2 X

For a 1+1 configuration the RAU2 X can be fitted directly to an Integrated Power Splitter (IPS).

An asymmetrical power splitter is mainly used for 1+1 hot standby configurations, that is, hardware protection only. The IPS provides one main channel with low attenuation and one standby channel with higher attenuation.

A symmetrical power splitter is mainly used for 1+1 working standby or 2+0 configurations, that is, hardware protection and frequency diversity. The IPS provides equal attenuation in both channels.


Figure 149 RAU2 X Fitted to an Integrated Power Splitter

The 0.3 m and 0.6 m integrated dual polarized antennas are used with two RAU2s.

In an integrated installation, ANT2 can be used with the RAU2 X, besides the antennas of type ANT2 with fixed radio interface or IPS2.

Using an antenna adapter plate also enables antennas of type ANT1 or IPS1 to be connected to the RAU2 X. The antenna adapter plate is illustrated in Figure 150.



Figure 150 The Antenna Adapter Plate for RAU2 X

15.19.3 MINI-LINK 6363, MINI-LINK 6364, MINI-LINK 6365, and RAU2 X Separate Installation

All antennas have a standardized waveguide interface and can be installed separately, by using a flexible waveguide to connect to the MINI-LINK 6363, MINI-LINK 6364, MINI-LINK 6365, or RAU2 X. The 2.4–3.7 m dual polarized and single polarized antennas are always installed separately.



Figure 151 MINI-LINK 6363, MINI-LINK 6364, or MINI-LINK 6365 Separate Installation in a 1+0 Configuration



Figure 152 RAU2 X Separate Installation in a 1+0 Configuration

16 Management

16.1 DCN

This section covers the Data Communication Network (DCN) functions provided by the NE.

Since both IPv4 and IPv6 (only in DCN VLAN mode) are supported, this would imply using two independent DCN networks, one IPv4 and one IPv6, using

different addressing and different routing. In one case, though, the two can interfere: when a node via IPv4 asks a DNS server for an IP address, the DNS server might respond with an IPv6 address.

16.1.1 IP Services

The following standard external IP network services are supported:

- All clocks, used, for example, for time stamping alarms and events, can be synchronized with a Network Time Protocol (NTP) server. NTP synchronization is supported.
- File Transfer Protocol (FTP) as well as SSH FTP (SFTP) are used as file transfer mechanisms for software upgrade, and for backup and restore of system configuration.
- Domain Name System (DNS) enables the use of host names.
- Dynamic Host Configuration Protocol (DHCP) is used to allocate IP addresses in the DCN. The NE has a DHCP relay agent for serving other equipment on the site LAN.
- Syslog is used to forward log messages in the network and log alarms and events to a central syslog server.

16.1.2 DCN Interfaces

MINI-LINK 6600 provides an IP-based DCN for transport of its O&M data. Each NE has an IP router for handling of the DCN traffic. A number of different alternatives to connect and transport DCN traffic are supported. This diversity of DCN interfaces provides the operator with a variety of options when deploying a DCN.

Each physical interface can support two IP hosts, one for IPv4 and one for IPv6.

The following connection options are supported:

- DCN over VLAN
- DCN over VLAN for L1 connections
- Routed DCN in default VRF
- DCN in L3VPN
- USB
- DCN over PPP
- DCN login to far-end via Radio link

16.1.2.1 DCCr/DCCm

Note: LTU 1002 is needed to provide STM-1 interfaces for DCCr/DCCm.

The DCCr/DCCm overhead sections in the STM-1 frame can be used to transport DCN traffic. A PPP connection is established over the overhead segments between two end points.

The default bandwidth is automatically established to DCCr=192 kbps and DCCm=192 kbps and DCCm is configurable to 384 kbps and 576 kbps.

For LTU 1002, the bandwidth is fixed to DCCr=192 kbps and DCCm=576 kbps.

The PPP connection in the overhead segments is implemented as PPP over bit synchronous HDLC. Any 3rd party equipment that complies with this and the channel bandwidth segmentation can interoperate with MINI-LINK 6600. DCCm can be used to connect MINI-LINK 6600 to MINI-LINK 6600 over an STM-1 connection. For this connection there can be no multiplexer between the two MINI-LINK 6600 NEs.



DCN in SDH



The following channels can be used for DCN transportation in SDH:

- 128 kbps default proprietary channel available on radio side only.
- 192 kbps channel available on line side and radio side by using EOC or DCC bytes of the Regenerator Section Overhead Frame (RSOH) of the SDH frame.

16.1.2.2 10/100BASE-T

Each NE has a 10/100BASE-T Ethernet interface for connection to a site LAN. This interface offers a high speed DCN connection.

16.1.2.3 DCN over VLAN

The management traffic can be transported in a logically separated VLAN together with the Ethernet traffic. An internal switch port in MINI-LINK 6600 forwards the management traffic to the IP DCN router.

In Provider Mode, it is possible to separate DCN over VLAN from the rest of the traffic by configuring a default mapping rule into the C-VID registration table of a port.

16.1.2.4 DCN over VLAN for L1 Connections

The management traffic can be transported in a VLAN over an L1 connection. In MINI-LINK 6600, ports of L1 connection can be added to one VLAN, thus enabling L2 switching for that VLAN in the L2 connection. By the adding LAN-DCN port to the same VLAN, DCN over VLAN can be used over the L1 connection.

16.1.2.5 Routed DCN in Default VRF

High speed routing between multiple DCN VLANs is supported, providing enhanced DCN (compared to DCN over PPP).

The main benefits are:

- Increased bandwidth
- Routing between multiple L3 sub-interfaces
- Better protection of L2 traffic storms.
- Migration can be done remotely without affecting user traffic.

L3 Interfaces

The following L3 interfaces are available for Routed DCN:

L3 VLAN subinterfaces

L3 VLAN subinterfaces can be defined on Ethernet ports and can use VLAN tagging. L2 switching is not available for L3 VLAN subinterfaces.

L3 VLAN interfaces - Bridge Virtual Interfaces (BVIs)

L3 VLAN (BVI) interfaces can be defined on VLANs. L2 switching will operate on the VLAN ports while L3 traffic is handled by the L3 interface. This enables building a L2 DCN with dynamic routing, for example, OSPF. — PPP interfaces

PPP interfaces can be used on LTU1002 boards, to create a DCN link over the STM-1 link.

16.1.2.6 DCN in L3VPN

DCN can also be reached from a user-defined VRF, that is, DCN can be routed not only in the transport infrastructure, but also in the service layer (VPN).

In the default configuration, DCN is attached to the default VRF. By configuration, DCN can be attached to one single user-defined L3VPN VRF. From that point only that VRF can be used for DCN, the default VRF cannot.

Main benefits of the DCN in L3VPN feature are the following:

- Since L3VPN services can pass through the network end-to-end, the DCN in L3VPN feature enables easy DCN provisioning within the network.
- PPP connections, since they can only be used for DCN purposes, are automatically moved to the selected DCN VRF. Host address selection algorithm does not change, be either automatic or manual, but will apply to the interfaces of the selected DCN VRF.

16.1.2.7 USB

The USB interface is used for an O&M connection (MINI-LINK Node GUI, CLI session, or CLI Tool) using an IP address, according to the following:

— IPv4

Local IP address.

— IPv6

Either local (but not link-local) or globally routable IP address.

16.1.3 IP Addressing

The DCN of MINI-LINK 6600 supports both IPv4 and IPv6 (in DCN VLAN mode).

16.1.3.1 IPv4

As for generic IP services, DCN for IPv4 supports numbered IP interfaces, used for the Ethernet interface and IP interfaces configured as ABR. All other IP interfaces should be set up with unnumbered IP interfaces.

The use of unnumbered interfaces has several advantages:

- The use of IP addresses is limited. For a large aggregation site, this would imply a lot of addresses.
- The planning of the IP addresses is simplified.
- The amount of configuration is reduced because only one IP address is configured upon installation.

16.1.3.2 IPv6 (in DCN VLAN Mode)

Interfaces support both globally routable IP addresses and link-local addresses on IPv6. With some exceptions, this corresponds to IPv4 according to Table 69.

	_
IPv4	IPv6
Unnumbered interfaces	Link-local addresses
Numbered interfaces	Globally routable addresses

Table 69Corresponding IPv4 and IPv6 Addressing

16.1.4 IP Router

The IP router supports the following routing mechanisms:

- Open Short Path First (OSPF), which is normally used for routers within the MINI-LINK domain. OSPFv2 (for IPv4) is supported.
- Static routing

There are two different ways to configure the IP router, Command Line Interface (CLI) (see Section 16.3.3 on page 237) and MINI-LINK Node GUI (see Section 16.3.1 on page 236).

Most configurations are done using CLI, and complex router configuration and troubleshooting can only be performed with CLI.

16.1.4.1 OSPF Features

The following summarizes the OSPF features:

- An NE can be a part of a non-stub area, stub area or totally stub area.
- An NE can act as an Internal Router (IR) or an Area Border Router (ABR).
- Virtual links are supported, which is useful when an area needs to be split in two parts.
- Link summarization is supported, which is used in the ABR to minimize the routing information distributed to the backbone and/or other areas.

16.2 Link Layer Discovery Protocol (LLDP)

This section gives a brief overview of the LLDP that is used by network elements (NE) for advertising their identity, capabilities, and neighbors.

LLDP is a simplex protocol where each node periodically transmits information about itself. The protocol is stateless, with no confirmation mechanism. LLDP is specified in IEEE 802.1 AB.

Supported TLVs are as follows:

- ChassisID
- PortID
- Time To Live
- System Name
- Management Address

16.3 Management Tools and Interfaces

This section gives a brief overview of the management tools and interfaces used for MINI-LINK 6600.

16.3.1 MINI-LINK Node GUI

MINI-LINK Node GUI provides tools for on-site installation, configuration management, fault management, performance management and software upgrade. It is also used to configure the traffic routing function, protection and DCN.

MINI-LINK Node GUI is used for local management, that is the NE is accessed locally by connecting a PC to the node with a USB cable.

The NE can also be accessed over the site LAN or remotely over the DCN.

To access an NE remotely through MINI-LINK Node GUI, a MINI-LINK Node GUI license is required on that specific NE to ensure full functionality in MINI-LINK Node GUI.

16.3.2 SNMP

Each NE provides an SNMP agent enabling easy integration with any SNMP-based management system. The SNMP agent can be configured to support SNMPv1/v2c/v3 for get and set operations. SNMPv3 is default. The SNMP agent sends SNMPv1, SNMPv2c and SNMPv3 traps.

The system is built on standard MIBs as well as some private MIBs.

16.3.3 Command Line Interfaces

A CLI is provided for advanced IP router configuration and troubleshooting. This interface is similar to Cisco's industry standard router configuration and is accessed from a Command Prompt window using Telnet or SSH.

16.3.4 Syslog

Logging of alarms and events to syslog servers can be managed through MINI-LINK Node GUI, and CLI.

16.3.5 NETCONF

NETCONF protocol enabling integration with NETCONF-based management systems.

For more information, see NETCONF Interface.

16.4 Configuration Handling

The configuration can be managed locally and from the O&M center provided that the DCN is set up. The following list gives examples of configuration areas:

- Transmission interface parameters
- Traffic routing
- Radio Link Bonding (RLB)
- DCN parameters, such as host name, IP address
- Security parameters, such as enabling Telnet and SSH, enabling SNMPv3 with encryption, and using RADIUS and TACACS+ for AAA
- Radio Terminal parameters, such as frequency, output power, ATPC, and protection

16.4.1 Configuration File

The configuration file is stored non-volatile both on the RMM and in a flash memory on the NPU.

If an NPU needs to be replaced, the RMM from the faulty NPU can be inserted in the new NPU. In this scenario the user copies the configuration file from the RMM to the flash memory of the new NPU. If an RMM is replaced, for example due to a license upgrade or downgrade, the user copies the configuration file from the flash memory to the new RMM.

16.4.2 Aligning the Node with the Network Management System

When the configuration of a node is changed, either through MINI-LINK Node GUI, or CLI, the node sends an event to the Network Management System (NMS).

It is possible to set a timer to decide how long after the configuration change that the event is sent to the NMS. If another configuration change is made before the event has been sent, the timer starts counting from zero again.

If the name of a node is changed, the node immediately sends a separate event to the NMS. Since a name change is also a configuration change, this triggers a configuration change event to be sent as well, but only after the timer value has been reached.

16.5 Software Management

The NE system software consists of different software modules for different applications. Software can be upgraded both locally and remotely. Software upgrade uses a local or remote FTP or SFTP server, to distribute the software to the NE. Using SFTP instead of FTP ensures that the entire session, including passwords, is encrypted.

All traffic continues while the software is being loaded. During the execution of the software download a progress indication is provided in the user interface. When the download is completed, the new software and the previous software versions are stored on the unit.

The downloaded software file is automatically checked for corruption with keyed-Hash Message Authentication Code (HMAC) and the Secure Hash Algorithm 1 (SHA-1), both to avoid erroneous transfer and to mitigate malware attacks.

The file integrity check monitors changes to the configuration file, password file, audit log, and Operating System (OS) files. Alarm notifications are generated and information about the captured changes is logged in a report. The operator can then investigate the report and take proper actions. The operator can view the report on the node or upload the report to an external secure server through SFTP.

Performing a restart of the NE activates the new software version. A warm restart only affects the control system. This restart can be performed immediately or scheduled at a later time. The restart, depending on the new functionality, may influence the traffic. When the restart with the new software is completed, the NE will wait for a "Commit" command from the management system. When the "Commit" command is received, the software upgrade process is completed.

The previous software revision remains stored on the unit in case a rollback is required. This may be the case if something goes wrong during the software



upgrade or if no "Commit" command is received within 15 minutes after the restart.

If plug-in units with old software versions are inserted into the NE, they can be automatically upgraded.

16.5.1 Uploading Local Logs to a Remote FTP Server

The Local Log Upload feature secures the local log files generated on the node by persistently uploading them to a central FTP or SFTP server and making them available for debugging even if the logs on the node are lost due to overwrite, restart, power loss, or any other reasons.

If the feature is enabled, the local log files are automatically uploaded to the server when they are rotated and compressed.

In addition, forced upload can also be initiated manually, where all pending log files are uploaded to the FTP or SFTP server without a delay.

16.6 License Handling

The MINI-LINK 6600 features are divided into three types:

- Base Package
- Value Packages
- Capacity Licenses

The Base Package provides a fully functional Network Element (NE) and is a part of the standard product offering. Value Packages and Capacity Licenses, when enabled, enhance the functionality and the capacity of the NE.

The basic features are licensed through the Base Package. The Base Package is mandatory and is valid for the full MINI-LINK 6600 release.

License-controlled features have two licensing modes:

- Locked Mode
- Unlocked Mode

Under certain circumstances, the system or the user is entitled to an unlock period, during which license controlled features can be used without sufficient license.

Alarms are raised if a license-controlled feature is used without sufficient license on the NE.

License handling allows the user to track the license usage on slot/port level, getting detailed information on the state of license controlled features. The user can:

- Identify which port(s)/slot(s) has a missing license when the node license status is degraded.
- Get detailed information about license allocation on slot/port level.

16.7 Fault Management

All software and hardware in operation is monitored by the control system. The control system locates and maps faults down to the correct replaceable hardware unit. Faults that cannot be mapped to one replaceable unit result in a fault indication of all suspect units (this may be the whole NE).

Hardware errors are indicated with a red LED found on each plug-in unit and radio.

The control system will generally try to repair software faults by performing warm restarts on a given plug-in unit or on the whole NE.

16.7.1 Alarm Handling

The NE uses SNMP traps to report alarms to SNMP-based management system. Alarms can also be sent to a syslog server. To enable a management system to synchronize alarm status, there is a notification log (alarm history log) where all traps are recorded. There is also a list of current active alarms. Both these can be accessed by the management system using SNMP, or MINI-LINK Node GUI. The alarm status of specific managed objects can also be read.

In general, alarms are correlated to prevent alarm flooding. This is especially important for high capacity links, where a defect on the physical layer can result in many alarms at higher layer interface, like E1. Correlation will cause physical defects to suppress alarms, like AIS, at these higher layers.

Alarm notifications can be enabled/disabled for an entire NE, for an individual plug-in unit, and for individual interfaces. Disabling alarm notification means that no new alarms or event notifications are sent to the management systems.

Alarm and event notifications are sent as SNMPv2c/v3 traps with a format according to Ericsson's Alarm IRP SNMP solution set version 1.2. The following fields are included in such a notification:

- Notification identifier: uniquely identifies each notification.
- Alarm identifier: only applicable for alarms, identifies all alarm notifications that relate to the same alarm.
- Managed object class: identifies the type of the source, for example, E1, DS1, and so on.

- Managed object instance: identifies the instance of the source, for example 1/4/6A for an E1 on the NPU.
- Event time: time when alarm/event was generated.
- Event type: X.73x compliant alarm/event type like communications alarm and equipment alarm.
- Probable cause: M.3100 and X.733 compliant probable cause, for example, Loss Of Signal (LOS).
- Perceived severity: X.733 compliant severity, for example, critical or warning.
- Specific problem: free text string detailing the probable cause.

The system can also be configured to send SNMPv1 traps. These traps are translated from the IRP format using co-existence rules for v1 and v2/v3 traps (RFC 2576).

Alarm and event notifications can also be sent to (up to 3) syslog servers in the network. The information content is the same as for the SNMP traps. The messages use a fixed syslog facility of LOG_LOCAL6 and severity mapping and message text is based on RFC 5674 – Alarms in syslog.

16.7.2 Ethernet Link OAM

Ethernet Link OAM supports fault management on Ethernet links according to IEEE 802.3ah and provides link monitoring, fault notification, and loopback test.

16.7.3 Ethernet Service OAM

Ethernet Service OAM supports fault management on Ethernet links according to IEEE 802.1ag and is used to manage networks comprising of multiple LANs.

16.7.4 Loops

Loops can be used to verify that the transmission system is working properly or they can be used to locate a faulty unit or interface.

Note: On MINI-LINK 6651/1 and on MINI-LINK 6651/3, only IF and RF loops on the radio is supported.

16.7.4.1 Loops for Circuit Switched Traffic

The following loops are available on units with an E1/DS1 line interface.

3

Connection Loop	This loop can be initiated for an E1/DS1. The traffic connection is looped in the TDM bus back to its origin, see Figure 154. If an E1/DS1 interface is traffic routed an AIS is sent to the other interface in the traffic routing.
	A Connection Loop can be used in combination with a BERT in another NE to test a network connection including the termination plug-in unit, in case a Local Loop cannot be used due to the lack of a traffic routing.
Line Loop	Loops an incoming line signal back to its origin. The loop is done in the plug-in unit, close to the line interface, see Figure 154. An AIS is sent to the TDM bus.
	A Line Loop in combination with a BERT in an adjacent NE is used to test the transmission link between the two NEs.
Local Loop	Loops a line signal received from the TDM bus back to its origin, see Figure 154. An AIS is sent to the line interface.
	A Local Loop in combination with a BERT in another NE can be used to test a connection as far as possible in the looped NE.
The following loop is only supported on the modem/MMU.	
Rx Loop	This loop is similar to the Connection Loop but the loop is done in the plua-in unit close to the TDM bus, where a group

of E1s in the traffic connection is looped back to its origin, Figure 154. Traffic on high-speed bus is not affected. An Rx Loop can be used on the far-end modem/MMU to verify the communication over the radio path, see Figure 155.



Figure 154 Loops

The following loops on the near-end Radio Terminal are supported in order to find out if the modem/MMU or radio is faulty.

- IF Loop In the modem/MMU the traffic signal to be transmitted is, after being modulated, mixed with the frequency of a local oscillator and looped back for demodulation (on the receiving side). See Figure 155.
- **RF Loop** In the radio, a fraction of the RF signal transmitted is shifted in frequency and looped back to the receiving side. See Figure 155.



Figure 155 Radio Link Loops

16.7.4.2 Loops Impact for Packet Switched Traffic

Note: The Ethernet protocol does not have a mechanism that automatically discards frames when a loop occurs. Without a loop prevention mechanism, all traffic is returned back to the port causing problems with MAC address switching tables and possibly affecting the traffic in the entire network.

Loops for Ethernet traffic are considered to be very harmful and should be avoided.

For IF and RF loops, the looped traffic is discarded to prevent Ethernet loops. Thus, IF and RF loops can be set with Ethernet traffic running.

Loops on physical layers are only employed after disconnecting Ethernet services in order to avoid Ethernet loops. During installation and validation of a hop on radio level, it is recommended to perform measurements by installing loops on (temporary) PDH connection and test the point-to-point connection by configuring an Rx loop. After the hop has been installed and validated the (temporary) PDH connection is removed and the Ethernet services reconnected.

Spanning Tree Protocol (STP) or Rapid STP (RSTP) is not able to prevent looped Ethernet traffic flooding into the network for physical loops on the LAN ports or the WAN ports since own BPDU packets are discarded and therefore the looped traffic is passed through.

Test loops installed on physical and logical interfaces could influence Ethernet traffic as follows:

Physical loop Traffic can be looped with an external physical Tx-Rx loop. MINI-LINK 6600 does not have any built-in mechanism to prevent loops on LAN ports. All traffic will be returned back to the network.

The following loop is part of the Ethernet Link OAM functionality:

RemoteLoops Ethernet traffic between two adjacent NEs, connected
through a LAN interface, and is used for fault localization and
link performance testing of Ethernet links. It is available on
Ethernet traffic units with support for Ethernet Link OAM.

Remote loopback can only be performed on LAN interfaces.

16.7.4.3 MAC Swap Loopback

The source and destination MAC addresses in frames with a specified VLAN ID are swapped at ingress and the frame is looped back. MAC swap loopback can be used for troubleshooting and performance measurement.

The node can be configured to loopback specified L2 frames based on the following configurable parameters:

- Ingress port (LAN/WAN)
- Egress port (LAN/WAN)
- VLAN
- Src MAC address
- Dst MAC address

The src and dst MAC addresses of the specified frames are swapped in linerate (GE/10G).

When Destination MAC address filtering is disabled, the frames with any destination MAC address are looped back and their source MAC address is set to the own MAC address of the local switch.

In case of egress-lookup configuration, the traffic on the ingress port is checked for frames to be egressed on the egress port configured with MAC Swap Loopback. The frames to be egressed are looped back on the ingress port.

Because the loopback always takes place on the ingress port, latency time measurements are the same regradless of whether the latency time is measured for the egress or ingress port.

The following prerequisites must be fulfilled to be able to use the egress-lookup configuration:

- The dst MAC address must already be learnt by the switch to be able to identify the egress port
- The configured egress port Oper Status must be Up
- The configured egress port must be a member of the configured VLAN

Only one MAC swap loopback session is supported for the node.

Loop is automatically deactivated after the configurable timeout. The default timeout value is 300 seconds.

16.8 Spectrum Diagnostics Scan for Potential Interferer

The Spectrum Diagnostics scan used when verifying an installation and can then be activated by a user to automatically scan the frequency range of the configured radio channel or the complete radio sub-band. The test results in Received Signal Strength Indicator (RSSI) values, which are used to verify that there are no interfering signals. If the result indicates the presence of an interferer, further investigation is needed to conclude if this has impact on the radio-link performance or not.

The Spectrum Diagnostics scan can be performed using either CLI commands or MINI-LINK Node GUI.

Note: Running a spectrum diagnostics scan on a hop that is already taken into service impacts the traffic and can also lead to lost connection with the node as the transmitters are turned off during the scan.

16.9 Performance Measurements

Performance counters are available for 5-minute, 15-minute, and 24-hour intervals. The start time of a 24-hour interval is configurable.

The following counters are stored in the NE:

- (only for Ethernet PM counters) current 5 minutes and the previous 12×5 minutes,
- current 15 minutes and the previous 96×15 minutes,
- current 24 hours and the previous 24 hours.

Performance data is stored in a volatile memory. Therefore, all gathered data is lost at a restart.

Performance data is available for the following major features:

- Radio link (see Section 16.9.2 on page 247)
- Ethernet traffic (see Section 16.9.3 on page 248)
- PDH (see Section 16.9.4 on page 249)
- Bit Error Testing (see Section 16.9.5 on page 249)
- Ethernet Service OAM (see Section 16.9.6 on page 250)
- PTP Performance (see Section 16.9.8 on page 251)
- Power measurement PM data (see Section 16.9.9 on page 252)



All historical performance measurement data is available in form of XML files through the Bulk PM feature, see Section 16.9.1 on page 247.

16.9.1 Bulk PM

The Bulk Performance Management (PM) feature can replace the Ethernet PM or SOAM PM solutions. The difference between Bulk PM and the other PM solutions is that Bulk PM collects all counters from a node into a single XML file instead of using separate XML files.

The Bulk PM solution does not require a separate request for each counter through an SNMP interface; instead, it uses a single SNMP request to ask for all PM data of a single interval. Once it receives the requested PM data, the Bulk PM feature uploads a file to the specified FTP or SFTP server.

For more information on Bulk PM and for the configuration methods, see Bulk Performance Measurement.

16.9.2 Radio Link Performance Measurements

Radio Link PM makes it possible to compare actual performance with planned performance. The configuration of each microwave link is normally based on radio link planning criteria and RAN requirements (quality, delay, and availability) with the help of planning tools.

RF Input and Output Power

RF Input power can be used to identify underperforming radio links in terms of received signal levels, which could indicate a problem caused by, for example, bad RF planning, interference, incorrect configuration, or hardware faults.

RF Output power can be monitored when Automatic Transmit Power Control (ATPC) is activated.

Threshold alarms can be defined for Input and Output Power.

Radio Link Quality

G.826 is originally a standard for measurement of the quality of PDH links. In MINI-LINK nodes this measurement is also available on the continuously running serial data stream below any link layer such as Ethernet.

The G.826 PM, such as ES, SES and BBE, give an indication of the quality of the radio link.

The channel conditions are continuously monitored on the Rx side by measurement of Signal to Noise and Interference Ratio (SNIR).

Adaptive Coding and Modulation

When Adaptive Coding and Modulation (ACM) is configured counters are available that measure how much time that is spent in each modulation.

See also Section 3.2 on page 12.

16.9.2.1 Radio Cable Monitoring

If the communication between the modem and the radio is disturbed, it can cause intermittent traffic loss and result in the transmitter on the radio getting switched off.

By checking for any alarms on the RAU-MMU communication channel, the connection between the modem and the radio can be monitored and problems with cables between the modem and the radio can be detected.

16.9.3 Ethernet PM Counters

The Performance Monitoring function is used by the network operator to analyze the performance and usage of the Ethernet traffic to find unused capacity, congestion points and traffic lost due to congestion or faults. The analysis result is used to re-plan the network for better usage. The Performance Monitoring is also used when troubleshooting and performing functional tests of the equipment.

For Bandwidth and Traffic Performance measurements, the Ethernet PM counters are divided into the following intervals: 5-minute current PM, 15-minute current PM, 24-hour current PM, 5-minute history PM (12 intervals), 15-minute history PM (96 intervals), and 24-hour history PM. When the elapsed time reaches 5 minutes, 15 minutes, or 24 hours respectively, the data is moved into the history intervals and the oldest history data is deleted.

Note: For measurements made on VLAN basis, 5-minute intervals are not supported.

Bandwidth Counters

Bandwidth measurements are done on LAN and WAN interfaces in Tx and Rx direction.

WAN interfaces support the measurement on separate TC queues in Tx direction.

Bandwidth measurements (except for TC queue measurements) are also available on a per interface per VLAN basis. Bandwidth measurements for VLANs are disabled by default. For how to enable them, see Configuring an Ethernet Layer 2 Connection.

Average, maximum, and minimum bandwidth, as well as bandwidth histograms are available.

Traffic Performance Counters

Traffic performance measurements are done on LAN and WAN interfaces in Tx and Rx direction.

Some of the traffic performance counters in Tx direction are also available for separate Traffic Class queues on WAN interfaces.

Threshold Crossing Alarms for Ethernet PM Counters

It is possible to configure threshold crossing alarms for Ethernet PM counters on used interfaces. The thresholds are configured per interface, where ordinary LAN interfaces have a subset of the thresholds available for a buffered LAN or a WAN interface. The availability of thresholds are based on which counters are provided by the different interfaces.

XML File

The Ethernet performance data can be transferred to MINI-LINK Node GUI or to an external system, such as Ericsson Network Manager (ENM), in XML files. Upon a file transfer request the NE creates an XML file with the requested data, compresses the file and sends it to the selected FTP or SFTP Server.

The content of the file is based on data available in the NE at the time of the request.

In MINI-LINK Node GUI, it is possible to display historical Ethernet performance data without having an FTP Server installed on the host where MINI-LINK Node GUI is running.

16.9.4 PDH

The following performance counters are used for the E1/DS1 line interfaces:

- Errored Seconds (ES)
- Severely Errored Seconds (SES)
- Background Block Error (BBE) (only structured interfaces)
- Unavailable Seconds (UAS)

16.9.5 Bit Error Testing

Each NE has a built-in Bit Error Ratio Tester (BERT) in all plug-in units carrying traffic. The BERT is used for measuring performance on E1/DS1 interfaces according to ITU standard 0.151. A Pseudo Random Bit Sequence (PRBS) with a test pattern 2¹⁵–1 is sent through the selected interface.

2



As with loop tests, bit error testing may be used for system verification or for fault location.

Figure 156 BERT in Combination with an External Loop

The BERT is started and stopped by the operator and the bit error rate as a function of the elapsed time is the test result. The test can be started and stopped locally or remotely using the management system.

Several BERTs can be executed concurrently, but only one BERT per plug-in unit.

16.9.6 Ethernet Service OAM

Ethernet Service OAM provides performance monitoring between MEPs, according to ITU-T Y.1731 and Y.1730. Ethernet Service OAM provides delay and delay variation information to the service provider, and ensures that the provided service is in accordance to the SLA.

Ethernet Service OAM supports the following PM functions:

- Frame Delay
- Frame Delay Variation Measurement
- Frame Loss Measurement
- XML File transfer
- Ethernet Bandwidth Notification

16.9.7 IP Performance Monitoring with TWAMP Reflector Light

Two-Way Active Measurement Protocol (TWAMP) is a Layer 3 protocol. It provides a flexible method for measuring two-way or round-trip IP performance between two nodes (sender and reflector) in an IP network.

TWAMP is specified in RFC 5357.

TWAMP can be used to measure two-way metrics like inter-packet delay (latency) and delay variation (jitter) between two endpoints, packet loss, far-end delay, far-end jitter, near-end delay, near-end jitter, duplicate packets, and so on, using UDP packets. The supported UDP port range is 863-65535 and the default UDP port number is 862.

TWAMP consists of two major parts; a control plane part and a data plane part. The control part sets up the TWAMP session by using TCP.

The test part is the an UDP based measurement. The sender transmits a test message and that is reflected from the reflector. Both endpoints timestamp the message.



The reflector never generates packets by itself, it only sends answers to packets. The reflector has a separate, configurable IP address. There is a special type of implementation for TWAMP with lower complexity that is called TWAMP Light

where the control plane is removed and only the test messages are exchanged.

16.9.8 PTP Performance Monitoring

PTP Performance Counters

When enabled, MINI-LINK can collect a set of performance data in a PTP network. The user can enable PTP performance monitoring (and, when needed, clear its counters) using MINI-LINK Node GUI or the CLI. \geq

PTP Performance Data Types

PTP performance monitoring collects the following types of data:

- ClockPerformanceMonitoringData measures data related to path delay on clock level. This data type indicates timestamping quality.
- PortPerformanceMonitoringData measures data related to messaging. This data type indicates network performance.

XML File

The performance data can be transferred to MINI-LINK Node GUI or to an external system, such as Ericsson Network Manager (ENM). Upon a file transfer request, the NE creates an XML file with the requested data, compresses the file, and sends it to the selected FTP or SFTP Server.

The content of the file is based on data available in the NE at the time of the request.

In MINI-LINK Node GUI, it is possible to display historical performance data without having an FTP Server installed on the host where MINI-LINK Node GUI is running.

16.9.9 Power Measurement PM

Power PM Counters

When enabled, MINI-LINK collects historical values of input voltage, power consumption, and consumed energy in 15-minute and 24-hour intervals. This feature is only available on MINI-LINK 6691, 6692, 6693, and 6694 nodes.

XML File

The Power PM data can be transferred to MINI-LINK Node GUI or to an external system, such as Ericsson Network Manager (ENM). During a file transfer request, the NE creates an XML file with the requested data, compresses the file, and sends it to the selected FTP or SFTP Server.

The content of the file is based on data available in the NE at the time of the request.

In MINI-LINK Node GUI, it is possible to display historical performance data without having an FTP Server installed on the host where MINI-LINK Node GUI is running.

16.10 Node Security

All management access to the NE is protected by a user name and a password.

The following user types are defined:

- view_user with read-only access
- oper_user with read and write access
- control_user with read and write access
- admin_user with read and write access
- **Note:** Only the admin_user has full read and write access, the oper_user and control_user have full read access, but limited write access.
 - Only the admin_user can change the system time.

All user types have an associated password. Passwords can be changed using MINI-LINK Node GUI, CLI, or the SNMPv3 interface. Each local user can change their own password. Only an administrator can change password for other local users.

If a local user has the default password when they log in through the CLI or Node GUI, then the system forces them to change this default password. This applies to default passwords after reverting the node to factory settings as well as one-time passwords given by an administrator to a local user.

Note: Strong password feature is available and is enabled by default on the MINI-LINK Node GUI and CLI interfaces. Strong passwords require minimum 12 characters in length, and at least three of the following character types: lowercase, uppercase, numeral, non-alphanumeric characters.

The following security mechanisms are used on the various O&M interfaces:

MINI-LINK Node GUI

Local and remote MINI-LINK Node GUI access requires a user name and password.

Note: The use of default password at USB connection can be allowed or disallowed by the user.

It is possible to add a legal notice shown before logging on, to highlight the legal consequences of unauthorized access.

— SNMP

For SNMPv3 access, the regular user name and password protection is used. In addition to this the User-based Security Model (USM) and View-based Access Model (VACM) are supported. This means that additional users and passwords might be defined by external SNMPv3 managers.

The default security level is authentication/encrypted privacy where SHA-1 is used for authentication. Message Digest 5 (MD5) or SHA-2 can also be

used for authentication. SNMPv3 can be configured to authentication with no privacy, using Data Encryption Standard (DES) or Advanced Encryption Standard (AES). An encrypted setting means that the management traffic is secure.

For SNMPv1/v2c access, the regular user name and password protection does not apply. Instead a community-based access protection is used. As default, a public and a private community are configured. The public community enables default read-access and the private community provides read and write access to MIB-II system information. These privileges can be extended through either MINI-LINK Node GUI, or SNMPv3 interface. The SNMPv1/v2c interface may be disabled.

— CLI

Access to the Telnet port using CLI commands is protected by the regular user name and password protection. The Telnet port can be disabled from MINI-LINK Node GUI.

Secure Shell (SSH) protocol can be used for more secure remote access and use of CLI commands. The SSH protocol is enabled as an alternative to the Telnet protocol, using MINI-LINK Node GUI or. When SSH is enabled telnet is automatically disabled and for security reasons can only be re-enabled if SSH is disabled.

It is possible to regenerate the automatically generated public-private key pairs in case the SSH keys have been compromised.

16.10.1 Authentication, Authorization, and Accounting

Authentication, Authorization, and Accounting (AAA) is a security architecture for distributed systems. The Authentication process makes sure that only accepted users can log on to the system, for example, using user names and passwords. The Authorization process gives authenticated users certain permissions, for example, based user roles. The Accounting process records information about access and use of the system.

There are three AAA policies in MINI-LINK: local, RADIUS, and TACACS+. RADIUS and TACACS+ are two types of centralized authentication.

Note: If the connection to the remote AAA server is interrupted, the NE falls back to local authentication. When the communication to the AAA server is working again, the centralized authentication (RADIUS or TACACS+) is automatically restored.

MINI-LINK 6600 devices support RADIUS or TACACS+ session accounting for all central and local users. RADIUS or TACACS+ servers for accounting are handled together with the authentication servers. The management sessions are logged by enabling the session accounting feature. When the management session is started, that is, the management user has logged on, the NE sends a RADIUS or TACACS+ accounting start message to the server. When the session is ended, that

is, the management user has logged off, the NE sends a RADIUS or TACACS+ accounting stop message to the server. Both the beginning and the end of a session can be recorded. Accounting messages contain the IP address of the client from which the NE is accessed by the user.

16.10.1.1 Local Authentication

The local policy supports the following features:

— Authentication

For local user authentication, it is necessary to supply authentication information in the form of a user name and password. During the authentication process, the NE searches its locally stored configuration for a user with a matching user name. If a matching user name cannot be found, the request is refused. The maximum number of local users supported is four.

- **Note:** When logging in through the CLI or Node GUI, if a local user has the default password, then the user is forced to change that default password.
- Authorization

The NE uses local authorization information to distinguish which privileges belong to a role.

The authorization process is based on the user role (system admin, network admin, operator, or guest).

16.10.1.2 Centralized Authentication

The two types of centralized authentication are RADIUS and TACACS+.

The system has two external interfaces, CLI and SNMPv3. Centralized users and authentication are fully supported for CLI and partially supported for SNMPv3. The local admin_user needs to set the AAA policy and configure the TACACS+ or RADIUS servers before attempting a centralized authentication.

The NE supports the User-based Security Model (USM) for SNMPv3. USM requires that the node knows the user before starting a SNMPv3 communication. Local users are always known by the NE but centralized users have to authenticate themselves on CLI first.

After a successful authentication, a user session is created and a SNMPv3 communication with the NE can start. The CLI connection can be terminated after the NE receives the first SNMPv3 message. If the CLI connection is terminated, the SNMPv3 messages keep the user session running.

SNMPv3 security protocols (authentication and privacy) are configurable by RADIUS/TACACS+ authorization parameters.

If centralized authentication is used, all local users (admin_user, view_user, control_user, and oper_user) are enabled. Passwords for centrally-authenticated user accounts cannot be changed using CLI, or MINI-LINK Node GUI.

It is possible to disable local users (admin_user, view_user, control_user, and oper_user) when TACACS+ is configured.

Note:

- Only admin_user can disable local users when centralized authentication is configured.
- If the connection to the remote AAA server is interrupted and the NE falls back to local authentication, then local users temporarily become available even if they were disabled.

It is recommended to set SSH as a secure CLI connection, to allow a fully-secure centralized authentication and authorization process.

A session timeout and a user message display, for example, welcome messages, legal banners, and so on, are supported. The timer value and the user messages must be properly configured in the external TACACS+ or RADIUS server. The user message is a text intended to be presented to the user when they connect to the node. The session timeout is the maximum time the user is allowed to stay connected. The audit logging functionality keeps track of the individual user name that has connected to the node.

RADIUS

The RADIUS protocol, which is based on a client-server model, enables remote access to networks and network services. When configured with the IP or host name of a RADIUS server, the NE can act as a RADIUS client. The format and validation of RADIUS packets is in accordance with the IETF protocol specification RFC 2865.

Both IPv4 and IPv6 are supported.

RADIUS uses UDP, which offers best-effort delivery.

RADIUS only encrypts the password in the Access-Request packet from the client to the server. The rest of the packet (for example, user name, authorized services, and accounting) is not encrypted.

RADIUS supports the following features:

Authentication

The NE supports centralized authentication using RADIUS. A user needs to have an account created on the external server before logging on. Once the account is created, the system can be configured to receive centralized authentication using RADIUS. For centralized authentication, all Local Users are enabled. The NE supports fetching the user roles through a RADIUS server. The NE uses local authorization information to distinguish which privileges belong to a role.

The authorization process is based on the user role (system admin, network admin, operator, or guest). For authorization using RADIUS, the RADIUS server provides the user role when the user logs on to the NE. The NE can be managed in situations when a RADIUS server is unreachable. Therefore it ensures there is always at least one locally-authenticated system administrator account.

The default is that all Local Users are enabled. The user must manually set them to Disabled, after choosing the centralized mode. If they are disabled, then they will be automatically re-enabled in case all RADIUS servers are down.

Accounting

The NE supports session accounting using RADIUS or TACACS+. The management sessions can be logged by enabling either the RADIUS or the TACACS+ accounting feature. When the session is started, that is, the management user is logged on, the NE sends an accounting start message to the RADIUS or TACACS+ server. When the session is ended, that is, the management user has logged off, the NE sends an accounting stop message to the RADIUS or TACACS+ server.

RADIUS Server-Client Feature

The NE supports up to six RADIUS servers. It connects to the servers one-by-one according to their priorities. If no server is reachable, the NE enables local authentication automatically. When the communication to any RADIUS server is working again, the centralized authentication is automatically restored.

The NE supports three RADIUS packet types: Access-Request, Access-Accept, and Access-Reject.

A RADIUS Access-Request message containing the authentication information is sent to a remote server. When the RADIUS server receives the request, it validates the client using a "shared secret". If the client is valid, the RADIUS server consults its user database to validate the access. The server responds to an Access-Request message with either an Access-Reject message or an Access-Accept message. On receipt of an Access-Reject message, the client refuses access to the user. On receipt of an Access-Accept message, the client grants access to the user.

If the NE does not receive a RADIUS response to an Access-Request message within the configured timeout, it keeps retransmitting the request until it receives a response, or until the configured number of maximum transmissions has been reached.

TACACS+

The TACACS+ protocol enables the building of a system that secures remote access to networks and network services. TACACS+ is based on a client/server architecture. The TACACS+ servers are configured on a per-context basis, with a limit of six servers.

Both IPv4 and IPv6 are supported.

TACACS+ uses the Authentication, Authorization, and Accounting (AAA) architecture. This allows separate authentication solutions that can still use TACACS+ for authorization and accounting.

TACACS+ uses TCP, which offers connection-oriented transport.

TACACS+ encrypts the entire body of the packet.

TACACS+ supports the following features:

- Authentication

The NE supports centralized authentication using TACACS+. A user needs to have an account created on the external server before logging on. Once the account is created, the system can be configured to receive centralized authentication using TACACS+.

Authorization

The NE supports fetching the user roles through a TACACS+ server, and the NE uses local authorization information to distinguish which privileges belong to a role.

The authorization process is based on the user role (system admin, network admin, operator, or guest). For authorization using TACACS+, the TACACS+ server provides the user role when the user logs on to the NE. The NE can be managed in situations when a TACACS+ server is unreachable. Therefore it ensures there is always at least one locally-authenticated system administrator account. When the communication to any TACACS+ server is working again, the centralized authentication is automatically restored.

The default is that all Local Users are enabled. The user must manually set them to Disabled, after choosing the centralized mode. If they are disabled, then they will be automatically re-enabled in case all TACACS+ servers are down.

— Accounting

The NE supports session accounting using RADIUS or TACACS+. The management sessions can be logged by enabling either the RADIUS or the TACACS+ accounting feature. When the session is started, that is, the management user is logged on, the NE sends an accounting start message to the RADIUS or TACACS+ server. When the session is ended, that is, the



The NE supports command accounting using TACACS+. User commands can be logged by enabling the TACACS+ command accounting feature. When the session is open, that is, the management user is logged on, the NE sends command accounting messages to the TACACS+ server.

16.10.2 NTP Authentication

To prevent manipulation of the time signal, MINI-LINK 6600 authenticates that the system is synchronized with the Network Time Protocol (NTP) server. An event is generated if the connection to the NTP server is lost.

The user is able to do the following:

- Enable/Disable authentication
- Set the external FTP server to download the Key File

The Key File is a file with a number of cryptographic NTP keys that can be downloaded via SFTP from an external server.

— Download the Key File

16.10.3 SFTP

An SFTP server can be used to upgrade NE system software, instead of just using an FTP server. Using SFTP instead of FTP ensures that the entire session, including passwords, is encrypted. When adding a new FTP or SFTP server, it is possible to reserve that server for security tasks, so that only the admin_user is allowed to modify the server. Other users, for example control_user, can add the same server and use it for purposes that are not security related.

For more information about software management, see Section 16.5 on page 238

16.10.4 File Integrity Violation

File integrity violation raises an alarm if the internal files of the node are modified. Each node has a file integrity check available, that enhances protection against viruses and Trojan Horses. The file integrity check monitors for changes to files and directories in the system. It uses some cryptographic means that basically protect the configuration files it uses for that scope. When a change is captured, then a report is automatically generated and the user can check the reason for those changes, and then investigate and take proper actions. The user can view the report or upload the report to an external secure server through SFTP. An alarm is raised in case of integrity violations.

16.10.5 Security Audit Logging

The NE has an audit log that traces user-initiated events in the node back to the responsible user/operator. The audit log is a text file comprising the following information:

- Sequence number
- Timestamp
- User name
- IP address of the host associated with the event
- Operation including values (when applicable)

The alarm is raised when the audit log transfer to the externally-configured FTP or SFTP server is unsuccessful.

The audit log is not affected if the NE is restarted and it cannot be changed in MINI-LINK Node GUI. The audit log can be transferred to an FTP server on a regular basis.

16.10.6 Monitoring Local Users and Active Sessions

It is possible to view and download reports about local users and active sessions from MINI-LINK Node GUI.

Note: For viewing and downloading reports, admin_user is required.

About local users generated by the node, information on the previous session (start date and time) can be monitored.

About active sessions, the following parameters can be monitored:

- Information on the user
- IP address
- Session ID
- Interface used (CLI, NETCONF, SNMP, or MINI-LINK Node GUI)
- Start time of the active session

16.10.7 Syslog for Security Events

It is possible to send Security events to an external Syslog server in syslog format to trace security events and user actions.

The selected existing security events are sent in Syslog format, independently of the audit log file. "Event" means "setting" in this respect. For example, the protocol setting Telnet/SSH is considered a security event, and the change between Telnet/SSH will be notified to the external Syslog server once it is configured to do so.

16.10.8 Disable Local O&M Ports

It is possible to disable local access to an NE by disabling the local ports in terms of the USB port and the LAN port.

If both the USB and LAN port are disabled, it is no longer possible to access the DCN and the NE on-site. It is possible to disable the LAN port and the USB port independently. The only way to access the node is to access the DCN remotely, either from an OMC or from a remote node.

16.10.9 Notification on Logon

Through Notification on Logon, the OMC can react on an unexpected logon. Notification on Logon is available for local (USB) connections.

An event is sent to the management systems each time a user logs on to an NE. The notification contains a timestamp, the user name, the IP address, and information on whether a logon is through CLI or SNMP. Upon successful logon, the information associated with the last successful logon is displayed up to 30 days. The Security Issue on Logon alarm, which can be enabled using MINI-LINK Node GUI, is raised in case of three consecutively failed logon attempts. This alerts the management system of a potential security threat.

16.10.10 Password Cracking Defense: Setting Brute Force Threshold

Delayed logon is activated to protect against password cracking. When a suspected attack is performed against CLI, MINI-LINK Node GUI, NETCONF or SNMP interface, a delay is inserted in case of an authentication error. The delay is inserted when a certain number of authentication errors is reached during a time window (activation threshold). The normal behavior is restored only when no error is detected during the defense mode for a certain period of time (defense period). The functionality can be managed through the activation threshold setting.

- **Note:** When brute force defense mode is activated, the following system response is applicable:
 - Users with blacklisted user names or blacklisted IP addresses cannot access the NE.
 - Non-blacklisted users can access the NE but can experience system slowdown.

It is possible to set a time limit for how long the brute force protection lockout is to be active after a failed login attempt. It is also possible to monitor and reset the locked sessions. These settings can be configured using either CLI or MINI-LINK Node GUI.

16.10.11 Firewall

A firewall is in place for packet filtering on the IP address and the range of the IP address. The packet filter option is protection from external traffic connections through each possible port or service by closing or opening commands. These options are only available to the admin_user.

Both IPv4 and IPv6 are supported.

16.10.12 Legal Notice

It is possible to add a legal notice shown before logging on, to highlight the legal consequences of unauthorized access. The legal notice can be customized according to local law.

16.10.13 Access Control Lists

IPv4 Access Control Lists (ACLs) can filter incoming or outgoing IPv4 packets on IP interfaces, based on the contents of the IPv4 header. Packets can be permitted or denied at line speed using hardware capabilities.

The supported filtering parameters are the following:

- Source IP address
- Destination IP address
- IP protocol type
- TCP/UDP source port
- TCP/UDP destination port

ACLs can be configured independently on each VLAN interface and VLAN subinterface. The total number of configurable ACLs is limited per node based on the type of the NPU.

Transit and originating or terminating IP packets are filtered by the access lists.

Note: ACLs are not supported for PPP and loopback interfaces.

17 Accessories

The MINI-LINK 6600 product program contains a comprehensive set of accessories for installation and operation. This section gives additional technical information for Small Form Factor Pluggable (SFP) and Small Form Factor Pluggable Plus (SFP+).

For a full list of approved SFP and SFP+, see MINI-LINK 6600 R1 Compatibility, Reference [1].

17.1 SFP

The SFP exists as electrical or optical transmitter/receiver for PDH/Ethernet, see Figure 158.



Figure 158 Electrical/Optical SFP

17.2 SFP+

The SFP+ exists as optical transmitter/receiver for Ethernet, see Figure 159.





17.3 Tunable DWDM SFP+

The tunable SFP+ is a Dense Wavelength Division Multiplexing (DWDM) transceiver for Ethernet. Tunable DWDM SFP+ allows adjusting the transmission wavelength within the C-band DWDM International Telecommunication Union (ITU) Grid. The use of tunable DWDM SFP+ is supported in the 10G port of the ETU 1002 plug-in unit.

17.4 Direct-Attach Copper Cable

The Direct-Attach Copper Cable (DAC) is an electrical traffic cable that can be used to directly connect MINI-LINK 6600 with another product placed physically close. It is equipped with SFP+ modules at both ends, and it provides a 10 Gbps rate.
1

Reference List

- [1] MINI-LINK 6600 R1 Compatibility, 1/221 02-HRA 901 20/11
- [2] Physical/Electrical Characteristics of Hierarchy Digital Interface, ITU-T G.703 (11/2001)





MINI-LINK 6363

MINI-LINK the Network Node

The increasing complexity of today's and future networks require flexible and well-integrated microwave nodes. Building an efficient microwave backhaul network with end-to-end performance in mind, requires high node capacity, compact and modular building practice, advanced packet functionality and features that are aligned and backward-compatible across different network nodes. The microwave nodes also need to be capable of handling single hops as well as advanced hub sites for larger networks. The MINI-LINK network node has shown to reduce product investment by 40 percent compared with a 'hop-by-hop' approach. By combining MINI-LINK outdoor units and indoor units, all network scenarios are supported with superior performance and lowest possible cost of ownership.

Ericsson is the market leader in microwave transmission and has over 40 years of microwave experience with more than 3.5 million radio units delivered to over 175 countries.

High capacities in a compact format

The MINI LINK 6363 is to be used in split systems together with indoor units MINI-LINK 6600, MIN-LINK TN and MINI-LINK CN. It connects to the indoor unit via a coaxial cable. In a very compact format, it supports gigabit capacities in traditional frequency bands as well as in E-band. The high output power caters for the need of high capacities and high availability, providing superior network performance.

World's smallest high power radio

MINI-LINK 6363 builds on the strengths from the world's most widely deployed microwave radio MINI-LINK RAU2 X. The footprint has been reduced by 65% and weight by 35% compared to RAU2 X. This enables easier and faster installations as well as less wind load on towers.



Superior output power

MINI-LINK 6363 has the highest output power in the split mount radio market, that is maintained also for higher modulations. For 24-42 GHz the output power has been increased with up to 5 dB compared to RAU2 X's already high specification. This means more capacity, higher availability and smaller antennas. In combination with functionality in the indoor units, a superior system gain is obtained. High output power is available as a SW license, which makes it possible to step up in modulation and capacity when needed, following a pay as you grow approach.

Best in class dynamic range

MINI-LINK 6363 has best in class dynamic range, which is crucial to reduce interference (using ATPC), reduce power consumption and also to be able to install short hops.

Reduced power consumption

The power consumption in MINI-LINK 6363 has been reduced compared to RAU2 X. For 6-11 GHz the power consumption has been reduced by up to 10 W.

High capacities in traditional frequency bands

The radio unit also supports the new 112 MHz wide channels and a modulation of 4096 QAM, which provides capacities over 1 Gbps.

World's first split mount E-band radio

MINI-LINK 6363 offers a cost efficient access to the Eband spectrum, with reuse of indoor units and coaxial cabling. It supports both TDM and packet transport. Gigabit capacity is provided through 1024 QAM modulation support. It enables Multi-band Booster in split mount configurations, in combination with a traditional frequency band

Increased ingress protection

The radio unit can be installed in very harsh environments as it fulfills IP66 protection against dust and water.

Backward compatibility

MINI-LINK 6363 is hop compatible with the MINI-LINK RAU2 X. If a radio unit needs to be upgraded, the antenna and radio cable can be reused.

Flat panel antennas

Ericsson is introducing high performing flat panel antennas. With high focus on visual appearance and minimized size, Ericsson has created the world's smallest outdoor unit (radio+antenna) in traditional bands. Since antenna performance is key to secure network performance, the antennas are ETSI class 3 compliant.

Technical Specifications MINI-LINK 6363

R

(ŀ

IN

IN

Ν

RADIO LINK	Capacity: 1.4 Gbps
	Channel: 7 – 112 MHz
	Modulation: C-QPSK and 4 – 4096 QAM
	TX power: -10 to +30 dBm
REFLECTOR ANTENNAS	0.2 – 3.7 m / 9 in – 12 ft HP and HPX SHP and SHPX
FLAT PANEL ANTENNAS	0.1 m HP 28 – 42 GHz, 30 – 33 dBi
FREQUENCIES	6 – 42 and 70/80 GHz
WEIGHT	2.8 kg / 6.2 lbs
DIMENSIONS (H × W × D)	179 × 197 × 79 mm (2.8 l) 7.0 × 7.8 × 3.1 in (170 in3)
POWER SUPPLY	+57 VDC
POWER CONSUMPTION	20 W
CONFIGURATIONS	1+0, 1+1, N+0
INTERFACES	Coaxial (modem) Waveguide (antenna) Alignment port
STANDARDS AND RECOMMENDATIONS	ETSI, ECC, FCC, IC, IEC, ITU
ENVIRONMENTAL SPECIFICATIONS	-45 to +60 °C / -49 to +140 °F IP 66
INDOOR UNITS	MINI-LINK 6600 MINI-LINK TN MINI-LINK CN
NETWORK MANAGEMENT	ServiceOn Element Manager



Philips Monitor LCD monitor

V Line 24 (23.8" / 60.5 cm diag.) 1920 × 1080 (Full HD)

241V8L

Vivid, crisp images beyond boundaries

Philips V line wide-view monitor gives viewing beyond boundaries, great value with essential features. Adaptive-Sync delivers smooth video display. Features like anti-glare, LowBlue mode, and flicker-free for easy-on-the eyes.

Vivid pictures every time

- VA display delivers awesome images with wide viewing angles
- SmartContrast for rich black details
- · SmartImage presets for easy optimized image settings
- Effortlessly smooth action with Adaptive-Sync technology

Easy on the eyes

- · Less eye fatigue with Flicker-free technology
- · LowBlue Mode for easy on-the-eyes productivity
- · EasyRead mode for a paper-like reading experience

Experience true multimedia

· HDMI ensures universal digital connectivity

Perfectly designed for your space

• VESA mount allows for flexibility



Highlights

VA display



Philips VA LED display uses an advanced multidomain vertical alignment technology which gives you super-high static contrast ratios for extra vivid and bright images. While standard office applications are handled with ease, it is especially suitable for photos, web-browsing, movies, gaming, and demanding graphical applications. It's optimized pixel management technology gives you 178/178 degree extra wide viewing angle, resulting in crisp images.

SmartContrast



SmartContrast is a Philips technology that analyzes the contents you are displaying, automatically adjusting colors and controlling backlight intensity to dynamically enhance contrast for the best digital images and videos or when playing games where dark hues are displayed. When Economy mode is selected, contrast is adjusted and backlighting fine-tuned for just-right display of everyday office applications and lower power consumption.

SmartImage



SmartImage is an exclusive leading edge Philips technology that analyzes the content displayed on your screen and gives you optimized display performance. This user friendly interface allows you to select various modes like Office, Photo, Movie, Game, Economy etc., to fit the application in use. Based on the selection, SmartImage dynamically optimizes the contrast, color saturation and sharpness of images and videos for ultimate display performance. The Economy mode option offers you major power savings. All in real time with the press of a single button!

Adaptive-Sync technology



Gaming shouldn't be a choice between choppy gameplay or broken frames. Get fluid, artifactfree performance at virtually any framerate with Adaptive-Sync technology, smooth quick refresh and ultra-fast response time.

Flicker-free technology



Due to the way brightness is controlled on LED-backlit screens, some users experience flicker on their screen which causes eye fatigue. Philips Flicker-free Technology applies a new solution to regulate brightness and reduce flicker for more comfortable viewing.

LowBlue Mode



Studies have shown that just as ultra-violet rays can cause eye damage, shortwave length blue light rays from LED displays can cause eye damage and affect vision over time. Developed for wellbeing, Philips LowBlue Mode setting uses a smart software technology to reduce harmful shortwave blue light.

EasyRead mode



EasyRead mode for a paper-like reading experience

















Specifications

Picture/Display

- LCD panel type: VA LCD
- Backlight type: W-LED system
- Panel Size: 23.8 inch / 60.5 cm
- Display Screen Coating: Anti-Glare, 3H, Haze 25%
 Effective viewing area: 527 (H) x 296.5 (V)
- Aspect ratio: 16:9
- Maximum resolution: 1920 x 1080 @ 75 Hz*
- Pixel Density: 93 PPI
- Response time (typical): 4 ms (GtG)*
- Brightness: 250 cd/m²
- Contrast ratio (typical): 3000:1
- SmartContrast: Mega Infinity DCR
- Pixel pitch: 0.2745 x 0.2745 mm
- Viewing angle: 178° (H) / 178° (V), @ C/R > 10
- Flicker-free
- Display colors: 16.7 M
- Scanning Frequency: 30 85 kHz (H) / 48 75 Hz (V)
- LowBlue Mode
- EasyRead
- sRGB
- Adaptive sync

Connectivity

- Signal Input: VGA (Analog), HDMI (digital, HDCP)
- Sync Input: Separate Sync, Sync on Green
- Audio (In/Out): Audio out

Convenience

- User convenience: Power On/Off, Menu/OK, Brightness/UP, Input/Down, SmartImage/Back
- OSD Languages: Brazil Portuguese, Czech, Dutch, English, Finnish, French, German, Greek, Hungarian, Italian, Japanese, Korean, Polish, Portuguese, Russian, Spanish, Simplified Chinese, Swedish, Turkish, Traditional Chinese, Ukrainian
- Other convenience: Kensington lock, VESA mount (100x100mm)
- Plug & Play Compatibility: DDC/CI, Mac OS X, sRGB, Windows 10 / 8.1 / 8 / 7

Stand

• Tilt: -5/20 degree

Power

- On mode: 20.4 W (typ.)
- Standby mode: 0.5 W (typ.)
- Off mode: 0.3 W (typ.)
- Energy Label Class: E
- Power LED indicator: Operation White, Standby mode- White (blinking)
- Power supply: Built-in, 100-240VAC, 50-60Hz

Dimensions

- Product with stand(max height): 540 x 416 x 220 mm
- Product without stand (mm): 540 x 322 x 51 mm
- Packaging in mm (WxHxD): 608 x 415 x 121 mm

Weight

- Product with stand (kg): 3.03 kg
- Product without stand (kg): 2.64 kg
- Product with packaging (kg): 4.25 kg

Operating conditions

- Temperature range (operation): 0°C to 40°C °C
- Temperature range (storage): -20°C to 60°C °C
- Relative humidity: 20%-80 %
- Altitude: Operation: +12,000ft (3,658m), Non-
- operation: +40,000ft (12,192m) • MTBF: 50,000 (Excluded backlight) hour(s)

Sustainability

- Environmental and energy: RoHS, Mercury Free
- Recyclable packaging material: 100 %

Compliance and standards

- Regulatory Approvals: UKRAINIAN, CB, TUV Ergo, TUV/GS, CU-EAC, EAEU RoHS, CE Mark



Issue date 2022-06-22 © 2022 Koninklijke Philips N.V. All Rights reserved. Version: 3.0.1 Specifications are subject to change without notice. Trademarks are the property of Koninklijke Philips N.V. or their respective owners. * The maximum resolution works for HDMI input only. 12 NC: 8670 001 70948 * Response time value equal to SmartResponse * The monitor may look different from feature images. EAN: 87 12581 77164 5 www.philips.com

Cabinet

- Color: Black
- Finish: Textured

What's in the box?

- · Monitor with stand
- Cables: HDMI cable, Power cable

User Documentation